# **Progress in Mathematical Physics**

Volume 42

## Editors-in-Chief

Anne Boutet de Monvel, Université Paris VII Denis Diderot Gerald Kaiser, The Virginia Center for Signals and Waves, Austin Texas, USA

Editorial Board Sir M. Berry, University of Bristol, UK C. Berenstein, University of Maryland, College Park, USA P. Blanchard, University of Bielefeld, Germany A.S. Fokas, University of Cambridge, UK D. Sternheimer, Université de Bourgogne, Dijon, France C. Tracy, University of California, Davis, USA Bozhidar Z. Iliev

# Handbook of Normal Frames and Coordinates

Birkhäuser Verlag Basel Boston Berlin Author:

Bozhidar Z. Iliev Institute for Nuclear Research and Nuclear Energy Blvd. Tzarigradsko Chaussée 72 1784 Sofia Bulgaria e-mail : bozho@inrne.bas.bg

2000 Mathematics Subject Classification 53B05, 53B99, 53C05, 53C80, 53C99, 53Z05, 57R25, 57R55, 81Q99, 83C99

A CIP catalogue record for this book is available from the Library of Congress, Washington D.C., USA

Bibliographic information published by Die Deutsche Bibliothek Die Deutsche Bibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data is available in the Internet at <a href="http://dnb.dbb.de">http://dnb.dbb.de</a>>.

## ISBN 3-7643-7618-X Birkhäuser Verlag, Basel – Boston – Berlin

This work is subject to copyright. All rights are reserved, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, re-use of illustrations, broadcasting, reproduction on microfilms or in other ways, and storage in data banks. For any kind of use whatsoever, permission from the copyright owner must be obtained.

© 2006 Birkhäuser Verlag, P.O. Box 133, CH-4010 Basel, Switzerland Part of Springer Science+Business Media Printed on acid-free paper produced of chlorine-free pulp. TCF ∞ Printed in Germany ISBN-10: 3-7643-7618-X ISBN-13: 987-3-7643-7618-5 e-ISBN: 3-7643-7619-X

987654321

www.birkhauser.ch

# **Table of Contents**

Li	st of (	Convent	ions	xi	
Pr	eface			xiii	
A	cknow	ledgme	nts	xvi	
I	Man	ifolds, I	Normal Frames and Riemannian Coordinates		
	1	Introd	uction	2	
	2	Differe	ntiable manifolds	4	
		2.1	Basic definitions	4	
		2.2	Differentiable mappings	7	
		2.3	Tangent vectors and vector fields	9	
		2.4	Covectors and covector fields	15	
		2.5	Tensors and tensor fields. Tensor algebras	17	
	3	Linear	connections on a differentiable manifold	19	
		3.1	Motivation	20	
		3.2	Basic definitions	21	
		3.3	Parallel transport	26	
		3.4	Geodesics and exponential mapping	30	
	4	Riema	nnian manifolds	33	
	5	Norma	l frames: general ideas, uniqueness and holonomicity	37	
	6	Norma	l coordinates on Riemannian manifolds	44	
	7	Examp	bles of normal coordinates for Riemannian connections	49	
	8	Termin	nology 1: Bases and frames. Holonomicity	68	
	9	Conclu	usion	71	
п	Exis	tence, I	Uniqueness and Construction of		
	Nori	Normal Frames and Coordinates for Linear Connections			
	1	Introd	uction	74	
	2	The ca	se at a single point	75	
		2.1	Old classical method $\ldots$	75	
		2.2	Complete description	79	

	2.3	Modern classical method	83
3	The c	ase along paths without self-intersections	84
	3.1	Fermi coordinates	85
	3.2	Complete description	93
4	The c	ase in a neighborhood	104
5	The c	ase on arbitrary submanifolds	111
	5.1	Conventional method $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	112
	5.2	Complete description	120
6	Exam	ples of normal frames and coordinates $\ldots \ldots \ldots \ldots$	128
7	Concl	usion	137
III No	ormal Fra	ames and Coordinates	
for	Derivat	ions on Differentiable Manifolds	
1	Introc	luction	142
2	Deriva	ations of the tensor algebra over a manifold	143
3	Gener	al overview	148
4	Frame	es and coordinates normal at a point	153
5	Frame	es and coordinates normal along paths	154
6	Frame	es and coordinates normal in a neighborhood	157
7	Frame	es and coordinates normal on submanifolds	158
8	Frame	es and coordinates normal along mappings	161
	8.1	Injective mappings	162
	8.2	Locally injective mappings	168
	8.3	Mappings between manifolds	169
9	Norm	al frames and coordinates for derivations along	1 = 0
	a fixe		172
	9.1	The case at a single point	170
	9.2	The case along paths	1/8
	9.3	I he case on the whole manifold	184
10	9.4 N		180
10	Norm	al frames and coordinates for derivations along paths	189
11	On fra	ames simultaneously normal for two derivations	197
12	Norm	al frames for linear connections (review)	202
13	Exam	pies	204
14	Termi	nology 2: Normal and geodesic frames	209
15	Concl		210

### **IV Normal Frames in Vector Bundles**

1	Introd	luction	214
2	Vector	r bundles	216
	2.1	Basic definitions	217
	2.2	Liftings of paths	218
	2.3	Derivations along paths	219
	2.4	Tensor bundles	222
3	Linear	r transports along paths in vector bundles	225
	3.1	Definition and general form	225
	3.2	Representations in frames along paths	228
	3.3	Linear transports and derivations along paths	230
4	Norma	al frames for linear transports	234
5	On the	e existence of normal frames	239
6	The ca	ase of a manifold as a base	247
7	Linear	r transports and normal frames in line bundles	254
8	Norma	al frames for derivations in vector bundles with	
	a man	ifold as a base	257
9	Curva	ture and normal frames $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	261
	9.1	Curvature of linear transport or derivation along paths	261
	9.2	On the curvature of Euclidean transports along paths	268
10	Torsio	on and normal coordinates	271
	10.1	Torsion of linear transport or derivation along paths	
		in the tangent bundle over a manifold	271
	10.2	Holonomic normal frames in the tangent bundle	276
11	Parall	el transports in tangent bundles	280
	11.1	The parallel transport as a transport along paths	281
	11.2	Normal frames for parallel transports along paths	285
12	Strong	g normal frames	289
13	Linear	${\bf r}$ transports assigned to derivations in tangent bundles	292
	13.1	Derivations along paths	293
	13.2	Derivations along vector fields	295
	13.3	Derivations along fixed vector field	295
	13.4	Normal frames	298
14	Links	with the theory of connections and parallel transports	301
	14.1	Parallelism structures, connections and	
		covariant derivatives	301
	14.2	Parallel transports in vector bundles	311
	14.3	Parallel transports and linear transports along paths	315
	14.4	Normal frames for parallel transports, connections and	_
		covariant derivatives	320
	14.5	On the role of the curvature	323

	15	Autoparallel paths
	16	On a fibre bundle view at quantum mechanics
	17	Conclusion
v	Nori	nal Frames for Connections on Differentiable Fibre Bundles
•	1	Introduction 334
	2	Preliminaries 336
	3	Connections on hundles 330
	0	3.1 Frames and coframes on the hundle space 330
		3.2 Connection theory 345
	4	Connections on vector bundles 350
	1	4.1 Vertical lifts 350
		4.2 Linear connections on vector bundles
		4.3 Covariant derivatives in vector bundles
		4.4 Affine connections
	5	General (co)frames
	6	Normal frames
		6.1 The general case
		6.2 Normal frames adapted to holonomic frames
		6.3 Normal frames on vector bundles
	7	Coordinates normal along injective mappings with
		non-vanishing horizontal component
	8	Links between connections and transports along paths in
		fibre bundles
	9	Conclusion
Bi	bliogr	aphy
ç.,	hight	List of Symbols (11)
Su	Sot t	hoory /11
	Line	ar/vactor spaces
	Mati	$\frac{11}{12}$
	Man	ifolds /1/
	Line	ar connections on manifolds
	Deri	vations on manifolds
	Fibr	e bundles 417
	Line	ar transports and derivations along paths
	Para	llel transports, connections and covariant derivatives 419
Aι	ıhtor	Index
No	otatio	n Index 425
Su	biect	Index

# List of Tables

## Chapter II

-		
7.1	Main contributions in the theory of normal coordinates for	
	torsionless linear connections $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	138

# Chapter V

9.1	Main contributions in axiomatizing the concept of "parallel	
	transport"	399

# List of Figures

# Chapter V

8.1	Mappings between the sets of parallel transports, connections and	
	parallel transports along paths	397

# List of Conventions

**References.** The book is divided into chapters which have a sequential Roman enumeration. The chapters are divided into sections with a sequential Arabic enumeration, which is independent in each chapter. Some sections are divided into subsections.

In each chapter the subsections, equations, propositions, theorems, lemmas, and so on have a double independent enumeration of the form  $\mathbf{m.n}$  or  $(\mathbf{m.n})$  for the equations,  $\mathbf{m,n=1,2,...}$  where  $\mathbf{m}$  is the number of the section in which the designated item appears and  $\mathbf{n}$  is its sequential number in it. So, Proposition 4.7 and (3.12) (or equation (3.12)) mean respectively Proposition 7 in Section 4 and equation 12 in Section 3 of the current chapter. A suitable item from a chapter different from the current one is referred as  $\mathbf{R.m.n}$  or  $(\mathbf{R.m.n})$  for equations, where  $\mathbf{R=I,II,...}$  is the Roman number of the chapter in which the item appears; e.g., Remark II.5.3 and IV.4 (or Section IV.4) mean respectively Remark 3 in Section 5 of Chapter II and Section 4 in Chapter IV.

The footnotes are indicated as superscripts in the main text and have independent Arabic enumeration in each section. When we refer to a footnote, it is on the current page if the page on which it appears is not explicitly indicated.

**Citations.** An Arabic number in square brackets, e.g., [27], directs the reader to the list of references, i.e., in this example [27] means the 27th item from the Bibliography list beginning on page 401.

The ends of the proofs are marked by empty square sign, viz. with  $\Box$ .

**Indices.** The Latin indices refer to an arbitrary linear (vector) space, in particular to the tangent and cotangent spaces. If in a given problem are presented the tangent and cotangent spaces to a manifold and other vector space(s), then the indices referring to the first two spaces are denoted with small Greek letters; for the rest one(s) the Latin letters will be used.

**Einstein's summation convention:** in a product of quantities or in a single expression, a summation over indices repeated on different levels is assumed over the whole range in which they change. Any exception of this rule is explicitly stated.

**Symmetrization and antisymmetrization.** On indices included in (or surrounded by) round (resp. square) brackets a symmetrization (resp. antisymmetrization) with coefficient one over the factorial of their number is assumed. If some indices in such a group have to be excluded from this operation, they are included in (surrounded by) vertical bars.

Matrix of linear mapping with respect to a given basis, or bases, or field of, possibly local, bases: the same symbol but the kernel letter is in **boldface**. Exception: the matrix of a derivation (derivative operator) is denoted by boldface capital Greek letter gamma, i.e., by  $\Gamma$ , possibly with some indices.

**Matrix elements.** When the elements of a (two-dimensional) matrix are labeled by superscript and subscript, the superscript is considered as a first index, numbering the matrix's rows, and the subscript as a second one, numbering the matrix's columns. In this way the matrix of composition of linear mappings is equal to the product of the matrices of the mappings in the same order in which they appear in the composition and this does not depend on the way the matrix's indices are situated.

**Free arguments.** If we want to show explicitly the argument(s) of some mapping or to single out it (them) as arbitrary while the other arguments, if any, are considered as fixed ones, we denote it (them) by (centered) dot, i.e., by  $\cdot$ . E.g., if  $f: A \to C$  and  $g: A \times B \to C$ , then  $f(\cdot) \equiv f$ ,  $g(\cdot, \cdot) \equiv g$ , and  $g(\cdot, b)$ ,  $b \in B$ , means  $g(\cdot, b): A \to C$  with  $g(\cdot, b): a \mapsto g(a, b)$  for all  $a \in A$ .

# Preface

The main subject of this book is an up-to-date and in-depth survey of the theory of normal frames and coordinates in differential geometry. The existing results, as well as new ones obtained lately by the author, on the theme are presented.

The text is so organized that it can serve equally well as a reference manual, introduction to and review of the current research on the topic. Correspondingly, the possible audience ranges from graduate and post-graduate students to scientists working in differential geometry and theoretical/mathematical physics. This is reflected in the bibliography which consists mainly of standard (text)books and journal articles.

The present monograph is the first attempt for collecting the known facts concerting normal frames and coordinates into a single publication. For that reason, the considerations and most of the proofs are given in details.

Conventionally local coordinates or frames, which can be holonomic or not, are called *normal* if in them the coefficients of a linear connection vanish on some subset, usually a submanifold, of a differentiable manifold. Until recently the existence of normal frames was known (proved) only for symmetric linear connections on submanifolds of a manifold. Now the problems concerning normal frames for derivations of the tensor algebra over a differentiable manifold are well investigate; in particular they completely cover the exploration of normal frames for arbitrary linear connections on a manifold. These rigorous results are important in connection with some physical applications. They may be applied for rigorous analysis of the equivalence principle. This results in two general conclusions: the (strong) equivalence principle (in its 'conventional' formulations) is a provable theorem and the normal frames are the mathematical realization of the physical concept of 'inertial' frames. The normal frames find other important physical application in the bundle formulation of quantum mechanics. It turns out that in a normal frame the bundle Heisenberg and Schrödinger pictures of motion coincide.

Applying some freedom of language, we can state the general physical idea: the normal frames are the most suitable ones for describing free objects and events, i.e., such that on them do not act any forces. Regardless of the different realizations of that idea in general relativity and its generalizations, quantum mechanics, gauge theories etc., there is an underlying mathematical background for the general description of such situations: the existence (or non-existence) of normal frames in vector bundles. This observation fixes to a great extend the mathematical tools required for the description of some fundamental physical theories.

In the book, formally, may be distinguished three parts: The first one includes Chapters I–III and deals with a variety of mathematical problems concerning normal frames and coordinates on differentiable manifolds. The second part consists of Chapters IV and V and investigates normal frames (and possibly coordinates) in vector bundles and differentiable bundles, respectively. The last part, involving the text after Chapter V, contains inquiry material.

The requisite mathematical language required for the description of normal frames is spread over the initial sections of the chapters. In particular, Sections I.2–I.4, III.2, IV.9, IV.2, IV.14.1 and V.2–V.5 can be collected into an introductory chapter under the title "Mathematical preliminaries"<sup>1</sup> but this is not done by pedagogical reasons.<sup>2</sup> The normal coordinates and frames, in the case of linear connections on a manifold, are initially introduced in Chapter I. It contains our basic preliminary material and a review of the Riemannian coordinates. Chapter II is devoted to the existence, uniqueness, construction and other related problems concerning normal frames and coordinates in manifolds endowed with linear connection. It presents, in historical order, a detailed review of the existing literature as well as generalization of a number of results, e.g., for connections with torsion. Further, in Chapter III, problems connected with the existence, uniqueness, holonomicity etc. of normal frames for arbitrary derivations of the tensor algebra over a manifold are investigated. Next (Chapter IV), the same range of problems is explored for normal frames for linear transports in vector bundles. This material covers completely the special case of normal frames for linear connections in vector bundles or on a differentiable manifold. The main aim of Chapter V is the exploration of normal frames (and coordinates, if any) for general connections on differentiable fibre bundles which, in particular, can be vector ones.

The general approach of the book is essentially coordinate-dependent or basis-dependent. This is due to its basic subject: frames, bases or coordinates with some special properties. However, if possible and suitable, the coordinatefree notation and methods are not neglected.

The basic mathematical prerequisites vary from chapter to chapter but generically they include the grounds of vector (linear) spaces, differentiable manifolds, vector bundles, connection theory, and a firm belief in the existence and uniqueness theorems of ordinary differential equations. Some of the corresponding concepts and results are reproduced in our text but the acquaintance with adequate literature is required. Appropriate references are given in the Introductions to the chapters and directly in the main text.

<sup>&</sup>lt;sup>1</sup>As (practically) any 'preliminary' knowledge requires for its understanding some other 'preliminary' to it knowledge, in the corresponding sections are cited a number of works containing this second kind of mathematical 'luggage'.

 $<sup>^{2}</sup>$ The material is so organized, that the required concepts and results appear in the logical order in which they are necessary for some particular purpose(s).

Preface

The material is so organized that a successive chapter generalizes the preceding one(s) and refers to it (them).

Any suggestions and comments are welcome. The author's postal address is

Bozhidar Zakhariev Iliev, Laboratory of Mathematical Modeling in Physics, Institute for Nuclear Research and Nuclear Energy, Bulgarian Academy of Sciences, Boul. Tzarigradsko chaussée 72, 1784 Sofia, Bulgaria,

his e-mail address is

bozho@inrne.bas.bg

and

http://theo.inrne.bas.bg/~bozho/

is the URL address of his personal World Wide Web site.

Sofia, Bulgaria 7 July, 2006

# Acknowledgments

The idea for writing a book on normal frames and coordinates in differential geometry was suggested by professor Dr. Michiel Hazewinkel (Centre for Mathematics and computer science, Amsterdam, The Netherlands) to the author in October 1998. I express to him my sincere gratitude for this.

I would like to thank Prof. Dr. Stancho Dimiev and mathematician Vladimir Aleksandrov (Mathematical Institute of Bulgarian Academy of Sciences) for the encouragement and fruitful conversations at the early stages of my investigations on normal frames. Special thanks to professor Dr. Sava Manov (1943–2005) (Institute for Nuclear Research and Nuclear Energy of Bulgarian Academy of Sciences). During 1991–1996 he read the manuscripts of some my papers on the subject. With him I have had many useful discussions concerning, first of all, the applications of normal frames to gravity physics. I am indebted to prof. N.A. Chernikov for his hospitality during my numerous visits in the Joint Institute for Nuclear Research, Dubna, Russia.

The (anonymous for me) referees from Journal of Physics A: Mathematical and General helped me with many suggestions and improvements. I would like to thank them also.

The entire manuscript was typeset by the author by means of the main  $\text{IAT}_{\text{EX}} 2_{\varepsilon}$  document preparation computer system together with a number of additional to it packages of programs, including first of all AMS-IATEX, AMS-fonts, BIBTEX, MakeIndex, Index, etc. My gratitude to all of the numerous persons who created and developed, maintain and distribute (free!) this valuable and high-quality typesetting system.

I would like to express my deep gratitude to my mother Dr. Lilyana Stefanova Shtereva for her understanding, support, love and faith in me.

The work on this monograph was partially supported by the National Science Fund of Bulgaria under Grant No. F 1515/2005.

Bozhidar Z. Iliev

11 July, 2006 Sofia, Bulgaria

# Chapter I

# Manifolds, Normal Frames and Riemannian Coordinates

The basic differentialgeometric concepts, such as differentiable manifolds and mappings, tensors and tensor fields. connections, on and linear which the book rests, are introduced. Partially the notation and terminology employed are fixed. The normal frames and coordinates defined are as ones in which the coefficients of a linear connection in them vanish on some set. Certain their general properties are mentioned. The Riemannian coordinates, which are normal at their origin, are described.

# 1. Introduction

The goal of this chapter is twofold: it introduces most of the basic preliminary definitions and results on which our investigation rests (Sections 2, 3, and 4) and it begins the study of the normal frames and coordinates (Sections 5 and 6).

The main concepts of differential geometry required for the understanding of the book are: differentiable manifolds and mappings, submanifolds, Riemannian manifolds, tangent vectors and vector fields, tensors and tensor fields, linear connections. The readers acquainted with them may only look over the corresponding sections for our notation, omitting the major text to which they may wish to return later, following the references to it.

In more details, the contents of the chapter is as follows.

The purpose of Section 2 is to fix our terminology and notation concerning differentiable manifolds and some typical to them natural structures. This is not a summary of the differential geometry, only certain basic concepts and particular relations between them required for our future aims are presented. At first the concepts of topological and differentiable manifolds are introduced, then tangent vectors, cotangent vectors, and tensors and the corresponding fields of them on a manifold are defined. Also some expressions in local bases (or frames) and coordinates are given. If the reader is acquainted with all this, he/she can simply look over this Section for our notation skipping the main text. A reader interested in deeper understanding of these concepts, as well as in differential geometry as a whole, should consult with the specialized literature. Here is a (random) selection of such titles. An elementary introduction to differential geometry, with 'physical' orientation, can be found in [1-6]. The same purpose can serve the books [7–10] which are more 'mathematically' oriented. Our text follows the excellent (text)books [11, 12]. At last, the advanced works [13–16] can be recommended. A brief synopsis of the mathematics preceding the introduction of manifolds is given it [9, 14, 16] while [12, 17] contain an expanded presentation of the 'preliminary' to manifolds material. Of course, the reading of all of the abovementioned serious books is not necessary for the understanding of what follows. For this end, the reading of Section 2 is sufficient and the references cited may be consulted for more detains and proofs of some assertions. The knowledge of the tensor analysis in coordinate-dependent language is desirable [18, 19]. It is almost sufficient for the most of this and subsequent chapters.

In Section 3, we introduce the concept of linear connection on a manifold. The approach chosen is, in a sense, middle between elementary books on general relativity, such as [20, 21], and pure mathematical ones on differential geometry, like [11, 22]. We have tried to follow closely [9, 11, 12] but the abstracting material is adapted to the goals of the present book. After a motivation for what the connections are needed for, we introduce the linear connections via a system of axioms for the covariant derivative of the algebra of tensor fields over a given manifold. We employ this method since the theory of vector bundles, which is not

required for Chapters I–III, will be involve into action only at the beginning of Chapter IV. In this connection, let us mention that the linear connections can be defined only on the algebra of vector fields on a manifold (i.e., to the tangent to it bundle), and then they admit a unique extension on the whole algebra of tensor fields [11, Chapter 3, Proposition 7.5]. A more advanced and deep treatment of the theory of linear connections on manifolds and vector bundles can be found in [10,13,15,22,23]. We also present the notion of a parallel transport (induced by a linear connection) which will practically step on scene in Chapter IV but here is a natural place for it to appear. It will be used in Chapters I–III for proving and formulating some results. Section 3 ends with a brief consideration of the geodesics and exponential mapping.

The concept of Riemannian metric and Riemannian connection are given in Section 4. If the reader is interested in essence of Riemannian geometry, he/she is referred, for example, to [8–12, 19, 24–27].

In Section 5, we introduce the main objects of our investigation, the *normal* frames and coordinates. We define them as ones in which the coefficients of a linear connection vanish on a given set. Some considerations on the uniqueness and (an)holonomicity of the normal frames are presented too.

Section 6 contains a complete description of normal frames at a given point on  $(C^{\infty})$  Riemannian manifolds. This is done on the base of Riemannian coordinates which turn to be normal at their origin. The geodesic coordinates are pointed as other example of coordinates normal at a point. Some general results, proved further in Chapter II, concerning the existence of normal frames on submanifolds are quoted. An expanded presentation of the problem of existence of normal coordinates at a point of a  $C^{\infty}$  Riemannian manifold is given in [19,24], where also a list of original early works on this topic can be found.

In Section 7 are presented a number of examples and exercises of concrete Riemannian connection and coordinates/frames normal for them on different sets. At first, the (locally) Euclidean and one-dimensional manifolds are considered. The (pseudo)spherical coordinates on (pseudo)spheres are (partially) investigated for sets on which they are normal for the Riemannian connection induced on them by the metric on them generated by the Euclidean one of the Euclidean space in which the (pseudo)spheres are embedded. Similar instance on the twodimensional torus is presented. The cosmological models of Einstein, de Sitter and Schwarzschild are considered (in concrete coordinates) from the view-point of normal frames/coordinates on them. Some peculiarities of the light cone in Minkowski spacetime are pointed too.

Section 8 deals with certain terminological problems concerning bases and frames. Some links between these concepts are explicitly formulated and/or derived.

The chapter ends with some general remarks and conclusions in Section 9.

### 2. Differentiable manifolds

In this section we introduce the basic concepts of differential geometry on which the present and next chapters rest. First of all, attention is paid on the differentiable manifolds and the algebras of tensors and tensor fields on them.

#### 2.1. Basic definitions

As we shall see a little further, the fields of real,  $\mathbb{R}$ , and complex,  $\mathbb{C}$ , numbers are the simplest examples of respectively real and complex (one-dimensional) differentiable manifolds. Correspondingly, their *n*th-Cartesian powers (direct products),

$$\mathbb{R}^n := \underbrace{\mathbb{R} \times \cdots \times \mathbb{R}}_{n-\text{times}}, \quad \mathbb{C}^n := \underbrace{\mathbb{C} \times \cdots \times \mathbb{C}}_{n-\text{times}}$$

where  $n \in \mathbb{N}$ ,  $\mathbb{N} := \{1, 2, ...\}$  being the set of integers,<sup>1</sup> and  $\times$  is the Cartesian product sign (of sets),<sup>2</sup> are the simplest examples of respectively real and complex *n*-dimensional manifolds.

The silent and, perhaps the most important idea in the concept of an *n*dimensional manifold M over a field  $\mathbb{K}$  is that it is a set which is 'locally Euclidean' or which 'locally looks like  $\mathbb{K}^n$ ' where, to save writing,  $\mathbb{K}$  denotes some of the fields  $\mathbb{R}$  or  $\mathbb{C}$ . The intuitive meaning of the last phrase is that M can be divided into (or, rewording, can be presented as a union of) sets each of which is into a bijective (one-to-one onto) correspondence with some subset of  $\mathbb{K}^n$ , in particular, possibly, with the whole  $\mathbb{K}^n$ .

Now, following the books [11, 12], the precise definitions are in order.

**Definition 2.1.** An *n*-dimensional topological  $\mathbb{K}$ -manifold is a Hausdorff topological space such that every its point has a neighborhood homeomorphic to an open subset of  $\mathbb{K}^{n,3}$ 

Notice, if we say that U is a neighborhood of a set  $V \subseteq M$ , M being a topological space, we mean that U is an open set in M containing V. Otherwise by a neighborhood (in M) we understand any open set in M (which set is a neighborhood of any its point in the just pointed sense).

For  $\mathbb{K} = \mathbb{R}$  (resp.  $\mathbb{K} = \mathbb{C}$ ) the manifold is called real (resp. complex). If  $\mathbb{K}$  is clear from the context, the *n*-dimensional  $\mathbb{K}$ -manifold is referred as an *n*-manifold. In this work only the finite-dimensional case will be considered, i.e.,  $n \in \mathbb{N}$ , with

<sup>&</sup>lt;sup>1</sup>When writing  $x \in X$ , X being a set, we mean "for all x in X" if the point x is not specified (fixed, given) and is considered as an argument or a variable.

<sup>&</sup>lt;sup>2</sup>For some purposes the sign  $\oplus$  is used instead of  $\times$ ; the result is called a direct sum or product in this case.

<sup>&</sup>lt;sup>3</sup>For more general purposes, the Hausdorffness can be dropped and/or  $\mathbb{K}$  can be (locally) replaced with an arbitrary dimensional vector space(s) [14, pp. 21–22], [10, pp. 2–4]. Some authors [13, p. 32] add the additional requirement for existence of a countable base. These cases will not be considered in the present book.

#### 2. Differentiable manifolds

 $n < \infty$ . If both n and K are evident from the context, we speak simply of a topological manifold.

The number  $n \in \mathbb{N}$  of an *n*-dimensional K-manifold M is called its *dimension* and is denoted by dim M. If  $\mathbb{K} = \mathbb{R}$  (resp.  $\mathbb{K} = \mathbb{C}$ ), the dimension of M is also denoted by dim<sub>R</sub> (resp. dim<sub>C</sub>). (See p. 7 for the meaning of the symbol dim<sub>R</sub> Mfor a complex manifold M.)

A chart of (for) a topological K-manifold M is an ordered pair  $(U, \varphi)$  of an open subset  $U \subseteq M$ , where  $\subseteq$  is the contained in or equal sign, domain of the chart, and homeomorphism  $\varphi \colon U \to V$  onto an open subset  $V \subseteq \mathbb{K}^n$ . Since any open set  $V \subseteq \mathbb{K}^n$  is homeomorphic to the whole  $\mathbb{K}^n$ , we can always put  $V = \mathbb{K}^n$ . By Definition 2.1, for every point  $p \in M$  there exists a chart in whose domain it is contained. This is the rigorous expression of the phrases that 'M locally looks like (a neighborhood of)  $\mathbb{K}^n$ ' or 'M is locally Euclidean'.

**Example 2.1.** The set  $\mathbb{K}^n$  is an *n*-manifold and  $(\mathbb{K}^n, \mathsf{id}_{\mathbb{K}^n})$  is a (global) chart for it. Here and henceforth  $\mathsf{id}_A$  denotes the identity mapping of a set A.

Let  $\{r^i : i = 1, \ldots, n\}$  be the standard Cartesian coordinate functions on  $\mathbb{K}^n$ , viz. if  $c = (c^1, \ldots, c^n) \in \mathbb{K}^n = \mathbb{K} \times \cdots \times \mathbb{K}$  (*n*-times), then  $r^i(c) := c^i$  is the *i*th Cartesian coordinate of *c*. If  $(U, \varphi)$  is a chart of *M*, then  $\{\varphi^1, \ldots, \varphi^n\}$ ,  $\varphi^i := r^i \circ \varphi : U \to \mathbb{K}$ ,  $\circ$  being the sign of mapping's composition, is called *local coordinate system* on *U* (with respect to the chart  $(U, \varphi)$ ) and we say that *U* is (local) coordinate neighborhood; the functions  $\varphi^i$  are called *coordinate functions*. As  $\{r^i\}$  are fixed, sometimes  $\varphi$  is also called (local) coordinate system [10, p. 2]. The reason is that if we define the *n*-tuple  $(\varphi^1, \ldots, \varphi^n)$  as a mapping  $U \to \mathbb{K}^n$  by  $(\varphi^1, \ldots, \varphi^n)(p) := (\varphi^1(p), \ldots, \varphi^n(p)), p \in U$  then  $\varphi \equiv (\varphi^1, \ldots, \varphi^n)$ .

The numbers  $\varphi^1(p) = r^1(\varphi(p)), \ldots, \varphi^n(p) = r^n(\varphi(p)) \in \mathbb{K}$  (or the ordered *n*-tuple  $(\varphi^1(p), \ldots, \varphi^n(p))$ ) are (is) called *local coordinates* of  $p \in U$  with respect to  $(U, \varphi)$ . Often  $\varphi^i(p)$  is abbreviated to  $p^i$ .

If for two charts  $(U, \varphi)$  and  $(U', \varphi')$  is fulfilled  $U \cap U' \neq \emptyset$ ,  $\emptyset$  being the empty set and  $\cap$  denotes intersections of sets, in  $U \cap U'$  are defined two coordinate systems,  $\{\varphi^i = r^i \circ \varphi\}$  and  $\{\varphi'^i = r^i \circ \varphi'\}$ . Obviously, the connection between them is  $\varphi'^i = (r^i \circ \varphi' \circ \varphi^{-1}) \circ \varphi$  which implies  $\varphi'^i(p) = (r^i \circ \varphi' \circ \varphi^{-1})(\varphi^1(p), \dots, \varphi^n(p))$ for  $p \in U \cap U'$ . So, we can write  $p'^i = p'^i(p^1, \dots, p^n)$ , the explicit dependence being just given.

A family  $\{(U_{\alpha}, \varphi_{\alpha}) | \alpha \in A\}$  of charts on M is called an *atlas* if  $\{U_{\alpha}\}$  is an open cover of M,  $M = \bigcup_{\alpha \in A} U_{\alpha}$ . Here the sign  $\cup$  mean union of sets. We speak of an atlas of class (of smoothness)  $C^k$ ,  $k \in \mathbb{N} \cup \{0\}$ , or of a  $C^k$  atlas if the homeomorphisms  $\varphi_{\alpha}$  are  $C^k$  compatible in a sense that if  $U_{\alpha} \cap U_{\beta} \neq \emptyset$  for some  $\alpha, \beta \in A$ , then the mappings  $\varphi_{\beta} \circ \varphi_{\alpha}^{-1} : \varphi_{\alpha}(U_{\alpha} \cap U_{\beta}) \to \varphi_{\beta}(U_{\alpha} \cap U_{\beta})$  are of class  $C^k$  as mappings, which actually are bijections, between open subsets of  $\mathbb{K}^n$ . A  $C^k$  mapping between open subsets of  $\mathbb{K}^n$  is a one having continuous partial derivatives for all orders  $r \leq k, \ k \in \{0\} \cup \mathbb{N} \cup \{\omega\}; \ C^0$  means continuous,  $C^{\omega}$  stands for (real or complex) analytic, and if  $k \in \mathbb{N}$  is arbitrary we speak of  $C^{\infty}$  mappings.

values k = 3, 2, 1 will be sufficient for the most problems to be considered in this book.

Obviously, any topological manifold M admits atlases. A chart  $(U, \varphi)$  of Mis said to be  $(C^k)$  compatible with an atlas  $\{(U_\alpha, \varphi_\alpha) | \alpha \in A\}$  if  $\varphi_\alpha \circ \varphi^{-1} : \varphi(U_\alpha \cap U) \to \varphi_\alpha(U_\alpha \cap U)$  is a  $C^k$  mapping for every  $\alpha$  such that  $U_\alpha \cap U \neq \emptyset$ . Two atlases are  $(C^k)$  compatible if each chart of one of them is  $(C^k)$  compatible with the other atlas. The compatibility relation between atlases is an equivalence relation. An equivalence class of atlases on M is called a  $C^k$  differentiable structure (on M). A differentiable structure on M can also be defined as the maximal (complete) atlas  $C^k$  compatible with (and containing) a given atlas  $\{(U_\alpha, \varphi_\alpha) | \alpha \in A\}$ . So, any atlas, when expanded to a complete atlas, gives rise to a  $C^k$  structure.

**Definition 2.2.** A differentiable manifold of class  $C^k$  is a topological manifold with fixed  $C^k$  differentiable structure on it.

Remark 2.1. To be quite precise, the object described in Definition 2.2 is called differentiable manifold without boundary of class  $C^{k}$  [14, p. 38]. There are a number of equivalent or similar definitions of a manifold with boundary (see, e.g., [14, Chapter II, § 4], [16, p. 139], [12, pp. 208–209]), one of them being the following. Let  $\lambda \colon \mathbb{K}^n \to \mathbb{R}$  be continuous linear mapping into  $\mathbb{R}$ . A Euclidean ('upper') half space or plane  $H^+_{\lambda}$  is the closed subset of  $\mathbb{K}^n$  given by the non-negative values of  $\lambda, H_{\lambda}^+ := \{h | h \in \mathbb{K}^n, \lambda(h) \ge 0\}$ . For instance, in the real case,  $\mathbb{K} = \mathbb{R}$ , one usually takes  $\lambda = r^n$ ,  $\lambda(c^1, \dots, c^n) = c^n$  for  $c^i \in \mathbb{R}$ , which results in  $H^+_{\lambda} = \mathbb{R}^n_+ := \{c | c \in \mathbb{R}^n, r^n(c) \ge 0\}$ , or  $\lambda : c \mapsto \sum_{i=1}^n a^i c^i$  for a fixed non-zero  $a \in \mathbb{R}^n$ . A chart (with boundary) in a K-topological manifold M is a pair  $(U, \varphi)$  of an open subset  $U \subseteq M$ and homeomorphism  $\varphi \colon U \to V$  from U on an open subset V of  $H_{\lambda}^+$ . There are charts  $(U, \varphi)$  for which  $\varphi(U)$  is open set in  $H_{\lambda}^+$  homeomorphic to  $\mathbb{K}^n$  and others for which  $\varphi(U)$  is open set in  $H_{\lambda}^+$  but not in  $\mathbb{K}^n$ . If we make this modification in all of the above text, i.e., take a chart with boundary for a chart, we obtain the notions of atlas and differentiable structure with boundary and correspondingly a manifold with boundary. An example of such a manifold, which is frequently met, is a real interval J closed from one of both ends with  $(J, \mathrm{id}_J)$  as a differentiable structure. The manifolds without boundary are evident special case of the ones with boundary. Most of the results concerning manifolds without boundary can be transferred on manifolds with boundary; general directions on how this can be done are given in [14, pp. 38–40]. In a general context, a manifold should mean manifold with boundary. However, in this book we shall deal exclusively with manifolds without boundary which, for brevity, will be called simply manifolds.

Often a differentiable manifold of class  $C^k$  is denoted as  $C^k$ -manifold. So, a *B*-manifold with  $B = n, \mathbb{K}, C^k$  means different things depending on the context. In this book only  $C^k$ -manifolds will be employed; they will, for brevity, be referred simply as manifolds, the class of smoothness, usually  $C^3$ , or  $C^2$ , or  $C^1$ , will be clear from the context.

#### 2. Differentiable manifolds

**Example 2.2.** The sets  $\mathbb{R}^n$  and  $\mathbb{C}^n$  are  $C^{\infty}$  manifold with  $(\mathbb{R}^n, \mathsf{id}_{\mathbb{R}^n})$  and  $(\mathbb{C}^n, \mathsf{id}_{\mathbb{C}^n})$ , respectively, as  $C^{\infty}$  differentiable structures.

It is well known [10,11,26], every *n*-dimensional *complex* manifold has also a structure of 2*n*-dimensional *real* manifold and, consequently, can be considered as real manifold of dimension 2*n*. Since the complex and real differentiable structures on a complex manifold are in bijective correspondence, both points of view, the real and complex ones, on it are equivalent. The dimension of a complex manifold M considered as a real manifold is denoted by dim<sub> $\mathbb{R}$ </sub> M. Hence, we have dim<sub> $\mathbb{R}$ </sub>  $M = \dim M$  if M is real and dim<sub> $\mathbb{R}$ </sub>  $M = 2 \dim M = 2 \dim_{\mathbb{C}} M$  if M is complex.

A subset N of m-dimensional manifold M is called an n-dimensional submanifold of M,  $n \leq m$ , if for every point  $p \in N$  there exists a chart  $(U, \varphi)$  of M with p in its domain,  $U \ni p$ , and such that

$$\varphi \colon N \cap U \to \mathbb{K}^n \times \{a\}, \qquad \varphi \colon q \mapsto (\varphi^1(q), \dots, \varphi^n(q), a^1, \dots, a^{m-n})$$

for some fixed  $a = (a^1, \ldots, a^{m-n}) \in \mathbb{K}^{m-n}$  and all  $q \in N \cap U$ . The set of charts  $\{(\bar{U}, \bar{\varphi})\}$ , given by  $\bar{U} := U \cap N$  and  $\bar{\varphi} \colon \bar{U} \to \mathbb{K}^n$  with  $\bar{\varphi}(q) := (\varphi^1(q), \ldots, \varphi^n(q))$ , is an atlas on N of the same class as the original atlas  $\{(U, \varphi)\}$  on M. So, in more precise terms, a subset  $N \subseteq M$  with just described differentiable stricture is an *n*-dimensional submanifold of the m-manifold  $M, m \geq n$ .<sup>4</sup> One should be aware of the fact that presently subtle versions of different definitions of a 'submanifold' are in current use.

**Example 2.3.** The open subsets of a manifold M (which will be referred as neighborhoods) and the sets consisting of finite number of its points are respectively dim M- and zero-dimensional submanifolds of M.

Important examples of submanifolds are generated according to the theorem in [12, p. 228].

### 2.2. Differentiable mappings

Let M and N be  $C^k$ -manifolds of dimension m and n, respectively, and a mapping  $f: M \to N$  be given. If  $(U, \varphi)$  and  $(V, \psi)$  are charts in the respective complete atlases on M and N such that  $f(U) \subseteq V$ , the mapping  $\overline{f} := \psi \circ f \circ \varphi^{-1}$  from  $\varphi(U) \subseteq \mathbb{K}^m$  into  $\psi(V) \subseteq \mathbb{K}^n$  (locally) represents f in them.

A mapping f is differentiable of class  $C^r$ ,  $r \leq k$  at  $p \in M$  if  $\psi \circ f \circ \varphi^{-1}$ is a  $C^r$  mapping at  $\varphi(p)$ . Such mappings are also called  $C^r$  differentiable or  $C^r$ mappings (at p).<sup>5</sup> Obviously, the  $C^r$  differentiability at a point is independent of the particular charts utilized above. If  $r^1, \ldots, r^m$  and  $v^1, \ldots, v^n$  are the standard

 $<sup>{}^{4}</sup>$ If N already has a manifold structure, we suppose it to be equivalent to the described submanifold one. Otherwise one should clearly specify with respect to which differentiable structure N is considered as a manifold.

<sup>&</sup>lt;sup>5</sup>For manifolds with boundary (see Remark 2.1) the following modifications must be done [16, p. 139]. If O is an open subset of  $H^+_{\lambda}$ , the mapping  $g: O \to H^+_{\mu}$ , with  $\lambda, \mu: \mathbb{K}^n \to \mathbb{R}$  being  $C^0$ 

coordinate functions on  $\mathbb{K}^m$  and  $\mathbb{K}^n$  respectively, in the local coordinate systems  $\{\varphi^i := r^i \circ \varphi, i = 1, ..., m\}$  and  $\{\psi^j := v^j \circ \psi, j = 1, ..., n\}$  the  $C^r$  differentiability means that the set of n functions

$$f^j:=\psi^j\circ f\circ \varphi^{-1}=v^j\circ \psi\circ f\circ \varphi^{-1}\colon \varphi(U)\to \mathbb{K}$$

are of class  $C^r$  at the point  $\varphi(p) = (p^1, \ldots, p^m) \in \mathbb{K}^m$ ,  $p^i := \varphi^i(p)$ . Sometimes this is expressed by the assertion that the set of functions  $\bar{q}^j = f^j(q^1, \ldots, q^m)$ ,  $j = 1, \ldots, n, q^i \in \mathbb{K}$ , called *expression* of f in coordinates, are of class  $C^r$  at  $q^i = p^i$ .

The mapping  $f: M \to N$  is of class  $C^r$ ,  $r \leq k$  (is a  $C^r$  mapping, is  $C^r$  differentiable) if it is  $C^r$  differentiable at every point  $p \in M$ . Analogously, f is a  $C^r$  mapping on  $M' \subset M$  if it is a  $C^r$  at every point  $p \in M'$ .

The above definitions are *mutatis mutandis* transferred on mappings like  $f: M' \to N'$  with  $M' \subset M$  and  $N' \subset N$ : at the corresponding places U and V have to be replace by  $U \cap M'$  and  $V \cap N'$  respectively.

Under a differentiable mapping or simply mapping between  $C^k$  manifolds, we shall understand a  $C^k$  mapping between them, i.e., a  $C^r$  mapping with maximal r, r = k.

**Example 2.4.** A mapping  $f: M \to \mathbb{K}^m$  is called function on M. If it is of class  $C^r$  (at  $p \in M$ ), it is called a  $C^r$  function (at  $p \in M$ ). It is real or complex depending on  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ . If  $(U, \varphi)$  is a chart of M and we take  $(\mathbb{K}^m, \mathrm{id}_{\mathbb{K}^m})$  as a natural atlas of  $\mathbb{K}^m$ , then f has the local representation  $\overline{f} = f \circ \varphi^{-1} : \varphi(U) \to \mathbb{K}^m$ . The algebra of  $C^r$ ,  $r \leq k$  (resp. all) functions on (one and the same neighborhood of) a subset  $U \subseteq M$  of a  $C^k$  manifold with  $k \geq 0$  will be denoted by  $\mathfrak{F}^r(U)$  (resp. by  $\mathfrak{F}(U)$ ). In particular,  $\mathfrak{F}^r(p), p \in M$  and  $\mathfrak{F}^r(M)$  are the algebras of  $C^r$  functions on a (fixed) neighborhood of p and on the whole M respectively. Let us note that the neighborhoods (if  $U \neq M$ ) in these definitions must be fixed as otherwise the sets  $\mathfrak{F}^r(U)$  and  $\mathfrak{F}(U)$  are not quite algebras as they will not have unique zero elements: two zero functions f + (-f) and g + (-g) coincide iff the domains of f and g are identical.

A mapping  $f: M \to N$  between  $C^k$  manifolds is called  $C^r$ ,  $r \leq k$ , diffeomorphism if it is bijective and f and  $f^{-1}$  are of class  $C^r$ . A  $C^k$  diffeomorphism is referred as simply diffeomorphism. With respect to the manifolds, the diffeomorphisms play the same role as the isomorphisms (resp. homeomorphisms) in the theory of vector (resp. topological) spaces.

linear functions, is of class  $C^r$  if it extends to a mapping  $g': O' \to H^+_{\mu}, g'|O = g$ , of class  $C^r$ , where O' is open subset of  $H^+_{\lambda}$  containing O. A mapping between manifolds with boundary is of class  $C^r$  if it is locally of class  $C^r$ , i.e., if  $\psi \circ f \circ \varphi^{-1}$  is such in the above notation.

#### 2.3. Tangent vectors and vector fields

A number of different but equivalent definitions of a tangent vector can be found in the literature [7, 11–13]. Below we reproduce the most direct one.

**Definition 2.3.** A tangent vector on a differentiable manifold M of class  $C^k$ , with  $k \ge 1$ , at a point  $p \in M$  is a linear mapping  $X_p : \mathfrak{F}^1(p) \to \mathbb{K}$  satisfying the Leibnitz rule:

$$X_p(af+bg) = aX_p(f) + bX_p(g), \qquad a, b \in \mathbb{K} \quad f, g \in \mathfrak{F}^1(p) \qquad (2.1)$$

$$X_p(fg) = g(p)X_p(f) + f(p)X_p(g), \qquad f, g \in \mathfrak{F}^1(p).$$
(2.2)

Thus a tangent vector at p is a derivation of the algebra  $\mathfrak{F}^1(p)$ .<sup>6</sup>

**Example 2.5.** A mapping  $\gamma: J \to M$ , J being  $\mathbb{R}$ -interval of arbitrary type (open or closed from one or the both ends), is called *path* (or *parameterized curve*).<sup>7</sup> If  $\gamma$  is a  $C^1$  path, the mapping  $\dot{\gamma}(s): \mathfrak{F}^1(\gamma(s)) \to \mathbb{K}$  given by

$$\left(\dot{\gamma}(s)\right)(f) := \frac{\mathrm{d}f(\gamma(t))}{\mathrm{d}t}\Big|_{t=s}, \qquad s, t \in J \quad f \in \mathfrak{F}^1(\gamma(s)), \tag{2.3}$$

where  $\cdots |_A$  means the restriction of  $\cdots$  to the set A, is an important example of a tangent vector at  $p = \gamma(s) \in \gamma(J)$ .<sup>8</sup> So,  $\dot{\gamma}(s)(f)$  is the derivative of f along  $\gamma$ at the parameter value s. The vector  $\dot{\gamma}(s)$  is called *tangent to*  $\gamma$  *at*  $\gamma(s)$  *or, more precisely, at the parameter value* s as it may happen that, for a non-injective path, there are  $s, t \in J$  such that  $s \neq t$ ,  $\gamma(s) = \gamma(t)$  and  $\dot{\gamma}(s) \neq \dot{\gamma}(t)$ . Sometimes  $\dot{\gamma}(s)$ is denoted by  $\frac{d}{ds}$  which can be confusing. As a result of (2.3), the vector  $\dot{\gamma}(s)$  is often denoted by  $\frac{d}{dt}\Big|_{t=s}\gamma(t)$ .

The set of all tangent vectors at  $p \in M$  is a linear space (over  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ ) called *tangent* (vector) space to M at p. It is denoted by  $T_p(M)$  or, for brevity, by  $M_p$  or  $T_p$  if there is no risk of ambiguity. The dimensions of  $T_p(M)$  and Mcoincide for every p, dim  $T_p(M) \equiv \dim M$ . Moreover, if  $(U, \varphi)$  is a chart with

<sup>&</sup>lt;sup>6</sup>Generally, a mapping  $\partial \colon R \to L$  from a ring R to a left R-module L is *derivation* if  $\partial(x \cdot y) = x\partial(y) + y\partial(x)$ ,  $x, y \in R$ . The case when R is an algebra, which in turn is a linear space over  $\mathbb{K}$ , is frequently met in the differential geometry. In the last case on  $\partial$  is added the restriction to be  $\mathbb{K}$ -linear.

<sup>&</sup>lt;sup>7</sup>We make a clear distinction between the concepts 'path' and 'curve'. By a *curve* or geometric curve we understand an equivalence class of paths with respect to some set of changes of the path's parameter. More precisely, let T be some set of bijective mappings between the  $\mathbb{R}$ -intervals, two paths  $\gamma_a: J_a \to A$ , with a = 1, 2 and A being a non-empty set, are equivalent (with respect to T) if there is  $\tau \in T$  such that  $\gamma_2 = \gamma_1 \circ \tau$  and  $J_2 = \tau(J_1)$ . This is an equivalence relation on the set of paths in A. Any equivalence class  $[\gamma]$  of paths equivalent to  $\gamma: J \to A$  is called a curve. Any path of a curve  $[\gamma]$  provides a 'parameterization' of the set  $\gamma(J) \subseteq A$  which sometimes is also called a curve. Most often as T is chosen the class of  $(C^0, C^1, \ldots)$  diffeomorphisms between the real intervals.

<sup>&</sup>lt;sup>8</sup>If J has end point(s) and s is such a point, the derivative in (2.3) is considered as one-sided, resp.  $\dot{\gamma}(s)$  is one-sided tangent vector to  $\gamma$  at the endpoint(s).

 $U \ni p$ ,  $\{x^i = r^i \circ \varphi\}$  is the corresponding coordinate system, and  $\{r^i\}$  are the standard Cartesian coordinates on  $\varphi(U)$ , then the mappings  $(i = 1, ..., \dim M)$ 

$$\frac{\partial}{\partial x^i}\Big|_p: \mathfrak{F}^1(p) \to \mathbb{K}, \qquad \frac{\partial}{\partial x^i}\Big|_p: f \mapsto \frac{\partial f}{\partial x^i}\Big|_p:= \frac{\partial (f \circ \varphi^{-1})}{\partial r^i}\Big|_{\varphi(p)} \tag{2.4}$$

form a basis  $\{\frac{\partial}{\partial x^i}|_p\}$  of  $T_p(M)$ , called associated (coordinate or holonomic [5, p. 110] basis) with the given chart or local coordinates.<sup>9</sup> Therefore every  $X_p \in T_p(M)$  has a unique decomposition

$$X_p = X_p^i \left. \frac{\partial}{\partial x^i} \right|_p := \sum_{i=1}^{i=n} X_p^i \left. \frac{\partial}{\partial x^i} \right|_p \tag{2.5}$$

where

$$X_p^i := X_p(x^i) = X_p(r^i \circ \varphi) \in \mathbb{K}$$
(2.6)

are the (local) components of  $X_p$  (with respect to  $(U, \varphi)$  or  $\{x^i\}$ ). Here, as well as throughout the whole book, the Einstein's summation convention is assumed (see the list of conventions, p. xi).

**Example 2.6.** Let  $\gamma: J \to M$  be a  $C^1$  path,  $s \in J$  and  $(U, \varphi)$  be a chart such that  $U \ni \gamma(s)$ . Then

$$\dot{\gamma}(s) = \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=s}\gamma(t) = \frac{\mathrm{d}\gamma^{i}(s)}{\mathrm{d}s}\frac{\partial}{\partial x^{i}}\Big|_{\gamma(s)},\tag{2.7}$$

where  $\gamma^{i}(s) = x^{i}(\gamma(s))$  are the local coordinates of  $\gamma(s)$  in  $\{x^{i}\}$ .

For every (tangent) vector  $X_p \in T_p(M)$  there exists a path  $\gamma: J \to M$  such that for some  $s \in J$  is fulfilled  $\gamma(s) = p$  and  $\dot{\gamma}(s) = X_p$ .<sup>10</sup> In such representation  $X_p^i = X_p(x^i) = \mathrm{d}x^i(\gamma(t))/\mathrm{d}t\Big|_{t=s}$ .

A change  $(U, \varphi) \mapsto (U', \varphi')$  of the chart containing  $p, p \in U \cap U' \neq \emptyset$ , or equivalently, of the local coordinates  $x^i \mapsto x'^i$  implies corresponding transformation of the basic vectors (2.4) and the vector's components in (2.5). For the components of  $X_p \in T_p(M)$ , due to (2.5) and (2.4), we have

$$X_{p}^{\prime i} = X_{p}(x^{\prime i}) = \sum_{j} X_{p}^{j} \left. \frac{\partial (x^{\prime i} \circ \varphi^{-1})}{\partial r^{j}} \right|_{\varphi(p)} = \left. \frac{\partial x^{\prime i}}{\partial x^{j}} \right|_{p} X_{p}^{j}$$
(2.8)

and, analogously,  $X_p^i = \partial x^i / \partial x'^j |_p X_p'^j$ . Very often the restrictions  $|_{\varphi(p)}$  and  $|_{\varphi'(p)}$  in these formulae are abbreviate to  $|_p$  or even are not written at all. According to (2.6) the components of  $\frac{\partial}{\partial x'^i}|_p$  with respect to  $\frac{\partial}{\partial x^i}|_p$  are  $(\frac{\partial}{\partial x'^i}|_p)^j$ 

<sup>&</sup>lt;sup>9</sup>Notice, in (2.4) enters the local representation  $\overline{f} := f \circ \varphi^{-1}$  of f with respect to the charts  $(U, \varphi)$  of M and  $(\mathbb{K}^n, \mathsf{id}_{\mathbb{K}^n})$  of  $\mathbb{K}^n$ .

 $<sup>^{10}</sup>$  This path is not unique. Actually, there is a family of such paths having a common tangent vector at p.

#### 2. Differentiable manifolds

$$\frac{\partial}{\partial x'^{i}}\Big|_{p}(x^{j}) = \frac{\partial(x^{j} \circ \varphi'^{-1})}{\partial r^{i}}\Big|_{\varphi'(p)} = \frac{\partial x^{j}}{\partial x'^{i}}\Big|_{p}, \text{ i.e.,}$$
$$\frac{\partial}{\partial x'^{i}}\Big|_{p} = \frac{\partial x^{j}}{\partial x'^{i}}\Big|_{p}\frac{\partial}{\partial x^{i}}\Big|_{p} = \frac{\partial(x^{j} \circ \varphi')}{\partial r^{i}}\Big|_{\varphi'(p)}\frac{\partial}{\partial x^{i}}\Big|_{p}.$$
(2.9)

More generally, if  $\{E_i|_p\}$  is an arbitrary basis of  $T_p(M)$  and the expansion of  $X_p \in T_p(M)$  over  $\{E_i|_p\}$  is

$$X_p = X_p^i E_i|_p, \qquad X_p^i \in \mathbb{K}, \tag{2.10}$$

the change

$$\{E_i|_p\} \mapsto \{E'_i|_p = A_i^j(p)E_j|_p\},$$
(2.11)

with a nondegenerate matrix  $A(p) := \left[A_i^j(p)\right], A_i^j(p) \in \mathbb{K}$ , implies

$$X_{p}^{i} \mapsto X_{p}^{\prime \, i} = \left(A^{-1}(p)\right)_{j}^{i} X_{p}^{j} \tag{2.12}$$

as one must have  $X_p = X_p^{\prime i} E_i^{\prime}|_p$ . Clearly, the equations (2.8) and (2.9) are special case of (2.12) and (2.11), respectively.

The basis  $\left\{\frac{\partial}{\partial x^i}\Big|_p\right\}$  is usually referred as a *coordinate* one (associated with or generated by the local coordinates  $\{x^i\}$ ) while  $\{E_i|_p\}$  is called non-coordinate. For any basis  $\{E_i|_p\}$  there exist local coordinates with respect to which it is coordinate (see Lemma II.5.2 on page 116).

The set of all tangent vectors on a manifold M is denoted by T(M), i.e.,  $T(M) := \bigcup_{p \in M} T_p(M)$ , and is called tangent bundle space.

**Definition 2.4.** A (*tangent*) vector field on  $U \subseteq M$  is a mapping X assigning to each  $p \in U$  a (tangent) vector  $X_p$  at p, i.e.,  $X : p \mapsto X_p \in T_p(M)$ .

If U is clear from the context or if U = M, we speak simply of a vector field.

**Example 2.7.** If  $\gamma: J \to M$  is a  $C^1$  path without self-intersections, the mapping  $\dot{\gamma}: \gamma(s) \mapsto \dot{\gamma}(s), \dot{\gamma}(s)$  being the vector tangent to  $\gamma$  at  $\gamma(s)$  (see (2.3)), is an important example of a vector field over  $\gamma(J)$ . It is called *tangent to*  $\gamma$  vector field.<sup>11</sup>

The set  $\mathfrak{X}(U)$  of the vector fields on U is a linear space over  $\mathbb{K}$ ; for the purpose we put  $(\alpha X + \beta Y)$ :  $p \mapsto \alpha X_p + \beta Y_p$  with  $\alpha, \beta \in \mathbb{K}, p \in U$ , and  $X, Y \in \mathfrak{X}(U)$ . Moreover, this set can naturally be turned into a (left) module over the ring (algebra in this case)  $\mathfrak{F}(U)$  of functions on U; this is done by defining (fX + gY):  $p \mapsto f(p)X_p + g(p)Y_p$  with  $f, g: U \to \mathbb{K}$ .

There is a naturally defined action of the vector fields on  $C^1$  functions, viz. a vector field X on  $U \subseteq M$  can be regarded as a mapping  $\mathfrak{F}^1(U) \to \mathfrak{F}(U)$  by setting  $X: f \mapsto Xf := X(f)$  with  $X(f): p \mapsto X_p(f)$  for all  $p \in U$  and  $f \in \mathfrak{F}^1(U)$ .

<sup>&</sup>lt;sup>11</sup>If  $\gamma$  is not injective, the mapping  $\dot{\gamma}: \gamma(s) \to \dot{\gamma}(s)$  may turn to be multiple-valued at the points of self-intersection.

In terms of local coordinates  $\{x^i\}$  defined via a chart  $(U, \varphi)$ , we have

$$X = X^{i} \frac{\partial}{\partial x^{i}}, \qquad \frac{\partial}{\partial x^{i}} \colon p \mapsto \frac{\partial}{\partial x^{i}}\Big|_{p} \coloneqq \frac{\partial(\cdot \circ \varphi^{-1})}{\partial r^{i}}\Big|_{\varphi(p)}, \tag{2.13}$$

where the functions

$$X^{i} := X(x^{i}) = X(r^{i} \circ \varphi) \colon U \to \mathbb{K}$$
(2.14)

are called *components* of X with respect to  $\{x^i\}$  or  $(U, \varphi)$ .

It is important to be noted that the set of vector fields  $\mathfrak{X}(U)$  is a finitely generated (left)  $\mathfrak{F}(U)$ -module of rank  $n = \dim M$  (see (2.13)) but it is also infinitely dimensional linear (vector) space over  $\mathbb{K}$ . A basis in the former case is formed by the vector fields  $\{\frac{\partial}{\partial x^i}\}$ . This basis is called associated to (generated by) the local coordinates  $\{x^i\}$  or the chart  $(U, \varphi)$ .

By virtue of (2.8), (2.9), (2.13), and (2.14), a change  $(U, \varphi) \mapsto (U', \varphi')$  of the chart implies the transforms

$$\frac{\partial}{\partial x^{i}} \mapsto \frac{\partial}{\partial x^{\prime i}} = \frac{\partial x^{j}}{\partial x^{\prime i}} \frac{\partial}{\partial x^{j}}, \qquad X^{i} \mapsto X^{\prime i} = \frac{\partial x^{\prime i}}{\partial x^{j}} X^{j}$$
(2.15)

where  $\frac{\partial x^j}{\partial x'^i} : p \mapsto \frac{\partial x^j}{\partial x'^i} \Big|_p = \frac{\partial (x^j \circ \varphi'^{-1})}{\partial r^i} \Big|_{\varphi'(p)}$  and  $\frac{\partial x'^i}{\partial x^j} : p \mapsto \frac{\partial x'^i}{\partial x^j} \Big|_p = \frac{\partial (x'^i \circ \varphi^{-1})}{\partial r^j} \Big|_{\varphi(p)}$ ,  $p \in U \cap U'$ .

We call a basis  $\{E_i\}$  of the module of vector fields on (over) U a frame on (over) U.<sup>12</sup> So, a frame on U is a collection of dim M vector fields on U such that at every point  $p \in U$  they form a basis of  $T_p(M)$ . At a single point  $p, U = \{p\}$ , the concepts 'basis in  $T_p(M)$ ' and 'frame on (over, at) p' are synonyms.

The frame  $\{\frac{\partial}{\partial x^i}\}$  on U is referred as a *coordinate* (or natural) one (associated with or generated by the local coordinates  $\{x^i\}$ ) while a general frame  $\{E_i\}$  is referred as a *noncoordinate* frame. A frame  $\{E_i\}$  is called *holonomic* on U if there exist local coordinates  $\{x^i\}$  on U such that  $E_i = \partial/\partial x^i$ , i.e., if  $\{E_i\}$  is the associated with  $\{x^i\}$  coordinate frame. Otherwise, if such  $\{x^i\}$  do not exist, the frame is called *anholonomic*.<sup>13</sup> Equivalently, the holonomic (resp. anholonomic) frames can be defined as ones whose basic vector fields commute (resp. do not commute); for details *vide infra* Section 8.

Any change

$$\{E_i\} \mapsto \{E'_i = A^j_i E_j\} \tag{2.16}$$

of a frame  $\{E_i\}$  on U with a non-degenerate matrix-valued function  $A = [A_i^j]$ , where  $A_i^j : U \to \mathbb{K}$ , implies the transformation

$$X^{i} \mapsto X^{\prime i} = \left(A^{-1}\right)^{i}_{j} X^{j} \tag{2.17}$$

of the local components of some vector field  $X = X^i E_i = X'^i E'_i$ .

 $<sup>^{12}</sup>$ For details, see Section 8.

<sup>&</sup>lt;sup>13</sup>A frame given only on U can be extended outside U, if  $U \subset M$ , in a holonomic as well as in anholonomic way. (Cf. Remark 5.1 on page 40 or see Lemma II.5.2 on page 116.)

#### 2. Differentiable manifolds

A vector field X on  $U \subseteq M$ , M being a  $C^k$  differentiable manifold, is said to be differentiable of class  $C^r$ , or simply is called  $C^r$  (vector) field, with  $r \leq k$  if for every  $C^{r+1}$  function  $f: U \to \mathbb{K}$  the function  $X(f): U \to \mathbb{K}$  is of class  $C^r$ . A vector field is of class  $C^r$  if and only if its local components are  $C^r$  functions in any local chart (see (2.13)). The (left) module of all  $C^r$  vector fields on U will be denoted by  $\mathfrak{X}^r(U)$ .

Define the bracket, called also Lie bracket or commutator, of two  $C^1$  vector fields  $X,Y\in\mathfrak{X}^1(U)$  by  $^{14}$ 

$$[X,Y]_{-} := X \circ Y - Y \circ X. \tag{2.18}$$

In local coordinates  $\{x^i\}$ , the local components of  $[X, Y]_{-}$  are  $[X, Y]_{-}^j = X^j \frac{\partial Y^i}{\partial x^j} - Y^j \frac{\partial X^i}{\partial x^j}$ ; in an arbitrary frame, we have  $[X, Y]_{-}^j = X(Y^i) - Y(X^i)$ . The set  $\mathfrak{X}^1(U)$  is an infinitely dimensional Lie algebra over the field  $\mathbb{K}$  under the operation  $[\cdot, \cdot]_{-}: \mathfrak{X}^1(U) \times \mathfrak{X}^1(U) \to \mathfrak{X}(U)$ .

A frame  $\{E_i\}$  on a neighborhood (open set)  $U \subseteq M$  is holonomic if and only if  $[E_j, E_k]_{-|U|} = 0$  (see Section 8 for some details).

Further, in Subsection IV.2.4, we shall look on the tangent vector fields as sections of the tangent bundle over a manifold.

Let  $X \in \mathfrak{X}^0(U)$  and  $\gamma: J \to U$  be a  $C^1$  path in a neighborhood  $U \subseteq M$ . The path  $\gamma$  is called *integral path* (*curve*) for X through  $p \in U$  if

$$\dot{\gamma}(s) = X_{\gamma(s)}, \quad \gamma(s_0) = p \tag{2.19}$$

for some fixed  $s_0 \in J$  end every  $s \in J$ . Through every  $p \in U$  passes a (locally) unique integral path for X; so if two integral paths have a common point, they (locally) coincide.<sup>15</sup> On the opposite, if U is filled with a family of non-intersecting  $C^1$  paths passing through every its point, there exists a unique  $C^1$  vector field on U whose integral paths form the given family of paths. It coincides with the field of vectors tangent to the paths of the family.

Let  $f: M \to N$  be a  $C^1$  mapping between  $C^1$  manifolds M and N and  $p \in M$ . The differential or induced tangent mapping  $f_*|_p$  of f at p, denoted also by  $df_p \equiv df|_p$  and  $f'_p \equiv f'|_p$ , is a linear mapping

$$f_*|_p \colon T_p(M) \to T_{f(p)}(N) \tag{2.20}$$

such that, for all  $X_p \in T_p(M)$  and all  $g: N \to \mathbb{K}$  differentiable at f(p),

$$(f_*|_p(X_p))(g) := X_p(g \circ f).$$
 (2.21)

Evidently,  $f_*|_p$  is a linear mapping. Its matrix, in the coordinate bases associated to charts  $(U, \varphi)$  of M and  $(V, \psi)$  of N with  $U \ni p$  and  $V \ni f(p)$  can be found

 $<sup>^{14}\</sup>textsc{Often}$  the Lie bracket is denoted by  $[\,\cdot\,,\,\cdot\,],$  omitting the minus sign as a subscript.

 $<sup>^{15}</sup>$  For details on integral paths, see, e.g., [9, p. 131ff], [7, Sections 1.46–1.53] or [8, Sections 8.5–8.7].

as follows. Let  $\{x^i | i = 1, \ldots, \dim M\}$  and  $\{y^j | j = 1, \ldots, \dim N\}$  be the respective coordinate systems and  $\{r^i\}$  and  $\{v^j\}$  be the standard Cartesian coordinate systems in  $\mathbb{K}^{\dim M}$  and  $\mathbb{K}^{\dim N}$ , respectively. Then

$$\begin{split} & \left(f_*|_p \left(\frac{\partial}{\partial x^i}\Big|_p\right)\right)(g) = \frac{\partial}{\partial x^i}\Big|_p (g \circ f) = \frac{\partial(g \circ f \circ \varphi^{-1})}{\partial r^i}\Big|_{\varphi(p)} \\ & = \sum_{j=1}^{\dim N} \frac{\partial(g \circ f \circ \varphi^{-1})}{\partial v^j}\Big|_{\varphi(p)} \frac{\partial(y^j \circ f \circ \varphi^{-1})}{\partial r^i}\Big|_{\varphi(p)} = \sum_{j=1}^{\dim N} \frac{\partial(g \circ \psi^{-1})}{\partial v^j}\Big|_{\psi(f(p))} \\ & \times \frac{\partial(y^j \circ f \circ \varphi^{-1})}{\partial r^i}\Big|_{\varphi(p)} = \left(\sum_{j=1}^{\dim N} \frac{\partial(y^j \circ f)}{\partial x^i}\Big|_p \frac{\partial}{\partial y^j}\Big|_{f(p)}\right)(g), \end{split}$$

so that

$$f_*|_p \left(\frac{\partial}{\partial x^i}\Big|_p\right) = \sum_{j=1}^{\dim N} \frac{\partial (y^j \circ f \circ \varphi^{-1})}{\partial r^i}\Big|_{\varphi(p)} \frac{\partial}{\partial y^j}\Big|_{f(p)} = \sum_{j=1}^{\dim N} \frac{\partial (y^j \circ f)}{\partial x^i}\Big|_p \frac{\partial}{\partial y^j}\Big|_{f(p)}.$$
(2.22)

Therefore the matrix of  $f_*|_p$  in the bases chosen has elements

$$\frac{\partial(y^{j}\circ f\circ\varphi^{-1})}{\partial r^{i}}\Big|_{\varphi(p)} = \frac{\partial}{\partial x^{i}}\Big|_{p}(y^{j}\circ f) = \frac{\partial(y^{j}\circ f)}{\partial x^{i}}\Big|_{p},$$
(2.23)

where  $i = 1, ..., \dim M$  and  $j = 1, ..., \dim N$ , and, consequently, coincides with the Jacobi matrix of the mapping f (with respect to the given coordinate systems).

**Example 2.8.** If  $\gamma \colon \mathbb{R} \to M$  is a  $C^1$  path in a  $C^1$  manifold  $M, s \in \mathbb{R}$ , and we choose a chart  $(U, \varphi)$  of M with  $U \ni \gamma(s)$  and the standard chart  $(\mathbb{R}, \mathsf{id}_{\mathbb{R}})$  of  $\mathbb{R}$ , with coordinate function  $\mathsf{id}_{\mathbb{R}}$  (as  $r^1 = \mathsf{id}_{\mathbb{R}}$  in the one-dimensional case), then the differential  $\gamma_*|_s$  at s maps the standard basic vector  $\frac{\partial}{\partial r^1}|_s$  of  $T_s(\mathbb{R})$  into the tangent vector  $\dot{\gamma}(s)$  of  $\gamma$  at s and the matrix of  $\gamma_*|_s$  consists of the components of  $\dot{\gamma}(s)$ :

$$\gamma_*|_s \left(\frac{\partial}{\partial r^1}\Big|_s\right) = \sum_i \frac{\partial (x^i \circ \gamma)}{\partial r^1}\Big|_s \frac{\partial}{\partial x^i}\Big|_{\gamma(s)} = \dot{\gamma}^i(s) \frac{\partial}{\partial x^i}\Big|_{\gamma(s)} = \dot{\gamma}(s).$$

Note that often  $r^1$  is identified with s and one writes  $\dot{\gamma}^i(s) = \frac{\partial (x^i \circ \gamma(s))}{\partial s}$ .

Analogously, one can prove that the matrix of the differential at  $p \in M$  of a function  $f: M \to \mathbb{K}$  is formed from the gradient components of f, i.e., they are  $\frac{\partial (f \circ \varphi^{-1})}{\partial r^i} \Big|_{\varphi(p)}$  in appropriate bases.

A  $C^1$  mapping  $f: M \to N$  between  $C^1$  manifolds is called *regular* if its differential  $f_*|_p$  at  $p \in M$  is one-to-one (injective) for every  $p \in M$ . The above example (with interval J for  $\mathbb{R}$ ) shows that a  $C^1$  path  $\gamma: J \to M$  is regular iff its tangent vector  $\dot{\gamma}$  is non-zero everywhere. Generally, a mapping  $f: M \to N$  is

#### 2. Differentiable manifolds

regular iff  $\dim_{\mathbb{R}} M \leq \dim_{\mathbb{R}} N$  and the Jacobi matrix of f in some (and hence in any) bases has maximum rank, equal to  $\dim_{\mathbb{R}} M$  (in real terms).

Details concerning the differential of  $C^1$  mappings can be found in most of the standard (text)books on differential geometry, for instance, in [10, 11, 15].

#### **2.4.** Covectors and covector fields

The vector space  $T_p^*(M)$  dual to the tangent space  $T_p(M)$ ,  $p \in M$ , consists of the linear forms on  $T_p(M)$ . It is called the *cotangent (vector) space* to M at pand is of dimension  $n = \dim M$ . The elements of  $T_p^*(M)$  are referred as *covectors*, *covariant vectors*, or *covariant tangent vectors* at p. For  $X_p \in T_p(M)$ , the covector  $\omega_p \in T_p^*(M)$  is a linear mapping<sup>16</sup>

$$\omega_p \colon X_p \mapsto \omega_p(X_p) \in \mathbb{K}$$

If  $\{E_i|_p\}$  is a basis in  $T_p(M)$ , its dual basis  $\{E^i|_p\}$  in  $T_p^*$  is uniquely defined by

$$E^i|_p \left( E_j|_p \right) := \delta^i_j \tag{2.24}$$

where  $\delta_i^j$  are the Kronecker (delta-)symbols,  $\delta_i^j = 1$  for i = j and  $\delta_i^j = 0$  for  $i \neq j$ . If  $\{x^i\}$  are local coordinates in a neighborhood of p, the (full) differentials  $dx^i|_p$  at p form the natural cobasis  $\{dx^i|_p\}$  with respect to  $\{\partial/\partial x^i|_p\}$ ;  $dx^j (\partial/\partial x^i|_p) = \delta_i^j$ . For the expansion of  $\omega|_p \in T_p^*(M)$  with respect to  $\{E^i|_p\}$ , we write  $\omega_p = (\omega_p)_i E^i|_p \equiv \omega_{p\,i} E^i|_p$ . So, the components of covectors/vectors are labeled by subscripts/superscripts.

The change  $\{E^i|_p\} \mapsto \{E'^i|_p = B^i_j(p)E^j|_p\}$  with a nondegenerate matrix  $B(p) = [B^i_j(p)]$  leads to  $\omega_{p\,i} \mapsto \omega'_{p\,i} = (B^{-1}(p))^j_i \omega_{p\,j}$ . Such a transformations are rarely used. The wide spread case is a transform  $\{E^i|_p\} \mapsto \{E'^i|_p\}$  induced by a change  $\{E_i|_p\} \mapsto \{E'_i|_p = A^j_i(p)E_j|_p\}$  of the dual basis in  $T_p(M)$ . In this situation, by (2.24), we have

$$E^{i}|_{p} \mapsto E^{\prime i}|_{p} = \left(A^{-1}(p)\right)_{j}^{i}(p)E^{j}|_{p}$$
 (2.25)

while the covectors' components transform like (cf. (2.12))

$$\omega_{p\,i} \mapsto \omega'_{p\,i} = A^j_i(p)\omega_{p\,j}.\tag{2.26}$$

Hence, under (2.11) the covectors' components transform like a basis of vectors and the vectors' components change like a cobasis (basis of covectors).

<sup>&</sup>lt;sup>16</sup>Sometimes the notation  $\omega_p \colon X_p \mapsto \langle \omega_p, X_p \rangle$  is used. It expresses the action of the Kronecker (unit) tensor (see below) on the ordered pair  $(\omega_p, X_p)$ . We prefer to reserve the symbol  $\langle \cdot, \cdot \rangle$  for the scalar products in quantum mechanics as accepted in this theory. More precisely, at present, in it the symbol  $\langle \cdot | \cdot \rangle$  for the Hermitian scalar product is used which sometimes is replaced by  $\langle \cdot, \cdot \rangle$  or  $(\cdot, \cdot)$ .

If local coordinate  $\{x^i\}$  and  $\{x'^i\}$  in charts  $(U, \varphi)$  and  $(U', \varphi')$ , respectively, are employed and  $U \cap U' \neq \emptyset$ , then it is fulfilled (see (2.9))<sup>17</sup>

$$A_i^j(p) = \frac{\partial x^j}{\partial x'^i}\Big|_p := \frac{\partial (x^j \circ \varphi'^{-1})}{\partial r^i}\Big|_{\varphi'(p)}, \qquad p \in U \cap U' \neq \varnothing.$$
(2.27)

Therefore (2.25) and (2.26) now read (cf. (2.9) and (2.8) resp.)

$$\mathrm{d}x^{i}|_{p} \mapsto \mathrm{d}x^{\prime i}|_{p} = \frac{\partial x^{\prime i}}{\partial x^{j}}\Big|_{p} \mathrm{d}x^{j}|_{p} := \frac{\partial (x^{\prime i} \circ \varphi^{-1})}{\partial r^{j}}\Big|_{\varphi(p)} \mathrm{d}x^{j}|_{p}$$
(2.28)

$$\omega_{p\,i} \mapsto \omega'_{p\,i} = \frac{\partial x^j}{\partial x'^i} \Big|_p \omega_{p\,j} := \frac{\partial (x^j \circ \varphi'^{-1})}{\partial r^i} \Big|_{\varphi'(p)} \omega_{p\,j}.$$
(2.29)

A covariant vector (covector) field  $\omega$  on  $U \subseteq M$  is a mapping assigning to each  $p \in U$  a covector at  $p, \omega: p \mapsto \omega_p \in T_p^*(M)$ . It is often called a *one-form* (abbreviated to 1-form), or differential form of degree 1.<sup>18</sup> Equivalently, a 1-form  $\omega$  on U can be considered (defined) as  $\mathfrak{F}(U)$ -linear mapping from the (left)  $\mathfrak{F}(U)$ module  $\mathfrak{X}(U)$  into  $\mathfrak{F}(U)$  given by

$$\omega \colon X \mapsto \omega(X), \quad \omega(X) \colon p \mapsto \omega_p(X_p), \qquad X \in \mathfrak{X}(U), \ p \in U.$$
(2.30)

In a chart  $(U, \varphi)$ , the full differentials  $dx^i$  of the local coordinates  $\{x^i\}$  form a natural basis  $\{dx^i\}$  in the set of 1-forms on U which is dual to the basis  $\{\partial/\partial x^i\}$ in  $\mathfrak{X}(U)$ :

$$\omega = \omega_i \mathrm{d}x^i, \qquad \mathrm{d}x^i \left(\partial/\partial x^j\right) := \delta^i_j. \tag{2.31}$$

More generally, if  $\{E_i\}$  is a frame on U, i.e., a basis in the module  $\mathfrak{X}(U)$ , then in the dual to it *coframe*  $\{E^i\}$  we have

$$\omega = \omega_i E^i, \quad E^i := (E_i)^*, \quad E^i(E_j) := \delta^i_j.$$
 (2.32)

In such a pair  $(\{E_i\}, \{E^i\})$  of frame and coframe, the mapping  $\omega \colon X \mapsto \omega(X)$  is given by  $\omega(X) = \omega_i X^i \in \mathfrak{F}(U)$ . The change (2.16) of the frame implies a change of the dual coframe and of the component of a covector field in it:

$$E^{i} \mapsto E^{\prime i} = \left(A^{-1}\right)^{i}_{j} E^{j}, \quad \omega_{i} \mapsto \omega^{\prime}_{i} = A^{j}_{i} \omega_{j}.$$

$$(2.33)$$

In a case of a coordinate frame  $\{\partial/\partial x^i\}$  and its dual coframe  $\{dx^i\}$  we, obviously, obtain

$$A_j^i = \frac{\partial x^i}{\partial x'^j} \colon p \mapsto \frac{\partial x^i}{\partial x'^j} \Big|_p = \frac{\partial (x^i \circ \varphi'^{-1})}{\partial r^j} \Big|_{\varphi'(p)}.$$
 (2.34)

A covector field  $\omega$  on U is said to be differentiable of class  $C^r$  if for  $X \in \mathfrak{X}^r(U)$ the function  $\omega(X)$  is of class  $C^r$  or, equivalently, the components of  $\omega$  (in one and hence in any coframe) are in  $\mathfrak{F}^r(U)$ . For brevity such a covector field is called  $C^r$ covector field (or  $C^r$  1-form).

<sup>&</sup>lt;sup>17</sup>Recall that  $x^i := r^i \circ \varphi$  and  $r^i \colon \mathbb{K}^n \to \mathbb{K}$  is defined by  $r^i(c^1, \dots, c^n) := c^i, c^i \in \mathbb{K}$ .

 $<sup>^{18}{\</sup>rm With}$  a few exceptions, in this work differential forms of degree greater than one (and the exterior algebra) will not be considered.

#### 2.5. Tensors and tensor fields. Tensor algebras

Denote by  $\otimes$  the tensor product sign (see, e.g., [11, Chapter 1, § 2]) and let  $\otimes^r V := V \otimes \cdots \otimes V$  where the vector or vector space V over K is taken  $r \in \mathbb{N}$  times; by definition  $\otimes^0 V := \mathbb{K}$ .

The tensor space of type  $(r, s), r, s \in \mathbb{N} \cup \{0\}$ , (and order (rank) r + s) at (over)  $p \in M$  is

$$T_{p_s}^{\ r}(M) := \left[ \otimes^r \left( T_p^*(M) \right) \right] \otimes \left[ \otimes^s \left( T_p(M) \right) \right]$$
(2.35)

and consists (for  $r + s \ge 1$ ) of all multilinear forms

$$\left[\times^{r}(T_{p}(M))\right]\times\left[\times^{s}(T_{p}^{*}(M))\right]\to\mathbb{K},$$
(2.36)

with  $\times^r V := V \times \cdots \times V$  (*r*-times). It is a K-linear space of dimension  $(\dim M)^{r+s}$ . Since the tensor product is not commutative, the order of the multipliers in (2.35) is essential.

Remark 2.2. More generally, the tensor space of type  $(r_1, \ldots, r_m; s_1, \ldots, s_m) \in \times^{2m} (\mathbb{N} \cup \{0\})$  at p is defined as

$$T_p^{r_1} \cdots r_m \atop \ldots \quad s_m \\ := \left[ \otimes^{r_1} T_p^*(M) \right] \otimes \left[ \otimes^{s_1} T_p(M) \right] \otimes \cdots \otimes \left[ \otimes^{r_m} T_p^*(M) \right] \otimes \left[ \otimes^{s_m} T_p(M) \right].$$

All such space with  $\sum_{a=1}^{m} (r_a + s_a) = \text{const}$  are isomorphic but different unless their types coincide. In the complex case,  $\mathbb{K} = \mathbb{C}$ , some of the space  $T_p(M)$  and  $T_p^*(M)$  can be replaced with their complex conjugate spaces  $\overline{T}_p(M)$  and  $\overline{T}_p^*(M)$ , respectively; for instance, a Hermitian metric on a complex manifold M at p is a Hermitian form on  $T_p(M) \times T_p(M)$  but, equivalently it is a bilinear mapping  $T_p(M) \times \overline{T}_p(M) \to \mathbb{C}$  whose transpose is equal to its complex conjugate mapping. In this book only tensor spaces of the type (2.35) will be involved. All results (and definitions) in it can *mutatis mutandis* be transferred to the above general cases which is a simple technical problem.

The elements of (2.35) are referred as tensors, or, if their type must be specified, tensors of type (r, s) (and rank r + s). They are also called *r*-contravariant (*r* times contravariant) and *s*-covariant (*s* times covariant) tensors, or tensors of contravariant degree *r* and covariant degree *s*.

In (2.35) can, of course, be introduced an arbitrary basis  $E_{i_1...i_r}^{j_1...j_s}|_p$ , all indices running from 1 to dim M, but this is done very rarely. The interesting case, which is practically the only one considered in the literature and in our book, is when the basis is induced by some basis  $\{E_i|_p\}$  in  $T_p(M)$  and its dual basis  $\{E^i|_p := (E_i|_p)^*\}$ in  $T_p^*(M)$ :

$$E_{i_1\dots i_r}^{j_1\dots j_s}\big|_p = E^{j_1}\big|_p \otimes \dots \otimes E^{j_s}\big|_p \otimes E_{i_1}\big|_p \otimes \dots \otimes E_{i_r}\big|_p.$$
(2.37)

The components  $K_{j_1...j_s}^{i_1...i_r}(p) \in \mathbb{K}$  of  $K_p \in T_{p_s}^r$  are defined by the expansion

$$K_p = K_{j_1\dots j_s}^{i_1\dots i_r}(p)E^{j_1}\big|_p \otimes \dots \otimes E^{j_s}\big|_p \otimes E_{i_1}\big|_p \otimes \dots \otimes E_{i_r}\big|_p.$$
(2.38)

where the Einstein's summation convention (see p. xi) is used. It is an elementary exercise to verify that a transform  $\{E_i|_p\} \mapsto \{E'_i|_p = A^j_i(p)E_j|_p\}$  of the basis in  $T_p(M)$  induces the changes:

$$E_{i_1\dots i_r}^{j_1\dots j_s}\Big|_p \mapsto E_{i_1\dots i_r}^{\prime j_1\dots j_s}\Big|_p = A_{i_1}^{k_1}(p)\cdots A_{i_r}^{k_r}(p) \left(A^{-1}(p)\right)_{l_1}^{j_1}\cdots \left(A^{-1}(p)\right)_{l_s}^{j_s} E_{k_1\dots k_r}^{l_1\dots l_s}\Big|_p$$
(2.39)

$$K_{j_1\dots j_s}^{i_1\dots i_r}(p) \mapsto K'_{j_1\dots j_s}^{i_1\dots i_r}(p) = \left(A^{-1}(p)\right)_{k_1}^{i_1}\cdots \left(A^{-1}(p)\right)_{k_r}^{i_r}A_{j_1}^{l_1}(p)\cdots A_{j_s}^{l_s}(p)K_{l_1\dots l_s}^{k_1\dots k_r}(p) \quad (2.40)$$

where  $A(p) = [A_i^j(p)]$  is a nondegenerate (constant) matrix.

The tensor algebra  $\mathbf{T}_p(M)$  at  $p \in M$  is the direct sum of all tensor spaces  $T_{p_s}^r$  for  $r, s \geq 0$  with the ordinary tensor multiplication<sup>19</sup> as the algebra's multiplication. This algebra is associative but non-commutative algebra.

A tensor field K of type (r, s) on  $U \subseteq M$  is a mapping  $K: p \mapsto K_p \in T_{p_s}^r(M), p \in U$ . The set  $\mathfrak{T}_s^r(U)$  of tensor fields on U consists of all  $\mathfrak{F}(U)$ -multilinear mappings from the Cartesian product of r copies if the set of covectors on U and s copies of  $\mathfrak{X}(U)$  into  $\mathfrak{F}(U)$ . It is a natural left  $\mathfrak{F}(U)$ -module: for  $K \in \mathfrak{T}_s^r(U)$  and  $f \in \mathfrak{F}(U)$ , we define  $fK: U \to \mathfrak{T}_s^r(U)$  with  $fK: p \mapsto (fK)_p := f(p)K_p$ . It is also a K-linear space: for  $\alpha, \beta \in \mathbb{K}$  and  $K, L \in \mathfrak{T}_s^r(U)$ , the field  $\alpha K + \beta L \in \mathfrak{T}_s^r(U)$  is defined pointwise, viz.  $(\alpha K + \beta L): p \mapsto (\alpha K + \beta L)_p := \alpha K_p + \beta L_p$ . As a linear space  $\mathfrak{T}_s^r(U)$  is infinitely dimensional but as a module its rank is  $(\dim M)^{r+s}$ .

A frame  $\{E_i\}$  on U induces the tensor frame (cf. (2.37))

$$E_{i_1\dots i_r}^{j_1\dots j_s} = E^{j_1} \otimes \dots \otimes E^{j_s} \otimes E_{i_1} \otimes \dots \otimes E_{i_r}.$$
(2.41)

where  $\{E^i = (E_i)^*\}$  is the coframe on U dual to  $\{E_i\}$ . With respect to (2.41) a field  $K \in \mathfrak{T}_s^r(U)$  has the representation

$$K = K_{j_1 \dots j_s}^{i_1 \dots i_r} E^{j_1} \otimes \dots \otimes E^{j_s} \otimes E_{i_1} \otimes \dots \otimes E_{i_r}.$$
 (2.42)

where  $K_{j_1...j_r}^{i_1...i_r} \in \mathfrak{F}(U)$  are called *components* of K with respect to (2.41) (or with respect to  $\{E_i\}$ , or to some local coordinate  $\{x^i\}$  on U if  $E_i = \frac{\partial}{\partial x^i}$ ). Analogously to (2.39) and (2.40), a change  $\{E_i\} \mapsto \{E'_i = A_i^j E_j\}$  of the frame  $\{E_i\}$  with a non-degenerate matrix-valued function  $A = [A_i^j]$  implies

$$E_{i_1\dots i_r}^{j_1\dots j_s} \mapsto E_{i_1\dots i_r}^{j_1\dots j_s} = A_{i_1}^{k_1} \cdots A_{i_r}^{k_r} (A^{-1})_{l_1}^{j_1} \cdots (A^{-1})_{l_s}^{j_s} E_{k_1\dots k_r}^{l_1\dots l_s}$$
(2.43)

<sup>&</sup>lt;sup>19</sup>See, e.g, [11, Chapter 1, § 2] for definition of tensor multiplication. Note that  $\alpha \otimes K_p = K_p \otimes \alpha = \alpha K_p$  for  $\alpha \in \mathbb{K}$  and a tensor  $K_p$  at p

3. Linear connections on manifolds

$$K_{j_1\dots j_s}^{i_1\dots i_r} \mapsto K_{j_1\dots j_s}^{i_1\dots i_r} = \left(A^{-1}\right)_{k_1}^{i_1} \cdots \left(A^{-1}\right)_{k_r}^{i_r} A_{j_1}^{l_1} \cdots A_{j_s}^{l_s} K_{l_1\dots l_s}^{k_1\dots k_r}.$$
(2.44)

A tensor field is of class  $C^r$ , or is a  $C^r$  (tensor) field, if its components are of class  $C^r$  in some, and hence in any, tensor frame. The set of all  $C^k$  tensor fields of type (r, s) on U will be denote by  $\mathfrak{T}_s^{r;k}(U)$ .

**Example 2.9.** An example of  $C^{\infty}$  tensor field on M is the *unit tensor (field)*, called also the *Kronecker tensor (field)*. In any local frame  $\{E_i\}$  it is given by

$$E_i \otimes E^i = \sum_{i=1}^{\dim M} E_i \otimes E^i.$$
(2.45)

and, consequently, its components are the Kronecker (delta-)symbols:

$$\delta_i^j = \begin{cases} 1 & \text{for } i = j \\ 0 & \text{for } i \neq j \end{cases}.$$
 (2.46)

The algebra  $\mathbf{T}(U)$  of the tensor fields on  $U \subseteq M$  is the direct sum of all  $\mathfrak{T}_s^r(U)$ for  $r, s \geq 0$  with the tensor product of tensor fields as algebra's multiplication.<sup>20</sup> This algebra over the field  $\mathbb{K} = \mathbb{R}, \mathbb{C}$  is associative but not commutative. If we restrict ourselves to  $C^r$  tensor fields, the corresponding algebra will be denoted by  $\mathbf{T}^r(U)$ .

At the end of this section, we define the contraction operator(s). Let the integers m, n, r, and s be such that  $1 \le m \le r$  and  $1 \le n \le s$ . The contraction operator  $C_n^m$  (of type (m, n), acting on the *m*th superscript and *n*th subscript) maps a tensor or tensor field K of type (r, s) into respectively a tensor or tensor field  $C_n^m K$  of type (r-1, s-1) such that in any local basis or frame respectively, we have

$$\left(C_n^m K\right)_{i_1\dots i_{r-1}}^{j_1\dots j_{r-1}} := \sum_{k=1}^{\dim M} K_{i_1\dots i_{m-1}k i_{m+1}\dots i_{r-1}}^{j_1\dots j_{r-1}k j_{n+1}\dots j_{r-1}}.$$
(2.47)

This definition is independent of the basis or frame in which the last equation is written; for example,  $C_1^1(E^i \otimes E_i) = \delta_i^i = \dim M$  and  $C_1^1(\omega \otimes X) = \omega(X)$  for vector (field) X and covector (field)  $\omega$ . Generally two contraction operators do not commute. If the numbers m and n are arbitrary (and insignificant) for some problem, we shall speak simply of a contraction operator and denote it by C.

## 3. Linear connections on a differentiable manifold

Below we define the linear connections on a differentiable manifold by means of axiomatic description of the properties of covariant derivative. We give also an idea of a parallel transport and the connected to it concepts of geodesics and exponential mapping.

<sup>&</sup>lt;sup>20</sup>Note that  $f \otimes K = K \otimes f = fK$  for  $f \in \mathfrak{F}(U)$  and a tensor field K on U. (Cf. footnote 19 on the preceding page.)

### **3.1.** Motivation

The comparison of objects defined or given at different points is the main idea and task of the theory of connections.<sup>1</sup> The mathematical structures by means of which this is done are known as *connections*. In more free and general language, by this term is understand any rule for implicit or explicit 'transportation' of objects from one point to another. A typical physical situation where these kind of problems arise is the following. Given an electromagnetic field in a region Uof the space-time M and two (e.g., identical) charged point particles moving in U. The problems are to compare the velocities of the particles, the forces acting on them, and to describe (in an invariant way) the relative motion of one of the particles with respect to the other one. Leaving aside the last problem, the first two ones reduce to the comparison of two vectors, say X and Y, defined at points  $p,q \in M$  and representing the particle's velocities or the forces acting on them. If for the space-time model M is taken  $\mathbb{R}^3$ , as in the classical physics, we can simply form the difference X - Y of  $X, Y \in \mathbb{R}^{3,2}$  But if M is a more general manifold, for instance the one of general relativity, then the difference X - Yis meaningless unless it is explicitly defined. Just for this purpose a connection is needed; in this case it must establish a link between the tangent spaces  $T_p(M) \ni X$ and  $T_q(M) \ni Y$ . If this is done appropriately, we can define X - Y as, e.g.,  $X - \overline{Y} \in T_p(M)$  where  $\overline{Y} \in T_p(M)$  is the vector corresponding to  $Y \in T_q(M)$ with respect to the connection. As one can expect, such a procedure is not unique (if it exists) and when physical problems, like the above one, are investigated, the correct correspondence between the theory's predictions and the Nature is the only criteria that can select the 'right' connection.

Without any doubts, the natural scene where the connections 'live' are the (fibre) bundles (see Section V.3) which is clearly reflected in the modern mathematical literature [10–12, 16] where the connection theory is primary described in terms of fibre bundles and only then is specified on manifolds. But we shall not follow this general approach for the following reasons. The historical order of events is just the opposite and for the main purpose of this book, the description of normal frames, is better if it is followed. This is justified and from pedagogical view-point: beginning with simple concepts, we step by step generalize them, the positive results at a lower level being a motivation for further developments of the theory. Besides, until recently (1992) the normal frames ware known only for symmetric linear connections and the presentation of general results (with a few applications at the moment) at the beginning may cause some phycological problems and push away the reader, especially if he/she is a physicists. At last, the

 $<sup>^{1}</sup>$ In the old literature (see, e.g., [19]) the word 'connexion' instead of 'connection' is used. Nowadays this is rarely done [10, 28].

<sup>&</sup>lt;sup>2</sup>More precisely, the velocities X and Y are elements of the tangent spaces  $T_p(\mathbb{R}^3)$  and  $T_q(\mathbb{R}^3)$  which are naturally identified with  $\mathbb{R}^3$ , where, after the identification is done, the difference X - Y is formed. In fact, this implicit convention is a connection on  $\mathbb{R}^3$ . The situation, when X and Y represent forces, is similar to the one when these vector fields represent velocities.

physical applications of the normal frames concern at present mainly the gravity physics.

In the connection theory there are two basic concepts, a 'connection' and 'parallel transport', the latter being called sometimes 'parallel translation' [22, Chapter VII, § 5]. In the majority of the literature, the connection is taken as a primitive one and the parallel transport is defined on its base. We shall follow this approach in the present and the next two chapters. It is also possible to take as an initial concept the parallel transport and on its ground to define what a connection is (see, e.g., [23]). This point of view is, by our opinion, more general (and difficult) and we utilize it in Chapter IV.

### **3.2.** Basic definitions

The shortest way to come to the concept 'connection' on a manifold M is to try to define what a 'derivative of a  $C^1$  vector field' is. The direct transferring of the definition of the derivative of a  $C^1$  function  $\mathbb{K} \to \mathbb{K}$  is impossible since the difference  $X_p - X_q$ ,  $X \in \mathfrak{X}^1(U)$ ,  $p, q \in U \subseteq M$  is not defined for  $q \neq p$ . If p and qbelong to one and the same coordinate neighborhood U, the quantities  $X_p^i - X_q^i$ and  $\partial X_p^i / \partial x^i$ , with  $\{x^i\}$  being local coordinate system in U, are well defined but are not components of a vector (field). It is well known that these quantities can be 'repaired' (redefined) in such a way that this results in the (equivalent in this context) concepts parallel transport [20] and covariant derivative( $\equiv$  connection here) [21], respectively. This approach is typical for the elementary textbooks on general relativity, as the above-cited. Below we present an equivalent to these methods axiomatical point of view which is almost standard for the modern differential geometry.<sup>3</sup>

**Definition 3.1.** A linear connection (or covariant derivative)  $\nabla$  in (on) a neighborhood  $U \subseteq M$ , M being  $C^1$  manifold, is a mapping assigning to every vector field X on U a mapping  $\nabla_X$ , called covariant derivative along X, of the tensor algebra of  $C^1$  tensor fields on U into the algebra of tensor fields on U, i.e.,  $\nabla \colon X \mapsto \nabla_X$ ,  $X \in \mathfrak{X}(U)$  with  $\nabla_X \colon \mathbf{T}^1(U) \to \mathbf{T}(U)$ , such that:

 $\begin{array}{ll} (\mathbf{i}) \ \nabla_{X+Y} = \nabla_X + \nabla_Y, & X, Y \in \mathfrak{X}(U); \\ (\mathbf{ii}) \ \nabla_{fX} = f \nabla_X, & f \in \mathfrak{F}(U); \\ (\mathbf{iii}) \ \nabla_X(K+L) = \nabla_X K + \nabla_X L, & K, L \in \mathfrak{T}_s^{r;1}(U); \\ (\mathbf{iv}) \ \nabla_X(K \otimes L) = (\nabla_X K) \otimes L + K \otimes (\nabla_X L), & K, L \in \mathbf{T}^1(U); \\ (\mathbf{va}) \ \nabla_X: \mathfrak{F}_1^{0;1}(U) \to \mathfrak{F}_0^{0}(U); \\ (\mathbf{vb}) \ \nabla_X: \mathfrak{T}_1^{0;1}(U) \to \mathfrak{T}_0^{1;1}(U); \\ (\mathbf{vc}) \ \nabla_X: \mathfrak{T}_1^{0;1}(U) \to \mathfrak{T}_{1;1}^{0}(U); \\ (\mathbf{vi}) \ \nabla_X(g) = X(g), & g \in \mathfrak{F}^1(U); \\ (\mathbf{vii}) \ \nabla_X(\omega(Z)) = C_1^1 \big( \nabla_X(\omega \otimes Z) \big), & \omega \in \mathfrak{T}_1^{0;1}(U), Z \in \mathfrak{T}_0^{1;1}(U). \end{array}$ 

 $<sup>^{3}</sup>$ Other equivalent definitions, as well as extended comments on them, can be found in the specialized literature on differential geometry; see, e.g., [9, 11, 12].

*Comments* 3.1. There exist a number of equivalent definitions of a linear connection on a manifold. For this reason, we make below some remarks on the above one:

- (1) Conditions (i) and (ii) specify that  $\nabla$  is  $\mathfrak{F}(U)$ -linear with respect to the vector fields along which (in the direction of which) it acts.
- (2) Condition (iii) means that  $\nabla_X$  is compatible with the additive structure of the tensor algebra on U.
- (3) According to the Leibnitz rule/formula (iv),  $\nabla_X$  is a derivation with respect to the tensor multiplication, i.e., it is a derivation of  $\mathfrak{T}^1(U)$ .
- (4) The conditions (v) mean that the covariant derivative maps  $C^1$  functions in functions and  $C^1$  vector/covector fields in vector/covector fields. Taking this into account, one can derive from (iii) and (iv) that  $\nabla_X$  preserves the types of the tensor fields. That is why often (va)–(vc) are replaced by the demand  $\nabla_X$  to be type-preserving, but this is partially contained in the preceding axioms.
- (5) Conditions (i)–(v) uniquely define  $\nabla_X$  up to its action on  $C^1$  functions, vector and covector fields. This arbitrariness in the definition of  $\nabla_X$  is considerably reduced by (vi) and (vii): Condition (vi) defines  $\nabla_X$  on  $C^1$  functions while from (vii) follows that it is uniquely defined if its action on  $C^1$  vector (or, equivalently, covector) fields is fixed.
- (6) Often condition (vii) is formulated as:  $\nabla_X$  commutes with the contraction operator C, i.e.,  $[\nabla_X, C_s^r] = 0$  for every  $r, s \in \mathbb{N}$ . Such a general demand is not needed as it follows from the presented axioms.

For the conditions under which a manifold admits linear connections, the reader is referred to the books on differential geometry cited in Section 1; in particular, to those problems are devoted [11, Chapter II, § 2] and [22, Chapter VII, § 3].

Below we suppose on a manifold M (of class at least  $C^1$ ) to be given a linear connection  $\nabla$ . Everything of what follows can be specified in an evident way on a neighborhood  $U \subset M$ .

Let  $\{E_i\}$  be a  $C^1$  frame on  $U \subseteq M$ . The (*local*) coefficients  $\Gamma^i_{jk}$  in  $\{E_i\}$  of a linear connection  $\nabla$  on U are defined by the expansion (see Definition 3.1, condition (v b))

$$\nabla_{E_k} E_j =: \Gamma^i_{\ jk} E_i \tag{3.1}$$

where all indices run from 1 to  $\dim M$  and summation on repeated indices is assumed.  $^4$ 

<sup>&</sup>lt;sup>4</sup>The functions  $\Gamma^i_{\ jk}$  are sometimes called Christoffel symbols but we shall preserve this term for a special kind of  $\Gamma^i_{\ jk}$  on a Riemannian manifold (see Section 4 and footnote 9 on page 36). Note the order of the indices j and k in the both sides of (3.1). In some works, like [11, 12], in the right-hand side of (3.1) instead of  $\Gamma^i_{\ jk}$  stands  $\Gamma^i_{\ kj}$ . This results in the change of signs before certain terms in some equations.
## 3. Linear connections on manifolds

If K is a  $C^1$  tensor field of type (r, s) with expansion (2.42) in  $\{E_i\}$ , then it is a simple exercise to verify the formula

$$\nabla_X K = K^{i_1 \dots i_r}_{j_1 \dots j_s; k} X^k E^{j_1} \otimes \dots \otimes E^{j_s} \otimes E_{i_1} \otimes \dots \otimes E_{i_r}.$$
(3.2)

where

$$K_{j_{1}\dots j_{s};k}^{i_{1}\dots i_{r}} := E_{k} \left( K_{j_{1}\dots j_{s}}^{i_{1}\dots i_{r}} \right) + \sum_{a=1}^{r} \Gamma^{i_{a}}{}_{lk} K_{j_{1}\dots j_{s}}^{i_{1}\dots i_{a-1}li_{a+1}\dots i_{r}} - \sum_{b=1}^{s} \Gamma^{l}{}_{j_{b}k} K_{j_{1}\dots j_{b-1}lj_{b+1}\dots j_{s}}^{i_{1}\dots i_{r}}$$

$$(3.3)$$

are the components of the so-called covariant differential  $\nabla K \in \mathfrak{T}_{s+1}^r(U)$  of K defined by (see (2.36))

$$(\nabla K) \big( \omega^{(1)}, \dots, \omega^{(r)}, X_{(1)}, \dots, X_{(s)}; X \big) := \big( \nabla_X K \big) \big( \omega^{(1)}, \dots, \omega^{(r)}, X_{(1)}, \dots, X_{(s)} \big)$$
(3.4)

with  $\omega^{(1)}, \dots, \omega^{(r)} \in \mathfrak{T}_1^0(U)$  and  $X_{(1)}, \dots, X_{(s)} \in \mathfrak{T}_0^1(U)$ .

If we make a change  $\{E_i\} \mapsto \{E'_i = A^j_i E_j\}$  of the frame with a non-degenerate  $C^1$  matrix-valued function  $A = [A^j_i]$ , equation (3.1) implies the transformation<sup>5</sup>

$$\Gamma^{i}_{\ jk} \mapsto \Gamma^{\prime i}_{\ jk} = \left(A^{-1}\right)^{i}_{l} A^{m}_{j} A^{n}_{k} \Gamma^{l}_{\ mn} + \left(A^{-1}\right)^{i}_{l} E^{\prime}_{k} \left(A^{l}_{j}\right)$$
(3.5)

of the coefficients of a linear connection  $\nabla$ . Because of the importance of this result for the present book, we will rewrite it in the case of frames associated with two local  $C^2$  coordinate systems  $\{x^i\}$  and  $\{x'^i\}$  in their common domain (see (2.34)):

$$\Gamma^{i}_{\ jk} \mapsto {\Gamma'}^{i}_{\ jk} = \frac{\partial x'^{i}}{\partial x^{l}} \frac{\partial x^{m}}{\partial x'^{j}} \frac{\partial x^{n}}{\partial x'^{k}} \Gamma^{l}_{\ mn} + \frac{\partial x'^{i}}{\partial x^{l}} \frac{\partial^{2} x^{l}}{\partial x'^{j} \partial x'^{k}}.$$
(3.6)

Note, in this result enter the second partial derivatives of the local coordinates, so it is meaningful if the manifold M is of class not less than  $C^2$  which will be supposed in this and the next chapters.

Consequently, a linear connection  $\nabla$  defines in any frame  $\{E_i\}$  a family of functions  $\{\Gamma^i_{\ jk}\}$  which transform according to (3.5). The opposite statement is also true (see, e.g., [11, Chapter III, Proposition 7.3]): if in every frame  $\{E_i\}$  is given a family of functions  $\{\Gamma^i_{\ jk}\}$  satisfying the transformation law (3.5), then there exists a unique linear connection  $\nabla$  whose local coefficients in  $\{E_i\}$  coincide with  $\{\Gamma^i_{\ jk}\}$ .

<sup>&</sup>lt;sup>5</sup>Notice, here and below we suppose the frames and the matrix-valued function A to be defined on a neighborhood (open set) in M of each point in U, i.e., on a neighborhood of U if U is not a neighborhood. Otherwise expressions like  $E_k(A_i^i)$  may turn to be not (uniquely) defined.

The set of linear connections on M is not  $\mathfrak{F}(M)$ -linear or even  $\mathbb{K}$ -linear space. In fact, if  $\nabla$  and  $\overline{\nabla}$  are linear connections with coefficients  $\Gamma^i_{\ jk}$  and  $\overline{\Gamma}^i_{\ jk}$  respectively and  $f, g \in \mathfrak{F}(M)$ , then we can define  $f \nabla + g \overline{\nabla}$  as a 'linear connection' whose local coefficients are  $f \Gamma^i_{\ jk} + g \overline{\Gamma}^i_{\ jk}$  but, by virtue of (3.5), one verifies easily (see the last term in (3.5)) that these functions form coefficients of a linear connection if and only if f + g = 1.6 So, the only  $\mathbb{K}$ -linear (resp.  $\mathfrak{F}(M)$ -linear) combination under which the set of linear connection is closed is

$$\Gamma^{i}_{jk}, \bar{\Gamma}^{i}_{jk} \mapsto {}^{f}\Gamma^{i}_{jk} := f\Gamma^{i}_{jk} + (1-f)\bar{\Gamma}^{i}_{jk}$$
(3.7)

where  $f \in \mathbb{K}$  (resp.  $f \in \mathfrak{F}(M)$ ). This result is often mentioned when f = 1/2. If we restrict ourselves to coordinate frames, i.e., to ones associated with local coordinates, then, due to (3.6), the quantities  $f\Gamma^i_{\ jk} + g\bar{\Gamma}^i_{\ kj}$  form coefficients of some linear connection iff f + g = 1. Hence, the mapping

$$\Gamma^{i}_{\ jk}, \bar{\Gamma}^{i}_{\ jk} \mapsto \ {}^{f}\bar{\Gamma}^{i}_{\ jk} = f\Gamma^{i}_{\ jk} + (1-f)\bar{\Gamma}^{i}_{\ kj} \tag{3.8}$$

defines a linear connection if only coordinate frames are considered. In particular, from every linear connection  $\nabla$  with local coefficients  $\Gamma^i{}_{jk}$  we can form a linear connection  ${}^{s}\nabla$  with coefficients

$${}^{s}\Gamma^{i}_{\ jk} := \frac{1}{2} \left( \Gamma^{i}_{\ jk} + \Gamma^{i}_{\ kj} \right) =: \Gamma^{i}_{\ (jk)}$$

$$(3.9)$$

in some coordinate frame. This is an example of symmetric linear connection. Generally a linear connection is called *symmetric* if in one (and hence in any – see (3.6)) coordinate frame  $\{\partial/\partial x^i\}$  its local coefficients  $\Gamma^i_{\ jk}$  are symmetric in their subscripts:

$$\Gamma^i_{\ jk} = \Gamma^i_{\ kj}.\tag{3.10}$$

Often these equations are written as

$$\Gamma^{i}_{kj} = \Gamma^{i}_{(kj)} \quad \text{or} \quad \Gamma^{i}_{[kj]} = 0 \tag{3.10'}$$

where symmetrization (resp. antisymmetrization) is understood over the indices included in round (resp. square) brackets (see the list of conventions, page xii). It is important to be emphasized, in the general case it is meaningful to be spoken on the symmetry properties of the coefficients of a linear connection only in *coordinate* frames. The same is valid with respect to the transform (3.8), or its special case (3.9): if (anholonomic) *non-coordinate* frames are involved, the quantities  ${}^{f}\bar{\Gamma}_{\ ik}^{i}$  generally do not define coefficients of some linear connection.

Now we want to define what a  $C^r$  linear connection, called also linear connection of class  $C^r$  or  $C^r$  differentiable linear connection, on a  $C^k$  manifold is.<sup>7</sup>

<sup>&</sup>lt;sup>6</sup>Notice, for g = -f we get the quantities  $f(\Gamma^{i}_{jk} - \overline{\Gamma}^{i}_{jk})$  which are components of a tensor field of type (1, 2).

<sup>&</sup>lt;sup>7</sup>In the literature the problem of defining the differentiability of a linear connection does not arise because usually only the  $C^{\infty}$  case is considered and/or only the class of coordinate frames is employed (see below and, e.g., [12, p. 295]).

Our intention is to call  $\nabla \in C^r$  differentiable if its local coefficients are of class  $C^r$  but this is not quite rigorous. Actually, if its coefficients  $\Gamma^i_{\ ik}$  in a frame  $\{E_i\}$ are  $C^r$  functions, then in another frame  $\{E'_i = A^j_i E_j\}$  the differentiability of the new coefficients  $\Gamma'_{ik}^{i}$  will depend, in conformity with (3.5), on of what class the transformation nondegenerate matrix  $A = [A_i^j]$  is. If  $A_i^j$  are of class  $C^{r'}$ , then that of  ${\Gamma'}^i{}_{jk}$  will be  $C^{\min(r,r'-1)}$ . Consequently the differentiability of the coefficients of a linear connection is essentially frame-dependent concept. For this reason, we adopt the following definition. Given a class of frames connected with linear transformations whose matrices are  $C^{r'}$ ,  $r' \geq 1$  matrix-valued functions, a linear connection is said to be  $C^r$  differentiable with  $r \leq r' - 1$  with respect to it if its local coefficients are of class  $C^r$  with respect to one (and hence relative to any) frame in the given set of frames. Beginning with this point henceforth, when speaking of a  $C^r$  linear connection, we shall suppose, by default if the opposite is not explicitly stated, that the above class of  $C^{r'}$ ,  $r' \ge r+1$ , frames is fixed and consists of all frames associated to (one or all systems of) local coordinates on the  $C^k$ ,  $k \ge r' + 1 \ge r + 2$ , manifold or some its open subset and all frames that can be obtained from them by means of  $C^{r'}$  linear transformations. For example, talking of a  $C^1$  linear connection, we have in mind that it is a linear connection on  $C^3$  manifold and such that its local coefficients in (one or) any frame, which is coordinate or obtainable from a coordinate one by means of  $C^2$  changes, are  $C^1$ functions.<sup>8</sup>

We shall now introduce two basic characteristics of the linear connections which will be mentioned frequently in this book.

The curvature tensor field (or simply curvature), R, and the torsion tensor field (or simply torsion), T, of a linear connection  $\nabla$  are tensor fields of types respectively (1,3) and (1,2), i.e.,

$$R:\mathfrak{T}_1^0(M)\times \left(\times^3\mathfrak{X}(M)\right)\to\mathbb{K}$$
$$T:\mathfrak{T}_1^0(M)\times \left(\times^2\mathfrak{X}(M)\right)\to\mathbb{K},$$

defined as follows. The curvature R for a  $C^1$  linear connection  $\nabla$  on a  $C^3$  manifold M is considered as an  $\mathfrak{F}(M)$ -trilinear mapping  $\times^3 \mathfrak{X}(M) \to \mathfrak{X}(M)$  such that  $(X, Y, Z) \mapsto R(X, Y)Z, X, Y, Z \in \mathfrak{X}(M)$ , where the *curvature operator* (called also simply *curvature*) R(X, Y) (along the pair (X, Y)) is

$$R(X,Y) := \nabla_X \circ \nabla_Y - \nabla_Y \circ \nabla_X - \nabla_{X \circ Y - Y \circ X} \equiv \left[\nabla_X, \nabla_Y\right] - \nabla_{[X,Y]}. \quad (3.11)$$

Analogously, the torsion T of an arbitrary linear connection  $\nabla$  on a manifold M is treated as an  $\mathfrak{F}(M)$ -bilinear mapping  $\mathfrak{X}(M) \times \mathfrak{X}(M) \to \mathfrak{X}(M)$  such that  $(X, Y) \mapsto T(X, Y), X, Y \in \mathfrak{X}(M)$ , where the *torsion operator* (called also simply *torsion*) T(X, Y) (along the pair (X, Y)) is

$$T(X,Y) := \nabla_X Y - \nabla_Y X - [X,Y]_{\bullet}.$$
(3.12)

 $<sup>^8{\</sup>rm For}$  a general and coordinate/frame independent definition of a differentiability class of a connection, see Section V.3.2.

*Remark* 3.1. Nevertheless that the torsion is defined for arbitrary linear connection on every manifold, this concept is intrinsically linked to the module of vector fields on the manifold, or, equivalently, to the tangent bundle to it, and can not be generalized on more general bundles. On the other hand, the concept of curvature survives a generalizations on vector bundles endowed with linear connections (or, more generally, linear transports along paths), as well as for connections on differentiable bundles.

In a local frame  $\{E_i\}$ , the components of the curvature R,  $R^i_{\ jkl}$ , and torsion T,  $T^i_{\ jk}$ , are defined by the expansions

$$R(E_k, E_l)E_j =: R^i_{\ ikl}E_i, \qquad T(E_j, E_k) =: T^i_{\ ik}E_i$$

and their explicit form is:

$$R^{i}_{\ jkl} = -2\Gamma^{i}_{\ j[k,l]} - 2\Gamma^{m}_{\ j[k]}\Gamma^{i}_{\ |m|l]} - \Gamma^{i}_{\ jm}C^{m}_{kl}$$
  
$$= -\Gamma^{i}_{\ ikl} + \Gamma^{i}_{\ ilk} - \Gamma^{m}_{\ ik}\Gamma^{i}_{\ ml} + \Gamma^{m}_{\ il}\Gamma^{i}_{\ mk} - \Gamma^{i}_{\ im}C^{m}_{kl}$$
(3.13)

$$T^{i}_{jk} = -2\Gamma^{i}_{[jk]} - C^{i}_{jk} = -\Gamma^{i}_{jk} + \Gamma^{i}_{kj} - C^{i}_{jk}.$$
(3.14)

Here the summation and (anti)symmetrization conventions are used (see the list of conventions),  $f_{,i} := E_i(f)$  for  $f \in \mathfrak{F}^1(M)$  and the (structure) functions  $C^i_{jk} \in \mathfrak{F}(M)$  define the commutators of the basis vector fields,

$$[E_j, E_k]_{-} \coloneqq C^i_{jk} E_i. \tag{3.15}$$

Notice, in a coordinate or holonomic frame  $C_{jk}^i \equiv 0$  (see Section 8 for some details).

A linear connection with zero curvature (resp. torsion) on  $\subseteq M$  is called *flat*, or *curvature free*, or *integrable* (resp. *torsionless* or *torsion free*) on U. If U = M it is call simply flat (resp. torsionless).

Comparing (3.10) with (3.14), we see that a linear connection is symmetric on  $U \subseteq M$  iff it is torsion free on U. So, the vanishment of the torsion is the *invariant* way to describe what a symmetric linear connection is.

The properties and geometric interpretation of curvature and torsion, as well as that of linear connections as a whole, are well known and investigated at length in the books on differential geometry pointed in Section 1.

# **3.3.** Parallel transport

The concept of a 'parallel transport', first introduced on surfaces by Levi-Civita in 1917 [29], will be employed in the present and subsequent chapters mainly for many purposes, e.g., for proving and formulating some results. Besides this, we include here a small subsection devoted to it also for the following reasons:

(i) The parallel transport is the practical realization of the ideas underlying the connection theory (see subsection 3.1);

## 3. Linear connections on manifolds

- (ii) Up to this point we have introduce the full 'machinery' required for describing what a parallel transport associated to (defined by) a linear connection is;
- (iii) This concept can be defined independently of the connection theory by a system of axioms and on its base the (linear) connection can (equivalently) be introduced [17, 23, 30–33] (see also Section IV.11, Definitions IV.11.2 and IV.11.3);
- (iv) As we shall see in Chapter IV, the 'transport theory' is extremely suitable for exploring normal frames in vector bundles, in particular for arbitrary linear connections in them.

Let  $\gamma: J \to M$ , J being  $\mathbb{R}$ -interval, be a  $C^1$  path in a manifold M endowed with a linear connection  $\nabla$ . Let  $V \in \mathfrak{X}(U)$  be arbitrary vector field on a neighborhood  $U \supset \gamma(J)$  and such that  $V|_{\gamma(J)} = \dot{\gamma}, \dot{\gamma}$  being the vector field tangent to  $\gamma$ (see (2.3)), i.e.,  $V_{\gamma(s)} = \dot{\gamma}(s), s \in J$ .<sup>9</sup> Consider the restriction of the action of  $\nabla_V$ on  $\gamma(J)$ :

$$\nabla_{\dot{\gamma}}(K) := (\nabla_V K)|_{\gamma(J)}, \qquad (\nabla_{\dot{\gamma}} K)|_{\gamma(s)} := \frac{DK}{\mathrm{d}s} := (\nabla_V K)|_{\gamma(s)}. \tag{3.16}$$

The operator  $\nabla_{\dot{\gamma}}$  is called *covariant derivative along the path*  $\gamma$ . In an arbitrary frame on U, due to (2.3), (3.2), and (3.3), we see that, for any tensor field  $K \in \mathfrak{T}_{r'}^{r;1}(U)$ , the tensor field  $\nabla_{\dot{\gamma}}K$  given by  $\nabla_{\dot{\gamma}}K \colon \gamma(s) \mapsto (\nabla_V K)_{\gamma(s)}$  is independent of U and of the values  $V_p$  and  $K_p$  for  $p \in U \setminus \gamma(J)$ :

$$(\nabla_{\dot{\gamma}}K)_{\gamma(s)} = \left\{ \frac{\mathrm{d}}{\mathrm{d}s} \left( K^{i_{1}\dots i_{r'}}_{j_{1}\dots j_{r}} \left(\gamma(s)\right) \right) + \sum_{a=1}^{r} \Gamma^{i_{a}}{}_{lk} (\gamma(s)) K^{i_{1}\dots i_{a-1}li_{a+1}\dots i_{r}}_{j_{1}\dots j_{s}} (\gamma(s)) \dot{\gamma}^{k}(s) - \sum_{b=1}^{r'} \Gamma^{l}{}_{j_{b}k} (\gamma(s)) K^{i_{1}\dots i_{r}}_{j_{1}\dots j_{b-1}lj_{b+1}\dots j_{s}} (\gamma(s)) \dot{\gamma}^{k}(s) \right\} \times \left( E^{j_{1}} \otimes \dots \otimes E^{j_{r'}} \otimes E_{i_{1}} \otimes \dots \otimes E_{i_{r}} \right) |_{\gamma(s)}.$$

$$(3.17)$$

**Definition 3.2.** The parallel transport (along paths or curves) assigned to a  $C^0$ linear connection  $\nabla$  is a mapping  $\mathsf{P} \colon \beta \mapsto \mathsf{P}^\beta$  which, for every  $C^1$  path  $\beta \colon [a,b] \to M^{10}, a, b \in \mathbb{R}, a \leq b$ , puts into correspondence a mapping  $\mathsf{P}^\beta \colon T_{\beta(a)} \to T_{\beta(b)}$ from the tensor algebra at  $\beta(a)$  into the one at  $\beta(b)$  such that if  $K_0 \in T_{\beta(a)}$ , then

<sup>&</sup>lt;sup>9</sup>Here we implicitly suppose  $\gamma$  to be injective as otherwise  $\dot{\gamma}: \gamma(s) \mapsto \dot{\gamma}(s)$  may not be a single-valued mapping (see footnote 11 on page 11). However (see Remark 3.3 on the following page), taking (3.17) as a definition for  $\nabla_{\dot{\gamma}} K$ , the next considerations hold for arbitrary, injective or non-injective, paths.

<sup>&</sup>lt;sup>10</sup>With  $[a, b] = \{x \in \mathbb{R} : a \le x \le b\}$  we denote a real closed interval with end points  $a, b \in \mathbb{R}$  such that  $a \le b$ . Similarly, by  $(a, b] = \{x \in \mathbb{R} : a < x \le b\}$  and  $[a, b) = \{x \in \mathbb{R} : a \le x < b\}$  we denote real intervals closed from right and left, respectively, and opened from left and right, respectively, and with end points  $a, b \in \mathbb{R}$  such that a < b. The notation  $(a, b) = \{x \in \mathbb{R} : a < x < b\}$  will stand for an open from both end points  $a, b \in \mathbb{R}$ , with a < b, real interval.

 $\mathsf{P}^{\beta}(K_0) := K_{\beta(b)}$ , where K is a tensor field on  $\beta([a, b])$  of the same type as  $K_0$  and is defined as the unique solution of the initial-value problem

$$\nabla_{\dot{\beta}}K = 0, \qquad K_{\beta(a)} = K_0.$$
 (3.18)

Remark 3.2. According to (3.17), the problem (3.18) reduces in a local frame to a first-order system of ordinary differential equations with respect to the local components of K with initial condition at s = a. This initial-value problem has a unique solution along  $\beta$  by virtue of the existence and uniqueness theorems for such systems [34].

Remark 3.3. By (3.17), the solution K of (3.18) depends, generally, explicitly on  $\beta$ , i.e., the value  $K_{\beta(s)}$  for  $s \in [a, b]$  generally depends on s and  $\beta$  separately, not only on the combination  $\beta(s)$ ; in particular, at the points of self-intersection of  $\beta$ , if any, K may be multiple-valued (as a tensor field). The right and natural interpretation of K is as a lifting of  $\beta$  from M to the bundle space of the tensor bundle of the same type as  $K_0$ . This fact will become important in Chapter IV where it will be treated appropriately. If  $\beta$  is injective, i.e., without self-intersections, K is ordinary (single-valued) tensor field over  $\beta([a, b])$ . So, to be more precise, one should write  $K_{\beta}$  and  $K_{\beta}(s)$  for K and  $K(\beta(s))$  respectively, and, according to (3.17), the symbol  $(\nabla_{\dot{\beta}}K)(s)$ , denoting the covariant derivative of (a lifting of paths) K along  $\beta$  at  $s \in [a, b]$ , should be defined by

$$(\nabla_{\dot{\beta}}K)(s) = \left\{ \frac{\mathrm{d}}{\mathrm{d}s} \left( K_{\beta j_{1}\dots j_{r}}^{i_{1}\dots i_{r'}}(s) \right) + \sum_{a=1}^{r} \Gamma^{i_{a}}{}_{lk} (\beta(s)) K_{\beta j_{1}\dots j_{s}}^{i_{1}\dots i_{a-1}li_{a+1}\dots i_{r}}(s) \dot{\beta}^{k}(s) - \sum_{b=1}^{r'} \Gamma^{l}{}_{j_{b}k} (\beta(s)) K_{\beta j_{1}\dots j_{b-1}lj_{b+1}\dots j_{s}}(s) \dot{\beta}^{k}(s) \right\} \times \left( E^{j_{1}} \otimes \dots \otimes E^{j_{r'}} \otimes E_{i_{1}} \otimes \dots \otimes E_{i_{r}} \right) |_{\beta(s)}.$$

$$(3.19)$$

The reader is referred to Chapter IV, in particular to Sections IV.11, IV.14, and IV.3 for further details concerning the definition of a parallel transport and its links with the concept of lifting (of paths).

*Remark* 3.4. Since in the present book, with an exception of Section IV.14 and Chapter V, only linear connection on manifolds and respectively parallel transports assigned to such connections will be employed, we shall implicitly understood, when talking about parallel transports, that they are with respect to (assigned to) some linear connection on a manifold, if this is not explicitly stated.

The properties of the parallel transport are well-known and can be found in practically every book on differential geometry covering the connection theory, e.g in [11,22]. We will mention only some of them: (i) The transport along  $\beta$  depends on the set  $\beta([a.b])$ , not on its particular parameterization; (ii) It depends only on

 $\beta(a)$  and  $\beta(b)$  iff  $\nabla$  is flat; (iii) To product of paths corresponds the composition of the corresponding transports; (iv) The mapping  $\mathsf{P}^{\beta}$  is a linear isomorphism.

Thus the parallel transport generated by a linear connection realizes the general ideas presented in Subsection 3.1. In particular, by its help we can give sense to operations with tensors defined at different points, say  $p, q \in M, p \neq q$ , by parallelly transporting, e.g., the tensors at p to q along some path connecting p and q.<sup>11</sup> For instance, we can compare two vectors  $X \in T_p(M)$  and  $Y \in T_q(M)$  by forming, e.g., the difference  $Y - \mathsf{P}^\beta(X)$  for some path  $\beta: [a, b] \to M$  such that  $\beta(a) = p$  and  $\beta(b) = q$  (see footnote 11).

*Remark* 3.5. Above we implicitly supposed the existence of a  $C^1$  path  $\beta$  in M connecting the (arbitrary) points p and q. This is true if M is simply connected. If M is multiply connected, then always exist paths joining every  $p, q \in M$  but such a path can be chosen in a smooth way, i.e., of class  $C^1$  or higher, iff p and q belong to a simply connected region of M. Since, by definition, the parallel transport is defined only along  $C^1$  paths, when talking of a parallel transport along a path joining two points in M, we always presuppose that they are situated in a simply connected subset of M. Moreover, when saying that the parallel transport between two points p and q is path-independent, we have in mind that this is with respect to reconcilable paths, i.e., ones forming (belonging to) a smooth,  $C^1$ , homotopy  $\eta: [a,b] \times W \to M$  where W is inessential for us non-empty set,  $^{12} \eta(a,W) = p$ , and  $\eta(b, W) = q$ . Said more freely, the paths connecting p and q are suppose such that any closed loop formed from every pair of them can smoothly be deformed to a point in M. This will be presupposed and further in this book. Otherwise the parallel transport may turn to be path-dependent even in the flat case: such situation may happen if M is multiply connected and there is a smooth closed path connection p and q which can not be smoothly contracted to a single point in M.

*Remark* 3.6. The concept of a parallel transport has a natural generalization in arbitrary vector bundles, called linear transport along paths. For details, see Sections IV.3 and IV.11, in particular Definition IV.11.1 and Proposition IV.11.1. For some links between (parallel) transports (along paths) and connections on differentiable bundles, see Section V.8.

**Definition 3.3.** A tensor field  $K \in \mathbf{T}^1(U)$ ,  $U \subseteq M$  is called *parallel along a*  $C^1$  path  $\gamma: J \to U$  with respect to a linear connection  $\nabla$  if  $\nabla_{\dot{\gamma}} K = 0$ . The field K is called parallel (with respect to  $\nabla$ ) on U if it is parallel along every path  $\gamma: J \to U$ .

A parallel (along  $\gamma$ ) field is uniquely defined by fixing its value  $K_0$  at a point  $p = \gamma(c)$  for some  $c \in J$ ; in fact we have  $K_{\gamma(s)} = \mathsf{P}^{\gamma|[c,s]}(K_0), s \in J$ , with  $\gamma|[c,s]$  being the restriction of  $\gamma$  to  $[c,s] \subseteq J$ , for  $c \geq s$  and  $K_{\gamma(s)} = (\mathsf{P}^{\gamma|[s,c]})^{-1}(K_0)$  for  $c \leq s$ . The field K is parallel along  $\gamma$  iff  $K_{\gamma(t)} = \mathsf{P}^{\gamma|[t,s]}(K_{\gamma(s)})$  for  $t \geq s$  and

<sup>&</sup>lt;sup>11</sup>This operation is path dependent unless the connection is flat.

<sup>&</sup>lt;sup>12</sup>The set W is a manifold whose *real* dimension is  $(\dim M - 1)$  if  $\mathbb{K} = \mathbb{R}$  and  $(2 \dim_{\mathbb{C}} M - 1)$  for  $\mathbb{K} = \mathbb{C}$ .

$$\begin{split} K_{\gamma(t)} &= \left(\mathsf{P}^{\gamma|[s,t]}\right)^{-1}(K_{\gamma(s)}) \text{ for } t \leq s \text{ for all } s,t \in J. \text{ A tensor field } K \text{ is parallel on } \\ U \subseteq M \text{ if and only if } (\nabla_X K)|_U &= 0 \text{ for all } X \in \mathfrak{X}(U). \end{split}$$

**Definition 3.4.** A frame  $\{E_i\}$  on  $U \subseteq M$  (resp. along a path  $\gamma: J \to M$ ) is called *parallel* (on U, resp. along  $\gamma$ ) if its basic vector fields  $E_1, \ldots, E_{\dim M}$  are parallel on U(resp. along  $\gamma$ ).

Frames parallel on  $(\dim M)$ -dimensional submanifolds U, i.e., on neighborhoods, exist iff  $\nabla$  is flat on U.

An almost standard way for construction of a (parallel) frame on some pathconnected flat set U (with respect to a linear connection) is to fix a basis in  $T_p(M)$  for some arbitrarily chosen point  $p \in U$  and then to transport parallelly its vectors, i.e., the whole basis, to every  $q \in U$  along some path connecting p and q. A frame parallel on U is analogous to a Cartesian coordinate system on  $\mathbb{R}^n$  (see Chapter IV).

# 3.4. Geodesics and exponential mapping

The geodesic paths, called in different contexts also geodesic curves, geodesic lines, or simply geodesics, in a manifold with connection are in many aspects analogues of the straight lines in  $\mathbb{R}^n$ . Historically the introduction of normal frames/coordinates is primary related to them.

**Definition 3.5.** A geodesic or a geodesic path in a  $C^2$  manifold endowed with  $C^0$  linear connection  $\nabla$  is a  $C^1$  path  $\gamma: J \to M$  whose tangent vector  $\dot{\gamma}$  is parallel along  $\gamma$ , i.e.,

$$\dot{\gamma}(s_2) = \begin{cases} \mathsf{P}^{\gamma | [s_1, s_2]} \dot{\gamma}(s_1) & \text{for } s_1, s_2 \in J, \quad s_1 \le s_2\\ \left(\mathsf{P}^{\gamma | [s_2, s_1]}\right)^{-1} \dot{\gamma}(s_1) & \text{for } s_1, s_2 \in J, \quad s_2 \le s_1 \end{cases}$$
(3.20)

where  $\mathsf{P}$  is the assigned to  $\nabla$  parallel transport, or, equivalently (however, see below Remark 3.8),

$$\nabla_{\dot{\gamma}}\dot{\gamma} = 0. \tag{3.21}$$

This equation, as well as its equivalent local forms (3.22) and (3.23) presented below, are known as the *geodesic equation* or *equation of the geodesics*.

Remark 3.7. In some works, like [18], the paths satisfying (3.21) are called autoparallel, the term 'geodesic' being reserved for paths satisfying certain Euler-Lagrange equations (derived from a variational principle) [12, 18]. This is more typical for the 'physically oriented' literature [35] which is connected with the development of theories based on manifolds endowed with (more or less independent) linear connection and Riemannian metric [36, 37]. The more widely accepted modern mathematical terminology is to call the former paths geodesics and the latter ones extremals [10, 11, 19]. That last convention is followed in the present book. We reserve the term "autoparallels" or "autoparallel paths" for a certain generalization of the geodesics to which is devoted Section IV.15.

## 3. Linear connections on manifolds

Remark 3.8. If the  $C^1$  path  $\gamma: J \to M$  is not injective, the mapping  $\dot{\gamma}: s \mapsto \dot{\gamma}(s) \in T_{\gamma(s)}(M)$  cannot be considered as a vector field over  $\gamma(J)$  assigning to  $\gamma(s)$  the value  $\dot{\gamma}(s)$  as the mapping  $\gamma(s) \mapsto \dot{\gamma}(s)$  is generally multiple-valued at the points of self-intersection of  $\gamma$ .<sup>13</sup> For this reason, the symbol  $\nabla_{\dot{\gamma}}\dot{\gamma}$  is not well-defined if  $\gamma$  is not injective. In that case, as well as in the general one, it should be defined as a path in T(M) whose local components are *defined* by the left-hand side of equation (3.22) below.

In the domain U of some local frame  $\{E_i\}$ , the geodesic equation (3.21) reads

$$\frac{\mathrm{d}\dot{\gamma}^{i}}{\mathrm{d}s} + \Gamma^{i}{}_{jk}\dot{\gamma}^{j}\dot{\gamma}^{k} = 0, \qquad s \in J$$
(3.22)

where (3.17) was used. If some local coordinate system  $\{x^i\}$  is employed and  $E_i = \partial/\partial x^i$ , then  $\dot{\gamma}^i = d\gamma^i/ds$  with  $\gamma^i = x^i \circ \gamma$ , so (3.22) reduces to

$$\frac{\mathrm{d}^2\gamma^i(s)}{\mathrm{d}s^2} + \Gamma^i_{\ jk}(\gamma(s))\frac{\mathrm{d}\gamma^j(s)}{\mathrm{d}s}\frac{\mathrm{d}\gamma^k(s)}{\mathrm{d}s} = 0 \qquad s \in J.$$
(3.23)

From here a number of immediate observations can be derived:

- (i) If  $\gamma: J \to M$  is a geodesic, then so is the path  $\gamma \circ \tau$ , with  $\tau: J' \to J$  and  $\tau: t \mapsto at + b, a, b \in \mathbb{R}, a \neq 0, t \in J'$ , i.e., a geodesic is mapped into geodesic under nondegenerate affine change of its parameter.<sup>14</sup>
- (ii) If  $\gamma: J \to M$  is a geodesic, its restriction  $\gamma|J'$  to any subinterval  $J' \subset J$  is also geodesic.
- (iii) If the connection is of class  $C^k$ ,  $k \ge 0$  (on  $C^{k+2}$  manifold), i.e., if  $\Gamma^i_{jk}$  are of class  $C^k$ , any geodesic, which is a  $C^1$  path by definition, is automatically a  $C^{k+2}$  path; in particular the geodesics of  $C^{\infty}$  linear connection on  $C^{\infty}$  manifold are  $C^{\infty}$  paths.
- (iv) Since (3.23) is a second-order system of ordinary differential equations, from the theorems of existence and uniqueness of the solutions of such systems [34] follows that for every point  $x \in M$  and every vector  $X \in T_x(M)$  there exists a geodesic  $\gamma: J \to M$  such that

$$\gamma(s_0) = x, \qquad \dot{\gamma}(s_0) = X \tag{3.24}$$

for a fixed  $s_0 \in J$  and some interval J. Besides, there is a subinterval  $J_0 \subseteq J$ such that  $J_0 \ni s_0$  and on  $J_0$  all such geodesics coincide. Hence locally, on  $J_0 \ni s_0$ , there is a unique geodesic with initial conditions (3.24). Generally  $J_0 \subset \mathbb{R}$ . If for every geodesic  $J_0 = \mathbb{R}$ , the connection/manifold is called (geodesically) complete.

<sup>&</sup>lt;sup>13</sup>In fact,  $\dot{\gamma}$  is a lifting of  $\gamma$  in the tangent bundle over M, not a vector field on  $\gamma(J)$ ; it can be considered as a vector field iff  $\gamma$  is injective. For details, see Section IV.2.

<sup>&</sup>lt;sup>14</sup>If  $\tau(J') = J$ , the sets  $\gamma(J)$  and  $(\gamma \circ \tau)(J')$  coincide, i.e., they represent one and the same (geometric, unparametrized geometric) curve  $\sigma = \gamma(J)$ ; see [11, Chapter III, § 6] and footnote 7 on page 9.

(v) The geodesics depend only on the 'symmetric' part of the connection, i.e., on the symmetric part  $\Gamma^i{}_{(jk)}$  of its coefficients  $\Gamma^i{}_{jk}$  in any frame, coordinate or not: in particular, since  $\dot{\gamma}^j \dot{\gamma}^k = \dot{\gamma}^k \dot{\gamma}^j$ , we have the equivalent to (3.23) equation

$$\frac{\mathrm{d}^2\gamma^i(s)}{\mathrm{d}s^2} + \Gamma^i_{(jk)}(\gamma(s))\frac{\mathrm{d}\gamma^j(s)}{\mathrm{d}s}\frac{\mathrm{d}\gamma^k(s)}{\mathrm{d}s} = 0 \qquad s \in J.$$
(3.23')

So, if two (or more) linear connections on M generate one and the same symmetric connection via (3.9), their geodesics coincide.

(vi) A geodesic is either regular or degenerate path: if  $\dot{\gamma}(s_0) = 0$  for some  $s_0 \in J$ , then  $\dot{\gamma}(s) \equiv 0$  for every  $s \in J$  as the parallel transport is an isomorphism.

Now we are going to formulate a result which is often used for the proof of existence of normal frames at a single point.

A geodesic is called *maximal* if it is not a restriction of other geodesic, i.e.,  $\gamma: J \to M$  is maximal geodesic if there does not exist a geodesic  $\bar{\gamma}: \bar{J} \to M$  such that  $J \subset \bar{J}$  and  $\gamma = \bar{\gamma}|J$ . For every point  $x \in M$ , every vector  $X \in T_x(M)$ , and any number  $s_0 \in \mathbb{R}$  there exists a unique maximal geodesic  $\gamma_{x,X}^{s_0}: J_m \to M$  defined on some interval  $J_m \ni s_0$  and such that<sup>15</sup>

$$\gamma_{x,X}^{s_0}(s_0) = x, \quad \dot{\gamma}_{x,X}^{s_0}(s_0) = X.$$
(3.25)

Using the uniqueness of the maximal geodesics, one easily verifies that

$$\gamma_{x,aX}^{s_0}(s) = \gamma_{x,X}^{s_0}(a(s-s_0)+s_0) \qquad a \in \mathbb{R}.$$
(3.26)

In what follows, we put  $s_0 = 0$  (see Remark 3.9 on the next page) and write  $\gamma_{x,X}$  for  $\gamma_{x,X}^0$ . For this choice, equation (3.26) reads

$$\gamma_{x,aX}(s) = \gamma_{x,X}(as) \qquad s \in J^m \ni 0 \in \mathbb{R}, \quad a \in \mathbb{R}.$$
(3.27)

Let  $V_x \subseteq T_x(M)$  be such that  $\gamma_{x,X}$ , with X in  $V_x$ , is defined on interval  $J_m$  containing the point s = 1,  $J_m \ni 1 \in \mathbb{R}$ . The set  $V_x$  contains, evidently, the zero vector; besides, by (3.27) it contains also non-zero vectors. We define a mapping  $\exp_x : V_x \to M$ , called *exponential mapping at the point x*, by<sup>16</sup>

$$\exp_x X := \gamma_{x,X}(1), \qquad X \in V_x. \tag{3.28}$$

Due to (3.27), it is fulfilled

$$\exp_x(sX) = \gamma_{x,X}(s), \qquad s \in J_m \ni 1, \quad X \in V_x.$$
(3.29)

The following important result is valid.

<sup>&</sup>lt;sup>15</sup>For the proof, see [38, p. 380]. Generally  $\gamma_{x,X}^{s_1}$  and  $\gamma_{x,X}^{s_2}$  are different unless  $s_1 = s_2$ .

<sup>&</sup>lt;sup>16</sup>The exponential mapping, exp, itself is defined via exp:  $x \mapsto \exp_x, x \in M$ , or by exp:  $X \mapsto \exp_x X$  for  $X \in V_x \subseteq T_x(M)$ .

**Proposition 3.1.** For every point  $x \in M$  there exists a neighborhood  $V_x^0$  of the zero vector in  $T_x(M)$  such that:

- (i) It is star-shaped (star-like), i.e., if  $X \in V_x^0$ , then  $aX \in V_x^0$  for  $0 \le a \le 1$ ;
- (ii) The exponential mapping  $\exp_x$  at x is defined on  $V_x^0 \subseteq V_x \subseteq T_x(M)$ ;
- (iii) There exists a neighborhood V(x) of x in M such that  $\exp_x : V_x^0 \to V(x)$  is diffeomorphism.

*Proof.* See [38, pp. 381–384], [11, Chapter III, § 8] or [39, Section 1.6] □

Remark 3.9. The choice  $s_0 = 0$  made above is not necessary, it only saves some writing, simplifies the formulae, and eliminates from the theory an arbitrary number (which is insignificant at present). For  $s_0 \neq 0$  the above construction goes like this: Let  $V_{s_0,x} \subseteq T_x(M)$  be such that  $\gamma_{x,X}^{s_0}$  with  $X \in V_{s_0,x}$  is defined on  $J_m \subseteq \mathbb{R}$  containing the point  $s = s_0 + 1$ ,  $J_m \ni (s_0 + 1) \in \mathbb{R}$ . The exponential mapping at x (depending on  $s_0$ ) is  $\exp_{s_0,x} : V_{s_0,x} \to M$  with  $\exp_{s_0,x} X := \gamma_{x,X}^{s_0}(s_0 + 1)$ ,  $X \in V_{s_0,x}$ . So,  $\exp_{s_0,x}(sX) = \gamma_{x,X}^{s_0}(s_0 + s)$ ,  $(s_0 + s) \in J_m$ . Proposition 3.3 remains true if we replace in it  $V_x^0$ ,  $V_x$ ,  $\exp_x$ , and V(x) respectively by  $V_{s_0,x}^0$ ,  $V_{s_0,x}$ ,  $\exp_{s_0,x}$ ,

**Definition 3.6.** A neighborhood  $V_x^0$  of the zero vector in  $T_x(M)$  having the properties (i)–(iii) described in Proposition 3.1 is called normal (neighborhood of  $0 \in T_x(M)$ ). A neighborhood V(x) of  $x \in M$  is called normal if it is the image of a normal neighborhood of the zero vector in  $T_x(M)$  under the exponential mapping (at x),  $V(x) = \exp_x(V_x^0)$ .

Therefore the essence of Proposition 3.3 is that every point  $x \in M$  admits normal neighborhoods.

The normal neighborhoods possess a number of remarkable properties described, for instance, in [11,38]. They have a straightforward relation to the existence of normal frames that will be revealed in Section 6 and in Chapter II.

# 4. Riemannian manifolds

Freely said, a Riemannian manifold is a real differentiable manifold on each tangent space, considered as a vector space, of which a scalar product is given. When metrics (scalar or inner products) are concerned, one should clearly distinguish the real,  $\mathbb{K} = \mathbb{R}$ , and complex,  $\mathbb{K} = \mathbb{C}$ , cases. In the former case the scaler product is understand as Riemannian metric [10,11], while in the latter case it is assumed to be a Hermitian metric [10,26,40].<sup>1</sup> That is why, usually a real manifold with a

<sup>&</sup>lt;sup>1</sup>Formally one can consider a 'Riemannian metric' on a complex manifold as a non-degenerate bilinear form on it. Such a form is complex-valued and does not define a metric or inner product in the accepted sense [16, p. 2], [10, Chapter I, Section 11]; in particular such a 'metric' does not define a distance function on the manifold and, correspondingly, there is no associated with it 'metric topology', etc. Regardless of this, if in what follows one considers such forms, all remains true *mutatis mutandis*.

Riemannian metric is called Riemannian manifold [11] while a complex manifold with a Hermitian metric is called Hermitian manifold [26].

In this book only Riemannian metrics will be considered as the Hermitian once require somewhat different treatment which is not primary related to its main subject.

**Definition 4.1.** A Riemannian metric g on a subset  $U \subseteq M$  of a real manifold M is a field of symmetric non-degenerate bilinear forms on the tangent spaces at the points in U, i.e., g is a symmetric non-degenerate tensor field in  $\mathfrak{F}_2^0(U)$ :<sup>2</sup>

$$g: x \mapsto g_x, \quad x \in U, \qquad g_x: T_x(M) \times T_x(M) \to \mathbb{R},$$
 (4.1a)

$$g_x(X_x, Y_x + Z_x) = g_x(X_x, Y_x) + g_x(X_x, Z_x) \qquad X_x, Y_x, Z_x \in T_x(M), \quad (4.1b)$$

$$g_x(X_x, aY_x) = ag_x(X_x, Y_x) \qquad a \in \mathbb{R}, \quad X_x, Y_x \in T_x(M),$$
(4.1c)

$$g_x(X_x, Y_x) = g_x(Y_x, X_x) \qquad X_x, Y_x \in T_x(M),$$
(4.1d)

$$g_x(X_x, Y_x) = 0$$
 for all  $X_x \in T_x(M)$  and some  $Y_x \in T_x(M) \iff Y_x = 0$ . (4.1e)

A real manifold endowed with a Riemannian metric is called *Riemannian manifold*. Below we suppose the existence of a Riemannian metric on the real manifold M.<sup>3</sup>

**Example 4.1.** The *n*-dimensional Euclidean space  $\mathbb{E}^n$ ,  $n \in \mathbb{N}$ , is an almost trivial example of a Riemannian manifold. It is defined as  $\mathbb{R}^n$  endowed with *Euclidean* metric *e* such that  $e(X,Y) := \sum_{i=1}^n X^i Y^i$  for  $X = (X^1, \ldots, X^n) \in \mathbb{R}^n$  and  $Y = (Y^1, \ldots, Y^n) \in \mathbb{R}^n$  (see footnote 2 on page 20 and Example 7.1 below).

Let g denotes a Riemannian metric on a real manifold M. For any  $U \subseteq M$ , it induces a Riemannian metric on the set  $\mathfrak{X}(U)$  of vector fields on U: we define  $g: \mathfrak{X}(U) \times \mathfrak{X}(U) \to \mathfrak{F}(U)$  by  $g: (X,Y) \mapsto g(X,Y) \in \mathfrak{F}(U), X,Y \in \mathfrak{X}(U)$  with  $g(X,Y): x \mapsto g_x(X_x,Y_x), x \in U$ .

If  $X, Y \in T_x(M)$  (resp.  $X, Y \in \mathfrak{X}(U)$ ), the number  $g_x(X,Y) \in \mathbb{R}$  (resp. the function  $g(X,Y) \in \mathfrak{F}(U)$ ) is called *scalar product* of the vectors (resp. vector fields) X and Y in  $T_x(M)$  (resp. in  $\mathfrak{X}(U)$ ). In [11, Chapter IV, § 1] is shown how the scalar product in  $T_x(M)$  (resp. in  $\mathfrak{X}(U)$ ) can be extended to a scalar product in  $T_{xs}^r$  (resp. in  $\mathfrak{T}_s^r$ ) for every  $r, s \geq 0$  with  $r + s \geq 1$  and that it induces an isomorphism  $T_{xs}^r \to T_{xs \mp 1}^{r \pm 1}$  for every  $r, s \geq 0$  such that  $r \pm 1, s \mp 1 \geq 0$  but these, otherwise useful, results will not find applications in our investigation.

A Riemannian metric g on U is called  $C^r$  metric ( $C^r$  differentiable, (differentiable) of class  $C^r$ ) if it is a  $C^r$  tensor field on U, i.e.,  $g \in \mathfrak{T}_2^{0,r}(U)$ .

The Riemannian manifolds of class  $C^2$  are interesting for us due to the existence on them of a 'natural' linear connection induced by the metric. This unique linear connection associate with a  $C^1$  Riemannian metric is called *Riemannian* 

 $<sup>^2{\</sup>rm The}$  condition (4.1e) is equivalent to the non-degeneracy only in the finite-dimensional case, as we presuppose here.

<sup>&</sup>lt;sup>3</sup>See, e.g., [11,12,26] for the existence of such metrics, in particular on paracompact manifolds.

# 4. Riemannian manifolds

connection, known also as Levi-Civita connection, and it is defined as a metricpreserving torsionless linear connection on the respective manifolds. Explicitly this means that the Riemannian connection  $\nabla$  on U for a  $C^1$  metric g on U is defined as a  $C^0$  linear connection satisfying the system of equations<sup>4</sup>

$$\nabla_X g = 0 \tag{4.2}$$

$$T(X,Y) := \nabla_X Y - \nabla_Y X - [X,Y] = 0$$
(4.3)

for all vector fields  $X, Y \in \mathfrak{X}(U)$ . Its solution with respect to  $\nabla_X Y$  is implicitly given by [11, Chapter IV, § 2]<sup>5</sup>

$$2g(\nabla_X Y, Z) = X(g(Y, Z)) + Y(g(X, Z)) - Z(g(X, Y)) + g([X, Y], Z) + g([Z, X], Y) + g(X, [Z, Y])$$
(4.8)

where  $X, Y, Z \in \mathfrak{X}(U)$  are arbitrary. The explicit solution of this equation for the Riemannian connection is easily found in some local frame  $\{E_i\}$  on U.

Let

$$g_{ij} := g(E_i, E_j) \tag{4.9}$$

be the *local* (*covariant*) components of g in  $\{E_i\}$ .<sup>6</sup> From Definition 4.1 follows that  $[g_{ij}]$  is a symmetric non-degenerate matrix-valued function on U:

$$g_{ij} = g_{ji} \qquad \det[g_{ij}] \neq 0, \infty. \tag{4.10}$$

<sup>5</sup> Since  $g(Y, Z) = C_1^1(C_1^1(g \otimes Y \otimes Z))$ , we have

$$X(g(Y,Z)) = \nabla_X(g(Y,Z)) = (\nabla_X g)(Y,Z) + g(\nabla_X Y,Z) + g(Y,\nabla_X Z).$$

$$(4.4)$$

Making here the cyclic permutations  $(X, Y, Z) \mapsto (Y, Z, X) \mapsto (Z, X, Y)$ , summing the first two equations obtained and subtracting the last one from the result, we get

$$2g(\nabla_X Y, Z) = X(g(Y, Z)) + Y(g(Z, X)) - Z(g(X, Y)) + (\nabla_Z g)(X, Y)) - (\nabla_X)(Y, Z)) - (\nabla_Y g)(Z, X)) + g([X, Y]_, Z) + g([Z, Y]_, X) + g([Z, X]_, Y), \quad (4.5)$$

where the torsionless condition (4.3) was used. Now the metricity condition (4.2) reduces this equation to (4.8). Equation (4.5) is useful, for example, in the Weyl spaces in which (4.2) is replaced with

$$\nabla_X g = \omega(X)g \tag{4.6}$$

for some one-form  $\omega$ . A torsionless connection  $\nabla$  satisfying the equation (4.6) is called *Weyl* connection [41, 42]; the choice  $\omega = 0$  reduce a Weyl connection to Riemannian one. Let us note that for a Weyl connection equation (4.13) below is replaced by

$$\Gamma^{i}_{\ jk} = \left\{ {i \atop jk} \right\} + \frac{1}{2} \left( g^{im} C^{l}_{mj} g_{lk} + g^{im} C^{l}_{mk} g_{lj} - C^{i}_{jk} \right) + \frac{1}{2} (g_{jk} g^{il} \omega_l - \delta^{i}_{j} \omega_k - \delta^{i}_{k} \omega_j) \tag{4.7}$$

where  $\omega_i := \omega(E_i)$  is the *i*th component of  $\omega$ .

<sup>6</sup>Note, in the real case  $g(X, Y) = g_{ij}X^iY^j$ , while in the complex case, for a Hermitian metric h, we have  $h(X, Y) = g_{ij}X^i\overline{Y^j}$  where  $\overline{Y^j}$  is the complex conjugate to  $Y^j$ .

<sup>&</sup>lt;sup>4</sup>The metric preserving condition (4.2) can be expressed in a number of equivalent ways; e.g., by  $\nabla_X(g(Y,Z)) = g(\nabla_X Y, Z) + g(Y, \nabla_X Z), Z \in \mathfrak{X}(U)$  or by saying that the scalar product of every two vectors does not change after they are parallelly transported along an arbitrary path in U. A linear connection satisfying only the condition (4.2) is called metric connection; generally it depends on g and the torsion tensor.

Define  $g^{ij}$ , the contravariant components of the metric g, as the elements of the matrix inverse to  $[g_{ij}], [g^{ij}] := [g_{ij}]^{-1}$ , i.e.,

$$g^{ij}g_{jk} = \delta^i_k. \tag{4.11}$$

The matrix  $[g^{ij}]$  is also symmetric and non-degenerate.

Now, if we set  $X = E_i$ ,  $Y = E_j$ , and  $Z = E_k$  in (4.8) and take into account (3.1) and (3.15), we get

$$2g_{lk}\Gamma^{l}_{\ ji} = g_{jk,i} + g_{ik,j} - g_{ij,k} + C^{l}_{ij}g_{lk} + C^{l}_{ki}g_{lj} + C^{l}_{kj}g_{il}.$$
 (4.12)

From here and (4.11), we obtain

$$\Gamma^{i}_{\ jk} = \begin{cases} i\\ jk \end{cases} + \frac{1}{2} \left( g^{im} C^{l}_{mj} g_{lk} + g^{im} C^{l}_{mk} g_{lj} - C^{i}_{jk} \right)$$
(4.13)

where

$${i \\ jk} := {i \\ jk}_g := \frac{1}{2}g^{il}(g_{lj,k} + g_{lk,j} + g_{jk,l})$$
(4.14)

are the so-called *Christoffel symbols* [19, p. 132] <sup>7</sup> of the metric g (induced by g, assigned to g).<sup>8</sup> If we restrict ourselves to the class of  $(C^0)$  holonomic frames on U (or on the whole  $C^2$  manifold M), then  $C_{jk}^i \equiv 0$  and the Christoffel symbols coincide with the local coefficients of the Riemannian connection in them.<sup>9</sup> That is why the Riemannian connection is often denoted by  $\nabla^{\{\}}$  (and also by  ${}^{\{\}}\nabla$  or  $\nabla^{\{\}}$  if the position of the superscripts to  $\nabla$  is reserved for other purposes).

If the Christoffel symbols are known, from (4.14) we can find [18, p. 81] the derivatives  $g_{ij,k} = E_k(g_{ij})$  of the metric tensor:

$$g_{ij,k} = g_{il} \begin{Bmatrix} l \\ jk \end{Bmatrix} + g_{jl} \begin{Bmatrix} l \\ ik \end{Bmatrix}.$$
(4.16)

This equality, which in the context of (4.14) is identity, can be interpreted in the following way. Let  $\nabla$  be a torsionless connection with coefficients  $\Gamma^i_{\ jk} = \Gamma^i_{\ kj}$ 

$${i \choose jk} = \frac{1}{2g_i} (\delta_{ij}g_{i,k} + \delta_{ik}g_{i,j} - \delta_{jk}g_{j,i}).$$

$$(4.15)$$

<sup>&</sup>lt;sup>7</sup>The right-hand side of (4.12) without the last three terms is known as Christoffel symbols of the first kind and is denoted by [ij, k]. That is why (4.14) are also known as Christoffel symbols of the second kind. For detains see, e.g., [18] and [24, p. 17]. Nowadays this old terminology, introduced by E. B. Christoffel is practically out of usage.

<sup>&</sup>lt;sup>8</sup>If the metric g happens to be diagonal in  $\{E_i\}$ , i.e.,  $[g_{ij}] = \operatorname{diag}(g_1, \ldots, g_{\dim M})$  where  $\operatorname{diag}(a_1, \ldots, a_n)$  means a diagonal matrix with diagonal elements  $a_1, \ldots, a_n$ , then  $[g^{ij}] = \operatorname{diag}(1/g_1, \ldots, 1/g_{\dim M})$  and (do not sum over i!)

<sup>&</sup>lt;sup>9</sup>For this reason, sometimes, the coefficients of an arbitrary linear connection are called Christoffel symbols. This is not quite fair as E.B. Christoffel introduced his symbols in 1869 [19, p. 132, footnote 1], [24, p. 17], while the general theory of linear (affine) connections was developed more then half a century later, mainly due to the works of E. Cartan.

in some coordinate frame. Define  ${i \atop jk} = \Gamma^i_{kj}$ , then all metrics g whose Christoffel symbols coincide with  $\Gamma^i_{kj}$  in the given frame are solutions of (4.16).

On a Riemannian manifold whose metric is of class  $C^2$  exist two types of 'privilege' paths [11,24]: the geodesics assigned to the Riemannian connection and the extremals.<sup>10</sup> A remarkable result of the Riemannian geometry [11,12,24] is that in it the geodesics and extremals (defined as paths of minimal, maximal, or zero length between two points) *coincide*. Consequently, by (3.23) and (4.13), they satisfy the equations

$$\frac{\mathrm{d}^2\gamma^i}{\mathrm{d}s^2} + \left\{ \begin{matrix} i\\ jk \end{matrix} \right\} \frac{\mathrm{d}\gamma^j}{\mathrm{d}s} \frac{\mathrm{d}\gamma^k}{\mathrm{d}s} = 0, \tag{4.17}$$

in any coordinate frame  $\{\partial/\partial x^i\}$ , with  $\{{i\atop ik}\}$  being defined by (4.14).

The geometry of Riemannian manifolds is one of the best developed branches of the differential geometry. To it is devoted a huge number of woks, some of which were mentioned in Section 1. The reader is referred to them for deeper concepts and results in this region.

# 5. Normal frames: general ideas uniqueness, and holonomicity

Any manifold, as we saw in Section 2, is locally Euclidean in a sense that every its point has a neighborhood homeomorphic to  $\mathbb{K}^n$  (or its open subset). In subsection 3.2, we introduced linear connections on a manifold by means of covariant derivatives along vector fields on it. Are the linear connections 'locally Euclidean'? Are the covariant derivatives 'locally looking' like ordinary derivatives (along vector fields or along paths)? At this point these questions are only heuristic and their rigorous meaning will be revealed further in the present book. Since in any local frame the components of the covariant derivative of a tensor field are formed from the ordinary derivatives (along the basic vector fields) of its components (see (3.2) and the first term in (3.3)) and a 'correction' which is linear in them (see (3.2)) and the sums in (3.3), we may pose a problem for choosing a local frame so that this 'correction' to vanish. Since a vector field along it's integral paths can be replace by the integral paths' tangent vectors, we, by virtue of (3.17), could expect that such special frames may, possibly, exist along  $C^1$  paths. We would like the above 'correction' to vanish for all  $C^1$  tensor fields of arbitrary type. Due to (3.2) and (3.3), this is equivalent to the demand the (local) coefficients of the linear connection to vanish in the mentioned special kind of frames. Thus we have come to the following definition.

**Definition 5.1.** Given a linear connection  $\nabla$  on a differentiable manifold M and a subset  $U \subseteq M$ . A frame  $\{E_i\}$ , define on an open subset of M containing U

<sup>&</sup>lt;sup>10</sup>The extremals are defined as paths for which some functional on the space of paths connecting two fixed points is stationary; for details see, e.g., [12, p. 309ff] or [24, p. 48ff].

or equal to it, is called normal for  $\nabla$  on U if in it the coefficients of  $\nabla$  vanish everywhere on U. Respectively,  $\{E_i\}$  is called normal for  $\nabla$  along  $g: Q \to M, Q$ being non-empty set, if it is normal for  $\nabla$  on g(Q).

We can rephrase the first part of this definition by writing

$$\{E_i\} \text{ is normal on } U \iff \Gamma^i_{jk}|_U = 0 \text{ in } \{E_i\}.$$
(5.1)

Therefore if  $K \in \mathfrak{T}_s^r(M)$ , in a frame normal on U, if it exists, according to (3.2) and (3.3), we have on U

$$\left(\nabla_X K\right)_{j_1\dots j_s}^{i_1\dots i_r} = \left[E_k\left(K_{j_1\dots j_s}^{i_1\dots i_r}\right)\right] X^k = X\left(K_{j_1\dots j_s}^{i_1\dots i_r}\right)$$
(5.2)

which looks exactly like a directional derivative (along X) in  $\mathbb{R}^n$ . Just in this sense one can say that the linear connections and covariant derivatives are locally Euclidean but this is not quite rigorous because, at this point, we still do not know anything on the existence of normal frames. That problem will be investigated at length further in the present monograph.

By virtue of (3.1), the equalities  $\Gamma^i_{jk}|_U = 0$ ,  $i, j, k = 1, \ldots, \dim M$ , on a neighborhood U are equivalent to  $(\nabla_X E_i)|_U = 0$  for every  $X \in \mathfrak{X}(M)$ , i.e., the basic vector fields  $E_i$  must be parallel on U (see Definition 3.3). Thus we have proved the following simple result.

**Proposition 5.1.** A frame is normal on a neighborhood U in M iff it is parallel on U.

Therefore the concepts 'parallel frame' and 'normal frame' are equivalent on neighborhoods.

**Example 5.1.** A frame normal for a linear connection on a neighborhood U on which the connection is flat is provided by a frame obtained from a basis at a fixed point in U by its parallel transportation to the other points in U along paths lying in U.

We want to emphasize on the fact that a frame  $\{E_i\}$  normal for  $\nabla$  on a set  $U \subseteq M$  is always supposed to be defined on a neighborhood  $\overline{U}$  containing or equal to  $U, \overline{U} \supseteq U$ , and the equality  $\overline{U} = U$  is admissible if U is itself a neighborhood. This is quite essential assumption as the property of  $\{E_i\}$  to be normal at p for  $\nabla$  for every  $p \in U$  depends on the properties of  $\{E_i\}$  and  $\nabla$  in a neighborhood of p, not only at p. (See, e.g., equation (5.4) below in which the action of  $E_i$  on A appears which is not defined if A is not given on a neighborhood of each  $p \in U$ .)

Let  $\{E_i\}$  be a frame on  $U \subseteq M$ . On U exists a normal frame  $\{E'_i\}$  iff there is a  $C^1$  matrix-valued function  $A := [A^j_i]$  transforming  $\{E_i\}$  into  $\{E'_i\}, E'_i = A^j_i E_j$ , and such that in  $\{E'_i\}$  the coefficients on  $\nabla$  are  ${\Gamma'}^i{}_{jk} = 0$  which is a system of partial differential equations with respect to A. To write it in a compact form, as well as for saving writing and for purposes that will be clear in the next chapters, we introduce the matrices of the connection coefficients  $\Gamma_k := [\Gamma^i{}_{jk}]^{\dim M}$ ,  $k = 1, \ldots, \dim M$ .

# 5. Normal frames: General ideas and uniqueness

In their terms, the equation (3.5) reads<sup>1</sup>

$$\Gamma_k \mapsto \Gamma'_k = A_k^l A^{-1} \big( \Gamma_l A + E_l(A) \big) \tag{5.3}$$

where  $E_l(A) := \left[E_l(A_i^j)\right]_{i,j=1}^{\dim M}$ . Consequently  $\{E'_i\}$  is normal on U if the first-order system of partial differential equations

$$\left(\Gamma_l A + E_l(A)\right)\Big|_U = 0 \tag{5.4}$$

has solution(s) on U with respect to the matrix A.

The above considerations show that normal frames  $\{E'_i\}$  exist on a set U if and only if the system (5.4) has solutions in some (and hence in any) frame  $\{E_i\}$ . Besides, if such solutions exist, their properties are completely responsible for the properties of the frames normal on U. Moreover, to any solution A of (5.4) there corresponds a unique frame normal on U and v. v., i.e., there is a bijective correspondence between the solutions of (5.4) and the frames normal on U. For these reasons, the equation (5.4) is called the equation of the normal (on U) frames or simply the normal frame equation.

To the problems of existence and uniqueness of the solutions of (5.4) for linear connections is devoted Chapter II. It contains a complete description of all frames and coordinates normal at a point, along path, or on submanifolds (if such exist at all in the last case).

Now we shall prove two simple propositions, the first concerning the uniqueness of normal frames while the second one reveals the role of the torsion for the existence of normal coordinates (see below Definition 5.2).

**Proposition 5.2.** Let a linear connection admits a frame normal on  $U \subseteq M$ . The set of all frames normal for this connection on U consists of all frames that can be obtained from a fixed normal frame by means of linear transformations whose matrices vanish on U under the action of the basic vector fields of the normal frames.

*Proof.* Let  $\{E_i\}$  be a frame normal on U, i.e.,  $\Gamma_l = 0$  and  $\{E'_i = A^j_i E_j\}$  be a frame on U. By (5.3), in  $\{E'_i\}$  the matrices of the connection's coefficients are  $\Gamma'_k = A^l_k A^{-1} E_l(A), A = [A^j_i]$ . Hence,  $\Gamma'_k|_U = 0$  iff  $E_l(A)|_U = 0$ .

Let us mention the following evident consequence of this proposition (or of its proof).

**Corollary 5.1.** All frames normal on U for a linear connection, if any, are connected via linear transformations whose coefficients vanish on U under the action of the basic vector fields of the normal frames.

**Proposition 5.3.** Let a linear connection admits frames normal on some neighborhood  $U \subseteq M$ . All of these frames are either holonomic or anholonomic on U depending on is the torsion of the connection zero or non-zero on U respectively.

<sup>&</sup>lt;sup>1</sup>When writing the elements of a matrix A in the form  $A_i^j$  (or  $A_i^j$ , or  $A_i^j$ ), we consider the superscript as a first index and the subscript as a second index (see the list of conventions, page xii). So, the product of A and B is  $AB = [A_i^j B_k^j]$  if  $A = [A_i^j]$  and  $B = [B_i^j]$ .

*Proof.* Suppose  $\{E_i\}$  is a normal frame on U. In it, by (3.14), the torsion's components are  $T^i_{jk}|_U = -C^i_{jk}|_U$  and, consequently (see (3.15)),  $[E_j, E_k]_{-}|_U = -(T^i_{jk}E_i)|_U$ . Therefore  $[E_j, E_k]_{-}|_U = 0$  iff  $T|_U = 0$ .

Remark 5.1. The holonomicity of the frames normal on  $U \subseteq M$  does not imply any conclusions for their holonomicity outside U if  $U \neq M$  and the frames are defined on a larger set  $V \supset U$ ,  $V \subseteq M$ . Moreover, on  $V \setminus U$  the frames (normal on U, but not on  $V \setminus U$ ) can be either holonomic or anholonomic (or both on different subsets of  $V \setminus U$ ). In fact, let  $\{E_i\}$  be a frame defined on  $V \subseteq M$  which is normal on  $U \subset V$  and  $A = [A_i^j]$  be non-degenerate matrix-valued function on V. According to (3.5), in the frame  $\{E'_i := A_i^j E_j\}$  the connection's coefficients on U are  ${\Gamma'}^i{}_{jk} = (A^{-1})^i_l E'_k(A^l_j)$ . Thereof  $\{E'_i\}$  is normal on U iff  $E'_k(A^l_j)|_U = 0$ or, equivalently, iff  $E_k(A^l_j)|_U = 0$  which do not imply any restrictions on A (or on  $\{E'_i\}$ ) on the set  $V \setminus U$ . Hence, choosing appropriately A on  $V \setminus U$ , we can force  $\{E'_i\}$ , which is normal on U, on the set  $V \setminus U$ , i.e., outside U, to be holonomic as well as anholonomic depending on the properties of A on  $V \setminus U$ .

Remark 5.2. Proposition 5.3 has the following generalization. Call a frame  $\{E_i\}$ , define on a neighborhood containing or equal to  $U \subseteq M$ , commuting on U if  $[E_j, E_k]_{-|U} = 0$ . (If U is a neighborhood, a frame commuting on U is holonomic on U and vice versa.) Then a frame normal for a connection  $\nabla$  is commuting on Uiff  $\nabla$  is torsion free on U. (The proof is identical with the one of Proposition 5.3.) This result will not be used further as the commutativity of  $\{E_i\}$  on U does not, generally, imply some assertions concerning the holonomicity of  $\{E_i\}$  on U.

Remark 5.3. If U is not a neighborhood, from the relation  $[E_j, E_k]_{U} = 0$ , generally, does not follow the holonomicity of the frame  $\{E_i\}$  on U as, by definition, this is a property defined on neighborhoods.

In particular, Proposition 5.3 means that only the symmetric, i.e., torsionless, linear connections may admit holonomic frames normal on a neighborhood and, consequently, normal coordinates. Moreover, these connections do not admit anholonomic normal frames.

**Definition 5.2.** Given a linear connection  $\nabla$  on a differentiable manifold M. A chart (V, x) of M and the associated to it coordinate system  $\{x^i\}$  are called normal on a subset  $U \subseteq V$  for  $\nabla$  if the coordinate frame  $\{\frac{\partial}{\partial x^i}\}$  is normal on U for  $\nabla$ .

Rewording, normal coordinates are those the associated to which local frames are normal. A trivial consequence of Definition 5.2 and Propositions 5.2 and 5.3 is the following result.

**Corollary 5.2.** Coordinates normal on a neighborhood may exist only for torsion free linear connections and they generate all normal frames for them, if such exist at all.

*Remark* 5.4. The theorem what spaces (with linear connection) admit normal coordinates (at a fixed point), which is a special case of Proposition 5.3 and Corollary 5.2, dates back to the first third of the twentieth century; see [43,44].

We can also paraphrase the above in the assertion that the linear connections with non-vanishing torsion do not admit holonomic frames normal on a neighborhood and, consequently, coordinates normal for them do not exist; if for them normal frames exist, these frames are anholonomic with necessity.

For future purposes, we reformulate equation (5.4) in terms of local coordinates. Let  $U \subseteq M$ , and  $\{x^i\}$  be local coordinates associated with a chart (V, x) such that  $V \cap U \neq \emptyset$ . Then, due to (3.6), the defined on V coordinates  $x'^i = x'^i(x^1, \ldots, x^{\dim M})$  are normal on  $U \cap V$  iff

$$\left(\frac{\partial^2 x^i}{\partial x'^j \partial x'^k} + \frac{\partial x^m}{\partial x'^j} \frac{\partial x^n}{\partial x'^k} \Gamma^i_{mn}\right)\Big|_{U \cap V} = 0.$$
(5.4')

Hence, normal coordinates exist iff the last equation has solution(s) with respect to  $\{x'^i\}$  provided  $\{x^i\}$  is given. Obviously, (5.4') is a special case of (5.4) for  $E_i = \frac{\partial}{\partial x^i}$ ,  $E'_i = \frac{\partial}{\partial x'^i}$ , and U replaced with  $U \cap V$ , the last reflecting the local character of the coordinates. We call (5.4') the normal coordinates equation or the (system of) equation(s) of normal coordinates. Thus we see that the normal coordinates can be found by solving a second order system of partial differential equations while the analogous system for the normal frames is of first order. Correspondingly, if we are interested only in normal frames, the smoothness class of the connection admitting them is generally with one less then the one required for the existence of normal coordinates.

If the subset  $U \subseteq M$  is not a neighborhood, results slightly weaker than Proposition 5.3 and Corollary 5.2 hold:

**Proposition 5.4.** If a linear connection  $\nabla$  admits coordinates normal on a set U, it is torsionless on U. Said differently, if  $\nabla$  admits frame normal and holonomic on U, it is torsion free on U.

*Proof.* If  $q \in U$  and (V, x) with  $V \ni q$  is a normal chart on U for  $\nabla$ , equation (5.4) holds and, taking its antisymmetric part with respect to j and k, we get  $\Gamma^{i}_{[mn]}|_{U\cap V} = 0$ . Since  $\left\{\frac{\partial}{\partial x^{i}}\right\}$  is a holonomic frame,  $C^{i}_{jk} = 0$ , the last equality is equivalent to  $T^{i}_{jk}|_{U\cap V} = 0$  (see (3.14)). Hence  $T^{i}_{jk}(q) = 0$  for every  $q \in U$ , i.e.,  $\nabla$  is torsion free on U.

**Corollary 5.3.** Normal coordinates and holonomic normal frames may exist on  $U \subseteq M$  only for linear connections which are torsionless on U.

*Proof.* See Proposition 5.4.

Further, we shall see that if frames normal on a submanifold exist, the vanishment of the torsion is also a sufficient conditions for the existence of normal coordinates.

Just in the above arguments we see the reasons, why in the majority of the literature only the case of normal coordinates for torsionless (symmetric) linear connections is investigated.

Let  $\{E_i\}$  be a frame normal on U, defined on larger set  $V \supset U$ ,  $V \subseteq M$ , and not normal on  $V \setminus U \neq \emptyset$ . In such a case it should be clearly understood that the connection coefficients in  $\{E_i\}$  vanish *solely* on U and generally are non-zero on  $V \setminus U$ . This implies the non-vanishment of the derivatives  $\Gamma^i_{jk,l} := E_l(\Gamma^i_{jk})$ ,  $\Gamma^i_{jk,lm} := E_m(E_l(\Gamma^i_{jk}))$ , etc., if they exist, i.e., some of the partial derivatives of the connection's coefficients may vanish on U but not all of them. If dim U =dim M, the equalities  $\Gamma^i_{jk}|_U = 0$ , due to (3.13), imply  $R|_U = 0$ , i.e., the flatness of the connection is flat; in Section II.4 we shall see that this condition is also sufficient. By virtue of the above remarks, the normal frames, if any, can sometimes be used to simplify certain calculations. For instance, in a normal frame the components of curvature and torsion are (in the domain where the frame is normal):

$$R^{i}_{\ jkl} = -2\Gamma^{i}_{\ j[k,l]} \qquad T^{i}_{\ jk} = -C^{i}_{\ jk}$$

Now, let us see what happens with the parallel transport in a set  $U \subseteq M$  admitting a normal frame  $\{E_i\}$ . In it, due to (3.17), the parallel transport initial-value problem (3.18) reads

$$\frac{\mathrm{d}}{\mathrm{d}s}K^{i_1\dots i_{r'}}_{j_1\dots j_r}(\gamma(s)) = 0, \qquad K^{i_1\dots i_{r'}}_{j_1\dots j_r}(\gamma(s_0)) = K^{i_1\dots i_{r'}}_{0j_1\dots j_r}$$
(5.5)

and hence  $K_{j_1...j_r}^{i_1...i_{r'}}(\gamma(s)) = K_{0j_1...j_r}^{i_1...i_{r'}} = \text{const.}$  This implies an interesting conclusion: the parallel transport along paths lying entirely in the domain of a normal frame preserves the tensors' components in such a frames. A trivial, but important, corollary from here is the following result.

**Proposition 5.5.** If on a subset  $U \subseteq M$  exist normal frames, then the parallel transport between every two points in U is independent of the path along with it is performed provided this path lies entirely in U.

So, in a normal frame the parallel transport looks exactly as the 'parallel transport' in  $\mathbb{R}^n$  which preserves the vectors' components while it changes only the points at which they are 'attached'. This rigorous result is the strict meaning of the assertion that 'locally the parallel transport is Euclidean'. Here the problem what exactly 'locally' means is open. Strictly speaking, by 'locally' one should understand a set on which normal frames or coordinates exist.

The assertion inverse to Proposition 5.5 is (locally) valid if U is a submanifold. This is important and highly untrivial result that will be proved in Subsection II.5.2 (see Corollary II.5.1 on page 123).

**Proposition 5.6.** If frames normal on  $U \subseteq M$  exist, then all of them are parallel on the set U.

Proof. Let the frame  $\{E_i\}$  be normal on a set U and  $X = X^i E_i \in \mathfrak{X}(U)$ . Applying condition (ii) of Definition 3.1, (3.1), and (5.1), we obtain  $(\nabla_X E_j)|_U = (X^k \nabla_{E_k} E_j)|_U = X^k|_U (\Gamma^i_{jk}|_U) E_i|_U = 0$ . So, according to Definition 3.4, the frame  $\{E_i\}$  is parallel on U.

The statement inverse to Proposition 5.6 is partially valid provided U is a submanifold of M: if on U exist parallel frames (with respect to paths in U), it admits also normal ones but, generally, not all parallel on U frames are normal. This essential assertion will be proved in Subsection II.5.2 (see Corollary II.5.2 on page 123). From more general positions, the links between parallel transports and normal frames will be studied in Chapter IV.

Running a few steps forward, we want briefly to stress on the importance of the normal frames for the physics. Above we saw that in a normal frame the mathematical structures related to a linear connection look like the ones in  $\mathbb{R}^n$  (or in the Euclidean space  $\mathbb{E}^n$ ). Similar situation is observed in the theoretical physics: the consideration of a physical phenomenon in a suitable *reference frame*, usually called *inertial*, makes it looking like a 'free' one, i.e., as in the absence of forces.<sup>2</sup> The analogy between normal frames and frames of reference is most obvious in the general theory of relativity: in it the gravitational field strength is (locally) described via the coefficients of some Riemannian linear connection, so the gravity force (locally) 'disappears' in a frame normal for this connection and, consequently, this frame is inertial. The same situation can be discovered in the classical physics but it is so natural and (almost?) trivial that it is practically nowhere mentioned in this context. Recently it was shown that the Heisenberg picture of (nonrelativistic) quantum mechanics, which is something like a 'quantum mechanics in an inertial frame', is identical to the representation of this theory in Schrödinger in an appropriate normal frame (in a Hilbert bundle – see Section IV.16 below). These and other examples push forward the idea of identifying the *mathematical* concept 'normal frame' with the *physical* concept 'inertial frame (of reference)'. Generally, in the identification

# NORMAL FRAME $\equiv$ INERTIAL FRAME

we see the reason why the manifolds with linear connection find a broad application in a lot of fundamental physical theories. The above explains why the investigation of the normal frames is essential for the theoretical physics, not only for pure mathematical purposes.

 $<sup>^{2}</sup>$ A simple, but typical, example is the representation of the Klein-Gordon equation in normal coordinates which leads to the decomposition of the initial scalar field as a sum of independent harmonic oscillators [45, Chapter XXI, § 2]. A role similar to the normal coordinates play the normal modes or, more generally, normal waves in some linear dynamical systems [46, pp. 360–362].

# 6. Normal coordinates on Riemannian manifolds

In local coordinates  $\{x^i\}$  on  $U \subseteq M$ , M being a Riemannian  $C^2$  manifold with Riemannian metric g, the local coefficients of the Riemannian connection  $\nabla$  are (see (4.13) and (4.14))

$$\Gamma^{i}{}_{jk} = \frac{1}{2}g^{il} \left( \frac{\partial g_{lj}}{\partial x^{k}} + \frac{\partial g_{lk}}{\partial x^{j}} - \frac{\partial g_{jk}}{\partial x^{l}} \right) = \begin{cases} i\\ jk \end{cases}$$
(6.1)

**Proposition 6.1.** Local coordinates  $\{x^i\}$  are normal on U if and only if in them

$$\left. \frac{\partial g_{ij}}{\partial x^k} \right|_U = 0. \tag{6.2}$$

*Proof.* If  $\{x^i\}$  are normal on U, then  $\Gamma^i_{jk}|_U = 0$  by definition and, from equation (4.16), we get  $g_{ij,k}(x) = 0$ ,  $x \in U$ . Conversely, if (6.2) holds, equation (6.1) implies  $\Gamma^i_{jk}|_U = 0$ .

Therefore the normal coordinates on a Riemannian manifold can be defined as ones in which the partial derivatives of the components of the Riemannian metric vanish. Notice, equation (6.2) generally does not imply the vanishment of (all of) the second derivatives of the metric's components on U.

Regardless that nowadays we have at our disposal refine and powerful methods for proving the existence and construction of normal frames/coodinates on spaces far more general than the Riemannian ones, considered in the next chapters, below we present, following [24, pp. 51–57], possibly the first known such method which goes back to B. Riemann in 1854 [19, 24, resp. p. 155 and p. 53]. Besides from historical positions, this method is interesting due to the fact that, at least at a level of ideas, it is explicitly or implicitly presented as a part or underling background of some of the modern methods in that field.

Let M be a  $C^{\infty}$  manifold endowed with a  $C^{\infty}$  Riemannian metric g. Take an arbitrary point  $p \in M$  and let  $\{x^i\}$  be local coordinate system in a neighborhood  $U' \ni p$ . Consider on some subneighborhood  $U \ni p$  of U' the unique geodesic  $\gamma: J \to M$  passing through  $p, \gamma(s_0) = p$  for some  $s_0 \in J$ , in the direction of arbitrarily chosen vector  $X \in T_p(M), \dot{\gamma}(s_0) = X$ . This geodesic is necessarily of class  $C^{\infty}$  (see Section 4) and we can expand  $\gamma^i(s), s \in J$  in a power series with respect to  $s - s_0$ :

$$\gamma^{i}(s) = \gamma^{i}(s_{0}) + \frac{\mathrm{d}\gamma^{i}}{\mathrm{d}s}\Big|_{s_{0}}(s - s_{0}) + \sum_{n=2}^{\infty} \frac{1}{n!} \frac{\mathrm{d}^{n}\gamma^{i}}{\mathrm{d}s^{n}}\Big|_{s_{0}}(s - s_{0})^{n}.$$
 (6.3)

Taking into account  $\gamma^i(s_0) = p^i = x^i(p)$ ,  $\dot{\gamma}^i(s_0) = X^i$  and the geodesic equation (4.17), as well as the infinite number of equations obtained from it by

differentiation with respect to s, we derive the expansion

$$\gamma^{i}(s) = p^{i} + X^{i}(s - s_{0}) - \frac{1}{2} \begin{Bmatrix} i \\ jk \end{Bmatrix} \Big|_{p} X^{j} X^{k} (s - s_{0})^{2} \\ - \sum_{n=3}^{\infty} \frac{1}{n!} \Gamma^{i}_{i_{1} \dots i_{n}}(p) X^{i_{1}} \cdots X^{i_{n}} (s - s_{0})^{n}.$$
(6.4)

Here the  $C^1$  functions  $\Gamma^i_{i_1...i_n} \colon U \to \mathbb{R}$ , which are symmetric in  $i_1, \ldots, i_n$ , are given via the recurrent relations

$$\Gamma^{i}_{i_{1}\dots i_{n+1}} = \frac{\partial}{\partial x^{(i_{1}}} \Gamma^{i}_{i_{2}\dots i_{n+1})} - n \Gamma^{k}_{(i_{1}i_{2}} \Gamma^{i}_{i_{3}\dots i_{n+1})k}, \qquad n \in \mathbb{N} \setminus \{1\}, \tag{6.5}$$

where

$$\Gamma^{i}_{\ jk} = \begin{cases} i\\ jk \end{cases} \tag{6.6}$$

and the round brackets denote symmetrization according to our convention on page xii.

If the series (6.4) is convergent and  $\gamma$  is without self-intersections, the quantities

$$y^{i} = X^{i}(s - s_{0}) \tag{6.7}$$

can be used as coordinates along  $\gamma$  (i.e., of the current point  $\gamma(s)$ ) since the mapping  $\varphi \colon \gamma(s) \mapsto (X^1(s-s_0), \ldots, X^{\dim M}(s-s_0))$  is homeomorphism (between 1-dimensional manifolds). These coordinates can be extended in a whole neighborhood of p. Let V(p) be a normal neighborhood of p (see Definition 3.6 and Proposition 3.1). Since  $U \cap V(p)$  is, evidently, normal neighborhood, every  $q \in U \cap V(p)$  can be connected with p with a unique geodesic, say  $\gamma \colon J \to U \cap V(p)$ , lying entirely in  $U \cap V(p)$  [38, p. 385], [11, Chapter IV, Proposition 3.4]. So, there exist unique  $s_0, s \in J$  such that  $\gamma(s_0) = p$  and  $\gamma(s) = q$ . In  $U \cap V(p)$  we define a coordinate system  $\{y^i\}$  such that

$$y^{i}(q) := \dot{\gamma}^{i}(s_{0})(s - s_{0}). \tag{6.8}$$

In particular  $y^i(p) = 0$ . By virtue of (6.4), the link between  $\{x^i\}$  and  $\{y^i\}$  in  $U \cap V(p)$  is

$$x^{i}(q) = x^{i}(p) + y^{i}(q) - \frac{1}{2} \left\{ \begin{matrix} i \\ jk \end{matrix} \right\} \Big|_{p} y^{j}(q) y^{k}(q) - \sum_{n=3}^{\infty} \frac{1}{n!} \Gamma^{i}_{i_{1}...i_{n}}(p) y^{i_{1}}(q) \cdots y^{i_{n}}(q).$$

$$(6.9)$$

The transition  $\{x^i\} \mapsto \{y^i\}$  is regular in a neighborhood of p as its Jacobian at p is  $\det[\partial y^i/\partial x^j|_p] = \det[\partial x^i/\partial y^j|_p]^{-1} = \det[\delta_i^j] = 1 \neq 0$ . Below we shall prove that  $\{y^i\}$  are normal at p.

The just introduced coordinates  $\{y^i\}$  are known as *Riemannian coordinates* with origin (pole) p.<sup>1</sup> Their domain coincides with that for which the series (6.9) may be inverted or, equivalently, it is a neighborhood of p in which no two geodesics through p meet again in it. The above results can be formulate as the following assertion.

**Proposition 6.2.** For every point of a  $C^{\infty}$  Riemannian manifold with  $C^{\infty}$  metric there exists its normal neighborhood in which Riemannian coordinates can be introduces.

Evidently, the equation (6.7) is nothing else but the equation of the unique geodesic with initial conditions (3.24). So, in Riemannian coordinates, the geodesic equation (4.17) looks exactly as the equation of the straight lines in  $\mathbb{R}^n$ . Denoting the Christoffel symbols with respect to  $\{y^i\}$  with  $\{{}^i_{jk}\}^y$ , from (6.7) and (4.17), we derive

$$\begin{cases} i\\ jk \end{cases}^y y^j y^k \equiv 0$$

in the domain of  $\{y^i\}$ . On the opposite, if these equations hold,  $\{y^i\}$  are Riemannian coordinates.

The following proposition expresses the most important for us result of this section.

**Proposition 6.3.** The Riemannian coordinates are normal at their origin.

*Proof.* Using (6.9), we calculate

$$\frac{\partial x^i}{\partial y^j}\Big|_p = \frac{\partial y^i}{\partial x^j}\Big|_p = \delta^i_j, \qquad \frac{\partial^2 x^i}{\partial y^j \partial y^k}\Big|_p = -\left. \begin{cases} i\\ jk \end{cases} \right|_p$$

The substitution of these equations in the transformation law (3.6) results in  $\begin{cases} i \\ jk \end{cases}^{y} \Big|_{p} = 0.$ 

**Corollary 6.1.** The partial derivatives of the metric's components  $g_{ij}^y$  in the Riemannian coordinate system  $\{y^i\}$  vanish at the their origin p:

$$\frac{\partial g_{ij}^y}{\partial y^k}\Big|_p = 0. \tag{6.10}$$

 $\Box$ 

*Proof.* The result is a consequence of Propositions 6.1 and 6.3

The Riemannian coordinates are 'more than normal' at their origin, viz. at it vanish not only the Christoffel symbols, but also all of the quantities (6.5) and

<sup>&</sup>lt;sup>1</sup>The Riemannian (normal) coordinates find a vast field of application in physics, e.g., for approximate calculation of the metric tensor [47].

## 6. Normal coordinates on Riemannian manifolds

their symmetrized partial derivatives:

$${}^{y}\Gamma^{i}_{i_{1}...i_{n+1}}(p) = 0, \qquad n \in \mathbb{N},$$
 (6.11a)

$$\frac{\partial}{\partial y^{(i_1}} {}^y \Gamma^i_{i_2 \dots i_{n+2})}(p) = 0, \qquad (6.11b)$$

$$\frac{\partial}{\partial y^{(i_1}} \cdots \frac{\partial}{\partial y^{i_n}} {}^y \Gamma^i_{\ jk)}(p) = 0.$$
(6.11c)

Here the left superscript y indicates that the corresponding quantities are computed in  $\{y^i\}$ . The equalities (6.11a) follow from (6.9) if we put in them  $x^i = y^i$ (the initial choice of  $\{x^i\}$  is completely arbitrary) and take into account the symmetry of  $\Gamma^i_{i_1...i_{n+1}}$  in the subscripts. The other equations, (6.11b) and (6.11c), are consequence of (6.11a) and (6.5). So, at the origin of the Riemannian coordinates vanish the Christoffel symbols together with their symmetrized derivatives of all orders.

If we  $(C^{\infty})$  change the initial local coordinates  $\{x^i\}, \{x^i\} \mapsto \{x'^i\}$ , then, due to (6.8) and (6.9), this results in the change of the Riemannian coordinates with a constant matrix:

$$y^{i}(q) \mapsto y'^{i}(q) = a^{i}_{j} y^{j}(q) \tag{6.12}$$

with  $a_j^i = \frac{\partial x^{\prime i}}{\partial x^j}\Big|_p$ . The converse is also true: a change (6.12) can be described via analytic change  $\{x^i\} \mapsto \{x^{\prime i}\}$  which is, of course, not unique.

If  $\{y^i\}$  are Riemannian coordinates with origin at some  $p \in M$  in a neighborhood  $U \ni p$ , the holonomic frame  $\{\partial/\partial y^i\}$  on U is normal at p. As we know from Proposition 5.2, all other frames  $\{E_i\}$  normal at p are given by  $E_i = A_i^j \frac{\partial}{\partial y^j}$  on U where the non-degenerate matrix-valued function  $A = [A_i^j]$  is such that  $\frac{\partial A}{\partial y^i}|_p = 0$ . According to Proposition 5.3, these frames are holonomic at p,  $[E_i, E_j]_{-|p|} = 0$ , but outside p they need not to be such, i.e., generally  $[E_i, E_j]_{-|q|} \neq 0$  for  $q \in U \setminus \{p\}$  (see Remark 5.1 on page 40).

In this way we have obtained a *complete* description of the frames normal at a single point of a  $C^{\infty}$  Riemannian manifold.<sup>2</sup>

According to [19, pp. 155, 158] and [24, p. 53] normal coordinates (in a sense of Riemannian ones) ware first introduce by B. Riemann in 1854 [48].

Let  $\{y^i\}$  be Riemannian coordinates with origin at  $p \in M$ . Since  $[g_{ij}^y(p)]$  is a constant symmetric non-degenerate matrix, it can be diagonalized via transformation of the form  $A[g_{ij}^y(p)]A^{\top}$  with constant orthogonal matrix  $A = [a_i^j]$ ,  $A^{\top} = A^{-1}, A^{\top}$  being the transposed to A matrix [49, Chapter 4, § 7]. So we can define new Riemannian coordinates  $y'^i = a_j^i y^j$  in which  $[g_{ij}^{y'}(p)] = A[g_{ij}^y(p)]A^{\top} =$  $\operatorname{diag}(\varepsilon_1, \ldots, \varepsilon_{\dim M})$  where  $\varepsilon_i = \pm 1$  and  $\operatorname{diag}(a_1, \ldots, a_n)$  means a diagonal matrix with diagonal elements  $a_1, \ldots, a_n$ . These particular Riemannian coordinates were

<sup>&</sup>lt;sup>2</sup>At the moment this description is implicit. The explicit formula for the matrix A transforming  $\{\partial/\partial y^i\}$  to arbitrary normal frame will be derived in Subsection II.2.2; see Theorem II.2.3.

first introduced by G.D. Birkhoff in 1923 [50] under the name 'normal coordinates' (see [19, p. 155] and [24, p. 55]). The associated to these coordinates frame  $\{\partial/\partial y^i\}$  is such that at p it is orthonormal,  $g(\partial/\partial y^i, \partial/\partial y^j)|_p = \pm \delta_{ij}$ , and at the same time it is normal,  $\nabla_{\partial/\partial y^i}(\partial/\partial y^j)|_p = 0$  (see [11, Chapter IV, § 3]). Nowadays these special Riemannian coordinates are constructed by means of the exponential mapping; see the end of Subsection II.2.3 and [8].

Besides the Riemannian coordinates, there are also other classes of coordinates normal at a single point. Their complete description on a manifold with arbitrary symmetric linear connection will be given in Subsection II.2.2. An example of such coordinates are the *geodesic coordinates*.<sup>3</sup> If  $\{x^i\}$  are local coordinates in a neighborhood U of  $p \in M$ , the geodesic coordinates  $\{z^i\}$  with origin at p of a point q belonging to some subneighborhood of U, are given via the series [24, p. 56]

$$x^{i}(q) = x^{i}(p) + z^{i}(q) - \frac{1}{2} \left\{ \begin{matrix} i \\ jk \end{matrix} \right\} \Big|_{p} z^{j}(q) z^{k}(q) - \sum_{n=3}^{\infty} \frac{1}{n!} c^{i}{}_{i_{1}\dots i_{n}} z^{i_{1}}(q) \cdots z^{i_{n}}(q)$$
(6.13)

where  $c^i_{i_1...i_n} = c^i_{(i_1...i_n)} \in \mathbb{R}$  and  $z^i(p) := 0$ . Analogously to (6.9), it is convergent and invertible with respect to  $z^i(q)$  in some neighborhood of p.

Repeating the proof of Proposition 6.3, we see that the geodesic coordinates are normal at their origin p,  ${i \atop jk}|_p = 0$ . In them the equation of the geodesics through the origin reads (see (6.4))

$$\gamma^{i}(s) = \gamma^{i}(s_{0})(s-s_{0}) - \sum_{n=3}^{\infty} \frac{1}{n!} \Gamma^{i}{}_{i_{1}\dots i_{n}}(p) X^{i_{1}} \cdots X^{i_{n}}(s-s_{0})^{n}$$

with  $p = \gamma(s_0)$ ,  $X = \dot{\gamma}(s_0)$ , and the  $\Gamma$ 's being defined by (6.5). Hence the Riemannian coordinates are geodesic ones in which the  $\Gamma$ 's vanish at the origin (cf. (6.11)).

A number of examples of coordinates/frames normal for Riemannian connections will be presented below in Section 7. Instances of some applications of the Riemannian and geodesic coordinates can be found in [19, 24, 51].

At this point a natural question arises: are there subsets of a Riemannian manifold, different from single points, on which normal coordinates/frames exist?

The first result in this field was obtained by E. Fermi in 1922 [52]: a coordinate system can be chosen so that in it the partial derivatives of the components of the metric tensor vanish along a given smooth path without self-intersections (see also [53,54]). Now the Fermi's proof of this theorem is only of historical interest and we are not going to reproduce it here. Further, in Section II.3, we will establish an analogous result for arbitrary linear connections from which the above assertion is an evident special case (see, in particular, Proposition II.3.2 and Corollary II.3.1).

 $<sup>^{3}</sup>$ In [51] the geodesic coordinates are defined as ones in which the partial derivatives of the metric vanish at a given point. By Proposition 6.1, the concepts 'geodesic coordinates' in Fock's sense and 'normal coordinates' in our sense are synonyms.

# 7. Examples of normal coordinates

Till 1958 the problem for existence of normal coordinates/frames on sets other than points and curves (without self-intersections) in a general Riemannian manifold, as well as for other manifolds with linear connection, was open. In 1958 O'Raifeartaigh [55] proved a general theorem concerning the existence of normal frames on arbitrary submanifolds of manifolds with symmetric, i.e., torsionless, linear connection. This result will be reviewed in Section II.5. When applied to the above problem, it asserts that the points and curves are the only submanifolds of a Riemannian manifold on which normal coordinates/frames always exists. Only special types of Riemannian connections admit frames normal on other submanifolds; in particular, coordinates/frames normal on the whole manifold or its open subset (neighborhood) exist iff the connection is flat on it.

# 7. Examples of normal coordinates for Riemannian connections

The purpose of this section is to exemplify the general theory of Sections 5 and 6 on manifolds endowed with Riemannian connections generated by Riemannian metrics. The section contains also several exercises.

Most of the examples presented below will follow the following scheme. Let M be a submanifold of  $\mathbb{R}^N$  for some  $N \in \mathbb{N}^{-1}$  and  $f : \mathbb{R}^N \times \mathbb{R}^N \to \mathbb{R}$  be a scalar product (see, e.g., (7.1) and (7.16) below) which induces a Riemannian metric  $\overline{f}$  on  $T(\mathbb{R}^N)$ ; one usually identifies f and  $\overline{f}$  as  $T(\mathbb{R}^N)$  and  $\mathbb{R}^N$  are isomorphic vector spaces. Suppose g is the Riemannian metric on M induced (generated) by  $\overline{f}$ , viz.  $g = \overline{f}|_M$  is the restriction of  $\overline{f}$  to M.<sup>2</sup> Then the metric g induces on M a Riemannian connection  $\nabla$  as described in Section 4 (see, in particular, equations (4.13) and (4.14)). We shall calculate the coefficients of  $\nabla$  in concrete coordinate systems and will look for subsets of M on which these systems are normal for  $\nabla$ . Besides, some constructions of geodesic coordinates will be pointed out.

Suppose the manifold under consideration has some symmetry and the coordinates/frames on it are chosen in a way that 'reflects' this symmetry. As we shall see below, in such a situation it is 'quite likely' that these coordinates/frames may happen to be normal on some ('natural' for the symmetry) subset for the connection induced on the manifold by a Riemannian metric which, in its turn, is generated by the Euclidean metric of some  $\mathbb{R}^n$  for some  $n \in \mathbb{N}$  in which the initial manifold is embedded.

<sup>&</sup>lt;sup>1</sup>Recall the Whitney embedding theorem [56, p. 44, Theorem 1.9.12]: every  $C^2$  *n*-dimensional manifold can be embedded in  $\mathbb{R}^{2n}$ , i.e., can be considered as a submanifold of  $\mathbb{R}^N$  for some  $N \in \mathbb{N}$ ; see also [56, p. 119, Remark 3.2.9].

<sup>&</sup>lt;sup>2</sup>The easiest way to find g explicitly is to use differential forms. Let d denotes the exterior derivative operator and  $\wedge$  is the wedge (exterior) product sign. If  $\bar{f} = f_{ij} dx^i \wedge dx^j$  in some coordinate system  $\{x^i\}$  on  $\mathbb{R}^N$ , then  $g = f_{ij}|_M dx^i|_M \wedge dx^j|_M$  and it is quite convenient to use as a coordinate system on M the restricted coordinate system  $\{x^i|_M\}$ , which agrees with the definition of a submanifold given on page 7.

**Example 7.1 ((Locally) (pseudo-)Euclidean spaces).** The Euclidean space  $\mathbb{E}^n$ ,  $n \in \mathbb{N}$ , is a collection of  $\mathbb{R}^n$  and (standard Euclidean) scalar product  $e \colon \mathbb{R}^n \times \mathbb{R}^n \to \mathbb{R}$  such that

$$e(p,q) = \sum_{i=1}^{n} p^{i} q^{i}$$
 (7.1)

for all  $p = (p^1, \ldots, p^n) \in \mathbb{R}^n$  and  $q = (q^1, \ldots, q^n) \in \mathbb{R}^n$ . If  $\{u^i\}$  is the standard coordinate system in  $\mathbb{R}^n$ ,  $u^i(p) := p^i$ , the linear mapping  $I_q: T_q(\mathbb{E}^n) \to \mathbb{E}^n$  with  $I_q(p^i \frac{\partial}{\partial u^i}|_q) = p$  is an isomorphism and transfers the scalar product e from  $\mathbb{E}^n$  to  $T_x(\mathbb{E}^n)$  by  $e \mapsto \bar{e}: p \mapsto \bar{e}_p := e \circ (I_p \times I_p)$ ; hence  $\bar{e}$  is a metric on  $\mathbb{E}^n$  (or  $\mathbb{R}^n$ ) such that

$$\bar{e}\left(p^{i}\frac{\partial}{\partial u^{i}}, p^{i}\frac{\partial}{\partial u^{i}}\right) = e(p,q)$$
(7.2)

and the components of  $\bar{e}$  in the global frame  $\left\{\frac{\partial}{\partial u^i}\right\}$  are  $\bar{e}_{ij} = \delta_{ij}$  (see (4.9)).<sup>4</sup> The metric  $\bar{e}$  induces on  $\mathbb{E}^n$  a (flat) Riemannian connection  $\nabla^E$  whose coefficients in  $\left\{\frac{\partial}{\partial u^i}\right\}$  are (see (4.13))

$$\Gamma^{i}_{\ jk} = \begin{cases} i\\ jk \end{cases} = 0. \tag{7.3}$$

Consequently the (global) coordinate system  $\{u^i\}$  and the (global) frame  $\{\frac{\partial}{\partial u^i}\}$  are normal for  $\nabla^E$  on the whole space  $\mathbb{E}^n$ .

Similarly, if on a manifold M is given a Riemannian metric g such that its components  $g_{ij}$  are constant functions relative to a frame  $\left(\frac{\partial}{\partial x^i}\right)$  induced by the coordinates  $x^i$  of a chart (V, x) of M, then the Riemannian connection  $\nabla^g$ generated by g is such that (see (4.13))

$$\Gamma^{i}_{\ jk}|_{V} = \left\{ \frac{i}{jk} \right\} \Big|_{V} = 0.$$
(7.4)

Consequently the coordinate system  $\{x^i\}$  and the corresponding to it natural frame  $\{\frac{\partial}{\partial x^i}\}$  are normal for  $\nabla^g$  on V; the afore considered case of  $\mathbb{E}^n$  corresponds to the choice  $(M,g) = (\mathbb{R}^n, \bar{e}), V = \mathbb{R}^n$  and  $x^i = u^i$ . All other frames  $\{E_i\}$  normal on V for  $\nabla^g$  are such that  $E_i = A_i^j \frac{\partial}{\partial x^j}$  with  $\det[A_i^j] \neq 0, \infty$  and  $\frac{\partial A_i^j}{\partial x^k}|_V = 0$  (see Proposition 5.2), so that  $A_i^j$  are constant on V. Evidently  $E_i = \frac{\partial}{\partial y^i}$  for some coordinates  $y^i$  normal on V and such that  $x^i = A_j^i y^j + A^i$  for some numbers  $A^i \in \mathbb{K}$ .

**Example 7.2 (One-dimensional manifolds).** Let M be one-dimensional manifold, dim M = 1, endowed with a  $C^1$  Riemannian metric g. Let (V, x) be a chart of M

<sup>&</sup>lt;sup>3</sup>In this section, we denote the standard coordinates on  $\mathbb{R}^n$  by  $u^1, \ldots, u^n$  instead by  $r^1, \ldots, r^n$  to distinguish them from some radii (e.g., of (pseudo)spheres or torii), typically denoted by r with possible indices, and their powers.

<sup>&</sup>lt;sup>4</sup>Usually one identifies  $T_p(\mathbb{E}^n)$  with  $\mathbb{E}^n$ ,  $p^i \frac{\partial}{\partial u^i}$  with p, and e with  $\bar{e}$ .

# 7. Examples of normal coordinates

and  $\{x^1\}$  be the corresponding local coordinate system. The sole coefficient of the Riemannian connection  $\nabla$  induced by g is (see (4.13) or (4.15))

$$\Gamma^{1}_{11} = \begin{cases} 1\\11 \end{cases} = \frac{1}{2} \frac{1}{g_{11}} \frac{\partial g_{11}}{\partial x^{1}} = \frac{1}{2} \frac{\partial \ln g_{11}}{\partial x_{1}}, \tag{7.5}$$

where  $g_{11} = 1/g^{11}$  is the sole component of g in  $\left\{\frac{\partial}{\partial x^1}\right\}$ . Therefore  $\{x^1\}$  and  $\left\{\frac{\partial}{\partial x^1}\right\}$  are normal on a subset  $U \subseteq V$ , i.e.,  $\Gamma^1_{11}|_U = 0$ , iff

$$\left. \frac{\partial g_{11}}{\partial x^1} \right|_U = 0. \tag{7.6}$$

Suppose U is an open set, i.e.,  $U = x^{-1}(J)$  for some open real interval J. Define a chart (V, x') with local coordinate  $x'^{1}$  such that, for a fixed  $p_{0} \in V$  and all  $p \in V$ ,

$$x'^{1}(p) = x^{1}(p_{0}) + \int_{p_{0}}^{p} |g_{11}|^{-\frac{1}{2}} dx^{1}$$
(7.7)

where  $|\lambda|$  is the absolute value of  $\lambda \in \mathbb{R}$ , viz.  $|\lambda| = (\operatorname{sign} \lambda)\lambda$  with  $\operatorname{sign} \lambda := \pm 1$  for  $\lambda \leq 0$  and  $\operatorname{sign} \lambda = 0$  for  $\lambda = 0$ . The transformation  $x^1 \mapsto x'^1$  is invertible in V as its Jacobian is  $\frac{\partial x'^1}{\partial x^1} = |g_{11}|^{-\frac{1}{2}} \neq 0$ . The component of g in  $\{\frac{\partial}{\partial x^1}\}$  is (see (2.44) with  $A_1^1 = \frac{\partial x^1}{\partial x'^1}$ )

$$g'_{11} = g\left(\frac{\partial}{\partial x'^1}, \frac{\partial}{\partial x'^1}\right) = \operatorname{sign} g_{11}.$$
 (7.8)

It is constant on V and consequently

$$\frac{\partial g'_{11}}{\partial x'^{1}}\Big|_{V} = 0 \qquad {\Gamma'}^{i}_{jk}\Big|_{V} = 0 \tag{7.9}$$

so that  $\{x^i\}$  and  $\{\frac{\partial}{\partial x^1}\}$  are normal on V for  $\nabla$ .

The main conclusion from the above is that for any point in one-dimensional Riemannian space there exists local coordinates in some its neighborhood which coordinates are normal on their domain for the Riemannian connection induced via the initial Riemannian metric. Since the curvature tensor identically vanishes for a  $C^1$  connection on  $C^3$  1-manifold (see (3.11) and (3.13)), this result is also a consequence of Theorem II.4.1, presented below on page 104, in a case of a  $C^2$  metric on a  $C^3$  1-dimensional Riemannian manifold.<sup>5</sup>

**Example 7.3 (The two-sphere**  $\mathbb{S}^2$ ). Let  $\mathbb{S}^2 := \{(v^1, v^2, v^3) \in \mathbb{R}^3 : (v^1)^2 + (v^2)^2 + (v^3)^2 = r^2\}$  be a two-dimensional sphere of radius  $r \in \mathbb{R}, r > 0$ , in  $\mathbb{R}^3$ . Let on  $\mathbb{S}^2$  minus one point (e.g., the 'North pole') be given the standard spherical coordinates  $(\theta, \varphi)$ , with range  $(0, \pi] \times [0, 2\pi)$ , obtained from the spherical coordinates in  $\mathbb{R}^3$  via restriction to  $\mathbb{S}^2$  (see equations (7.12) below or/and [57, Section 3.1-6]).

 $<sup>^5 \</sup>mathrm{See}$  also Example II.6.7.

The Euclidean metric in  $\mathbb{R}^3$  induces on  $\mathbb{S}^2$  a metric g whose components  $g_{ij}$ , i, j = 1, 2, in the coordinates  $\{x^1 = \theta, x^2 = \varphi\}$  are such that [57, Table 6.5-1]

$$[g_{ij}] = \operatorname{diag}(r^2, r^2 \sin^2 \theta) \quad [g^{ij}] = \operatorname{diag}\left(\frac{1}{r^2}, \frac{1}{r^2 \sin^2 \theta}\right).$$
(7.10)

Let  $\nabla$  be the Riemannian connection generated by g (see Section 4). An elementary calculation by means of (4.15) shows that the non-vanishing coefficients of  $\nabla$  in  $\{x^1, x^2\}$  are (see also [57, Table 6.5-1])

$$\Gamma^{1}_{22} = -\sin\theta\cos\theta \quad \Gamma^{2}_{12} = \Gamma^{2}_{21} = \cot\theta.$$
(7.11)

Since  $\cos \frac{\pi}{2} = \cot \frac{\pi}{2} = 0$ , the spherical coordinates  $\{x^1 = \theta, x^2 = \varphi\}$  and hence the frame  $\left\{\frac{\partial}{\partial \theta}, \frac{\partial}{\partial \varphi}\right\}$  on  $\mathbb{S}^2$  are normal for  $\nabla$  along the path  $\gamma \colon [0, 2\pi) \to \mathbb{S}^2$  such that  $\theta \circ \gamma = \frac{\pi}{2}$  and  $\varphi \circ \gamma = \mathrm{id}_{[0,2\pi)}$ , i.e., they are normal on the great circle obtained by intersecting  $\mathbb{S}^2$  with the equatorial plane  $\{u^3(p) = 0 : p \in \mathbb{R}^3\}$ . The general frame  $\{E_1, E_2\}$  normal for  $\nabla$  along  $\gamma$  is such that  $E_i = A_i^j \frac{\partial}{\partial x^j}$  where the  $2 \times 2$  matrix-valued function A satisfies the equation (see Proposition 5.2)  $\frac{\partial A_i^j}{\partial x^k}|_{\gamma([0,2\pi))} = 0$ . The explicit form of A will be presented in Example II.6.2 below; for details concerning frames/coordinates normal along paths for linear connections, see Section II.3 in the Chapter II.

Due to the symmetry of the sphere  $\mathbb{S}^2$ , it is clear that for any great circle on it there exist (local) coordinates on  $\mathbb{S}^2$  which are normal on this circle for the Riemannian connection on  $\mathbb{S}^2$  generated by the metric g (induced from  $\mathbb{R}^3$ ).

**Example 7.4 (The spheres**  $\mathbb{S}^n$  for any  $n \in \mathbb{N}$ ). The circle  $\mathbb{S}^1 := \{(v^1, v^2) \in \mathbb{R}^2 : (v^1)^2 + (v^2)^2 = r^2\}$  of radius  $r \in \mathbb{R}$ , r > 0, is a 1-dimensional manifold. The polar coordinate function  $\varphi$ , with range  $[0, 2\pi)$  and induced from polar coordinates in  $\mathbb{R}^2$ , provides a coordinate function on a subset V equal to  $\mathbb{S}^1$  without some arbitrarily fixed point. The standard Euclidean metric on  $\mathbb{E}^2$  induces on  $\mathbb{S}^1$  a metric g whose only component in  $\{x^1 = \varphi\}$  is  $g_{11} = r^2 = \text{const.}$  Hence the metric g generates on  $\mathbb{S}^2$  a Riemannian connection  $\nabla$  whose only coefficient in  $\{x^1\}$  is  $\Gamma^1_{11}|_V = 0$ . Consequently, the polar coordinate system  $\{x^1 = \varphi\}$  on  $\mathbb{S}^2$  is normal for  $\nabla$  on V. This result agrees with the results of Examples 7.1 (g is a 'constant' metric in  $\{x^1\}$ ) and 7.2 ( $\mathbb{S}^1$  is a one-dimensional manifold).

Consider now the general case of an *n*-dimensional,  $n \ge 2$ , sphere  $\mathbb{S}^n$  in  $\mathbb{R}^{n+1}$  of radius r > 0, viz.

$$\mathbb{S}^{n} := \{ (v^{1}, \dots, v^{n}) \in \mathbb{R}^{n+1} : (v^{1})^{2} + \dots + (v^{n})^{2} = r^{2} \}.$$

Let  $(R, \theta_{n-1}, \ldots, \theta_1, \varphi)$  be the (hyper-)spherical coordinates in  $\mathbb{R}^{n+1}$ ; the range of R is  $(0, \infty)$ , the one of  $\theta_{n-1}, \ldots, \theta_1$  is  $(0, \pi]$  and  $\varphi$  takes values in  $[0, 2\pi)$ . They are connected with the standard coordinate system  $\{u^1, \ldots, u^{n+1}\}$  in  $\mathbb{R}^{n+1}$  by the

#### 7. Examples of normal coordinates

following equations:

The equation of  $\mathbb{S}^n$  in these coordinates is  $R^2 = r^2$  or R = r (as r > 0 and R takes non-negative values). The functions

$$\{x^1 = \theta_{n-1}, x^2 = \theta_{n-2}, \dots, x^{n-1} = \theta_1, x^n = \varphi\}$$
(7.13)

provide an internal coordinates system on  $\mathbb{S}^n$  (without one point) in which the metric g induced on  $\mathbb{S}^n$  from  $\mathbb{E}^n$  has components  $g_{ij}$ ,  $i, j = 1, \ldots, n$ , such that  $(2 \le k \le n-1)$ 

$$[g_{ij}] = \operatorname{diag}(r^2, r^2 \sin^2 x^1, \dots, r^2 \sin^2 x^1 \dots \sin^2 x^{k-1}, \dots, r^2 \sin^2 x^1 \dots \sin^2 x^{n-1}).$$
(7.14)

The metric g induces on  $\mathbb{S}^n$  a Riemannian metric  $\nabla$ . Using equations (4.15) and (4.13), we can calculate that  $\nabla$  has the following non-vanishing coefficients in the coordinate system (7.13) (do not sum over i and k!):

$$\Gamma^i_{\ ik}|_{k< i} = \Gamma^i_{\ ki}|_{k< i} = \cot x^k \tag{7.15a}$$

$$\Gamma^{i}_{kk}|_{k>i} = -\sin x^{i} \cos x^{i} \prod_{l=i+1}^{k-1} \sin^{2} x^{l}$$
(7.15b)

for  $1 \leq i, k \leq n$ . (We set  $\prod_{l=a}^{b} (\cdots) := 1$  for b < a.) For instance, these coefficients for n = 2 (resp. for n = 3) are given by (7.11) (resp. by (7.43) below).

Let  $\gamma: [0, 2\pi) \to \mathbb{S}^n$  be a path in  $\mathbb{S}^n$  such that  $x^i \circ \gamma = \frac{\pi}{2}$  for  $i = 1, \ldots, n-1$ and  $x^n \circ \gamma = \operatorname{id}_{[0,2\pi]}$ . Since  $\cos \frac{\pi}{2} = \cot \frac{\pi}{2} = 0$  and  $\sin \frac{\pi}{2} = 1$ , we have

$$\Gamma^i_{\ jk}|_{\gamma([0,2\pi))} = 0$$

and, consequently, the coordinate system  $\{x^i\}$  and the frame  $\{\frac{\partial}{\partial x^i}\}$  are normal for  $\nabla$  along the path  $\gamma$ . From (7.12) is clear that the set  $\gamma([0, 2\pi))$  is a circle obtained by intersecting  $\mathbb{S}^n \subset \mathbb{R}^{n+1}$  with the  $(u^1, u^2)$ -plane  $\{v \in \mathbb{R}^{n+1} : u^i(p) = 0 \text{ for } i \geq 3\}$  in  $\mathbb{R}^{n+1}$ ; in fact, we have

$$\gamma([0,2\pi)) = \{ (v^1, v^1, \underbrace{0, \dots, 0}_{(n-1)-\text{times}}) \in \mathbb{R}^{n+1} : (v^1)^2 + (v^2)^2 = r^2 \} = \mathbb{S}^1 \times (\underbrace{0, \dots, 0}_{(n-1)-\text{times}}).$$

It is clear, if C is a circle on  $\mathbb{S}^n$  obtained by intersecting  $\mathbb{S}^n$  by a 2-plane through its origin, then there are coordinates on  $\mathbb{S}^n$  normal along C for the Riemannian connection considered above; this is a consequence of the afore-presented material in which  $u^1$  and  $u^2$  should be standard coordinates in the 2-plane to which C belongs.

The case of  $\mathbb{S}^2$ , investigated in Example 7.3, corresponds to the choice n = 2.

**Example 7.5 (The pseudospheres**  $\mathbb{S}_q^n$ ). The Euclidean space  $\mathbb{R}_q^n := \mathbb{R}_{p,q}^n := (\mathbb{R}^n, e_q^n)$  of index q, where  $p, q \in \mathbb{N} \cup \{0\}$  and  $p + q = n \in \mathbb{N}$ , consists of the space  $\mathbb{R}^n$  endowed with a scalar product (metric) (of index q)  $e_q^n := e_{p,q}^n$  such that

$$e_q^n(v,w) := \sum_{a=1}^p v^a w^a - \sum_{b=p+1}^n v^b w^b$$
(7.16)

for  $v = (v^1, \dots, v^n) \in \mathbb{R}^n$  and  $w = (w^1, \dots, w^n) \in \mathbb{R}^n.^6$ 

Special cases of the spaces  $\mathbb{R}_q^n$  are well know in physics. For instance, the (Lorentzian) space  $\mathbb{R}_4^5 = \mathbb{R}_4^4 \times \mathbb{R}$  and the spherical space  $\mathbb{S}^3 \subset \mathbb{R}^4$  of (special) radius R find application in the geometrical interpretation of the Einstein model of the Universe [58, § 138] (see also Example 7.9 below). However, the most famous example is the Minkowski spacetime  $M_4 := \mathbb{R}_3^4$  (or the isomorphic to it space  $\mathbb{R}_1^4$ ) on which the special theory of relativity and the whole relativistic physics rests; for instance, see [20, 21, 51, 59–64].

The pseudosphere (pseudo-Riemannian spherical manifold)  $\mathbb{S}_q^n$  of index q, with  $n \geq 1$  and  $0 \leq q \leq n$ , of radius  $r \in \mathbb{R}$ , r > 0, in  $\mathbb{R}_q^{n+1}$  is defined by [39, Section 2.4], [27, Chapter 1, § 4]

$$\mathbb{S}_q^n := \{ v \in \mathbb{R}_q^{n+1} : e_q^{n+1}(v, v) = r^2 \}.$$

It is a  $C^{\omega}$  2-connected manifold. Since the pseudospheres reduce in the case q = 0 to ordinary spheres,  $\mathbb{S}_0^n = \mathbb{S}^n$ , investigated in Example 7.4, further we shall suppose that  $q \geq 1$ . Since the pseudosphere  $\mathbb{S}_1^1$  is a 2-connected 1-dimensional manifold, which case is covered by Example 7.2, we shall suppose below  $n \geq 2$ .<sup>7</sup>

By g below will be denoted the Riemannian metric on  $\mathbb{S}_q^n$  induced by the metric  $e_q^{n+1}$ ,  $g = e_q^{n+1}|_{\mathbb{S}_q^n}$ . Respectively, by  $\nabla$  will be denoted the Riemannian connection on  $\mathbb{S}_q^n$  generated by g as described in Section 4.

$$u^{1} = \rho \cosh \chi \qquad u^{2} = \rho \sinh \chi. \tag{7.17}$$

In them  $\mathbb{S}_1^1 = \{v \in \mathbb{R}_1^2 : \rho(v) = \pm r\}$ . The function  $\chi$  provides a coordinate system  $\{\chi\}$  with domain  $\{v \in \mathbb{R}_1^2 : \rho(v) = +r\}$  or  $\{v \in \mathbb{R}_1^2 : \rho(v) = -r\}$  on  $\mathbb{S}_1^1$ . In it the metric g induced on  $\mathbb{S}_1^1$  by  $e_1^1$  has a single component equal to  $r^2$ . Hence g induces a flat Riemannian connection on  $\mathbb{S}_1^1$  for which  $\{\chi\}$  is normal on its domain. Cf. the last part of Example 7.1.

<sup>&</sup>lt;sup>6</sup>By replacing e and  $\bar{e}$  in (7.2) by  $e_q^n$  and  $\bar{e}_q$ , respectively, we obtain a Riemannian (pseudo-Euclidean) metric  $\bar{e}_q$  of index q on  $\mathbb{R}^n$  induced by  $e_q^n$ ; for evident reasons, one usually writes  $e_q^n$  for  $\bar{e}_q^n$ .

<sup>&</sup>lt;sup>7</sup>Éxample 7.1 is also applicable in the concrete case. Briefly this can be shown as follows. On  $\mathbb{R}^2_1$  can be introduced pseudospherical coordinates  $(\rho, \chi)$ , with respective ranges  $\mathbb{R} \setminus \{0\}$  and  $\mathbb{R}$ , connected with the standard coordinates  $(u^1, u^2)$  on  $\mathbb{R}^2$  by

# 7. Examples of normal coordinates

Let  $n \ge 2$ ,  $q \ge 1$  and  $p := n - q \ge 1$ ; hence  $q \le n - 1$ . Then in  $\mathbb{R}_q^{n+1}$  can be introduced pseudospherical coordinates

$$(\rho, \chi_q, \dots, \chi_1, \theta_{p-1}, \dots, \theta_1, \varphi) \tag{7.18}$$

where for p = 1 the  $\theta$ 's and the terms containing them below should be missing/deleted. The range of  $\rho$  is  $\mathbb{R} \setminus \{0\}$ , the one of the  $\chi$ 's is  $\mathbb{R}$ , the range of the  $\theta$ 's is  $(0, \pi]$ , and  $\varphi$  takes values in  $[0, 2\pi)$ . The connection of the coordinates (7.18) with the standard ones  $(u^1, \ldots, u^{n+1})$  in  $\mathbb{R}^{n+1}$  (or  $\mathbb{R}^{n+1}_q$ ) is:

$$u^{1} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \sin \theta_{p-1} \cdots \sin \theta_{2} \sin \theta_{1} \cos \varphi$$

$$u^{2} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \sin \theta_{p-1} \cdots \sin \theta_{2} \sin \theta_{1} \sin \varphi$$

$$u^{3} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \sin \theta_{p-1} \cdots \sin \theta_{2} \cos \theta_{1} \quad (\text{if } p \ge 3)$$

$$\cdots$$

$$u^{2+k} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \sin \theta_{p-1} \cdots \sin \theta_{k+1} \cos \theta_{k} \quad \text{for } 1 \le k \le p-2$$

$$\cdots$$

$$u^{p} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \sin \theta_{p-1} \cos \theta_{p-2} \quad (\text{if } p \ge 3)$$

$$u^{p+1} = \rho \cosh \chi_{q} \cdots \cosh \chi_{1} \cos \theta_{p-1} \quad (\text{if } p \ge 2)$$

$$u^{p+1+1} = \rho \cosh \chi_{q} \cdots \cosh \chi_{2} \sinh \chi_{1} \quad (\text{if } q \ge 3)$$

$$\cdots$$

$$u^{p+1+k} = \rho \cosh \chi_{q} \cdots \cosh \chi_{k+1} \sinh \chi_{k} \quad \text{for } 2 \le k \le q-1 \text{ (if } q \ge 3)$$

$$\cdots$$

$$u^{n} = \rho \cosh \chi_{q} \sinh \chi_{q-1} \quad (n = p+1+q-1)$$

$$u^{n+1} = \rho \sinh \chi_{q}.$$
(7.19)

The comparison between (7.19) and (7.12) reveals that  $(\theta_p, \ldots, \theta_1, \varphi)$  are spherical coordinates on  $\mathbb{S}^p \subset \mathbb{R}^{p+1} \subset \mathbb{R}^{p+1} \times \mathbb{R}^q_q \cong \mathbb{R}^{n+1}_q$ , where  $\cong$  is the 'isomorphic to' sign.

Suppose now that  $n \ge 2$  and  $q \ge 2$ ; hence  $p := n - q \ge 0$  and this case covers the possibility q = n, i.e., the pseudosphere  $\mathbb{S}_n^n$  in the Lorentz manifold  $\mathbb{R}_n^{n+1}$ . Then in  $\mathbb{R}_q^{n+1}$  can be introduced physical coordinates

$$(\rho, \theta_{p-1}, \dots, \theta_1, \chi, \tau_{q-1}, \dots, \tau_1, \varphi)$$

$$(7.20)$$

where for p = 1 (resp. p = 0) the  $\theta$ 's (resp. the  $\theta$ 's and  $\tau_{q-1}$ ) and the terms containing them below should be missing/deleted. The range of  $\rho$  is  $\mathbb{R} \setminus \{0\}$ , the one of  $\chi$  is  $\mathbb{R}$ , the range of the  $\theta$ 's and  $\tau$ 's is  $(0, \pi]$ , and  $\varphi$  takes values in  $[0, 2\pi)$ . The explicit connection of these coordinates with the standard ones  $\{u^1, \ldots, u^{n+1}\}$  on  $\mathbb{R}^{n+1}$  (or  $\mathbb{R}^{n+1}_{a+1}$ ) is (cf. (7.12) and (7.19)):  $u^{p+1} = \rho \sin \theta_{n-1} \cdots \sin \theta_1 \sinh \chi \sin \tau_{q-1} \cdots \sin \tau_2 \sin \tau_1 \cos \varphi$  $u^{p+2} = \rho \sin \theta_{p-1} \cdots \sin \theta_1 \sinh \chi \sin \tau_{q-1} \cdots \sin \tau_2 \sin \tau_1 \sin \varphi$  $u^{p+3} = \rho \sin \theta_{p-1} \cdots \sin \theta_1 \sinh \chi \sin \tau_{q-1} \cdots \sin \tau_2 \cos \tau_1 \qquad (\text{if } q \ge 3)$  $u^{p+k} = \rho \sin \theta_{p-1} \cdots \sin \theta_1 \sinh \chi \sin \tau_{q-1} \cdots \sin \tau_{k-1} \cos \tau_{k-2} \qquad \text{for } 3 \le k \le q$  $u^n = \rho \sin \theta_{n-1} \cdots \sin \theta_1 \sinh \chi \sin \tau_{q-1} \cos \tau_{q-2} \qquad (p+q=n; \text{ if } q \ge 3)$  $u^{n+1} = \rho \sin \theta_{n-1} \cdots \sin \theta_1 \sinh \chi \cos \tau_{q-1}$  $u^1 = \rho \sin \theta_{n-1} \cdots \sin \theta_1 \cosh \chi$  $u^2 = \rho \sin \theta_{p-1} \cdots \sin \theta_2 \cos \theta_1 \qquad (\text{if } p \ge 3)$  $u^k = \rho \sin \theta_{p-1} \cdots \sin \theta_k \cos \theta_{k-1}$  for  $2 \le k \le p-1$ . . . . . . . . . . . . . . . . . . .  $u^{p-1} = \rho \sin \theta_{p-1} \cos \theta_{p-2} \qquad (\text{if } p > 3)$  $u^p = \rho \cos \theta_{n-1}.$ (7.21)

The only peculiarity here is the case  $q = n \geq 2$  and p = 0, when we have (see above; for q = n = 2, the terms containing the  $\tau$ 's should be missing/deleted):

$$u^{2} = \rho \sinh \chi \sin \tau_{q-2} \cdots \sin \tau_{1} \cos \varphi$$

$$u^{3} = \rho \sinh \chi \sin \tau_{q-2} \cdots \sin \tau_{1} \sin \varphi$$

$$u^{4} = \rho \sinh \chi \sin \tau_{q-2} \cdots \sin \tau_{2} \cos \tau_{1} \quad (\text{if } q \ge 3)$$

$$\cdots$$

$$u^{k} = \rho \sinh \chi \sin \tau_{q-2} \cdots \sin \tau_{k-1} \cos \tau_{k-2} \quad \text{for } 4 \le k \le q-1 \quad (7.22)$$

$$\cdots$$

$$u^{n} = \rho \sinh \chi \sin \tau_{q-2} \cos \tau_{q-3} \quad (q = n; \text{ if } q \ge 4)$$

$$u^{n+1} = \rho \sinh \chi \cos \tau_{q-2} \quad (\text{if } q \ge 3)$$

$$u^{1} = \rho \cosh \chi.$$

The coordinates (7.20) are also a modification of the spherical coordinates given via (7.12); indeed, the coordinates  $(\tau_{q-1}, \ldots, \tau_1, \varphi)$  are spherical coordinates on  $\mathbb{S}_q^q \subset \mathbb{R}_q^{q+1} \subset \mathbb{R}_q^{q+1} \times \mathbb{R}^p \cong \mathbb{R}_q^{n+1}$ .

The pseudosphere  $\mathbb{S}_q^n$  has the following representation in any one of the coordinates (7.18) or (7.20) (when they are applicable for given n, q and p = n - q)

$$\mathbb{S}_{q}^{n} = \{ v \in \mathbb{R}_{q}^{n+1} : \rho(v) = \pm r \}.$$
(7.23)

# 7. Examples of normal coordinates

Consider now the case  $n \ge 2$  and  $q \ge 2$  when the coordinates (7.20) are applicable; to exclude the special case (p,q) = (0,n), one can impose the additional restriction  $p \ge 1$ , but this does not change the essence of the calculations and the final results. They induce on the pseudosphere  $\mathbb{S}_q^n$  the coordinate system

$$\{x^{1} = \theta_{p-1}, \dots, x^{p-1} = \theta_{1}, x^{p} = \chi, x^{p+1} = \tau_{q-1}, \dots, x^{p+q-1} = \tau_{1}, x^{n} = \varphi\}.$$
(7.24)

The Riemannian metric g induced on  $\mathbb{S}_q^n$  by  $e_q^{n+1}$  has in  $\{x^i\}$  components  $g_{ij}$ ,  $i, j = 1, \ldots, n$ , forming the diagonal matrix

$$[g_{ij}] = \operatorname{diag}(g_1, \dots, g_n) = r^2 \operatorname{diag}(1, \sin^2 x^1, \sin^2 x^1 \sin^2 x^2, \dots, \sin^2 x^1 \cdots \sin^2 x^{p-1}, \underbrace{0, \dots, 0}_{q\text{-times}}) - r^2 \sin^2 x^1 \cdots \sin^2 x^{p-1} \sinh^2 x^p \times \operatorname{diag}(\underbrace{0, \dots, 0}_{p\text{-times}}, 1, \sin^2 x^{p+1}, \sin^2 x^{p+1} \sin^2 x^{p+2}, \dots, \sin^2 x^{p+1} \cdots \sin^2 x^{p+q-1}).$$
(7.25)

Let  $\nabla$  be the Riemannian connection on  $\mathbb{S}_q^n$  generated by g. The coefficients  $\Gamma^i_{jk}$  of  $\nabla$  in  $\{x^i\}$  can be calculated via equations (4.13) and (4.15). The non-vanishing of them are (cf. equations (7.15), which remain true for  $1 \leq i, j, k \leq n$  with  $i, j, k \neq p$ )

$$\Gamma^{i}_{\ ik}\Big|_{\substack{k < i \\ i,k \neq p}} = \Gamma^{i}_{\ ki}\Big|_{\substack{k < i \\ i,k \neq p}} = \cot x^{k} \qquad \Gamma^{p}_{\ pk}\Big|_{k > p} = \Gamma^{p}_{\ kp}\Big|_{k > p} = \cot x^{k}$$
(7.26a)

$$\Gamma^{i}_{kk}\Big|_{\substack{k>i\\i,k\neq p}} = -\sin x^{i}\cos x^{i}\prod_{l=i+1}^{k-1}\sin^{2}x^{l} \qquad \Gamma^{i}_{pp}\Big|_{i< p} = +\sin x^{i}\cos x^{i}\prod_{l=i+1}^{p-1}\sin^{2}x^{l}$$
(7.26b)

$$\Gamma^{p}_{kk}\big|_{k>p} = -\coth x^{p} \qquad \Gamma^{i}_{ip}\big|_{i>p} = \Gamma^{i}_{pi}\big|_{i>p} = +\coth x^{p}$$
(7.26c)

These formulae show that the pseudospherical coordinates (7.24) on  $\mathbb{S}_q^n$  are nowhere normal for  $\nabla$  as  $\coth s \neq 0$  for all  $s \in \mathbb{R}$ . (This result could be expected as the coordinates employed are 'quite near' to a direct sum of two sets of spherical coordinates, which are suitable of other kind of a symmetry.) However, on the subset  $C := \{v \in \mathbb{S}_q^n : x^i(v) = \frac{\pi}{2} \text{ for } i \neq p\}$  (which is 2-connected 2-manifold), the coordinate system  $\{x^i\}$  is 'very near' to a normal one as only the coefficients (7.26c) of  $\nabla$  are non-vanishing on C. For this reason, one can expect that coordinates normal on a subset of C can be found by a suitable change of the coordinate system  $\{x^i\}$ .

Let us look for geodesic coordinate system  $\{z^i\}$  on  $\mathbb{S}_q^n$  in a neighborhood of a point  $v_0 \in C \subset \mathbb{S}_q^n$ . We set (see (6.13) with  $c_{\dots}^i = 0$ )

$$x^{i}(v) = x^{i}(v_{0}) + z^{i}(v) - \frac{1}{2}\Gamma^{i}{}_{jk}z^{j}(v)z^{k}(v)$$

where  $v \in C$  and  $z^i(v_0) := 0$ .

The equations (7.26) reduce this system of equations to:

$$\begin{aligned} x^{i}(v) &= x^{i}(v_{0}) + z^{i}(v) & \text{for } i p. \end{aligned}$$

We get from here:

$$z^{i}(v) = x^{i}(v) - x^{i}(v_{0}) \quad \text{for } i 
$$z^{p}(v)[1 + \coth(x^{p}(v_{0}))z^{p}(v)]^{2} - [x^{p}(v) - x^{p}(v_{0})][1 + \coth(x^{p}(v_{0}))z^{p}(v)] - \frac{1}{2}\coth(x^{p}(v_{0}))\sum_{k=p+1}^{n} [x^{i}(v) - x^{i}(v_{0})]^{2} = 0$$

$$(7.27b)$$$$

$$z^{i}(v) = \frac{x^{i}(v) - x^{i}(v_{0})}{1 + \coth(x^{p}(v_{0}))z^{p}(v)} \quad \text{for } i > p.$$
(7.27c)

Since (7.27b) is an algebraic cubic equation relative to  $z^p(v)$ , it has at least one real solution that can explicitly be found (in radicals) by the known methods [57, Section 1.8-3], [65]. Substituting this solution into (7.27a) and (7.27c), we can obtain a geodesic coordinate system  $\{z^i\}$  on some subset of the pseudosphere  $\mathbb{S}_q^n$ and which is normal at a point  $v_0 \in C := \{v \in \mathbb{S}_q^n : x^i(v) = \frac{\pi}{2} \text{ for } i \neq p\}.$ 

**Exercise 7.1.** Let  $\mathbb{H}_{q-1}^n := \{v \in \mathbb{R}_q^{n+1} : e_q^{n+1}(v, v) = -r^2\}$  be a hyperbolic space of radius  $r \in \mathbb{R}, r > 0$ , and index q-1 and g be the metric on it induced by  $e_q^{n+1}$ . Suppose the coordinates (7.20) are defined via (7.21) in which the replacements  $\sin \theta_i \mapsto \cosh \theta_i$  for  $i = 1, \ldots, p-1$  and  $\cosh \chi \mapsto \sinh \chi$  are made. Show that  $\mathbb{H}_{q-1}^n := \{v \in \mathbb{R}_q^{n+1} : \rho(v) = \pm r\}$ . Calculate the components of g and the coefficients of  $\nabla$  in the coordinates (7.24). Construct geodesic coordinate system  $\{z^i\}$  on  $\mathbb{H}_{q-1}^n$  using the results obtained and following the above considerations for  $\mathbb{S}_q^n$ .

We shall now return to the coordinates (7.18), defined via (7.19), to study in them coordinates normal on the pseudosphere  $\mathbb{S}_q^n$  when  $n \ge 2$ ,  $q \ge 1$  and  $p = n - q \ge 1$ . First of all, we notice that the set of functions

$$\{x^1 = \theta_{p-1}, \dots, x^{p-1} = \theta_1, x^p = \varphi, x^{p+1} = \chi_q, \dots, x^{p+q-1} = \chi_2, x^n = \chi_1\}$$
(7.28)

provides a coordinate system on  $\mathbb{S}_q^n$  (precisely on any one of its two connected components). The metric g has in these coordinates components  $g_{ij}$  forming the
#### 7. Examples of normal coordinates

diagonal matrix

$$[g_{ij}] = \operatorname{diag}(g_1, \dots, g_n) = r^2 \cosh^2 x^{p+1} \cdots \cosh^2 x^{p+q} \times \operatorname{diag}(1, \sin^2 x^1, \sin^2 x^1 \sin^2 x^2, \dots, \sin^2 x^1 \cdots \sin^2 x^{p-1}, \underbrace{0, \dots, 0}_{q-\operatorname{times}}) - r^2 \operatorname{diag}(\underbrace{0, \dots, 0}_{p-\operatorname{times}}, 1, \cosh^2 x^{p+1}, \cosh^2 x^{p+1} \cosh^2 x^{p+2}, \dots, \underbrace{0, \dots, 0}_{p-\operatorname{times}})$$
(7.29)

The Riemannian connection  $\nabla$  generated by g has the following non-vanishing components in  $\{x^i\}$  (cf. (7.15) and (7.26)):

$$\Gamma_{ik}^{i}|_{\substack{k < i \\ i,k \le p}} = \Gamma_{ki}^{i}|_{\substack{k < i \\ i,k \le p}} = \cot x^{k} \prod_{l=p+1}^{p+q} \cosh^{2} x^{l} 
\Gamma_{kk}^{i}|_{\substack{k > i \\ i,k \le p}} = -\sin x^{i} \cos x^{i} \left( \prod_{l=i+1}^{k-1} \sin^{2} x^{l} \right) \prod_{l=p+1}^{p+q} \cosh^{2} x^{l} 
\Gamma_{ik}^{i}|_{\substack{k < i \\ i,k \ge p+1}} = \Gamma_{ki}^{i}|_{\substack{k < i \\ i,k \ge p+1}} = \coth x^{k} 
\Gamma_{kk}^{i}|_{\substack{i < k \\ k \ge p+1}} = -\coth x^{i} \prod_{l=i+1}^{k-1} \cosh^{2} x^{l} 
\Gamma_{ik}^{i}|_{\substack{i \le p \\ k \ge p+1}} = \Gamma_{ki}^{i}|_{\substack{i \le p \\ k \ge p+1}} = \coth x^{k} 
\Gamma_{kk}^{i}|_{\substack{i \le p \\ k \ge p+1}} = \cosh x^{i} \sinh x^{i} \left( \prod_{l=i+1}^{p+q} \cosh^{2} x^{l} \right) \prod_{l=1}^{k-1} \sin^{2} x^{l}.$$
(7.30)

Since  $\cos \frac{\pi}{2} = \cot \frac{\pi}{2} = 0$  and  $\sinh 0 = \tanh 0 = 0$ , the coordinate system  $\{x^i\}$  is normal for  $\nabla$  on the subset

$$\{v \in \mathbb{S}_q^n : x^1(v) = \dots = x^{p-1}(v) = \frac{\pi}{2} \text{ and } x^{p+1}(v) = \dots = x^{p+q}(v) = 0\} \subset \mathbb{S}_q^n.$$

We can also say that the coordinate system  $\{x^i\}$  is normal for  $\nabla$  along the path  $\gamma \colon [0, 2\pi) \to \mathbb{S}_q^n$  such that  $x^k \circ \gamma = \frac{\pi}{2}$  for  $k = 1, \ldots, p-1, x^k \circ \gamma = 0$  for  $k = p+1, \ldots, p+q$  and  $x^p \circ \gamma = \operatorname{id}_{[0,2\pi)}$ ; we can equivalently express this by writing  $\Gamma^i_{jk} \circ \gamma = 0$  in  $\{x^i\}$ . Since

$$\gamma([0,2\pi)) = \{(r\cos s, r\sin s, \underbrace{0,\dots,0}_{(n-1)\text{-times}}) : s \in [0,2\pi)\} = \mathbb{S}^1 \times (\underbrace{0,\dots,0}_{(n-1)\text{-times}}) \subset \mathbb{R}^{n+1},$$

the circle  $\gamma([0, 2\pi))$  can be obtained by intersection  $\mathbb{S}_q^n$  with the  $(u^1, u^2)$ -plane in  $\mathbb{R}^{n+1}$ .

**Example 7.6 (The torus**  $\mathbb{T}^2$ ). The torus  $\mathbb{T}^2$  of radii  $r_1, r_2 \in \mathbb{R}$ ,  $r_1 > 0$  and  $r_2 > 0$ , in  $\mathbb{R}^3$  is a product of two circles  $\mathbb{S}^1$  and  $\mathbb{S}^1$  of radii  $r_1 > 0$  and  $r_2 > 0$ , respectively,  $\mathbb{T}^2 = \mathbb{S}^1 \times \mathbb{S}^{1,8}$  If  $\varphi_1$  and  $\varphi_2$  are the respective polar coordinates, with range  $[0, 2\pi)$ , on these circles, then

$$\mathbb{T}^{2} = \{ (r \cos \varphi_{1}, r \sin \varphi_{1}, r_{2} \sin \varphi_{2}) \in \mathbb{R}^{3} : \varphi_{1}, \varphi_{2} \in [0, 2\pi) \} 
r := r_{1} + (1 + \cos \varphi_{2}) r_{2}.$$
(7.31)

The standard Euclidean metric on  $\mathbb{R}^3$  (see Example 7.1) induces on  $\mathbb{T}^2$  a metric g whose components in the coordinate system  $\{x^1 = \varphi_1, x^2 = \varphi_2\}$  of  $\mathbb{T}^2$  has components  $g_{ij}, i, j = 1, 2$ , forming the diagonal matrix

$$[g_{ij}] = \operatorname{diag}(r^2, r_2^2). \tag{7.32}$$

The Riemannian connection  $\nabla$  induced by g on  $\mathbb{T}^2$  has the following non-vanishing components in  $\{x^1, x^2\}$  (see (4.13) and (4.15))

$$\Gamma^{1}_{12} = \Gamma^{1}_{21} = -\Gamma^{2}_{11} = -\frac{r_2}{r}\sin x^2.$$
(7.33)

From here immediately follows that the coordinates system  $\{x^1 = \varphi_1, x^2 = \varphi_2\}$  is normal along the path  $\gamma \colon [0, 2\pi) \to \mathbb{T}^2$  with  $x^1 \circ \gamma = \mathrm{id}_{[0,2\pi)}$  and  $x^2 \circ \gamma = 0$  (or  $\gamma(s) = (r \sin s, r \cos s, 0)|_{r=r_1+2r_2} \in \mathbb{T}^2 \subset \mathbb{R}^3$  with  $s \in [0, 2\pi)$ ). Obviously,  $\gamma([0, 2\pi))$  is the the greater circle obtained by intersecting  $\mathbb{T}^2$  with the so-called equatorial plane (i.e.,  $\{(v^1, v^2, 0) \in \mathbb{R}^2 : v^1, v^2 \in \mathbb{R}\}$  in the representation we are using).

Exercise 7.2. Generalize the above results for multidimensional tori, i.e., for

$$\mathbb{T}^n := \underbrace{\mathbb{S}^1 \times \cdots \times \mathbb{S}^1}_{n\text{-times}}$$

where  $n \geq 2$  and the *i*th, i = 1, ..., n, circle has radius  $r_i \in \mathbb{R}$  with  $r_i > 0$ . Hint: (i) parameterize the *i*th circle with a polar angle  $\varphi_i$  with range  $[0, 2\pi)$  and obtain a representation similar to (7.31),<sup>9</sup> (ii) then, in the coordinate system  $\{x^1 = \varphi_1, ..., x^n = \varphi_n\}$  on  $\mathbb{T}^n$ , find the components of the metric g induced on  $\mathbb{T}^n$  by the standard one in  $\mathbb{R}^{n+1}$  and (iii) at last, calculate in  $\{x^i\}$  the components of the Riemannian connection generated by g and investigate the existence of a subset of  $\mathbb{T}^n$  on which  $\{x^i\}$  is normal for this connection.

**Example 7.7 (Surfaces of revolution).** A surface of revolution in  $\mathbb{R}^3$  is obtained by rotating a plane path  $\gamma: J \to \mathbb{R}^2 \subset \mathbb{R}^3$  around an axis in the plane of the path. If

$$(n-2)$$
-times

<sup>&</sup>lt;sup>8</sup>For more details regarding tori, see [42].

<sup>&</sup>lt;sup>9</sup>For  $n \geq 3$ , a representation like  $\mathbb{T}^n := \mathbb{T}^2 \times \mathfrak{S}^1 \times \cdots \times \mathfrak{S}^1$  may turn to be useful.

#### 7. Examples of normal coordinates

 $s \in J$  is the path's parameter and  $\rho(s) \ge 0$  is the distance of  $\gamma(s)$  from the axis, say the third axis in  $\mathbb{R}^3$ , then the surface of revolution is [66, p. 767]

$$S = \{ (\rho(s) \cos \alpha, \rho(s) \sin \alpha, z(s)) \in \mathbb{R}^3 : s \in J, \ \alpha \in [0, 2\pi) \}$$
(7.34)

where z(s) defines  $\gamma$  by  $\gamma(s) = (\rho(s), 0, z(s))$  and  $\alpha$  is the angle of rotation; below we shall suppose that  $\rho$  and z are  $C^2$  functions. The functions  $x^1 : v \to x^1(v) = s$ and  $x^2 : v \to x^2(v) = \alpha$  for  $v = (\rho(s) \cos \alpha, \rho(s) \sin \alpha, z(s))$  provide a coordinate system  $\{x^1, x^2\}$  on S.

The restriction of the Euclidean metric in  $\mathbb{R}^3$  to S results into a tensor field g which has in  $\{x^1, x^2\}$  components  $g_{ij}$ , i, j = 1, 2, such that

$$[g_{ij}] = \operatorname{diag}(\rho'^2 + z'^2, \rho^2), \tag{7.35}$$

where the prime means derivative relative to s, i.e.,  $\rho' := \frac{d\rho}{ds}$  and  $z' := \frac{dz}{ds}$ . Consequently, the tensor field g defines on the set  $S \setminus \{\text{points at which } \rho'(s) = z'(s) = 0\}$  a Riemannian metric g. The metric g induces on S a Riemannian connection  $\nabla$ . The non-vanishing coefficients of  $\nabla$  in  $\{x^1, x^2\}$  are (see (4.13) and (4.15))

$$\Gamma^{1}_{11} = \frac{\rho' \rho'' + z' z''}{\rho'^{2} + z'^{2}} \quad \Gamma^{1}_{22} = -\frac{\rho \rho'}{\rho'^{2} + z'^{2}} \quad \Gamma^{2}_{12} = \Gamma^{2}_{21} = \frac{\rho'}{\rho}.$$
 (7.36)

Thus the coordinate system  $\{x^1, x^2\}$  is normal for  $\nabla$  on the set

$$C := \{ v = (\rho(s) \cos \alpha, \rho(s) \sin \alpha, z(s)) \in S : \rho'(s) = 0, \ z'(s) \neq 0 \text{ and } z''(v) = 0 \}$$
(7.37)

if it is non-empty.

For instance,  $\{x^1, x^2\}$  is normal on the whole surface of revolution S if, and only if, it is a cylinder defined via  $\rho(s) = c$  and z(s) = as + b for some  $a, b, c \in \mathbb{R}$ with  $a \neq 0$  and  $c \neq 0$ . This result can be expected from the considerations in Example 7.1.

However, the set (7.37) may turn to be the empty set in which case the coordinates  $\{x^1, x^2\}$  are nowhere normal on S for  $\nabla$ . Examples of surfaces of revolution on which this happens are provided by the choices  $\rho(s) = e^s, s^3 + s$  and/or  $z(s) = e^s$ .

**Exercise 7.3.** Construct explicitly geodesic coordinates system  $\{z^1, z^2\}$  with  $c_{\dots}^i = 0$  (see (6.13)) from  $\{x^1, x^2\}$  such that  $\{z^1, z^2\}$  is normal for  $\nabla$  at an arbitrarily chosen point  $v_0 \in S$ . (This construction is completely independent from is the set (7.37) empty or non-empty.)

**Exercise 7.4.** Obtain the results of Example 7.3 from the above ones. Hint: a sphere can be obtained by rotating a semi-circle around the axis passing through its ends.

**Example 7.8 (Geodesic coordinates in Schwarzschild spacetime).** The manifold of Schwarzschild is a 4-dimensional (pseudo-)Riemannian manifold endowed with a Riemannian connection  $\nabla$  induced by the Schwarzschild metric g which in suitably

chosen (Schwarzschild) coordinates  $(x^1 = r, x^2 = \theta, x^3 = \varphi, x^4 = ct)$ , c being the velocity of light in vacuum, has components  $[g_{ij}], i, j = 1, \ldots, 4$ , such that<sup>10</sup>

$$[g_{ij}] = \text{diag}(-e^{-\lambda}, -r^2, -r^2\sin^2\theta, e^{\nu})$$
(7.38)

where<sup>11</sup>

$$e^{\nu} := 1 - \frac{r_g}{r} \quad e^{\lambda} := \left(1 - \frac{r_g}{r}\right)^{-1}$$
 (7.39)

with  $r_g$  being a constant (known as the (Schwarzschild) gravitational radius). Using (4.15), we find after some tedious calculations the following non-vanishing coefficients of the Riemannian connection induced by g [58, eq. (83.2)]

$$\Gamma^{1}_{11} = \frac{1}{2} \frac{d\lambda}{dr} \qquad \Gamma^{1}_{22} = -re^{-\lambda} \quad \Gamma^{1}_{33} = -re^{-\lambda} \sin^{2}\theta \quad \Gamma^{1}_{44} = \frac{1}{2}e^{\nu-\lambda}\frac{d\nu}{dr} 
\Gamma^{2}_{12} = \Gamma^{2}_{21} = \frac{1}{r} \quad \Gamma^{2}_{33} = -\sin\theta\cos\theta 
\Gamma^{3}_{13} = \Gamma^{3}_{31} = \frac{1}{r} \quad \Gamma^{3}_{23} = \Gamma^{3}_{32} = \cot\theta 
\Gamma^{4}_{14} = \Gamma^{4}_{41} = \frac{1}{2}\frac{d\nu}{dr},$$
(7.40)

where  $\frac{d\nu}{dr} = +\left(1 - \frac{r_g}{r}\right)^{-1} \frac{r_g}{r^2}$  and  $\frac{d\lambda}{dr} = +\left(1 - \frac{r_g}{r}\right)^{-1} \frac{r_g}{r^2}$ , due to (7.39). The coordinate system  $\left[\sigma^i\right]$  is not normal at all points of the S

The coordinate system  $\{x^i\}$  is not normal at all points of the Schwarzschild spacetime. However, from it can be constricted Riemannian or geodesic coordinates (with origin at any arbitrarily fixed point p) according to the procedures of Section 6. In particular, setting  $c_{i_1...i_n}^i \equiv 0$  for all  $n \geq 3$  in (6.13), we see that the equations

$$x^{1}(q) = x^{1}(p) + z^{1}(q) - \frac{1}{4} \frac{d\lambda}{dr} (z^{1}(q))^{2} + \frac{1}{2} r e^{-\lambda} (z^{2}(q))^{2} + \frac{1}{2} r e^{-\lambda} \sin^{2} (x^{2}(p)) (z^{3}(q))^{2} - \frac{1}{4} e^{\nu - \lambda} \frac{d\nu}{dr} (z^{4}(q))^{2}$$
(7.41a)

$$x^{2}(q) = x^{2}(p) + z^{2}(q) - \frac{1}{r}z^{1}(q)z^{2}(q) + \frac{1}{2}\sin(x^{2}(p))\cos(x^{2}(p))(z^{3}(q))^{2} \quad (7.41b)$$

$$x^{3}(q) = x^{3}(p) + z^{3}(q) - \frac{1}{r}z^{1}(q)z^{3}(q) - \cot(x^{2}(p))z^{2}(q)z^{3}(q)$$
(7.41c)

$$x^{4}(q) = x^{4}(p) + z^{4}(q) - \frac{1}{2} \frac{\mathrm{d}\nu}{\mathrm{d}r} z^{1}(q) z^{4}(q), \qquad (7.41d)$$

<sup>10</sup>For details and the physical significants of the Schwarzschild metric, see (text)books on general relativity and gravitation, like [61, Chapter VII, § 4], [67, Chapter 3], [21, Chapter 8, § 2 and § 8], [68, § 100], or [58, § 83 and § 83].

<sup>&</sup>lt;sup>11</sup>If we admit that  $\lambda$  and  $\nu$  depend on r and do not fix this dependence, we obtain the class of the so-called spherically symmetric metrics [21, Chapter 13, § 5 B]. In particular, the choice  $e^{\nu} = e^{-\lambda} = 1 - \frac{r_g}{r} + \frac{Q^2}{r^2}$ , for some constant  $Q \in \mathbb{R}$  (having a sense of electrical charge), selects the Reissner-Nordström metric [67, Chapter 5, in particular, § 39]. The particular setting (7.39) will not be used below; see, for instance (7.40) and (7.41).

where  $z^i(p) := 0$ , define  $z^1, \ldots, z^4$  as geodesic coordinates with origin at a point p in some neighborhood of p. Equations (7.41c) and (7.41d) yield

$$z^{3}(q) = (x^{3}(q) - x^{3}(p)) \left(1 - \frac{1}{r}z^{1}(q) - \cot(x^{2}(p))z^{2}(q)\right)^{-1}$$
$$z^{4}(q) = (x^{4}(q) - x^{4}(p)) \left(1 - \frac{1}{2}\frac{\mathrm{d}\nu}{\mathrm{d}r}z^{1}(q)\right)^{-1}$$

which, when inserted into (7.41a) and (7.41b), give

$$\begin{split} x^{1}(q) - x^{1}(p) &= z^{1}(q) - \frac{1}{4} \frac{\mathrm{d}\lambda}{\mathrm{d}r} (z^{1}(q))^{2} + \frac{1}{2} r \mathrm{e}^{-\lambda} (z^{2}(q))^{2} \\ &+ \frac{1}{2} r \mathrm{e}^{-\lambda} \sin^{2} (x^{2}(p)) \frac{(x^{3}(q)) - x^{3}(p))^{2}}{\left(1 - \frac{1}{r} z^{1}(q) - \cot(x^{2}(p)) z^{2}(q)\right)^{2}} \\ &- \frac{1}{4} \mathrm{e}^{\nu - \lambda} \frac{\mathrm{d}\nu}{\mathrm{d}r} \frac{(x^{4}(q)) - x^{4}(p))^{2}}{\left(1 - \frac{1}{2} \frac{\mathrm{d}\nu}{\mathrm{d}r} z^{1}(q)\right)^{2}} \\ x^{2}(q) - x^{2}(p) &= \left(1 - \frac{1}{r} z^{1}(1)\right) z^{2}(q) \\ &+ \frac{1}{2} \sin(x^{2}(p)) \cos(x^{2}(p)) \frac{(x^{3}(q) - x^{3}(p))^{2}}{\left(1 - \frac{1}{r} z^{1}(q) - \cot(x^{2}(p)) z^{2}(q)\right)^{2}} \,. \end{split}$$

The latter equation is a cubic algebraic equation relative to  $z^2(q)$  and hence it has at least one real solution that can be found by the known methods [57,65,69]. Substituting this solution into the former equation, one obtains an equation for only  $z^1(q)$  which has at least one real solution  $z^1$  (proof this!). Substituting this last function into the previous expressions for  $z^2$ ,  $z^3$  and  $z^4$ , one derives the required geodesic coordinate system  $\{z^i\}$  as a function of the initial one  $\{x^i\}$ .

**Example 7.9 (The Einstein Universe).** The Einstein Universe [58, §§ 134–141] is a homogeneous and static model of the Universe<sup>12</sup> whose underlying geometrical structure is the (4-dimensional) Einstein manifold. This manifold is a Riemannian 4-dimensional manifold with metric g which, in suitable coordinates  $(x^1 = \chi, x^2 =$  $\theta, x^3 = \varphi, x^4 = ct)$  (c is the velocity of light in vacuum), with respective ranges  $\mathbb{R}$ ,  $(0, \pi]$ ,  $[0, 2\pi)$  and  $(-\infty, +\infty)$ , has components  $g_{ij}, i, j = 1, \ldots, 4$ , forming the diagonal matrix (see, e.g., [58, eq. (138.6)] or [62, eq. (12.135)])

$$[g_{ij}] = \operatorname{diag}(-R^2, -R^2 \sin^2 x^1, -R^2 \sin^2 x^1 \sin^2 x^2, 1)$$
(7.42)

where R is a constant (known as the radius of the spherical space representing the spacial part of the Einstein Universe). This metric induces a Riemannian

 $<sup>^{12}</sup>$  There are possible only three versions of a homogeneous and static Universe in general relativity [58, § 134]: Einstein, de Sitter and Minkowski spacetimes.

connection  $\nabla$  whose nonvanishing coefficients in  $\{x^i\}$  are (see (4.13) and (4.15); cf. (7.15))

$$\Gamma^{1}_{22} = -\sin x^{1} \cos x^{1} \quad \Gamma^{1}_{33} = -\sin x^{1} \cos x^{1} \sin^{2} x^{2}$$

$$\Gamma^{2}_{12} = \Gamma^{2}_{21} = \cot x^{1} \quad \Gamma^{2}_{33} = -\sin x^{2} \cos x^{2}$$

$$\Gamma^{3}_{13} = \Gamma^{3}_{31} = \cot x^{1} \quad \Gamma^{3}_{23} = \Gamma^{3}_{32} = \cot x^{2}.$$

$$(7.43)$$

(Notice, these coefficients are exactly the coefficients (7.15) of the sphere  $\mathbb{S}^3$  in  $\mathbb{R}^4$ .) Consequently, the coordinate system  $\{x^i\}$  is normal for  $\nabla$  on the 2-dimensional subspace  $\{v: x^1(v) = x^2(v) = \frac{\pi}{2}\}$  of the Einstein Universe.

**Example 7.10 (The de Sitter Universe).** The de Sitter Universe [58, §§ 136, 142–145], [62, § 12.7] is also a homogeneous and static model of the Universe. It is based on a 4-dimensional Riemannian manifold, called the de Sitter manifold, whose metric g in suitable coordinates  $(x^1 = \chi, x^2 = \theta, x^3 = \varphi, x^4 = ct)$  has components  $g_{ij}$  such that [58, eq. (142.3)]

$$[g_{ij}] = \operatorname{diag}(-R^2, -R^2 \sin^2 x^1, -R^2 \sin^2 x^1 \sin^2 x^2, \cos^2 x^1)$$
(7.44)

for some constant R. Applying (4.13) and (4.15) (see also (7.43)), one can verify that the Riemannian connection  $\nabla$  induced by g has in  $\{x^i\}$  the following non-vanishing coefficients (cf. (7.43))

$$\Gamma^{1}_{22} = -\sin x^{1} \cos x^{1} \quad \Gamma^{1}_{33} = -\sin x^{1} \cos x^{1} \sin^{2} x^{2}$$

$$\Gamma^{1}_{44} = -\frac{1}{R^{2}} \sin x^{1} \cos x^{1}$$

$$\Gamma^{2}_{12} = \Gamma^{2}_{21} = \cot x^{1} \quad \Gamma^{2}_{33} = -\sin s^{2} \cos x^{2}$$

$$\Gamma^{3}_{13} = \Gamma^{3}_{31} = \cot x^{1} \quad \Gamma^{3}_{23} = \Gamma^{3}_{32} = \cot x^{2}$$

$$\Gamma^{4}_{14} = \Gamma^{4}_{41} = -\cot x^{1}.$$

$$(7.45)$$

Evidently, the coordinate system  $\{x^i\}$  is normal for  $\nabla$  on the 2-dimensional submanifold  $\{v: x^1(v) = x^2(v) = \frac{\pi}{2}\}$  of the de Sitter Universe.

Via an appropriate change  $\{x^i\} \mapsto \{y^i\}$  of the local coordinates, one can transform the metric's components  $g_{ij}$  to  $g_{ij}^y$  such that (see, e.g., [58, § 142, eq. (142.11)] or [62, eq. (12.161)])

$$[g_{ij}] = \operatorname{diag}(-e^{2\varkappa y^4}, -e^{2\varkappa y^4}, -e^{2\varkappa y^4}, y^4)$$
(7.46)

for some number  $\varkappa \neq 0$ . The non-vanishing coefficients  ${}^{y}\Gamma^{i}{}_{ik}$  of  $\nabla$  in  $\{y^{i}\}$  are

$${}^{y}\Gamma^{i}_{i4} = {}^{y}\Gamma^{i}_{4i} = -\varkappa \quad \text{for } i = 1, 2, 3$$
  
$${}^{y}\Gamma^{4}_{kk} = \varkappa e^{2\varkappa y^{4}} \quad \text{for } k = 1, 2, 3.$$
(7.47)

Thus the coordinates  $\{y^i\}$  are nowhere normal for  $\nabla$  regardless that the metric looks 'simpler' in them.

#### 7. Examples of normal coordinates

The above conclusion is due to the main difference between the coordinates systems  $\{x^i\}$  and  $\{y^i\}$ : the former reflects the symmetry of the de Sitter spacetime (its spacial part is spherically symmetric) while the latter one 'tends to represent' this space 'as near as possible' to the flat Minkowski space  $\mathbb{R}_3^4$ .

**Exercise 7.5.** Using (6.13) with  $c_{\dots}^i = 0$ , construct from  $\{y^i\}$  a geodesic coordinate system  $\{z^i\}$  normal for  $\nabla$  at some fixed point.

**Example 7.11 (Non-static spacially homogeneous Universes).** If we admit the constant R to change with the time,  $R^2 = R_0^2 e^{f(x^4)}$  for some  $R_0 \in \mathbb{R}$  and  $C^1$  function f, in the Einstein metric (7.42), we obtain a 4-dimensional Riemannian manifold with a metric g which in some fixed coordinates system  $\{x^i\}$  has components  $g_{ij}$ ,  $i, j = 1, \ldots, 4$ , forming the diagonal matrix

$$[g_{ij}] = \operatorname{diag}(-R_0^2 \mathrm{e}^{f(x^4)}, -R_0^2 \mathrm{e}^{f(x^4)} \sin^2 x^1, -R_0^2 \mathrm{e}^{f(x^4)} \sin^2 x^1 \sin^2 x^2, 1)$$
  
=: diag(g\_1, g\_2, g\_3, g\_4). (7.48)

This manifold represents a non-static (and spacially homogeneous) model of the Universe in the general theory of relativity [58, §§ 147–149; eq. (149.7)], [62, § 12.8]. The Riemannian connection generated by g has in  $\{x^i\}$  the following non-vanishing coefficients (see (4.13) and (4.15); cf. (7.43))

$$\Gamma^{1}_{22} = -\sin x^{1} \cos x^{1} \quad \Gamma^{1}_{33} = -\sin x^{1} \cos x^{1} \sin^{2} x^{2} \quad \Gamma^{1}_{14} = \Gamma^{1}_{41} = \frac{1}{2} f'(x^{4})$$

$$\Gamma^{2}_{12} = \Gamma^{2}_{21} = \cot x^{1} \quad \Gamma^{2}_{33} = -\sin s^{2} \cos x^{2} \qquad \Gamma^{2}_{24} = \Gamma^{2}_{42} = \frac{1}{2} f'(x^{4})$$

$$\Gamma^{3}_{13} = \Gamma^{3}_{31} = \cot x^{1} \quad \Gamma^{3}_{23} = \Gamma^{3}_{32} = \cot x^{2} \qquad \Gamma^{3}_{34} = \Gamma^{3}_{43} = \frac{1}{2} f'(x^{4})$$

$$\Gamma^{4}_{kk} = -\frac{1}{2} f'(x^{4}) g_{k} \qquad \Gamma^{i}_{i4} = \Gamma^{i}_{4i} = \frac{1}{2} f'(x^{4}) \qquad \text{for } i, k = 1, 2, 3,$$

$$(7.49)$$

where  $f'(x^4) := \frac{df(x^4)}{dx^4}$ . Therefore the function g (which together with the cosmological constant determine the pressure and energy density [58, § 150, eqs. (150.7) and (150.8)]) is responsible for the existence of set(s) on which the coordinate system  $\{x^i\}$  is normal. Indeed, if the equation  $f'(x^4) = 0$  has a real solution  $x_0^4$ ,  $f'(x_0^4) = 0$ , then  $\{x^i\}$  is normal on the 1-dimensional submanifold  $\{v : x^1(v) = x^2(v) = \frac{\pi}{2}, x^4(v) = x_0^4\}$ .

**Exercise 7.6.** If  $f'(a) \neq 0$  for all  $a \in \mathbb{R}$ , construct from  $\{x^i\}$  a coordinate system  $\{z^i\}$  which is normal at a fixed spacetime point.

**Example 7.12 (The light cone in Minkowski spacetime).** The 4-dimensional manifold  $M_4 = \mathbb{R}^4_3$  (or the isomorphic to it manifold  $\mathbb{R}^4_1$ ) is know as the Minkowski spacetime and it is the geometrical base for the special theory of relativity and the whole relativistic physics [20, 21, 51, 59–64]. The causal structure of the physical theories based on it is determined by the so-called light cone  $C_3^3$  which is defined as a 'pseudosphere' of index 3 and zero radius in it,

$$C_3^3 = \{ v \in \mathbb{R}_3^4 : e_3^4(v) = 0 \}$$
  
=  $\{ (v^1, v^2, v^3, v^4) \in \mathbb{R}_3^4 : (v^1)^2 - (v^2)^1 - (v^3)^2 - (v^4)^2 = 0 \}.$  (7.50)

The light cone is a 3-dimensional 1-connected  $C^{\omega}$  manifold whose geometry is very well explored due to its physical significants. From the view-point of the considerations in Example 7.5, it is described by the choice n = 3 (n + 1 = 4), q = 3, p = 0, and r = 0. The metric  $e_3^4$ , when restricted to  $C_3^3$ , induces a (1-time) degenerate tensor field on  $C_3^3$  which can further be restricted on other subsets of  $C_3^3$  to give a Riemannian metric g, which generates a connection  $\nabla$  on them.

The different kinds of pseudospherical coordinates in  $\mathbb{R}^4_3$  are not suitable for studding the light cone due to its vanishing radius when it is consider as a pseudosphere. For instance, in the coordinates  $(\rho, \chi, \tau_1, \varphi)$  in  $\mathbb{R}^4_3$ , provided by (7.22) and such that

$$u^{2} = \rho \sinh \chi \sin \tau_{1} \cos \varphi \quad u^{4} = \rho \sinh \chi \cos \tau_{1}$$
$$u^{3} = \rho \sinh \chi \sin \tau_{1} \sin \varphi \quad u^{1} = \rho \cosh \chi,$$

we have  $C_3^3 = \{v \in \mathbb{R}^4_3 : \rho(v) = 0\}$  and the components of  $e_3^4$  in it are given by the diagonal matrix diag $(1, -\rho^2, -\rho^2 \sinh \chi, -\rho^2 \sinh \chi \sin \tau_1)$ . Therefore  $u^1|_{C_3^3} \equiv 0$ and the 'metric' induced by  $e_3^4$  will have components forming the degenerate matrix diag(1, 0, 0, 0); the cause for this is that the change  $(u^1, u^2, u^3, u^4) \mapsto (\rho, \chi, \tau_1, \varphi)$ is degenerate on the light cone.

Below we shall investigate two concrete coordinates systems in  $\mathbb{R}^4_3$  which are suitable for description of the light cone  $C_3^3$ .

Define on  $\mathbb{R}^4_3$  coordinates  $(H, R, \theta, \varphi)$ , with respective ranges  $\mathbb{R}$ ,  $(0, \infty)$ ,  $(0, \pi]$  and  $[0, 2\pi)$ , such that

$$u^{2} = R \sin \theta \cos \varphi$$
  

$$u^{3} = R \sin \theta \sin \varphi$$
  

$$u^{4} = R \cos \theta$$
  

$$u^{1} = H.$$
  
(7.51)

The Jacobian of the change to the new coordinates equals  $-R^2 \sin \theta \neq 0$ , so that they are well defined. These coordinates agree with the splitting  $\mathbb{R}_3^4 = \mathbb{R}^1 \times \mathbb{R}_3^3$  and  $(R, \theta, \varphi)$  are spherical coordinates in  $\mathbb{R}^3$ . In them  $C_3^3 = \{v \in \mathbb{R}_3^4 : H = -R \text{ or } H = +R\}$  and the components of  $e_3^4$  form the diagonal matrix diag $(1, -1, -R^2, -R^2 \sin \theta)$ . The set of functions

$$\{x^1 = R, x^2 = \theta, x^3 = \varphi\}$$
(7.52)

#### 7. Examples of normal coordinates

is a coordinate system on the future part  $\{v \in \mathbb{R}_3^4 : H = +R\}$  or the past part  $\{v \in \mathbb{R}_3^4 : H = -R\}$  of the light cone  $C_3^3$ . The components of the tensor g obtained by restricting  $e_3^4$  to  $C_3^3$  has in  $\{x^1, x^2, x^3\}$  components  $g_{ij}$ , i, j = 1, 2, 3, such that<sup>13</sup>

$$[g_{ij}] = \operatorname{diag}(0, -(x^1)^2, -(x^1)^2 \sin^2 x^2).$$
(7.53)

Consequently g is degenerate and hence does not define on the light cone a Riemannian metric; one can prove that this result is independent of the particular coordinates used. However, on the set  $C_3^3|_{R=R_0} = \{v \in \mathbb{R}_3^4 : H = \pm R_0, R = R_0\} \subset C_3^3$ , where  $R_0$  is a fixed positive real number, the restriction of  $e_3^4$  or of g results into a Riemannian metric  $\bar{g}$  which in the coordinates  $\{x^1 = \theta, x^2 = \varphi\}$  on  $C_3^3|_{R=R_0}$  has components  $\bar{g}_{ij}$ , i, j = 1, 2 such that

$$[\bar{g}_{ij}] = \operatorname{diag}(-R_0^2, -R_0^2 \sin^2 x^1) = -\operatorname{diag}(R_0^2, R_0^2 \sin^2 x^1).$$
(7.54)

Consequently  $\bar{g} = -g_{\mathbb{S}^2}$ , where  $g_{\mathbb{S}^2}$  is the metric on the 2-sphere  $\mathbb{S}^2$  of radius  $R_0$  in  $\mathbb{R}^3$  (see (7.10)). So, if  $\bar{\nabla}$  is the Riemannian connection on  $C_3^3|_{R=R_0}$  induced by  $\bar{g}$ , then its non-vanishing coefficients in  $\{x^1, x^2\}$  are (7.11). Hence  $\{x^1, x^2\}$  is normal along the path  $\gamma \colon [0, 2\pi) \to C_3^3|_{R=R_0}$  such that  $x^1 \circ \gamma = \frac{\pi}{2}$  and  $x^2 \circ \gamma = \mathrm{id}_{[0, 2\pi)}$ . Geometrically  $C_3^3|_{R=R_0}$  is the intersection of  $C_3^3$  with the 3-plane  $u^1 = \pm R_0$ ; physically this 3-plane represents the spacial part of the spacetime at the moment  $t = \pm r_0/c$ , c being the velocity of light in vacuum.

Let us now look on the light cone  $C_3^3$  from the view-point of coordinates  $(\rho, \chi, R, \varphi)$  in  $\mathbb{R}^4_3$  such that

$$u^{1} = \rho \cosh \chi$$
  

$$u^{2} = \rho \sinh \chi$$
  

$$u^{3} = R \cos \varphi$$
  

$$u^{4} = R \sin \varphi.$$
  
(7.55)

The range of  $\rho$  is  $\mathbb{R} \setminus \{0\}$ , the one of  $\chi$  is  $\mathbb{R}$ , R ranges in  $(0, \infty)$ , and  $\varphi$  takes values in  $[0, 2\pi)$ . The Jacobian of the change to the new coordinates equals  $\rho R$  and hence they are well defined. The light cone has in these coordinates the representation  $C_3^3 = \{v \in \mathbb{R}_3^4 : \rho^2 = R^2\}$  and the components of  $e_3^4$  in them form the diagonal matrix diag $(1, -\rho^2, -1, -R^2)$ . The set  $\{\rho, \chi, \varphi\}$  is a coordinates system on  $C_3^3$  in which the tensor g, to which  $e_3^4$  reduces on  $C_3^3$ , has components  $g_{ij}$ ,  $i, j = \rho, \chi, \varphi$ , such that

$$[g_{ij}] = \operatorname{diag}(0, -\rho^2, -\rho^2). \tag{7.56}$$

(The same result can be obtained in the coordinate system  $\{\pm R, \chi, \varphi\}$ .) Thus g is degenerate and does not define a Riemannian metric on the light cone. However, when restricted to the set  $C_3^3|_{\rho=\rho_0} = \{v \in \mathbb{R}^4_3 : \rho(v) = \rho_0 \text{ and } R = |\rho_0|\} \subset C_3^3$  for

<sup>&</sup>lt;sup>13</sup>To derive (7.53), write  $e_3^4$  as a differentials form,  $e_3^4 = dH^2 - dR^2 - R^2(d\theta^2 + \sin^2\theta d\varphi^2)$  and restrict it to the light cone where  $H^2 = R^2$ .

a fixed number  $\rho_0 \in \mathbb{R} \setminus \{0\}$ , this tensor (or  $e_3^4$ ) reduces to a Riemannian metric  $\bar{g}$  whose components  $\bar{g}_{ij}$ , i, j = 1, 2, in  $\{x^1 = \chi, x^2 = \varphi\}$  are given by

$$[\bar{g}_{ij}] = \operatorname{diag}(-\rho_0^2, -\rho_0^2) = -\rho_0^2 \operatorname{diag}(1, 1).$$
(7.57)

Hence  $\bar{g} = -\rho_0^2 e_0^2$ , where  $e_0^2$  is the standard Euclidean metric in  $\mathbb{R}_0^2 = \mathbb{R}^2$  (see Example 7.1). In this way, we see that, if  $\bar{\nabla}$  is the Riemannian connection induced by  $\bar{g}$  on  $C_3^3|_{\rho=\rho_0}$ , then the coordinate system  $\{x^1 = \chi, x^2 = \varphi\}$  is everywhere normal on  $C_3^3|_{\rho=\rho_0}$  for  $\bar{\nabla}$ . This conclusion is clear also from the geometrical interpretation of  $C_3^3|_{\rho=\rho_0}$  as an intersection of the '3-cylinder'  $\mathbb{R}_1^2 \times \mathbb{S}^1 \subset \mathbb{R}_3^4$ , of 'radius'  $\sqrt{-1}\rho_0$  with the pseudosphere  $\mathbb{S}_1^2 \subset \mathbb{S}_1^2 \times \mathbb{R}_2^2 \subset \mathbb{R}_3^4$  of radius  $\rho_0$ .

**Exercise 7.7.** Generalize the above considerations for a general cone  $C_q^n = \{v \in \mathbb{R}_q^{n+1} : e_q^{n+1}(v,v) = 0\}$  of index  $q, 1 \leq q \leq n$ , in  $\mathbb{R}_q^{n+1}$ . (Hint: Relying on the  $\mathbb{R}_q^{n+1} \cong \mathbb{R}^{p+1} \times \mathbb{R}_q^q$  for p = n-q, use standard coordinates on  $\mathbb{R}^{p+1}$  and spherical coordinates on  $\mathbb{R}_q^q$ .) Show that the different pseudospherical coordinates in  $\mathbb{R}_q^{n+1}$  are not applicable for investigating the cones. Find different analogues in this case of the coordinates given via (7.55).

# 8. Terminology 1: Bases and frames. Holonomicity

It is beyond any doubt, all authors define a *basis* of an *n*-dimensional,  $n \in \mathbb{N}$ , vector space as a set of *n* linearly independent vectors in it (see, for instance, [1,12, resp. Section 1.5 and p. 9] or any book on vector spaces<sup>1</sup>). Sometimes the term 'frame' is used as a synonym of basis [13, p. 8]. But when (tangent) vector fields over a subset  $U \subseteq M$  of a differentiable manifold of dimension *n* are concerned, the situation slightly changes: a frame (*n*-frame, comoving (or moving) frame, or vielbein)<sup>2</sup> is defined as a set of *n* linearly independent vector fields over *U* [5, Sections 2.2 and 7.6] (see also [7]). It is easily seen that a frame over *U* is, in fact, a basis in the module  $\mathfrak{X}(U)$  of vector fields on *U* over the ring (algebra)  $\mathfrak{F}(U)$  of real or complex, if the complex case is considered, functions on *U* [15, p. 10]. Furthermore, when restricted to a single point  $x \in U$  (or if  $U = \{x\}$ ), any frame becomes a basis of  $T_x(M)$ . An evident example of a frame on a coordinate neighborhood *U* is the frame  $\{\partial/\partial x^i\}$  associated with some coordinates  $\{x^i\}$  on *U* (see [15, p. 10], [5, p. 10], or Section 2).

Taking into account the above-said, as well as the argumentation in the references cited, we accept the following definitions (cf. Subsection 2.3). If a single vector space of finite dimension n is concerned, the concepts 'basis' and 'frame' are synonyms and mean an arbitrary set on n linearly independent vectors in it. A *frame* on (over)  $U \subseteq M$ , M being manifold, is a set of dim M vector fields on U

 $<sup>^1\</sup>mathrm{Note},$  the given in [12, p. 9] definition (of Hamel) basis covers the case of infinite, countable or not, dimension too.

<sup>&</sup>lt;sup>2</sup>For n = 3 (resp. n = 4) a frame is often called triad (resp. tetrad) or dreibein (resp. vierbein) depending one prefers Greek or German.

such that at each  $x \in U$  they form a basis in  $T_x(M)$ , i.e., it is a field of bases over U.<sup>3</sup> So, a frame at (on, over) x and basis in  $T_x(M)$  are synonyms, i.e., equivalent concepts. Further, in Chapter IV, we shall see that the notion of 'frame' admits a natural generalization on vector bundles.<sup>4</sup>

Let  $\{E_i, i = 1, ..., \dim M\}$  be a frame on an open subset U in M. It is called *holonomic* (resp. *anholonomic*) [19, p. 99ff], [11] if the basic fields  $E_i$  commute (resp. do not commute), i.e., if  $C_{jk}^i = 0$  (resp. if  $C_{jk}^i \neq 0$  for at least one triple (i, j, k)) where  $C_{ik}^i$  define the commutators

$$[E_j, E_k]_{-} := E_j \circ E_k - E_k \circ E_j =: C^i_{jk} E_i.$$

$$(8.1)$$

The functions  $C_{jk}^i$  are known as structure functions of the frame  $\{E_i\}$  (of the Lie algebra of vector fields). The frame  $\{\partial/\partial x^i\}$ , associated with local coordinates  $\{x^i\}$  in a neighborhood U, is holonomic,

$$\left[\frac{\partial}{\partial x^i}, \frac{\partial}{\partial x^j}\right] \equiv 0. \tag{8.2}$$

Therefore, if we expand a frame  $\{E_i\}$  on U over  $\{\partial/\partial x^i\}$ ,  $E_i = B_i^j \partial/\partial x^j$  with a  $B := [B_i^j]$  being nondegenerate matrix-valued  $C^1$  function on U, the holonomicity of  $\{E_i\}$  depends entirely on the functions

$$C_{jk}^{i} = (B^{-1})_{m}^{i} \left( B_{j}^{l} \frac{\partial}{\partial x^{l}} B_{k}^{m} - B_{k}^{l} \frac{\partial}{\partial x^{l}} B_{j}^{m} \right)$$
  
$$= (B^{-1})_{m}^{i} 2B_{[j}^{l} \frac{\partial}{\partial x^{|l|}} B_{k]}^{m} = (B^{-1})_{m}^{i} 2E_{[j} B_{k]}^{m}$$
  
$$= 2B_{[j}^{m} E_{k]} (B^{-1})_{m}^{i} = -2B_{j}^{m} B_{k}^{n} \frac{\partial}{\partial x^{[m}} (B^{-1})_{n]}^{i} \quad (8.3)$$

where over indices included in square brackets antisymmetrization is assumed, which up to a constant, equal to (-2), coincides with the components of the (field of the) object of anholonomicity (anholonomy) [19, p. 100], [70, Chapter IV, § 7].<sup>5</sup>

From (8.1) one can easily find the transformation law of the functions  $C_{jk}^i$ under a change  $\{E_i\} \mapsto \{E'_i = A_i^j E_j\}, A = [A_i^j]$  being non-degenerate matrixvalued function, of the frame  $\{E_i\}$ :

$$C_{jk}^{i} \mapsto C_{jk}^{\prime \, i} = (A^{-1})_{l}^{i} [A_{j}^{m} A_{k}^{n} C_{mn}^{l} + 2E_{[j}^{\prime} (A_{k}^{l})]$$

$$(8.4)$$

<sup>&</sup>lt;sup>3</sup>Said differently, in terms of vector bundles, a frame on U is a section of the frame bundle (= bundle of frames) restricted to U. The frame bundle over M is (cf. [23, 27], see also [5, p. 139 and Section 9.11]) a vector bundle with base M and whose fibre over  $x \in M$  consists of all bases in  $T_x(M)$ .

<sup>&</sup>lt;sup>4</sup>The just presented definition of frame corresponds to the case of the tangent bundle  $(T(M), \pi, M)$  over M (see Subsection IV.2.4). Here  $T(M) := \bigcup_{x \in M} T_x(M)$  and  $\pi : T(M) \to M$  is such that  $\pi(X) := x$  if  $X \in T_x(M)$ .

<sup>&</sup>lt;sup>5</sup>The (index) notation of J.A. Schouten is practically out of usage our days.

where  $[E'_j, E'_k]_{-} := E'_j \circ E'_k - E'_k \circ E'_j =: C'^i_{jk} E_i$ . Notice, (8.3) corresponds to (8.4) with  $E_i = \frac{\partial}{\partial x^i}$ ,  $C^i_{jk} = 0$ , A = B, and  $C^i_{jk}$  for  $C'^i_{jk}$ .

We have to emphasize that the holonomicity is a property of the frames, not of a particular basis. This is clear from (8.3) in which partial derivatives with respect to the local coordinates (or to frame's vector fields) enter. The holonomicity characterizes the frames with respect to the local existence of coordinates the associated to which (holonomic) frame coincides with the initial one (in some open subset of the manifold). Evidently (see (8.2)), only the holonomic frames can admit such coordinates. The opposite assertion is (locally) true too.

**Proposition 8.1.** A frame  $\{E_i\}$  on a neighborhood U is holonomic iff locally exist coordinates  $\{y^i\}$  such that locally  $E_i = \partial/\partial y^i$ .

Proof. Let  $\{E^i\}$  be the frame dual to  $\{E_i\}$ ,<sup>6</sup> i.e.,  $E^i = (E_i)^*$ ,  $E^i(E_j) := \delta^i_j$ . Hence, if locally  $E_i = A^j_i \partial/\partial x^j$ , then  $E^j = (A^{-1})^j_k dx^k$ ,  $dx^k := (\partial/\partial x^k)^*$ . We are interested in the existence of local coordinate  $\{y^i\}$  such that  $E_i = \partial/\partial y^i$  which is equivalent to  $E^i = dy^i$ , i.e.,  $E^i$  must be locally exact 1-forms [13, p. 108], [2, p. 55]. A necessary and sufficient condition for this is  $dE^i = 0$  (cf. Poincaré's lemma [2, p. 55], [13, p. 121]), i.e.,

$$\frac{\partial}{\partial x^l} \left( A^{-1} \right)_k^i - \frac{\partial}{\partial x^k} \left( A^{-1} \right)_l^i = 0$$

which expresses the fact that we must have  $\partial y^i / \partial x^k = (A^{-1})_k^i$ . A simple matrix computation verifies that the last conditions hold iff  $C_{ik}^i = 0$ .

Sometimes a frame is called holonomic if it coincides with the coordinate frame generated by some local coordinates. Respectively, a frame is called *locally holonomic* if every point in its domain has a neighborhood in which the frame coincides with the coordinate frame generated by some local coordinates in this neighborhood. By Proposition 8.1, the concept 'holonomic frame' in our sense and 'locally holonomic frame' in the above sense are identical. and will be used further. The difference is that the latter emphasizes on the local link between the holonomic frames and the local coordinates.

Proposition 8.1 suggests a definition of the property 'holonomicity' on arbitrary set U, not only on neighborhoods.

**Definition 8.1.** A frame  $\{E_i\}$  defined on a neighborhood  $\overline{U}$  containing or equal to a set  $U, \overline{U} \supseteq U$ , is called *holonomic on* U, if there exists a chart (V, x) of M such that  $V \supseteq U$  and  $E_i|_U = \frac{\partial}{\partial x^i}|_U$ , i.e., the frame  $\{E_i\}$  coincides on U with the coordinate frame generated by some local coordinates.

If U is a neighborhood and  $\overline{U} = U$ , by virtue of Proposition 8.1, this definition agrees with the afore-presented one.

<sup>&</sup>lt;sup>6</sup>It is a frame in the cotangent bundle  $(T^*(M), \pi, M)|_U$  with  $\pi^{-1}(p) := T_p^*(M), p \in M$  (see Subsection IV.2.4 below).

**Definition 8.2.** A frame  $\{E_i\}$  defined on an open subset of M containing or equal to a given set U is called *locally holonomic on* U if for every  $q \in U$  exists a chart (V, x) of M such that  $V \ni q$  and  $E_i|_{U \cap V} = \frac{\partial}{\partial x^i}|_{U \cap V}$ , i.e., on U the frame  $\{E_i\}$  locally coincides with the frame assigned to some local coordinates.

If U is a neighborhood, the local holonomicity agrees with the holonomicity, as defined above. But if U is not a neighborhood, e.g., if it is a submanifold of M, the local holonomicity is weaker concept than the holonomicity.

The concept of a frame is especially useful in physics where the concept of a basis (of a single space) is rarely utilized. More precisely, in physics the term frame of reference or reference frame is applied. This is a set of (real or mathematical) objects with respect to which is described the behavior, e.g., the evolution in time, of some physical system(s). Such reference objects are practically always defined on some subset of the space(-time) which is frequently a neighborhood or a curve (path) and very seldom chosen as a single point. Taking into account that the frames are used for referring (describing) of vector fields,<sup>7</sup> which in turn may represent some physical fields, the analogy between frames and reference frame there corresponds the *mathematical* concept frame (in some vector bundle). (See, e.g., [71], where a good analysis of different kinds of reference frames can be found.)

# 9. Conclusion

This chapter, as we saw, has an introductory character. It does not contain new original material except the implicit description of the frames normal at a single point of a Riemannian manifold (Section 6) and partially the investigation of normal frames/coordinates in Section 7.

After the presentation of the minimum knowledge from the differential geometry, required for our work, we started with the initial ideas concerning normal frames and coordinates. The basic results here are: only torsionless linear connections (may) admit normal coordinates; if the torsion is non-zero, normal frames (may) exit, but normal coordinates do not. If normal frames exist, they are parallel and are connected with linear transformations whose matrices are constant under the action of their basic vector fields.

The Riemannian and geodesic coordinates, which are normal at their origins, were pointed out as first examples of normal coordinates.

<sup>&</sup>lt;sup>7</sup>In the general case they are replaced with sections of vector bundles.

# Chapter II

# Existence, Uniqueness and Construction of Normal Frames and Coordinates for Linear Connections

An indepth investigation of existence, uniqueness construction of frames and coordinates and normal for linear connections on manifolds is given. Detailed review of the literature dealing with normal coordinates is presented. Some proofs are improved/generalized which entails a number of new results. Similar problems in the case with non-zero torsion are studied. Main results: For arbitrary (resp. torsionless) connections frames (resp. coordinates) normal at a single point and along path exist; they exist on submanifolds of higher dimensions iff the parallel transport along paths lying in them is path-independent. Complete constructive description of all, if any, frames and coordinates normal for arbitrary linear connections.

 $\heartsuit$ 

# 1. Introduction

This chapter presents a complete exploration of the problems linked to the existence, uniqueness, and construction of normal coordinates and frames for manifolds endowed with a linear connection, with or without torsion. The review of the literature dealing with normal coordinates is mixed with new results. Such are, first of all, the ones concerning normal frames, connections with non-vanishing torsion, and the complete constructive description of the normal coordinates, if any.

The methods for description of normal coordinates/frames on Riemannian manifolds can *mutatis mutandis* be transferred on arbitrary manifolds, real or complex  $(\mathbb{K} = \mathbb{R}, \mathbb{C})$ ,<sup>1</sup> endowed with linear connection. The possibility for this is hidden in the fact that the existence and properties of the normal coordinates/frames on a Riemannian manifold is intrinsically connected with the properties of the Christoffel symbols, i.e., with the Riemannian connection, not with the particular metric generating them. After this situation was clearly understood, somewhere in 1922–1927 [50, 72–74] (see [19, p. 155] for other references), the attention of the mathematicians, working in the field, was completely switched to the exploration of normal coordinates on manifolds with linear connections. Practically only the symmetric (torsionless) case has been investigate (see the comments after Remark I.5.4 on page 41). Some random works, like [44,75], dealing with the asymmetric case (non-zero torsion) do not add nothing new as they simply note that the symmetric parts (I.3.9) of the connection coefficients (in coordinate frame) are coefficients of a symmetric linear connection to which the known results for torsionless connections are applicable.

Below in this chapter, in more or less modern terms and notation, are reviewed all results concerning the existence of normal coordinates/frames on manifolds endowed with symmetric linear connection. It contains a number of original new results too.

At first (Section 2), we concentrate on coordinates or frames normal at a single point. We present the known classical methods in this field [18, 19, 70] and then, modifying the methods that will be given in Chapter III in full generality, we present a full description of these coordinates/frames.

In Section 3 the attention is turned on the coordinates or frames normal along paths without self-intersections. For symmetric linear connections, we give a detailed description of the Fermi coordinates as the first known coordinates of this kind with [19] being our basic reference. Then, modifying the methods developed for similar but more general problems (see Chapter III and [76]), we derive a complete description of all coordinates or frames normal along paths without self-intersections or along locally injective paths in manifolds with symmetric or, respectively, arbitrary linear connections.

Several pages deal with problems concerning normal frames and coordinates on submanifolds with maximum dimensionality (Section 4), in particular on neigh-

 $<sup>^1\</sup>mathrm{In}$  the literature is often supposed  $\mathbb{K}=\mathbb{R}$  but this does not influence the results.

borhoods and on the whole manifold. We prove that such frames or coordinates exist iff the connection is (locally) respectively flat or flat and torsionless. A complete description of the normal frames and coordinates in these cases is presented. We also point to some links between normal frames and parallel transports for flat linear connections.

Section 5 explores the problems of existence, uniqueness, and construction of frames or coordinates normal on arbitrary submanifolds. The classical results of [55] are reproduced in details using modern notation. Meanwhile, the corresponding proofs are improved, some results are generalized for arbitrary connections, with or without torsion, and new ones are presented. Next, we provide a complete constructive description of all frames (resp. coordinates) normal on submanifolds of a manifold with arbitrary (resp. torsionless) linear connection. Amongst a number of general results, we prove that normal on a submanifold frames (resp. coordinates) exist iff the parallel transport is path-independent along paths lying entirely in it (resp. and the connection is torsionless).

Section 6 contains instances and exercises illustrating the general theory of this chapter. Explicit expressions for frames and coordinates normal at a single point in and along a great circle on a two-dimensional sphere are presented in a case of the Riemannian connection induced from the Euclidean space in which the sphere is embedded. Some problems connected with frames/coordinates normal for Weyl connections are investigated. All frames/coordinates normal in the onedimensional case are explicitly described. A similar problem is solved along a geodesic path in a 2-dimensional manifold. All coordinates normal at a point in Einstein-de Sitter spacetime are found.

A brief recapitulation of the above items can be found in Section 7.

# 2. The case at a single point

The coordinates/frames that are normal at a single point are the most simple and widely known ones. The proof of existence of coordinates/frames normal at a single point of Riemannian manifold presented in Section I.6 can be transferred, practically without changes, to manifolds endowed with symmetric linear connections. The main steps of the so-obtained proof are outlined below in Subsection 2.1; for details see [19, p. 155–159], [70, Chapter V, Section 3], or [18, Section 4.3].<sup>1</sup>

# 2.1. Old classical method

Let M be a  $C^{\infty}$  K-manifold,  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ , endowed with  $C^{\infty}$  linear connection  $\nabla$ . Let  $p \in M, V(p)$  be a normal neighborhood of p, (U, x) be a chart with  $U \ni p$ , and  $\{x^i\}$  be the local coordinates on U associated with (U, x). For every

 $<sup>^1\</sup>mathrm{In}$  the last book, the symmetry condition is dropped but this does not effect the general conclusions.

 $q \in U \cap V(p)$  there exists a unique geodesic  $\gamma \colon J \to U \cap V(p)$  such that  $\gamma(s_0) = p$ and  $\gamma(s) = q$  for some (unique)  $s_0, s \in J$ . As the connection is of class  $C^{\infty}$ , such are all geodesics, in particular  $\gamma$ . Therefore in  $\{x^i\}$  the expansion (I.6.3) with  $\gamma^i = x^i \circ \gamma$  is valid. Successively differentiating the equation of geodesics (I.3.23), we can express the derivatives  $\frac{\mathrm{d}^n \gamma^i}{\mathrm{d} s^n} \big|_{s_0}$ ,  $n \in \mathbb{N} \setminus \{1\}$  via  $\dot{\gamma}(s_0)$  and the partial derivatives of the connection's coefficients  $\Gamma^i_{jk}$  at  $p = \gamma(s_0)$ . The result is (cf. (I.6.4))

$$\gamma^{i}(s) = \gamma^{i}(s_{0}) + \dot{\gamma}^{i}(s_{0})(s - s_{0}) - \frac{1}{2}\Gamma^{i}{}_{jk}(\gamma(s_{0}))\dot{\gamma}^{j}(s_{0})\dot{\gamma}^{k}(s_{0})(s - s_{0})^{2} - \sum_{n=3}^{\infty}\frac{1}{n!}\Gamma^{i}{}_{i_{1}\dots i_{n}}(\gamma(s_{0}))\dot{\gamma}^{i_{1}}(s_{0})\dots\dot{\gamma}^{i_{n}}(s_{0})(s - s_{0})^{n} \quad (2.1)$$

where the  $\Gamma$ 's are defined through the recurrent relations (I.6.5) in which  $\Gamma^{i}_{jk}$  must be replaced with  $\Gamma^{i}_{(jk)} = \frac{1}{2} (\Gamma^{i}_{jk} + \Gamma^{i}_{kj})$ , i.e., with the symmetrized connection coefficients.

Define a coordinate system  $\{y^i\}$  on  $U \cap V(p) \ni p = \gamma(s_0), q = \gamma(s)$  by putting

$$y^{i}(q) = \dot{\gamma}(s_0)(s - s_0).$$
 (2.2)

The coordinates  $y^i$  are similar to the Riemannian coordinates of Section I.6 and are called *Riemannian normal coordinates* with origin at the point p; obviously  $y^i(p) = 0$ . The explicit relations between  $\{y^i\}$  and  $\{x^i\}$  is (see (2.1) and cf. (I.6.9))

$$x^{i}(q) = x^{i}(p) + y^{i}(q) - \frac{1}{2}\Gamma^{i}_{\ jk}(p)y^{j}(q)y^{k}(q) - \sum_{n=3}^{\infty}\frac{1}{n!}\Gamma^{i}_{\ i_{1}\dots i_{n}}(p)y^{i_{i}}(q)\dots y^{i_{n}}(q).$$
(2.3)

Since the Jacobian of the change  $\{x^i\} \mapsto \{y^i\}$  at p is

$$\det\left[\frac{\partial y^{i}}{\partial x^{j}}\right]\Big|_{p} = \det\left[\frac{\partial x^{i}}{\partial y^{j}}\right]^{-1}\Big|_{p} = \det\left[\delta_{j}^{i}\right]^{-1} = 1$$

and  $y^i(p) = 0$ , the transition  $\{x^i\} \mapsto \{y^i\}$  is regular in some neighborhood W of p in which the series (2.3) is convergent and invertible with respect to the y's.<sup>2</sup> Consequently the Riemannian normal coordinates are well-defined in the domain  $W \cap U \cap V(p)$  which is a normal neighborhood of p. In it the only common point of every two geodesics though p is the point p itself. Thus we have proved the following result which is an evident generalization of Proposition I.6.2.

**Proposition 2.1.** Every point of a  $C^{\infty}$  manifold with  $C^{\infty}$  linear connection has a normal neighborhood in which Riemannian normal coordinates with origin at that point can be introduced.

<sup>&</sup>lt;sup>2</sup>This is a consequence of the implicit function theorem; e.g., see [77, Chapter III, § 8], [7, Sections 1.37 and 1.38], [78, Chapter 10, Section 2], [79, Theorem 9.18].

#### 2. The case at a single point

In the domain of Riemannian normal coordinates, any geodesic through their origin has an equation like (2.2), i.e., the geodesics in these coordinates are locally described by linear equations like the straight lines in  $\mathbb{R}^n$ .

The adjective 'normal' in the Riemannian normal coordinates is justified by the following results.

**Proposition 2.2.** The symmetric parts of the connection coefficients vanish at the origin of the Riemannian normal coordinates

*Proof.* Using (2.1), we get:

$$\frac{\partial x^{i}}{\partial y^{j}}\Big|_{p} = \frac{\partial y^{i}}{\partial x^{j}}\Big|_{p} = \delta^{i}_{j}, \qquad \frac{\partial^{2} x^{i}}{\partial y^{j} \partial y^{k}}\Big|_{p} = -\Gamma^{i}_{(jk)}(p) = -\frac{1}{2} \left(\Gamma^{i}_{jk}(p) + \Gamma^{i}_{kj}(p)\right). \tag{2.4}$$

Inserting these equations in (I.3.6), we find the coefficients  ${}^{y}\Gamma^{i}_{\ jk}(p)$  of  $\nabla$  in  $\{y^{i}\}$  at p:

$${}^{y}\Gamma^{i}{}_{jk}(p) = \Gamma^{i}{}_{jk}(p) - \Gamma^{i}{}_{(jk)}(p) = \Gamma^{i}{}_{[jk]}(p) = \frac{1}{2} \left( \Gamma^{i}{}_{jk}(p) - \Gamma^{i}{}_{kj}(p) \right).$$

Thereof  ${}^{y}\Gamma^{i}_{(jk)}(p) = \Gamma^{i}_{([jk])}(p) = 0.$ 

**Corollary 2.1.** If a  $C^{\infty}$  manifold is endowed with symmetric  $C^{\infty}$  linear connection, the connection's coefficients vanish at the origin of Riemannian normal coordinates.

Proof. See Proposition 2.2.

Thus the existence of normal coordinates/frames at a single point in the symmetric  $C^{\infty}$  case is proved.

**Example 2.1.** Consider a  $C^{\infty}$  symmetric linear connection on  $C^{\infty}$  manifold M. Let  $p \in M$  and in a chart (U, x) with  $U \ni p$  the only non-vanishing coefficient of the connection at p to be  $\Gamma^{1}_{11}(p) = 1$ . By (I.6.5), the only non-zero value at p of the functions  $\Gamma^{i}_{i_{1}...i_{n}}$ , with  $n \ge 3$ , is  $\Gamma^{1}_{\underbrace{1...1}_{n-\text{times}}}(p) = (-1)^{n}1 \times \cdots \times (n - 1)^{n}$ 

1) =  $(-1)^n(n-1)!$ . Substituting these equations into (2.3), we get the following explicit connection between the coordinates  $\{x^i\}$ , associated with (U, x), and the Riemannian normal coordinates  $\{y^i\}$  with origin at p:

$$\begin{aligned} x^{1}(q) &= x^{1}(p) + \ln(1 + y^{1}(q)) \\ x^{i}(q) &= x^{i}(p) + y^{i}(q) \quad \text{for } i \geq 2, \end{aligned}$$

from which the Riemannian normal coordinates can be expressed as

$$y^{1}(q) = \exp(x^{1}(q) - x^{1}(p)) - 1$$
  

$$y^{i}(q) = x^{i}(q) - x^{i}(p) \quad \text{for } i \ge 2.$$

 $\square$ 

Notice, the Riemannian normal coordinates are 'more than normal' at their origin: at that point, the equations (I.6.11) hold with  $\Gamma^{i}_{jk}$  replaced by  $\Gamma^{i}_{(jk)}$ . The proof of this assertion is identical with the one of (I.6.11) in Section I.6.

Taking the above into account, we see that the equations of geodesics through p in Riemannian normal coordinates is

$$\gamma^{i}(s) = \gamma^{i}(s_{0}) + \dot{\gamma}^{i}(s_{0})(s - s_{0}).$$
(2.5)

**Exercise 2.1.** Prove that a path given via the last equation in Riemannian normal coordinates is a geodesic path through their origin.

A conclusion follows from the above: a path through the origin of the Riemannian normal coordinates is (locally) a geodesic iff in them it is represented with linear equations with respect to its parameter. Said differently, such paths are geodesics iff they are inverse images (with respect to the local coordinates) of the straight lines through a fixed point in  $\mathbb{K}^{\dim M}$ .

According to [19, p. 158] and [53, p. 59], the Riemannian normal coordinates for manifolds with symmetric linear connections, considered on them as coordinates normal at a given point, were first introduced in 1922 by O. Veblen [72].

Let  $\{y^i\}$  be Riemannian normal coordinates on  $C^{\infty}$  manifold endowed with symmetric  $C^{\infty}$  linear connection and p and U be their origin and domain respectively. By the definition of normal coordinates (see Section I.5), the defined on U frame  $\{\frac{\partial}{\partial y^i}\}$  is normal at p. Consequently, according to Proposition I.5.2, the set of all frames on U normal at p is  $\{E_i = A_i^j \frac{\partial}{\partial y^j}\}$  where the non-degenerate matrix-valued function  $A := [A_i^j]$  is such that  $\frac{\partial A}{\partial y^i}|_p = 0$ . (If required, the frames  $\{E_i\}$  can be extended outside U in completely arbitrary ways.) As the point pis arbitrary, in this way we have obtained a *complete* description of the frames normal at a single point of a  $C^{\infty}$  manifold endowed with symmetric  $C^{\infty}$  linear connection.<sup>3</sup>

It is almost self-evident, the Riemannian normal coordinates are not the only local coordinates normal at some point in the symmetric case. For example, we can define in a coordinate neighborhood U of  $p \in M$  the geodesic normal coordinates  $\{z^i\}$  through the series (cf. (I.6.13))

$$x^{i}(q) = x^{i}(p) + z^{i}(q) - \frac{1}{2}\Gamma^{i}_{jk}(p)z^{j}(q)z^{k}(q) - \sum_{n=3}^{\infty}\frac{1}{n!}c^{i}_{i_{1}\dots i_{n}}z^{i_{1}}(q)\cdots z^{i_{n}}(q) \quad (2.6)$$

where  $\{x^i\}$  are some coordinates in  $U, q \in U, z^i(p) := 0$ , and  $c^i_{i_1...i_n} = c^i_{(i_1...i_n)} \in \mathbb{K}$ . As  $\frac{\partial x^i}{\partial y^j}|_p = \delta^i_j$ , there is neighborhood  $V \subseteq U$  of p in which (2.6) is convergent and invertible, i.e.,  $\{z^i\}$  are really coordinates in V. Repeating the proof of Proposition 2.2, we get (cf. (2.4))  ${}^z\Gamma^i_{jk}(p) = \Gamma^i_{[jk]}(p)$ . Therefore  ${}^z\Gamma^i_{jk}(p) = 0$ 

 $<sup>^{3}\</sup>mathrm{This}$  description of the frames normal at a single point is implicit; for the explicit one, see Theorem 2.3 on page 82.

iff  $\Gamma^{i}_{[jk]}(p) = 0$ , etc. Obviously, the choice  $c^{i}_{i_1...i_n} = \Gamma^{i}_{i_1...i_n}(p)$  returns us to the Riemannian normal coordinates.

**Example 2.2.** Let the only non-vanishing values of  $\Gamma^{i}_{\ jk}(p)$  and  $c^{i}_{\ i_{1}...i_{n}}$  be  $\Gamma^{1}_{\ 11} = c^{1}_{\ 1...1} = 1$ . Then (2.6) reduces to the system

$$x^{1}(q) = x^{1}(p) + 1 + 2z^{1}(q) - e^{z^{1}(q)}$$
  
$$x^{i}(q) = x^{i}(p) + z^{i}(q) \quad \text{for } i \ge 2,$$

which is invertible relative to  $\{z^i\}$  everywhere in the domain U of  $\{x^i\}$ .

# 2.2. Complete description

The above-presented way for introduction of normal coordinates is the historically established one. Now we are going to modify it in order to obtain a full description of all coordinates normal at a given point. The first step in this direction is to notice that, due to the definition of normal frames/coordinates, the utilization of the geodesics in the construction of coordinates normal at a single point is not necessary. This is a useful tool but it does not always work! For instance, as we shall see, the geodesics may not exist (e.g., if the connection is not continuous) while coordinates normal at a point exist. Also, we find too strong the requirement for the underlying manifold to be of class  $C^{\infty}$ .

**Proposition 2.3.** Every point p of a  $C^3$  manifold with linear connection has a neighborhood in which exist coordinates such that the symmetric parts of the connection coefficients in them vanish at p. All such coordinates are given via equation (2.11') below in which the coordinates  $\{x^i\}$  are arbitrary,  $[b_j^i]$  is non-degenerate constant matrix, and the  $C^3$  functions  $b_{jkl}^i$  together with their partial derivatives are bounded in the domain of  $\{y^i\}$ . The inverse transformation  $\{y^i\} \to \{x^i\}$  is given by equation (2.11) below with  $[a_j^i] = [b_j^i]^{-1}$ .

*Proof.* Let M be a  $C^3$  manifold endowed with linear connection  $\nabla$  on which we do not impose any differentiability conditions. Choose an arbitrary point  $p \in M$  and charts (U, x) and (V, y) such that  $p \in U \cap V \neq \emptyset$ . As we know from Subsection I.2.1, the local coordinates  $x^i(q)$  and  $y^i(q)$  of every point  $q \in U \cap V$  are connected via  $C^3$ functions  $y^i(q) = f^i(x^1(q), \ldots, x^{\dim M}(q))$ , or  $f^i = r^i \circ y \circ x^{-1}$  with  $r^i$  being the standard coordinate functions on  $\mathbb{K}^{\dim M}$ . Since  $f^i$  and  $(f^i)^{-1}$  are  $C^3$  functions, there exist numbers  $a^i_j, a^i_{jk} \in \mathbb{K}$  and  $C^3$  functions  $a^i_{jkl} : U \cap V \to \mathbb{K}$ , which together with their partial derivatives are bounded on  $U \cap V$ , such that

$$x^{i}(q) = x^{i}(p) + a^{i}_{j}[y^{j}(q) - y^{j}(p)] + a^{i}_{jk}[y^{j}(q) - y^{j}(p)][y^{k}(q) - y^{k}(p)] + a^{i}_{jkl}(q)[y^{j}(q) - y^{j}(p)][y^{k}(q) - y^{k}(p)][y^{l}(q) - y^{l}(p)]. \quad (2.7)$$

By virtue of the above-said, this formula is invertible on  $U \cap V$  with respect to  $y^i(q)$  which can be expressed in a similar way as functions of  $\{x^i(q)\}$ :

$$y^{i}(q) = y^{i}(p) + b^{i}_{j}[x^{j}(q) - x^{j}(p)] + b^{i}_{jk}[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)] + b^{i}_{jkl}(q)[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)][x^{l}(q) - x^{l}(p)], \quad (2.8)$$

where  $b_j^i, b_{jk}^i \in \mathbb{K}$  are constants and  $b_{jkl}^i: U \cap V \to \mathbb{K}$  are of class  $C^3$  and they and their first partial derivatives are bounded on  $U \cap V$ .

Now the problem, which is central for us, is: given a point p and coordinates  $\{x^i\}$ , can a chart (V, y) be chosen so that  $\{y^i\}$  are normal at p? As one can expect, the answer is positive if the connection is symmetric.

From (2.7) and (2.8), we derive:

$$\frac{\partial x^{i}}{\partial y^{j}}\Big|_{p} = a^{i}_{j}, \quad \frac{\partial y^{i}}{\partial x^{j}}\Big|_{p} = b^{i}_{j}, \qquad \frac{\partial^{2} x^{i}}{\partial y^{j} y^{k}}\Big|_{p} = 2a^{i}_{(jk)}, \quad \frac{\partial^{2} y^{i}}{\partial x^{j} x^{k}}\Big|_{p} = 2b^{i}_{(jk)}$$
(2.9)

where the matrices  $[a_j^i]$  and  $[b_j^i]$  are non-degenerate as a consequence of the invertability of (2.7) and (2.8) at p. Using the equality  $\frac{\partial x^i}{\partial y^k} \frac{\partial y^k}{\partial x^j} = \delta_j^i$  and the one obtained from it by differentiation with respect to  $x^l$ , we find from (2.9)

$$\begin{aligned}
a_{k}^{i}b_{j}^{k} &= \delta_{i}^{j} \iff [b_{k}^{i}] = [a_{k}^{i}]^{-1}, \\
a_{(mn)}^{i}b_{l}^{m}b_{j}^{m} + a_{k}^{i}b_{(jl)}^{k} &= 0 \iff b_{(jk)}^{i} = -b_{l}^{i}a_{(mn)}^{l}b_{j}^{m}b_{k}^{n}.
\end{aligned}$$
(2.10)

Let  ${}^{x}\Gamma_{jk}^{i}$  and  ${}^{y}\Gamma_{jk}^{i}$  be the coefficients of  $\nabla$  in  $\{x^{i}\}$  and  $\{y^{i}\}$  respectively. Applying the transformation laws (I.3.6) and (2.9), we get at p:

$${}^{y}\Gamma^{i}_{jk}(p) = \left( [a^{i}_{j}]^{-1} \right)^{i}_{l} \left( a^{m}_{j} a^{n}_{k} {}^{x}\Gamma^{l}_{mn}(p) + 2a^{l}_{(jk)} \right).$$

So, at p we can obtain  ${}^{y}\Gamma^{i}{}_{(jk)}(p) = 0$  if and only if  $2a^{l}_{(jk)} = -a^{m}_{j}a^{n}_{k}{}^{x}\Gamma^{l}{}_{mn}(p)$  which, due to (2.10), is equivalent to  $2b^{i}_{(jk)} = b^{i}_{l}{}^{x}\Gamma^{l}{}_{jk}(p)$ .<sup>4</sup> Substituting the last equation into (2.7), we see that all coordinates  $\{y^{i}\}$  for which  ${}^{y}\Gamma^{i}{}_{(jk)}(p) = 0$ , if such exist, are obtainable from fixed coordinates  $\{x^{i}\}$  by inverting the equation

$$\begin{aligned} x^{i}(q) &= x^{i}(p) + a^{i}_{j}[y^{j}(q) - y^{j}(p)] \\ &- {}^{x}\Gamma^{i}_{(mn)}(p)a^{m}_{j}a^{n}_{k}[y^{j}(q) - y^{j}(p)][y^{k}(q) - y^{k}(p)] \\ &+ a^{i}_{jkl}(q)[y^{j}(q) - y^{j}(p)][y^{k}(q) - y^{k}(p)][y^{l}(q) - y^{l}(p)] \end{aligned}$$
(2.11)

on some subneighborhood  $W \ni p$  of U. By virtue of (2.8), this inversion results in

$$y^{i}(q) = y^{i}(p) + b^{i}_{j}[x^{j}(q) - x^{j}(p)] + b^{i}_{l}{}^{x}\Gamma^{l}_{(jk)}(p)[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)] + b^{i}_{jkl}(q)[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)][x^{l}(q) - x^{l}(p)]. \quad (2.11')$$

<sup>4</sup>We cannot set  ${}^{y}\Gamma^{i}_{jk}(p) = 0$  or  ${}^{y}\Gamma^{i}_{[jk]}(p) = 0$  unless  $\nabla$  is torsionless at p.

Here  $a_j^i, b_j^i \in \mathbb{K}$ ,  $\det[a_j^i] \neq 0, \infty, [b_j^i] = [a_j^i]^{-1}$ , and the  $C^3$  functions  $a_{jkl}^i, b_{jkl}^i \colon W \to \mathbb{K}$  together with their partial derivatives are bounded on W. Since the matrix (the Jacobian at p)  $\left[\frac{\partial x^i}{\partial y^j}|_p\right] = \left[a_j^i\right]$  is non-degenerate, by the implicit function theorem (see [77, Chapter III, § 8], [7, Sections 1.37 and 1.38], or [78, Chapter 10, Section 2]), there exists a neighborhood W of p in which (2.11) defines  $y^i$  as unique  $C^3$  functions of  $x^i$ . Hence, in the neighborhood  $V = U \cap W \ni p$ , the mappings  $\{x^i\} \mapsto \{y^i\}$  and  $\{y^i\} \mapsto \{x^i\}$ , defined via (2.11) and (2.11') in which  $x^i(p)$  and  $y^i(p)$  are fixed numbers, are  $C^3$  diffeomorphisms. This ends to proof of Proposition 2.3.

If we specify the connection to be symmetric, from Proposition 2.3, we obtain a theorem describing all local coordinates normal at a given point.

**Theorem 2.1.** Every point p of a  $C^3$  manifold endowed with symmetric linear connection has a neighborhood on which coordinates normal at p exist. All normal coordinates  $\{y^i\}$  in the mentioned neighborhood are given via equation (2.11') in which the coordinates  $\{x^i\}$  are arbitrary,  $[b_j^i]$  is non-degenerate constant matrix, and the  $C^3$  functions  $b_{jkl}^i$  together with their partial derivatives are bounded in the domain of  $\{y^i\}$ . The inverse transformation  $\{y^i\} \to \{x^i\}$  is given by (2.11) with  $[a_j^i] = [b_j^i]^{-1}$ .

This theorem gives a *complete* description of all normal coordinates at a single point of a  $C^3$  manifold with symmetric linear connection. Analogous result concerning the normal frames is provided by the following theorem.

**Theorem 2.2.** Let  $\{y^i\}$  be coordinates normal at a point p in a  $C^3$  manifold with symmetric linear connection. In the domain U of  $\{y^i\}$  all frames normal at p have the form  $\{E_i = A_i^j \frac{\partial}{\partial y^j}\}$ , where the non-degenerate matrix-valued function  $A = [A_j^i]$  is such that  $\frac{\partial A}{\partial y^j}\Big|_p = 0.5$  Outside U the frames can be extended arbitrarily. All of these frames normal at p are holonomic at p but in  $U \setminus \{p\}$  they need not to be such.

*Proof.* See Propositions I.5.2 and I.5.3.

Remark 2.1. In the notation of Theorem 2.3 on the next page below, the explicit form of A is given via (2.14) with  $y^i$  for  $x^i$  and  $\Gamma^i_{jk}(p) = 0$  (as  $\{y^i\}$  is normal at p), i.e.,

$$A(q) = A_0 + A_{jk}(q)[y^j(q) - y^j(p)][y^k(q) - y^k(p)].$$
(2.12)

For the proof, see the proof of Theorem 2.3 below. The frames  $\{E_i\}$  holonomic on U and normal at p are such that  $E_i = \frac{\partial}{\partial z^i}$  for some coordinates  $z^i$  on U which are

<sup>&</sup>lt;sup>5</sup>In the notation of Theorem 2.3 on the following page, the explicit form of A is given via (2.14) with  $y^i$  for  $x^i$ ; for the proof, see the proof of Theorem 2.3 below.

normal at p. In this case  $A_j^i = \frac{\partial y^i}{\partial z^j}$  which, due to (2.12), implies

$$z^{i}(q) = y^{i}(p) + (A_{0}^{-1})^{i}_{j}(q)[y^{j}(q) - y^{j}(p)] + b^{i}_{jkl}(q)[y^{j}(q) - y^{j}(p)][y^{k}(q) - y^{k}(p)][y^{l}(q) - y^{l}(p)], \quad (2.13)$$

where  $b_{jkl}^i: U \to \mathbb{K}$  are of class  $C^1$  and they and their partial derivatives are bounded on U. Of course, this result is a special case of (2.11') as  $\{y^i\}$  are normal at p and the connection considered is symmetric.

If the manifold and its connection are of class  $C^{\infty}$ , the choice  $a_j^i = \delta_j^i$  and the expansion of  $a_{jkl}^i$  into a power series, with suitable coefficients, brings us back to the results of Subsection 2.1.

Theorem 2.2 can easily be generalized to arbitrary linear connections, with or without torsion:

**Theorem 2.3.** Let M be a  $C^2$  manifold endowed with linear connection. For every point  $p \in M$  there exist frames normal at p. Moreover, if (U, x) is a chart with  $U \ni p$ , then in U all frames normal at p are  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  where  $A := [A_j^i]: U \rightarrow$  $\operatorname{GL}(\dim M, \mathbb{K})$ , with  $\operatorname{GL}(n, \mathbb{K})$  being the group of all  $n \times n$  invertible matrices with entries in  $\mathbb{K}$ , is non-degenerate, of class  $C^1$ , and its general form is

$$A(q) = \left\{ \mathbb{1} - \Gamma_j(p)[x^j(q) - x^j(p)] \right\} A_0 + A_{jk}(q)[x^j(q) - x^j(p)][x^k(q) - x^k(p)] \qquad q \in U.$$
(2.14)

Here  $\Gamma_i$  are the matrices of the connection coefficients,  $A_0$  is constant and nondegenerate matrix, and  $A_{jk}$  are  $C^1$  matrix-valued functions on U such that they and their partial derivatives are bounded at p.

*Proof.* A frame  $\{E'_i\}$  is normal at p iff A satisfies (I.5.4) with  $\{p\}$  for U and  $\frac{\partial}{\partial x^i}$  for  $E_i$ , i.e.,  $\frac{\partial A}{\partial x^i}\Big|_p + \Gamma_i(p)A(p) = 0$ . Supposing A to be of class  $C^2$ , we can write the expansion

$$A(q) = A_0 + A_k[x^k(q) - x^k(p)] + A_{jk}(q)[x^j(q) - x^j(p)][x^k(q) - x^k(p)],$$

where  $A_0, A_i \in \operatorname{GL}(\dim M, \mathbb{K})$  are constant,  $A_0$  is non-degenerate, and the  $C^1$  matrix-valued functions  $A_{jk}^i$  on U together with their first partial derivatives are bounded at p. From the last equation, we get  $A(p) = A_0$  and  $\frac{\partial A}{\partial x^i}\Big|_p = A_i$  and hence (see above)  $A_i = -\Gamma_i(p)A_0$ , the substitution of which into the last displayed equation gives (2.14).

Theorem 2.3 gives a *complete* constructive description of all frames normal at a single point.

The above description of the normal coordinates/frames is an adaptation of the developed in [80] methods (see also Chapter III) for the case of manifolds with linear connections.

### 2.3. Modern classical method

In the modern books on differential geometry, the normal coordinates as described in Subsection 2.1 are most often not mentioned at all; see, e.g.: [11, Chapter III, § 8], [12, pp. 313–314], [10, p. 110], [28, p. 133], [1, § 6.7], and [8, § 3.8, (ii)]. Nowadays the coordinates normal at a point are introduced by means of the exponential mapping (see Subsection I.3.4), like in the references just cited. This method is practically identical with the introduction of Riemannian normal coordinates, presented in Subsection 2.1, only other terminology and concepts are involved. It is briefly outlined below.

Let M be a  $C^2$  manifold,  $p \in M$ , and  $(U, \varphi)$  be a chart with  $U \ni p$ . The chart  $(U, \varphi)$  or the associated with it local coordinates  $\{\varphi^i\}$  are called *normal at* p (with respect to p) if the inverse images of the straight lines through  $\varphi(p) \in \mathbb{K}^{\dim M}$  are geodesics in M through p. This definition, as we shall see further, agrees with our previous terminology (see Definition I.5.2).

Let a basis  $\{e_i\}$  in  $T_p(M)$  be given. Since the mapping  $\nu: T_p(M) \to \mathbb{K}^{\dim M}$ defined by  $X \mapsto (X^1, \ldots, X^{\dim M})$  for  $X = X^i e_i \in T_p(M)$  is a linear isomorphism, the exponential mapping can be used to define normal coordinates in a neighborhood of p. The chart  $(T_p(M), \nu)$  of  $T_p(M)$  provides a natural coordinate system on  $T_p(M)$ . Let  $V_p^0 \subseteq T_p(M)$  and  $V(p) \subseteq M$  be normal neighborhoods such that the exponential mapping  $\exp_p: V_p^0 \to V(p)$  is a diffeomorphism (see Definition I.3.6). Consider a local chart (V(p), x) with

$$r := \nu \circ \exp_p^{-1} \colon V(p) \to \mathbb{K}^{\dim M}.$$
(2.15)

If  $\gamma_{p,X}: J \to V(p), J \ni 0 \in \mathbb{R}$  is the unique geodesic with  $\gamma_{p,X}(0) = p$  and  $\dot{\gamma}_{p,X}(0) = X$ , its current coordinates in the coordinate system  $\{x^i\}$  are

$$\begin{aligned} \gamma_{p,X}^{i}(s) &:= x^{i}(\gamma_{p,X}(s)) = (r^{i} \circ x)(\gamma_{p,X}(s)) = (r^{i} \circ \nu \circ \exp_{p}^{-1}) \circ \exp_{p}(sX) \\ &= r^{i}(\nu(sX)) = r^{i}(sX^{1}, \dots, sX^{\dim M}) = sX^{i}, \qquad s \in J \end{aligned}$$

where (I.3.29) is used and  $\{r^i\}$  are the standard coordinate functions on  $\mathbb{K}^{\dim M}$ . Reversing the last equalities, we see that any path given by  $\gamma^i(s) = sX^i$  in  $\{x^i\}$  is a geodesic through p. Consequently  $\{x^i\}$  are normal coordinates at p according to the last definition. In case of symmetric connection, the coordinates  $\{x^i\}$  are also normal according to Definition I.5.2. Indeed, if  $\gamma: J \to M$  is a geodesic through p, in  $\{x^i\}$  we have  $\gamma^i(s) = sX^i$  for some  $X^i \in \mathbb{K}$ , the substitution of which in the geodesic equation (I.3.23') on page 32 results in  $\Gamma^i_{(jk)}(x^{-1}(sX^1,\ldots,sX^{\dim M}))X^jX^k = \Gamma^i_{(jk)}(\gamma(s))X^jX^k \equiv 0$  with  $\Gamma^i_{jk}$  being the connection coefficients in  $\{x^i\}$ . In particular we have

$$0 = \Gamma^{i}_{(jk)}(p)X^{j}X^{k} = \Gamma^{i}_{(jk)}(x^{-1}(\mathbf{0}))X^{j}X^{k}$$

as  $p = x^{-1}(\mathbf{0}), \mathbf{0} \in \mathbb{K}^{\dim M}$ . Since X is completely arbitrary, such are  $X^i$ , and consequently

$$\Gamma^{i}_{(jk)}(p) := \frac{1}{2} \left( \Gamma^{i}_{jk}(p) + \Gamma^{i}_{kj}(p) \right) = 0.$$

So, if the connection is symmetric, then  $\Gamma^i{}_{jk}(p) = 0$ , i.e.,  $\{x^i\}$  are normal coordinates at p in the sense of Definition I.5.2.

Comparing the above results with the ones in Subsection 2.1, we see that in the case of a  $C^{\infty}$  manifold with  $C^{\infty}$  linear connection the coordinates  $\{x^i\}$ are just the Riemannian normal coordinates introduced in Subsection 2.1. For this reason, we find it appropriate and convenient to call the coordinates defined by (2.15) *Riemannian normal coordinates*, not simply normal as in the literature cited, with an exception of [8, § 3.8, (ii)], and at the beginning of this subsection; thus we reserve the adjective 'normal' for frames and coordinates described via Definitions I.5.1 and I.5.2.

At the end, let us note that in [8, § 3.8, (ii)] the local coordinates, described by (2.15) (with  $i := \nu^{-1}$  being a linear isometry), are applied to a proper Riemannian manifold (the metric being positively definite) which results to a new construction of the particular Riemannian coordinates obtained by G. D. Birkhoff [50] (see p. 48). These coordinates appear in [8, § 3.8, (ii)] under the name Riemannian normal coordinates.

# 3. The case along paths without self-intersections

The first proof of existence of coordinates normal along curve without self-intersections on  $(C^{\infty})$  Riemannian manifold was given by E. Fermi in 1922 [52]. The explicit formula for transition to the coordinates discovered by E. Fermi was given by Levi-Civita in 1926 [81]. Analogous result on  $(C^{\infty})$  manifold with arbitrary symmetric  $(C^{\infty})$  connection was originally published by L. P. Eisenhart in 1927 [53, p. 64].<sup>1</sup> These special types of coordinates are widely known as *Fermi coordinates* [19, Chapter III, § 8], [8, § 3.8, (ii)]. Sometimes they are referred as Fermi geodesic coordinates [12, p. 327]<sup>2</sup> and very rarely as geodesic (along a curve) coordinates [25, § 91], [44].

Since all of the proofs, known to the author, of the existence of such coordinates are more or less identical at a level of ideas, i.e., a construction of particular class of coordinates with the property required, we suggest to call *Fermi coordinates* the special kind of coordinates normal along a path described below in Subsection 3.1, thus reserving the term 'coordinates normal along a path' for a particular realization of Definition I.5.2.

<sup>&</sup>lt;sup>1</sup>For other original papers on this topic, see [19, p. 166].

<sup>&</sup>lt;sup>2</sup>More precisely, in [12, p. 327] is done the following. Let  $\gamma: J \to M$  be without self-intersections and for some  $s_0 \in J$  a basis  $\{E_i^0\}$  in  $T_{\gamma(s_0)}(M)$  be fixed. Define along  $\gamma$  a frame  $\{E_i\}$  such that  $E_i|_{\gamma(s_0)} = E_i^0$  and  $(\nabla_{\dot{\gamma}} E_i)|_{\gamma(J)} = 0$ , i.e.,  $\{E_i\}$  is obtained from  $\{E_i^0\}$  by means of parallel propagation along  $\gamma$ . Notice,  $\{E_i\}$  is defined only on  $\gamma(J)$ ; outside  $\gamma(J)$  it can be extended arbitrarily. The Fermi geodesic coordinates with domain  $U, U \cap \gamma(J) \neq \emptyset$ , are local coordinates  $\{x^i\}$  such that  $\frac{\partial}{\partial x^i}|_p = E_i|_p$  for  $p \in U \cap \gamma(J)$ . Generally these coordinates are not normal along  $\gamma$  in U.

#### 3. The case along paths

One must be aware of the fact that as 'Fermi coordinates' can be found coordinate systems entirely different from the ones we are dealing with. For instance, in the literature on general relativity under the name 'Fermi coordinates' are known completely different coordinates which generally are not normal at all but along a geodesic in a Riemannian manifold they are normal at their origin [62, Chapter II, § 10]. More precisely, in the last case the coordinates mentioned, as defined in [62, Chapter II, § 10], coincide with the normal coordinates introduced by Birkhoff (see p. 48). The cause for this is that these coordinates are defined by means of the Fermi-Walker transport [62, Chapter I, § 4] which along a geodesic coincides with the parallel transport along it. Another example of 'Fermi coordinates', which generally are not normal, may be found in [28, pp. 133–134]. In [55] the term 'Fermi coordinates' is a synonym of our notion of 'coordinates normal on a submanifold'.

There are two basic methods for proving the existence of coordinates normal along paths without self-intersections. The first one is to construct a specific coordinate system in a neighborhood of the path and then, by explicit calculation, to show that along the path given the coordinates constructed are normal. The second one consist in finding a particular (class of) solution of equation (I.5.4') on page 41 along a given path. A typical example of the former method is given in Subsection 3.1, while a modification of the latter method is presented in Subsection 3.2

Below in this section, the manifold M will be considered as real one, i.e., if it is complex, it will be regarded as real one of dimension  $2 \dim M = 2 \dim_{\mathbb{C}} M = \dim_{\mathbb{R}} M$  (see page 7).<sup>3</sup> Formally we shall reflect this by writing  $\dim_{\mathbb{R}} M (= \dim M$ if M is real and  $= 2 \dim_{\mathbb{C}} M$  if M is complex) instead of  $\dim M$  for the dimension of M; respectively, all Latin indices, whose range is not specified, run from 1 to  $\dim_{\mathbb{R}} M$  and the values of the coordinate homeomorphisms will be in  $\mathbb{R}^{\dim_{\mathbb{R}} M}$ .

# 3.1. Fermi coordinates

Below, following [19, pp. 166–169], we shall prove the existence of coordinates normal along a (part of a given) path without self-intersections by explicit construction of concrete such coordinates. Regardless of the technical difficulties, the idea of the proof is quite simple and consists in the following. In a neighborhood of the path  $\gamma: J \to M$ , the manifold M is (locally) represented as a direct sum of suitable (dim<sub>R</sub> M - 1)-dimensional submanifolds  $V_s$ , one for each point  $\gamma(s), s \in J$ . So, every point p of this neighborhood belongs to a single  $V_{s(p)}$  for some unique  $s(p) \in J$ . Then the Fermi coordinates of p are  $(s(p), \xi^2, \ldots, \xi^{\dim_R} M)$ where  $(\xi^2, \ldots, \xi^{\dim_R} M)$  are the Riemannian normal coordinates of p with respect

<sup>&</sup>lt;sup>3</sup>In this way we avoid some completely technical problems connected with the fact that the domain of every path is a *real* interval by definition. Roughly speaking, if  $\gamma: J \to M$  is  $C^1$  regular injective path in a complex manifold M, the set  $\gamma(J)$  is a submanifold of M with real dimension  $\dim_{\mathbb{R}} \gamma(J) = 1$  and complex one  $\dim_{\mathbb{C}} \gamma(J) = 1/2$ , the last case, of non-integer dimension, being out of the range of our work.

to  $\gamma(s(p))$  in  $V_{s(p)}$  considered as a  $(\dim_{\mathbb{R}} M - 1)$ -dimensional submanifold of M. The details of this construction are presented below and their aim is the formulation and proof of Proposition 3.1 below and of other results presented after it. If the reader is not interested in the technical details that follow below, he/she can jump to the definition of Fermi coordinates, given just after equation (3.9') below, and next to proceed with Proposition 3.1 and the text after it.

Let  $\gamma: J \to M$  be a  $C^1$  regular<sup>4</sup> path without self-intersections<sup>5</sup> in a  $C^{\infty}$ manifold M endowed with  $C^{\infty}$  connection  $\nabla$ . Let  $s_0 \in J$  be arbitrarily fixed inner, i.e., not boundary, if any,<sup>6</sup> point and (U, x) be a chart in whose domain is  $\gamma(s_0), U \ni \gamma(s_0)$ . Let  $X \in \mathfrak{X}(\gamma(J))$  be a vector field parallel along  $\gamma$  with zero component along  $\dot{\gamma}(s_0)$  at  $\gamma(s_0)$ , i.e.,

$$\nabla_{\dot{\gamma}} X = 0, \qquad X_{\gamma(s_0)} \in T_{\gamma(s_0)}(M) \setminus \left\{ a \dot{\gamma}(s_0) | a \in \mathbb{R} \setminus \{0\} \right\}$$

or, equivalently, we can write  $X_{\gamma(s_0)} = \sum_{i=2}^{\dim_{\mathbb{R}} M} X^i_{\gamma(s_0)} E^0_i$  for every basis  $\{E^0_i\}$  in  $T_{\gamma(s_0)}(M)$  with  $E^0_1 = \dot{\gamma}(s_0)$ .

At every point  $\gamma(s) \in \gamma(J) \cap U$ ,  $s \in J_U := \{u | u \in J, \gamma(u) \in U\}$ , we consider the geodesics  $\beta_s : J_X \to U$  with initial conditions  $\beta_s(t_0) = \gamma(s)$  and  $\dot{\beta}_s(t_0) = X_{\gamma(s)}$  for  $s \in J_U$ , some fixed  $t_0 \in J_X$ , and arbitrary X as defined above (i.e., for arbitrary its value  $X_{\gamma(s_0)} \in T_{\gamma(s_0)}(M) \setminus \{a\dot{\gamma}(s_0) | a \in \mathbb{R} \setminus \{0\}\}$ ). According to (2.1), the expansion

$$\beta_s^i(t) = \gamma^i(s) + X^i_{\gamma(s)}(t - t_0) - \sum_{n=2}^{\infty} \frac{1}{n!} \Gamma^i_{i_1 \dots i_n}(\gamma(s)) X^{i_1}_{\gamma(s)} \dots X^{i_n}_{\gamma(s)}(t - t_0)^n \quad (3.1)$$

is valid in the associated to (U, x) local coordinates  $\{x^i\}$ . Here  $\beta_s^i := x^i \circ \beta_s$ ,  $\gamma^i := x^i \circ \gamma$ ,  $X = X^i \frac{\partial}{\partial x^i} |_{\gamma(J)}$ ,  $\Gamma^i_{\ jk}$  are the coefficients of  $\nabla$  in  $\{x^i\}$ , and  $\Gamma^i_{i_1...i_n}$  for  $n \geq 3$  are defined via (I.6.5).

Let  $\{E_i^0\}$  be a basis in  $T_{\gamma(s_0)}(M)$  with  $E_1^0 = \dot{\gamma}(s_0)$  and  $\{E_i'\}$  be a frame along  $\gamma$  obtained from  $\{E_i^0\}$  by parallel transportation along  $\gamma$ :

$$\nabla_{\dot{\gamma}} E'_i = 0 \quad E'_i |_{\gamma(s_0)} = E^0_i, \quad E^0_1 = \dot{\gamma}(s_0).$$

<sup>&</sup>lt;sup>4</sup>A  $C^1$  path  $\gamma \colon J \to M$  is regular if  $\dot{\gamma}(s) \neq 0, \infty$  for all  $s \in J$  (see p. 14).

<sup>&</sup>lt;sup>5</sup>A point  $p \in \gamma(J)$  is called self-intersection point of  $\gamma: J \to M$  if there exist  $s_1, s_2 \in J$  such that  $s_1 \neq s_2$  and  $\gamma(s_1) = \gamma(s_2) = p$ . Given  $s_0 \in J$ , the number k(p) of the different values  $s \in J$  with  $s \neq s_0$  and  $\gamma(s) = \gamma(s_0) = p$  is called self-intersection number of  $\gamma$  at  $p = \gamma(s_0)$ ; we also say that  $\gamma$  self-intersects (itself) k(p) times at p. If k(p) = 0 for all  $p \in \gamma(J)$ , the path  $\gamma$  is said to be without self-intersections, i.e.,  $\gamma$  is without self-intersections if for every  $s_1 \in J$  there does not exist  $s_2 \in J$ ,  $s_1 \neq s_2$  with  $\gamma(s_1) = \gamma(s_2)$ . In other words,  $\gamma$  is without self-intersections if  $\gamma$  is an injective mapping.

<sup>&</sup>lt;sup>6</sup>If J has left or/and right end (boundary) point(s) and  $s_0$  happens to be its left or right end, some complications arise; e.g., with the definition of  $\dot{\gamma}(s_0)$  as a corresponding one-sided (left or right) limit, the implicit function theorem [77, Chapter III, § 8] can not be applied in a neighborhood of  $(s_0, \gamma(s_0))$ , etc. Generally this is not a typical situation and we exclude it for a moment from our investigation. Later we will drop this assumption. These complications are due to the fact that if J is closed from one or both ends, the set  $\gamma(J)$  is a manifold with boundary (see Remark I.2.1 on page 6) which we treat with methods for manifolds without boundary. They will not appear if from the beginning one starts with manifolds with boundary.

#### 3. The case along paths

By construction, we have (see Subsection I.3.3)  $X = X'^{i}E'_{i} = \sum_{i=2}^{\dim_{\mathbb{R}} M} X'^{i}E'_{i}$ where  $X'^{i}$  are constants,  $X'^{i} \in \mathbb{R}$ , and  $X'^{1} = 0$ .

In  $U \cap \gamma(J)$  the expansion  $E'_i = A^j_i \frac{\partial}{\partial x^i}$  is valid for some non-degenerate matrix-valued function  $A = [A_i^j], A_i^j: U \cap \gamma(J) \to \mathbb{R}$ . Applying (I.2.12), we get  $X^i = A_i^i X'^j$  with  $X'^1 = 0$ , so (3.1) can be rewritten as

$$\beta_{s}^{i}(t) = \gamma^{i}(s) + \sum_{j=2}^{\dim_{\mathbb{R}} M} A_{j}^{i}(\gamma(s)) X'^{j}(t-t_{0}) - \sum_{n=2}^{\infty} \frac{1}{n!} \sum_{i_{1},\dots,i_{n}=2}^{\dim_{\mathbb{R}} M} A_{i_{1}}^{j_{1}}(\gamma(s)) \cdots A_{i_{n}}^{j_{n}}(\gamma(s)) \Gamma^{i}_{j_{1}\dots j_{n}}(\gamma(s)) \times X'^{i_{1}}_{\gamma(s)} \dots X'^{i_{n}}_{\gamma(s)}(t-t_{0})^{n}.$$
(3.2)

Now the idea is in (a subneighborhood of) U to be constructed local coordinates  $\{x'^i\}$  such that  $\frac{\partial}{\partial x'^i}\Big|_p = E'_i\Big|_p$  and  $\Gamma'^i_{jk}(p) = 0$  for p in (a subset of)  $U \cap \gamma(J)$ . As  $E'_i|_p = A^j_i(p) \frac{\partial}{\partial x^j}|_p$ ,  $p \in U \cap \gamma(J)$ , the only restriction on the connection  $x'^i = x'^i(x^1, \ldots, x^{\dim_{\mathbb{R}} M})$  is  $\frac{\partial x^j}{\partial x'^i}\Big|_p = A^j_i(p), p \in U \cap \gamma(J)$ . Below we will construct a particular realization of this link.

Let  $f^i: J_U \to \mathbb{R}$ , with  $s \in J_U := \{u | u \in J, \gamma(u) \in U\}$ , be  $C^1$  functions and  $f^1$  be  $C^1$  diffeomorphism from  $J_U$  on the image  $f^1(J_U)$ . We shall find these functions from the requirement that<sup>7</sup>

$$x'^{1}(\beta_{s}(t)) := f^{1}(s), \qquad x'^{i}(\beta_{s}(t)) := f^{i}(s) + (t - t_{0})X'^{i}, \quad i \ge 2$$
(3.3)

which is a special realization of the above general connection between  $\{x^i\}$  and  $\{x'^{i}\}.$ 

*Remark* 3.1. Here and below we implicitly suppose that the family of geodesic paths  $\{\beta_s\}$  forms (for different X) an  $(\dim_{\mathbb{R}} M - 1)$ -dimensional foliation along the path  $\gamma$ , i.e., that the sets  $V_s := \{\beta_s(J_X) \text{ where } X, \text{ as defined above, is arbitrary} \}$  are such that:  $\bigcup_{\gamma(s)\in U} V_s = U, V_s \cap V_{s'} = \emptyset$  for  $s \neq s', s, s' \in J$  and  $V_s \cap \gamma(J) = \gamma(s)$ . Then for every  $p \in U$  there exists only one  $s \in J$  such that  $p \in V_s$ . Equations (3.3) mean that at first we define the coordinates of p with respect to  $\gamma(s)$  in  $V_s$  and then of  $\gamma(s)$  along  $\gamma$ . The existence of such a foliation  $\{V_s\}$  along  $\gamma$ , which is overlooked in [19], is natural but not trivial and requires corresponding proof. Rigorously we have to do the following. Let  $\{y^i\}$ be any coordinate system in U. In Subsection 3.2 (see the text preceding equation (3.12)) will be proved that there exist an interval  $J_1 \subseteq J$  and neighborhood  $U_1 \subseteq U$  such that  $\dot{\gamma}_y^1|_{J_1} \neq 0$  and  $\gamma_y^1|_{J_1}: J_1 \to \gamma_y^1(J_1)$  is diffeomorphism  $(\gamma_y^k:=y^k \circ \gamma)$ . Now, in a neighborhood  $U_1'$  of  $\gamma(J_1)$  in  $U_1$ , choose  $\{y^i\}$  such that  $\frac{\partial}{\partial y^i}|_{\gamma(s)} = E_i'|_{\gamma(s)}$  for  $s \in J_1$  and define  $V_s := \{p \in U_1'|y^1(p) = \gamma^1(s)\}$ .<sup>8</sup> Then we have  $U_1' = \bigcup_{s \in J_1} V_s, V_s \cap V_{s'} = \emptyset$ 

<sup>&</sup>lt;sup>7</sup>Generally the last condition on  $f^1$  can not be satisfied on the whole interval  $J_U$  if equations (3.3) hold (see below the comment after (3.5)). If this happens to be the case, the interval  $J_U$  must be replaced by some open subinterval  $J'_U \subset J_U$  containing the initial point  $s_0, J'_U \ni s_0$ . <sup>8</sup>Such coordinates  $\{y^i\}$  always exist – see [76, Lemma 4.1] or Lemma 5.2 on page 116 for

 $<sup>\</sup>dim_{\mathbb{R}} N = 1.$ 

for  $s \neq s'$ ,  $s, s' \in J_1$  and  $V_s \cap \gamma(J_1) = \gamma(s)$  for  $s \in J_1$ . The sets  $V_s$  are  $(\dim_{\mathbb{R}} M - 1)$ dimensional submanifolds of M. At last, in each  $V_s$  we take a normal neighborhood  $V_s^0$ of  $\gamma(s) \in V_s$  and put  $U_1^0 := \bigcup_{s \in j_1} V_s^0$ . It is evident that  $U_1^0$  is a neighborhood of  $\gamma(J_1)$ and for every  $p \in U_1^0$  there exists a unique  $s(p) \in J_1$  such that there is unique geodesic  $\beta_{s(p)}$  joining p and  $\gamma(s(p))$  in  $V_{s(p)}^0$ . Thus, generally, we are forced to consider instead of U and J smaller sets  $U_1^0 \subseteq U$  and  $J_1 \subseteq J$ , respectively, on which the foliation mentioned exists. Hence, to be quite precise, below we have to replace U with  $U_1^0$  and J with  $J_1$ where  $U_1^0$  and  $J_1$  are defined above. Since these 'details' do not change the main ideas and formulae, we leave them and follow [19] directly.

As  $\beta_s(t_0) = \gamma(s)$ , now the equation of  $\gamma$  is  $x'^i(\gamma(s)) = f^i(g(\xi)), i \ge 2$ , where  $g := (f^1)^{-1}$  and  $\xi := x'^1(\beta_s(t)) = x'^1(\gamma(s))$ . Hence along  $\gamma$  the *A*'s and  $\Gamma$ 's may be considered as functions of  $\xi = f^1(s)$  as  $s = (f^1)^{-1}(\xi) =: g(\xi)$ . Consequently, using (3.3), we transform (3.2) into

$$\beta_{s}^{i}(t) = \gamma^{i}(g(\xi)) + \sum_{j=2}^{\dim_{\mathbb{R}} M} A_{j}^{i}(\gamma(g(\xi))) \left[ x'^{j}(\beta_{s}(t)) - f^{j}(g(\xi)) \right] - \sum_{n=2}^{\infty} \frac{1}{n!} \sum_{i_{1},\dots,i_{n}=2}^{\dim_{\mathbb{R}} M} A_{i_{1}}^{j_{1}}(\gamma(g(\xi))) \cdots A_{i_{n}}^{j_{n}}(\gamma(g(\xi))) \times \Gamma^{i}_{j_{1}\dots j_{n}}(\gamma(g(\xi))) \left[ x'^{i_{1}}(\beta_{s}(t)) - f^{i_{1}}(g(\xi)) \right] \cdots \left[ x'^{i_{n}}(\beta_{s}(t)) - f^{i_{n}}(g(\xi)) \right].$$
(3.4)

We will find the functions  $f^i$  from the condition  $E'_i|_p = \frac{\partial}{\partial x'^i}|_p$  for  $p \in U \cap \gamma(J)$ . In this case  $\frac{\partial}{\partial x'^i}|_p = A^j_i(p)\frac{\partial}{\partial x^j}|_p$ , so that  $A^j_i(p) = \frac{\partial x^j}{\partial x'^i}|_p$ . Taking this into account, by differentiating (3.4) with respect to  $\xi = x'^1$  and  $x'^i$ ,  $i \geq 2$ , and restricting the results on  $\gamma(J_U) = U \cap \gamma(J)$ , we obtain:

$$A_1^i = \frac{\mathrm{d}f^i}{\mathrm{d}s} \left(\frac{\mathrm{d}f^1}{\mathrm{d}s}\right)^{-1} - \sum_{j\geq 2} A_j^i \frac{\mathrm{d}f^j}{\mathrm{d}s} \left(\frac{\mathrm{d}f^1}{\mathrm{d}s}\right)^{-1}, \qquad i = 1, \dots, \dim_{\mathbb{R}} M$$

and the identities  $A_i^j = A_i^j$  for  $i \ge 2, j \ge 1$ . Hence the functions  $f^i$  must be chosen such that  $\dot{\gamma}^i(s) = A_j^i(\gamma(s)) \frac{\mathrm{d}f^j(s)}{\mathrm{d}s}, i, j \ge 1$ . Since in  $\{E'_i = A_i^j \frac{\partial}{\partial x^j} = \frac{\partial}{\partial x'^i}\}$  and  $\{\frac{\partial}{\partial x^i}\}$  the tangent vector field  $\dot{\gamma}$  has the expansions  $\dot{\gamma}(s) = \dot{\gamma}'^i(s)E'_i|_{\gamma(s)} = \dot{\gamma}^i(s)\frac{\partial}{\partial x^i}|_{\gamma(s)} = A_j^i(\gamma(s))\frac{\mathrm{d}f^j}{\mathrm{d}s}|_s \frac{\partial}{\partial x^i}|_{\gamma(s)}$ , we find  $\frac{\mathrm{d}f^i(s)}{\mathrm{d}s} = \dot{\gamma}'^i(s)$ , i.e.,

$$f^{i}(s) = \int_{s_{0}}^{s} \dot{\gamma}'^{i}(\sigma) \,\mathrm{d}\sigma + c^{i} = \gamma'^{i}(s) - \gamma'^{i}(s_{0}) + c^{i}$$
(3.5)

with  $c^i \in \mathbb{R}$  being constants. In particular, if  $\gamma$  is geodesic, then  $\dot{\gamma}'^i(s) = \dot{\gamma}'^i(s_0) = \delta_0^1$  as now  $E'_1|_{\gamma(s)} = \dot{\gamma}(s)$ , so that  $f^i(s) = \delta_1^i(s - s_0) + c^i$ .

In (3.5)  $f^1: J_U \to f^1(J_U) \subseteq \mathbb{R}$  is  $C^1$  (and also  $C^{\infty}$ ) diffeomorphism but this is not the typical situation. Generally the invertability and differentiability

#### 3. The case along paths

of  $f^1(s) = \gamma'^{1}(s) - \gamma'^{1}(s_0) + c^1$ ,  $s \in J_U$  may be a problem. If we can find some  $i_0 \in \{1, \ldots, \dim_{\mathbb{R}} M\}$  for which  $f^{i_0} \colon J_U \to f^{i_0}(J_U)$ , given by (3.5), is  $C^1$  diffeomorphism, we can simply renumber the coordinates and take  $f^{i_0}$  for  $f^1$ . But this is also an exception. In the most general case, we, on the base of the implicit function theorem,<sup>9</sup> can only assert that if  $\gamma'^1 \neq 0$ ,  $s \in J_U$ , then there exists an open interval  $J'_U \subseteq J_U$  such that the restriction  $f^1|_{J'_U} \colon J'_U \to f^1(J'_U)$  is a  $C^1$  diffeomorphism.<sup>10</sup> Thereof the special coordinates  $\{x'^i\}$ , given via (3.3), are defined only for  $s \in J'_U$  with  $f^i$  given by (3.5).

Let us summarize the above discussion. We started with some  $s_0 \in J$  and a chart (U, x) with  $U \ni \gamma(s_0)$ . Then a parallel frame  $\{E'_i\}$  along  $\gamma$  was constructed,  $\nabla_{\dot{\gamma}}E'_i = 0$ , with  $E'_1|_{\gamma(s_0)} = \dot{\gamma}(s_0)$ , and coordinates  $\{x'^i\}$  in U such that  $E'_i|_p = \frac{\partial}{\partial x'^i}|_p$  for  $p \in U \cap \gamma(J) = \gamma(J_U)$  and for which (3.3) holds with  $f^i$  given by (3.5). Below we are going to prove that there exists a neighborhood of  $\gamma(s_0)$  in

which the coordinates  $\{x'^i\}$  are normal along  $\gamma$ .

As the coordinate system  $\{x^i\}$  is completely arbitrary, we can take  $\{x'^i\}$  for it,  $x^i = x'^i$ . This results in  $A(p) = [A_i^j(p)] = 1$ , with  $1 = [\delta_i^j]$  being the dim<sub> $\mathbb{R}$ </sub>  $M \times$ dim<sub> $\mathbb{R}$ </sub> M identity (called also unit) matrix, and  $\frac{\partial}{\partial x'^i} = \frac{\partial}{\partial x^i} = E'_i$ . Inserting these equalities into (3.4) and omitting the primes, we get

$$\beta_{s}^{i}(t) = \gamma^{i}(g(\xi)) + \sum_{j=2}^{\dim_{\mathbb{R}} M} \delta_{j}^{i} \left[ x^{j}(\beta_{s}(t)) - f^{j}(g(\xi)) \right] \\ - \sum_{n=2}^{\infty} \frac{1}{n!} \sum_{i_{1},\dots,i_{n}=2}^{\dim_{\mathbb{R}} M} \Gamma^{i}_{i_{1}\dots i_{n}}(\gamma(g(\xi))) \\ \times \left[ x^{i_{1}}(\beta_{s}(t)) - f^{i_{1}}(g(\xi)) \right] \cdots \left[ x^{i_{n}}(\beta_{s}(t)) - f^{i_{n}}(g(\xi)) \right].$$
(3.6)

From the last equation, we get, by differentiation with respect to  $\beta_s^i(t)$ ,  $i \ge 2$ , and restricting the result to  $\gamma$  (i.e., putting  $t = t_0$ ), the equation

$$\sum_{k=2}^{\dim_{\mathbb{R}} M} \Gamma^i_{(jk)}(\gamma(s))[x^k(\gamma(s)) - f^k(g(\xi))] = 0$$

for  $i \ge 1, j \ge 2$  and every  $\gamma(s)$ . Hence

6

$$\Gamma^{i}_{(jk)}(\gamma(s)) = 0 \quad \text{for } i \ge 1, \, j, k \ge 2, \text{ and } s \in J'_{U}.$$
 (3.7)

Moreover, by construction

$$0 = \nabla_{\dot{\gamma}} E'_i|_{\gamma(s)} = \dot{\gamma}^j(s) \nabla_{E'_j} E'_i = \dot{\gamma}^j(s) \Gamma^k_{\ ij}(\gamma(s)) E'_k|_{\gamma(s)}$$

<sup>&</sup>lt;sup>9</sup>For instance, see: [77, Chapter III, § 8], [7, Sections 1.37 and 1.38], [78, Chapter 10, Section 2], [79, Theorem 9.18]. Notice, at that precise place, the assumption that  $s_0$  is not an end point of J, if any, is explicitly used. Also the assumption of regularity of  $\gamma$  at (every)  $s_0$  is essential here.

<sup>&</sup>lt;sup>10</sup>Here we suppose  $\gamma'^1(s) \neq 0$  for all  $s \in J'_U$ . Below we point out that this may not be the case, but there always exist intervals  $J_{\delta} \ni s_0$ ,  $J_{\delta} \subseteq J_U$  such that  $\gamma'^1|_{J_{\delta}} \neq 0$ . One can define  $J'_U$  as the union of all such intervals, i.e., to take the maximal such interval for it.

or  $\dot{\gamma}^{j}(s)\Gamma^{k}_{\ ij}(\gamma(s)) \equiv 0, \ s \in J.$  Consequently, for symmetric connection  $\nabla$ , we find

$$\dot{\gamma}^{1}(s)\Gamma^{k}_{11}(\gamma(s)) + \sum_{j=2}^{\dim_{\mathbb{R}}M} \dot{\gamma}^{j}(s)\Gamma^{k}_{1j}(\gamma(s)) = 0, \quad \dot{\gamma}^{1}(s)\Gamma^{k}_{1j}(\gamma(s)) = 0, \quad j \ge 2 \quad (3.8)$$

where equation (3.7) was used, i.e.,  $\Gamma_{ij}^{k} = \Gamma_{(ij)}^{k} = 0$  for  $i, j \ge 2$  and  $k \ge 1$ . From the last equations, we conclude that  $\Gamma_{1j}^{k}(s) = 0$  for  $j \ge 1$  if  $\dot{\gamma}^{1}(s) \ne 0$ . (If  $\gamma$  is a geodesic, then  $E_{1}'|_{\gamma(s)} = \dot{\gamma}(s)$  and hence  $\dot{\gamma}^{1}(s) = \dot{\gamma}'^{1}(s) = 1$  for all  $s \in J$ , but this is not the general case.) Now it is time to use the choice  $E_{1}'|_{\gamma(s_{0})} = \dot{\gamma}(s_{0})$  again: from this it follows  $\dot{\gamma}^{1}(s_{0}) = 1$  and  $\dot{\gamma}^{j}(s_{0}) = 0$  for  $j \ge 2$ , as now  $\frac{\partial}{\partial x^{i}}|_{p} = \frac{\partial}{\partial x'^{i}}|_{p} = E_{i}'|_{p}$ . Since  $\gamma: J \to M$  is supposed to be of class  $C^{1}$ , the function  $\dot{\gamma}^{1}: J \to \mathbb{R}$  is of class  $C^{0}$ , i.e., continuous. Thereof for every  $\varepsilon \in \mathbb{R}$  with  $\varepsilon > 0$  there exists  $\delta \in \mathbb{R}, \delta > 0$  $(\delta$  may depend on  $\varepsilon$ ) such that  $|\dot{\gamma}^{1}(s) - \dot{\gamma}^{1}(s_{0})| < \varepsilon$  for  $s \in J$  and  $|s - s_{0}| < \delta$ . So, in the interval  $J_{\delta} := \{s \in J, |s - s_{0}| < \delta\} \subseteq J$  is fulfilled  $1 - \varepsilon < \dot{\gamma}^{1}(s) < 1 + \varepsilon$  as  $\dot{\gamma}^{1}(s_{0}) = 1$ .

At the end, we choose some  $\varepsilon \in \mathbb{R}$ ,  $0 < \varepsilon < 1$  and fix corresponding  $\delta \in \mathbb{R}$ ,  $\delta > 0$ . Then for  $s \in J_0 := J'_U \cap J_\delta$ , we have  $0 < 1 - \varepsilon < \dot{\gamma}^1(s)$  and hence  $\dot{\gamma}^1(s) \neq 0$ . So equations (3.8) imply  $\Gamma^k_{1j}(s) = 0$  for all  $j \ge 1$ . Combining this with (3.7) and taking into account that now the connection considered is symmetric, we finally obtain

$$\Gamma^{k}_{ij}(\gamma(s)) \equiv 0 \qquad \text{for } s \in J_0 := J'_U \cap J_\delta \tag{3.9}$$

or, equivalently,

$$\Gamma^{k}_{\ ij}\big|_{\gamma(J_{0})} = 0, \qquad J_{0} := J'_{U} \cap J_{\delta} \ni s_{0}.$$
(3.9')

The so-constructed coordinates  $\{x'^i\}$  are called *Fermi coordinates* along  $\gamma$  (with a reference path  $\gamma$ ). By definition, their domain is a subneighborhood  $\overline{U} \subseteq U$  such that  $\overline{U} \cap \gamma(J) \subseteq \gamma(J_0)$  as we supposed that  $s_0$  is not a boundary point of J, if any.

Now a few words are in order for the case when the interval J is closed from one or both ends and  $s_0$  is its end point. In such a case, the above investigation can be modified as follows. Let  $s_0$  be the left (or, resp., right) end of J. Define  $\dot{\gamma}(s_0) = \dot{\gamma}_+(s_0)$  (resp.  $\dot{\gamma}(s_0) = \dot{\gamma}_-(s_0)$ ) via (cf. (I.2.3))  $(\dot{\gamma}_\pm(s_0))(f) := \lim_{\varepsilon \to \pm 0} \frac{1}{\varepsilon} [f(\gamma(s_0 + \varepsilon)) - f(\gamma(s_0))], f \in \mathfrak{F}(\gamma(s_0))$  where  $\varepsilon \to +0$  (resp.  $\varepsilon \to -0$ ) means that  $\varepsilon$  tends to  $0 \in \mathbb{R}$  with values greater (resp. less) than zero. Now take some interval  $J' \supset J$  for which  $s_0$  is inner, not boundary, point and certain regular  $C^1$  path  $\gamma' : J' \to M$ without self-intersections such that  $\gamma'|_J = \gamma$  and  $\dot{\gamma}'(s_0) = \dot{\gamma}(s_0)$ .<sup>11</sup> All of the above results and conclusions are completely valid for  $\gamma'$  at  $s = s_0 \in J'$ . Consequently, applying them to  $\gamma'$  and restricting the results to  $\gamma = \gamma'|_J$ , we obtain their true versions for any point along  $\gamma$ , including the one(s) corresponding to the end(s) of

<sup>&</sup>lt;sup>11</sup>The choice of J' and  $\gamma'$  can be made in infinitely many ways. The procedure just described is actually the definition of  $\gamma^i$  (or of  $\gamma$ ) as  $C^1$  mappings on a manifold with boundary, the interval J in the particular case, according to the definition in Footnote I.5 on page 7.

J. Of course, the construction of the Fermi coordinates at the end point(s) of J depends on the particular choice of  $\gamma': J' \to M$  but this is not essential as we are interested only in their existence, not in their concrete properties.

Generally the Fermi coordinates are local and are not defined along the whole path  $\gamma$ . Since the point  $s_0$  is arbitrarily fixed in the above considerations, the main moral from them is:

**Proposition 3.1.** Let  $\gamma: J \to M$  be regular  $C^1$  path without self-intersections in  $C^{\infty}$  manifold endowed with  $C^{\infty}$  linear connection. For every point  $p \in \gamma(J)$  there is its neighborhood on which Fermi coordinates exist.

**Proposition 3.2.** In Fermi coordinates, the coefficients of a symmetric linear connection vanish along the part of their reference path lying in their domain.

**Corollary 3.1.** If  $\gamma: J \to M$  is a regular  $C^1$  path without self-intersections in  $C^{\infty}$ manifold endowed with  $C^{\infty}$  symmetric linear connection, then for every  $s \in J$ exists a neighborhood U of  $\gamma(s)$  and coordinates in it, in which the connection coefficients vanish on  $\gamma(J) \cap U$ .

Remark 3.2. When proving the existence of the Fermi coordinates, we have used implicitly several times the absence of self-intersection points of  $\gamma$ . This is most evident in (3.3): if  $\gamma$  has self-intersection points, at them  $x'^i(\beta_s)$  and  $x^i(\beta_s)$  are not injective functions of s and, hence,  $f^1$  can not be a diffeomorphism from J on  $f^1(J)$ . It is evident that the above considerations hold on any 'part' of  $\gamma$  without self-intersections, if any.

Remark 3.3. Our construction of the Fermi coordinates is essentially more precise and, correspondingly, longer than the one in [19, pp. 166–169]. In this reference a number of implicit conditions are presupposed. For example:  $\gamma$  is regular and without self-intersections,  $\gamma$  is contained in a single coordinate neighborhood (one and the same for  $\{x^i\}$  and  $\{x'^i\}$ ),  $\dot{\gamma}^1(s) \neq 0$  for all  $s \in J$ , etc.

Remark 3.4. Often, especially in the physical literature, the Fermi coordinates are referred as ones in which the coefficients of a symmetric linear connection vanish along the *whole* path  $\gamma$ . This is true if  $\gamma$  is geodesic (or a path 'near' to geodesic), contained in a single coordinate neighborhood, but in the general case this is a wrong assertion. Generally along  $\gamma$  we can construct a family of Fermi coordinates such that  $\gamma(J)$  is contained in the union of their domains and the connection coefficients in them vanish along the corresponding pieces of  $\gamma$ . Said in other words, as a rule there is not a single Fermi system of coordinates in whose domain  $\gamma(J)$  is contained and in which the connection coefficients vanish on the whole set  $\gamma(J)$ .

Remark 3.5. If the connection is non-symmetric, in the constructed Fermi coordinates vanish the symmetric parts of some, but generally not all, of its coefficients – see (3.7) and (3.8). As the symmetrized connection coefficients are coefficients of a symmetric linear connection (see (I.3.9) and the comments before it), we can construct Fermi coordinates for this symmetric connection. In these particular co-

ordinates, the symmetric parts of the initial connection coefficients vanish along their reference path.

For torsionless linear connection, in the Fermi coordinates, hold not only equations (3.7), but we also have (cf. (I.6.11))

$$\Gamma^{i}_{i_{1}...i_{n+1}}(p) = 0 \qquad n \in \mathbb{N}, \ p \in \gamma(J'_{U}), \ i \ge 1, \ i_{1}, \dots, i_{n+1} \ge 2 \quad (3.10a)$$

$$\frac{\partial}{\partial x^{(i_1}} \Gamma^i_{i_2\dots i_{n+2}}(p) = 0 \qquad n \in \mathbb{N}, \ p \in \gamma(J'_U), \ i \ge 1, \ i_1, \dots, i_{n+2} \ge 2$$
(3.10b)

which equalities are obtained from (3.6) by repeated differentiation.

From the construction of the Fermi coordinates, a conclusion can be made that these coordinates are not the only ones (locally) normal along pieces of a given path. For instance, one can construct some analogue of the geodesic normal coordinates at a given point (see page 48) by replacing in (3.1) and in the next formulae  $\Gamma^i_{i_1,\ldots,i_n}$  by some constants  $c^i_{i_1,\ldots,i_n}$ . We are not going to describe here such modifications because further a complete description of the coordinates normal along a path will be presented.

On the contrary to the Fermi coordinates along  $\gamma$ , which are essentially local, one can construct global frames along  $\gamma$  which are normal along the *whole* path  $\gamma$ . Actually, let  $\gamma: J \to M$ , J being open  $\mathbb{R}$ -interval, be regular  $C^1$  path without self-intersections and  $\{J_0^{\alpha} | \alpha \in A\}$ , A being a non-empty set, be an open cover of J such that for every  $p \in \gamma(J)$  the set  $A_p := \{\alpha | \alpha \in A, \gamma(J_0^{\alpha}) \ni p\}$  consists of one or two elements and for each  $\alpha \in N$  exist Fermi coordinates  $\{x_{\alpha}^i\}$  with domain  $U_{\alpha}$  such that  $U_{\alpha} \cap \gamma(J) = \gamma(J_0^{\alpha})$ . Define along  $\gamma$  a frame  $\{E_i\}$  such that, for every  $p \in \gamma(J)$ ,  $E_i|_p = \frac{\partial}{\partial x_{\alpha}^i}|_p$  if  $\gamma(J_0^{\alpha}) \in p$  for a single  $\alpha \in A$  and if the set  $A_p$  consists of two elements, we arbitrarily choose some  $\beta \in A_p$  and set  $E_i|_p = \frac{\partial}{\partial x_{\beta}^i}|_p$ . The soobtained frame  $\{E_i\}$  is normal along  $\gamma$  but generally it is not smooth, even it may not be continuous, along the whole path  $\gamma$ . Hence the holonomicity problem for this frame is globally ill-posed.<sup>12</sup> All other frames normal along  $\gamma$  can be obtained from  $\{E_i\}$  according to the recipe of Proposition I.5.2.

At the end, we want to mention one special type of Fermi coordinates in a neighborhood of (injective normal) geodesic in Riemannian manifold.

**Example 3.1.** Let  $\gamma: J \to M$  be geodesic in a Riemannian manifold M with metric g. Let  $\{E_i\}$  be a parallel frame along  $\gamma$  with  $E_1 = \dot{\gamma}$ , i.e.,  $\nabla_{\dot{\gamma}} E_i = 0$  and  $E_1 = \dot{\gamma}$ ; we can also say that  $\{E_i|_{\gamma(s)}\}$ ,  $s \in J$  is obtained from some fixed basis  $\{e_i\}$  in  $T_{\gamma(s_0)}(M)$ , with  $s_0 \in J$  and  $e_1 = \dot{\gamma}(s_0)$ , by a parallel transporting it from  $\gamma(s_0)$  to the other points in  $\gamma(J)$ . For  $(s, t) = (s, t^2, \ldots, t^{\dim_{\mathbb{R}} M}) \in J \times \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ , we define  $i: (s, t) \mapsto \sum_{i \geq 2} t^i E_i|_{\gamma(s)} \in T_{\gamma(s)}(M)$ . If  $\gamma$  is injective, i.e., without self-intersections, and normal, i.e.,  $g(\dot{\gamma}, \dot{\gamma}) = 1$ , in [8, § 3.8 (ii)] it is proved that there

<sup>&</sup>lt;sup>12</sup>With a little attention, one can see that locally  $\{E_i\}$  is smooth and holonomic on the part of  $\gamma$  corresponding to a single interval  $J_0^{\alpha}$ , but in the regions of intersections of such intervals the smoothness of the frame may be lost.

exists a neighborhood  $V_0$  of the zero vector in  $\mathbb{R}^{\dim_{\mathbb{R}} M-1}$  and a neighborhood U of  $\gamma(J)$  in M such that the mapping  $y := (\exp \circ i|_{J \times V_0})^{-1} \colon U \to J \times V_0 \subseteq \mathbb{R}^{\dim_{\mathbb{R}} M-1}$  is diffeomorphism. The associated to the chart (U, y) coordinates  $\{y^i\}$  are called in [8, § 3.8 (ii)] Fermi coordinates. In the same reference it is proved that the coordinates  $\{y^i\}$  so-defined are normal along  $\gamma$ ,  $\Gamma^i_{\ jk} \circ \gamma = 0$ , and orthonormal,  $g_{ij}(\gamma(s)) = \pm \delta_{ij}, \ s \in J$ , at the same time. Thereof we conclude, the Fermi coordinates along injective (without self-intersections) normal geodesic are similar to the orthonormal Riemannian coordinates found by Birkhoff (see page 48), but when the latter are considered 'along paths'.

**Exercise 3.1.** Prove that the coordinates of Example 3.1 agree with our definition of 'Fermi coordinates'. (Hint:  $\beta_s(t) = x^{-1}(s, \hat{t})$  where one of the components of  $\hat{t} \in \mathbb{R}^{\dim_{\mathbb{R}} M-1}$  is equal to t, the others being constants.)

## **3.2.** Complete description

As we said earlier, the second method for proving the existence of coordinates or frames normal along a path  $\gamma: J \to M$  is via direct solving of the equation (I.5.4) along  $\gamma$ :

$$\left(\Gamma_k A + E_k(A)\right)\Big|_{\gamma(J)} = 0 \tag{3.11}$$

or

$$\Gamma_k(\gamma(s))A(\gamma(s)) + \left(E_k(A)\right)\Big|_{\gamma(s)} = 0.$$
(3.11')

Let us recall the notation here. The coefficients of a linear connection  $\nabla$  are  $\Gamma^{i}_{jk}$ and  $\Gamma_{k} := [\Gamma^{i}_{jk}]_{i,j=1}^{\dim_{\mathbb{R}}M}$  are their matrices in an arbitrarily chosen frame  $\{E_{i}\}$  in a neighborhood of  $\gamma(J)$ . We want to find a  $C^{1}$  matrix-valued function  $A := [A_{i}^{j}]$ such that the frame(s)  $\{E'_{i} = A_{i}^{j}E_{j}\}$  is (are) normal along  $\gamma$ , i.e., on  $\gamma(J)$ . A necessary and sufficient condition for this is A to be a solution of (3.11).

Now we pose a problem: given a  $C^2$  manifold M endowed with symmetric  $C^0$  connection  $\nabla$  and a regular  $C^1$  path  $\gamma: J \to M$  without self-intersection, describe all frames and local coordinates normal respectively along  $\gamma$  and on (some part(s) of)  $\gamma(J)$ . Below we describe the general solution of this problem by adapting the method of [76, Section 3] to the particular situation just outlined; we also employ the initial idea underlying the construction of coordinates normal along a path in [25, pp. 428–431]. We shall see that the solution in terms of frames is easier and more simple than the one in terms of local coordinates.<sup>13</sup> This is not occasional: to a coordinate system  $\{x^i\}$  corresponds the frame  $\{\partial/\partial x^i\}$ , but to obtain the local coordinates corresponding to a holonomic frame  $\{A_i^j\partial/\partial x^j\}$ , one has to solve a system of first-order partial differential equations. Said differently, the normal frames are obtained by solving a first-order system of partial differential equations.

<sup>&</sup>lt;sup>13</sup>The same situation will be met in Chapters III–V for much more general problems.

(see (I.5.4)) while the corresponding system for the normal coordinates is of second order (see (I.5.4') on page 41).<sup>14</sup>

Below we are going to prove the existence of solutions of (3.11) and to find its general solution, which results in a complete description of all frames normal along a regular  $C^1$  path. The main results are formulated below as Theorem 3.1 on page 97 to which theorem the reader can jump right now if he/she is not interested in the details of its proof.

To begin with, we shall construct local coordinates in a neighborhood of a point in  $\gamma(J)$  such that a point in their domain has the path's parameter as its first coordinate.

**Lemma 3.1.** Let  $\gamma: J \to M$  be a regular  $C^1$  injective path in a  $C^3$  manifold M. For every  $s_0 \in J$ , there exists a chart  $(U_1, x)$  of M such that  $\gamma(s_0) \in U_1, x: U_1 \to J_1 \times \mathbb{R}^{\dim_{\mathbb{R}} M-1}$  for some open subinterval  $J_1 \subseteq J$ ,  $s_0 \in J_1$  and  $x(\gamma(s)) = (s, \mathbf{t}_0)$  for all  $s \in J_1$  and some fixed  $\mathbf{t}_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ .

Proof. Let  $s_0 \in J$  be a point in J which is not an end point of J, if any, and (U, y) be a chart with  $\gamma(s_0)$  in its domain,  $U \ni \gamma(s_0)$ , and  $y: U \to \mathbb{R}^{\dim_{\mathbb{R}} M}$ . From the regularity of  $\gamma, \dot{\gamma} \neq 0$ , follows that at least one of the numbers  $\dot{\gamma}_y^1(s_0), \ldots, \dot{\gamma}_y^{\dim_{\mathbb{R}} M}(s_0)$ , where  $\gamma_y^i := y^i \circ \gamma$ , is non-zero. We, without lost of generality, choose this non-vanishing component to be  $\dot{\gamma}_y^1(s_0)^{.15}$  Then, due to the continuity of  $\dot{\gamma}$  ( $\gamma$  is of class  $C^1$ ) and according to the implicit function Theorem [77, Chapter III, § 8], [7, Sections 1.37 and 1.38], [78, Chapter 10, Section 2], there exists an *open* subinterval  $J_1 \subseteq J$  containing  $s_0, J_1 \ni s_0$ , and such that  $\dot{\gamma}^1|_{J_1} \neq 0$  and the restricted mapping  $\gamma_y^1|_{J_1}: J_1 \to \gamma_y^1(J_1)$  is a  $C^1$  diffeomorphism on its image. Define a neighborhood

$$U_1 := \left\{ p | p \in U, \ y^1(p) \in \gamma_y^1(J_1) \right\} = y^{-1} \left( \gamma_y^1(J_1) \times \mathbb{R}^{\dim_{\mathbb{R}} M - 1} \right) \ni \gamma(s_0)$$

and a chart  $(U_1, x)$  with local coordinate functions

$$x^{1} := (\gamma_{y}^{1}|_{J_{1}})^{-1} \circ y^{1}$$
  

$$x^{k} := y^{k} - \gamma_{y}^{k} \circ x^{1} + t_{0}^{k} \qquad k = 2, \dots, \dim_{\mathbb{R}} M$$
(3.12)

where  $t_0^k \in \mathbb{R}$  are arbitrarily fixed constant numbers. Since  $\frac{\partial x^1}{\partial y^j} = \frac{1}{\dot{\gamma}_y^1} \delta_j^1$ ,  $\frac{\partial x^k}{\partial y^1} = -\frac{\dot{\gamma}_y^k}{\dot{\gamma}_y^1}$  for  $k \ge 2$ , and  $\frac{\partial x^k}{\partial y^l} = \delta_l^k$  for  $k, l \ge 2$ , the Jacobian of the change  $\{y^i\} \to \{x^i\}$  at  $p \in U_1$  is  $\frac{1}{\dot{\gamma}^1(p)} \ne 0, \infty$ . Consequently  $x: U_1 \to J_1 \times \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$  is really a coordinate homeomorphism with coordinate functions  $x^i$ .

 $<sup>^{14}{\</sup>rm This}$  is realized implicitly in [25, p. 429] in the method developed in this work for finding a class of coordinates normal along a path.

<sup>&</sup>lt;sup>15</sup>If it happens that  $\dot{\gamma}_y^1(s_0) = 0$  and  $\dot{\gamma}_y^{i_0}(s_0) \neq 0$  for some  $i_0 \neq 1$ , we have simply to renumber the local coordinates to get  $\dot{\gamma}_y^1(s_0) \neq 0$ . Practically this is a transition to new coordinates  $\{y^i\} \to \{z^i\}$  with  $z^1 = y^{i_0}$  and, for instance,  $z^{i_0} = y^1$  and  $z^i = y^i$  for  $i \neq 1, i_0$ , in which the first component of  $\dot{\gamma}$  is non-zero. We suppose that, if required, this coordinate change is already done. If occasionally it happens that  $\dot{\gamma}_y^{j_0}(s) \neq 0$  for all  $s \in J$  and fixed  $j_0$ , it is extremely convenient to take this particular component of  $\dot{\gamma}$  as  $\dot{\gamma}_y^1$  – see the next sentence.

In the new chart  $(U_1, x)$ , the coordinates of  $\gamma(s)$ ,  $s \in J_1$  are

$$\gamma^{1}(s) := (x^{1} \circ \gamma)(s) = s, \quad \gamma^{k}(s) := (x^{k} \circ \gamma)(s) = t_{0}^{k}, \ k \ge 2,$$
(3.13)

i.e.,

$$x(\gamma(s)) = (s, \boldsymbol{t}_0)$$

for some  $\mathbf{t}_0 = (t_0^2, \dots, t_0^{\dim_{\mathbb{R}} M}) \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$  and all  $s \in J_1$ .

Lemma 3.1 means that the chart  $(U_1, x)$  is so luckily chosen that the first coordinate in it of a point along  $\gamma$  coincides with the value of the corresponding path's parameter, the other coordinates being constant (along  $\gamma$ ) numbers. Moreover, in  $U_1$  the path  $\gamma$  can be considered as a representative of a family of paths  $\eta(\cdot, t): J_1 \to M$ ,  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ , defined by  $\eta(s, t) := x^{-1}(s, t)$  for  $(s, t) \in J_1 \times \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ ; indeed,  $\gamma = \eta(\cdot, t_0)$  or  $\gamma(s) = \eta(s, t_0), s \in J_1 \subseteq J$ .

Remark 3.6. In this way we have obtained a natural foliation of  $U_1$  along  $\gamma$ : putting  $V_s := \{x^{-1}(s, t) | t \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}\}$ , we see that  $\bigcup_{s \in J_1} V_s = U_1, V_s \cap V_{s'} = \emptyset$ for  $s \neq s', s, s' \in J_1$ , and  $V_s \cap \gamma(J_1) = \gamma(s)$  for  $s \in J_1$ . Having this in mind, the main idea of the next considerations in this subsection is in each  $(\dim_{\mathbb{R}} M - 1)$ dimensional submanifolds  $V_s$  (see p. 7) to be constructed frames or coordinates normal at the single point  $\gamma(s)$  (as a point of  $V_s$ ) and then to joint them smoothly along  $\gamma$ . We shall see in Section 5 below that this is possible due to the onedimensionality of  $\gamma(J_1)$  considered as a submanifold of M.

Remark 3.7. The vector field  $\dot{\gamma}$  tangent to  $\gamma$  has in the coordinates  $\{x^i\}$ , associated with the chart  $(U_1, x)$  provided by Lemma 3.1, the components  $(1, 0, \ldots, 0)$ . Consequently, if  $\gamma$  is a geodesic path (see Subsection I.3.4), the geodesic equation (I.3.23) reduces to

$$\Gamma^i_{11} \circ \gamma = 0. \tag{3.14}$$

Hence  $\{x^i\}$  is generally not normal along a geodesic path, but along such a path the last equation holds.

*Remark* 3.8. We have met examples of concrete kinds of coordinates systems  $\{x^i\}$  in Examples I.7.3, I.7.4 and I.7.5 in which they happen to be normal along the paths considered in them.

Now we are ready to solve equation (3.11) in the neighborhood  $U_1$ . Let us take an arbitrary point  $p \in U_1$  with coordinates  $(s, t) \in J_1 \times \mathbb{R}^{\dim_{\mathbb{R}} M-1}$  in  $(U_1, x)$ , i.e., x(p) = (s, t) or  $p = \eta(s, t) = x^{-1}(s, t)$ . The mapping  $p \mapsto p_0 := \gamma(x^1(p)) \in \gamma(J_1)$ defines a useful for our purposes 'projection' of p on  $\gamma$  (or on  $\gamma(J)$ ) along the family  $\eta$ ; in coordinates, if  $p = \eta(s, t)$ , then  $p_0 = \gamma(s) = \eta(s, t_0)$ , i.e.,  $x(p_0) = (s, t_0)$  for x(p) = (s, t).

Suppose A is of class  $C^3$ . Then [57, Section 4.10-5] there exist  $C^1$  matrixvalued functions  $B_{kl}$  on  $U_1$  such that they and their partial derivatives are bounded
on  $\gamma(J_1)$  and the following Taylor formula holds:

$$A(\eta(s, t)) = A(\eta(s, t_0)) + \sum_{k=2}^{\dim_{\mathbb{R}} M} \frac{\partial A(\eta(s, t))}{\partial t^k} \Big|_{t=t_0} (t^k - t_0^k) + \sum_{k,l=2}^{\dim_{\mathbb{R}} M} B_{kl}(\eta(s, t))(t^k - t_0^k)(t^l - t_0^l). \quad (3.15)$$

In (3.11) we choose  $E_j|_p = \frac{\partial}{\partial x^j}|_p$ ,  $p \in U_1$ , so that  $E_1(A)|_{\gamma(s)} = \frac{dA(\gamma(s))}{ds}$  and  $E_k(A)|_{\gamma(s)} = \frac{\partial A(\eta(s,t))}{\partial t^k}|_{t=t_0}$ ,  $k \ge 2$  as  $\gamma(s) = \eta(s, t_0)$ . Substituting (3.15) into (3.11'), we split (3.11') into equivalent to it system of equations

$$\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} = -\Gamma_1(\gamma(s))A(\gamma(s)), \qquad (3.16a)$$

$$\frac{\partial A(\eta(s, t))}{\partial t^k}\Big|_{t=t_0} = -\Gamma_k(\gamma(s))A(\gamma(s)), \qquad k \ge 2.$$
(3.16b)

**Lemma 3.2.** Let J be  $\mathbb{R}$ -interval,  $s_0 \in J$  be fixed, and Z be continuous, square matrix-valued, and non-zero function on J. On J exists a unique (square) non-degenerate matrix-valued function Y satisfying the initial-valued problem:

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = ZY, \qquad Y|_{s=s_0} = 1, \quad Y = Y(s, s_0; Z)$$
(3.17)

where  $s \in J$  and  $\mathbb{1}$  is the identity (unit) matrix of the corresponding size.

*Proof.* See [34, Chapter IV, Section 1] or [82, pp.125–127].<sup>16</sup>  $\Box$ 

Remark 3.9. This lemma and its multidimensional version (see Lemma 4.1) play a crucial role in our general approach for investigating normal frames. They are simple corollaries of the theorems of existence and uniqueness for normal systems of ordinary or partial differential equations of first order. By means of Y, we can express every solution of the matrix equation

$$\frac{\mathrm{d}X}{\mathrm{d}s} = ZX \tag{3.18}$$

with the  $C^1$  matrix-valued function X being of the same size (order) as Z in (3.17). The general solution of (3.18) is

$$X = Y(s, s_0; Z) (3.19)$$

with arbitrary  $s_0 \in J$ , or

$$X = Y(s, s_0; Z) X_0 (3.20)$$

<sup>&</sup>lt;sup>16</sup>The solution of (3.17) is called *matricant*.

with fixed  $s_0 \in J$  and arbitrary  $X_0$ . The particular solution of (3.18) satisfying the initial condition

$$X|_{s=s_0} = X_0 \tag{3.21}$$

for some  $s_0 \in J$  and  $X_0$  is given by (3.20). At present, special properties of Y, such as  $Y^{-1}(s, s_0; Z) = Y^{-1}(s_0, s; Z)$  or  $Y(s, s_1; Z)Y(s_1, s_0; Z) = Y(s, s_0; Z)$ , will not be needed.

Remark 3.10. If the order of the matrices in (3.17) is important, we write  $\mathbb{1}_n$  for  $\mathbb{1}$  with  $\mathbb{1}_n$  being the  $n \times n$ ,  $n \in \mathbb{N}$ , identity matrix; respective labels may be attached to Y and Z if required.

The general solution of equation (3.16a), which immediately follows from Lemma 3.2 and Remark 3.9, is

$$A(\gamma(s)) = Y(s, s_0; -\Gamma_1 \circ \gamma)B \tag{3.22}$$

where Y is defined by the initial-value problem (3.17), with  $1 = 1_{\dim_{\mathbb{R}} M}$ , and B is constant non-degenerate  $\dim_{\mathbb{R}} M \times \dim_{\mathbb{R}} M$  matrix. Inserting (3.16b) in (3.15) and using (3.22), we find:

$$A(p) = \left\{ \mathbb{1} - \Gamma_k(p_0) [x^k(p) - x^k(p_0)] \right\} Y(x^1(p), s_0; -\Gamma_1 \circ \gamma) B + B_{kl}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]$$
(3.23)

where we have used  $p = \eta(s, t)$ ,  $\eta = x^{-1}$ , and  $p_0 := \gamma(x^1(p)) = \gamma(s)$ . Note, in (3.23) the terms corresponding to k = 1 and/or l = 1 vanish due to  $x^1(p_0) = x^1(\gamma(x^1(p))) = (x^1 \circ \gamma)(x^1(p)) = x^1(p) = s.^{17}$ 

In the above investigation we have supposed  $s_0$  not to be an end-point of J, if any. This restriction is completely inessential for the final results, like (3.22) and (3.23). Actually, if J has end point(s), we, analogously to the scheme described on page 90, can proceed as follows. If  $s_0$  is the left/right end of J, we define  $\dot{\gamma}(s_0)$  such that  $(\dot{\gamma}f)(s_0), f \in \mathfrak{F}(\gamma(s_0))$  is the left/right derivative of  $f \circ \gamma$  at  $s = s_0$ . Take an open interval  $J' \supset J$  and let  $\gamma': J' \to M$  be such that  $\gamma'|_J = \gamma$  and  $\dot{\gamma}'|_J = \dot{\gamma}$ . Applying all of the above considerations for  $\gamma'$  instead of  $\gamma$  and restricting the results to  $\gamma$  (or to  $\gamma(J)$ ), we see that all of the above investigation remain valid with the only changes that now the interval  $J_1$  (after the restriction) may turn to be closed from its left/right end and correspondingly  $U_1$  may be neighborhood with boundary, i.e., a (dim<sub> $\mathbb{R}$ </sub> M)-dimensional submanifold with boundary (see Remark I.2.1 and Footnote I.5 on page 7).

Let us explicitly formulate the results obtained.

**Theorem 3.1.** Let  $\gamma: J \to M$  be a  $C^1$  regular path without self-intersections in  $C^3$ manifold M endowed with (arbitrary)  $C^0$  linear connection  $\nabla$ . For every  $s_0 \in J$ , there exist a neighborhood  $U_1$  of  $\gamma(s_0)$  and a frame  $\{E'_i\}$  on  $U_1$  which is normal for

<sup>&</sup>lt;sup>17</sup>If we would have started with some  $i_0 \neq 1$ , for which  $\dot{\gamma}^{i_0} \neq 0$ , the result will be (3.22) and (3.23) with  $\Gamma_{i_0}$  instead of  $\Gamma_1$ . In the intermediary steps, e.g., starting from (3.12), the index 1 must be replace with  $i_0$ . For instance,  $p_0$  will be defined by  $p_0 := \gamma(x^{i_0}(p))$ , so  $x^{i_0}(p_0) = x^{i_0}(p)$  and the corresponding terms with  $k = i_0$  and/or  $l = i_0$  in (3.23) will vanish.

 $\nabla$  along  $\gamma$ , i.e., it is normal on  $U_1 \cap \gamma(J)$ . Moreover, in  $U_1$  exist coordinates  $\{x^i\}$ such that  $\gamma^1(s) = s$  and  $\gamma^i(s) = \text{const}$ , for  $i \geq 2$  and  $s \in J_1$ , with  $J_1 \subseteq J$  being a subinterval of J and  $\gamma(J_1) \subset U_1$ . All such normal frames are  $\{E'_i|_p = A^j_i(p)\frac{\partial}{\partial x^j}|_p\}$ ,  $p \in U_1$ , where the  $C^3$  non-degenerate matrix-valued function  $A = [A^j_i]$  is given via (3.23).

*Remark* 3.11. This result is considerably more general than we expected at the beginning: it concerns *all* linear connections, with or without torsion.

*Proof.* The first part of the theorem follows from the above-constructed solutions (3.22) for every  $s_0 \in J$  and *arbitrary*, symmetric or not, connection  $\nabla$ , as we did not use any assumption on the symmetry of its coefficients. The second assertion is a consequence of the fact that (3.22), with arbitrary B, is the general solution of (3.11) in the coordinates  $\{x^i\}$  in  $U_1$  defined by (3.12).

Remark 3.12. If we replace the  $C^3$  differentiability of A with  $C^1$  differentiability, then (3.23) will be a solution of (3.11) but we cannot assert that it will be the general one. So, if M is of class  $C^2$  and A is of class  $C^1$ , then Theorem 3.1 remains valid with an exception that there may exist matrices of class  $C^1$  but not  $C^3$  that are solutions of (3.11) and cannot be obtained by (3.23) for suitable choice of B and  $B_{kl}$ .

Said in a more free language, Theorem 3.1 asserts that along a path without self-intersections there exist frames locally normal for any linear connection. The assumption for the absence of points of self-intersections is important at the moment: at such points, if any, the coordinates  $\{x^i\}$  given via (3.12) do not exist and, correspondingly, our proof fails. It is easily seen, this is a general feature, not one characterizing our particular proof: since the mapping  $\gamma: J \to M$  is at least two-to-one at the points of self-intersections, there do not exist normal coordinates in their neighborhoods along  $\gamma$ . But, if we restrict  $\gamma$  to any subinterval of J on which it is injective, Theorem 3.1 is completely valid to the obtained path.

The constancy of B in (3.22) agrees with Proposition I.5.2. For, e.g., B = 1, we obtain a particular frame locally normal along  $\gamma$ , say  $\{E_i\}$ . Any other normal frame, say  $\{E'_i\}$ , is given by  $E'_i = A^j_i E_j$  with  $(E_i A)|_{\gamma(J)} = 0$ ,  $A = [A^j_i]$ , according to Proposition I.5.2. Combining this with (3.22) for some B, we get  $A|_{\gamma(J)} = B$ .

Let us turn now our attention to (3.23). Its meaning is that in  $U_1$  we have constructed a frame  $\{E'_i|_p = A^j_i(p)E\frac{\partial}{\partial x^j}|_p\}$  which is normal along  $\gamma$ , i.e., on  $\gamma(J) \cap U_1$ . By virtue of Proposition I.5.2, every frame on  $U_1$  which is normal along  $\gamma$  is of the form  $\{C^j_iE'_j\}$  where  $C = [C^j_i]$  is  $C^1$  non-degenerate matrix valued function on  $U_1$  such that  $E'_i(C)|_{\gamma(J)} = 0$ . The holonomicity of these frames on  $U_1 \setminus (\gamma(J) \cap U_1)$  may be completely arbitrary (see Remark I.5.1 on page 40), even it may change from point to point, but on  $\gamma(J) \cap U_1$  part of them are holonomic if the connection is torsion free on this set (Corollary I.5.3);<sup>18</sup> one can check this directly, if required, by using (3.23).

 $<sup>^{18}\</sup>mathrm{As}~\gamma(J)$  is 1-dimensional submanifold of M, the condition in Corollary I.5.3 is identically satisfied.

**Proposition 3.3.** Let  $\gamma: J \to M$  be  $C^1$  regular injective path in a  $C^2$  manifold M endowed with  $C^1$  linear connection  $\nabla$ . There exist frames normal along the whole path  $\gamma$  for  $\nabla$ .

Proof. If it happens that  $\gamma(J)$  lies entirely in  $U_1$ , the frame normal along  $\gamma$  constructed above is global, i.e., defined on the whole set  $\gamma(J)$ . But this may not be the case. Generally, we can only build an open cover  $\{J_1^{\alpha} | \alpha \in A\}$  of J and neighborhoods  $U_1^{\alpha} \supset \gamma(J_1^{\alpha})$  on which frames  $\{E_i^{\alpha}\}$  normal along  $\gamma$  exist. From these frames can be constructed global frames  $\{E_i^{\alpha}\}$ , normal along  $\gamma$ , as follows. If for  $s \in J$  exists a unique  $\alpha \in A$  with  $J_1^{\alpha} \ni s$ , we put  $E_i|_{\gamma(s)} = E_i^{\alpha}|_{\gamma(s)}$ ; if the set  $P_s := \{\alpha | \alpha \in A, \ \gamma(s) \in \gamma(J_1^{\alpha})\}$  consists of more than one element, we choose some  $\beta \in P_s$  and put  $E_i|_{\gamma(s)} = E_i^{\beta}|_{\gamma(s)}$ .

Generally the frames  $\{E_i\}$  will not be differentiable, or even not continuous, in the regions of intersection of two or more neighborhoods  $U_1^{\alpha}$ . Since on every  $U_1^{\alpha}$  Theorem 3.1 gives a complete description of the frames (locally) normal along  $\gamma$ , we can combine this with the just described construction of a frame along the whole path  $\gamma$  to obtain a *full description of all frames globally normal along*  $\gamma$ .

At that moment, without any effort, we can make one generalization by partially omitting the assumption of the lack of self-intersection points of  $\gamma$ .

**Theorem 3.2.** Let  $\gamma: J \to M$  be locally injective  $C^1$  regular path in a  $C^2$  manifold M with  $C^0$  linear connection  $\nabla$ . There exists (global, on  $\gamma(J)$ ) frames normal along the whole path  $\gamma$  for  $\nabla$ . Locally these frames are of class  $C^1$  but globally they need not to be such. The frames are (locally) holonomic iff  $\nabla$  is torsionless on  $\gamma(J)$ .

*Proof.* Let  $\gamma: J \to M$  be locally injective path, i.e., for every  $s \in J$  there is a subinterval  $J' \subseteq J$  such that the restricted path  $\gamma|_{J'}: J' \to M$  is injective (without self-intersections). Let  $J_I := \{s | s \in J, \gamma(s) \text{ is self-intersection point}\} \subset J$ . From our assumption follows that  $J_I$  has no condensation points. Hence, if  $J_I \neq \emptyset$ , for every  $s \in J$  there are three possibilities: (i) there exist  $s', s'' \in J_I, s' < s''$  such that  $s' < s \le s''$ ; (ii) there is  $s' \in J_I$  such that  $s \le s'$  and if  $s'' \in J_I$ , then  $s'' \ge s'$ ; (iii) there is  $s'' \in J_I$  such that s > s'' and if  $s' \in J_I$ , then  $s' \leq s''$ . Therefore, in any case, for  $s \in J$  there exists a maximal (in J) interval  $J_s \subseteq J$  containing s and such that  $J_s \cap J_I$  is the empty set or consists of only one element; in the above notation, we, respectively, have  $J_s = \{s_1 | s_1 \in J, s' < s_1 \le s''\}, J_s = \{s_2 | s_2 \in J, s_2 \le s'\},\$  $J_s = \{s_3 | s_3 \in J, s_3 > s''\}$ . Evidently,  $J_s = J_t$  iff  $s \in J_t$  or  $t \in J_s$ . Therefore J can be represented as  $J = \bigcup_{\lambda \in \Lambda} J^{\lambda}$  where  $\Lambda \neq \emptyset$  for  $J_I \neq \emptyset$  and the intervals  $J^{\lambda}$  are such that  $J^{\lambda} \cap J_I$  is the empty set or contains only one element and if  $s \in J^{\lambda}$ , then  $J^{\lambda} = J_s$ . Consequently, the above construction of a global frame normal along a path without self-intersections is applicable to every restricted path  $\gamma|_{I\lambda}$ , viz., for every  $\lambda \in \Lambda$ , along  $\gamma|_{J^{\lambda}}$  there is a (global) normal frame  $\{E_i^{\lambda}\}$ . At the end, for every  $p \in \gamma(J)$  we define a basis  $\{E_i|_p\}$  such that  $E_i|_p = E_i^{\lambda}|_p$  if  $p \in \gamma(J^{\lambda})$  for some (unique)  $\lambda \in \Lambda$ . The bases  $\{E_i|_p | p \in \gamma(J)\}$  form a frame along the whole

path  $\gamma$  which is normal by construction. Obviously, any global frame normal along the whole path  $\gamma$  can be obtained in this way.

Notice, all frames normal along  $\gamma$  can be constructed in the pointed way or from a single such frame according to the rule reflected in Proposition I.5.2.

Now we turn our attention to the coordinates normal along a given path which, as we know from Proposition I.5.4, could exist only for connections that are torsion free along  $\gamma$ . Our main result is formulated below as Theorem 3.3 on page 103 and the reader who is not interested in its proof and the construction of coordinates (locally) normal along  $\gamma$  can skip the text till it.

Let M be a  $C^3$  manifold endowed with symmetric linear connection and  $\gamma: J \to M$  be a regular  $C^1$  path without self-intersections in it. The problem is to be found coordinates  $\{z^i\}$  (locally) normal along  $\gamma$ . Such coordinates exist due to the symmetry of the connection and Corollary 3.1.

We start with the already employed chart  $(U_1, x)$  with coordinate functions  $x^i$  given by (3.12). Let  $(U_z, z)$  be a chart with  $U_z \cap U_1 \supset \gamma(J_1)$ . Since the transition functions  $z^{-1} \circ x$  and  $x^{-1} \circ z$  on  $U_z \cap U_1$  are of class  $C^3$ , for  $p \in U_z \cap U_1$ , we can write the expansion (cf. (2.7))

$$z^{i}(p) = z^{i}(p_{0}) + a^{i}_{j}(p_{0})[x^{j}(p) - x^{j}(p_{0})] + a^{i}_{jk}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})].$$
(3.24)

Here:  $p = \eta(s, t), p_0 = \gamma(x^1(p)) = \eta(s, t_0) = \gamma(s), a_j^i \colon \gamma(J_1) \to \mathbb{R}$  are  $C^1$  functions, and  $a_{jk}^i \colon U_z \cap U_1 \to \mathbb{R}$  are of class  $C^1$  and together with their partial derivatives they are bounded. Now we invert the situation: we shall try to find  $a_j^i$  and  $a_{jk}^i$ such that the functions  $z^i$  given by (3.24) generate a normal frame  $\{E_i'\}$ ,

$$\frac{\partial}{\partial z^i}\Big|_{\gamma(s)} = E'_i|_{\gamma(s)} := A^j_i(\gamma(s))\frac{\partial}{\partial x^j}\Big|_{\gamma(s)}$$

with A given by (3.22).<sup>19</sup> So,  $\frac{\partial z^i}{\partial x^j}\Big|_{p_0} = (A^{-1}(p_0))_j^i$  as, for  $p = x^{-1}(s, t)$ , we have  $p_0 = \gamma(s)$ . Inserting here (3.24), we get

$$\begin{aligned} \frac{\partial z^{i}(p_{0})}{\partial x^{1}} &= \frac{\partial z^{i}}{\partial x^{1}}\Big|_{p_{0}} = \frac{\partial z^{i}}{\partial x^{1}}\Big|_{\gamma(s)} = \left(A^{-1}(p_{0})\right)_{1}^{i} = \left(A^{-1}(\gamma(s))\right)_{1}^{i} \\ &\iff z^{i}(p_{0}) = \int_{s_{0}}^{x^{1}(p_{0})} \left(A^{-1}(\gamma(\sigma))\right)_{1}^{i} \mathrm{d}\sigma + a^{i} \\ a^{i}_{j}(p_{0}) &= \frac{\partial z^{i}(p)}{\partial x^{j}}\Big|_{p_{0}} = \frac{\partial z^{i}}{\partial x^{j}}\Big|_{\gamma(s)} = \frac{\partial z^{i}}{\partial t^{j}}\Big|_{t=t_{0}} = \left(A^{-1}(p_{0})\right)_{j}^{i}, \quad \text{ for } j \ge 2 \end{aligned}$$

<sup>&</sup>lt;sup>19</sup>The idea is to restrict a frame normal along  $\gamma$  to  $\gamma(J)$  and then to extend the resulting frame, defined solely on  $\gamma(J)$ , in a neighborhood of  $\gamma(J)$  in a holonomic way. (For a general result regarding such extensions – see Lemma III.10.1 on page 194.)

### 3. The case along paths

where  $a^i \in \mathbb{R}$  are constants representing the coordinates of  $\gamma(s_0)$  in  $\{z^i\}$ . Therefore, by virtue of (3.22) and (3.24), the looked for functions  $z^i$  have the form

$$z^{i}(p) = a^{i} + \int_{s_{0}}^{x^{1}(p)} \left(A^{-1}(\gamma(\sigma))\right)_{1}^{i} d\sigma + \left(A^{-1}(\gamma(x^{1}(p)))\right)_{j}^{i} [x^{j}(p) - x^{j}(p_{0})] + a^{i}_{jk}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})], \quad A(\gamma(s)) = Y(s, s_{0}; -\Gamma_{1} \circ \gamma)B,$$
(3.25)

where  $p_0 = \gamma(x^1(p))$  and  $a_{jk}^i \colon U_1 \to \mathbb{R}$  are  $C^1$  functions which together with their partial derivatives are bounded on  $U_1$ . Notice, in (3.25) the terms with j, k = 1vanish as  $x^1(p) = x^1(p_0) = s$ . Since the Jacobian of the change  $\{x^i\} \to \{z^i\}$  along  $\gamma$  is  $\det\left[\frac{\partial z^i}{\partial x^j}|_{\gamma(s)}\right] = \left(\det\left[\frac{\partial x^i}{\partial z^j}|_{\gamma(s)}\right]\right)^{-1} = \det\left(Y(s, s_0; -\Gamma_1 \circ \gamma)B\right)^{-1} \neq 0, \infty$  for every  $s \in J_1$ , there is a neighborhood  $U_z \subseteq U_1$  of  $\gamma(J), U_z \supset \gamma(J)$ , such that  $(U_z, z)$  is a chart of M with coordinate functions  $z^i$ .

At the moment we can not assert that the chart  $(U_z, z)$  is normal along  $\gamma$ : we proved that  $\frac{\partial}{\partial z^i}|_{\gamma(J_1)} = E'_i|_{\gamma(J_1)}$  where  $\{E'_i\}$  is normal on  $\gamma(J_1)$  from where it generally does not follow that  $\{\frac{\partial}{\partial z^i}\}$  is normal on  $\gamma(J_1)$ , but from here the conclusion can be made that every coordinate system  $\{z^i\}$  normal along  $\gamma$  has the form (3.25). We shall try to choose the functions  $a^i_{jk}: U_z \to \mathbb{R}$  so that to make  $\{z^i\}$  normal coordinates system. For this purpose, the equation (I.5.4') on page 41 with  $U_z$  for  $U \cap V$  must be satisfied:

$$\left(\frac{\partial^2 x^i}{\partial z^j \partial z^k} + \frac{\partial x^r}{\partial z^j} \frac{\partial x^n}{\partial z^k} \Gamma^i{}_{rn}\right)\Big|_{\gamma(s)} = 0 \qquad s \in J_1.$$
(3.26)

Multiplying this equality with  $\frac{\partial z^{j}}{\partial x^{l}} \frac{\partial z^{k}}{\partial x^{m}} \Big|_{\gamma(s)}$ , summing over j and k, and using

$$\frac{\partial^2 x^i}{\partial z^j z^k} \frac{\partial z^j}{\partial x^l} \frac{\partial z^k}{\partial x^m} + \frac{\partial x^i}{\partial z^j} \frac{\partial^2 z^j}{\partial x^l x^m} = 0,$$

which can be obtained from  $\frac{\partial x^i}{\partial z^k} \frac{\partial z^k}{\partial x^m} = \delta^i_m$  by differentiation with respect to  $x^l$ , we, after some simple algebra, find an equivalent to (3.26) equation

$$\frac{\partial z^{i}}{\partial x^{l}}\Big|_{\gamma(s)}\Gamma^{l}{}_{jk}(\gamma(s)) = \frac{\partial^{2} z^{i}}{\partial x^{j} x^{k}}\Big|_{\gamma(s)}, \qquad s \in J_{1}$$
(3.26')

since  $\frac{\partial^2 z^i}{\partial x^j x^k}$  is symmetric in j and k (the manifold is of class  $C^3$ ). From here an immediate observation follows: normal along  $\gamma$  coordinates (may) exist only for symmetric (torsionless) along  $\gamma$  connections. This is in a full agreement with Corollary I.5.3. Correspondingly, below we deal with the symmetric case as until now the symmetry of the connection was not used. From (3.25), the derivatives in (3.26') can easily be calculated:

$$\begin{split} \frac{\partial z^i}{\partial x^l}\Big|_{\gamma(s)} &= \left(A^{-1}(\gamma(s))\right)_l^i,\\ \frac{\partial^2 z^i}{\partial x^1 x^k}\Big|_{\gamma(s)} &= \left(\frac{\mathrm{d}A^{-1}(\gamma(s))}{\mathrm{d}s}\right)_k^i, \quad \frac{\partial^2 z^i}{\partial x^m x^n}\Big|_{\gamma(s)} = 2a^i_{(mn)}(\gamma(s)), \quad m,n \ge 2 \end{split}$$

with A given by (3.22). Since  $\frac{d}{ds}A^{-1} = -A^{-1}\frac{dA}{ds}A^{-1} = A^{-1}\Gamma_1$  on  $\gamma(J)$ , where equation (3.16a) was used, the substitution of the above equalities into (3.26') results in

$$\begin{split} \left(A^{-1}(\gamma(s))\right)_l^i \Gamma^l{}_{1m}(\gamma(s)) &= \left(A^{-1}(\gamma(s))\Gamma_1(\gamma(s))\right)_m^i \\ \left(A^{-1}(\gamma(s))\right)_l^i \Gamma^l{}_{jk}(\gamma(s)) &= 2a^i_{(jk)}(\gamma(s)), \qquad j,k \ge 2. \end{split}$$

As  $(\Gamma_1)_m^n := \Gamma_{m1}^n = \Gamma_{1m}^n$ , the first of these equalities is an identity. This reflects the vanishment of the terms with j, k = 1 in (3.25). The second equation expresses  $a_{(jk)}^i$  along  $\gamma$  through known quantities. For  $p \in U_1 \setminus \gamma(J_1)$ , the values of  $a_{(jk)}^i(p)$ can not be determined from (3.26), so they are left completely arbitrary from the requirement  $\{z^i\}$  to be normal along  $\gamma$ . Since we supposed M to be of class  $C^3$ , there are  $C^0$  bounded functions  $a_{jkl}^i(p): U_z \to \mathbb{R}$  such that

$$a_{jk}^{i}(p) = a_{jk}^{i}(p_{0}) + a_{jkl}^{i}(p)[x^{l}(p) - x^{l}(p_{0})]$$

where  $p = x^{-1}(s, t) \in U_z \subseteq U_1$  and  $p_0 = \gamma(s) = x^{-1}(s, t_0)$ . Inserting this expansion in (3.25) and using the above ones for  $a^i_{(jk)}, j, k \ge 2$ , we finally get:

$$z^{i}(p) = a^{i} + \int_{s_{0}}^{x^{1}(p)} \left(A^{-1}(\gamma(\sigma))\right)_{1}^{i} d\sigma + \left(A^{-1}(\gamma(x^{1}(p)))\right)_{j}^{i} [x^{j}(p) - x^{j}(p_{0})] + \left(A^{-1}(\gamma(x^{1}(p)))\right)_{l}^{i} \Gamma^{l}_{jk}(\gamma(x^{1}(p)))[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})] + a^{i}_{jkl}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})]$$
(3.27)

where

$$a^{i} \in \mathbb{R}, \quad A(\gamma(s)) = Y(s, s_{0}; -\Gamma_{1} \circ \gamma)B, \ s \in J_{1}, p = x^{-1}(s, t) \in U_{1}, \quad p_{0} = x^{-1}(s, t_{0}) = \gamma(s),$$
(3.28)

*B* is constant non-degenerate matrix, and *Y* and *x* are defined via (3.17) and (3.12), respectively. Recall, the coordinates  $\{x^i\}$  have been defined via (3.12).

In this way, after long and tedious computation, we have almost proved the following theorem.

### 3. The case along paths

**Theorem 3.3.** Let  $\gamma: J \to M$  be a regular  $C^1$  path without self-intersections in a  $C^3$ manifold endowed with  $C^0$  symmetric linear connection. For every  $s_0 \in J$ , there exist an interval  $J_1 \subseteq J$  containing  $s_0$  and chart  $(U_z, z)$  such that  $\gamma(J_1) \subset U_z$  and the associated with it local coordinates  $\{z^i\}$  are normal along  $\gamma$ , or, more precisely, on  $\gamma(J_1)$ . All such local coordinates, which are normal along  $\gamma$  in  $U_z$ , are given via (3.27).

Remark 3.13. The theorem remains true if  $\gamma$  is fixed and the connection is symmetric only on  $\gamma(J)$ , i.e., for a fixed path the symmetry on the whole manifold is not necessary, it is required only along  $\gamma$ .

Proof. Since in the derivation of (3.27) the point  $s_0 \in J$  was completely arbitrary, the theorem is proved when J is an open  $\mathbb{R}$ -interval. If J is closed from one or both ends, one should proceed exactly as was pointed before the formulation of Theorem 3.1 (see page 97). Take some regular  $C^1$  path  $\gamma' : J' \to M$  without selfintersections such that  $J' \supset J$  is open,  $\gamma'|_J = \gamma$  and the tangent vectors to  $\gamma'$  and  $\gamma$  coincide on J (at the end point(s)  $\dot{\gamma}$  is defined as one-sided derivative). Since for  $\gamma'$  the assertions of the theorem are valid, the restriction of them to  $\gamma$ , i.e., to  $J \subset J'$ , completes the proof.  $\Box$ 

Remark 3.14. The assumption of absence of self-intersection points is important. If such points exist, at them the theorem fails as the mapping  $\gamma: s \mapsto \gamma(s)$  is at least two-to-one at them and in no way the parameter  $s \in J$  can be taken as one of the local coordinates in their neighborhood. Generally, normal frames can be introduced on any piece of  $\gamma$  without self-intersections, but at these points, if any, the normal coordinates can not be joint smoothly.

Remark 3.15. If we have started with some fixed  $i_0 \neq 1$  for which  $\dot{\gamma}_y^{i_0}(s_0) \neq 0$  instead of our choice  $\dot{\gamma}_y^1(s_0) \neq 0$ , the result will be given again via (3.27) and (3.28) in which the index 1 (not in the power -1 of  $A^{-1}$ !) must be replaced with  $i_0$ . Correspondingly, the terms with  $j, k = i_0$  will vanish.

**Corollary 3.2.** Let  $\gamma: J \to M$  be regular  $C^1$  path without self-intersections in a  $C^3$  manifold endowed with  $C^0$  symmetric linear connection. Let  $\{E_i\}$  be a frame on U, with  $U \cap \gamma(J) \neq \emptyset$ , normal along  $\gamma|_{J_1}$  for some subinterval  $J_1 \subseteq J$ . Then all frames  $\{E'_i = A^i_j E_j\}$  normal along  $\gamma|_{J_1}$  are such that

$$A(p) = B + B_{kl}(p)[x^k(p) - x^k(p_0)][x^l(p) - x^l(p_0)]$$
(3.29)

for a constant matrix B and  $C^1$  matrix-valued functions  $B_{kl}$  which together with their partial derivatives are bounded on U. Besides, the normal holonomic frames  $\{E'_i\}$  are such that  $E'_i = \frac{\partial}{\partial z^i}$ , where the coordinates  $z^i$  are normal along  $\gamma|_{J_1}$  and

$$z^{i}(p) = a^{i} + \delta_{1}^{i}[x^{1}(p) - s_{0}] + [x^{i}(p) - x^{i}(p_{0})] + a^{i}_{jkl}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})].$$
(3.30)

*Proof.* Apply Theorems 3.1 and 3.3 for  $\Gamma^i_{\ jk} \circ \gamma|_{J_1} = 0$  (and hence  $Y(s, s_0; -\Gamma_1) = \mathbb{1}$ ) and  $A \circ \gamma|_{J_1} = \mathbb{1}$  as  $\{x^i\}$  and  $\{E_i = \frac{\partial}{\partial x^i}\}$  are normal on  $\gamma(J_1)$ .

**Exercise 3.2.** Prove Corollary 3.2 by applying only Theorem 3.1. Hint: use (3.23) and follow the scheme outlined in Remark 2.1.

**Example 3.2.** We want to emphasize once again that the coordinates normal along a path are local by their essence. Globally, i.e., on the whole path, they exist only as an exception. Such a rare case is when the path  $\gamma$  is a geodesic lying entirely in a normal neighborhood in which Riemannian coordinates exist and their origin lies on the given geodesic path. In this case, taking the Riemannian coordinates for  $\{y^i\}$ , the equation of  $\gamma$  is (see Section I.6)  $\gamma^i(s) = (y^i \circ \gamma)(s) = X^i(s - s_0)$  with  $X = \dot{\gamma}(s_0) \neq 0$ . Letting  $X^1 \neq 0$ , we get  $\dot{\gamma}^1(s) \neq 0$ ,  $s \in J$  and  $J_1 = J$  which results in  $\gamma(J) \in U_1 = U$ , and, at the end, that  $\{z^i\}$  are defined in a neighborhood of  $\gamma(J)$ , i.e.,  $\{z^i\}$  are globally normal along  $\gamma$ .

**Example 3.3.** Now we can look back on the Fermi coordinates constructed in the previous subsection. At first, choose the initial coordinates  $\{y^i\}$  as pointed in Remark 3.1 on page 87. Then, comparing (3.2) with (3.24) or (3.4) with (3.27), we see that these particular coordinates correspond along a given path to the choice  $\beta_s(t) = \eta(s, \hat{t}) = \exp_{\gamma(s),X}(t - t_0)$ , where one of the components of  $\hat{t}$  is equal to  $t \in J_X$ , the others being constants, i.e., the paths  $\beta_s \colon J_X \to M$  are geodesics. Besides, in the case of Fermi coordinates, by virtue of the assumption that all structures are of class  $C^{\infty}$ , the appearing in (3.27) functions are expanded into power series with respect to  $x^k(p) - x^k(p_0) = t^k - t_0^k = (t - t_0)\delta_{i_0}^k$  for some fixed  $i_0$ . (Note, the matrix A has completely different meanings in (3.27) and in (3.2) or (3.4)!)

## 4. The case in a neighborhood

The flat linear connections are usually associated with the path-independence of the parallel transport assigned to them (see p. 29 and especially Remark I.3.5) rather than with the normal coordinates or frames [8, § 2.6 (iii)], [11, Chapter II, § 9 and Chapter V, § 4], [1, Section 6.10].<sup>1</sup> Below we shall see that these are different sides of one and the same problem.

**Theorem 4.1.** Let M be a  $C^3$  manifold endowed with  $C^1$  linear connection  $\nabla$  and U be  $(\dim M)$ -dimensional submanifold of M. Frames normal for  $\nabla$  on U exist if and only if  $\nabla$  is flat on U,  $R|_U = 0$ .

*Remark* 4.1. Notice, the theorem covers equally well the cases without and with torsion.

Remark 4.2. In particular, the submanifold U can be a neighborhood, possibly coordinate one, or the whole manifold M.

<sup>&</sup>lt;sup>1</sup>For the special case of Riemannian manifolds, see [12, p. 286 and p. 303] where, equivalently, a manifold is called (locally) flat if it is (locally) isometric to  $\mathbb{R}^n$  with a metric  $a, b \mapsto \sum_i \varepsilon_i a^i b^i$  for  $a, b \in \mathbb{R}^n$  and  $\varepsilon_i = \pm 1$ .

### 4. The case in a neighborhood

*Proof.* Let  $\{E_i\}$  be a frame on U. To save some writing, we introduce the matrices  $\Gamma_k$  and  $\mathbf{R}_{kl}$  of, respectively, the coefficients and curvature of  $\nabla$ :<sup>2</sup>

$$\Gamma_k := \left[\Gamma^i_{jk}\right]_{i,j=1}^{\dim M}, \qquad \boldsymbol{R}_{kl} := \left[R^i_{jkl}\right]_{i,j=1}^{\dim M}.$$

In this notation, equation (I.3.13) reads

$$\boldsymbol{R}_{kl} = -2\Gamma_{[k,l]} + 2\Gamma_{[k}\Gamma_{l]} - C_{kl}^m\Gamma_m \tag{4.1}$$

where  $\Gamma_{k,l} := E_l(\Gamma_k)$  and  $C_{kl}^m$  are defined via (I.3.15).

NECESSITY. If  $\nabla$  admits a frame  $\{E'_i\}$  normal on U, there is a non-degenerate  $C^1$  matrix-valued function  $A = [A^j_i]$  such that  $E'_i = A^j_i E_j$  and (see (I.5.4))

$$\Gamma_k = -A_{,k}A^{-1}$$
 on  $U$ ,  $k = 1, \dots, \dim U = \dim M$ . (4.2)

The substitution of this equality into (4.1) results in  $R|_U = 0$ , due to (I.3.15).

SUFFICIENCY. As we know from the considerations in Section I.5 (p. 39), a connection  $\nabla$  admits normal frames iff (I.5.4) has solutions with respect to the matrix A in a given frame  $\{E_i\}$ . Rewriting (I.5.4) in the form

$$E_k(A)|_p = -\Gamma_k(p)A(p), \qquad p \in U, \tag{4.3}$$

we shall find its general solution, under the condition  $R|_U = 0$ , by applying the following lemma.

**Lemma 4.1.** Let N be a manifold and there be given  $C^1$  matrix-valued functions  $Z_a: N \to \operatorname{GL}(m, \mathbb{K}), a = 1, \ldots, \dim N, \operatorname{GL}(m, \mathbb{K})$  being the group of  $m \times m$  nondegenerate matrices on  $\mathbb{K}$  for some  $m \in \mathbb{N}$ . Suppose  $\{e_a | a = 1, \ldots, \dim N\}$  is a (global) frame on N and consider the initial-value problem

$$e_a(Y)|_q = Z_a(q)Y, \qquad q \in N, \ a = 1, \dots, \dim N,$$

$$(4.4a)$$

$$Y|_{q=q_0} = 1$$
 (4.4b)

with respect to the  $m \times m$  matrix-valued function Y on N. Here  $q_0 \in N$  is fixed and  $\mathbb{1} = \mathbb{1}_m$ . Then:

(i) The integrability conditions for (4.4a) are

$$e_a(e_b(Y)) - e_b(e_a(Y)) = c_{ab}^d e_d(Y), \qquad a, b, d = 1, \dots, \dim N$$
 (4.5)

where  $[e_a, e_b]_- =: c_{ab}^d e_d$ , or, equivalently,

$$R_{ab}(Z_1, \dots, Z_{\dim N}) := e_b(Z_a) - e_a(Z_b) + Z_a Z_b - Z_b Z_a + c_{ab}^d Z_d = 0, \quad (4.6)$$

i.e., (4.4a) has solutions with respect to  $Y: N \to \operatorname{GL}(m, \mathbb{K})$  under these conditions.

<sup>&</sup>lt;sup>2</sup>Since the Ricci tensor will not appear in this book, the quantities  $\mathbf{R}_{kl}$  can not be confused with its components.

(ii) The initial-value problem (4.4) has a solution, which is of class C<sup>2</sup>, unique and smoothly depends on its arguments, if and only if the integrability conditions (4.6) are valid. This solution will be denoted by

$$Y = Y(q, q_0; Z_1, \ldots, Z_{\dim N}).$$

Remark 4.3. In this section, Lemma 4.1 will be used only for N = U, U being a neighborhood in M, resp. dim  $N = m = \dim M$ . We formulate it in the above general form because we shall need this generalization further. The proofs in the both cases are practically identical.

Remark 4.4. The choice N = J, dim N = 1,  $q = s \in J$ ,  $e_1 = \frac{d}{ds}$ , and  $q_0 = s_0 \in J$  returns us to Lemma 3.2. Moreover, for any N with dim N = 1 the integrability conditions are identically valid, so the problem (4.4) always has a unique smooth solution in the one-dimensional case.

Remark 4.5. It is easily seen that the general solution of the matrix system of differential equations

$$e_a(X) = Z_a X \tag{4.7}$$

with  $X \in \operatorname{GL}(m, \mathbb{K})$  is

$$X = Y(q, q_0; Z_1, \dots, Z_{\dim N})$$
(4.8)

for arbitrary  $q \in N$ , or, equivalently,

$$X = Y(q, q_0; Z_1, \dots, Z_{\dim N}) X_0$$
(4.9)

for fixed  $q_0 \in N$  and arbitrary  $X_0 \in GL(m, \mathbb{K})$ . Also, the particular solution satisfying the initial condition

$$X|_{q=q_0} = X_0 \tag{4.10}$$

for some  $q_0 \in N$  and  $X_0 \in GL(m, \mathbb{K})$  is given by (4.9).

Proof of Lemma 4.1. Let (V, z) be a chart of N and  $e_a|_V = D_a^b \frac{\partial}{\partial z^b}$  for some  $C^1$  non-degenerate matrix-valued function  $D = [D_a^b]$ . In the coordinates  $\{z_a\}$ , the equation (4.4a) reads  $\frac{\partial Y}{\partial z^a} = (B^{-1})_a^b Z_b Y$ . The integrability conditions for the last system are [34, Chapter VI, § 1]  $\frac{\partial^2 Y}{\partial z^a \partial z^b} - \frac{\partial^2 Y}{\partial z^b \partial z^a} = 0$ . Due to  $e_a(e_b(Y)) - e_b(e_a(Y)) = D_a^{a'} D_b^{b'} \left(\frac{\partial^2 Y}{\partial z^{a'} \partial z^{b'}} - \frac{\partial^2 Y}{\partial z^{b'} \partial z^{a'}}\right) + c_{ab}^d e_d(Y)$ , which is obtainable by direct calculation (cf. equations (I.8.1)–(I.8.3) in Section I.8), these conditions are equivalent to (4.5).

Expressing the derivatives  $e_a(Y)$  from (4.4a) and inserting the results into equation (4.5), we get (4.6) (cf. [34, Chapter VI, equation (1.4)]). Reversing this process, we can derive (4.5) from (4.6). Hence (4.5) and (4.6) are equivalent (provided (4.4a) is valid, as we supposed).

The second part of the lemma's assertion is a corollary of its first part and the corresponding theorems of existence and uniqueness in the theory of differential equations; see, e.g., [34, Chapter VI, Theorem 6.1].  $\Box$ 

#### 4. The case in a neighborhood

Let us return now to equation (4.3). It corresponds to (4.7) with N = U, dim  $N = m = \dim M$ ,  $e_i = E_i$ , and  $Z_k = -\Gamma_k$ . So, the integrability conditions for (4.3) are (4.6) with  $Z_k = -\Gamma_k$ :

$$0 = R_{kl}(-\Gamma_1, \dots, -\Gamma_{\dim M}) = -2\Gamma_{[k,l]} + 2\Gamma_{[k}\Gamma_{l]} - C_{kl}^m\Gamma_m = \mathbf{R}_{kl}$$

where (4.1) was used. These conditions are fulfilled due to the flatness of  $\nabla$  on U,  $R|_U = 0$ . Consequently, according to Lemma 4.1 (ii) and Remark 4.5, the general solution of (4.3) is

$$A(p) = Y(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M})B$$

$$(4.11)$$

where  $p \in U$ ,  $p_0 \in U$  is fixed point, B is non-degenerate constant matrix in  $GL(\dim M, \mathbb{K})$ , and Y is the unique solution of the initial-value problem

$$E_i(Y)|_p = -\Gamma_i(p)Y, \quad Y|_{p=p_0} = \mathbb{1}_{\dim M}.$$
 (4.12)

Thereof, by construction, the frames  $\{E'_i = A^j_i E_j\}$ , with the matrices A given via (4.11), are normal on U.

**Corollary 4.1.** A  $C^3$  manifold with  $C^1$  linear connection admits a (global) normal frame on it iff it is flat.

*Proof.* See Theorem 4.1.

**Theorem 4.2.** Given a  $C^1$  linear connection on  $C^3$  manifold M. If the connection is flat on a (dim M)-dimensional submanifold U of M, then all frames normal on U for it are  $\{E'_i = A^j_i E_j\}$  where  $\{E_i\}$  is arbitrarily chosen frame on U and A is given by (4.11) in which  $p \in U$ ,  $p_0 \in U$  is fixed,  $\Gamma_k$  are the connection's matrices in  $\{E_i\}$ , and B is constant non-degenerate matrix.

*Proof.* See the proof of the sufficiency of Theorem 4.1 in which is proved that (4.11) is the general solution of (I.5.4) in U.

*Remark* 4.6. The constancy of B in (4.11) agrees with Proposition I.5.2 and Corollary I.5.1.

**Corollary 4.2.** Let M be  $C^3$  manifold endowed with a symmetric flat linear connection. On every  $(\dim M)$ -dimensional submanifold of M, in particular on an open set in M or on the whole M, exist normal frames whose general form is  $\{E'_i = A^j_i E_j\}$  with  $A = [A^j_i]$  given by (4.11) and  $\{E_i\}$  being an arbitrary frame on it.

Proof. See Theorem 4.2

Theorems 4.1 and 4.2 give a *complete* description of the frames normal on submanifolds of maximum dimensionality, in particular on neighborhoods or on the whole manifold. On their base an analogous description for the normal coordinates can be given.

Let M be  $C^3$  manifold endowed with  $C^1$  symmetric (torsionless) linear connection  $\nabla$  which is flat in the domain U of some local chart (U, x) of M,  $R|_U = 0$ . The problem is to be described all local coordinates normal on U for  $\nabla$ . Its general solution is given by the following result.

**Theorem 4.3.** Given a  $C^3$  manifold M endowed with  $C^1$  linear connection. If (U, x) is a chart of M on whose domain the connection is flat and torsionless,  $R|_U = 0$  and  $T|_U = 0$ , then on U exist coordinates  $\{x'^i\}$  normal for the connection given. All such coordinates are obtained from  $\{x^i\}$  according to equation (4.14) below.

*Proof.* According to Proposition I.5.3 and Theorem 4.1, coordinates  $\{x'^i\}$  normal on U exist and they locally generate all frames normal on U. By Corollary I.5.2 and Theorem 4.2, all coordinates normal on U are such that  $\frac{\partial}{\partial x'^i} = A_i^j E_j$  with  $A = [A_i^j]$  given by (4.11) and  $\{E_i\}$  being arbitrary frame on U. Below, for simplicity, we choose  $E_i = \frac{\partial}{\partial x^i}$ . So  $\{x'^i\}$  must be such that  $\frac{\partial}{\partial x'^i} = A_i^j \frac{\partial}{\partial x^j}$  with A given by (4.11). Consequently they must be solutions of the following system of partial differential equations

$$\frac{\partial x'^{i}}{\partial x^{j}} = \left(A^{-1}\right)^{i}_{j}, \qquad A(p) = Y(p, p_{0}; -\Gamma_{1}, \dots, -\Gamma_{\dim M})B, \qquad (4.13)$$

where  $p \in U$ ,  $p_0 \in U$  is fixed,  $\Gamma_k$  are the connection coefficients' matrices in  $\{\frac{\partial}{\partial x^i}\}$ , and B is constant non-degenerate matrix. The integrability conditions for this system are [34, Chapter VI, §§ 1–6]

$$0 = \frac{\partial^2 x'^i}{\partial x^{[j} \partial x^{k]}} = \frac{\partial}{\partial x^{[j}} \left(A^{-1}\right)^i_{k]} = \left(\frac{\partial}{\partial x^{[j}} A^{-1}\right)^i_{k]} = \left(A^{-1} \frac{\partial A}{\partial x^{[j}} A^{-1}\right)^i_{k]}$$
$$= \left(A^{-1} \Gamma_{[j)}\right)^i_{k]} = \left(A^{-1}\right)^i_l \Gamma^l_{[kj]} ,$$

where (4.13) and (4.3) were used, which equations are identically satisfied, due to the symmetry of the connection. Hence the system (4.13) is completely integrable and defines  $x'^{i}$  as  $C^{3}$  functions of  $x^{i}$  (as A is of class  $C^{2}$ ).<sup>3</sup> The general solution of (4.13) can be written as

$$x'^{i}(p) = a^{i} + \int_{p_{0}}^{p} \left( B^{-1}Y^{-1}(q, p_{0}; -\Gamma_{1}, \dots, -\Gamma_{\dim M}) \right)_{k}^{i} \mathrm{d}q^{k}$$
(4.14)

or

$$x'^{i}(p) = a^{i} + \int_{s_{0}}^{s} \left( B^{-1} Y^{-1}(\gamma(t), p_{0}; -\Gamma_{1}, \dots, -\Gamma_{\dim M}) \right)_{k}^{i} \dot{\gamma}^{k}(t) \, \mathrm{d}t \qquad (4.14')$$

<sup>&</sup>lt;sup>3</sup>The same result follows from the Poincaré Lemma [13, p. 121], [2, p. 55]: writing (4.13) as  $dx'^{i} = (A^{-1})^{i}_{j} dx^{j}$ , a necessary and sufficient condition for the existence of  $x'^{i}$  is  $0 = d[(A^{-1})^{i}_{j} dx^{j}] = 2(A^{-1})^{i}_{[j,k]} dx^{j} \wedge dx^{k}$ . For details concerning Pfaff systems, see, e.g., [12, Chapter IV, Section C].

Here:  $p \in U$ ,  $p_0 \in U$  is fixed,  $a^i \in \mathbb{K}$  are constants representing the coordinates of  $p_0$  in  $x'^i$ ,  $\Gamma_k$  are the matrices of the connection coefficients in  $\{\frac{\partial}{\partial x^i}\}$ , the integrals are taken along paths lying entirely in U,  $\gamma: J \to U$  is a  $C^1$  path such that  $\gamma(s_0) = p_0$  and  $\gamma(s) = p$  for some  $s_0, s \in J$ , B is constant non-degenerate matrix, and Y is the solution of (4.4) with N = U,  $m = \dim M$ , and  $e_k = \frac{\partial}{\partial x^k}$ . i.e.,

$$\frac{\partial Y}{\partial x^k} = -\Gamma_k Y, \qquad Y|_{p=p_0} = \mathbb{1}, \quad Y = Y(p.p_0; -\Gamma_1, \dots, -\Gamma_{\dim M}).$$
(4.15)

Since the connection is supposed flat in U,  $R|_U = 0$ , it is a simple exercise to be shown that the integrals in (4.14) and (4.14') are well-defined, i.e., independent of the concrete paths, like  $\gamma$ , lying in U along which the integration is performed. This ends the proof of Theorem 4.3.

**Exercise 4.1.** Show that the choice  $E_i = \frac{\partial}{\partial x^i}$  in the proof of Theorem 4.3 does not restrict the generality of our constructions. For the purpose, prove that the choice  $E_i = D_i^j \frac{\partial}{\partial x^j}$ , with a  $C^1$  non-degenerate matrix-valued function  $D = [D_i^j]$ , does not influence the final results expressed by Theorem 4.3. (Hint: The matrix (4.11), which is a solution of (4.3), transforms  $\{E_i\}$  into  $\{E_i\}$  and  $\Gamma_k$ , appearing in (4.3) and (4.11), are the connection's matrices in  $\{E_i\}$ ; so, the particular form (and meaning) of A and  $\Gamma_k$  is different for arbitrary  $E_i$  and for  $E_i = \frac{\partial}{\partial x^i}$ .)

**Corollary 4.3.** Let M be a  $C^3$  manifold and  $\nabla$  be a  $C^1$  flat torsionless linear connection on it. On every coordinate neighborhood of M exist coordinates normal for  $\nabla$  whose general form is given via (4.14).

*Proof.* See Theorem 4.3.

Theorem 4.3 and its Corollary 4.3 give a *complete* description of the normal coordinates on a  $C^3$  manifold with  $C^1$  (flat torsionless) linear connection.

If the manifold and the connection considered are of class  $C^{\infty}$ , the integral in (4.14) (or (4.14')) can be calculated explicitly in terms of the connection coefficients (in  $\{\frac{\partial}{\partial x^i}\}$ ) as a power series in some neighborhood of a fixed point  $p_0 \in U$ . Indeed, in this case we can write

$$Y^{-1}(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M}) = Y^{-1}(p_0, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M}) \\ + \sum_{n=1}^{\infty} \frac{1}{n!} \sum_{i_1, \dots, i_n=1}^{\dim M} \frac{\partial^n Y^{-1}(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M})}{\partial p^{i_1} \cdots \partial p^{i_n}} \Big|_{p=p_0} \\ \times [p^{i_1} - p_0^{i_1}] \cdots [p^{i_n} - p_0^{i_n}]$$

where  $p^k := x^k(p), p_0^k := x^k(p_0)$ , and the series is convergent in some neighborhood  $V \subseteq U$  of  $p_0$ .

The values of  $Y^{-1}$  and its derivatives in the right-hand side of the last equation can be evaluated by means of (4.15) as it is equivalent to

$$\frac{\partial Y^{-1}}{\partial x^k} = Y^{-1} \Gamma_k \tag{4.16a}$$

$$Y^{-1}|_{p=p_0} = \mathbb{1}, \qquad Y^{-1} = \left(Y(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M})\right)^{-1}.$$
 (4.16b)

In fact, by successive differentiation of equation (4.16a), we get  $\frac{\partial^n Y^{-1}}{\partial x^{(i_1} \dots \partial x^{i_n})} = Y^{-1}G_{i_1,\dots,i_n}$  where the matrices  $G_{i_1,\dots,i_n}$  are symmetric in the subscripts and are defined through the relation

$$G_{i_1,\dots,i_{n+1}} := \Gamma_{(i_1}G_{i_2,\dots,i_{n+1})} + \frac{\partial}{\partial x^{(i_1}}G_{i_2,\dots,i_{n+1})}, \qquad n \in \mathbb{N}, \quad G_k := \Gamma_k.$$
(4.17)

Hence the expansion of  $Y^{-1}$  takes the form

$$Y^{-1}(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M}) = \mathbb{1} + \Gamma_k(p_0)[x^k(p) - x^k(p_0)] + \sum_{n=1}^{\infty} \frac{1}{n!} \sum_{i_1, \dots, i_n=1}^{\dim M} G_{i_1, \dots, i_n}(p_0)[x^{i_1}(p) - x_0^{i_1}(p_0)] \cdots [x^{i_n}(p) - x_0^{i_n}(p_0)].$$

Substituting this into (4.14) and taking into account that

$$\int_{p_0}^{p} [q^{i_1} - p_0^{i_1}] \cdots [q^{i_n} - p_0^{i_n}] \, \mathrm{d}q^j = \left(1 + \sum_{a=1}^{n} \delta^{i_a j}\right)^{-1} [p^{i_1} - p_0^{i_1}] \cdots [p^{i_n} - p_0^{i_n}] [p^j - p_0^j],$$

we get

$$x'^{i} = a^{i} + (B^{-1})^{i}_{j} \left\{ x^{j}(p) - x^{j}(p_{0}) + \sum_{k,l=1}^{\dim M} \frac{1}{1 + \delta^{kl}} \Gamma^{j}_{kl} \right. \\ \times \left[ x^{k}(p) - x^{k}(p_{0}) \right] \left[ x^{l}(p) - x^{l}(p_{0}) \right] \\ + \sum_{n=1}^{\infty} \frac{1}{n!} \sum_{k,i_{1},\dots,i_{n}=1}^{\dim M} \frac{1}{1 + \sum_{a=1}^{n} \delta^{i_{a}k}} \left( G_{i_{1},\dots,i_{n}}(p_{0}) \right)^{j}_{k} \\ \times \left[ x^{i_{1}}(p) - x^{i_{1}}(p_{0}) \right] \cdots \left[ x^{i_{n}}(p) - x^{i_{n}}(p_{0}) \right] \left[ x^{k}(p) - x^{k}(p_{0}) \right] \right\}.$$
(4.18)

Since  $\frac{\partial x'^i}{\partial x^j}\Big|_{p_0} = (B^{-1})^i_j$  and  $B = [B^j_i]$  is non-degenerate, this series is convergent in some neighborhood  $V \subseteq U$  of  $p_0 \in U$  and, consequently, the transition  $\{x^i\} \to \{x'^i\}$  is  $(C^{\infty})$  well-defined and  $(C^{\infty})$  invertible.

### 5. The case on arbitrary submanifolds

Now we want to say a few words on the links between the normal frames/coordinates and parallel transport on manifolds with flat linear connection. (For simplicity and brevity, we consider only the global case below.)

It is a classical result that the parallel transport is (locally) path-independent iff the manifold M (or the connection) is (locally) flat [3,8,11]. Sometimes for such manifolds is said that they possess teleparallelism or absolute parallelism [19, p. 142], admit distant parallelism [55], or that they are parallelizable [28, p. 68]. A parallelization of M is called the choice of a covariantly-constant frame  $\{E_i\}$ on M, i.e.,  $\nabla_X E_i = 0$  for all  $X \in \mathfrak{X}(M)$ . Hence  $\{E_i\}$  is a parallelization of M iff  $\nabla_{E_j} E_i = 0$  which is equivalent to  $\Gamma^k_{ij} = 0$  (see (I.3.1)), i.e., iff  $\{E_i\}$  is normal. Thereof, the parallelization is simply a concrete choice of some normal frame and the manifold is parallelizable iff it admits normal frames or, equivalently (see Theorem 4.1 and Corollary 4.1), iff it is flat. The following result, which agrees with Proposition I.5.5, expresses the same in another form.

**Proposition 4.1.** A  $C^3$  manifold endowed with  $C^1$  linear connection admits (global) normal frames if and only if the parallel transport associated to the connection is path-independent.

Proof. If M admits a normal frame  $\{E_i\}$ , in it the parallel transport is pathindependent according to Proposition I.5.5. Conversely, let the parallel transport be path-independent and  $\{E_i^0\}$  be a fixed basis at an arbitrarily chosen point  $p \in M$ . Define a frame  $\{E_i\}$  on M such that  $E_i^0|_p \to E_i|_q$  is the result of the parallel transport of  $E_i^0$  from p to q for all  $q \in M$ . By construction, this frame is path-independent and satisfies the equation (see (I.3.18))  $\nabla_{\dot{\gamma}} E_i = 0$  for every path  $\gamma$  joining p and q. So, choosing  $\gamma$  such that at q the tangent vector  $\dot{\gamma}$  coincides with  $E_j|_q$ , we get  $0 = (\nabla_{E_j} E_i)|_q = (\Gamma_{ij}^k E_k)|_q$ , i.e.,  $\Gamma_{ij}^k(q) = 0, q \in M$  which means that  $\{E_i\}$  is normal on M.

*Remark* 4.7. A trivial consequence of Corollary 4.1 and Proposition 4.1 is that the parallel transport is path-independent iff the generating it connection is flat. Conversely, if this result is proved by another way [3,11,28], from Proposition 4.1 immediately follows Corollary 4.1.

### 5. The case on arbitrary submanifolds

The problem for existence of coordinates normal on a submanifold of dimension higher than one was first posed by J. A. Schouten and D. J. Struik in 1935 [43, p. 106]. They showed that if N is n-dimensional submanifold of m-dimensional,  $m \ge n$ , manifold M endowed with symmetric linear connection, then on N exist (in our terminology) normal coordinates provided on N exist n linearly independent covariantly constant (with respect to every vector field) vector fields. In our notation this is translated as: on N exist normal coordinates if on it the connection admits normal frames. The same result is quoted in [19, p. 169] where a new problem is put (in our terminology): if on N, considered as a manifold, a (global) normal frame exists, are there coordinates on M which are normal on N considered as a submanifold of M? This problem was completely solved (for symmetric linear connections) by L. O'Raifeartaigh in 1958 [55].

### 5.1. Conventional method

In this subsection, we shall review (in our notation) and partially generalize the results of the paper [55]. Its idea [55, pp. 18 and 21, paragraphs 1 and 3] is quite simple and analogous to the one of the construction of Fermi coordinates (Subsection 3.1). If N is an n-dimensional submanifold of the m-dimensional,  $m \ge n$ , manifold M, in a neighborhood of N the manifold M can be represented (locally) as a direct sum of suitable (m - n)-dimensional submanifolds  $L_q$ , one for each  $q \in N$ , such that  $L_q$  is (m - n)-dimensional normal neighborhood of q (see Definition I.3.6) and for every p in the mentioned neighborhood of N there is unique  $p_0 \in N$  such that  $L_{p_0} \ni p$ . If on N exist normal coordinates, such are the coordinates  $(\varphi^1, \ldots, \varphi^n, \xi^1, \ldots, \xi^{m-n})$  where  $\varphi^1(p), \ldots, \varphi^n(p)$  are the Riemannian normal coordinates of p with respect to  $p_0$  in  $L_{p_0}$ . Special cases of this construction, corresponding to n = 0, 1, m, were investigated in Sections 2, 3, and 4, respectively. As we shall see, for  $n \ge 2$  (if  $m \ge 2$ ) coordinates normal on N exist iff some conditions, derived below, are satisfied.

Now the rigorous statement and solution of the problem outlined are in order.

Let M be  $C^3$  manifold provided with  $C^1$  torsionless linear connection and N be submanifold of M (see Subsection I.2.1, p. 7). The problem, we are going to investigate, is to find conditions under which on M exist local coordinates normal on (an open subset of) N and, when they are valid, to construct a particular example of such coordinates.

**Lemma 5.1.** Let N be a submanifold of a  $C^3$  manifold M endowed with  $C^1$  linear connection  $\nabla$ . The parallel transport with respect to  $\nabla$  along paths lying entirely in N is path-independent if and only if

$$(R(X,Y))|_q = 0, \qquad X_q, Y_q \in T_q(N), \quad q \in N$$
 (5.1)

where R is the curvature tensor of  $\nabla$  in M and X and Y are vector fields tangent to N (as a manifold),  $X, Y \in \mathfrak{X}(N)$ .

Remark 5.1. This lemma is true for arbitrary linear connections, without or with torsion,<sup>1</sup> and for N = M it reduces to the wide-known result that the parallel transport is path-independent only for flat linear connections. Note, from (5.1) one can not conclude  $R|_N = 0$  unless dim  $N = \dim M$ . (In this context, see Remark IV.10.2 on page 276 concerning the same problem on vector bundles.)

<sup>&</sup>lt;sup>1</sup>In [55] it is proved in the torsionless case.

Remark 5.2. The equation (5.1) does not concern separately M, N, or  $\nabla$ . It involves simultaneously all of them: the manifold M is a carrier of  $\nabla$  and a set containing N and N is the set on which the parallel transport and the curvature of  $\nabla$  are restricted.

*Proof.* Let (U, x) be a chart of M with  $\overline{U} := U \cap N \neq \emptyset$ . By definition of a submanifold (page 7), the pair  $(\overline{U}, \overline{x})$  with  $\overline{x}(q) := (x^1(q), \dots, x^{\dim N}(q)), q \in N$ , is a chart of N with  $\{\overline{x}^a := x^a|_{\overline{U}}|_a = 1, \dots, \dim N\}$  as associated coordinate system. So  $\{\frac{\partial}{\partial \overline{x}^a} = \frac{\partial}{\partial x^a}|_{\overline{U}}\}$  is a frame on  $\overline{U}$ , i.e.,  $\{\frac{\partial}{\partial x^a}|_q\}$  is a basis in  $T_q(N)$ ,  $q \in N$ .

Let  $\gamma: J \to N$  and  $Z \in \mathfrak{X}^1(N) \subseteq \mathfrak{X}^1(M)$  be  $C^1$  vector field tangent to M over N and parallel along  $\gamma$ , viz.  $\dot{\gamma}(s) \in T_{\gamma(s)}(N)$  and  $Z_{\gamma(s)} \in T_{\gamma(s)}(M)$  for  $s \in J$  and

$$0 = \nabla_{\dot{\gamma}} Z = \dot{\gamma}^j \Big( \frac{\partial Z^k}{\partial x^j} + \Gamma^k_{\ lj} Z^l \Big) \frac{\partial}{\partial x^k}$$

Here and further in this proof, all components are with respect to  $\left\{\frac{\partial}{\partial x^k}\right\}$  or  $\left\{\frac{\partial}{\partial \bar{x}^k}\right\}$ ; in particular  $\Gamma^k_{\ lj}$  are the coefficients of  $\nabla$  in the former frame. Since  $\gamma$  lies entirely in N, we have  $\dot{\gamma}(s) \in T_{\gamma(s)}(N)$  and hence  $\dot{\gamma}^j = 0$  for  $j > \dim N$ . The parallel transport in N is path-independent if<sup>2</sup> there is a vector field  $Z \in \mathfrak{X}(N) \subseteq \mathfrak{X}(M)$ such that  $\nabla_{\dot{\gamma}} Z = 0$  for every path  $\gamma: J \to N$ , which is equivalent to the existence of a solution of the system of differential equations

$$\frac{\partial Z^k}{\partial x^a} = -\Gamma^k{}_{la}Z^l, \qquad a = 1, \dots, \dim N, \quad k, l = 1, \dots, \dim M$$

with respect to  $Z^a$  (as  $\dot{\gamma}^j = 0$  for  $j > \dim N$ ). The integrability conditions for this system are [34, Chapter VI, § 1]

$$0 = \frac{\partial^2 Z}{\partial x^{[a} \partial x^{b]}} = \frac{\partial}{\partial x^{[a}} \left( -\Gamma^k_{\ |l|b]} Z^l \right) = \left( -Z^l \frac{\partial}{\partial x^a} \Gamma^k_{\ lb} - \Gamma^k_{\ jb} \frac{\partial}{\partial x^a} Z^j \right)_{[ab]}$$
$$= \left( -\frac{\partial}{\partial x^a} \Gamma^k_{\ lb} + \Gamma^k_{\ jb} \Gamma^j_{\ la} \right)_{[ab]} Z^l = -R^k_{\ lab} Z^l$$

where (I.3.13) was used (with  $C_{jk}^i = 0$  as  $\left\{\frac{\partial}{\partial x^i}\right\}$  is a coordinate frame). By virtue of the arbitrariness of (the initial value of) Z, these conditions are equivalent to

$$R^{k}_{\ \ lab}|_{N} = 0, \qquad k, l = 1, \dots, \dim M, \quad a, b = 1, \dots, \dim N$$
 (5.1')

in every chart (U, x) of M with  $U \cap N \neq \emptyset$ . If X and Y are vector fields tangent to N, then  $X_q, Y_q \in T_q(N) \subseteq T_q(M), q \in N$ , so that  $X^i = Y^i = 0$  for  $i > \dim N$ and consequently (5.1') is equivalent to (5.1).  $\Box$ 

<sup>&</sup>lt;sup>2</sup>However, see Remark I.3.5 on page 29.

*Remark* 5.3. Using the notation of (4.6), we can rewrite (5.1') in more compact matrix form

$$R_{ab}(-\Gamma_1, \dots, -\Gamma_{\dim N})|_p = 0, \qquad a, b = 1, \dots, \dim N,$$
 (5.1")

which is valid in every chart of N.

**Proposition 5.1.** Let N be a submanifold of a  $C^3$  manifold M provided with  $C^1$  linear connection  $\nabla$ . If the connection  $\nabla$  admits frames normal on N, then (5.1) holds, i.e., equation (5.1) is a necessary condition for the existence of frames normal on N.

*Proof.* If  $\nabla$  admits a normal frame on N, by Proposition I.5.5 the parallel transport in N along paths in N is path-independent and, hence, according to Lemma 5.1, equation (5.1) holds.

We will prove in the next subsection (p. 120) that the condition (5.1) is also sufficient for the existence to frames normal on N for arbitrary, torsionless or not, linear connections. Below, following [55], we will prove only the existence of particular coordinates which are normal on N in the torsionless case provided (5.1)is valid.

**Proposition 5.2.** Given a submanifold N of a  $C^3$  manifold M endowed with  $C^1$  torsionless linear connection  $\nabla$  for which equation (5.1) holds. Then  $\nabla$  admits frames and coordinates locally normal on N, i.e., for every  $q \in N$  there exists a chart  $(U^0, y)$  of M with q in its domain,  $U^0 \ni q$ , such that the associated with it coordinate system  $\{y^i\}$  is normal on  $\overline{U} := U^0 \cap N$ . A particular example of such coordinate is provided by equation (5.5) below.

Proof. Let  $\nabla$  be torsion free, (U, x) be a chart of M such that  $\overline{U} := U \cap N \neq \emptyset$ and  $q_0 \in \overline{U}$ . So  $(\overline{U}, \overline{x})$  with  $\overline{x}^a = x^a|_N$ ,  $a = 1, \ldots, \dim N$ , is a chart of N in a neighborhood of  $q_0$ . Let  $\{E_i^0\}$  be a basis in  $T_{q_0}(M)$  such that  $E_a^0 \in T_{q_0}(N)$ ,  $a = 1, \ldots, \dim N$ . Consequently, the  $(\dim M - \dim N)$ -dimensional space spanned by  $\{E_i^0|i > \dim N\}$  does not contained vectors tangent to N except the zero vector.<sup>3</sup>

Suppose the equation (5.1) holds. Since the parallel transport in N along paths lying in N is path-independent (Lemma 5.1), the frame  $\{E_i\}$  on N obtained from  $\{E_i^0\}$  by parallel translation from  $q_0$  to every point  $q \in N$  along paths lying entirely in N is uniquely defined:  $\nabla_X E_i = 0$ , with X being tangent to N vector field, and  $E_i|_{q_0} = E_i^0$ .

By means of the frame  $\{E_i\}$ , we shall define the looked for coordinates. For the purpose, we write the expansion  $E_i = D_i^j \frac{\partial}{\partial x^j}$ . The non-degenerate matrixvalued function  $D = [D_j^i]$  can be obtained from the parallel character of  $\{E_i\}$ on N. Since  $\nabla_X E_i = 0$  for all vector fields X tangent to N, for  $X = \frac{\partial}{\partial x^a}$ ,

<sup>&</sup>lt;sup>3</sup>For example, we can put  $E_i^0 = \frac{\partial}{\partial x^i} \Big|_{q_0}$ ; see the definition of a submanifold on page 7.

#### 5. The case on arbitrary submanifolds

 $a = 1, \ldots, \dim N$ , we get

$$0 = \nabla_{\partial/\partial x^a} E_j = \nabla_{\partial/\partial x^a} \left( D_j^n \frac{\partial}{\partial x^n} \right) = \left( \frac{\partial}{\partial x^a} D_j^n + \Gamma_{ka}^n D_j^k \right) \frac{\partial}{\partial x^n} \\ \iff \frac{\partial D}{\partial x^a} = -\Gamma_a D.$$

As the integrability conditions for the last equation are (5.1''), which are equivalent to (5.1), in turn supposed to be valid, by Lemma 4.1 with  $m = \dim M$ , we have (see (4.4))

$$D(q) = Y(q, q_0; -\Gamma_1, \dots, -\Gamma_{\dim N})D_0$$
(5.2)

where the constant non-degenerate dim  $M \times \dim M$  matrix  $D_0$  is such that  $E_i|_{q_0} = E_i^0 = D_i^{0j} \frac{\partial}{\partial x^j}|_{q_0}$ , i.e., it represents the expansion of  $\{E_i^0\}$  over  $\{\frac{\partial}{\partial x^i}\Big|_{q_0}\}$ .<sup>4</sup>

Now we shall define new coordinates  $\{y^i\}$  on the domain of the chart (U, x). At first we define them only on  $U := U \cap N$  (cf. (3.25)) by

$$y^{i}(q) := a^{i} + \int_{q_{0}}^{q} \left( D^{-1}(p) \right)_{j}^{i} \mathrm{d}p^{j}, \qquad q \in \bar{U} := U \cap N \subseteq N$$
(5.3)

where  $p^j := x^j(p)$  and the integration is performed along some path connecting  $q_0$ and q and lying entirely in N. The integral in this definition is path-independent as the corresponding integrability conditions are identically satisfied on U:

$$\frac{\partial^2 y^i}{\partial \bar{x}^{[a} \bar{x}^{b]}} = \frac{\partial}{x^{[a}} \left( D^{-1} \right)^i_{b]} = \left( \frac{\partial D^{-1}}{\partial x^{[a}} \right)^i_{b]} = \left( -D^{-1} \frac{\partial D}{\partial x^{[a}} D^{-1} \right)^i_{b]} = \left( D^{-1} \Gamma_{[a} \right)^i_{b]} = \left( D^{-1} \right)^i_k \Gamma^k_{[ba]} \equiv 0$$

where  $\frac{\partial}{\partial \bar{x}^a}|_{\bar{U}} = \frac{\partial}{\partial x^a}|_{\bar{U}}$ ,  $\frac{\partial D}{\partial x^a} = -\Gamma_a D$  (see above), and the symmetry of the connection coefficients were employed. The functions  $y^1, \ldots y^{\dim N}$  define a coordinate system in a neighborhood of  $q_0$  in  $\overline{U}$ . Indeed, the Jacobian of the change  $\{\bar{x}^a\} \to \{\bar{y}^a = y^a|_{\bar{U}}\}$  on  $\bar{U}$  is  $J(q) = \det[(D^{-1}(q))_b^a]$ . Since  $D^{-1}(q_0) = D_0^{-1}$  and  $E_a^0 \in T_{q_0}(N)$ , we have  $J(q_0) = \det\left[\left(D_0^{-1}\right)_b^a\right] \neq 0, \infty$  as D is continuous (and also of class  $C^1$ ). Hence there is a neighborhood  $\overline{U}_{q_0} \subseteq \overline{U}$  of  $q_0$  such that  $J(q) \neq 0$  for  $q \in \overline{U}_{q_0}$ .<sup>5</sup> Therefore the functions  $y^i$  given via (5.3) define a coordinate system  $\{\bar{y}^a\}$  on  $\bar{U}_{q_0}$ .

<sup>&</sup>lt;sup>4</sup>The choice  $E_i^0 = \frac{\partial}{\partial x^i}\Big|_{q_0}$  results in  $D_0 = 1$ ; see Footnote3 on the preceding page. <sup>5</sup>The choice  $E_i^0 = \frac{\partial}{\partial x^i}\Big|_{q_0}$  results in  $D_0 = 1$  and  $J(q_0) = 1$ . So, from the continuity of  $J: \overline{U} \to \mathbb{K}$  follows that for every  $\varepsilon \in \mathbb{R}$ ,  $0 < \varepsilon < 1$ , there is a neighborhood  $U_{\varepsilon}$  of  $q_0$  such that  $0 < 1 - \varepsilon < |J(q)| < 1 + \varepsilon$  for  $q \in U_{\varepsilon}$ . Thus  $J(q) \neq 0$  for  $q \in U_{\varepsilon}$ .

From (5.3), we derive:

$$\begin{split} \frac{\partial}{\partial \bar{y}^a}\Big|_q &= \frac{\partial}{\partial y^a}\Big|_q = \frac{\partial x^i}{\partial y^a}\Big|_q \frac{\partial}{\partial x^i}\Big|_q = \left(\left[\frac{\partial x^k}{\partial y^j}\Big|_q\right]\right)_a^i \frac{\partial}{\partial x^i}\Big|_q \\ &= \left(\left[\frac{\partial y^k}{\partial y^j}\Big|_q\right]^{-1}\right)_a^i \frac{\partial}{\partial x^i}\Big|_q = \left((D(q)^{-1})^{-1}\right)_a^i \frac{\partial}{\partial x^i}\Big|_q = D_a^i(q)\frac{\partial}{\partial x^i}\Big|_q = E_a|_q \end{split}$$

for  $q \in \overline{U}_{q_0}$ . So,  $\{E_1, \ldots, E_{\dim M}\}$  is a frame holonomic on  $\overline{U}_{q_0}$  and it is generated by the coordinate system  $\{\overline{y}^a\}$  of the chart  $(\overline{U}_{q_0}, y)$  of N.

At this point we shall need the following lemma.

**Lemma 5.2.** Let N be submanifold of a  $C^3$  manifold M endowed with  $C^1$  torsionless linear connection whose curvature tensor satisfies (5.1). Every parallel frame  $\{E_i\}$ defined on N (resp. in a neighborhood of N) can locally be expanded (resp. be redefined) outside N in a holonomic way, i.e., if the vector fields  $E_i$  are parallel on N, for every point  $p_0 \in N$ , there is a chart (V,z) of M with  $V \ni p_0$  such that  $\overline{V} := V \cap N$  is a coordinate neighborhood in N and the associated with it coordinate system  $\{z^i\}$  generates  $\{E_i\}$  on  $\overline{V}: E_i|_q = \frac{\partial}{\partial z^i}|_q$ ,  $q \in \overline{V}$ . If dim N = 0or if N is real and dim N = 1, the assertion is valid for any frame, parallel or not, regardless of the existence of some connection on M, i.e., in the zero- and real one-dimensional cases any frame defined solely at a single point or along a given path can be (locally) generated by some local coordinates.

Remark 5.4. From this lemma, generally, does not follow that any (parallel) frame is holonomic on N! This is valid if dim  $N = \dim M$ . The meaning of the lemma is that any field of bases parallel on N, with respect to path-independent parallel transport generated by a linear connection, can be generated (locally) by some local coordinate system.

Remark 5.5. For dim N = 1, when (5.1) is identically valid, and  $\mathbb{K} = \mathbb{R}$  this lemma reduces to [76, Lemma 4.1]. See also Lemma III.10.1 on page 194.

Remark 5.6. The lemma means that any continuous frame defined solely on a submanifold and parallel on it can locally be extended to a holonomic frame on its neighborhood provided equation (5.1) holds. Of course, if required, such an extension can be done in an anholonomic way too (provided dim  $N < \dim M$ ). Consequently, the holonomicity problem for a (parallel) frame defined *only* on N (under the condition (5.1)) depends on the way this frame is extended outside the set N.

Proof of Lemma 5.2. Let  $p_0 \in N$ , (U, x) be a chart of M with  $U \ni p_0$ , and  $(\bar{U}, \bar{x})$  be the associated chart of N, viz.  $\bar{U} := U \cap N$ ,  $\bar{x}(q) := (x^1(q), \ldots, x^{\dim N}(q))$  for  $x(q) := (x^1(q), \ldots, x^{\dim N}(q), b^1, \ldots, b^{\dim M - \dim N})$  where  $q \in \bar{U}$  and the b's are constants in  $\mathbb{K}$ . Let  $U_N := \{p | p \in U, x^a(p) = x^a(q), a = 1, \ldots, \dim N, q \in \bar{U}\} \subseteq U$ . This set is a neighborhood of  $\bar{U}$  such that every  $p \in U_N$  has a 'projection  $p_0 \in \bar{U}$  along  $x': p_0 := \bar{x}^{-1}(x^1(p), \ldots, x^{\dim N}(p))$ . Let the expansion  $E_i|_q = D_i^j(q)\frac{\partial}{\partial x^j}|_q$  in

#### 5. The case on arbitrary submanifolds

 $\overline{U}$  be valid for some non-degenerate matrix-valued function  $D := [D_i^j]$ . We define  $z^i \colon U_N \to \mathbb{K}$  by (cf. (5.3))

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \left( D^{-1}(p) \right)_{j}^{i} dp^{j} + \left( D^{-1}(p_{0}) \right)_{j}^{i} \left[ x^{j}(p) - x^{j}(p_{0}) \right] + f_{jk}^{i}(p) \left[ x^{j}(p) - x^{j}(p_{0}) \right] \left[ x^{k}(p) - x^{k}(p_{0}) \right].$$
(5.4)

Here:  $p \in U_N$ ,  $a^i \in \mathbb{K}$  are constants,  $q_0 \in \overline{U}$  is fixed,  $p_0$  is the above-defined 'projection' of p on  $\overline{U}$ , the integral is taken along some path lying entirely in  $\overline{U}$ , and  $f_{jk}^i: U_N \to \mathbb{K}$  are of class  $C^1$  and they and their first partial derivatives are bounded on  $\overline{U} = U_N \cap N$ . As we saw above (see (5.3) and the lines below it), the integral in (5.4) is path-independent due to the parallel character of  $\{E_i\}$  on N and the symmetry of the connection. If N is real and dim N = 1, this integral is welldefined for any initial frame  $\{E_i\}$ , regardless of the existence of some connection on M.

**Exercise 5.1.** Prove the last assertion! (Hint: in this case, N has locally the form  $\gamma(J)$  where J is open  $\mathbb{R}$ -interval and  $\gamma: J \to \gamma(J)$  is a diffeomorphism.)

For dim N = 0, we take for  $p_0$  some point in  $\overline{U}$  and the above integral identically vanishes. The Jacobian of the transformation  $\{x^i\} \to \{z^i\}$  at  $p_0$  is  $J(p_0) = \det\left(\frac{\partial z^i}{\partial x^j}\Big|_{p_0}\right) = \det\left(D^{-1}(p_0)\right) = 1/\det(D(p_0)) \neq 0, \infty$  for every  $p_0 \in \overline{U}$ . Consequently  $\{z^i\}$  is a coordinate system on some neighborhood  $V \subseteq U_N$  of  $\overline{U}$ .

At last, we calculate the basic vectors of the frame  $\left\{\frac{\partial}{\partial z^j}\right\}$ , defined on V, at  $p_0 \in \overline{U} = V \cap N = U_N \cap N = U \cap N$ :

$$\begin{aligned} \frac{\partial}{\partial z^{i}}\Big|_{p_{0}} &= \frac{\partial x^{j}}{\partial z^{i}}\Big|_{p_{0}} \frac{\partial}{\partial x^{j}}\Big|_{p_{0}} = \left(\left[\frac{\partial z^{k}}{\partial x^{l}}\Big|_{p_{0}}\right]^{-1}\right)^{j}_{i} \frac{\partial}{\partial x^{j}}\Big|_{p_{0}} \\ &= \left(\left(D^{-1}(p_{0})\right)^{-1}\right)^{j}_{i} \frac{\partial}{\partial x^{j}}\Big|_{p_{0}} = D^{j}_{i}(p_{0}) \frac{\partial}{\partial x^{j}}\Big|_{p_{0}} = E_{i}|_{p_{0}}.\end{aligned}$$

Hence the chart (V, z) with coordinates functions  $z^i$  is the one we are looking for. This conclusion ends the proof of Lemma 5.2

**Exercise 5.2.** Prove that equation (5.4) defines the most general local coordinates generating locally a given (parallel) frame on a submanifold. From this view-point, the coincidence of  $\{z^i\}$  on  $\overline{U}$  with  $\{y^i\}$  given via (5.3) (as  $p = p_0$  for  $p \in \overline{U}$ ) is evident as all such coordinate can differ only at points not lying in N.

Using the chart (V, z),  $V \cap N = \overline{U}$ , provided by Lemma 5.2, we can construct a foliation of V 'along' N: defining an  $(\dim M - \dim N)$ -dimensional manifolds  $V_q$  by

$$V_q := \{ p | p \in V, \ z^a(p) = z^a(q), \ a = 1, \dots, \dim N \}, \qquad q \in \overline{U},$$

we see that

$$\bigcup_{q\in\bar{U}} V_q = V, \quad V_q \cap V_{q'} = \emptyset \text{ for } q \neq q', q, q' \in \bar{U}, \quad \text{and } V_q \cap \bar{U} = q \in \bar{U}$$

Besides, by construction (see Lemma 5.2 and its proof), the tangent space  $T_p(V_q)$ ,  $p \in V_q$ , does not contain vectors tangent to N; in particular, we have the direct decomposition  $T_q(M) = T_q(V_q) \oplus T_q(N), q \in \overline{U} \subseteq N$ .

Consequently, for every  $p \in V$ , there is a unique point  $\bar{p} \in \bar{U}$  such that  $p \in V_{\bar{p}}$ .<sup>6</sup>

Now in each  $V_q$  we take a  $(\dim M - \dim N)$ -dimensional normal neighborhood  $V_q^0$  of q (see Definition I.3.6 on page 33). The union  $V^0 := \bigcup_{q \in \overline{U}} V_q^0 \subseteq V$  forms a neighborhood of  $\overline{U}$  such that, for every  $p \in V^0$  in the submanifold  $V_{\overline{p}}^0$ , there exists a unique geodesic  $\beta_p : J \to V_{\overline{p}}, J \ni 0$  joining  $\overline{p}$  and p in  $V_{\overline{p}}$ , i.e., more precisely,  $p = \exp_{\overline{p}}(tX_p) = \beta_p(t)$  where  $\beta_p(0) = \overline{p}, \dot{\beta}_p(0) = X_p$  for some unique  $X_p \in T_{\overline{p}}(V_{\overline{p}}) \subseteq \{\sum_{k=\dim N+1}^{\dim M} \alpha_k E_k |_{\overline{p}} : \alpha_{\dim N+1}, \ldots, \alpha_{\dim M} \in \mathbb{K}\}$  and  $t \in J$ . Just to prove the existence of this unique  $X_p$  corresponding to every p in a neighborhood of  $\overline{U}$  were needed all of the considerations after (5.3). With its help, we define the coordinates  $\{y^i\}$  on  $V^0$  (cf. (5.4)) by

$$y^{i}(p) = a^{i} + \int_{q_{0}}^{\bar{p}} \left( D^{-1}(p) \right)_{j}^{i} \mathrm{d}p^{j} + t \left( D^{-1}(\bar{p}) \right)_{j}^{i} X_{p}^{j}.$$
(5.5)

Here:  $p \in V^0$ ,  $a^i \in \mathbb{K}$  are constants,  $q_0 \in \overline{U}$  is fixed,  $\overline{p}$  is the unique element in  $\overline{U}$  for which  $V_{\overline{p}} \ni p$ ,  $p^i := x^i(p)$ , the integral is taken along some path in  $\overline{U}$ , t is the increment of the parameter when passing from  $\overline{p}$  to p along the unique geodesic connecting them in  $V_{\overline{p}}$ , and  $X_p$  is the tangent vector to this geodesic at  $\overline{p}$ (corresponding to t = 0). The quantities  ${}^{y}X_p^i := (D^{-1}(\overline{p}))_j^i X_p^j$  are the components of  $X_p$  in  $\{E_i|_{\overline{p}}\}$  and, since  $X_p \in T_{\overline{p}}(V_{\overline{p}})$ , we have  ${}^{y}X_p^a = 0$  for  $a = 1, \ldots$ , dim N.

Since (5.5) corresponds to (5.4) with  $p_0 = \bar{p}$ ,  $x^j(p) = x^j(p_0) + t^y X_p^j$ ,  $f_{jk}^i = 0$ , and  $V = V^0$ , from the considerations after (5.4) in the proof of Lemma 5.2, it follows that there exist a neighborhood  $U^0 \subseteq V^0$  of  $\bar{U}$  (=  $U^0 \cap N = V^0 \cap N =$  $\cdots$ ) on which the transformation  $\{x^i\} \to \{y^i\}$ , given via (5.5), is a well-defined diffeomorphism.<sup>7</sup>

Ending these long considerations, we will prove that the chart  $(U^0, y)$  of M is normal on N, i.e., on  $\overline{U} = U^0 \cap N$ . Till now the chart (U, x) was complete arbitrary. Choosing  $(U, x) = (U^0, y)$ , we get  $\frac{\partial}{\partial x^i}\Big|_q = \frac{\partial}{\partial y^i}\Big|_q = E_i\Big|_q = D_i^j(q)\frac{\partial}{\partial x^j}\Big|_q$ ,  $q \in \overline{U}$ . Thus  $D|_{\overline{U}} = \mathbb{1}$  or  $D_i^j(q) = \delta_i^j$  for  $q \in \overline{U}$ . From the parallel character of  $\{E_i\}$  on N, we derive  $0 = \nabla_{\partial/\partial x^a} E_j = \nabla_{\partial/\partial x^a} \frac{\partial}{\partial x^j} = \Gamma_{ja}^i E_i$  on  $\overline{U}$ , so that  $\Gamma_{ja}^i\Big|_{\overline{U}} = 0$ 

 $<sup>^{6}\</sup>mathrm{In}$  [55] this assertion is mentioned and used but its rigorous proof, which is not trivial, is only partially presented, the main details are missing.

<sup>&</sup>lt;sup>7</sup>For another proof of this fact, see [55, pp. 21–22].

for  $a = 1, \ldots, \dim N$ . Since  $p \in U^0$  and  $\bar{p} \in \bar{U}$  are situated on a geodesic, the coordinates (5.5) must satisfy the geodesic equation (I.3.23) with p for  $\gamma(s), y^j(p)$  for  $\gamma^j(s)$ , and t for s, which results in  $\Gamma^i_{jk}(p)X^j_pX^k_p = 0$ . Taking into account that p is arbitrary,  $\Gamma^i_{jk} = \Gamma^i_{kj}$ , and  $X^a = 0$  for  $a = 1, \ldots, \dim N$  as  $X_p \in T_{\bar{p}}(V_{\bar{p}})$  and  $T_{\bar{p}}(V_{\bar{p}}) \cap T_{\bar{p}}(N) = \emptyset$ , from here we find  $\Gamma^i_{jk}|_{\bar{U}} = 0$  for  $j, k > \dim N$ . Combining this with  $\Gamma^i_{ja}|_{\bar{U}} = 0$  for  $a < \dim N$ , we, finally obtain  $\Gamma^i_{jk}|_{\bar{U}} = 0$  for all values of all indices.

Consequently the chart  $(U^0, y)$  is normal on  $\overline{U}$ , which ends the proof of Proposition 5.2

The coordinates (5.5) are direct analogue and multidimensional generalization of the Fermi coordinates along paths without self-intersections (see Subsection 3.1), to which they reduce in the one-dimensional case. For this reason, we propose, as in [55], to call *Fermi coordinates on a submanifold* this special kind of coordinates normal on a submanifold.

It should be emphasized, the Fermi coordinates on a submanifold are local by their nature. Generally global such coordinates, i.e., on the whole submanifold, do not exist. But, it is almost evident, we can construct a family of overlapping such coordinate systems whose domains form a neighborhood of the submanifold given.

From (5.5) (see also (5.4) for  $x^i(p) = x^i(p_0) + (D^{-1}(p_0))^i_j X^j_p$ ), one can derive that the basic vector fields of the normal frames associated to the Fermi coordinates are

$$\frac{\partial}{\partial y^i}\Big|_q = \frac{\partial x^j}{\partial y^i}\Big|_q \frac{\partial}{\partial x^j}\Big|_q = \left(\left[\frac{\partial x^k}{\partial y^l}\Big|_q\right]^{-1}\right)_i^j \frac{\partial}{\partial x^j}\Big|_q = D_i^j(q)\frac{\partial}{\partial x^j}\Big|_q = E_i|_q$$

for  $q \in \overline{U} \subseteq N$ . Consequently  $\left\{\frac{\partial}{\partial y^j}\right\}$  on  $\overline{U}$  coincides with the initial frame  $\{E_i\}$  on  $\overline{U}$ . Since  $\{E_i\}$  is parallel by definition, so is  $\left\{\frac{\partial}{\partial y^j}\right\}$ . This result completely agrees with Proposition I.5.6.

Combining Propositions 5.1 and 5.2, we obtain the following fundamental result.

**Theorem 5.1.** Let N be a submanifold of a  $C^3$  manifold M endowed with  $C^1$  torsionless linear connection. There exist coordinates (in M) normal on N for the connection given if and only if the conditions (5.1) are valid. If (5.1) holds, the above-constructed Fermi coordinates are normal on the intersection of their domain with N.

If M admits Fermi coordinates on its submanifold N, we can easily describe all frames normal on N. If  $(U^0, y)$  is a chart normal on N, i.e., on  $\overline{U} = U^0 \cap N$ , with  $\{y^i\}$  being Fermi coordinates on  $U^0$ , the frame  $\{\frac{\partial}{\partial y^j}\}$  is, by Definition I.5.2, normal on  $\overline{U}$ . According to Proposition I.5.2, in  $U^0$  all frames normal on  $\overline{U}$  have the form  $\{E_i = A_i^j \frac{\partial}{\partial y^j}\}$  where the non-degenerate  $C^2$  matrix-valued function  $A := [A_i^j]$  on  $U^0$  is such that  $\frac{\partial A}{\partial y^j}|_{\overline{U}} = 0$ . **Exercise 5.3.** Prove that the general form of the matrix-valued function A is (cf. (5.11) in Subsection 5.2)

$$A(p) = B(p_0) - \frac{\partial B}{\partial y^j}\Big|_{p_0} [y^j(p) - y^j(p_0)] + B_{jk}(p)[y^j(p) - y^j(p_0)][y^k(p) - y^k(p_0)]$$
(5.6)

where  $p \in U^0$ , B is non-degenerate  $C^2$  matrix-valued function on  $U^0$ ,  $p_0$  is the 'projection' of p on  $\overline{U}$  as defined before (5.4), the matrix-valued functions  $B_{jk}$  on  $\overline{U}$  are of class  $C^1$ , and B,  $\frac{\partial B}{\partial y^i}$ , and  $B_{jk}$  together with their partial derivatives are bounded on  $\overline{U}$ .

According to Theorem 5.1, between the normal frames exist holonomic ones. The *full local description* of all coordinates normal on a submanifold will be given in the next subsection. In the particular case considered, they are defined on  $U^0$ and normal on  $\bar{U} = U^0 \cap N$ .

If we cover N with overlapping systems of Fermi coordinates  $(U_{\lambda}^{0}, y_{\lambda}), \lambda \in \Lambda \neq \emptyset$ , we can construct on N a global (on N) frame normal on the whole submanifold N. Let  $\{E_{i}^{\lambda}\}$  be a frame defined on  $U_{\lambda}^{0}$  and normal on  $\bar{U}_{\lambda} = U_{\lambda}^{0} \cap N$ . If  $q \in N$  is such that  $q \in \bar{U}_{\lambda}$  for a single  $\lambda \in \Lambda$ , we put  $E_{i}|_{p} := E_{i}^{\lambda}|_{p}, p \in U_{\lambda}^{0}$ ; if such  $\lambda \in \Lambda$  are more than one, we arbitrarily choose some  $\mu \in \Lambda$  and put  $E_{i}|_{p} := E_{i}^{\mu}|_{p}$ ,  $p \in U_{\lambda}^{0}$  of N and is normal on N. Obviously, in this way can be constructed all frames normal on N. Generally these global on N normal frames are not smooth in the regions of overlapping of two or more local Fermi coordinates but locally, on each particular  $U_{\lambda}^{0}$ , they can be constructed in a smooth way.

### 5.2. Complete description

Below, following the main ideas of Subsection 3.2 and [83], we are going to give a full constructive description of the frames (resp. coordinates) normal on a submanifold of manifold with arbitrary linear connection, if such frames (resp. coordinates) exist.

Let M be  $C^2$  manifold endowed with  $C^0$  linear connection (of arbitrary torsion) and N be its submanifold. The problem is to find necessary and sufficient conditions for the existence of frames defined on a neighborhood of N and normal on N. If such frames exist, we want to give their complete description. From Proposition 5.1, we know that the conditions (5.1) are necessary for the existence of frames normal on N. Our first goal now is to prove the sufficiency of these conditions, which assertion is a part of the following theorem (cf. Theorem 3.1).

**Theorem 5.2.** Let N be submanifold of  $C^3$  manifold M provided with  $C^1$  linear connection, with or without torsion. Then:

(i) The equation (5.1) is a necessary and sufficient condition for the existence of frames normal on N.

(ii) If (5.1) holds, every point  $q_0 \in N$  has a neighborhood  $U_N$  (in M) such that on  $U_N$  exist  $C^1$  frames normal on N, i.e., on  $\overline{U} = U_N \cap N \ni q_0$ . Moreover, on  $U_N$  exist coordinates  $\{x^i\}$  such that every  $p = x^{-1}(s, t') \in U_N$ ,  $(s, t') \in W_N \times W' \subseteq \mathbb{K}^{\dim N} \times \mathbb{K}^{\dim M-\dim N}$ , has a unique 'projection'  $p_0 = x^{-1}(s, t'_0) \in \overline{U} \subseteq N$  for fixed  $t'_0 \in W'$  which depends on x but not on t'. In these local coordinates, all frames defined on  $U_N$  and normal on  $\overline{U} = U_N \cap N$  are  $\{E'_i = A^j_i \frac{\partial}{\partial x^j}\}$  with  $A = [A^j_i]$  given via equation (5.11) below in which  $s_0 = \overline{x}(q_0)$  are the coordinates of  $q_0$  in the chart  $(\overline{U}_N, \overline{x})$  of N induced by the chart  $(U_N, x)$  of M.

*Proof.* Let  $\{E_i\}$  be a frame defined in a neighborhood of N. From Section I.5, we know that all frames  $\{E'_i\}$  defined on the same neighborhood and normal on N are  $\{E'_i = A^j_i E_j\}$  where the  $C^1$  non-degenerate matrix-valued function  $A := [A^j_i]$  is a solution of the normal frame equation (I.5.4) with N for U, viz.

$$\Gamma_k(q)A(q) + (E_k(A))|_q = 0$$
(5.7)

where  $q \in N$  and  $\Gamma_k := [\Gamma_{jk}^i]_{i,j=1}^{\dim M}$  are the matrices of the connection coefficients in  $\{E_i\}$ . Hence, the finding of the general solution of (5.7) with respect to A is equivalent to the complete description of all frames normal on N, if any.

Let  $q \in N$  and (U, x) be a chart of M with  $\overline{U} := U \cap N \ni q$ . So  $x \colon U \to W$  is homeomorphism, which actually is a  $C^1$  diffeomorphism, for some open subset Wof  $\mathbb{K}^{\dim M}$ . Representing W as a Cartesian product of two open sets  $W'' \subseteq \mathbb{K}^{\dim N}$ and  $W' \subseteq \mathbb{K}^{\dim M - \dim N}$ , i.e.,  $W = W'' \times W'$ , from the definition of a submanifold (p. 7) follows the existence of  $t'_0 = (t'_0^{\dim N+1}, \ldots, t'_0^{\dim M}) \in W'$  such that

$$x(q) = (x^1(q), \dots, x^{\dim N}(q)) \times t'_0$$
 for all  $q \in N$ 

and  $(\bar{U}, \bar{x})$ , with  $\bar{x}(q) := (x^1(q), \ldots, x^{\dim N}(q))$  for  $q \in N$ , provides a coordinate system  $\{\bar{x}^i\}$  on  $\bar{U}$ . Let  $W_N := \bar{x}(\bar{U}) \subseteq W''$ , i.e.,  $\bar{x} : \bar{U} \to W_N$  is the local coordinate homeomorphism from  $\bar{U}$  on the open subset  $W_N$  of  $\mathbb{K}^{\dim N}$ .

Consider the chart  $(U_N, x)$  of M with  $U_N := x^{-1}(W_N \times W')$ .<sup>8</sup> Evidently  $U_N \cap N = \overline{U} \ni q$  and, for every  $p \in U_N$ , we have x(p) = (s, t') for some unique  $s = (s^1, \ldots, s^{\dim N}) \in W_N \subseteq \mathbb{K}^{\dim N}$  and  $t' = (t'^{(\dim N+1}, \ldots, t'^{\dim M}) \in W' \subseteq \mathbb{K}^{\dim M-\dim N}$ ; in particular, if  $q \in \overline{U}$ , then  $x(q) = (s, t'_0)$ . These facts give us a possibility to define a natural 'projection'  $\pi \colon U_N \to \overline{U}$  such that if  $p = x^{-1}(s, t') \in U_N$ , then  $\pi \colon p \mapsto p_0 := \pi(p) := x^{-1}(s, t'_0)$ , i.e.,  $p_0$  is the unique point in  $\overline{U}$  with  $\overline{x}(p_0) = s$ .<sup>9</sup>

Suppose the matrix-valued function A is of class  $C^3$ . There exist  $C^1$  matrixvalued functions  $B_{kl}$  on  $U_N$  such that they and their partial derivatives are

<sup>&</sup>lt;sup>8</sup>Strictly speaking, x in  $(U_N, x)$  has to be replace by the restriction  $x_N := x|_{U_N}$ ; we ignore this to save some writing.

<sup>&</sup>lt;sup>9</sup>The triple  $(U_N, \pi, \overline{U})$  is a fibre bundle with base  $\overline{U}$ , (total) bundle space  $U_N$ , and projection  $\pi$  (see further Section IV.2). From here the name 'projection' for  $\pi$  comes from.

bounded on  $\overline{U}$  and (cf. (3.15))

$$A(x^{-1}(s,t')) = A(x^{-1}(s,t'_{0})) + \sum_{k=\dim N+1}^{\dim M} \frac{\partial A(x^{-1}(s,t'))}{\partial t'^{k}} \bigg|_{t'=t'_{0}} (t'^{k} - t'^{k}_{0})$$
  
+ 
$$\sum_{k,l=\dim N+1}^{\dim M} B_{kl}(x^{-1}(s,t'))(t'^{k} - t'^{k}_{0})(t'^{l} - t'^{l}_{0})$$
(5.8)

Choosing  $E_k|_p = \frac{\partial}{\partial x^k}|_p$  for  $p \in U_N$ , we get

$$E_a(A)|_{\pi(p)} = \frac{\partial A\left(x^{-1}(\boldsymbol{s}, \boldsymbol{t}')\right)}{\partial s^a}\Big|_{\boldsymbol{t}'=\boldsymbol{t}'_0}, \qquad E_k(A)|_{\pi(p)} = \frac{\partial A\left(x^{-1}(\boldsymbol{s}, \boldsymbol{t}')\right)}{\partial t'^k}\Big|_{\boldsymbol{t}'=\boldsymbol{t}'_0},$$

where  $a = 1, \ldots, \dim N$ ,  $k = \dim N + 1, \ldots, \dim M$ ,  $p = x^{-1}(s, t')$  for  $(s, t') \in W_N \times W'$ . Taking this into account and substituting equation (5.8) into (5.7), we find the equivalent to (5.7) system of equations for A on  $\overline{U}$ :

$$\frac{\partial A(x^{-1}(\boldsymbol{s},\boldsymbol{t}'_0))}{\partial s^a} = -\Gamma_a(x^{-1}(\boldsymbol{s},\boldsymbol{t}'_0))A(x^{-1}(\boldsymbol{s},\boldsymbol{t}'_0))$$
(5.9a)

$$\frac{\partial A\left(x^{-1}(\boldsymbol{s},\boldsymbol{t}')\right)}{\partial t'^{k}}\Big|_{\boldsymbol{t}'=\boldsymbol{t}'_{0}} = -\Gamma_{k}\left(x^{-1}(\boldsymbol{s},\boldsymbol{t}'_{0})\right)A\left(x^{-1}(\boldsymbol{s},\boldsymbol{t}'_{0})\right),\tag{5.9b}$$

where  $a = 1, \ldots, \dim N$  and  $k = \dim N + 1, \ldots, \dim M$ . Since equation (5.9a) corresponds to (4.4a) with  $W_N$  for N,  $m = \dim M$ ,  $e_a = \frac{\partial}{\partial s^a}$ ,  $q = x^{-1}(s, t'_0)$ , and  $Z_a = -\Gamma_a$ , by Lemma 4.1, Assertion (i) it has solutions if and only if the conditions (5.1") hold on  $U_N$ . But as equation (5.1") is equivalent to (5.1) (see Remark 5.3 on page 114), this means that (5.7) has solutions on N with respect to A iff (5.1) is valid.

Now we suppose (5.1) to be true. Applying Lemma 4.1, Assertion (ii) and taking into account Remark 4.5 on page 106, we find the general solution of (5.9a) in the form (cf. (3.22))

$$A(x^{-1}(s, t'_0)) = Y(s, s_0; -\Gamma_1 \circ (x^{-1}(\cdot, t'_0)), \dots, -\Gamma_{\dim N} \circ (x^{-1}(\cdot, t'_0))) B(t'_0).$$
(5.10)

Here Y is the unique solution of (5.9a) with A = Y satisfying the initial conditions

$$Y|_{\boldsymbol{s}=\boldsymbol{s}_0} = \mathbb{1}_{\dim M} \tag{5.9c}$$

for some fixed  $\mathbf{s}_0 \in W_N \subseteq \mathbb{K}^{\dim N}$ , and *B* is non-degenerate dim  $M \times \dim M$  matrix-valued function on W' of class  $C^1$ . Substituting (5.9b) into (5.8) and using (5.10), we get:

$$A(p) = \left\{ \mathbb{1} - \Gamma_k(p_0) [x^k(p) - x^k(p_0)] \right\} \\ \times Y(\bar{x}(p_0), \boldsymbol{s}_0; -\Gamma_1 \circ (x^{-1}(\cdot, \boldsymbol{t}'_0)), \dots, -\Gamma_{\dim N} \circ (x^{-1}(\cdot, \boldsymbol{t}'_0))) B(\boldsymbol{t}'_0) \\ + B_{kl}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]$$
(5.11)

where  $\mathbf{s}_0 \in W_N \subseteq \mathbb{K}^{\dim N}$  is fixed and we used  $p = x^{-1}(\mathbf{s}, \mathbf{t}') \in U_N$ ,  $p_0 = \pi(p) = x^{-1}(\mathbf{s}, \mathbf{t}'_0) \in \overline{U} = U_N \cap N$ , and  $\overline{x}(p_0) = \mathbf{s}$ . Notice, in this equality all terms corresponding to  $k, l = 1, \ldots, \dim N$  are zero as  $x^a(p) = x^a(p_0) = \overline{x}^a(p_0) = \mathbf{s}^a$  for  $a = 1, \ldots, \dim N$ . Since (5.11) is the general solution of (5.7) on  $\overline{U}$  by construction, the above results, together with Proposition 5.1 complete the proof of Theorem 5.2.

Remark 5.7. Since  $t'_0$  in (5.8) depends only on the choice of the initial coordinates  $\{x^i\}$ , the matrix  $B(t'_0)$  is constant on N for every such particular choice. This agrees with Proposition I.5.2 and Corollary I.5.1.

*Remark* 5.8. Notice, equation (5.11) corresponds to (5.6) for  $x^i = y^i$  ( $\{y^i\}$  are Fermi coordinates) and  $B(p) = Y(\cdots)$ . The terms corresponding to the index values from 1 to dim N are zero in the both equations.

*Remark* 5.9. Theorem 5.2 is a first in a group of similar Theorems III.7.1, III.8.1, IV.6.1, and IV.9.1 generalizing it. (See also the comments after the proof of Theorem IV.9.1 on page 269.)

**Corollary 5.1.** Let N be a submanifold of  $C^3$  manifold with  $C^1$  linear connection. On N (locally) exist normal frames if and only if the parallel transport in N along paths lying in N is path-independent.<sup>10</sup>

*Proof.* See Theorem 5.2 and Lemma 5.1 on page 112

**Corollary 5.2.** Let N be a submanifold of  $C^3$  manifold with  $C^1$  linear connection. On N (locally) exist normal frames iff on N exists a parallel frame with respect to the paths in N.

*Proof.* If  $\{E_i\}$  is frame normal on N, then it is also parallel on N, due to Proposition I.5.6. Conversely, let  $\{E_i\}$  be parallel frame on N, i.e.,  $(\nabla_X E_i)|_N = 0$  for all  $X \in \mathfrak{X}(N)$ . The parallel transport of some  $Y_0 = Y_0^i E_i|_{p_0} \in T_{p_0}(M)$ ,  $p_0 \in N$ , along  $\gamma: J \to N$  from  $p_0 = \gamma(s_0)$  to every  $p = \gamma(s)$  for some  $s_0, s \in J$  gives the vector  $Y_p = Y_{\gamma(s)}$  where Y is such that (see Definition I.3.2)  $(\nabla_{\dot{\gamma}}Y)|_{\gamma(s)} = 0$  and  $Y_{p_0} = Y_0$ . But  $\nabla_{\dot{\gamma}}Y|_{\gamma(s)} = (\dot{\gamma}^k \nabla_{E_k}(Y^j E_j))|_{\gamma(s)} = \dot{\gamma}^k(s)(\nabla_{E_k}Y^j)|_{\gamma(s)}E_j|_{\gamma(s)} = \dot{\gamma}(s)(Y_{\gamma(s)}^j)E_j|_{\gamma(s)}$ , so that  $\dot{\gamma}(s)(Y_{\gamma(s)}^j) = 0$ , i.e.,  $Y_{\gamma(s)}^j = Y_0^j$  = const. Hence the parallel transport of  $Y_0$  from  $p_0$  to p along any path connecting them in N gives the vector  $Y_0^j E_j|_p$  and, consequently, it is path-independent in N. Now Corollary 5.1 implies the existence of frames normal on N. □

Theorem 5.2 gives a complete local description of the frames normal on a submanifold. On this base a complete global description of all frames normal on a submanifold can be given. Indeed, let  $\{\bar{U}^{\lambda}|\lambda \in \Lambda\}, \Lambda \neq \emptyset$ , be an open cover of N such that for every  $\lambda \in \Lambda$  there is a neighborhood  $U_{\lambda}^{\lambda}$  in M such that  $\bar{U}^{\lambda} = U_{\lambda}^{\lambda} \cap N$  and on  $U_{\lambda}^{\lambda}$  exist frames  $\{E_{i}^{\lambda}\}$  normal on  $\bar{U}^{\lambda}$ , like the ones constructed above. In the neighborhood  $V_{N} := \bigcup_{\lambda \in \Lambda} U_{\lambda}^{\lambda}$  of N we define a frame  $\{E_{i}\}$  by putting

<sup>&</sup>lt;sup>10</sup>If M is multiply connected, see Remark I.3.5 on page 29.

 $E_i|_p = E_i^{\lambda}|_p$ ,  $p \in V_N$  for some  $\lambda \in \Lambda$  for which  $p \in U_N^{\lambda}$ ; if there are more than one such  $\lambda$ , we arbitrary choose some of them. By construction  $\{E_i\}$  is a global frame which is normal on the whole submanifold N. Generally  $\{E_i\}$  is not smooth, even not continuous, on the sets (neighborhoods) of intersection (overlapping) of two or more neighborhoods  $U_N^{\lambda}$ . Relying on Theorem 5.2 (ii), we conclude that in this way can be constructed *all* frames (globally) normal on any submanifold.

Now we have at our disposal the full machinery required for the *complete* description of all local coordinates normal on a given submanifold. Meanwhile we shall find a necessary and sufficient condition for the existence of coordinates normal on a submanifold.

The following theorem provides a *complete constructive description of all local coordinates normal on a submanifold.* 

**Theorem 5.3.** Let N be submanifold of  $C^3$  manifold M endowed with  $C^1$  torsion free linear connection for which the condition (5.1) holds. Then for every point  $q \in N$  exists a chart  $(U_z, z)$  of M with  $U_z \ni q$  which is normal on  $U_z \cap N$ . The coordinate functions  $z^i : U_z \to \mathbb{K}$  of all such charts normal on N are given via equation (5.16) below.

*Proof.* Suppose M is a  $C^3$  manifold endowed with  $C^0$  torsionless linear connection  $\nabla$  for which the (integrability) conditions (5.1) are valid on some submanifold N of M. Theorem 5.1 ensures the existence of coordinates normal on N, the Fermi coordinates providing an example. The problem is to find *all* such coordinates. For its solution we shall follow the scheme of Subsection 3.2 for the complete description of the local coordinates normal along a path (see p. 100 and further).

Consider the chart  $(U_N, x)$  of M, defined in the proof of Theorem 5.2, for which each point  $p = x^{-1}(\mathbf{s}, \mathbf{t}') \in U_N$  has a unique projection  $p_0 := \pi(p) = x^{-1}(\mathbf{s}, \mathbf{t}'_0)$  on N. By Theorem 5.2, Assertion (ii) on  $U_N$  all frames normal on  $\overline{U} = U_N \cap N$  are  $\{E'_i = A^j_i \frac{\partial}{\partial x^j}\}$  with  $A = [A^j_i]$  given via (5.11). Therefore if local coordinates  $\{z^i\}$ , defined in a neighborhood  $U_z$  of  $\overline{U}$  in  $U_N, U_z \cap N = \overline{U}$  and normal on  $\overline{U}$ , exist, then they must satisfy the conditions

$$\frac{\partial}{\partial z^{i}}\Big|_{q} = E'_{i}|_{q} = A^{j}_{i}(q)\frac{\partial}{\partial x^{j}}\Big|_{q} \iff \frac{\partial z^{i}}{\partial x^{j}}\Big|_{q} = \left(A^{-1}(q)\right)^{i}_{j}.$$
(5.12)

Here  $q = x^{-1}(s, t'_0) \in \overline{U}$  and A is some matrix-valued function given by equation (5.11). Since the manifold is of class  $C^3$ , the Taylor expansion

$$z^{i}(p) = z^{i}(p_{0}) + a^{i}_{j}(p_{0})[x^{j}(p) - x^{j}(p_{0})] + a^{i}_{jk}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})]$$
(5.13)

is valid for every  $p \in U_z$ ,  $p_0 = \pi(p)$ . Here  $a_j^i : \overline{U} \to \mathbb{K}$  are of class  $C^1$  and the  $C^1$  functions  $a_{jk}^i : \overline{U} \to \mathbb{K}$  and their first partial derivatives are bounded on  $\overline{U}$ . Notice, due to  $p_0 = \pi(p)$ , we have  $x^a(p) = x^a(p_0) = s^a$  for  $a = 1, \ldots, \dim N$ ; so in (5.13)

the terms with  $j, k = 1, ..., \dim N$  are zero. From (5.12) and (5.13), we derive:

$$\frac{\partial z^{i}(p_{0})}{\partial x^{a}} = \frac{\partial z^{i}}{\partial x^{a}}\Big|_{p_{0}} = \left(A^{-1}(p_{0})\right)_{a}^{i}, \qquad a = 1, \dots, \dim N$$
$$\iff z^{i}(p_{0}) = a^{i} + \int_{q_{0}}^{p_{0}} \sum_{a=1}^{\min N} \left(A^{-1}(q)\right)_{a}^{i} dq^{a}$$
$$a_{j}^{i}(p_{0}) = \frac{\partial z^{i}(p)}{\partial x^{j}}\Big|_{p_{0}} = \frac{\partial z^{i}}{\partial x^{j}}\Big|_{p_{0}} = \frac{\partial z^{i}}{\partial t^{\prime j}}\Big|_{t^{\prime}=t_{0}^{\prime}} = \left(A^{-1}(p_{0})\right)_{j}^{i},$$

where  $j = \dim N+1, \ldots, \dim M, a^i \in \mathbb{K}$  are constants representing the coordinates of  $p_0$  in  $\{z^i\}$  with respect to some fixed point  $q_0 \in \overline{U}$ , and the integral is taken along some path lying entirely in  $\overline{U}$ . The integral in the above equality is pathindependent. This can be proved in the same way we proved the path-independence of the integral in (5.3). As  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  is normal, by Proposition I.5.6 it is also parallel in U which, in particular, means that  $\nabla_{\partial/\partial x^a} E_k = 0$ . This equation is equivalent to  $\frac{\partial A}{\partial x^a} = -\Gamma_a A$  or  $\frac{\partial A^{-1}}{\partial x^a} = A^{-1}\Gamma_a$ , by virtue of which and the torsionless of the connection the integrability conditions for the path-independence of the above integral are identically valid on  $\overline{U}$ :

$$\frac{\partial^2 z^i}{\partial x^{[a} x^{b]}} = \frac{\partial}{x^{[a}} \left(A^{-1}\right)^i_{b]} = \left(\frac{\partial A^{-1}}{\partial x^{[a}}\right)^i_{b]} = \left(A^{-1} \Gamma_{[a}\right)^i_{b]} = \left(A^{-1}\right)^i_k \Gamma^k_{\ [ba]} \equiv 0.$$

Substituting the values of  $z^i(p_0)$  and  $a^i_j(p_0)$  obtained above in (5.13), we derive the following general form of the coordinates  $\{z^i\}$ :

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \sum_{a=1}^{\dim N} \left( A^{-1}(q) \right)_{a}^{i} dq^{a} + \left( A^{-1}(p_{0}) \right)_{j}^{i} \left[ x^{j}(p) - x^{j}(p_{0}) \right] + a^{i}_{jk}(p) \left[ x^{j}(p) - x^{j}(p_{0}) \right] \left[ x^{k}(p) - x^{k}(p_{0}) \right], \quad (5.14)$$

with A given by (5.11). Since the Jacobian of the transformation  $\{x^i\} \to \{z^i\}$  at  $q \in \overline{U}$  is

$$\det\left[\frac{\partial x^{i}}{\partial z^{j}}\Big|_{q}\right] = \left(\det\left[\frac{\partial z^{i}}{\partial x^{j}}\Big|_{q}\right]\right)^{-1} = \left(\det(A^{-1}(q))\right)^{-1} = \det(A(q)) \neq 0, \infty,$$

there exists a neighborhood  $U_z$  of  $\overline{U}$  in  $U_N$  such that  $U_z \cap N = \overline{U}$  and  $(U_z, z)$  is a chart of M with coordinate functions  $z^i$ .

The charts  $(U_z, z)$  (locally) generate the restrictions to N of all frames normal on  $U_z$  but generally these charts are not normal on N, i.e., on  $\overline{U}$ . To make them such, we are going to choose the functions  $a_{jk}^i \colon U_z \to \mathbb{K}$  in such a way as to satisfy equation (I.5.4') on page 41 with  $U_z$  for  $U \cap V$  and z for x. Repeating the arguments leading from (3.26) to (3.26') with  $q \in \overline{U}$  for  $\gamma(s)$ , we get<sup>11</sup>

$$\frac{\partial z^{i}}{\partial x^{l}}\Big|_{q}\Gamma^{l}{}_{jk}(q) = \frac{\partial^{2} z^{i}}{\partial x^{j} x^{k}}\Big|_{q}, \qquad q \in \bar{U}.$$
(5.15)

The derivatives entering here can be calculated from (5.14):

$$\frac{\partial z^{i}}{\partial x^{l}}\Big|_{q} = \left(A^{-1}(q)\right)_{l}^{i},$$
$$\frac{\partial^{2} z^{i}}{\partial x^{a} x^{k}}\Big|_{q} = \left(\frac{\partial A^{-1}}{\partial x^{a}}\Big|_{q}\right)_{k}^{i}, \quad \frac{\partial^{2} z^{i}}{\partial x^{m} x^{n}}\Big|_{q} = 2a^{i}_{(mn)}(q)$$

where  $q \in \overline{U}$ ,  $a = 1, \ldots, \dim N$ ,  $m, n = \dim N + 1, \ldots, \dim M$ , and A is provided by (5.11). (Notice, in (5.14) the terms with  $j, k = 1, \ldots, \dim N$  vanish as  $p_0 = \pi(p)$ , so that  $x^a(p) = \overline{x}^a(p) = \overline{x}^a(p_0)$  for  $a = 1, \ldots, \dim N$ .) Substituting these expressions into (5.15) and using  $\frac{\partial A^{-1}}{\partial x^a} = A^{-1}\Gamma_a$  (see the paragraph preceding the one containing equation (5.14)), we find:

$$(A^{-1}(q))_l^i \Gamma^l_{ak}(q) = (A^{-1}(q)\Gamma_a(q))_k^i, \qquad q \in \bar{U}, \quad a = 1, \dots, \dim N$$
$$(A^{-1}(q))_l^i \Gamma^l_{mn}(q) = 2a_{(mn)}^i(q), \qquad m, n = \dim N + 1, \dots, \dim M.$$

The first of these equalities is an identity as  $(\Gamma_a)_k^l := \Gamma_{ka}^l = \Gamma_{ak}^l$  (the connection is symmetric); this reflects the vanishment of the terms with  $j, k = 1, \ldots, \dim N$ in (5.14). The second equation determines  $a_{(mn)}^i$  on  $\bar{U}$ ; on  $U_z \setminus \bar{U}$  the values of  $a_{(mn)}^i$ , as well as  $a_{[mn]}^i$ , are left arbitrary from the condition that  $\{z^i\}$  is normal on  $\bar{U}$ . Since M is supposed to be of class  $C^3$ , there exist continuous bounded functions  $a_{ikl}^i: \bar{U} \to \mathbb{K}$  such that

$$a_{jk}^{i}(p) = a_{jk}^{i}(p_{0}) + a_{jkl}^{i}(p)[x^{l}(p) - x^{l}(p_{0})].$$

At last, inserting in this expansion the obtained expressions for  $a_{jk}^i(p_0)$  and the result into (5.14), we, finally, get:

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \sum_{a=1}^{\dim N} \left(A^{-1}(q)\right)_{a}^{i} dq^{a} + \left(A^{-1}(p_{0})\right)_{j}^{i} \left[x^{j}(p) - x^{j}(p_{0})\right] + \left(A^{-1}(p_{0})\right)_{l}^{i} \Gamma^{l}{}_{jk}(p_{0}) \left[x^{j}(p) - x^{j}(p_{0})\right] \left[x^{k}(p) - x^{k}(p_{0})\right] + a^{i}_{jkl}(p) \left[x^{j}(p) - x^{j}(p_{0})\right] \left[x^{k}(p) - x^{k}(p_{0})\right] \left[x^{l}(p) - x^{l}(p_{0})\right].$$
(5.16)

Here:

$$p = x^{-1}(s, t') \in U_z, \quad p_0 = x^{-1}(s, t'_0) \in \bar{U} = U_z \cap N, a^i \in \mathbb{K}, \quad \bar{x}(p_0) = s, \quad q_0 \in \bar{U} \text{ is fixed},$$
(5.17)

 $<sup>^{11}\</sup>mathrm{From}$  (5.15), once again, follows that normal coordinates exist only for torsion free linear connections.

and the matrix-valued  $C^1$  function  $A: U_z \to \operatorname{GL}(\dim M, \mathbb{K})$  is given by (5.11) in which B is dim  $M \times \dim M$  non-degenerate matrix-valued function, the matrixvalued functions  $B_{lk}$  on  $U_z$  are of class  $C^1$  and they and their first partial derivatives are bounded on  $\overline{U}$ , and Y is the unique solution of the initial-value problem (4.4) with  $\frac{\partial}{\partial s^a}, -\Gamma_a, \bar{x}(p_0), s_0 = \bar{x}(q_0)$ , and s for  $e_a, Z_a, p, p_0$ , and  $q_0$ , respectively. This ends the proof of Theorem 5.3.

*Remark* 5.10. Theorem 5.3 remains valid if N is fixed and the connection is torsionless only on N (see Proposition 5.3).

*Remark* 5.11. Generally, under the conditions of Theorem 5.3, not all of the normal frames provided by Theorem 5.2, point (ii), are holonomic.

**Proposition 5.3.** Let N be submanifold of a  $C^3$  manifold endowed with  $C^1$  linear connection  $\nabla$  admitting frames normal on N. Holonomic frames normal on N or, equivalently, coordinates normal on N for  $\nabla$  exist if and only if the torsion T of  $\nabla$  vanishes on N,

$$(T(X,Y,))|_q = 0, \qquad X_q, Y_q \in T_q(N), \quad q \in N$$
 (5.18)

where X and Y are vector fields on N,  $X, Y \in \mathfrak{X}(N) \subseteq \mathfrak{X}(M)$ .

*Proof.* The result follows from Proposition I.5.4 and the observation that in the proof of Theorem 5.3, as well as in the one of Theorem 5.1, only the torsionless on N of the connection was used.

Now, we leave to the reader as exercises the next four problems:

**Exercise 5.4.** Show by explicit calculation that the normal frame  $\left\{\frac{\partial}{\partial z^i}\right\}$  induced by  $\{z^i\}$  (see (5.16)) is parallel on  $\overline{U}$ . (Hint: prove that  $\frac{\partial}{\partial z^i} = A_i^j \frac{\partial}{\partial x^j}$  on  $\overline{U}$ .) This agrees with Proposition I.5.6.

**Exercise 5.5.** Give a description of the Fermi coordinates on N in the framework of Theorem 5.3. (Hint: compare (5.5) with (5.16) and construct the submanifolds  $V_q$  using the chart  $(U_z, z)$  provided by (5.16).)

**Exercise 5.6.** Obtain the results of Section 4 from the ones of the present subsection. (Hint: for dim  $N = \dim M$ , we have  $p = p_0 = \pi(p)$ .)

**Exercise 5.7.** Find the explicit formula expressing the initial coordinates  $\{x^i\}$  via the normal coordinates  $\{z^i\}$  on N, i.e., invert equation (5.16) with respect to  $x^i$ . (Hint: use (5.16) or repeat the procedure of constructing  $\{z^i\}$  but starting with expansion of  $\{x^i\}$  with respect to  $\{z^i\}$  analogous to (5.13).)

Once again, we would like to notice that the coordinates normal on a submanifold N of M are local by their essence, they can be global, i.e., defined on a neighborhood of the whole N, only as an exception. But, generally, we can construct a family of overlapping coordinates normal on N such that the union of their domains forms a neighborhood of N in M. At the end, going rather ahead, we want to mention that the condition (5.1), ensuring the existence of frames normal on a submanifold N, has a quite natural interpretation in the theory of linear transports along paths (see Chapter IV, in particular Proposition IV.11.2 on page 282): it expresses the flatness on N of the parallel transport assigned to the linear connection under consideration.

# 6. Examples of normal frames and coordinates

Similarly to Section I.7, the present one contains different examples of frames and/or coordinates that are normal for some concrete connections on some sets. Their role is to illustrate the general theory presented in the preceding sections of this chapter.

Since the complete and explicit description of all coordinates normal at a point is expressible in a simple way via the coefficients of a symmetric linear connection (see Theorem 2.1), the reader can easily obtain (by means of (2.11')) as exercise this description for the Riemannian connections considered in Examples I.7.3–I.7.11 in which the coefficients of the corresponding connections are explicitly written.

Unfortunately, a 'more realistic' example is, the more difficult are the concrete calculations; besides, the problems of finding explicitly frames/coordinates normal on (sub)sets different from a single point cannot be solved completely in closed form (in elementary functions or in radicals); for instance, to find a frame or coordinate system (locally) normal along a path, one needs the solution of the initial-value problem (3.17) (for concrete Z) and the coordinates constructed in Lemma 3.1.

**Example 6.1 (Coordinates normal at a point in**  $\mathbb{S}^2$ ). Consider the 2-sphere  $\mathbb{S}^2$ , investigated in Example I.7.3, and a point  $p \in \mathbb{S}^2$ . Choose spherical coordinates  $(\theta, \varphi)$  on  $\mathbb{S}^2$  such that  $\theta(p) = \frac{\pi}{2}$  and  $\varphi(p) = 0$ ; so p will be in the great equatorial circle of  $\mathbb{S}^2$ . The coordinates  $(\theta, \varphi)$  are normal at p (as well as at any point in the equatorial circle) for the Riemannian  $\nabla$  according to Example I.7.3 in which  $\nabla$  is defined. Taking into account Remark 2.1, we can assert that all coordinate systems  $\{z^1, z^2\}$  on  $\mathbb{S}^2$ , which are normal at p, are given via the equations (see (2.13))

$$z^{1}(q) = \frac{\pi}{2} + a_{1}^{1} \Big[ \theta(q) - \frac{\pi}{2} \Big] + a_{2}^{1} \varphi(q) + \sum_{j,k,l=1,2} b_{jkl}^{1} [y^{j}(q) - y^{j}(p)] [y^{k}(q) - y^{k}(p)] [y^{l}(q) - y^{l}(p)] \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}}$$

$$z^{2}(q) = a_{1}^{2} \Big[ \theta(q) - \frac{\pi}{2} \Big] + a_{2}^{2} \varphi(q) + \sum_{j,k,l=1,2} b_{jkl}^{2} [y^{j}(q) - y^{j}(p)] [y^{k}(q) - y^{k}(p)] [y^{l}(q) - y^{l}(p)] \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}}$$

$$(6.1)$$

where the point q belongs to some open subset containing the point p.

**Example 6.2 (Frames/coordinates normal along a great circle on**  $\mathbb{S}^2$ ). Let us suppose that the path  $\gamma: [0, 2\pi) \to \mathbb{S}^2$  is a great circle on the 2-sphere  $\mathbb{S}^2$  (see Example I.7.3) and  $(\theta, \varphi)$  are spherical coordinates on  $\mathbb{S}^2$  such that  $\gamma$  lies on the equatorial plane, i.e.,  $\theta \circ \gamma = \frac{\pi}{2}$  and  $\varphi \circ \gamma = \operatorname{id}_{[0,2\pi)}$ . We know from Example I.7.3 that the coordinates system  $\{\theta, \varphi\}$  is normal along the whole path  $\gamma$  for the Riemannian connection  $\nabla$  describe in the example mentioned.

Due to Corollary 3.2 on page 103, all frames  $\{E'_1, E'_2\}$  normal for  $\nabla$  along  $\gamma$  are given by (see (3.29))

$$E_{1}' = B_{1}^{1} \frac{\partial}{\partial \varphi} + B_{1}^{2} \frac{\partial}{\partial \theta} + \sum_{j,k,l=1,2} B_{1kl}^{j} \Big\{ [y^{k} - y^{k}(p_{0})][y^{l} - y^{l}(p_{0})] \frac{\partial}{\partial y^{j}} \Big\} \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}}$$

$$E_{2}' = B_{2}^{1} \frac{\partial}{\partial \varphi} + B_{2}^{2} \frac{\partial}{\partial \theta} + \sum_{j,k,l=1,2} B_{2kl}^{j} \Big\{ [y^{k} - y^{k}(p_{0})][y^{l} - y^{l}(p_{0})] \frac{\partial}{\partial y^{j}} \Big\} \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}},$$

$$(6.2)$$

where  $B_j^i$  are real constants,  $B_{jkl}^i$  are  $C^1$  functions that together with their partial derivatives are bounded, and  $p_0 \in \gamma([0, 2\pi))$  is fixed and hence  $\theta(p_0) = \frac{\pi}{2}$  and  $\varphi(p_0)$  is a fixed number in  $[0, 2\pi)$ . By the same corollary, all coordinate systems  $\{z^1, z^2\}$  on  $\mathbb{S}^2$ , which are normal for  $\nabla$  along  $\gamma$ , have the representation (see (3.30))

$$z^{1} = a^{1} - s_{0} - \varphi(p_{0}) + 2\varphi + \sum_{j,k,l=1,2} a_{jkl}^{1} \left\{ [y^{j} - y^{j}(p_{0})][y^{k} - y^{k}(p_{0})][y^{l} - y^{l}(p_{0})] \right\} \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}} z^{2} = a^{2} - \theta(p_{0}) + \theta + \sum_{j,k,l=1,2} a_{jkl}^{2} \left\{ [y^{j} - y^{j}(p_{0})][y^{k} - y^{k}(p_{0})][y^{l} - y^{l}(p_{0})] \right\} \Big|_{\substack{y^{1} = \theta \\ y^{2} = \varphi}},$$
(6.3)

where  $a^1$  and  $a^2$  are real constants and  $a^i_{jkl}$  are  $C^1$  functions that together with their partial derivatives are bounded.

**Example 6.3 (Weyl connections: general considerations).** Recall (see Footnote I.5 on page 35), if M is a Riemannian manifold with metric g and  $\nabla$  is a Weyl connection on M characterized by a 1-form  $\omega$ , the coefficients of  $\nabla$  in an arbitrary (local) frame  $\{E_i\}$  are (see (I.4.7))

$$\Gamma^{i}_{\ jk} = \begin{cases} i \\ jk \end{cases} + \frac{1}{2} \left( g^{im} C^{l}_{mj} g_{lk} + g^{im} C^{l}_{mk} g_{lj} - C^{i}_{jk} \right) + \frac{1}{2} (g_{jk} g^{il} \omega_{l} - \delta^{i}_{j} \omega_{k} - \delta^{i}_{k} \omega_{j}) \\
= {}^{\mathrm{R}} \Gamma^{i}_{\ jk} + W^{i}_{jk},$$
(6.4)

where the Christoffel symbols  $\begin{cases} i\\ ik \end{cases}$  are given by (I.4.14), the functions

$${}^{\mathrm{R}}\Gamma^{i}_{\ jk} = \begin{cases} i\\ jk \end{cases} + \frac{1}{2} \left( g^{im} C^{l}_{mj} g_{lk} + g^{im} C^{l}_{mk} g_{lj} - C^{i}_{jk} \right)$$
(6.5)

are the coefficients in  $\{E_i\}$  of the Riemannian connection  $\nabla^{\mathbf{R}}$  induced by g, and the functions

$$W_{jk}^{i} = W_{kj}^{i} = \frac{1}{2} (g_{jk}g^{il}\omega_l - \delta_j^{i}\omega_k - \delta_k^{i}\omega_j)$$

$$(6.6)$$

are the components in  $\{E_i\}$  of a tensor field W of type (1, 2), which field characterizes the concrete Weyl connection relative to the Riemannian one with coefficients (6.5).

Suppose the components  $g_{ij}$  of g in  $\{E_i\}$  form a diagonal matrix

$$[g_{ij}] = \operatorname{diag}(g_1, \dots, g_{\dim M}). \tag{6.7}$$

Then the Christoffel symbols can be calculated via equation (I.4.15) and the components (6.6) of the tensor W reduce to (do not sum over i!)

$$W_{jk}^{i} = \frac{1}{2} \left( \delta_{jk} \frac{g_k}{g_i} \omega_i - \delta_j^i \omega_k - \delta_k^i \omega_j \right)$$
(6.8)

as  $[g^{ij}] = [g_{ij}]^{-1} = \text{diag}(1/g_1, \ldots, 1/g_{\dim M})$ . An elementary calculations shows that the non-vanishing components of W are

$$W_{ik}^{i} = W_{ki}^{i} = -\frac{1}{2}\omega_{k} \quad \text{for all } i, k = 1, \dots, \dim M$$
  
$$W_{kk}^{i}|_{k \neq i} = \frac{1}{2}\frac{g_{k}}{g_{i}}\omega_{i}.$$
(6.9)

Applying the identity (do not sum over i, j and k!)

$$W_{jk}^{i} \equiv W_{jk}^{i}\delta_{ij} + W_{jk}^{i}(1 - \delta_{ij})\delta_{jk} + W_{jk}^{i}(1 - \delta_{ij})(1 - \delta_{jk})$$

and (6.9), we get

$$W_k := [W_{jk}^i] = -\frac{1}{2}\omega_k \mathbb{1} + \frac{1}{2}g_k \Big[\frac{1}{g_i}\omega_i(1-\delta_{ij})\delta_{jk}\Big] - \frac{1}{2}[(1-\delta_{ij})(1-\delta_{jk})\delta_{ik}\omega_j].$$
(6.10)

In particular, for  $\dim M = 2$ , the last equation reduces to

$$W_1 = [W_{j1}^i] = \frac{1}{2} \begin{pmatrix} -\omega_1 & -\omega_2 \\ \frac{g_1}{g_2}\omega_2 & -\omega_1 \end{pmatrix} \qquad W_2 = [W_{j2}^i] = \frac{1}{2} \begin{pmatrix} -\omega_2 & \frac{g_2}{g_1}\omega_1 \\ -\omega_1 & -\omega_2 \end{pmatrix} .$$
(6.11)

Below in this and in the next two examples, we shall consider some problems concerning frames normal for Weyl connections.

First of all, we notice that, if a frame  $\{E_i\}$  on  $V \subseteq M$  is normal on  $U \subseteq V$  for the Riemannian connection  $\nabla^{\mathbb{R}}$  (with coefficients (6.5)), then it is normal for the Weyl connection  $\nabla$  on the set

$$U^W := \{ p \in U : \omega_i |_p = 0 \text{ for } i = 1, \dots, \dim M \} = \{ p \in U : \omega |_p = 0 \}$$
(6.12)

which may be empty. This is a consequence from (6.9) and

$$\Gamma^i_{\ jk}|_U = W^i_{jk}|_U \tag{6.13}$$

due to (6.4), (6.5) and  ${}^{\mathrm{R}}\Gamma^{i}_{jk}|_{U} = 0$  as the frame  $\{E_i\}$  is normal for  $\nabla^{\mathrm{R}}$  on U. As a result of the simplification (6.13), a frame like  $\{E_i\}$  can be used as an initial one from which frames normal for  $\nabla$  on subset(s) U may be constructed.

**Example 6.4 (Weyl connections on**  $\mathbb{S}^2$ ). Let  $\nabla^{\mathbb{R}}$  be the Riemannian connection on the 2-sphere  $\mathbb{S}^2$  investigated in Example I.7.3. Its coefficients in spherical coordinates  $(x^1 = \theta, x^2 = \varphi)$  are given by (I.7.11). The coordinate system  $\{x^1, x^2\}$  is normal along the equatorial circle  $\gamma: [0, 2\pi) \to \mathbb{S}^2$  with  $x^1 \circ \gamma = \frac{\pi}{2}$  and  $x^2 \circ \gamma = \operatorname{id}_{[0, 2\pi)}$ .

Suppose  $\nabla$  is a Weyl connection on  $\mathbb{S}^2$  whose Riemannian part is  $\nabla^{\mathbf{R}}$ . Then, due to (6.13), we have (i, j = 1, 2)

$$\Gamma^{i}_{\ jk} \circ \gamma = W^{i}_{jk} \circ \gamma \tag{6.14}$$

with  $W_k = [W_{jk}^i]$  given by (6.11). Thus, if there exist  $p \in \gamma([0, 2\pi))$  such that  $\omega|_p = 0$ , then  $\{x^1, x^2\}$  is normal for  $\nabla$  at p. From here immediately follows that, if  $\omega|_p = 0$  for some  $p \in \mathbb{S}^2$ , then one can construct spherical coordinates (with p in their equatorial plane) which are normal at p for  $\nabla$ .

In the general case, frames/coordinates normal for  $\nabla$  at an arbitrarily fixed point  $p \in \mathbb{S}^2$  can be constructed as follows. Take spherical coordinates  $(x^1 = \theta, x^2 = \varphi)$  on  $\mathbb{S}^2$  such that p is in their equatorial plane, e.g., we can choose them in such a way that  $\theta(p) = \frac{\pi}{2}$  and  $\varphi(p) = 0$ . Then  $\{x^1, x^2\}$  is normal for  $\nabla^{\mathbb{R}}$  along the equatorial circle and, in particular, at p.

According to Theorem 2.3, all frames normal at p for  $\nabla$  (but not for  $\nabla^{\mathbf{R}}$ ) are  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  with  $A = [A_j^i]$  given by (2.14) in which one should put  $\Gamma_1(p) = W_1(p)$  and  $\Gamma_2(p) = W_2(p)$  (see (6.13)), where  $W_1$  and  $W_2$  are given by (6.11) with  $\omega_1 = \omega(\frac{\partial}{\partial x^1})$  and  $\omega_2 = \omega(\frac{\partial}{\partial x^2})$ , viz.

$$A(q) = \left\{ \mathbb{1} - \frac{1}{2} [x^1(q) - x^1(p)] \begin{pmatrix} -\omega_1 & -\omega_2 \\ \frac{g_1}{g_2} \omega_2 & -\omega_1 \end{pmatrix} - \frac{1}{2} [x^2(q) - x^2(p)] \begin{pmatrix} -\omega_2 & \frac{g_2}{g_1} \omega_1 \\ -\omega_1 & -\omega_2 \end{pmatrix} \right\} A_0 + A_{jk}(q) [x^j(q) - x^j(p)] [x^k(q) - x^k(p)].$$
(6.15)

Similarly, to obtain all coordinates normal at p, one has to substitute the equality  $\Gamma^i_{(jk)}(p) = W^i_{jk}(p)$ , with  $W^i_{jk}$  given by (6.11), in (2.11'). The reader may wish to write as an exercise the explicit form of the so-obtained coordinates normal at p.

**Example 6.5 (Weyl connection on Minkowski spacetime).** Let now  $\nabla$  be a Weyl connection on the Minkowski spacetime  $\mathbb{R}^4_3$  and  $\gamma: J \to \mathbb{R}^4_3$  be a straight line; e.g., it may be a trajectory of a photon,  $\dot{\gamma} = 0$ , and hence lying on the light cone (see Example I.7.12). We shall look for a coordinate system normal for  $\nabla$  along  $\gamma$ .

From the standard coordinate system  $\{u^i : i = 1, ..., 4\}$ , we can obtain via a pseudo-orthogonal transformation, which preserves the components of the metric  $e_3^4$  of  $\mathbb{R}_3^4$ , a new coordinates system  $\{x^i\}$  such that  $x^1 \circ \gamma = \mathrm{id}_{\mathbb{R}}$ , i.e., the  $x^1$ -axis to coincide with the line  $\gamma$ .<sup>1</sup> The coordinate system  $\{x^i\}$  is of the type constructed in Lemma 3.1; it corresponds to  $U_1 = \mathbb{R}_3^4$  and  $\mathbf{t}_0 = (0, 0, 0)$ .

<sup>&</sup>lt;sup>1</sup>Here we shall not discuss the physical possibilities for realization of the coordinate system  $\{x^i\}$ .
Since in  $\{x^i\}$  the components of  $e_3^4$  form the matrix diag(1, -1, -1, -1), in it the coefficients of  $\nabla$  are

$$\Gamma^{i}_{\ jk} = W^{i}_{jk}$$

with  $W_{jk}^i$  given by (6.9) (or (6.10)). Consequently, by Theorem 3.1 (resp. 3.3), all frames  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  (resp. coordinate systems  $\{z^i\}$ ) on  $\mathbb{R}_3^4$  normal along the (whole) path  $\gamma$  are given via (3.23) (resp. (3.27)–(3.28)) in which the matrix  $\Gamma_1$ has the form (see (6.10))

$$\Gamma_1 = W_1 := [W_{j1}^i] = -\frac{1}{2}\omega_1 \mathbb{1} + \frac{1}{2}g_1 \Big[\frac{1}{g_i}\omega_i(1-\delta_{ij})\delta_{j1}\Big] - \frac{1}{2}[(1-\delta_{ij})(1-\delta_{j1})\delta_{i1}\omega_j].$$

The matrix-valued function  $Y(s, s_0; -\Gamma_1 \circ \gamma)$  is now the solution of the initial-value problem  $(s \in \mathbb{R})$ 

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = -(W_1 \circ \gamma)Y \qquad Y|_{s=s_0} = \mathbb{1}.$$

Unfortunately, an explicit and finite expression for Y can be obtained only in some exceptional cases. For instance, if  $\omega_i = \omega(\frac{\partial}{\partial x^i})$  are constant functions, then  $W_1$  is a constant matrix. This immediately implies

$$Y(s, s_0; -\Gamma_1 \circ \gamma) = e^{-(s-s_0)W_1 \circ \gamma}.$$

**Exercise 6.1.** Generalize the above results for  $\mathbb{R}^n_q$ ,  $0 \le q \le n \in \mathbb{N}$ , and when  $\gamma$  is a straight line in  $\mathbb{R}^n_q$ .

**Example 6.6 (Flat Riemannian connections).** Example I.7.1 will be re-considered below from the view-point of the results of Section 4.

Let M be a  $C^3$  Riemannian manifold with  $C^2$  metric g and  $\nabla$  be the Riemannian connection generated by g. Suppose  $\nabla$  is flat on an open set  $U \subseteq M$ . By Theorem 4.3, there exist normal coordinates  $x^i$  with domain  $V \subseteq U$  for  $\nabla$ . The coefficients of  $\nabla$  in these coordinates are  $\Gamma^i_{jk} = {i \\ ji} = 0$ , which implies  $g_{ij,k} = 0$  in them (see (I.4.16)). Consequently, the components  $g_{ij}$  of g in  $\{x^i\}$  are constant on V. In this way we have inverted the implication/conclusions of Example I.7.1.

The coordinates  $x^i$  can be calculated by means of (4.14), but the finding of their explicit form may turn to be a difficult problem. However, if one finds a coordinate system  $\{x^i\}$  on  $V \subseteq U$  in which the metric components are constant, then it will be also normal for  $\nabla$  on V (see (I.4.16)). For instance, such are the standard coordinates on the Euclidean space  $\mathbb{E}^n$  and on the pseudo-Euclidean space  $\mathbb{R}^n_q$  (see Example I.7.1 and (I.7.16)). In particular, the standard (Cartesian) coordinates on the Minkowski spacetime  $M_4 = \mathbb{R}^4_3$  (or  $\mathbb{R}^4_1$ ) are normal on  $M_4$  for the flat Riemannian connection on it induced by the metric  $e_3^4$ .

**Example 6.7 (One-dimensional real manifolds).** There are at lest two reasons for the global (local) existence of normal frames (coordinates) on 1-dimensional  $C^3$  real manifolds endowed with a linear connection  $\nabla$ , viz.:

#### 6. Examples of normal frames and coordinates

- (i) The curvature and torsion tensors identically vanish in the 1-dimensional case (see (I.3.13) and (I.3.14) with all indices taking the sole value 1). We can rephrase this by saying that any linear connection on 1-dimensional manifold is flat and torsionless. Consequently Theorems 4.1 and 4.3 imply the above result. Respectively, Theorem 4.2 and Theorem 4.3 give a complete description of all normal frames and coordinates.
- (ii) A chart (U, x) on 1-dimensional manifold M consists of an open set  $U \subseteq M$ and homomorphism  $x: U \to J$  with J being an open real interval. Thus  $x^{-1}: J \to M$  is a path for which the results of Section 3 are applicable. In particular, Theorems 3.1 and 3.2 (resp. Theorem 3.3) describe(s) all frames (resp. coordinates) normal along  $x^{-1}$ . Besides, since M can be represented as a union of curves like  $x^{-1}(J)$ , frames globally normal on M for  $\nabla$  can be constructed in a way similar to the one described in the proof of Theorem 3.2.

Let us now write some explicit formulae regarding coordinates/frames normal for a  $C^1$  connection  $\nabla$  on a  $C^3$  1-dimensional real manifold M. Let (U, x) be a chart of M with  $x \colon U \to J \subseteq \mathbb{R}$  and  $\{x^1\}$  be the corresponding coordinate system consisting of the sole coordinate function  $x^1 = u^1 \circ x = \mathrm{id}_{\mathbb{R}} \circ x = x$ . The connection  $\nabla$  has in  $\{x^1\}$  a sole coefficient  $\Gamma^1_{11}$  coinciding with the matrix  $\Gamma_1 = [\Gamma^1_{11}] = \Gamma^1_{11}$ .

To construct coordinates/frames normal for  $\nabla$ , we need, according to the scheme describe in Section 4, the solution  $Y(p, p_0; -\Gamma_1)$ , for fixed  $p_0 \in U$  and any  $p \in U$ , of the initial-value problem

$$\frac{\mathrm{d}Y}{\mathrm{d}x^1} = -\Gamma_1 Y \qquad Y_{p=p_0} = \mathbb{1}_1 = 1.$$

This solution is

$$Y(p, p_0; -\Gamma_1) = \exp\left(-\int_{p_0}^p \Gamma_1(q) \,\mathrm{d}q\right) = \exp\left(-\int_{s_0}^s \Gamma_1(\gamma(s))\dot{\gamma}^1(s) \,\mathrm{d}s\right) \tag{6.16}$$

where  $\gamma: J' \to U$  is any path in U such that  $\gamma(s_0) = p_0$  and  $\gamma(s) = p$  for some  $s_0, s \in J$ ; in particular, one can set  $\gamma = x^{-1}$  and J' = J.

According to Theorem 4.3, all coordinate systems  $\{x'^1\}$  on U which are normal for  $\nabla$  on U are such that

$$x'^{1}(p) = a^{1} + \frac{1}{B} \int_{p_{0}}^{p} Y^{-1}(q, p_{0}; -\Gamma_{1}) dq^{1}$$
  
$$= a^{1} + \frac{1}{B} \int_{p_{0}}^{p} \exp\left(\int_{p_{0}}^{q} \Gamma_{1}(r) dr\right) dq \qquad (6.17)$$
  
$$= a^{1} + \frac{1}{B} \int_{s_{0}}^{s} \exp\left(\int_{s_{0}}^{t} \Gamma_{1}(\gamma(u))\dot{\gamma}^{1}(u) du\right) \dot{\gamma}^{1}(t) dt$$

where  $a^i$  are constant numbers (representing the coordinates of  $p_0$  in  $\{x^1\}$ ) and B is a non-zero constant.

**Exercise 6.2.** If  $\nabla$  is a Riemannian connection, derive the results of Example I.7.2 from (6.17). (Hint: use equation (I.7.5).)

Similarly, by Theorem 4.2, all frames  $\{E'_1\}$  normal on U for  $\nabla$  are such that

$$E_{1}'|_{p} = B \exp\left(\int_{p_{0}}^{p} \Gamma_{1}(q) \,\mathrm{d}q\right) \frac{\partial}{\partial x^{1}}\Big|_{p}$$

$$= B \exp\left(\int_{s_{0}}^{s} \Gamma_{1}(\gamma(t))\dot{\gamma}^{1}(t) \,\mathrm{d}t\right) \frac{\partial}{\partial x^{1}}\Big|_{p}.$$
(6.18)

Comparing (6.18) and (6.17), we get

$$E_1' = \frac{\partial}{\partial x'^1} \tag{6.19}$$

which means that any normal frame is holonomic as it should be in 1-dimensional manifolds.<sup>2</sup>

**Exercise 6.3.** Derive equations (6.17)–(6.19) by applying the results of subsection 3.2. Hint: use Theorems 3.3 and 3.1.

**Example 6.8 (Frames/coordinates normal along geodesic in 2-manifold).** This example demonstrates that frames/coordinates normal along geodesic paths in a 2-dimensional manifold can always be found explicitly via quadratures.

Suppose M is a  $C^3$  2-dimensional manifold,  $\nabla$  is a  $C^0$  connection on M and  $\gamma: J \to M$  is a geodesic. Due to Remark 3.7 on page 95, the equations  $\Gamma^1_{11} \circ \gamma = 0$  and  $\Gamma^2_{11} \circ \gamma = 0$  hold in the coordinates  $\{x^1, x^2\}$  provided by Lemma 3.1. Therefore the matrix  $\Gamma_1$  along  $\gamma$  is

$$\Gamma_1 \circ \gamma = [\Gamma^i_{\ j1} \circ \gamma]_{i,j=1,2} = \begin{pmatrix} 0 & \Gamma^1_{\ 21} \circ \gamma \\ 0 & \Gamma^2_{\ 21} \circ \gamma \end{pmatrix} .$$
(6.20)

The matrix-valued function  $Y(s, s_0; -\Gamma_1 \circ \gamma)$ , entering into the main results of Subsection 3.2, is the solution of the matrix initial-value problem

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = -(\Gamma_1 \circ \gamma(s))Y \qquad Y_{s=s_0} = \mathbb{1}$$

<sup>&</sup>lt;sup>2</sup>It is an elementary exercise to be proved that all  $C^0$  frames on  $C^1$  1-dimensional manifold are (locally) holonomic.

which in an expanded form reads (see (6.20);  $Y = [Y_{ij}]$ ):

$$\frac{\mathrm{d}Y_{11}(s)}{\mathrm{d}s} = -\Gamma^{1}_{21}(\gamma(s))Y_{21} \qquad Y_{11}|_{s=s_{0}} = 1$$
$$\frac{\mathrm{d}Y_{12}(s)}{\mathrm{d}s} = -\Gamma^{1}_{21}(\gamma(s))Y_{22} \qquad Y_{12}|_{s=s_{0}} = 0$$
$$\frac{\mathrm{d}Y_{21}(s)}{\mathrm{d}s} = -\Gamma^{2}_{21}(\gamma(s))Y_{21} \qquad Y_{21}|_{s=s_{0}} = 0$$
$$\frac{\mathrm{d}Y_{22}(s)}{\mathrm{d}s} = -\Gamma^{2}_{21}(\gamma(s))Y_{22} \qquad Y_{22}|_{s=s_{0}} = 1.$$

The solutions of the last two initial-value problems are

$$Y_{21} = 0$$
  $Y_{22} = \exp\left(-\int_{s_0}^{s} \Gamma_{21}^2(\gamma(t)) dt\right)$ .

Inserting these functions into the previous two initial-value problems, we get

$$Y_{11} = 1$$
  $Y_{12} = 0$ 

and consequently

$$Y(s, s_0; -\Gamma_1 \circ \gamma) = \operatorname{diag}\left(1, \exp\left(-\int_{s_0}^s \Gamma_{21}^2(\gamma(t)) \,\mathrm{d}t\right)\right).$$
(6.21)

According to Theorem 3.1, all frames  $\{E'_i\}$  normal for  $\nabla$  along the geodesic path  $\gamma$  are such that  $E'_i(p) = A^j_i(p) \frac{\partial}{\partial x^j}\Big|_p$ , where  $p \in U_1$  and the matrix-valued function  $A = [A^j_i]$  is given by (3.23) and (6.21), i.e.,

$$A(p) = \left\{ \mathbb{1} - \sum_{k=1,2} \Gamma_k(p_0) [x^k(p) - x^k(p_0)] \right\} \operatorname{diag} \left( 1, \exp\left( - \int_{s_0}^{x^1(p)} \Gamma^2_{21}(\gamma(t)) \, \mathrm{d}t \right) \right) B + \sum_{k,l=1,2} B_{kl}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)] \quad (6.22)$$

with  $\Gamma_1(p_0)$  given via (6.20) (as  $p_0 = \gamma(s)$ ).

Similarly, by Theorem 3.3, all coordinate systems  $\{z^I : i = 1, 2\}$  normal for a symmetric connection  $\nabla$  on some subset  $U_z \subseteq U_1$  are given via (3.27), (3.28), (6.20) and (6.21), that is we have the explicit expression (i = 1, 2).

$$z^{i}(p) = a^{i} + (B^{-1})_{1}^{i} [x^{1}(p) - s_{0}]$$

$$+ (B^{-1})_{1}^{i} [x^{1}(p) - x^{1}(p_{0})] + (B^{-1})_{2}^{i} \exp\left(-\int_{s_{0}}^{x^{1}(p)} \Gamma^{2}_{21}(\gamma(t)) dt\right) [x^{2}(p) - x^{2}(p_{0})]$$

$$+ (B^{-1})_{1}^{i} [x^{2}(p) - x^{2}(p_{0})] \sum_{j=1,2} \Gamma^{1}_{j2}(\gamma(x^{1}(p))) [x^{j}(p) - x^{j}(p_{0})]$$

$$+ (B^{-1})_{2}^{i} [x^{2}(p) - x^{2}(p_{0})] \exp\left(-\int_{s_{0}}^{x^{1}(p)} \Gamma^{2}_{21}(\gamma(t)) dt\right)$$

$$\times \sum_{j=1,2} \Gamma^{2}_{j2}(\gamma(x^{1}(p))) [x^{j}(p) - x^{j}(p_{0})]$$

$$+ \sum_{j,k,l=1,2} a^{i}_{jkl}(p) [x^{j}(p) - x^{j}(p_{0})] [x^{k}(p) - x^{k}(p_{0})] [x^{l}(p) - x^{l}(p_{0})]. \quad (6.23)$$

The formula (6.22) (resp. (6.23)) gives a complete local description of all frames (resp. coordinates) normal along a geodesic path  $\gamma$  for arbitrary (resp. symmetric) linear connection  $\nabla$  on a 2-dimensional manifold. If one needs frames globally normal along  $\gamma$ , the scheme proposed in the proof of Theorem 3.2 can be applied for their construction and/or complete description.

**Example 6.9 (Open Einstein-de Sitter Universe).** The (open) Einstein-de Sitter Universe is a concrete type of non-static homogeneous cosmological model (see [58] and Example I.7.11). Its geometrical base is the Einstein-de Sitter manifold which is a 4-dimensional Riemannian manifold with Riemannian metric g which in a suitable coordinate system  $\{x^1 = x, x^2 = y, x^3 = z, x^4 = ct\}$  has components  $g_{ij}$ ,  $i, j = 1, \ldots, 4$ , forming the diagonal matrix [58, § 164]

$$[g_{ij}] = \operatorname{diag}\left(-\mathrm{e}^{f(x^4)}, -\mathrm{e}^{f(x^4)}, -\mathrm{e}^{f(x^4)}, 1\right)$$
(6.24)

for some  $C^1$  function f; the range of  $x^i$  is  $\mathbb{R}$ . The metric g induces a Riemannian connection  $\nabla$  whose non-zero coefficients in  $\{x^i\}$  are (see (I.4.13) and (I.4.15))

$$\Gamma^{i}_{i4} = \Gamma^{i}_{4i} = \frac{1}{2}f'(x^{4}) \quad \text{for } i = 1, 2, 3$$

$$\Gamma^{4}_{kk} = \frac{1}{2}f'(x^{4})e^{f(x^{4})} \quad \text{for } k = 1, 2, 3$$
(6.25)

where  $f'(x^4) = \frac{\mathrm{d}f(x^4)}{\mathrm{d}x^4}$ .

#### 7. Conclusion

These equations yield the following coefficients' matrices

$$\Gamma_{k} = [\Gamma_{jk}^{i}] = \frac{1}{2}f'(x^{4}) \begin{pmatrix} 0 & 0 & 0 & \delta_{1k} \\ 0 & 0 & 0 & \delta_{2k} \\ 0 & 0 & 0 & \delta_{3k} \\ \delta_{1k}e^{f(x^{4})} & \delta_{2k}e^{f(x^{4})} & \delta_{3k}e^{f(x^{4})} & 0 \end{pmatrix} \quad \text{for } k = 1, 2, 3$$
$$\Gamma_{4} = [\Gamma_{j4}^{i}] = \frac{1}{2}f'(x^{4}) \operatorname{diag}(1, 1, 1, 0). \tag{6.26}$$

Consequently, the coordinate system  $\{x^i\}$  is normal on the set  $\{p : f'(x^4(p)) = 0\}$ , which may be empty if f' is nowhere vanishing function. However, if, for instance, there is a point  $p_0$  such that  $f'(x^4(p_0)) = 0$ , then  $\{x^i\}$  is normal on the 3-dimensional submanifold  $\{p : x^4(p) = x^4(p_0)\}$ .

Due to Theorem 2.1, all coordinates  $y^i$  normal for  $\nabla$  at an arbitrarily chosen point p are given by (2.11'). So, substituting (6.25) into (2.11'), we get their explicit form as:

$$y^{i}(q) = y^{i}(p) + b^{i}_{j}[x^{j}(q) - x^{j}(p)] + \frac{1}{2}f'(x^{4})[x^{4}(q) - x^{4}(p)] \sum_{l=1}^{3} b^{i}_{l}[x^{l}(q) - x^{l}(p)] + \frac{1}{2}f'(x^{4})e^{f(x^{4})}b^{i}_{4}\sum_{l=1}^{3}[x^{l}(q) - x^{l}(p)]^{2} + b^{i}_{jkl}(q)[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)][x^{l}(q) - x^{l}(p)]. \quad (6.27)$$

Similarly, all frames normal at a point p are described via Theorem 2.3. Thus, to obtain an explicit expression for them, one simply has to substitute (6.26) into (2.14).

**Exercise 6.4.** Investigate frames/coordinates normal along a  $C^1$  path in the Einstein-de Sitter spacetime. For the purpose, the coefficients of  $\nabla$  should be calculated in the coordinates provided by Lemma 3.1 (see (I.3.6), (3.12) and (6.25)) and then Theorems 3.1 and 3.3 should be applied.

## 7. Conclusion

The main subject of this chapter was the existence and (non-)uniqueness of normal frames and coordinates for linear connections on differentiable manifolds. The existing literature on the problem was reviewed (in modern notation). It deals only with normal coordinates which exist only for torsionless connections. The few works investigating the asymmetric case do not add nothing new as they treat the symmetric part of the connection, thus transferring the problem to the exploration of some other torsion free linear connection.

The major classical results concerning the normal coordinates for linear connections are summarized in the Table 7.1 on the next page.

Year	Person	Result and original reference
1854	B. Riemann	Existence and construction of ('Riemannian') coordinates in a Riemannian manifold which are normal at a single point. [48]
1922	O. Veblen	Existence and construction of ('Riemannian normal') coordi- nates in a manifold with torsionless linear connection which are normal at a single point. [72]
1922	E. Fermi	Existence of ('Fermi') coordinates in a Riemannian manifold which are normal along a path without self-intersections. [52]
1926	T. Levi-Civita	Explicit transformation to the Fermi coordinates along paths without self-intersections. [81]
1927	L.P. Eisenhart	Existence and construction of particular kind of ('Fermi') coordinates on a manifold with torsionless linear connection which are normal along a path without self-intersections. [53]
1958	L. O'Raifeartaigh	Necessary and sufficient conditions for existence of coordi- nates normal on submanifold of a manifold with torsionless linear connection. If such coordinates exist, a particular ex- ample of them ('Fermi coordinates') is constructed. [55]

Table 7.1: Main contributions in the theory of normal coordinates for torsionless linear connections.

Besides the detailed review of these results, we have presented a number of their generalizations and wide discussion of related topics and methods, most of which seem to be new ones and appearing in the present book for the first time.<sup>1</sup> They include mainly: (i) Description of *all* frames normal for symmetric connections; (ii) Complete description of *all* coordinates normal for torsionless linear connections (cases at a given point, along paths, and on submanifolds); (iii) Existence and complete constructive description of *all* frames normal at a single point, along (locally injective) paths, and on submanifolds for arbitrary linear connections, with or without torsion. Moreover, with respect to the references cited, we have made a number of improvements of the existing proofs of known results, a lot of facts are formulated more precisely, and some assertions are (partially) generalized. The reader interested in what exactly is new in Subsections 2.1, 3.1, and 5.1 and in Section 4 may wish to compare them with the references given at their beginnings and the ones of Sections 2–5. Subsections 2.2, 3.2, and 5.2, entitled 'Complete description' contain original material.

Here we want to emphasize on the following facts. First, normal coordinates (may) exist only for torsionless linear connections while normal frames (may) exist for arbitrary connections; the normal frames for connections with non-zero (resp.

<sup>&</sup>lt;sup>1</sup>Preliminary (implicit) versions of some of these results are contained in the author's papers [80], [76, Section 6], and [83, Section 5].

zero) torsion are with necessity anholonomic (resp. holonomic). Second, at a single point or along a path normal coordinates (resp. frames) always exist for torsionless (resp. arbitrary) linear connections; on submanifolds of dimension greater than one they exist if and only if the parallel transport on it along paths lying entirely in it is path-independent; in particular on the whole manifold normal coordinates/frames exist iff it is flat. Third, the normal coordinates, if any, are essentially local; globally they exist only as an exception. If local normal frames exist, then global normal frames always exist but they are smooth generally only locally.

Looking over the Riemannian coordinates, Riemannian normal coordinates, and Fermi coordinates (along a path or on submanifold, if any in the last case), we see that they are constructed in a uniform common way by employing the (local) existence of geodesics. From this point of view, the above coordinates are realization of one special kind of normal coordinates on submanifolds of zero, one, or higher dimension respectively.

## Chapter III

# Normal Frames and Coordinates for Derivations on Differentiable Manifolds

The existence, uniqueness, and construction of frames and coordinates normal for derivations (along vector fields, fixed vector field, paths, and fixed path) of the tensor algebra over a manifold are explored in details. For arbitrary vector fields or paths, normal frames (resp. coordinates) exist always (resp. if the on other submantorsion vanishes); along more general mapifolds  $\mathbf{or}$ pings necessary and sufficient conditions for such existence derived. For are fixed vector field derivations along path normal frames and coordior With a few exnates exist always. ceptions, a complete constructive description of the normal frames and coordinates, if any, is presented. Frames simultaneously normal for two derivations are studied. With respect to the normal frames, the unique role of the linear connections amongst the other derivations is pointed out.

## 1. Introduction

The aim of this chapter is the investigation of frames and coordinates normal for different kinds of derivations of the tensor algebra over a differentiable manifold. Since the linear connections are a particular example of such derivations, the presented here material is a direct continuation and generalization of the one in Chapter II. But, as we shall see, a number of problems concerning normal frames and charts for general derivations are 'locally' reduced to the same problems for linear connections and, consequently, their (local) solutions could be found, in more or less ready form, in Chapter II.

Some of the results in the present chapter are partially based on the ones in the series of works [76,80,83–87] and are completely revised and generalized their versions. But most of the material is new and original.

Section 2 has an introductory character. The concepts of derivations and derivations along vector fields of the tensor algebra over a manifold are introduced. Their components, coefficients (if they are linear), curvature, and torsion are defined. Next, in Section 3, the normal frames and coordinates are defined as ones in which the components of a derivation along vector fields vanish (on some set). The equations describing the transition to normal frames or coordinates are derived and the linearity of a derivation along vector fields is pointed as a necessary conditions for their existence.

In Section 4 (resp. Section 5) is proved that at a single point (resp. along a (locally injective) path) frames normal for a linear at it (resp. along it) derivation along vector fields always exist and their complete descriptions are given. Besides, if the derivation is torsionless, all normal coordinates are found. In Sections 6–8, the problems of existence, uniqueness, and complete description of frames and local charts (or coordinates) on neighborhoods, on submanifolds, and along (injective or locally injective) mappings, respectively, for derivations along vector fields are studied in details and solved.

To the problems concerning frames or coordinates normal for derivations along fixed vector field is devoted Section 9. The existence of normal frames and coordinates in this case is proved. A complete description of the frames normal at a single point, along a path, and on the whole manifold are presented. The local charts (or coordinates) normal at a point are completely described. Along a path the explicit system of differential equations, which the normal coordinates must satisfy and which always have (local) solutions, is derived. A method for obtaining the coordinates (locally) normal on the whole manifold is pointed in the  $C^{\infty}$  case.

Normal frames for derivations along paths are investigated in Section 10. After the introduction of the basic definitions and notation, it is proved that frames normal for a derivation along a given (fixed) path always exist and their general form is found. A (local) holonomic extension of such frames, as well as of any other frame defined only along a path, is constructed. For derivations along arbitrary paths is proved that they admit normal frames iff they are covariant derivatives along paths induced by linear connections for which normal frames exist. Since the normal frames for the derivations and connections turn to be identical, all problems for these frames are transferred to similar ones considered in Chapter II.

Section 11 deals with problems connected with frames simultaneously normal for two derivations along arbitrary/fixed vector field or path. Necessary and sufficient conditions for the existence of such frames are found. In particular, in the case of arbitrary vector field or path, they exist iff the two derivations coincide. Normal frames for mixed linear connections are explored. It is shown that this range of problems is completely and equivalently reduced to similar one for two, possibly identical, linear connections, the contra- and co-variant 'parts' of the initial mixed connection.

In Section 12 are collected and commented some results concerning linear connections obtained in the preceding sections of this chapter.

Section 13 illustrates the theory of the preceding sections with concrete examples.

Section 14 contains a discussion of some terminological problems linked to the normal frames or coordinates.

The chapter ends with certain general remarks in Section 15.

## 2. Derivations of the tensor algebra over a manifold

The idea of a derivation of the algebra T(M) of tensor fields over a manifold Mis in  $T^{1}(M)$  to be introduced an operator analogous to the (ordinary or partial) derivative of scalar functions and 'compatible' with the tensor structure of  $T^{1}(M)$ . The covariant derivative (along vector fields), introduced via Definition I.3.1 is, as we shall see below, a particular example of a derivation of the tensor algebra over a differentiable manifold. The analysis of the system of axioms in Definition I.3.1 reveals that not all of them are equally important for an abstract definition of derivation and leads to the following definition (see [11, Chapter I, § 3] and [12,88]).

**Definition 2.1.** A derivation of the tensor algebra  $\mathbf{T}^{1}(U)$  (or of  $\mathbf{T}(U)$ ), U being an open set in M, is a mapping  $D: \mathbf{T}^{1}(U) \to \mathbf{T}^{0}(U)$  possessing the following properties:

(i) Linearity,

$$D(aK + bL) = aD(K) + bD(L), \qquad a, b \in \mathbb{K}, \quad K, L \in \mathfrak{T}_s^{r;1}(U).$$

(ii) Leibnitz rule (relative to tensor multiplication),

$$D(K \otimes L) = (D(K)) \otimes L + K \otimes D(L) \quad K, L \in \mathbf{T}^{1}(U).$$

(iii) Type preservation,

$$D: \mathfrak{T}^{r;1}_s(U) \to \mathfrak{T}^{r;0}_s(U).$$

(iv) Commutativity with every contraction operator,

$$[D,C]_{\_} := D \circ C - C \circ D = 0.$$

**Example 2.1.** The comparison of Definitions I.3.1 and 2.1 (see also comments I.3.1) shows that every covariant derivative  $\nabla_X$  along  $X \in \mathfrak{X}(U)$  is a derivation of  $T^1(U)$ . Other examples of derivations are provided by the Lie derivative  $\mathcal{L}_X$  along  $X \in \mathfrak{X}(U)$  and by arbitrary tensor field  $S \in \mathfrak{T}_1^1$  of type (1, 1) [11]. This is easily seen from the following local expansions in the tensor frame with basic fields (I.2.41) induced by a frame  $\{E_i\}$  on U [11, 19, 89]:

$$(\mathcal{L}_X(K))_{j_1\dots j_s}^{i_1\dots i_r} = X\left(K_{j_1\dots j_s}^{i_1\dots i_r}\right) + \sum_{a=1}^r [-E_k(X^{i_a}) + C_{lk}^{i_a}X^l]K_{j_1\dots j_s}^{i_1\dots i_{a-1}ki_{a+1}\dots i_r} - \sum_{b=1}^s [-E_{j_b}(X^k) + C_{lj_b}^kX^l]K_{j_1\dots j_{b-1}kj_{b+1}\dots j_s}^{i_1\dots i_r},$$

$$(2.1)$$

$$(S(K))_{j_1\dots j_s}^{i_1\dots i_r} = \sum_{a=1}^r S_k^{i_a} K_{j_1\dots j_s}^{i_1\dots i_{a-1}ki_{a+1}\dots i_r} - \sum_{b=1}^s S_{j_b}^k K_{j_1\dots j_{b-1}kj_{b+1}\dots j_s}^{i_1\dots i_r}, \qquad (2.2)$$

where  $K \in \mathfrak{T}_s^{r;1}(U)$  is arbitrary  $C^1$  tensor field of type (r, s) on U.

The following proposition describes the general structures of the derivations of  $T^{1}(U)$  [11, Chapter I, Proposition 3.3].

**Proposition 2.1.** For every derivation D of  $T^1(U)$ , there exist unique vector field  $X \in \mathfrak{X}^1(U)$  and tensor field  $S_X \in \mathfrak{T}^1_1(U)$ , generally depending on X, such that

$$D = \mathcal{L}_X + S_X,\tag{2.3}$$

 $\Box$ 

*i.e.*, every derivation admits a unique decomposition (2.3).

*Proof.* See [11, Chapter I, proof of Proposition 3.3].

**Example 2.2.** In particular, a covariant derivative  $\nabla_X$  along  $X \in \mathfrak{X}(U)$  has a representation (2.3) with  $S_X$  given via

$$S_X(Y) = \Sigma_X(Y) := \nabla_X Y - [X, Y]_{-} = \nabla_Y X + T(X, Y)$$
(2.4)

where  $Y \in \mathfrak{X}(U)$  and T is the torsion of  $\nabla$  (see (I.3.12)).

Evidently, every pair (X, S) with  $X \in \mathfrak{X}(U)$  and  $S: \mathfrak{X}(U) \to \mathfrak{T}_1^1(U), S: X \mapsto S_X$ , defines a unique derivation of  $T^1(U)$  by (2.3). There is a one-to-one onto correspondence, given via (2.3), between the sets of derivations of  $T^1(U)$  and the set of pairs (X, S). If some derivation D has a decomposition (2.3), we shall write  $D_X^S$  instead of D; if the dependence on X is essential and the one on S is not, we reduce  $D_X^S$  to  $D_X$ . We say that  $D_X$  is a *derivation along* X.

#### 2. Derivations on differentiable manifolds

Using (2.1)–(2.3), we find the local components of  $D_X(K), K \in \mathfrak{T}_s^{r;1}$ , in some local frame as

$$(D_X(K))_{j_1\dots j_s}^{i_1\dots i_r} = X(K_{j_1\dots j_s}^{i_1\dots i_r}) + \sum_{a=1}^r \Gamma_X{}^{i_a}{}_k K_{j_1\dots j_s}^{i_1\dots i_{a-1}ki_{a+1}\dots i_r} - \sum_{b=1}^s \Gamma_X{}^k{}_{j_b} K_{j_1\dots j_{b-1}kj_{b+1}\dots j_s}^{i_1\dots i_r}.$$

$$(2.5)$$

Here the functions  $\Gamma_X{}^i{}_j \in \mathfrak{F}(U)$  are defined via the expansion

$$D_X(E_k) =: \Gamma_X{}^i{}_k E_i \tag{2.6}$$

and their explicit form is

$$\Gamma_X{}^i{}_j = (S_X)^i_j - E_j(X^i) + C^i_{kj}X^k$$
(2.7)

where  $C_{jk}^i \in \mathfrak{F}(U)$  define the commutators of the basic vector fields according to (I.3.15). We call  $\Gamma_X{}^i{}_j$  the (local) *components* of  $D_X$  in  $\{E_i\}$ ; respectively, we call  $\Gamma_X := [\Gamma_X{}^i{}_j]$  matrix (of the components) or components' matrix of  $D_X$ .

An important result follows from the comparison of equations (2.6) and (I.3.1) (or (2.5) and (I.3.2) and taking into account (I.3.3)).

**Proposition 2.2.** A derivation  $D_X$  of  $\mathbf{T}^1(U)$  with local components  $\Gamma_X{}^i{}_j$  in a frame  $\{E_i\}$  is a covariant derivative along X iff

$$\Gamma_X{}^i{}_j = \Gamma^i{}_{jk} X^k \tag{2.8}$$

for some functions  $\Gamma^i_{\ ik} \colon U \to \mathbb{K}$ , which are coefficients of a linear connection on U.

Hence the covariant derivatives are derivations whose components depend linearly on the generating them via (2.3) vector field X.

**Definition 2.2.** A derivation  $D_X$  of  $T^1(M)$  is said to be linear on (in) a set  $U \subseteq M$  or along a mapping  $\eta: Q \to M$  for a set  $Q \neq \emptyset$ , if in some frame (and hence in all frames)  $\{E_i\}$  the relation

$$\boldsymbol{\Gamma}_{X}(p) = [\Gamma_{X}{}^{i}{}_{j}(p)] = \Gamma_{k}(p)X^{k}(p)$$
(2.9)

holds for some matrix-valued functions  $\Gamma_1, \ldots, \Gamma_{\dim M}$  on U or on  $\eta(Q)$  and every  $p \in U$  or  $p \in \eta(Q)$ , respectively.

According to Proposition 2.2, the covariant derivatives are derivations which are linear at every point.

If we transform the frame  $\{E_i\}$  on U into a frame  $\{E'_i = A^j_i E_j\}$  by means of a  $C^1$  non-degenerate matrix-valued function  $A = [A^j_i]$ , we see from (2.6) and Definition 2.1 that the components  $\Gamma_X{}^i{}_j$  of  $D_X$  transform as (cf. (I.3.5))

$$\Gamma_X{}^i{}_j \mapsto \Gamma'_X{}^i{}_j = (A^{-1})^i_k [\Gamma_X{}^k{}_m A^m_j + X(A^k_j)].$$
(2.10)

Introducing the matrix  $\mathbf{\Gamma}_X := [\Gamma_X{}^i{}_j]_{i,j=1}^{\dim M}$  of the components of the derivation  $D_X$ , we rewrite (2.10) in the form (cf. (I.5.3))

$$\Gamma_X \mapsto \Gamma'_X = A^{-1}[\Gamma_X A + X(A)] \tag{2.11}$$

where  $X(A) := [X(A_{j}^{i})]_{i,j=1}^{\dim M}$ .

**Definition 2.3.** A derivation D along vector fields of the tensor algebra  $T^1(U)$ ,  $U \subseteq M$ , is a mapping assigning to every  $X \in \mathfrak{X}(U)$  a derivation  $D_X$  of  $T^1(U)$  with decomposition  $D_X = \mathcal{L}_X + S_X$  for some  $S \colon \mathfrak{X}(U) \to \mathfrak{T}_1^1(U), S \colon X \mapsto S_X$ .

The concept of a derivation along vector fields<sup>1</sup> generalizes the one of a linear connection (see Definition I.3.1) to which it reduces if the derivative  $D_X$  along Xis a covariant one. In a given frame  $\{E_i\}$ , the main difference between the linear connections and derivations along vector fields is expressed by equations (I.3.1) and (2.6). The former one defines the *coefficients*  $\Gamma^i{}_{jk}: U \to \mathbb{K}$  of a linear connection  $\nabla$  irrespectively of some vector fields, while the latter one defines the *components*  $\Gamma^i{}_j: U \to \mathbb{K}$  of a derivation D along vector fields only with respect to some vector field X and, generally, one can not get rid of the dependence on Xunless  $D_X$  is linear in X.

**Definition 2.4.** A derivation D along vector fields is linear on (in) a set  $U \subseteq M$ or along a mapping  $\eta: Q \to M, Q \neq \emptyset$ , if in some frame (and hence in all frames)  $\{E_i\}$  the matrix  $\Gamma_X := [\Gamma_j^i]_{i,j=1}^{\dim M}$  of the components of D along every  $X \in \mathfrak{X}(M)$ admits a decomposition (2.9) for some matrix-valued functions  $\Gamma_1, \ldots, \Gamma_{\dim M}$  on U or on  $\eta(Q)$  and every  $p \in U$  or  $p \in \eta(Q)$ , respectively.

**Example 2.3.** It is trivial to verify that the linear connections are derivation along vector fields linear at every point. Conversely, a derivation D along vector fields linear on U is a linear connection on U. In fact, equation (2.8) holds for the components of such a derivation<sup>2</sup> and, by Proposition 2.2,  $D_X$  is a covariant derivative along X. So, by Definition I.3.1, D is a linear connection on U.

The elements  $\Gamma_{jk}^{i}$  of the matrices  $\Gamma_{k}$ , corresponding via (2.9) to a linear derivation D along vector fields, will be called *coefficients* of D on U. It is clear, the coefficients of D exist only on sets on which D is linear and they transform as coefficients of a linear connection by (I.3.5). The matrices  $\Gamma_{k} = [\Gamma_{jk}^{i}]_{i,j=1}^{\dim M}$  will be called *matrices of the derivation coefficients*, or *derivation's matrices*, or *coefficients' matrices* of the derivation.

The definition of a  $C^r$  derivation along vector fields (on  $C^k$  manifold with  $k \ge r$ ) is similar to the one of  $C^r$  linear connection (see Subsection I.3.2) and it is based on the transformation law (2.11). Given a class of frames connected via linear transformations whose matrices are  $C^{r'}$ ,  $r' \ge 1$ , matrix-valued functions, a

 $<sup>^1\</sup>mathrm{In}$  the series of papers [76, 80, 83] the name S-derivation is used instead of derivation along vector fields.

<sup>&</sup>lt;sup>2</sup>If (2.8) is valid, the equality (2.5) is equivalent on U to (I.3.2) and (I.3.3) with  $\nabla = D$ .

#### 2. Derivations on differentiable manifolds

derivation along vector fields is said to be of class  $C^r$ ,  $r \leq r'-1$  with respect to it if its local components in one (and hence in all) frame(s) in the above set of frames are of class  $C^r$ . In what follows, we shall suppose by default that a given class of  $C^{r'}$ ,  $r' \leq k-1$ , frames is fixed on a  $C^k$ ,  $k \geq 2$ , manifold M and it consists of all frames associated to one (or all) systems of local coordinates on M or its open subsets and all frames obtainable from them by means of linear transformations with  $C^{r'}$ matrices. For instance, by a  $C^1$  derivation along vector fields we understand one on a  $C^3$  manifold whose local components are  $C^1$  functions in any frame which is coordinate or is obtainable from such by means of  $C^2$  transformations.

Analogously to the case of linear connections (see the end of Subsection I.3.2), the concepts of curvature and torsion can be defined for derivations along vector fields.

The (operators of) curvature  $R^D$  and torsion  $T^D$  of a  $C^1$  or arbitrary, respectively, derivation D along vector fields on  $C^3$  or  $C^2$  manifold M, resp., are mappings

$$R^{D}: \mathfrak{X}(M) \times \mathfrak{X}(M) \times \boldsymbol{T}(M) \to \boldsymbol{T}(M)$$
$$T^{D}: \mathfrak{X}^{1}(M) \times \mathfrak{X}^{1}(M) \to \mathfrak{X}^{0}(M)$$

such that

$$R^D \colon (X,Y) \mapsto R^D(X,Y) := D_X \circ D_Y - D_Y \circ D_X - D_{[X,Y]}$$

$$(2.12)$$

$$T^{D}: (X,Y) \mapsto T^{D}(X,Y) := D_{X}(Y) - D_{Y}(X) - [X,Y]$$
(2.13)

for all vector fields  $X, Y \in \mathfrak{X}(M)$ . By abuse of the language, the quantities  $R^D(X,Y)$  and  $T^D(X,Y)$  will also be called (operators of) curvature and torsion, respectively, of D (along the pair (X,Y)). The mapping  $R: D \mapsto R^D$  (resp.  $T: D \mapsto T^D$ ), assigning to a derivation D along vector fields its curvature (resp. torsion), will be called curvature (resp. torsion) operator, or simply curvature (resp. torsion).

Using (2.5), it is a simple exercise to verify that  $R^D(X, Y)$  is a tensor field of type (1, 1), which we consider as a derivation, while  $T^D(X, Y)$  is a vector field and their components in a local frame  $\{E_i\}$  are given respectively by

$$\left[\left(R^{D}(X,Y)\right)_{j}^{i}\right] = X(\Gamma_{Y}) - Y(\Gamma_{X}) + \Gamma_{X}\Gamma_{Y} - \Gamma_{Y}\Gamma_{X} - \Gamma_{[X,Y]}, \qquad (2.14)$$

$$\left(T^{D}(X,Y)\right)^{i} = \Gamma_{X}{}^{i}{}_{j}Y^{j} - \Gamma_{Y}{}^{i}{}_{j}X^{j} - C^{i}_{jk}X^{j}Y^{k}$$
(2.15)

where  $\mathbf{\Gamma}_X := [\Gamma_X^{i_j}]$  is the matrix of the components of the derivation D along a vector field X in  $\{E_i\}$  (see (2.6)) and the functions  $C_{jk}^i$  are defined via  $[E_i, E_j]_{-} := C_{ij}^k E_k$ .

Since the equivalent equations (2.8) and (2.9) hold on U for a linear on  $U \subseteq M$  derivation D along vector fields, the equations (2.14) and (2.15) take

respectively the  $forms^3$ 

$$\left(R^D(X,Y)\right)_i^i = R^i_{jkl} X^k Y^l \tag{2.16}$$

$$(T^D(X,Y))^i = T^i_{jk} X^j Y^k.$$
 (2.17)

Here  $R_{jkl}^i$  and  $T_{jk}^i$  are given by (I.3.13) and (I.3.14), respectively, i.e., they are the components of the ordinary curvature and torsion tensors of a linear connection  $\nabla$  on U with coefficients coinciding with the ones of D on U.

A derivation along vector fields with vanishing curvature (resp. torsion) on a set  $U \subseteq M$  will be called *flat*, or *curvature free*, or *integrable* (resp. *torsionless* or *torsion free*) on U. For U = M, we call it simply flat (resp. torsionless).

Details on part of the above material, as well as other general results concerning derivations of the tensor algebra over a manifold, the reader can find in [11].

The symmetry of the coefficients,

$$\Gamma^i_{\ jk} = \Gamma^i_{\ kj},\tag{2.18}$$

of a torsion free derivation in coordinate frames on the set(s) on which it is linear is a trivial corollary from (2.15) and (2.9).

## 3. General overview

The heuristic arguments at the beginning of Section I.5, concerning linear connections, are completely applicable to the derivations along vector fields. According to them, one can expect the existence of (local) frames in which all terms except the first one in the right-hand side of (2.5) vanish. Obviously, this is equivalent to the vanishment of the local components of a given derivation along vector fields in some frame. Further in this chapter, we shall investigate at length when such special frames exist. Below in the present section, some general properties of these frames will be found provided they exist.

**Definition 3.1.** Given a manifold M, a subset  $U \subseteq M$  and a derivation D along vector fields of the tensor algebra  $T^1(M)$ . A frame  $\{E_i\}$ , defined on an open subset of M containing U or equal to it, is called normal on U for D if in it the components of D along every vector field  $X \in \mathfrak{X}(M)$ , i.e., of  $D_X$ , vanish everywhere on U. Respectively, if  $g: Q \to M$ , Q being non-empty set, a frame  $\{E_i\}$ , defined on an open subset of M containing g(Q) or equal to it, is called normal for D along g if it is normal on g(Q).

Similarly to (I.5.1), we can rewrite the first part of this definition as

 $\{E_i\}$  is normal on  $U \iff \Gamma_X{}^i{}_j|_U = 0$  in  $\{E_i\}$ . (3.1)

<sup>&</sup>lt;sup>3</sup>It should be emphasized, equations (2.16) and (2.17) are valid only on the set(s) on which the derivation D is linear.

Consequently, in a normal frame  $\{E_i\}$ , due to (2.5), the components of  $D_X K$ ,  $K \in \mathfrak{T}_s^{r;1}(U)$ , are

$$(D_X(K))_{j_1\dots j_s}^{i_1\dots i_r} = X\left(K_{j_1\dots j_s}^{i_1\dots i_r}\right) = X^l E_l\left(K_{j_1\dots j_s}^{i_1\dots i_r}\right).$$
(3.2)

If  $\{E_i\}$  is an arbitrary frame on (a neighborhood of)  $U \subseteq M$ , then on U exists a normal frame  $\{E'_i\}$  if and only if there is a  $C^1$  non-degenerate matrixvalued function  $A = [A^i_j]$  transforming  $\{E_i\}$  into  $\{E'_i\}, E'_i = A^j_i E_j$ , and such that, according to (3.1) and (2.11), satisfies the normal frame equation

$$(\mathbf{\Gamma}_X A + X(A))|_U = 0 \tag{3.3}$$

which in component form reads (see (2.10))

$$(\Gamma_X{}^i{}_m A^m_j + X^m E_m(A^i_j))|_U = 0. ag{3.3'}$$

Therefore a derivation D along vector fields admits normal frames on U iff (3.3) has solutions with respect to A in some frame  $\{E_i\}$  and every vector field X. Practically all properties, including the existence, of the normal frames are, explicitly or implicitly, related to the equation (3.3). Below we shall make some simple general conclusions from it provided it has (non-degenerate) solution(s).

**Proposition 3.1.** Let a derivation D along vector fields of  $T^1(M)$  admits a frame normal on  $U \subseteq M$ . Then the derivation D is linear on U, i.e., equation (2.9) (or (2.8) in component form) holds for every  $p \in U$ .

*Proof.* Let  $\{E'_i\}$  be a frame normal for D on U and  $\{E_i\}$  be arbitrary frame on U. There is a  $C^1$  non-degenerate matrix-valued function  $A = [A^i_j]$  such that  $E'_i = A^j_i E_j$  for which (3.3) is valid. Hence, we have  $\Gamma_X|_U = -[(X(A))A^{-1}]|_U = -[X^k(E_k(A))A^{-1}]|_U$ , i.e., equation (2.9) holds for

$$\Gamma_k(p) = -[E_k(A)A^{-1}]|_p, \qquad p \in U,$$
(3.4)

and hence, by Definition 2.4, the derivation D is linear on U.

Thus the linearity on U of a derivation D along vector fields is a necessary condition for the existence of frames normal for D on U. As we shall see in the next sections, this condition is generally not sufficient, exceptions being the zero-and one-dimensional cases.

Proposition 3.1 allows an essential simplification of equation (3.3) and its equivalent version (3.3'): the dependence on the arbitrary vector field X can be removed. Indeed, if D admits normal frame(s), then  $\Gamma_X = \Gamma_l X^l$  on U for some matrix-valued functions  $\Gamma_1, \ldots, \Gamma_{\dim M}$  which, when inserted into (3.3), reduces (3.3) to

$$(\Gamma_l A + E_l(A))|_U = 0 \tag{3.5}$$

 $\square$ 

due to the arbitrariness of  $X^1, \ldots, X^{\dim M}$ . Correspondingly (3.3') takes the form

$$\Gamma^{i}_{\ mk}A^{m}_{j} + E_{k}(A^{i}_{j}))|_{U} = 0.$$
(3.5')

where  $\Gamma^i_{jk} \colon U \to \mathbb{K}$  are the components of  $\Gamma_k$ ,  $\Gamma_k = [\Gamma^i_{jk}]_{i,j=1}^{\dim M}$ , i.e., they are the coefficients of the initial derivation.

Equation (3.5) is identical with equation (I.5.4), as it should be: according to the above-said, the normal frames on U, if any, for a derivation D along vector fields and for a linear connection coinciding on U with D are identical. This explains why most of the properties of the normal frames for derivations along vector fields and linear connections are similar or identical. Regardless of this, below some results, concerning normal frames for derivations, will be proved independently, for others a reference to the case of linear connections will be given, and, at last, to some peculiarities of the case of derivations attention will be paid.

According to the aforesaid, if frames normal on U for D exist, they are uniquely defined by the coefficients  $\Gamma^i_{ik}: U \to \mathbb{K}$  of D.

The following proposition and the first corollary of it describe the set of normal frames, in any, for a given derivation along vector fields.

**Proposition 3.2.** Let a derivation D along vector fields admits a frame normal on  $U \subseteq M$ . The set of all frames normal for D on U consists of the frames that can be obtained from a fixed frame normal for D on U by means of linear transformations whose matrices vanish on U under the action of the basic vector fields of the normal frames.

*Proof.* Suppose the frames  $\{E_i\}$  and  $\{E'_i\}$  are normal for D on U. Then, due to (3.1),  $\Gamma_X|_U = \Gamma'_X|_U = 0$  for every vector field X, so equation (3.3) reduces to  $(X(A))|_U = 0$ ,  $A = [A^i_j]$ . The choice  $X = E_i$  implies  $E_i(A)|_U = 0$ . The converse assertion is almost evident: if  $\{E_i\}$  is normal on U, i.e.,  $\Gamma_X|_U = 0$ , and  $E'_i = A^j_i E_j$  with  $E_i(A)|_U = 0$ , then, from the transformation law (2.11), we get  $\Gamma'_X|_U = 0$ , i.e., the frame  $\{E'_i\}$  is normal for D on U.

**Corollary 3.1.** All frames normal on U for some derivation along vector fields, if any, are connected via linear transformations whose coefficients vanish on U under the action of the basic vector fields of the normal frames.

Proof. See Proposition 3.2 or its proof.

**Corollary 3.2.** If a derivation D along vector fields admits a frame normal on  $U \subseteq M$ , then (2.9) holds in it with

$$\Gamma_k|_U = 0. \tag{3.6}$$

*Proof.* See Proposition 3.1, equation (3.4), and Corollary 3.1.

Remark 3.1. Here and below, we suppose the derivation D and the vector field X to be defined on the whole manifold M, i.e.,  $D: X \mapsto D_X$  with  $D_X$  being derivation of  $\mathbf{T}^1(M)$  and  $X \in \mathfrak{X}(M)$ . If D and X are defined on a smaller set, e.g., on a submanifold N of M with dim  $N < \dim M$ , then all of the above (and next) results are valid *mutatis mutandis* provided M is replaced with N; in particular, all frames will be defined on (subsets of) N and, correspondingly, the Latin indices  $i, j, \ldots$  should run from 1 to dim N, not to dim M.

Due to equation (3.6), Definition I.5.1 is a special case of Definition 3.1 when applied to linear connections. In fact, if  $\nabla$  is a linear connection, its components  $\Gamma_X{}^i{}_j$  are defined by (2.6) with  $\nabla$  for D and satisfy (2.8) at every point in U and for every  $X \in \mathfrak{X}(U)$ . So, if  $\nabla$  admits frames normal on U, equation (3.6) implies that a frame normal for  $\nabla$  on U according to Definition 3.1 is normal for  $\nabla$  on U by Definition I.5.1. Conversely, if  $\nabla$  admits frames normal on U according to Definition I.5.1, by equation (2.8) they are also normal by Definition 3.1. Consequently, Definitions 3.1 and I.5.1 are equivalent when linear connections are concerned.

Now we shall turn our attention to the holonomicity of the normal frames, if such exist.

**Proposition 3.3.** Let a derivation D along vector fields admits  $C^1$  frames normal on a neighborhood  $U \subseteq M$ . All of these frames are either holonomic or anholonomic depending on is the torsion of D zero of non-zero on U, respectively.

Proof. Let  $\{E_i\}$  be a frame normal for D on U. In it, by virtue of (2.15), the torsion along the pair  $(E_i, E_j)$  is  $(T^D(E_i, E_j))|_U = -(C_{ij}^k E_k)|_U = -([E_i, E_j]_)|_U$  and, consequently (see (2.15)), we have  $(T^D(X, Y))|_U = -(X^i Y^j [E_i, E_j]_)|_U$  for every  $X, Y \in \mathfrak{X}(U)$ . Therefore  $[E_i, E_j]_|_U = 0$  iff  $T^D|_U = 0$ .

*Remark* 3.2. Remarks I.5.2 and I.5.3 are valid *mutatis mutandis* in a case of derivations along vector fields.

Recalling Remark I.5.1 on page 40, which is completely valid in a case of general derivations along vector fields, we have to emphasize that the (an)holonomicity of a frame  $\{E_i\}$  normal on U does not imply any conclusions concerning the (an)holonomicity of the frame  $\{E_i\}$  outside U if it is defined on a set larger than U. Moreover, if  $\{E_i\}$  is normal on U but not outside it, the derivation's components in  $\{E_i\}$  must vanish solely on U and generally are non-zero outside U.

Proposition 3.3 simply means that only the torsionless derivations along vector fields admit holonomic frames normal on a neighborhood. Moreover, for these derivations anholonomic normal frames do not exist.

**Definition 3.2.** Let *D* be a derivation along vector fields admitting normal frames. A chart (V, x) of *M* and the associated to it coordinate system  $\{x^i\}$  are called normal (for *D*) on a set  $U \subseteq V$  if the coordinate frame  $\{\frac{\partial}{\partial x^i}\}$  is normal for *D* on *U*.

So, the normal coordinates are the ones for which the associated local frames are normal. They may exist only in the torsionless case, as stated in the following assertion. **Corollary 3.3.** Normal coordinates may exist on a neighborhood only for torsionless derivations along vector fields and they (locally) generate all frames normal for them, if any.

*Proof.* See Definition 3.2 and Proposition 3.3.

We can equivalently restate Corollary 3.3 by saying that derivations (along vector fields) with non-vanishing torsion do not admits holonomic frames normal on a neighborhood and if for them normal frames exist, they are anholonomic.

Now we want to turn our attention to coordinates normal on a subset  $U \subseteq M$  which may not be a neighborhood.

The normal coordinates, if any, for a derivation (along vector fields) admitting normal frames can be found as solutions of (3.3'). If on  $U \subseteq M$  normal frames exist and (V, x) is a chart with  $V \cap U \neq \emptyset$  and associated coordinates  $\{x^i\}$ , the coordinates  $x'^i = x'^i(x^1, \ldots, x^{\dim M})$  defined on V are normal on  $U \cap V$  if and only if (3.3') holds for  $E_i = \frac{\partial}{\partial x^i}$  and  $A_i^j = \frac{\partial x^j}{\partial x'^i}$ , i.e., iff

$$\left(\Gamma_X{}^i{}_m\frac{\partial x^m}{\partial x'{}^j} + X^m\frac{\partial}{\partial x^m}\left(\frac{\partial x^i}{\partial x'{}^j}\right)\right)\bigg|_{U\cap V} = 0$$

which, by virtue of  $\Gamma_{\mathbf{X}} = \Gamma_k X^k$ , is equivalent to<sup>1</sup>

$$\left(\frac{\partial^2 x^i}{\partial x'^j \partial x'^k} + \frac{\partial x^m}{\partial x'^j} \frac{\partial x^n}{\partial x'^k} \Gamma^i_{mn}\right)\Big|_{U \cap V} = 0.$$
(3.7)

As one can expect, this equation is identical with the normal frame equation (I.5.4') on page 41 defining the normal coordinates for linear connections.

**Proposition 3.4.** If a derivation D along vector fields admits coordinates normal on a set  $U \subseteq M$ , then it is torsionless on U. In other words, if D admits a holonomic frame normal on U, it is torsion free on U.

*Proof.* If  $p \in U$  and  $\{x'^i\}$  are coordinates in a neighborhood of p normal on U, then (3.7) is valid. Antisymmetrizing this equation with respect to j and k, we find  $\Gamma^i_{[mn]}|_U = 0$ . Combining the last result with (2.17) and (I.3.14), we get  $T|_U = 0$  as in a holonomic frame, like  $\{\frac{\partial}{\partial x^i}\}$  in (3.7), is fulfilled  $C^i_{jk} = 0$ .

The last proposition immediately implies the following assertion.

**Corollary 3.4.** If  $U \subseteq M$ , only the torsionless on U derivations along vector fields may admit frames which are normal and holonomic on U.

Later it will be proved that the vanishment of the torsion on a submanifold is also a sufficient condition for the existence of normal coordinates for derivations along vector fields admitting frames normal on them.

<sup>&</sup>lt;sup>1</sup>The same result follows also from (3.5') for  $E_i = \partial/\partial x^i$  and  $A_i^j = \partial x^j/\partial x^i$ .

If a derivation D along vector fields admits frames normal on  $U \subseteq M$ , it is a linear connection on U, as pointed above. To this connection corresponds a parallel transport (see Subsection I.3.3) for which *in ex.* are valid the results concerning parallel transports and normal frames of Section I.5 (see page 42 and further). In particular, Propositions I.5.5 and I.5.6 remain true for the parallel transport in U associate to D.

## 4. Frames and coordinates normal at a point

From the considerations in the previous section and Subsection II.2.2, we can expect the existence of frames normal for every derivation D along vector fields at every fixed point p in a  $C^2$  manifold M provided D is linear at p, i.e.,

$$\Gamma_X(p) = X^k(p)\Gamma_k \tag{4.1}$$

in every frame for some *constant* dim  $M \times \dim M$  matrices  $\Gamma_1, \ldots, \Gamma_{\dim M}$ . Below we shall see that just this is the situation, which can be proved in a number of ways.

**Theorem 4.1.** Let M be a  $C^2$  manifold, p be a given point in M, and D be a derivation along vector fields of  $\mathbf{T}^1(M)$ . There exist frames normal for D at p if and only if D is linear at p. Moreover, if D is linear at p and (V, x) is a chart with  $V \ni p$ , then in V all frames normal for D at p are  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  where  $A := [A_i^j]: V \to \operatorname{GL}(\dim M, \mathbb{K})$  is of class  $C^1$ , non-degenerate, and its general form is

$$A(q) = \left\{ \mathbb{1} - \Gamma_j [x^j(q) - x^j(p)] \right\} A_0 + A_{jk}(q) [x^j(q) - x^j(p)] [x^k(q) - x^k(p)], \qquad q \in V \quad (4.2)$$

where  $\Gamma_1, \ldots, \Gamma_{\dim M}$  are the (constant) matrices of the coefficients of D given via (4.1),  $A_0$  is a constant non-degenerate matrix, and  $A_{jk}$  are  $C^1$  matrix-valued functions on U such that they and their first partial derivatives are bounded at p.

Proof. If D admits frames normal at p, by Proposition 3.1, it is linear at p, i.e., (4.1) holds. Conversely, if D is linear at p and (V, x) is a chart with p in its domain, a frame  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  is normal for D at p iff  $A = [A_i^j]$  satisfies (3.5) with  $U = \{p\}$ . In the proof of Theorem II.2.3 (p. 82), we proved that this equation always has solutions in V, the general one being given by (4.2) (see (II.2.14) with  $\Gamma_j$  for  $\Gamma_j(p)$  and V for U).

Theorem 4.1 provides a *complete* description of all frames normal at a single point for a given derivation (along vector fields) linear at this point. By Corollary 3.1, these frames are connected via linear transformations whose matrices vanish at the given point under the action of the basic vector fields of the normal frames.<sup>1</sup> According to Corollary 3.4 for  $U = \{p\}$ , the frames normal at a single point p and holonomic at p exist if the derivation is torsionless at p.

In the torsion free case, the linear derivations along vector fields admit normal coordinates at every fixed point p, which coordinates are solutions of (3.7) for  $U = \{p\}$  with respect to  $x'^{i,2}$  Equivalently, in this case, coordinates  $\{y^i\}$  normal at p can be found by transforming some local coordinates  $x^i$ , defined in a neighborhood of p, in such a way that the transformation  $\{x^i\} \to \{y^i\}$  leads to the vanishment at p of the derivation's coefficients in  $\{y^i\}$ . Since they change according to (I.3.5), this problem is solved in the discussion preceding Proposition II.2.3 (see the paragraph containing equation (II.2.11) on page 80). So, the general form of the coordinates  $\{y^i\}$  normal at p is given by equation (II.2.11') on page 80 in which  ${}^x\Gamma^l_{(ij)}(p)$  should be replace with the coefficients  ${}^x\Gamma^l_{ij}$  of the derivation (along vector fields) at p. In this way we have proved the following theorem generalizing Theorem II.2.1.

**Theorem 4.2.** Let M be a  $C^3$  manifold,  $p \in M$  and D be torsionless and linear at p derivation along vector fields. There exist coordinates  $\{y^i\}$  normal at p for D whose general form is

$$y^{i}(q) = y^{i}(p) + b^{i}_{j}[x^{j}(q) - x^{j}(p)] + b^{i}_{l}{}^{x}\Gamma^{l}_{jk}[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)] + b^{i}_{jkl}(q)[x^{j}(q) - x^{j}(p)][x^{k}(q) - x^{k}(p)][x^{l}(q) - x^{l}(p)], \quad (4.3)$$

where  $y^i(p)$  are constant numbers,  $x^i$  are arbitrarily chosen local coordinates in some neighborhood V of  $p, q \in V, [b^j_i]$  is constant non-degenerate matrix,  ${}^x\Gamma^l_{jk}$ are the coefficients of D at p in  $\{x^i\}$ , and the C<sup>3</sup> functions  $b^i_{jkl}: V \to \mathbb{K}$  together with their first partial derivatives are bounded at p.

This result gives a *complete* description of *all* coordinates normal at a single point for torsionless and linear at it derivations along vector fields.

Following the discussion before Proposition II.2.3, Theorem 4.2 can be slightly generalized: if D is arbitrary, with or without torsion, derivation linear at p, there exist local coordinates  $\{y^i\}$  in which the symmetric part of the coefficients of D, i.e.,  ${}^{y}\Gamma^{l}_{(jk)} := \frac{1}{2} ({}^{y}\Gamma^{l}_{jk} + {}^{y}\Gamma^{l}_{kj})$ , vanish at p and their general form is (4.3) (in which  ${}^{x}\Gamma^{l}_{jk}$  may be replaced with  ${}^{x}\Gamma^{l}_{(jk)}$ ). But these coordinates are not normal for D at p unless D is torsionless at p.

## 5. Frames and coordinates normal along paths

Relaying on the results of Section 3 and Subsection II.3.2, we may assume the existence of normal frames along a path  $\gamma: J \to M$  for a derivation D along

<sup>&</sup>lt;sup>1</sup>This result is also a consequence of (3.5) and (3.6) for  $U = \{p\}$ .

<sup>&</sup>lt;sup>2</sup>The vanishment of the torsion plays a role of an integrability condition for (3.7) with  $U = \{p\}$ .

#### 5. Frames and coordinates normal along paths

vector fields provided D is linear along  $\gamma$ , i.e., on the set  $\gamma(J)$ :

$$\Gamma_X(\gamma(s)) = \Gamma_k(\gamma(s)) X^k(\gamma(s)), \qquad s \in J$$
(5.1)

for some matrix-valued functions  $\Gamma_k$  on  $\gamma(J)$ ,  $\Gamma_X$  being the matrix (of the components) of D in a frame  $\{E_i\}$  defined in a neighborhood of  $\gamma(J)$ . (Notice, we suppose D to be linear on  $\gamma(J)$  but outside  $\gamma(J)$  it could not be such.) The truthfulness of this supposition is verified by the results presented below.

In this section, as in Section II.3 (see p. 85), the manifold M will be considered as real one, i.e., as real manifold of dimension  $\dim_{\mathbb{R}} M$ . (Recall,  $\dim_{\mathbb{R}} M = \dim M$ if M is real and  $\dim_{\mathbb{R}} M = 2 \dim M = 2 \dim_{\mathbb{C}} M$  if M is complex; see also p. 7.)

**Theorem 5.1.** Let M be  $C^2$ -manifold,  $\gamma: J \to M$  be  $C^1$  regular path without self-intersections, and D be  $C^0$ , i.e., continuous, derivation along vector fields of  $\mathbf{T}^1(M)$ . There exist frames (locally) normal for D along  $\gamma$  if and only if D is linear along  $\gamma$ , that is on  $\gamma(J)$ . Moreover, if D is linear along  $\gamma$ , for every  $s_0 \in J$ , there exist neighborhood  $U_1$  of  $\gamma(s_0)$  and a frame  $\{E'_i\}$  on  $U_1$  which is normal for D along  $\gamma$  in  $U_1$ , i.e., on  $U_1 \cap \gamma(J)$ . Besides, there exist subinterval  $J_1 \subseteq J$ with  $\gamma(J_1) = U_1 \cap \gamma(J)$  and a chart  $(U_1, x)$  with associated coordinates  $\{x^i\}$  in  $U_1$  in which  $\gamma^1(s) = s$ , the other components of  $\gamma(s)$  being constants, such that all of the mentioned normal frames are  $\{E'_i|_p = A^j_i(p)\frac{\partial}{\partial x^j}|_p\}$ , where  $p \in U_1$  and the  $C^1$  non-degenerate matrix-valued function  $A = [A^j_i]: U_1 \to \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$  is given by

$$A(p) = \left\{ \mathbb{1} - \Gamma_k(p_0) [x^k(p) - x^k(p_0)] \right\} Y(x^1(p), s_0; -\Gamma_1 \circ \gamma) B + B_{kl}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)].$$
(5.2)

Here 1 is the identity (unit) matrix,  $\Gamma_k$  are the coefficients' matrices of D (see equation (5.1)),  $p = x^{-1}(s, t) \in U_1$  with  $s \in J_1$  and  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ ,  $p_0 = \gamma(s) = x^{-1}(s, t_0)$  for fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M-1}$ , Y is the unique solution of the initial-value problem (II.3.17) with  $s = x^1(p)$  and  $Z = -\Gamma_1 \circ \gamma$ , B is constant non-degenerate matrix, and the matrix valued functions  $B_{kl}$  on  $U_1$  are of class  $C^1$  and they and their first partial derivatives are bonded on  $\gamma(J_1)$ .

*Proof.* If frames (locally) normal for D along  $\gamma$  exist, by Proposition 3.1 the derivation D is linear along  $\gamma$ , i.e., (5.1) holds.

Suppose D is linear along  $\gamma: J \to M$ . Let  $\{E_i\}$  be a frame defined on a neighborhood of  $\gamma(J)$ . As it was proved in Section 3, D admits frames  $\{E'_i = A^j_i E_j\}$  normal along  $\gamma$  if there exist non-degenerate matrix-valued  $C^1$  functions  $A = [A^j_i]$  which satisfy equation (3.5) for  $U = \gamma(J)$ , i.e.,

$$\Gamma_k(\gamma(s))A(\gamma(s)) + E(A)|_{\gamma(s)} = 0$$
(5.3)

for  $s \in J$ . So, D admits frames normal along  $\gamma$  iff this equation has solutions with respect to A for some  $\{E_i\}$ . Besides, these solutions, if any, describe all such normal

frames. In Subsection II.3.2 (see the paragraph containing equation (II.3.12) on page 94), we proved that for every  $s_0 \in J$  there exist a subinterval  $J_1 \subseteq J$ with  $J_1 \ni s_0$ , neighborhood  $U_1 \supset \gamma(J_1) = \gamma(J) \cap U_1$ , and local chart  $(U_1, x)$ with coordinate functions (II.3.12) in which the components of  $\gamma(s)$ ,  $s \in J_1$ , are  $\gamma^1(s) = s$  and  $\gamma^k(s) = t_0^k$  for some constant numbers  $t^k$  for  $k \ge 2$  (see (II.3.13)) and every point  $p \in U_1$  has coordinates of the form x(p) = (s, t) for some  $s \in J_1$ and  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$ .

Further, it was proved in Subsection II.3.2 that (5.3) for  $E_k = \frac{\partial}{\partial x^k}$  reduces to the system (II.3.16) which always has solutions, the general one being given by (5.2) (see the discussion before (II.3.23)).

Now the discussion between Theorems II.3.1 on page 97 and II.3.2 on page 99 can be repeated *in extenso* for derivations (along vector fields) linear along a given path  $\gamma$ . Below we summarize the main results of it.

The holonomicity of the (local) frames normal along  $\gamma$  for D depends on the torsion of D on  $\gamma(J)$ : if D is torsion free on  $\gamma(J)$ , i.e., along  $\gamma$ , these frames are holonomic and, correspondingly, along  $\gamma$  exist normal coordinates; otherwise, if D has non-vanishing torsion along  $\gamma$ , the mentioned frames are anholonomic. It should be emphasized that the holonomicity of the normal frames on  $\gamma(J)$  does not imply some conclusions of their holonomicity outside the set  $\gamma(J)$  where they can be holonomic or anholonomic depending on the free parameters B and  $B_{kl}$  in (5.2).

The normal frames provided by Theorem 5.1 are essentially *local*: they exist in the corresponding local coordinate neighborhoods and only as an exception such a neighborhood may contain the whole path  $\gamma$ . Covering the set  $\gamma(J)$  with coordinate neighborhoods of this kind and taking a normal frame in each of them,<sup>1</sup> we can construct all frames *globally* normal along  $\gamma$ . These global normal frames may turn not to be smooth in the region(s) of intersection of the domains of the local coordinate neighborhoods.

The above concerns injective paths, i.e., ones without self-intersections. This restriction can be weakened considerably: the results are valid *mutatis mutandis* for locally injective paths. For this end, the interval J has to be presented as a union of subintervals  $J_{\lambda}$ ,  $\lambda \in \Lambda$ , such that the restricted paths  $\gamma|_{J_{\lambda}}$  are without self-intersections. So the afore-given construction of, possibly global, normal frames is applicable for  $\gamma|_{J_{\lambda}}$ . From the frames normal along  $\gamma|_{J_{\lambda}}$  all frames globally normal along the whole path  $\gamma$  can be constructed. The details of this scheme are given before Theorem II.3.2 on page 99 whose version for derivations along vector fields reads:

**Theorem 5.2.** Let M be  $C^2$  manifold,  $\gamma: J \to M$  be locally injective,  $C^1$ , and regular path, and D be  $C^0$  derivation along vector fields. There exist frames (globally, i.e., on  $\gamma(J)$ ) normal for D along the whole path  $\gamma$  iff D is linear along  $\gamma$ . Locally these frames, described via Theorem 5.1, are of class  $C^1$  but globally they may not be such. Frames normal and holonomic along  $\gamma$  exist iff D is torsion free on  $\gamma(J)$ .

<sup>&</sup>lt;sup>1</sup>In the region(s) of intersection, we arbitrarily fix (choose) some frame normal in it (them).

As was proved above, a derivation D along vector fields which is linear and torsion free along regular locally injective path  $\gamma$  admits holonomic frames normal along  $\gamma$ . Since these frames coincide with the ones for a linear connection  $\nabla$  with the same coefficients along  $\gamma$  as those of D, the corresponding normal coordinates, locally generating the holonomic normal frames, for  $\nabla$  and D coincide. The normal coordinates for a symmetric  $C^0$  linear connection are described by Theorem II.3.3 on page 103. Hence the normal coordinates for derivations along vector fields are given via the same theorem with the only change that the words "symmetric linear connection" must be replace with "derivation along vector fields which is linear and torsionless along  $\gamma$ ".

## 6. Frames and coordinates normal in a neighborhood

The main result concerning the existence of frames normal for derivations along vector fields in a neighborhood, or submanifold of maximum dimensionality, is that the flat linear connections are the only derivations admitting normal frames on them.

**Theorem 6.1.** Let M be  $C^3$  manifold, D be  $C^1$  derivation along vector fields on M, and U be  $(\dim M)$ -dimensional submanifold of M. There exist frames normal for D on U if and only if the restriction of D to U, i.e., to  $\mathbf{T}^1(U)$ , is a flat linear connection.

*Proof.* Suppose there is a frame  $\{E'_i\}$  normal for D on U. By Proposition 3.1 and its proof, D is linear on U and in an arbitrary frame  $\{E_i = (A^{-1})_i^j E'_j\}$  its coefficients' matrices are given by (3.4). Hence D reduces on U to a linear connection  $\nabla$  with coefficients' matrices (3.4) in  $\{E_i\}$  and whose curvature, according to (II.4.1) and (3.4), has in  $\{E_i\}$  components  $R^i_{jkl}$  such that on U is fulfilled

$$R_{kl} := [R^{i}_{jkl}]$$
  
=  $\{2E_{l}[+(E_{k}(A))A^{-1}] + 2(E_{k}(A))A^{-1}(E_{l}(A))A^{-1} + C^{m}_{kl}(E_{m}(A))A^{-1}\}_{[kl]}$   
=  $\{2(E_{l}E_{k})(A) + C^{m}_{kl}E_{m}(A)\}A^{-1} \equiv 0$ 

where  $[E_i, E_j] = 2E_{[i} \circ E_{j]} = C_{ij}^k E_k$  was used. Consequently  $\nabla$  (and also D) is flat on U.

Conversely, if D is a flat linear connection on U, by Theorem II.4.1, on U exist frames normal for this linear connection which, by virtue of (2.9), are also normal for D on U.

Theorem 6.1 reduces all problems concerning normal frames, if any, for derivations (along vector fields) on submanifolds of maximum dimension to the same problems for (flat) linear connections. Since the last range of problems was completely investigated in Section II.4, we are not going to repeat them and the corresponding discussion here. We shall only summarize the main points of them as the following exercises.

**Exercise 6.1.** Prove that every  $C^1$  flat linear derivation along vector fields on a  $C^3$  manifold admits (global) normal frames as, in fact, it is a flat linear connection. (See Theorem 6.1 and Corollary II.4.1.)

**Exercise 6.2.** Prove that all frames  $\{E'_i\}$  normal for a  $C^1$  flat linear derivation on an (dim M)-dimensional submanifold U are  $\{E'_i = A^j_i E_j\}$ , where  $\{E_i\}$  is a frame on U and  $A = [A^j_i]: U \to \operatorname{GL}(\dim M, \mathbb{K})$  is given by (II.4.11) in which  $p \in U$ ,  $p_0 \in U$  is fixed,  $B \in \operatorname{GL}(\dim M, \mathbb{K})$ ,  $\Gamma_k$  are the coefficients' matrices in  $\{E_i\}$ , and Y is the solution of (II.4.4) with q = p,  $q_0 = p_0$ , N = M,  $e_a = E_a$ , and  $Z_a = -\Gamma_a$  where  $a = 1, \ldots, \dim M$ . (See Theorems 6.1 and II.4.2.)

**Exercise 6.3.** Prove that, if a  $C^1$  derivation which is linear, flat and torsionless in the domain U of some local chart (U, x) of a  $C^3$  manifold, then it admits in U normal coordinates which can be obtained from  $\{x^i\}$  via (II.4.14) with  $a^i \in \mathbb{K}$ .

## 7. Frames and coordinates normal on submanifolds

As a consequence of the general constructions in Section 3, the equations describing frames normal on a submanifold U, if any, for derivations along vector fields coincide with the ones for a linear connection with the same coefficients on U. This allows an almost obvious direct transferring of the results obtained for linear connections to the general case of arbitrary derivations along vector fields. Here is a list of the main of them.

**Theorem 7.1.** Let N be submanifold of  $C^3$  manifold M and D be  $C^1$  derivation along vector fields of  $\mathbf{T}^1(M)$ . Then:

(i) There exist frames normal for D on N if and only if it is linear on N and it's curvature vanishes on N along vector fields tangent to N as a manifold:

$$\left(R^{D}(X,Y)\right)\Big|_{q} = 0, \qquad X_{q}, Y_{q} \in T_{q}(N), \quad q \in N \subseteq M.$$

$$(7.1)$$

- (ii) If D admits frames normal on N, every point q<sub>0</sub> ∈ N has a neighborhood U<sub>N</sub> in M such that on U<sub>N</sub> exist C<sup>1</sup> frames normal for D on N, i.e., on Ū = U<sub>N</sub> ∩ N ∋ q<sub>0</sub>. Moreover, on U<sub>N</sub> exist coordinates {x<sup>i</sup>} such that:
  - (a) to every  $p = x^{-1}(\boldsymbol{s}, \boldsymbol{t}'), \ (\boldsymbol{s}, \boldsymbol{t}') \in W_N \times W' \subseteq \mathbb{K}^{\dim N} \times \mathbb{K}^{\dim M \dim N}$ corresponds a unique point  $p_0 = x^{-1}(\boldsymbol{s}, \boldsymbol{t}'_0) \in \overline{U}$  for some  $\boldsymbol{t}'_0 \in W'$ independent of  $\boldsymbol{t}'$  and

#### 7. Frames normal on submanifolds

(b) all frames defined on  $U_N$  and normal on  $\overline{U}$  are  $\left\{E'_i = A^j_i \frac{\partial}{\partial x^j}\right\}$  with

$$A(p) = [A_i^j] = \left\{ \mathbb{1} - \Gamma_k(p_0) [x^k(p) - x^k(p_0)] \right\}$$
  
 
$$\times Y(\bar{x}(p_0), \boldsymbol{s}_0; -\Gamma_1 \circ (x^{-1}(\cdot, \boldsymbol{t}'_0)), \dots, -\Gamma_{\dim N} \circ (x^{-1}(\cdot, \boldsymbol{t}'_0))) B(\boldsymbol{t}'_0)$$
  
 
$$+ B_{kl}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]. \quad (7.2)$$

Here  $\Gamma_k$  are the coefficients' matrices of D, Y is the unique solution of equation (II.5.9a) with A = Y and initial condition (II.5.9c) on page 122,  $(\bar{U}_N, \bar{x})$  is the chart of N induced by  $(U_N, x)$ ,  $s_0 = \bar{x}(q_0)$ , B is bounded matrix-valued function, and the matrix-valued functions  $B_{kl}$  on  $U_N$  are of class  $C^1$  and they and their first partial derivatives are bounded on  $\bar{U}_N$ .

*Proof.* Let  $\{E_i\}$  be a frame on (a neighborhood of) N. Suppose D admits a frame  $\{E'_i = A^j_i E_j\}$  normal on N. By Proposition 3.1, the derivation D is linear on N and, according to the results after it, the matrix  $A = [A^j_i]$  satisfies the normal frame equation (3.5) with U = N, viz.

$$\Gamma_k(p)A(p) + (E_k(A))|_p = 0 \qquad p \in N.$$
 (7.3)

In Subsection II.5.2, we proved that this equation, written there as (II.5.7), has solutions iff the integrability conditions (II.5.1) hold with R being the curvature tensor of a linear connection on M whose coefficients, when restricted to N, coincide with the ones of D. Because of (2.16), the conditions (II.5.1) are equivalent to (7.1). Consequently, as A satisfies (II.5.7) with  $\Gamma_k$  being the coefficients' matrices of D, the equality (7.1) holds. Conversely, let the derivation D be linear with coefficients' matrices  $\Gamma_k$  and (7.1) be valid. Then, as we just said, the equation (7.3) admits solutions with respect to A. If A is such a solution, the frame  $\{E'_i = A^j_i E_j\}$  is normal for D on N due to (2.11) and (2.9). This completes the first part of the theorem's proof.

Now suppose D admits frames normal on N. Then D is linear on N and equation (7.1) and its equivalent version (II.5.1) hold. Then Theorem II.5.2 is valid for a connection whose coefficients on N coincide with the ones of D. Hence the coordinates with the properties (a) exist and in them, i.e., for  $E_i = \frac{\partial}{\partial x^i}$ , the general solution of (7.3) is given via (7.2). So, the frames  $\{E'_i = A^j_i E_j\}$  on  $U_N$  with  $A = [A^j_i]$  given by (7.2) are normal on  $\overline{U} = U_N \cap N$  for D and on  $U_N$  there are not other frames with this property.

Combining Theorem 7.1 with Corollaries II.5.1 and II.5.2, we can formulate the following result.

**Corollary 7.1.** Let N be a submanifold of a  $C^3$  manifold M endowed with  $C^1$  derivation D along vector fields which is linear on N. Suppose  $\nabla$  is a linear connection on M and the coefficients of  $\nabla$  on N coincide with the ones of D. Then on (a neighborhood of) N exist frames normal for D if and only if on N exists a frame parallel with respect to  $\nabla$  or, equivalently, iff the parallel transport generated by  $\nabla$  is path-independent on N along all paths lying entirely in N.

If a derivation along vector fields admits frames normal on a submanifold N, Assertion (ii) of Theorem 7.1 gives a complete *local* description of all such frames.

**Exercise 7.1.** Using the explicit form (7.2) of the transformation matrix A, prove the validity of Proposition 3.2 and Corollary 3.1 on submanifolds; this is reflected in the constancy of the matrix  $B(t'_0)$  in (7.2) for a fixed choice of the local coordinates  $\{x^i\}$ .

**Exercise 7.2.** Construct a frame globally normal on the whole N from the frames locally normal on N. The idea is to take a system of overlapping (coordinate) neighborhoods, forming a neighborhood of N, on each of which  $C^1$  normal frames exist and from these local normal frames to be constructed a global frame normal on N. In this way all frames globally normal on N can be obtained, the details of this construction were explained after the proof of Corollary II.5.2.

If a derivation D along vector fields admits frames normal on a submanifold N, there are holonomic frames normal on N iff D has zero torsion on N (see Corollary 3.4). Since the normal frames for derivations, linear on N for which (7.1) holds, and linear connections, with the same coefficients on N, coincide in the torsionless case, the corresponding normal coordinates for them coincide too. All of these normal coordinates  $\{z^i\}$  can be found explicitly by integrating the equation  $\frac{\partial z^i}{\partial x^j}\Big|_q = (A^{-1}(q))_j^i$ , where  $q \in \overline{U}$  and A is given by (7.2), with respect to  $z^i$  in some neighborhood  $U_z$  of  $\overline{U}$  in  $U_N$  with  $U_z \cap N = \overline{U}$  (see (II.5.12)); for the notation see Theorem 7.1. This system of equations was completely solved in Subsection II.5.2 (see pages 124–126) and its general solution is (II.5.16). This result, combined with the just said in this paragraph, completes the proof of the following theorem (cf. Theorem II.5.3 on page 124).

**Theorem 7.2.** Let N be a submanifold of a  $C^3$  manifold M and D be a  $C^1$  derivation along vector fields such that on N it is linear, torsionless and (7.1) holds. For every point  $q \in N$  exists a chart  $(U_z, z)$  of M, with q in its domain  $U_z$ , which is normal for D on  $U_z \cap N$ . Moreover, the coordinate functions  $z^i : U_z \to \mathbb{K}$  of all such charts normal on N are given by

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \sum_{a=1}^{\min N} \left( A^{-1}(q) \right)_{a}^{i} dq^{a} + \left( A^{-1}(p_{0}) \right)_{j}^{i} \left[ x^{j}(p) - x^{j}(p_{0}) \right] + \left( A^{-1}(p_{0}) \right)_{l}^{i} \Gamma^{l}_{jk}(p_{0}) [x^{j}(p) - x^{j}(p_{0})] [x^{k}(p) - x^{k}(p_{0})] + a^{i}_{jkl}(p) [x^{j}(p) - x^{j}(p_{0})] [x^{k}(p) - x^{k}(p_{0})] [x^{l}(p) - x^{l}(p_{0})], \quad (7.4)$$

the notation being explained in Subsection II.5.2.

## 8. Frames and coordinates normal along mappings

In Section 5, we have investigated the problems of existence, uniqueness, and holonomicity of frames normal along a path  $\gamma: J \to M$  for a derivation D along vector fields. The aim of the present section is a generalization of these results for frames normal along sufficiently general mappings  $\gamma: J^n \to M$  or  $f: N \to M$ where  $J^n$  is a neighborhood in  $\mathbb{R}^n$  (or *n*-dimensional submanifold of  $\mathbb{R}^n$ ) and N is *n*-dimensional manifold for some integer  $n \in \{1, \ldots, \dim M\}$ . Obviously, the choice n = 1 and an open interval for  $J^1$  returns us to Section 5.<sup>1</sup> If we suppose  $\gamma(J^n)$  to be a single point or, more generally, an *n*-dimensional submanifold of M (and  $\gamma$  to be injective in the last case),<sup>2</sup> the results of Sections 4 and 7, respectively, should be reproduced. Relaying on these special cases, we can expect that for n = 0, 1the linearity of the derivation should be a necessary and sufficient condition for the existence of frames normal along  $\gamma$ , while for  $n \ge 2$  (if dim  $M \ge 2$ ) this might not be the general case.

Now the rigorous statements are in order.

Recall, according to Definition 3.1, we say that a frames  $\{E_i\}$  is normal along a mapping  $g: Q \to M$ , Q being non-empty set, for a derivation D along vector fields of  $\mathbf{T}^1(M)$  if it is defined in a neighborhood of g(Q) (or on M if g(Q) = M) and in it the components of D along every vector field vanish everywhere on the set g(Q).

By Proposition 3.1, a necessary condition for the existence of frames normal along a mapping  $\gamma: J^n \to M$  is the linearity of D along  $\gamma$ , i.e.,

$$\Gamma_X(\gamma(s)) = \Gamma_k(\gamma(s)) X^k(\gamma(s)), \qquad s \in J^n$$
(8.1)

in an arbitrary frame  $\{E_i\}$ . Here  $\Gamma_X$  is the components' matrix of D in it,  $X \in \mathfrak{X}(M)$ , and  $\Gamma_k$  are matrix-valued functions on  $\gamma(J^n)$  which are the coefficients' matrices of D.

From Section 3, we know that a linear derivation D admits normal frames along  $\gamma: J^n \to M$  iff for some (and hence for every) frame  $\{E_i\}$  there exists a non-degenerate matrix-valued function  $A = [A_j^i]$  such that the normal frame equation (3.5) holds along  $\gamma$ , viz.

$$(\Gamma_k A + E_k(A))|_{\gamma(J^n)} = 0 \tag{8.2}$$

or

$$\Gamma_k(\gamma(s))A(\gamma(s)) + E_k(A))|_{\gamma(s)} = 0, \qquad s \in J^n.$$
(8.2)

<sup>&</sup>lt;sup>1</sup>In Section 5 the interval  $J^1 \equiv J$  is of arbitrary type, i.e., it may be open or closed from one or both ends. To incorporate the last case, when  $J^1$  is closed from the left or/and right, as a special case of the next considerations, we must admit  $J^n$  to be *n*-dimensional submanifold with boundary in  $\mathbb{R}^n$ . The following results can be modified to cover this more general case, which we leave to the reader as an exercise. (See Remark I.2.1 on page 6 for the notion of a manifold with boundary.)

<sup>&</sup>lt;sup>2</sup>The set  $\gamma(J^n)$  is a submanifold if, for instance,  $\gamma$  is a regular embedding [12, p. 230].

If (8.2) has solutions with respect to A, all frames  $\{E'_i\}$  normal for D along  $\gamma$  are  $\{E'_i = A^j_i E_j\}$  for some A satisfying (8.2).

Since the system of equations (8.2) is a multidimensional generalization of (II.3.11), corresponding to n = 1, we shall partially follow the scheme of Subsection II.3.2.

The problem is to find, if any, all frames normal along a mapping  $\gamma: J^n \to M$ for a  $C^1$  linear derivation D along vector fields on a  $C^3$  manifold M. The next two subsections deal with the case when  $\gamma$  is respectively injective or locally injective while Subsection 8.3 investigates the case of a mapping  $f: N \to M$  between manifolds instead of  $\gamma$ .

In the present section, to detour some technical problems,<sup>3</sup> we shall look on the manifold M as on a real one, i.e., if M is real, no changes are necessary and, if M is complex, we consider it as a real manifold of dimension  $\dim_{\mathbb{R}} M =$  $2\dim M \equiv 2\dim_{\mathbb{C}} M$  (see p. 7). Below this will be formally reflected in writing  $\dim_{\mathbb{R}} M$  instead of dim M. (Recall,  $\dim_{\mathbb{R}} M$  is equal to dim M if M is real or to  $2\dim_{\mathbb{C}} M \equiv 2\dim M$  if M is complex.) Respectively, all Latin indices, with not explicitly specified range, run from 1 to dim<sub> $\mathbb{R}</sub> M$  and the ranges of the coordinate homeomorphisms lie in  $\mathbb{R}^{\dim_{\mathbb{R}} M}$ . In short, in the present section M is real manifold or complex manifold which we consider as real manifold of dimension dim<sub> $\mathbb{R}</sub> M$  and endowed with a real differentiable structure.</sub></sub>

#### 8.1. Injective mappings

At first we suppose the mapping  $\gamma: J^n \to M$  to be injective, i.e., without selfintersections: if  $s_1, s_2 \in J^n$  and  $s_1 \neq s_2$ , then  $\gamma(s_1) \neq \gamma(s_2)$ . This requires  $n = \dim J^n \leq \dim_{\mathbb{R}} M$ ; otherwise  $\gamma$  can not be 1:1. Below we are going to study and solve equation (8.2'), if it has solutions, in a frame  $\{E_i\}$  associated to a special kind of local coordinates. More precisely, we are going to show that  $\gamma(J^n)$  is *n*dimensional submanifold of M (in a sense of our definition on page 7) and then to apply the results of Subsection II.5.2 or of Section 7.

Let  $\gamma: J^n \to M$  be  $C^1$  regular injective mapping from a neighborhood  $J^n \subseteq \mathbb{R}^n$  on a  $C^3$  manifold M with  $\dim_{\mathbb{R}} M \ge n$ . Consider  $J^n$  as an n-dimensional manifold with a global chart  $(J^n, \operatorname{id}_{J^n})$  and standard coordinate functions  $r^a, a = 1, \ldots, n$ , such that  $r^a(s) = s^a$  for  $s = (s^1, \ldots, s^n) \in J^n$ . If  $(U, y), y: U \to \mathbb{R}^{\dim_{\mathbb{R}} M}$ , is a chart of M with  $U \cap \gamma(J^n) \neq \emptyset$  and associated coordinate functions  $y^i$ , the regularity of  $\gamma$  implies (see p. 15) that the Jacobi matrix of  $\gamma$  in the charts introduced, i.e.,  $\left[\frac{\partial(y^i \circ \gamma)}{\partial s^a}\Big|_s\right] = \left[\frac{\partial \gamma_y^i}{\partial s^a}\Big|_s\right]$  where  $\gamma_y^i := y^i \circ \gamma: J^n \to \mathbb{R}$  and here and

<sup>&</sup>lt;sup>3</sup>For instance, if  $\gamma$  is (locally) injective,  $C^1$  and regular, the set  $\gamma(J^n)$  is a submanifold of M and, roughly speaking, its real dimension is  $\dim_{\mathbb{R}}(\gamma(J^n)) = n$  while its complex dimension, if M is complex, is  $\dim_{\mathbb{C}}(\gamma(J^n)) = n/2$  which is not an integer for odd n. Besides, for complex M and odd n, the below-constructed chart  $(U_1, x)$ , which plays a crucial role in proving most of the results in this section, requires a complicated redefinition if we do not look on M as on a real manifold. (If n is even, the almost only change in the complex case is to replace  $\dim_{\mathbb{R}} M - n$  with  $\dim M - n/2$  and to identify the neighborhoods  $J^n$  in  $\mathbb{R}^n$  with some neighborhoods in  $\mathbb{C}^{n/2}$ .)

below a, b, c = 1, ..., n and  $i, j, k, l = 1, ..., \dim_{\mathbb{R}} M$  if the index ranges are not explicitly written, has a rank equal to  $n \leq \dim_{\mathbb{R}} M$  whenever s is such that  $\gamma(s) \in U \cap \gamma(J^n)$ .

**Lemma 8.1.** Let  $n \in \mathbb{N}$ , M be a  $C^3$  manifold with dim  $M \ge n$ , and  $\gamma: J^n \to M$ be  $C^1$  regular injective mapping. For every  $s_0 \in J^n$ , there exists a chart  $(U_1, x)$  of M such that  $\gamma(s_0) \in U_1$ ,  $x: U_1 \to J_1^n \times \mathbb{R}^{\dim_{\mathbb{R}} M-n}$  for some open subset  $J_1^n \subseteq J^n$ ,  $s_0 \in J_1^n$ , and  $x(\gamma(s)) = (s, t_0)$  for some fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$  and all  $s \in J_1^n$ .

*Remark* 8.1. This lemma is a multidimensional generalization of Lemma II.3.1 on page 94, which is its special case for n = 1.

*Proof.* Let us arbitrarily choose some  $s_0 \in J^n$  and a chart (U, y) with  $U \ni \gamma(s_0)$ and  $y \colon U \to \mathbb{R}^{\dim_{\mathbb{R}} M}$ . Since the regularity of  $\gamma$  at  $s_0$  means that  $\left[\frac{\partial \gamma_y^i}{\partial s^a}\Big|_{s_0}\right]$  has maximal rank, equal to n, we, without loss of generality, can suppose the coordinates  $\{y^i\}$  to be taken such that  $\det\left[\frac{\partial \gamma_y^u}{\partial s^b}\Big|_{s_0}\right] \neq 0, \infty$ .<sup>4</sup> Then the implicit function Theorem [7,77,78] implies the existence of a subneighborhood  $J_1^n \subseteq J^n$  with  $J_1^n \ni s_0$ and such that the matrix  $\left[\frac{\partial \gamma_y^u}{\partial s^b}\Big|_s\right]$  is non-degenerate for  $s \in J_1^n$  and the mapping

$$(\gamma_y^1,\ldots,\gamma_y^n)|_{J_1^n}\colon J_1^n\to (\gamma_y^1(J_1^n),\ldots,\gamma_y^n(J_1^n))\subseteq \mathbb{R}^n,$$

with  $(\gamma_y^1, \ldots, \gamma_y^n)|_{J_1^n} : s \mapsto (\gamma_y^1(s), \ldots, \gamma_y^n(s))$  for  $s \in J_1^n$ , is a  $C^1$  diffeomorphism. Define a chart  $(U_1, x)$  of M with domain

$$U_{1} := \{ p | p \in U, \ y^{a}(p) \in \gamma^{a}_{y}(J_{1}^{n}), \ a = 1, \dots, n \}$$
  
=  $y^{-1} ((\gamma^{1}_{y}(J_{1}^{n}), \dots, \gamma^{n}_{y}(J_{1}^{n})) \times \mathbb{R}^{\dim_{\mathbb{R}} M - n}) \ni \gamma(s_{0})$ (8.3a)

and local coordinate functions  $x^i$  given via

$$y^{a} := (\gamma_{y}^{a}|_{J_{1}^{n}}) \circ (x^{1}, \dots, x^{n}), \qquad a = 1, \dots, n$$
  
$$y^{k} := x^{k} + (\gamma_{y}^{k}|_{J_{1}^{n}}) \circ (x^{1}, \dots, x^{n}) - t_{0}^{k}, \qquad k = n + 1, \dots, \dim_{\mathbb{R}} M$$
(8.3b)

where  $(x^1, \ldots, x^n)$ :  $p \mapsto (x^1(p), \ldots, x^n(p))$ ,  $p \in U_1$ , and  $t_0^k \in \mathbb{R}$  are arbitrarily fixed constant numbers. Since  $\frac{\partial y^a}{\partial x^b} = \frac{\partial \gamma^a_y}{\partial s^b}$ ,  $\frac{\partial y^a}{\partial x^k} = \delta^a_k$  for  $k \ge n+1$ ,  $\frac{\partial y^k}{\partial x^a} = \frac{\partial \gamma^k_y}{\partial s^a}$  for  $k \ge n+1$ , and  $\frac{\partial y^k}{\partial x^l} = \delta^k_l$  for  $k, l \ge n+1$ , the Jacobian of the change  $\{y^i\} \to \{x^i\}$ on  $U_1$  is det $\left[\frac{\partial x^i}{\partial y^j}\right] = \left(\det\left[\frac{\partial \gamma^a_y}{\partial s^b}\right]\right)^{-1} \ne 0, \infty$ . Consequently  $x^i$  are really coordinate functions and  $x: U_1 \to J_1^n \times \mathbb{R}^{\dim_{\mathbb{R}} M-n}$  is in fact coordinate homeomorphism.<sup>5</sup>

<sup>&</sup>lt;sup>4</sup>If we start from a chart (U, z) for which the matrix  $\left[\partial \gamma_z^a / \partial s^b \right|_{s_0}$  is degenerate, we can make a coordinate change  $\{z^i\} \to \{y^i\}$  with  $y^i = z^{\alpha_i}$ , where the integers  $\alpha_1, \ldots, \alpha_{\dim_{\mathbb{R}} M}$  form a permutation of  $1, \ldots, \dim_{\mathbb{R}} M$ , such that  $\left[\partial \gamma_y^a / \partial s^b \right|_{s_0}$  is non-degenerate. (For the proof, see any book on matrices, e.g., [49,90].) Further, we suppose that such a renumbering of the local coordinates is already done if required. (Cf. Footnote II.15 on page 94.)

<sup>&</sup>lt;sup>5</sup>The so-constructed chart  $(U_1, x)$  is, obviously, a multidimensional generalization of a similar chart defined in Subsection II.3.2 – see the paragraph containing equation (II.3.12) on page 94.

The coordinates  $\{x^i\}$  can be expressed through  $\{y^i\}$  explicitly. Indeed, writing the first raw of (8.3b) as

$$(y^{1}, \dots, y^{n}) = (\gamma_{y}^{1}|_{J_{1}^{n}}, \dots, \gamma_{y}^{n}|_{J_{1}^{n}}) \circ (x^{1}, \dots, x^{n}) = (\gamma_{y}^{1}, \dots, \gamma_{y}^{n})|_{J_{1}^{n}} \circ (x^{1}, \dots, x^{n})$$

and using that  $(\gamma_y^1, \ldots, \gamma_y^n)|_{J_1^n}$  is a  $C^1$  diffeomorphism and the second raw of (8.3b), we find (cf. (II.3.12))

$$(x^{1}, \dots, x^{n}) = \left( (\gamma_{y}^{1}, \dots, \gamma_{y}^{n}) |_{J_{1}^{n}} \right)^{-1} \circ (y^{1}, \dots, y^{n})$$
$$x^{k} = y^{k} - (\gamma_{y}^{k} |_{J_{1}^{n}}) \circ \left( (\gamma_{y}^{1}, \dots, \gamma_{y}^{n}) |_{J_{1}^{n}} \right)^{-1} \circ (y^{1}, \dots, y^{n}) + t_{0}^{k}, \qquad k \ge n + 1.$$

$$(8.3b')$$

Using (8.3), we see that the local coordinates of  $\gamma(s)$  for  $s = (s^1, \ldots, s^n) \in J_1^n$ in  $(U_1, x)$  are

$$\gamma^{a}(s) := x^{a}(\gamma(s)) = s^{a}, \quad \gamma^{k}(s) := x^{k}(\gamma(s)) = t_{0}^{k}, \qquad k \ge n+1,$$
 (8.4)

i.e.,  $x(\gamma(s)) = (s, t_0)$  for some fixed  $t_0 = (t_0^{n+1}, \dots, t_0^{\dim_{\mathbb{R}} M}) \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ .

Thus, in the chart  $(U_1, x)$  or in the coordinates  $\{x^i\}$  constructed above, the first *n* coordinates of a point lying in  $\gamma(J^n)$ , i.e., in  $\gamma(J_1^n)$ , coincide with the corresponding parameters  $s^1, \ldots, s^n$  of  $\gamma$ , the remaining coordinates, if any, being constant numbers. This conclusion allows locally, in  $U_1$ , the mapping  $\gamma$ to be considered as a representative of a family of mappings  $\eta(\cdot, t) : J_1^n \to M$ ,  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ , defined by  $\eta(s, t) := x^{-1}(s, t)$  for  $(s, t) \in J_1^n \times \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ . In fact, we have  $\gamma = \eta(\cdot, t_0)$  or  $\gamma(s) = \eta(s, t_0)$ .<sup>6</sup>

It is almost evident, for  $C^1$  regular injective mapping  $\gamma: J^n \to M$ , the set  $\gamma(J^n)$  is an *n*-dimensional submanifold of M (in a sense of the definition on page 7). Indeed, for every point  $p = \gamma(s_0) \in \gamma(J^n)$ ,  $s_0 \in J^n$ , we can construct the aforedescribed chart  $(U_1, x)$  which is such that (vide supra)  $p \in U_1$  and

$$\gamma(J^n) \cap U_1 = \gamma(J_1^n) \xrightarrow{x} J_1^n \times \{\boldsymbol{t}\},$$
$$x \colon q \mapsto (x^1(q), \dots, x^n(q), t_0^{n+1}, \dots, t_0^{\dim_{\mathbb{R}} M}) = s \times \boldsymbol{t}_0$$

for every point  $q = \gamma(s) \in \gamma(J^n) \cap U = \gamma(J^n_1)$ , where  $s = (s^1, \ldots, s^n)$ ,  $x^a(q) = s^a$ , and some  $\mathbf{t}_0 = (t_0^{n+1}, \ldots, t_0^{\dim_{\mathbb{R}} M}) \in \mathbb{R}^{\dim_{\mathbb{R}} M - n}$ .

Let us return to our initial problem, the existence and description of normal frames and coordinates, if any, along the mapping  $\gamma$ . One can solve it by continuing the analogy with the material of Subsection II.3.2 and its multidimensional generalization. We leave this method to the reader as an exercise; in this way he/she can independently verify the below-presented assertions.<sup>7</sup> Alternatively, we are going to apply the already obtained results of Section 7 (or Section II.5).

<sup>&</sup>lt;sup>6</sup>In [83] the existence of  $\eta$  is taken as a given fact without proof.

 $<sup>^7\</sup>mathrm{If}$  one follows this way, Lemma II.4.1 should be used instead of Lemma II.3.2.

#### 8. Frames normal along mappings

Since  $\gamma(J^n)$  is *n*-submanifold of M, if the  $C^1$  regular mapping  $\gamma: J^n \to M$  is injective, for the problems concerning frames normal along  $\gamma$  are completely valid and applicable the results of Section 7. We shall partially reproduce them for the mentioned range of problems utilizing the above-introduced special chart  $(U_1, x)$ .

**Theorem 8.1.** Let M be  $C^3$  manifold endowed with  $C^1$  derivation D along vector fields and the mapping  $\gamma: J^n \to M$  be injective,  $C^1$ , and regular. Then:

 (i) There exist frames normal for D along γ if and only if D is linear along γ, i.e., on γ(J<sup>n</sup>), and it is curvature free on γ(J<sup>n</sup>) as a manifold,

$$\left(R^D(X,Y)\right)\Big|_q = 0, \qquad X_q, Y_q \in T_q(\gamma(J^n)), \quad q \in \gamma(J^n), \tag{8.5}$$

the last condition being locally, in the chart  $(U_1, x)$  introduced above, equivalent to

$$[R_{ab}(-\Gamma_1 \circ \gamma, \dots, -\Gamma_n \circ \gamma)](s) = 0, \qquad a, b = 1, \dots, n$$
(8.6)

where  $s = (s^1, \ldots, s^n) \in J_1^n$  with  $\gamma(J_1^n) = U_1 \cap \gamma(J^n)$  and

$$[R_{ab}(\Gamma_1 \circ \gamma, \dots, \Gamma_n \circ \gamma)]|_s := \left(\frac{\partial(\Gamma_a \circ \gamma)}{\partial s^b} - \frac{\partial(\Gamma_b \circ \gamma)}{\partial s^a}\right)\Big|_s + (\Gamma_a \Gamma_b - \Gamma_b \Gamma_a)|_{\gamma(s)}$$
(8.7)

with  $\Gamma_1, \ldots, \Gamma_{\dim_{\mathbb{R}} M}$  being the matrices of the coefficients of D in the frame  $\{\frac{\partial}{\partial x^i}\}$ .

- (ii) If D admits frames normal along  $\gamma$ , for every  $s_1 \in J^n$ , there exists a chart  $(U_1, x)$ , the same as in Assertion (i), with  $U_1 \ni \gamma(s_1)$  such that:
  - (a) In the associated coordinates  $\{x^i\}$ , every  $p \in U_1$  has coordinates x(p) = (s, t) for some  $s \in J_1^n$ , where  $J_1^n \subseteq J^n$  and  $U_1 \cap \gamma(J^n) = \gamma(J_1^n)$ ,  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ , and the coordinates of  $\gamma(s)$ ,  $s \in J_1^n$ , are  $x(\gamma(s)) = (s, t_0)$  for some constant (independent of s) vector  $\mathbf{t}_0 = (t_0^{n+1}, \dots, t_0^{\dim_{\mathbb{R}} M}) \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ ;
  - (b) All frames normal for D along  $\gamma$  in  $U_1$ , i.e., normal on  $\gamma(J_1^n)$ , are  $\{E'_i|_p = A^j_i(p)E_j|_p\}, p = x^{-1}(s, t) \in U_1$ , where

$$A(x^{-1}(s, t)) = [A_{j}^{i}(x^{-1}(s, t))] = \left\{ \mathbb{1} - \sum_{k=n+1}^{\dim_{\mathbb{R}} M} \Gamma_{k}(\gamma(s))(t^{k} - t_{0}^{k}) \right\}$$
  
  $\times Y(s, s_{0}; -\Gamma_{1} \circ \gamma, \dots, -\Gamma_{n} \circ \gamma) B(t_{0})$   
  $+ \sum_{k,l=n+1}^{\dim_{\mathbb{R}} M} B_{kl}(x^{-1}(s, t))(t^{k} - t_{0}^{k})(t^{l} - t_{0}^{l}).$  (8.8)

Here:  $s_0 \in J_1^n$  is fixed, the non-degenerate matrix  $B(\mathbf{t}_0) \in \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$  may depend only on  $\mathbf{t}_0$ , the matrix-valued functions  $B_{kl}$  on  $U_1$  are of class  $C^1$  and they

and their first partial derivatives are bounded when  $\mathbf{t} \to \mathbf{t}_0$ , and Y is the unique solution of the initial-value problem

$$\frac{\partial Y}{\partial s^a}\Big|_s = -\Gamma_a(\gamma(s))Y, \quad a = 1, \dots, n, \quad Y|_{s=s_0} = \mathbb{1}.$$
(8.9)

Remark 8.2. The fact that in (8.8) the summations are only over the range  $n + 1, \ldots, \dim_{\mathbb{R}} M$  and in Y enter only  $\Gamma_1, \ldots, \Gamma_n$  is a consequence of the special choice of the coordinates  $\{y^i\}$  used in the construction of  $\{x^i\}$ : we have admitted  $\det\left[\frac{\partial \gamma^a_y}{\partial s^b}\Big|_{s_0}\right] \neq 0, \infty$ . If we have chosen another non-degenerate *n*th-order submatrix of  $\left[\frac{\partial \gamma^i_y}{\partial s^b}\Big|_{s_0}\right]$ , this will result in the corresponding changes in the summation ranges and in the indices of the  $\Gamma$ 's in (8.8).

*Proof.* Since  $\gamma(J^n)$  is *n*-dimensional submanifold of M, which was proved earlier, by Theorem 7.1 (i) the condition (8.5) and the linearity of D on  $\gamma(J^n)$  are necessary and sufficient for the existence of frames normal for D along  $\gamma$ . By (2.16), equation (8.5) is equivalent to  $R^i_{jkl}|_{\gamma(J^n)} = 0$  (cf. (II.5.1') on page 113 with  $N = \gamma(J^n)$ ) in any chart (U, y) of M with  $U \cap \gamma(J^n) \neq \emptyset$ . So, according to Remark II.5.3 on page 114, the condition (8.5) is equivalent to (II.5.1'') on page 114 with  $N = \gamma(J^n)$ , which in the particular chart  $(U_1, x)$  takes the form (8.6).

Suppose now D admits frames normal along  $\gamma$ ; so it is linear along  $\gamma$  and (8.6) holds.

The existence and explicit construction of a chart  $(U_1, x)$  with the properties described in (ii), point (a), was presented in the proof of Lemma 8.1 and it is given by (8.3) in terms of some other chart (U, y). To prove the rest of the assertion, i.e., point (b) of (ii), we have to demonstrate that (8.8) is the general solution of (8.2') with  $E_i = \frac{\partial}{\partial x^i}$ ,  $\{x^i\}$  being the associated to  $(U_1, x)$  local coordinates. It is almost evident, this proof is contained in the discussion preceding Theorem II.5.2 in Subsection II.5.2. Indeed, in it we proved that (II.5.11) is the general solution of (II.5.7) on a submanifold N of M in a chart  $(U_N, x)$  described there. If we put  $N = \gamma(J^n)$ , then equation (II.5.7) reduces to (8.2') and the chart  $(U_1, x)$  introduced above is a concrete realization of  $(U_N, x)$  for  $\mathbf{t}'_0 = \mathbf{t}_0$ . Therefore (II.5.11), with  $p = x^{-1}(s, \mathbf{t})$ ,  $p_0 = x^{-1}(s, \mathbf{t}_0) = \gamma(s)$ ,  $\mathbf{t}'_0 = \mathbf{t}_0$ , and  $\bar{x}(p_0) = s$ , is the general solution of (8.2'). Taking into account (8.4) and  $x(p) = (s, \mathbf{t}) = (s^1, \ldots, s^n, t^{n+1}, \ldots, t^{\dim_{\mathbb{R}}M})$ , we conclude that in this case (8.8) is an equivalent representation of the matrix-valued function  $A: U_1 \to \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$  given by (II.5.11).

Remark 8.3. An independent proof of Theorem 8.1 is presented in [83].

**Corollary 8.1.** Let  $\gamma: J^n \to M$  be a  $C^1$  regular injective mapping in  $C^3$  manifold Mendowed with linear on  $\gamma(J^n) C^1$  derivation D along vector fields and  $\nabla$  be a linear connection coinciding on  $\gamma(J^n)$  with D. For D exist frames normal along  $\gamma$  iff there exists a frame parallel on  $\gamma(J^n)$  with respect to  $\nabla$  or iff the parallel transport generated by  $\nabla$  is path-independent on  $\gamma(J^n)$  along the paths lying entirely in  $\gamma(J^n)$ . *Proof.* Use that  $\gamma(J^n)$  is a submanifold of M and see Corollary 7.1 for  $N = \gamma(J^n)$  or Theorem 8.1 (ii) and Lemma II.5.1.

If a derivation D along vector fields admits frames normal along regular  $C^1$  injective mapping  $\gamma: J^n \to M$ , Theorem 8.1 (ii) provides a complete local description of all such frames. A frame  $\{E_i^g\}$  globally normal along the whole mapping  $\gamma$  can be constructed by covering the set  $\gamma(J^n)$  by a set  $\{U^{\lambda}|\lambda \in \Lambda\}$  of overlapping coordinate neighborhoods, like  $U_1$  above, in which exist frames  $\{E_i^{\lambda}\}$  normal along the 'part' of  $\gamma$  in them and then letting  $E_i^g|_p = E_i^{\lambda}|_p$  if p is in a single neighborhood  $U^{\lambda}$  and, if the sets  $U^{\lambda}$  with  $U^{\lambda} \ni p$  are more than one, we arbitrary choose some of them. The obtained in this way frame globally normal along  $\gamma$  is defined in a neighborhood of  $\gamma(J^n)$  equal to the union of the neighborhoods  $U^{\lambda}$  on which the local normal frames forming it are defined. In this way all frames globally normal along  $\gamma$  are generally not smooth on the sets of intersection of two or more neighborhoods  $U^{\lambda}$ .

If frames normal along  $\gamma: J^n \to M$  exist for a derivation D along vector fields, then, according to Corollary 3.4, these frames can be (locally) holonomic if D is torsionless on  $\gamma(J^n)$  in which case normal coordinates for it can exist.

**Theorem 8.2.** Let M be  $C^3$  manifold and  $\gamma: J^n \to M$  be injective,  $C^1$  and regular. Let D be  $C^1$ , linear along  $\gamma$  and torsionless derivation along vector fields for which (8.5) holds. Then for every  $s_1 \in J^n$  there exists a subneighborhood  $J_1^n \ni s_1$ of  $J^n$  and a chart  $(U_z, z)$  of M, with  $U_z \subseteq U_1$   $((U_1, x)$  is the above-constructed chart, see (8.3)) and  $U_z \cap \gamma(J^n) = \gamma(J_1^n)$ , which is normal for D along  $\gamma|_{J_1^n}$ , i.e., on  $\gamma(J_1^n)$ . The coordinate functions  $z^i: U_z \to \mathbb{R}^{\dim_{\mathbb{R}} M}$  of all such charts are given through the equation

$$z^{i}(x^{-1}(s,\boldsymbol{t})) = a^{i} + \int_{s_{0}}^{s} \sum_{a,b=1}^{n} \left(A^{-1}(\gamma(\sigma_{1},\ldots,\sigma_{n}))\right)_{a}^{i} \frac{\mathrm{d}\gamma^{a}}{\mathrm{d}\sigma^{b}} \mathrm{d}\sigma^{b}$$
  
+ 
$$\sum_{k=n+1}^{\dim_{\mathbb{R}}M} \left(A^{-1}(\gamma(s))\right)_{k}^{i}(t^{k}-t_{0}^{k}) + \sum_{l=1}^{\dim_{\mathbb{R}}M} \left(A^{-1}(\gamma(s))\right)_{l}^{i} \sum_{j,k=n+1}^{\dim_{\mathbb{R}}M} \Gamma^{l}_{jk}(\gamma(s))(t^{j}-t_{0}^{j})$$
  
 
$$\times (t^{k}-t_{0}^{k}) + \sum_{j,k,l=n+1}^{\dim_{\mathbb{R}}M} a^{i}_{jkl}(x^{-1}(s,\boldsymbol{t}))(t^{j}-t_{0}^{j})(t^{k}-t_{0}^{k})(t^{l}-t_{0}^{l}). \quad (8.10)$$

Here  $(s, t) \in J_1^n \times \mathbb{R}^{\dim_{\mathbb{R}} M-n}$  are the coordinates of an arbitrary point  $p \in U_z \subseteq U_1$ in the chart  $(U_1, x)$ , i.e., x(p) = (s, t), in which  $\gamma = x^{-1}(\cdot, t_0)$  for some fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M-n}$ ,  $s_0 \in J_1^n$  is fixed,  $a^i$  are constant numbers, A is given by (8.8), and  $a^i_{ikl}: U_z \to \mathbb{R}$  are bounded functions.

*Proof.* In fact, we have to solve the system  $\frac{\partial z^i}{\partial x^j}\Big|_q = (A^{-1}(q))^i_j, q \in U_z$  (cf. (II.5.12)) with respect to  $z^i$  in a neighborhood  $U_z$  of an arbitrarily chosen point in  $\gamma(J^n)$ . In

Subsection II.5.2 (see pp. 122–126) we solved this equation on arbitrary submanifold N of M for A given by (II.5.11), the general solution being (II.5.16). Since in the concluding part of the proof of Theorem 8.1 we proved that in the chart  $(U_1, x)$  the equality (II.5.11) reduces to (8.8) for  $N = \gamma(J^n)$ ,  $\mathbf{t}'_0 = \mathbf{t}_0$ ,  $x(p) = (s, \mathbf{t})$ ,  $p_0 = x^{-1}(s, \mathbf{t}_0) = \gamma(s)$ , and  $\bar{x}(p_0) = s$ , we can assert that the general form of the looked for coordinates  $\{z^i\}$  is (II.5.16) for the pointed special choice of the parameters in it. One can easily verify that, in the last case, equation (8.8) is simply an equivalent form of (II.5.16).

Theorem 8.2 gives a full constructive description of all local coordinates normal along an injective mapping  $\gamma: J^n \to M$  for a torsionless derivation along vector fields. Hence it completes the exploration of the normal frames and coordinates along such mappings.

#### 8.2. Locally injective mappings

Relying on the proof of Theorem II.3.2 on page 99, one can expect that the requirement the mapping  $\gamma: J^n \to M$  to be injective, imposed in the previous subsection, can be weakened. Indeed, this happens to be the case.

Call a mapping  $f: X \to Y$ , X and Y being sets, locally injective if for every  $x \in X$  there exists a subset  $X_x \subseteq X$  with  $X_x \ni x$  such that the restricted mapping  $f|_{X_x}: X_x \to Y$  is injective. If X has some structure, e.g., if it is n-manifold, we further admit that  $X_x$  possesses the same structure; e.g.,  $X_x$  to be n-manifold too in the particular example. For instance, the mapping  $\gamma: J^n \to M, J^n$  being neighborhood of  $\mathbb{R}^n$ , is locally injective if for every  $s \in J^n$  there is a subneighborhood  $I_s^n \subseteq J^n$  with  $I_s^n \ni s$  such that  $\gamma|_{I_s^n}: I_s^n \to M$  is injective. Obviously,  $\gamma$  is locally injective if the set  $J_I^n := \{s|s \in J^n, \gamma(s) \text{ is self-intersection point}\} \subset J^n$  has not condensation points. The maximum neighborhood of s on which  $\gamma$  is injective will be denoted by  $J_s^n$ . It can be represented as a union of all neighborhoods of s, like  $I_s^n$ , on which  $\gamma$  is injective.

**Proposition 8.1.** Theorems 8.1 and 8.2 are valid if in them the condition on  $\gamma$  to be injective is replaced with the requirement it to be locally injective and  $J_1^n$  and  $(U_1, x)$  are constructed for the restricted mapping  $\gamma|_{J_{s_0}^n}$  instead for  $\gamma$ .

*Proof.* Since  $\gamma|_{J_{s_0}^n}$  is injective, for it Theorems 8.1 and 8.2 hold. Consequently, they are also valid for the whole mapping  $\gamma$  if the pointed changes are made in their formulations, as in the considered version they concern only the local properties of  $\gamma$  on subneighborhoods on which it is injective.

Following the proof of Theorem II.3.2 and the lines after the proof of Corollary II.5.2, we are going to prove the following theorem.

**Theorem 8.3.** Let M be  $C^3$  manifold endowed with  $C^1$  derivation D along vector fields and the mapping  $\gamma: J^n \to M$  be locally injective,  $C^1$ , and regular. Suppose Dis linear along  $\gamma$  and its curvature satisfies (8.5). Then there exist frames globally
normal for D along the whole mapping  $\gamma$ , i.e., normal on the whole set  $\gamma(J^n)$ . Locally holonomic such frames exist iff D is torsion free on  $\gamma(J^n)$ .

Proof. By Proposition 8.1 and the discussion after the proof of Corollary 8.1, for every  $I^n \subseteq J^n$  for which  $\gamma|_{I^n}$  is injective there exist frames normal for D along  $\gamma|_{I^n}$ . Let  $\{I^n_{\lambda}|\lambda \in \Lambda\}$  be an open cover of  $J^n$  such that the restricted mappings  $\gamma|_{I^n_{\lambda}}$  are injective.<sup>8</sup> For each mapping  $\gamma|_{J^n_{\lambda}}$ ,  $\lambda \in \Lambda$ , there exists a frame  $\{E^{\lambda}_i\}$ globally normal for D, i.e., on  $\gamma(J^n_{\lambda})$  and all of them can be constructed in the way pointed in the mentioned discussion. From the frames  $\{E^{\lambda}_i\}$ ,  $\lambda \in \Lambda$ , a frame  $\{E_i\}$  globally normal along  $\gamma$  can be constructed: at every point  $p \in \gamma(J^n)$ , we put  $E_i|_p = E^{\lambda}_i|_p$  if there is a single  $\lambda \in \Lambda$  for which  $p \in \gamma(I^n_{\lambda})$ , otherwise, if the set  $\{\mu|\mu \in \Lambda, \gamma(I^n_{\mu}) \ni p\}$  consists of at least two elements, we arbitrary choose some  $\mu_0 \in \Lambda$  for which  $\gamma(I^n_{\mu_0}) \ni p$  and define  $E_i|_p = E^{\mu_0}_i|_p$ . The domain of  $\{E_i\}$  is a union of the neighborhoods on which the frames  $\{E^{\lambda}_i\}$  forming it are defined. It is obvious that all frames globally normal along  $\gamma$  can be constructed by the just describe method.

The last assertion of the theorem is a consequence of Proposition 3.4, Theorem 8.2, and Proposition 8.1  $\hfill \Box$ 

Notice, the proof of Theorem 8.3 provides a complete *global* description of all frames normal along locally injective mappings, if such frames exist at all.

### 8.3. Mappings between manifolds

Since the differentiable manifolds are 'locally Euclidean' (Subsection I.2.1), the results obtained until now in the present section have natural generalization for mappings like  $f: N \to M$ , N being n-manifold. Below we briefly sketch this situation. Without loss of generality, N and M will be considered as real manifolds (see p. 7).

Let  $f: N \to M$  be  $C^1$  mapping from a  $C^1$  manifold N on a  $C^3$  manifold M with  $\dim_{\mathbb{R}} M \ge \dim_{\mathbb{R}} N$  and D be  $C^1$  derivation along vector fields of  $\mathbf{T}^1(M)$ . We want to explore the problems of existence, uniqueness, construction, etc. of frames (or, possibly, coordinates) normal for D along f. From the considerations in Section 3, we know that D admits frames normal along f iff it is linear along f, i.e., on  $f(N) \subseteq M$ , and its coefficients' matrices  $\Gamma_k$  in a frame  $\{E_i\}$  defined on (a neighborhood of) f(N) satisfy the normal frame equation

$$\Gamma_k(f(q))A(f(q)) + E_k(A)|_{f(q)} = 0, \qquad q \in N$$
(8.11)

for some non-degenerate matrix-valued function A whose domain is the same as the one of  $\{E_i\}$ . Moreover, all frames normal along f have the form  $\{E'_i = A^j_i E_j\}$  where  $A = [A^j_i]$  is a non-degenerate solution of (8.11), if such exists. Hence the above

<sup>&</sup>lt;sup>8</sup>Let  $J_I^n := \{s | s \in J^n, \exists s' \in J^n, s' \neq s, \gamma(s) = \gamma(s')\}$ . If  $J_I^n = \emptyset$ , we put  $\Lambda = \{1\}$  and  $I_I^n = J^n$ . If  $J_I^n \neq \emptyset$ , we define  $\Lambda = J_I^n$  and  $I_s^n = J_s^n$ ,  $J_s^n$  being the maximal neighborhood of  $s \in I_s^n$  for which  $\gamma|_{I_s^n}$  is injective.

range of problems (for a linear along f derivation) is reduced to the investigation of the equation (8.11) in some frame  $\{E_i\}$ .

Locally (8.11) is equivalent to (8.2) for a suitable choice of the mapping  $\gamma: J^n \to M$ ,  $J^n$  being neighborhood in  $\mathbb{R}^n$  for  $n = \dim_{\mathbb{R}} N$ . To demonstrate this, for arbitrary  $q_0 \in N$ , we take a chart  $(V, \varphi)$  of N with  $V \ni q_0$  and  $\varphi: V \to J^n$  for some neighborhood  $J^n$  in  $\mathbb{R}^n$ . Since  $\varphi$  is, by definition, homeomorphism, for every  $q \in V$  exists a unique  $s = (s^1, \ldots, s^n) \in J^n$  such that  $s = \varphi(q)$  or  $q = \varphi^{-1}(s)$ . Therefore in  $(V, \varphi)$  equation (8.11) is equivalent to

$$\Gamma_k(f(\varphi^{-1}(s)))A(f(\varphi^{-1}(s))) + E_k(A)|_{f(\varphi^{-1}(s))} = 0, \qquad s \in J^n$$
(8.12)

which corresponds to (8.2) with

$$\gamma = f \circ \varphi^{-1}. \tag{8.13}$$

Consequently in each chart  $(V, \varphi)$  of N the problems connected to the frames normal along  $f: N \to M$  are equivalent to the same problems for the frames normal along the mapping  $\gamma = f \circ \varphi^{-1} \colon J^n \to M$  with  $J^n = \varphi(V) \subseteq \mathbb{R}^n$  and  $n = \dim_{\mathbb{R}} N$ . Moreover, the normal frames, if any, for the both mappings, f and  $\gamma$ , are identical. Since  $\gamma$  is (resp. locally) injective iff f is (resp. locally) injective (as  $\varphi$  is bijection), the results in Subsection 8.1 and 8.2 have corresponding variants concerning mappings between manifolds.

**Theorem 8.4** (cf. Theorem 8.1). Let  $f: N \to M$  be a mapping from a  $C^1$  manifold N into a  $C^3$  manifold M, both considered as real manifolds, if some of them is/are complex, with  $\dim_{\mathbb{R}} N \leq \dim_{\mathbb{R}} M$ , and D be  $C^1$  derivation along vector fields of  $T^1(M)$ . Suppose f is locally injective, of class  $C^1$ , and regular. Then:

(i) There exist frames normal along f for D if and only if D is linear along f, i.e., on f(N), and it is locally curvature free on f(N) in a sense that for every (dim N)-submanifold N' of N on which the restricted mapping f|<sub>N'</sub>: N' → M is injective, the derivation D is curvature free on f(N') as a manifold:

$$(R^D(X,Y))|_q = 0, \qquad X_q, Y_q \in T_q(f(N')), \quad q \in f(N').$$
(8.14)

For every chart  $(V,\varphi)$  of N',  $\varphi: V \to J^n \subseteq \mathbb{R}$  with  $n = \dim_{\mathbb{R}} N$ , there exists a chart  $(U_1, x)$  of M with  $U_1 \cap f(N') = f(V_1)$ ,  $V_1 = \varphi^{-1}(J_1^n)$  being subneighborhood of V, such that in it (8.14) is equivalent to (8.6) with  $\gamma = f \circ \varphi^{-1}$ .

- (ii) If D admits frames normal along f, for every p<sub>1</sub> ∈ N, there exists a chart (U<sub>1</sub>, x) of M, the same as in (i) above, with U<sub>1</sub> ∋ p<sub>1</sub> and such that:
  - (a) The coordinates of every  $p \in U_1$  in the associated coordinates  $\{x^i\}$  are x(p) = (s, t) for some  $s \in J_1^n$ , where  $J_1^n$  is a subneighborhood of  $J^n$  and  $U_1 \cap V = (f \circ \varphi^{-1})(J_1^n)$  (for the notation, see (i)), and  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M n}$ , and the coordinates of  $q \in U_1 \cap V$  are  $x(q) = (\varphi(q), t_0) = (s, t_0)$  for some  $s \in J_1^n$  and constant, independent of s, vector  $\mathbf{t}_0 = (t_0^{n+1}, \dots, t_0^{\dim_{\mathbb{R}} M}) \in \mathbb{R}^{\dim_{\mathbb{R}} M n}$ ;

#### 8. Frames normal along mappings

(b) All frames normal for D along f in  $U_1$ , i.e., normal on  $U_1 \cap V = (f \circ \varphi^{-1})(J_1^n)$ , are  $\{E'_i|_p = A_i^j(p)\frac{\partial}{\partial x^j}|_p\}$ ,  $p = x^{-1}(s,t) \in U_1$ , where A is given by (8.8) with  $\gamma = f \circ \varphi^{-1}$ .

Remark 8.4. The formulation of the theorem implicitly uses the fact that f(N') is *n*-dimensional,  $n = \dim_{\mathbb{R}} N$ , real submanifold of M. (Recall, all manifolds are considered as real ones in the present section.) This follows from the proved in Subsection 8.1 assertion that  $\gamma(J^n)$  is *n*-submanifold of M for regular  $C^1$  mapping  $\gamma: J^n \to M, J^n$  being neighborhood in  $\mathbb{R}^n$ . In fact, since in every chart  $(V, \varphi), \varphi: V \to J^n$ , of N we have  $f(V) = (f \circ \varphi^{-1})(J^n)$  and, consequently, f(N') is *n*-submanifold of M as  $\gamma = f \circ \varphi^{-1}$  is locally injective ( $\varphi$  is bijection), the image f(N') is *n*-submanifold of M as it can be represented as a union of *n*-submanifolds like f(V).

Proof. The first part of Assertion (i) is a trivial corollary of Theorem 7.1 (i) and the fact that the frames globally normal along f can be constructed from the frames normal along the restricted mappings  $f|_{N'}$  in a way analogous to the one described in the proof of Theorem 8.3 (with N for  $J^n$  and f for  $\gamma$ ). To prove the equivalence of (8.14) and (8.6) for  $\gamma = f \circ \varphi^{-1}$ , we notice that for every chart  $(V, \varphi)$  of N',  $\varphi: V \to J^n \subseteq \mathbb{R}^n$ , Theorem 8.1 (i) is valid for the mapping  $\gamma = f \circ \varphi^{-1}$ :  $J^n \to M$ . So, equation (8.14) in  $(V, \varphi)$  is equivalent to (8.6) with  $\gamma = f \circ \varphi^{-1}$  and there is a chart  $(U_1, x)$  of M, described in Subsection 8.1, in which (8.5) is equivalent to (8.6) for  $\gamma = f \circ \varphi^{-1}$  and  $U_1 \cap \gamma(J^n) = U_1 \cap f(N') = (f \circ \varphi^{-1})(J_1^n) = f(V_1)$ .

The proof of Assertion (ii) rests on Theorem 8.1 (ii). Let  $p_1 \in N$  and N' be  $(\dim N)$ -submanifold of N with  $N' \ni p_1$  and such that  $f|_{N'}$  is injective. Let  $(V, \varphi), \varphi \colon V \to J^n \subseteq \mathbb{R}^n$ , be a chart of N' and  $s_1 = \varphi(p_1) \in J^n$ . Now it is evident that the second part of the theorem is a simple restatement of the second part of Theorem 8.1 for the mapping  $\gamma = f \circ \varphi^{-1} \colon J^n \to M$ .

**Corollary 8.2** (cf. Corollary 8.1). Let the manifolds N and M, regarded as real ones, if some of them is/are complex, be of classes  $C^1$  and  $C^3$ , respectively, and the mapping  $f: N \to M$  be locally injective, of class  $C^1$ , and regular. Let a derivation D along vector fields of  $\mathbf{T}^1(M)$  be of class  $C^1$  and linear along f and  $\nabla$  be a  $C^1$  linear connection on M coinciding on f(N) with D. For D exist frames normal along f iff there is a frame locally parallel on N with respect to  $\nabla$  or iff the parallel transport generated by  $\nabla$  is locally path-independent on N along paths locally lying entirely in N. Here by 'locally' is understood on any (dim N)-dimensional submanifold N' of N such that  $f|_{N'}$  is injective.

*Proof.* See Corollary 8.1 for  $\gamma = f \circ \varphi^{-1}$ , where  $\varphi$  is the coordinate homeomorphism of a chart  $(V, \varphi)$  of N', and note that N can be covered by the domains of charts like  $(V, \varphi)$ .

If the frames normal along a mapping  $f: N \to M$  exist, the ones provided by Theorem 8.4 (ii) are local in a sense that they are defined only on some neighborhood (in M) of every point in f(N). From these neighborhoods one can construct a neighborhood of f(N) in M on which a frame globally normal along f can be defined by identifying it in each coordinate neighborhood with the corresponding local normal frame; in a case of overlapping of two or more neighborhoods, one arbitrary can fix some of the local frames to represent the global one. (For details, see the paragraph after the proof of Corollary 8.1.) It is clear, all frames globally normal along f can be constructed in this way.

If a derivation is torsionless, some of the frames normal for it, if any, are holonomic and hence locally they are generated by (normal) local coordinates. Of course, this is true for the frames normal along mappings between manifolds as the following result states explicitly.

**Theorem 8.5** (cf. Theorem 8.2). Let N and M be manifolds, considered as real ones if some of them is/are complex, of classes  $C^1$  and  $C^3$ , respectively, and  $f: N \to M$ be locally injective,  $C^1$ , and regular. Let a derivation D along vector fields of  $T^1(M)$  be torsionless, of class  $C^1$ , linear along f, and its curvature to satisfies equation (8.14) on every (dim N)-dimensional submanifold N' of N on which  $f|_{N'}$ is injective. Then for every point  $p_1 \in N$  there exist a chart  $(V_1, \varphi)$  of N, with  $\varphi: V_1 \to J_1^n \subseteq \mathbb{R}^n, V_1 \ni p_1, n = \dim_{\mathbb{R}} N$  and  $f|_{V_1}$  injective, and a chart  $(U_z, z)$  of M, with  $U_z \subseteq U_1$  ( $(U_1, x)$  is the chart provided by Theorem 8.4 (ii)) and  $U_z \cap N =$  $f(V_1) = (f \circ \varphi^{-1})(J_1^n)$ , which is normal for D along f in  $f(V_1)$ , i.e., along  $f|_{V_1}$ or on  $f(V_1)$ . The associated coordinates functions  $z^i: U_z \to \mathbb{R}^{\dim_{\mathbb{R}} M}$  of all such normal charts are given via equation (8.10) for  $\gamma = f \circ \varphi^{-1}$ .

*Proof.* By Theorem 8.4 and Corollary 3.3, the derivation D admits normal coordinates. Let  $p_1 \in N$  and  $(V, \varphi)$  and  $(U_1, x)$  be the charts described in the Theorem 8.4 with  $U_1 \ni f(p_1) \in V$ . In the notation of Theorem 8.4, define  $V_1 = U_1 \cap V = (f \circ \varphi^{-1})(J_1^n)$  and  $\gamma = f \circ \varphi^{-1} \colon J_1^n \to M$ . Now the rest of the proof follows from the remark that Theorem 8.2 is valid for the mapping  $\gamma = f \circ \varphi^{-1}$ .  $\Box$ 

# 9. Normal frames and coordinates for derivations along a fixed vector field

In Sections 3–8, we have explored problems connected with frames normal for derivations along vector fields of the tensor algebra over a manifold (see Definition 2.3). In the present section, we want briefly to pay attention to the same range of problems for derivations along a fixed vector field or, equivalently, for a given derivation of the same algebra (see Definition 2.1).

Throughout this section X will denote a fixed vector field over a manifold M,  $X \in \mathfrak{X}(M)$ . Respectively,  $D_X$  means a derivation along this concrete X of  $\mathbf{T}^1(M)$  or, in other words,  $D_X$  is a derivation of  $\mathbf{T}^1(M)$  with decomposition like (2.3) with fixed X and arbitrary S.

**Definition 9.1** (cf. Definition 3.1). Let U be a subset of a manifold M and  $D_X$  be a derivation along some fixed vector field  $X \in \mathfrak{X}(M)$  of  $T^1(M)$ . A frame  $\{E_i\}$ ,

defined on an open subset of M containing U or equal to it, is called *normal on* U for  $D_X$  if the components of  $D_X$  in  $\{E_i\}$  vanish on U (for the fixed X).

**Proposition 9.1.** The only derivation  $D_0$  along the zero vector field for which frames normal on  $U \subseteq M$  exist is the zero derivation, assigning to a  $C^1$  tensor field the zero tensor field of the same type. Every frame defined on (a neighborhood of) U is normal for  $D_0$  on its domain.

*Proof.* If  $D_{\mathbf{0}}$  admits a frame  $\{E_i\}$  normal on U,  $\Gamma_{\mathbf{0}i}^i{}_j|_U = 0$ , equation (2.5) implies  $D_{\mathbf{0}}K = 0$  for every tensor field K, so  $D_{\mathbf{0}}$  is the zero derivation, with decomposition (2.3) with  $X = \mathbf{0}$  and  $S_{\mathbf{0}} = 0$ . If  $D_{\mathbf{0}}$  is the zero derivation, from (2.6) follows that every frames  $\{E_i\}$  is normal for it.

Proposition 9.1 has a slightly more general local version:

**Proposition 9.2.** Let M be manifold,  $U \subseteq M$ ,  $X \in \mathfrak{X}(M)$ ,  $X|_U = 0$ , and  $D_X$  be derivation along X. Frames normal on U for  $D_X$  exist if and only if  $D_X$  is the zero derivation on U,  $D_X|_{T^1(U)} = D_0|_{T^1(U)}$ , assigning to every tensor field on U the zero tensor field of the same type on U. If on U the derivation  $D_X$  is the zero one, then every frame defined on U or on a larger set is normal for  $D_X$  on U.

*Proof.* If  $\{E_i\}$  is a frame normal for  $D_X$  on U, i.e.,  $\Gamma_X|_U = 0$ , from (2.5) we get  $(D_X K)|_U = 0$  for every tensor field K as  $X|_U = 0$ . Conversely, if  $D_X$  is the zero derivation on U, from (2.6), we derive  $\Gamma_X|_U = 0$  in every frame  $\{E_i\}$ .

Nevertheless the above two results are completely trivial, they give a full description of the frames, if any, normal for derivations along vector fields on the sets on which these vector fields vanish.<sup>1</sup> For this reason, further in this section we assume X to be a non-zero vector field on  $U, X|_U \neq 0$ , but there could exist points in U at which X vanishes. Notice, from (2.6) follows that if  $D_X$  is the zero derivation along X on  $U \subseteq M$ , then its components vanish identically in any frame defined on U. Combining this with (2.11), we get  $X|_U = 0$ . Hence, the zero derivation on U along fixed vector field can be such if this vector field vanished on U. So, assuming  $X|_U \neq 0$ , we automatically exclude the case when  $D_X$  is the zero derivation on U from further considerations.

One may call a derivation  $D_X$  along a fixed vector field X linear on  $U \subseteq M$ if in some (and hence in any) frame its components' matrix has the form  $\Gamma_X(p) =$  $\Gamma_k(p)X^k(p), p \in U$ , for some matrix-valued functions  $\Gamma_k$  on U. Since the components' matrix of every  $D_X$  for every fixed X possess (generally infinitely many) such representations,<sup>2</sup> this concept is useless for exploring the normal frames for  $D_X$ . Paraphrasing, we may say that every derivation along a fixed vector field is linear on every set.

<sup>&</sup>lt;sup>1</sup>Evidently, if  $D_X$  is the zero derivation on U and  $\{E_i\}$  is a frame defined on V with  $V \cap U \neq \emptyset$ , the frame  $\{E_i\}$  is normal on  $V \cap U$  for  $D_X$ . Consequently, every chart or coordinate system is normal for the zero derivation on the intersection of their domains.

<sup>&</sup>lt;sup>2</sup>For example, if  $X^1(p) \neq 0$ , we put  $\Gamma_1(p) = \Gamma_X(p)/X^1(p)$  and  $\Gamma_k(p) = 0$  for  $k \geq 2$ .

Let  $\{E_i\}$  be a frame defined on a (neighborhood of)  $U \subseteq M$ . A frame  $\{E'_i = A^j_i E_i\}$ , where  $A = [A^j_i]$  is a  $C^1$  non-degenerate matrix valued-function, is normal for  $D_X$  on U iff A is a solution of the normal frame equation (see (2.11), cf. (3.3))

$$(\mathbf{\Gamma}_X A + X(A))|_U = 0 \quad \text{for fixed } X \in \mathfrak{X}(M).$$
(9.1)

So,  $D_X$  admits frames normal on U iff this equation (for fixed X) has solutions; if they exist, all such frames are  $\{E'_i = A^j_i E_j\}$  with A being a solution of (9.1). Below we shall prove that equation (9.1) (for fixed X) always has (non-degenerate) solutions (in a neighborhood of the non-singular points of X) which results in the general conclusion that frames normal for a derivation along a fixed vector field exist on arbitrary subsets of M, in particular on the whole manifold M. The set of frames normal for a derivation along fixed vector field is described by the following result.

**Proposition 9.3.** Let  $D_X$  be derivation along some fixed  $X \in \mathfrak{X}(M)$  and  $U \subseteq M$ . The frames normal on U for  $D_X$  (if any) are connected by linear transformations whose matrices vanish under the action of X on U. Equivalently, the set of all frames normal for  $D_X$  (if any) consists of frames that can be obtained from a fixed normal frame for  $D_X$  by transformations of the described type.

*Proof.* If  $\{E_i\}$  and  $\{E'_i = A^j_i E_j\}$  are normal for  $D_X$  on U, then  $\Gamma_X = \Gamma'_X = 0$ , so (9.1) (or (2.11)) implies  $X(A)|_U = 0$ ,  $A = [A^j_i]$ .

Since the concepts of curvature and torsion can not be introduced for derivations along a fixed vector fields,<sup>3</sup> for them there are no analogues of Proposition 3.3 and Corollary 3.3. Regardless of that, they always admit normal coordinates (in a neighborhood of the non-singular points of X) which are defined as follows.

**Definition 9.2** (cf. Definition 3.2). A chart (V, x) of M and the associated to it coordinate functions or system are called normal on  $U \subseteq V$  for a derivation  $D_X$  along fixed  $X \in \mathfrak{X}(M)$  if the coordinate frame  $\left\{\frac{\partial}{\partial x^i}\right\}$  is normal for  $D_X$  on U.

If  $U \subseteq M$  and (V, x) is a chart of M such that  $V \cap U \neq \emptyset$ , the coordinates  $x'^i = x'^i(x^1, \ldots, x^{\dim M})$  are normal on  $U \cap V$  iff (9.1) holds for  $A_i^j = \frac{\partial x^j}{\partial x^i}$ , i.e., iff they are solutions of the normal frame/coordinates equation

$$\left(\Gamma_X{}^i{}_m\frac{\partial x^m}{\partial x'{}^j} + X^m\frac{\partial}{\partial x^m}\left(\frac{\partial x^i}{\partial x'{}^j}\right)\right)\Big|_{U\cap V} = 0.$$
(9.2)

Using the equalities

$$\frac{\partial x^i}{\partial x'{}^l}\frac{\partial x'{}^l}{\partial x^k} = \delta^i_k, \qquad \frac{\partial}{\partial x^j} \left(\frac{\partial x^i}{\partial x'{}^k}\right) = -\frac{\partial x^i}{\partial x'{}^l}\frac{\partial^2 x'{}^l}{\partial x^j \partial x^m}\frac{\partial x^m}{\partial x'{}^k},$$

<sup>&</sup>lt;sup>3</sup>In some sense the curvature and torsion for every  $D_X$  are identically zero as seen from (2.12) and (2.13) for Y = X.

the second of which is a consequence of the first one and its derivative with respect to  $x^{j}$ , we rewrite (9.2) as

$$\left(X^m \frac{\partial^2 x'^i}{\partial x^m \partial x^k} - \frac{\partial x'^i}{\partial x^m} \Gamma_X^m{}_k\right)\Big|_{U \cap V} = 0.$$
(9.2')

This equation is more suitable than (9.2) for finding  $x'^{i}$  if  $x^{i}$  are known.

Now we turn our attention to the problems of existence and construction of normal frames and coordinates for derivations along fixed vector fields. Formally they are equivalent to the solutions of equations (9.1) and (9.2), respectively, under certain conditions.

#### 9.1. The case at a single point

Let  $p_0 \in M$  be a given point in a manifold M and  $D_X$  be a derivation along some fixed vector field  $X \in \mathfrak{X}(U)$  of  $\mathbf{T}^1(M)$  with components' matrix  $\Gamma_X$  in a frame  $\{E_i\}$  defined on a neighborhood of  $p_0$ . The frames normal for  $D_X$  at  $p_0$ , if any, are  $\{E'_i = A^j_i E_j\}$ , where  $A = [A^j_i]$  is a non-degenerate solution of (9.1) for  $U = \{p_0\}$ :

$$\Gamma_X(p_0)A(p_0) + (X(A))|_{p_0} = 0.$$
(9.3)

**Theorem 9.1.** Let M be  $C^1$  manifold,  $p_0 \in M$  and  $D_X$  be a derivation along fixed  $X \in \mathfrak{X}(M)$ . If  $X_{p_0} = 0$ , then either every frame in a neighborhood of  $p_0$  is normal at  $p_0$  for  $D_X$  (if  $D_X$  is the zero derivation at  $p_0$ ) or such frames do not exist (if  $D_X$  is not the zero derivation at  $p_0$ ). If  $X_{p_0} \neq 0$ , then  $D_X$  always admits frames  $\{E'_i\}$  normal at  $p_0$  and, if X is continuous, all such frames are  $\{E'_i|_p = A^j_i(p)E_j|_p\}$ , where  $A: U_1 \to \operatorname{GL}(\dim M, \mathbb{K})$  is given by equation (9.8) below and  $p \in U_1$ , with  $(U_1, x)$  being the special chart constricted in Subsection II.3.2 for  $U_1 \ni p_0$  and  $\gamma$  being the integral path for X through  $p_0$ .<sup>4</sup>

*Proof.* If  $X_{p_0} = 0$ , then equation (9.3) has solution(s) with respect to A iff  $\Gamma_X(p_0) = 0$  in which case  $D_X$  is the zero derivation at p and every non-generate matrix-valued function A is its solution. This trivial conclusion, which follows also from Proposition 9.2 for  $U = \{p\}$ , means that for  $X_{p_0} = 0$  either every frame in a neighborhood of  $p_0$  is normal for  $D_X$  at  $p_0$  or frames normal for  $D_X$  at  $p_0$  do not exist. To check the case, one should take some frame and calculate the matrix of the components of  $D_X$  in it. If it happens to be non-vanishing at  $p_0$ , frames normal at  $p_0$  do not exist, otherwise every frame is normal for  $D_X$  at  $p_0$ .

Suppose now  $X_{p_0} \neq 0$ . Since (9.3) implies a single condition at  $p_0$  on A and its directional derivative along X, one can expect that it admits infinitely many solutions which will be proved below.

<sup>&</sup>lt;sup>4</sup>Equivalently, in an arbitrary chart (U, y) with  $U \ni p_0$ , the matrix-valued function A has the form (9.4) under the condition (9.5). Formula (9.7) provides a class of such A which is its general form iff det $(\mathbf{\Gamma}_X(p_0)) \neq 0$ .

Let (U, y) be a chart of M with  $U \ni p_0$ ,  $E_i = \frac{\partial}{\partial y^i}$ , and  $A: U \to \operatorname{GL}(\dim M, \mathbb{K})$ be of class  $C^2$ . In U is valid the Taylor expansion

$$A(p) = A_0 + A_k[y^k(p) - y^k(p_0)] + A_{kl}(p)[y^k(p) - y^k(p_0)][y^l(p) - y^l(p_0)]$$
(9.4)

where  $A_0 \in \operatorname{GL}(\dim M, \mathbb{K})$ ,  $A_k$  are constant matrices and the matrix-valued functions  $A_{kl}$  on U are of class  $C^1$  and they and their first partial derivatives are bounded at  $p_0$ . Inserting (9.4) into (9.3), we get

$$\Gamma_X(p_0)A_0 + X_y^i(p_0)A_i = 0, \qquad \det A_0 \neq 0$$
(9.5)

with  $X_y^i$  being the components of X in  $\left\{\frac{\partial}{\partial y^i}\right\}$ .

A simple verification shows that

$$A_0 = X_y^i(p_0)B_i, \qquad A_k = -\Gamma_X(p_0)B_k, \tag{9.6}$$

with arbitrary constant matrices  $B_1, \ldots, B_{\dim M}$  for which  $\det(X_y^i(p_0)B_i) \neq 0$ , is a solution of (9.5) to which corresponds a solution of (9.3) of the form<sup>5</sup>

$$A(p) = X_y^i(p_0)B_i - \mathbf{\Gamma}_X(p_0)B_k[y^k(p) - y^k(p_0)] + A_{kl}(p)[y^k(p) - y^k(p_0)][y^l(p) - y^l(p_0)].$$
(9.7)

If  $\Gamma_X(p_0)$  is non-degenerate, then, evidently, (9.6) is the general solution of (9.5) and, consequently, (9.7) is the general solution of (9.3). A method for obtaining the general solution of (9.5), and hence of (9.3), for arbitrary  $\Gamma_X(p_0)$  and continuous X is the following one.<sup>6</sup>

Suppose X is of class  $C^0$ . Since  $X_{p_0} \neq 0$ , there is a unique integral path  $\gamma: J \to M$  through  $p_0$ , i.e., (see equation (I.2.19) on page 13)  $\dot{\gamma}(s) = X_{\gamma(s)}$ ,  $s \in J$ , and  $\gamma(s_0) = p_0$  for some fixed  $s_0 \in J$ . In Subsection II.3.2, we saw that there is a chart  $(U_1, x)$  with  $p_0 \in U_1 \subseteq U$  and coordinates functions (II.3.12) in which the coordinates of  $\gamma(s)$  are given via (II.3.13). Therefore in  $(U_1, x)$  we have  $X_{\gamma(s)}^k = \dot{\gamma}^k(s) = \delta_1^k$  for  $s \in J_1$ , where  $J_1 \subseteq J$  is such that  $J_1 \ni s_0$  and  $\gamma(J_1) = \gamma(J) \cap U_1$ . Hence

$$X_{\gamma(s)} = \dot{\gamma}(s) = \frac{\partial}{\partial x^1}\Big|_{\gamma(s)},$$

and, in particular,  $X_{p_0} = \frac{\partial}{\partial x^1}\Big|_{p_0}$ .

Repeating the aforesaid with  $(U_1.x)$  for (U, y), we get

$$\Gamma_X(p_0)A_0^{(x)} + A_1^{(x)} = 0 \tag{9.5'}$$

<sup>&</sup>lt;sup>5</sup>This solution was first derived in [84, Section IV].

<sup>&</sup>lt;sup>6</sup>If X is not of class  $C^0$  in a neighborhood of  $p_0$  and  $\det(\Gamma_X(p_0)) = 0$ , the methods of matrix algebra should be used for explicit solving of (9.5). Equivalently, one can take any  $C^1$  vector field Y with  $Y_{p_0} = X_{p_0}$  and  $\Gamma_Y(p_0) = \Gamma_X(p_0)$  for X in the next lines.

#### 9. Derivations along fixed vector field

instead of (9.5), with the index (x) indicating that the corresponding quantities are computed in  $\{\frac{\partial}{\partial x^i}\}$ . So  $A_1^{(x)} = -\mathbf{\Gamma}_X(p_0)A_0^{(x)}$ , with arbitrary non-degenerate  $A_0^{(x)}$ , is the general solution of (9.5). Consequently, the general solution of (9.3) in  $(U_1, x)$  is (see (9.4))

$$A^{(x)}(p) = A_0^{(x)} - \Gamma_X(p_0) A_0^{(x)} [x^1(p) - x^1(p_0)] + \sum_{k=2}^{\dim M} A_k^{(x)} [x^k(p) - x^k(p_0)] + A_{kl}^{(x)}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]$$
(9.8)

where  $A_0^{(x)}, A_2^{(x)}, \ldots, A_{\dim M}^{(x)}$  are constant matrices, the matrix  $A_0^{(x)}$  is non-degenerate, and the matrix-valued functions  $A_{kl}$  on  $U_1$  are of class  $C^1$  and they and their first partial derivatives are bounded at  $p_0$ .

The problem for existence and description of charts (or coordinates) normal at a single point  $p_0$  can be attacked by two independent but equivalent ways. The first one is to select the free parameters in (9.8) so that there to exist local coordinates  $\{z^i\}$  with  $\frac{\partial}{\partial z^i} = E'_i = (A^{(x)}(p))^j_i \frac{\partial}{\partial x^j}$ . The second one, which we will developed below, is to solve directly the normal frame equation (9.2') for  $U = \{p_0\}$ , i.e.,

$$X^{m}(p_{0})\frac{\partial^{2}z^{i}}{\partial y^{m}\partial y^{k}}\Big|_{p_{0}} - \frac{\partial z^{i}}{\partial y^{m}}\Big|_{p_{0}}\Gamma_{X}{}^{m}{}_{k}(p_{0}) = 0$$

$$(9.9)$$

where  $\{y^i\}$  are some given local coordinates in a neighborhood of  $p_0$ , the quantities  $\Gamma_X{}^k{}_m(p_0)$  and  $X^m(p_0)$  are computed in  $\{\frac{\partial}{\partial y^i}\}$ , and  $\{z^i\}$  are the looked for normal coordinates, if any.

**Theorem 9.2.** Let M be a  $C^3$  manifold,  $p_0 \in M$  and  $D_X$  be some derivation along fixed  $X \in \mathfrak{X}(M)$ . If  $p_0$  is a singular point for X, i.e.,  $X_{p_0} = 0$ , then either every chart with  $p_0$  in its domain is normal for  $D_X$  (in a case  $D_X$  is the zero derivation at  $p_0$ ) or charts normal at  $p_0$  for  $D_X$  do not exist (in a case  $D_X$  is not the zero derivation at  $p_0$ ). If  $X_{p_0} \neq 0$ , then for  $D_X$  always exist charts  $(U_z, z)$ , with  $U_z \ni p_0$ , normal at  $p_0$  and the general form of their coordinate functions  $z^i$  is given via equation (9.13) below in which:  $a^i, a^i_j, a^i_{jk} \in \mathbb{K}$ ,  $\det[a^i_j] \neq 0$ ,  $a^i_{jkl}: U_z \cap V \to \mathbb{K}$ are of class  $C^1$  and they and their first partial derivatives are bounded at  $p_0$ ,  $(U_1, x)$ , with  $U_1 \supseteq U_z$ , is the special chart constructed above in this subsection, and  $\Gamma_X$  is the matrix of  $D_X$  in  $\{\frac{\partial}{\partial x^i}\}$ .

Proof. If  $X_{p_0} = 0$ , then (9.9) holds iff  $\Gamma_X(p_0) = 0$  (see also (9.2) for  $U \cap V = \{p_0\}$ ). Hence if  $X_{p_0} = 0$ , then charts normal at  $p_0$  either do not exist or every chart with  $p_0$  in its domain is normal. (Recall, in the last case  $D_X$  must be the zero derivation at  $p_0$ ; see Proposition 9.2.)

Let M be  $C^3$  manifold,  $p_0 \in M$ ,  $X_{p_0} \neq 0$ , and (U, y) be a given chart in M with  $U \ni p_0$ . We search for a chart  $(U_z, z)$  with  $U_z \ni p_0$  which is normal at  $p_0$  for

 $D_X$ . It is valid the Taylor expansion

$$z^{i}(p) = a^{i} + a^{i}_{j}[y^{j}(p) - y^{j}(p_{0})] + a^{i}_{jk}[y^{j}(p) - y^{j}(p_{0})][y^{k}(p) - y^{k}(p_{0})] + a^{i}_{jkl}(p)[y^{j}(p) - y^{j}(p_{0})][y^{k}(p) - y^{k}(p_{0})][y^{l}(p) - y^{l}(p_{0})], \quad (9.10)$$

where  $a^i, a^i_j, a^i_{jk} \in \mathbb{K}$  with  $\det[a^i_j] \neq 0$  and  $a^i_{jkl} \colon U \cap V \to \mathbb{K}$  are of class  $C^1$  and they and their first partial derivatives are bounded at  $p_0$  (see the lines preceding equation (II.3.24) on page 100).

From (9.10), we get  $\frac{\partial z^i}{\partial y^j}\Big|_{p_0} = a_j^i$  and  $\frac{\partial^2 z^i}{\partial y^j \partial y^k}\Big|_{p_0} = 2a_{(jk)}^i$ . Substituting these equalities into (9.9), we find

$$a_j^i \Gamma_X^j{}_k(p_0) - X^m(p_0) 2a_{(mk)}^i = 0, \quad \det[a_j^i] \neq 0.$$
 (9.11)

This equation always has solutions, a class of which for  $\Gamma_X(p_0) \neq 0$  is

$$a_{j}^{i} = c^{i}b_{j}X^{m}(p_{0})\left(b_{l}\Gamma_{X}{}^{l}{}_{m}(p_{0})\right), \quad 2a_{(mk)}^{i} = c^{i}\left(b_{l}\Gamma_{X}{}^{l}{}_{m}(p_{0})\right)\left(b_{n}\Gamma_{X}{}^{n}{}_{k}(p_{0})\right) \quad (9.12)$$

where  $b_i, c^i \in \mathbb{K}$  are such that  $\det[a_j^i] \neq 0$ . The general solution of (9.11) can be found analogously to the one of (9.5) above. For this end, we take for (U, y) the afore-constructed chart  $(U_1, x)$  in which  $X = \frac{\partial}{\partial x^1}$  along the integral path of Xthrough  $p_0$  in  $U_1$ . In it (9.11) reduces to  $2a_{(1k)}^i = a_l^i \Gamma_X^{\ l}{}_k(p_0)$ ,  $\det[a_j^i] \neq 0$ , the remaining quantities being left arbitrary. Consequently, due to (9.10), the general form of the looked for coordinates  $\{z^i\}$  normal at  $p_0$  is

$$z^{i}(p) = a^{i} + a^{i}_{j}[x^{j}(p) - x^{j}(p_{0})] + a^{i}_{l}\Gamma_{X}{}^{l}_{k}(p_{0})[x^{k}(p) - x^{k}(p_{0})][x^{1}(p) - x^{1}(p_{0})] + \sum_{j,k=2}^{\dim M} a^{i}_{jk}[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})] + a^{i}_{jkl}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})].$$
(9.13)

Since the Jacobian of the transformation  $\{x^i\} \to \{z^i\}$  at  $p_0$  is  $\det\left[\frac{\partial z^i}{\partial x^j}\Big|_{p_0}\right] = \det[a^i_j] \neq 0$ , there is a subneighborhood  $U_z \subseteq U_1$  with  $U_z \ni p_0$  in which  $x^i$  are well-defined coordinate functions of a chart  $(U_z, z)$  which is normal by construction.<sup>7</sup>

### 9.2. The case along paths

The first step for generalizing the results form the previous subsection is to investigate the problems concerning frames or coordinates normal along paths for derivations along fixed vector field.

<sup>&</sup>lt;sup>7</sup>The constructed in [80, proof of Proposition 4.1] chart normal at a single point corresponds to the choices  $a_j^i = \delta_j^i$ ,  $a_{jk}^i = 0$  for  $j, k \ge 2$ , and  $a_{jkl}^i \equiv 0$ .

#### 9. Derivations along fixed vector field

In this section, as in Sections II.3, 5 and 8, we shall look on the manifold M as on a real one, i.e., if M is real, no changes are necessary and, if M is complex, we consider it as a real manifold of dimension  $\dim_{\mathbb{R}} M = 2 \dim M \equiv 2 \dim_{\mathbb{C}} M$  (see p. 7). We shall formally reflected this in writing  $\dim_{\mathbb{R}} M$  instead of  $\dim M$ . Respectively, the Latin indices with not explicitly specified range will run from 1 to  $\dim_{\mathbb{R}} M$  and the ranges of the coordinate homeomorphisms will lie in  $\mathbb{R}^{\dim_{\mathbb{R}} M}$ .

Suppose  $\gamma: J \to M$  is a locally injective  $C^1$  regular path in a  $C^2$  manifold M endowed with  $C^0$  derivation  $D_X$  along a fixed non-zero vector field X of class  $C^0$ , i.e.,  $X \in \mathfrak{X}^0(M)$  with  $X \neq 0$ . The problem is to study the equation (9.1) for  $U = \gamma(J)$ , i.e.,

$$\Gamma_X(\gamma(s))A(\gamma(s)) + X(A)|_{\gamma(s)} = 0$$
(9.14)

for  $s \in J$ . The idea for investigating (9.14) is to be constructed a chart along  $\gamma$  such that the first (or the first two) coordinate(s) of a point in its domain to be equal to some value of the parameter of  $\gamma$  (or to it and a function of the parameter of an integral path of X) and in which X coincides with or is proportional to certain basic coordinate vector.

**Theorem 9.3.** Let M be  $C^3$  manifold (which should be considered as a real one of dimension  $\dim_{\mathbb{R}} M$  if it is complex),  $\gamma: J \to M$  be locally injective  $C^1$  and regular,  $s_0 \in J$ , and  $D_X$  be a  $C^1$  derivation along a fixed  $C^0$  vector field X on M. If  $\gamma(s_0)$  is a singular point for  $X, X|_{\gamma(s_0)} = 0$ , then either every frame in a neighborhood of  $\gamma(s_0)$  is normal at  $\gamma(s_0)$  for  $D_X$  (in which case  $D_X$  is the zero derivation at  $\gamma(s_0)$ ) or such frames do not exist (in which case  $D_X$  is not the zero derivation at  $\gamma(s_0)$ ). If  $\gamma(s_0)$  is not a singular point for  $X, X|_{\gamma(s_0)} \neq 0$ , then there exist a subinterval  $J_1 \ni s_0$  of J and a frame  $\{E'_i\}$  such that  $\{E'_i\}$  is normal for  $D_X$  along  $\gamma|_{J_1}$ , i.e., on  $\gamma(J_1)$ . Moreover, there is a neighborhood  $V_1$  of  $\gamma(J_1)$  on which all frames normal for  $D_X$  are  $\{E'_i = A^j_i \frac{\partial}{\partial x^j}\}$ , where  $A = [A^j_i] \colon V_1 \to \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R}), V_1 \subseteq U_1$ , and  $(U_1, x)$  is a chart of M with associated coordinates  $\{x^i\}$ . The chart  $(U_1, x)$ , transformation matrix A, and neighborhood  $V_1 \supseteq J_1$  valid or not:

- (a) If (9.15) holds,  $(U_1, x)$  is the special chart described in Subsection II.3.2 for the path  $\gamma|_{J_1}$ , A is given by equation (9.19) below, and  $V_1 = x^{-1}(J_1, W)$  where W is a neighborhood of  $\mathbf{t}_0$  in  $\mathbb{R}^{\dim_{\mathbb{R}} M-1}$  on which  $A_0$  is non-degenerate;
- (b) If (9.15) is not valid, (U<sub>1</sub>, x) is the special chart constructed at the beginning of Subsection 8.1 for the mapping α, introduced below on page 181 in the proof of this theorem, A is given by equation (9.22) below, and V<sub>1</sub> = x<sup>-1</sup>(J<sub>1</sub>, J'<sub>0</sub>, ℝ<sup>dim<sub>ℝ</sub>M-2</sup>) where J'<sub>0</sub> is such that A|<sub>γ(J'<sub>0</sub>)</sub> is non-degenerate.

*Proof.* If  $X|_{\gamma(s)} = 0$  for some  $s \in J$ , then (9.14) implies  $\Gamma_X(\gamma(s)) = 0$  as A is non-degenerate (cf. a similar situation explored in Subsection 9.1). Hence at the singular points of X along  $\gamma$ , if any, either every frame is normal ( $D_X$  is the zero derivation at them) or frames normal at them do not exist ( $D_X$  is not the zero derivation at them). In particular, if X is singular along  $\gamma$ ,  $X|_{\gamma(J)} = 0$ , then either every frame along  $\gamma$  is normal for  $D_X$  or such frames do not exist at all. Below we shall suppose the existence of  $s_0 \in J$  such that  $X_{\gamma(s_0)} \neq 0$ , i.e., that X is not singular along the whole path  $\gamma$ . Since X is of class  $C^0$ , there is a neighborhood  $U_0$  of  $\gamma(s_0)$  in which X is non-singular,  $X|_{U_0} \neq 0$ .

Let  $\beta_p: J'_p \to M$ ,  $p \in \gamma(J)$  be the integral path of X through p (see p. 13), i.e.,  $\beta_p(\sigma_{p,0}) = p \in \gamma(J)$  for some  $\sigma_{p,0} \in J'_p$  and  $\dot{\beta}_p(\sigma_p) = X_{\beta_p(\sigma_p)}, \sigma_p \in J'_p$ . For the path  $\beta_{\gamma(s_0)}$  (in a neighborhood of  $\gamma(s_0)$ ) there are two possibilities: it (locally) intersects  $\gamma$  only at  $\gamma(s_0) = \beta_{\gamma(s_0)}(\sigma_{\gamma(s_0),0})$  for fixed  $\sigma_{\gamma(s_0),0} \in J'_{\gamma(s_0)}$  or it is (locally) obtainable from  $\gamma$  by a possible change of its parameter. We shall write that in terms of X. The second possibility means the existence of a subinterval  $J^0$ of J with  $J^0 \ni s_0$  and function  $g: \gamma(J^0) \to \mathbb{R} \setminus \{0\}$  such that  $\gamma(J^0) \subset U_0, \gamma|_{J^0}$  is injective (recall that  $\gamma$  is supposed locally injective), and on  $\gamma(J^0)$  the field X is proportional to the vector field  $\dot{\gamma}$  tangent to  $\gamma$  with factor g,

$$X|_{\gamma(J^0)} = g\dot{\gamma},\tag{9.15}$$

while the first possibility is equivalent to the nonexistence of such connection between X and  $\dot{\gamma}$ . (Notice, (9.15) with g = 1 means that  $\gamma$  is, possibly locally, an integral path for X.)

Let the relation (9.15) hold. Then, by virtue of equation (I.2.3), we have  $X(A)|_{\gamma(s)} = g(\gamma(s))[\dot{\gamma}(s)(A)] = g(\gamma(s))\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s}$ , so (9.14) reduces to

$$\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} + \frac{1}{g(\gamma(s))} \mathbf{\Gamma}_X(\gamma(s)) A(\gamma(s)) = 0, \qquad s \in J^0.$$
(9.16)

By Remark II.3.9 on page 96, the general solution of this equation, defining A only on  $\gamma(J^0)$ , is

$$A(\gamma(s)) = Y\left(s, s_0; -\frac{1}{g \circ \gamma} \Gamma_X \circ \gamma\right) A_0 \tag{9.17}$$

where  $A_0 \in \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R}), s \in J^0, s_0 \in J^0$  is fixed as above, and Y is the unique solution of the initial-value problem

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = -\frac{1}{g(\gamma(s))} \mathbf{\Gamma}_X(\gamma(s)) Y, \quad Y|_{s=s_0} = \mathbb{1}.$$
(9.18)

If  $\dim_{\mathbb{R}} M = 1$ , which is possible only if M is real, then (9.15) always holds<sup>8</sup> and (9.17) gives the general solution of (9.14) with respect to A in  $\gamma(J^0) \ni \gamma(s_0)$ .<sup>9</sup> For  $\dim_{\mathbb{R}} M \ge 2$ , the general form of A on some neighborhood of  $\gamma(s_0)$  could be found as follows. Since the integral paths are without self-intersections, i.e.,

<sup>&</sup>lt;sup>8</sup>For dim<sub>R</sub> M = 1, the vector  $\dot{\gamma}(s) \neq 0$ ,  $s \in J^0$ , can be taken as a basis of the one-dimensional tangent space  $T_{\gamma(s)}(M)$ , respectively  $\{\dot{\gamma}\}$  is a frame on  $\gamma(J^0)$ . Therefore (9.15) is simply the expansion of X over  $\{\dot{\gamma}\}$  in this case.

<sup>&</sup>lt;sup>9</sup>For dim<sub>R</sub> M = 1, the 1 × 1 matrix  $\Gamma_X(\gamma(s))$  is simply a number, so that the explicit form of Y is  $Y(s, s_0; -\frac{1}{g \circ \gamma} \Gamma_X \circ \gamma) = \exp\left(-\int_{s_0}^s \frac{\Gamma_X(\gamma(\sigma))}{g(\gamma(\sigma))} \,\mathrm{d}\sigma\right).$ 

injective [9, 11], such is the restricted path  $\gamma|_{J^0}$ . Construct for  $\gamma|_{J^0}$  the special chart  $(U_1, x)$  described in Subsection II.3.2 (see p. 94 with  $\gamma|_{J^0}: J^0 \to M$  for  $\gamma$ ). Then, for  $p \in U_1$ , we have x(p) = (s, t) for some  $s \in J_1 \subseteq J^0$  (recall that  $\gamma(J_1) = U_1 \cap \gamma(J^0)$ ) and  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$  and  $x(\gamma(s)) = (s, t_0)$  for a fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$ . Expanding A as

$$A(x^{-1}(s, t)) = A(\gamma(s)) + \sum_{k=2}^{\dim_{\mathbb{R}} M} A_k(x^{-1}(s, t))(t^k - t_0^k)$$

and using (9.17), we, finally, get the general solution of (9.14) in  $U_1$  as

$$A(x^{-1}(s, t)) = Y(s, s_0; -\frac{1}{g \circ \gamma} \Gamma_X \circ \gamma) A_0 + \sum_{k=2}^{\dim_{\mathbb{R}} M} A_k(x^{-1}(s, t))(t^k - t_0^k), \quad (9.19)$$

where  $A_0$  is constant non-degenerate matrix and the matrix-valued functions  $A_2, \ldots, A_{\dim_{\mathbb{R}} M}$  on  $U_1$  are of class  $C^1$  and they and their first derivatives are bounded when  $\mathbf{t} \to \mathbf{t}_0$ , i.e., on  $\gamma(J_1)$ . Since A is continuous (it is supposed of class  $C^1$ ), there is a neighborhood of  $\mathbf{t}_0$  in  $\mathbb{R}^{\dim_{\mathbb{R}} M-1}$  in which  $A(x^{-1}(s, \mathbf{t}))$  is non-degenerate.

Let us now suppose dim<sub>R</sub>  $M \geq 2$  and the integral path  $\beta_{\gamma(s_0)} \colon J'_{\gamma(s_0)} \to M$ to intersect  $\gamma \colon J \to M$  at the single point  $\gamma(s_0) = \beta_{\gamma(s_0)}(\sigma_{\gamma(s_0),0})$  for some unique  $\sigma_{\gamma(s_0),0} \in J'_{\gamma(s_0)}$ .<sup>10</sup> Since  $\gamma$  is locally injective, there is a subinterval  $J_3 \subseteq J$  with  $J_3 \ni s_0$  such that  $\gamma|_{J_3} \colon J_3 \to M$  is without self-intersections. Let  $J_2 := \{s|s \in J_3, \gamma(s) \in U_0\}$ . Then  $\gamma|_{J_2}$  is injective path and along it X does not vanish. At last, since X and  $\gamma$  are continuous, there exists a subinterval  $J_0 \subseteq J_2$  with  $J_0 \ni s_0$ such that the integral path  $\beta_{\gamma(s)}, s \in J_2$ , intersects  $\gamma|_{J_0}$  only once (at the point  $\gamma(s) = \beta_{\gamma(s)}(\sigma_{\gamma(s),0})$ ). A 'local' solution of (9.14) for the restricted path  $\gamma|_{J_0}$  can be found as follows.

Let, for every  $p \in \gamma(J)$ , there be fixed (chosen) some  $C^1$  bijective mapping  $(C^1$  diffeomorphism)  $\tau_p \colon J' \to J'_p$  from some open  $\mathbb{R}$ -interval J' to the domain of the integral path  $\beta_p$  such that  $\tau_p(\sigma_0) = \sigma_{p,0}$  for fixed  $\sigma_0 \in J'$ .<sup>11</sup> Consider a mapping  $\alpha \colon J_0 \times J' \to M$  with  $\alpha(s, \sigma) \coloneqq \beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma))$  for  $(s, \sigma) \in J_0 \times J'$ . Due to the aforesaid,  $\alpha$  is injective, of class  $C^1$  and regular. Consequently, for it are applicable the results at the beginning of Subsection 8.1 with  $\alpha$ , 2, and  $J_0 \times J'$  for  $\gamma$ , n, and  $J^n$  respectively. Therefore for  $\alpha$  can be constructed a chart  $(U_1, x)$ , explicitly given by (8.3) for n = 2,  $\gamma = \alpha$ , and  $J_1^2 \subseteq J^2 = J_0 \times J'$ , with  $U_1 \ni \alpha(s_0, \sigma_0) = \gamma(s_0)$  and  $U_1 \subseteq U_0$ , such that there is a subneighborhood  $J_1^2 \coloneqq J_1 \times J'^0 \subseteq J_0 \times J'$  with  $\alpha(J_1^2) = U_1 \cap \gamma(J_2)$ , the coordinates of  $p \in U_1$  are

<sup>&</sup>lt;sup>10</sup>This is only a local property if (9.15) does not hold. But in this way we do not loose generality since if  $\sigma_{-} \leq \sigma_{\gamma(s_0),0}$  and  $\sigma_{+} \geq \sigma_{\gamma(s_0),0}$  are respectively the maximal and minimal elements of the set  $\{\sigma | \sigma \in J'_{\gamma(s_0)}, \beta_{\gamma(s_0)}(\sigma_{\gamma(s_0)}) = \gamma(s) \text{ for some } s \in J\} \ni \sigma_{\gamma(s_0),0} \text{ and } \sigma_{-} \neq \sigma_{+}, \text{ then we can take the interval } (\sigma_{-}, \sigma_{+}) \ni \sigma_{\gamma(s_0),0} \text{ for } J'_{\gamma(s_0)}.$ 

<sup>&</sup>lt;sup>11</sup>Since all open real intervals are ( $C^{\infty}$ -) diffeomorphic, mappings like  $\tau_p$  always exist.

 $x(p) = (s, \sigma, t)$  for some  $(s, \sigma) \in J_1^2$  and  $t \in \mathbb{K}^{\dim_{\mathbb{R}} M - 2}$ , and  $x(\alpha(s, \sigma)) = (s, \sigma, t_0)$  for a fixed  $t_0 \in \mathbb{K}^{\dim_{\mathbb{R}} M - 2}$  (see (8.4)).

Since in the special chart  $(U_1, x)$  we have

$$x(\beta_{\gamma(s)}(\sigma_{\gamma(s)})) = x(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma))) = x(\alpha(s,\sigma)) = (s,\sigma,t_0)$$

for  $\sigma_{\gamma(s)} = \tau_{\gamma(s)}(\sigma) \in J'_s, \ \sigma \in J'$ , the components of  $X_{\beta_{\gamma(s)}(\sigma_{\gamma(s)})}$  in the frame  $\left\{\frac{\partial}{\partial x^i}\right\}$  are  $X_{\beta_{\gamma(s)}(\sigma_{\gamma(s)})} = \frac{\partial(x^k(\beta_{\gamma(s)}(\sigma_{\gamma(s)})))}{\partial \sigma_{\gamma(s)}} = \frac{\partial(\delta_2^k \sigma)}{\partial \sigma_{\gamma(s)}} = \delta_2^k \left(\frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}\right)^{-1}$ . Hence  $X_{\beta_{\gamma(s)}(\sigma_{\gamma(s)})} = \frac{1}{\frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}} \frac{\partial}{\partial x^2}\Big|_{\beta_{\gamma(s)}(\sigma_{\gamma(s)})}.$  (9.20)

So, in  $(U_1, x)$ , i.e., for  $s \in J_1$  and  $E_i = \frac{\partial}{\partial x^i}$ , the basic equation (9.14) reads

$$\frac{\partial A(x^{-1}(s,\sigma,\boldsymbol{t}))}{\partial \sigma}\Big|_{\substack{\sigma=\sigma_0\\\boldsymbol{t}=\boldsymbol{t}_0}} + \frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}\Big|_{\sigma=\sigma_0} \mathbf{\Gamma}_X(\gamma(s))A(\gamma(s)) = 0.$$
(9.21)

The general solution of this equation with respect the matrix-valued function  $A: U_1 \to \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$  can be found by expanding A with respect to  $(\sigma - \sigma_0)$  up to second order terms. In this way, we find

$$A(x^{-1}(s,\sigma,\boldsymbol{t})) = A_0(x^{-1}(s,\sigma_0,\boldsymbol{t})) - \frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}\Big|_{\sigma=\sigma_0} \mathbf{\Gamma}_X(\gamma(s))A_0(\gamma(s))(\sigma-\sigma_0) + B(x^{-1}(s,\sigma,\boldsymbol{t}))(\sigma-\sigma_0)^2$$
(9.22)

where  $A_0: x^{-1}(J_1, \sigma_0, \mathbb{R}^{\dim M-2}) \to \operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$  and the matrix-valued function B on  $U_1$  is of class  $C^1$  and it and its first partial derivatives are bounded on  $\gamma(J_1)$ , i.e., when  $\sigma \to \sigma_0$  and  $t \to t_0$ . Since we have  $\det A(x^{-1}(s, \sigma_0, t)) =$  $\det A_0(x^{-1}(s, \sigma_0, t)) \neq 0, \infty$  and A is continuous (as it is of class  $C^1$ ), there is a subinterval  $J'_0 \subseteq J'$  containing  $\sigma_0$  such that  $A: x^{-1}(J_1, J'_0, \mathbb{R}^{\dim M-2}) \to$  $\operatorname{GL}(\dim_{\mathbb{R}} M, \mathbb{R})$ . This last result completes the proof of Theorem 9.3.  $\Box$ 

The Theorem 9.3 provides a complete local description of the frames normal along a path  $\gamma: J \to M$  for a derivation  $D_X$  along fixed vector field  $X \in \mathfrak{X}(M)$ at the points in which X does not vanish. In particular, if  $X_{\gamma(J)} \neq 0$ , then every point along  $\gamma$  has a neighborhood on which frames normal along  $\gamma$  exist and from these frames all frames globally normal along  $\gamma$  can be constructed.

In a neighborhood of every point  $\gamma(s_0)$ ,  $s_0 \in J$ , for which  $X_{\gamma(s_0)} \neq 0$  coordinates  $\{z^i\}$  normal along  $\gamma$  exist. They can be found by integrating the system  $\frac{\partial}{\partial z^i} = A_i^j \frac{\partial}{\partial x^j}$  for some A given by (9.19) or by (9.22) (with corresponding  $\{x^i\}$  as specified above) or via direct solving of the normal frame equation (9.2') along  $\gamma$ , viz. of

$$X^{m}(\gamma(s))\frac{\partial^{2} z^{i}}{\partial x^{m} \partial x^{k}}\Big|_{\gamma(s)} + \frac{\partial z^{i}}{\partial x^{m}}\Big|_{\gamma(s)} \Gamma_{X}{}^{m}{}_{k}(\gamma(s)) = 0.$$
(9.23)

Since the results obtained considerably simplify the second method, we shall follow it. **Proposition 9.4.** Let M be  $C^3$  manifold,  $\gamma: J \to M$  be locally injective,  $C^1$  and regular path, and  $D_X$  be  $C^1$  derivation along a fixed  $X \in \mathfrak{X}^0(M)$ . For every  $s_0 \in J$  for which  $X|_{\gamma(s_0)} \neq 0$ , there exist local coordinates in a neighborhood of  $\gamma(s_0)$  which are normal along  $\gamma$ . Their general form is given by equation (9.24) below under the conditions (9.26) (resp. (9.27)) below if (9.15) holds (resp. does not hold).

*Proof.* Let  $s_0 \in J$ ,  $X_{\gamma(s_0)} \neq 0$ , and  $(U_1, x)$  with  $U_1 \ni \gamma(s_0)$  be the special chart constructed above with reference mapping  $\gamma|_{J_0}$  or  $\alpha$  if (9.15) holds or not, respectively. We shall look for a chart  $(U_z, z)$  with  $\gamma(s_0) \in U_z \subseteq U_1$  in which (9.23) is valid. Since M is supposed of class  $C^3$ , we can write the expansion

$$z^{i}(p) = a^{i}(p_{0}) + a^{i}_{j}(p_{0})[x^{j}(p) - x^{j}(p_{0})] + a^{i}_{jk}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})].$$
(9.24)

Here:  $p \in U_z$ ,  $p_0$  is the 'projection' of p along  $\gamma$  or  $\alpha$  (i.e.,  $p_0 = x^{-1}(s, t_0) = \gamma(s)$  or  $p_0 = x^{-1}(s, \sigma, t_0) = \beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma))$  if  $p = x^{-1}(s, t)$  or  $p = x^{-1}(s, \sigma, t)$  respectively),  $a^i$  and  $a^i_j$  are  $C^1$  functions of  $p_0$  and  $a^i_{jk} \colon U_z \to \mathbb{R}$  are  $C^1$  and they and their first partial derivatives are bounded when  $p \to p_0$ .

In the chart  $(U_1, x)$ , the fixed vector field X has the expansion  $X|_{\gamma(s)} = g(\gamma(s))\frac{\partial}{\partial x^1}|_{\gamma(s)}$  or (9.20) if (9.15) holds or not, respectively. So, in these two cases, equation (9.23) takes respectively the form (cf. (9.16) and (9.21))

$$\frac{\partial^2 z^i}{\partial x^1 \partial x^k} \Big|_{\gamma(s)} - \frac{1}{g(\gamma(s))} \frac{\partial z^i}{\partial x^j} \Big|_{\gamma(s)} \Gamma_X^{\ j}{}_k(\gamma(s)) = 0$$
(9.25a)

$$\frac{\partial^2 z^i}{\partial x^2 \partial x^k} \Big|_{\substack{p=p_0\\\sigma=\sigma_0}} - \frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma} \Big|_{\sigma=\sigma_0} \frac{\partial z^i}{\partial x^j} \Big|_{\substack{p=p_0\\\sigma=\sigma_0}} \Gamma_X{}^j{}_k(\gamma(s)) = 0.$$
(9.25b)

The entering here derivatives of the z's with respect to the x's can be expressed through the unknown functions  $a^i$  and  $a^i_j$  by employing (9.24):

$$\frac{\partial z^{i}}{\partial x^{j}}\Big|_{\gamma(s)} = \begin{cases} \frac{\mathrm{d}a^{i}(\gamma(s))}{\mathrm{d}s} & \text{for } j = 1\\ a^{i}_{j}(\gamma(s)) & \text{for } j \ge 2 \end{cases} \qquad \frac{\partial^{2} z^{i}}{\partial x^{1} \partial x^{k}}\Big|_{\gamma(s)} = \begin{cases} \frac{\mathrm{d}^{2} a^{i}(\gamma(s))}{\mathrm{d}s^{2}} & \text{for } k = 1\\ \frac{\mathrm{d}a^{i}_{k}(\gamma(s))}{\mathrm{d}s} & \text{for } k \ge 2 \end{cases}$$

$$\frac{\partial z^{i}}{\partial x^{j}}\Big|_{p=p_{0}} = \begin{cases} \frac{\partial a^{i}}{\partial s}\Big|_{p=p_{0}} & \text{for } j=1\\ \frac{\partial a^{i}}{\partial \sigma}\Big|_{p=p_{0}} & \text{for } j=2\\ a^{i}_{j}(p_{0}) & \text{for } j\geq3 \end{cases} \qquad \frac{\partial^{2}z^{i}}{\partial x^{2}\partial x^{k}}\Big|_{p=p_{0}} = \begin{cases} \frac{\partial^{2}a^{i}}{\partial s\partial \sigma}\Big|_{p=p_{0}} & \text{for } k=1\\ \frac{\partial^{2}a^{i}}{\partial \sigma^{2}}\Big|_{p=p_{0}} & \text{for } k=2\\ \frac{\partial a^{i}_{k}}{\partial \sigma}\Big|_{p=p_{0}} & \text{for } k\geq3 \end{cases}$$

where the first (resp. second) raw is valid when (9.15) holds (resp. does not hold) and we have used that  $p_0 = x^{-1}(s, \mathbf{t}_0) = \gamma(s)$  (resp.  $p_0 = x^{-1}(s, \sigma, \mathbf{t}_0) = \beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma))$ ) if  $p = x^{-1}(s, \mathbf{t})$  (resp.  $p = x^{-1}(s, \sigma, \mathbf{t})$ ). The substitution of these equations into (9.25) results in the following two systems of linear differential equations for the functions  $a^i$  and  $a^i_j$ :

$$\frac{1}{g(\gamma(s))} \left[ \frac{\mathrm{d}a^{i}(\gamma(s))}{\mathrm{d}s} \Gamma_{X}{}^{1}{}_{k}(\gamma(s)) + \sum_{j=2}^{\dim_{\mathbb{R}}M} a^{i}_{j}(\gamma(s)) \Gamma_{X}{}^{j}{}_{k}(\gamma(s)) \right] \\
= \begin{cases} \frac{\mathrm{d}^{2}a^{i}(\gamma(s))}{\mathrm{d}s^{2}} & \text{for } k = 1\\ \frac{\mathrm{d}a^{i}_{k}(\gamma(s))}{\mathrm{d}s} & \text{for } k \geq 2 \end{cases}$$
(9.26)

$$\frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}\Big|_{\sigma=\sigma_{0}} \left[\frac{\partial a^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial s}\Big|_{\sigma=\sigma_{0}}\Gamma_{X}^{1}{}_{k}(\gamma(s)) + \frac{\partial a^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial \sigma}\Big|_{\sigma=\sigma_{0}}\Gamma_{X}^{2}{}_{k}(\gamma(s)) + \frac{\partial a^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial \sigma}\Big|_{\sigma=\sigma_{0}}\Gamma_{X}^{2}{}_{k}(\gamma(s)) + \sum_{j=3}^{\mathrm{dim}_{\mathbb{R}}M}a_{j}^{i}(\gamma(s))\Gamma_{X}{}^{j}{}_{k}(\gamma(s))\Big] = \begin{cases} \frac{\partial^{2}a^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial \sigma^{2}}\Big|_{\sigma=\sigma_{0}} & \text{for } k=1\\ \frac{\partial^{2}a^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial \sigma^{2}}\Big|_{\sigma=\sigma_{0}} & \text{for } k=2 \\ \frac{\partial a_{k}^{i}(\beta_{\gamma(s)}(\tau_{\gamma(s)}(\sigma)))}{\partial \sigma}\Big|_{\sigma=\sigma_{0}} & \text{for } k\geq3 \end{cases}$$

We are not going to solve these systems of differential equations, but we must note that they always have solutions according to the general theorems of the theory of differential equations [34]. The substitution of these solutions into (9.24) gives the corresponding coordinates normal along the path  $\gamma$ . The domain  $U_z$  of the normal chart  $(U_z, z)$  should be chosen such that in it the Jacobian det $\left[\frac{\partial z^i}{\partial x^j}\right]$ be non-vanishing.

**Problem 9.1.** Explore the system (9.26)-(9.27) of differential equations and find its general solution.

## 9.3. The case on the whole manifold

The purpose of this section is to be shown that frames normal for a derivation along fixed vector field exist on the whole manifold. The general form of these frames will be found too.

Let M be  $C^1$  manifold,  $X \in \mathfrak{X}^0(M)$ ,  $X \neq 0$ ,  $\{E_i\}$  be a frame on M, and  $D_X$  be a  $C^0$  derivation along X. As we know, a frame  $\{E'_i = A^j_i E_j\}$  is normal for  $D_X$  (on M) iff (9.1) holds for U = M,

$$\Gamma_X A + X(A) = 0 \tag{9.28}$$

where  $\Gamma_X$  is the matrix of  $D_X$  in  $\{E_i\}$ . We intend to demonstrate that this equation always has (local) non-degenerate solutions (at the non-singular points for X) in contrast to the similar (system of) equation(s) for linear connections (Section II.4) or for derivations along vector fields (Section 6).

**Theorem 9.4.** Let M be  $C^1$  manifold, X be continuous vector field on M, X be not identically zero on M, and  $D_X$  be a  $C^0$  derivation along X. For every  $p_0 \in M$ for which  $X_{p_0} \neq 0$ , there exists a chart (U, x) with associated coordinate system  $\{x^i\}$  such that:  $U \ni p_0$ ,  $X|_U = \frac{\partial}{\partial x^i}|_U$ , there exist frames normal on U for  $D_X$ and all such frames are  $\{E'_i = A^j_i \frac{\partial}{\partial x^j}\}$  where  $A = [A^j_i] \colon U \to \operatorname{GL}(\dim M, \mathbb{R})$ is given by equation (9.31) below in which Y is defined by (9.32) below and  $A_0$ is a non-degenerate matrix-valued function. If  $p_0 \in M$  and  $X_{p_0} = 0$ , at  $p_0$  exist frames normal for  $D_X$  iff  $D_X$  is the zero derivation at  $p_0$ ,  $(D_X K)|_{p_0} = 0$  for every tensor field K, in which case every frame (in a neighborhood of  $p_0$ ) is normal at  $p_0$  for  $D_X$ .

*Proof.* Let us choose some  $p_0 \in M$  such that  $X_{p_0} \neq 0$ . According to [7, Proposition 1.53], there is a chart (U, x) of M with  $U \ni p_0$  and local coordinates  $\{x^i\}$  for which<sup>12</sup>

$$X|_U = \frac{\partial}{\partial x^1}\Big|_U \neq 0. \tag{9.29}$$

So, in (U, x) the normal frame equation (9.28) (on U) with  $E_i = \frac{\partial}{\partial x^i}$  reads

$$\frac{\partial A}{\partial x^1}\Big|_p = -\Gamma_X(p)A(p). \tag{9.30}$$

Its general solution on U is

$$A(p) = Y(x^{1}(p), x^{1}(p_{0}); -\Gamma_{X}(x^{-1}(\cdot, x^{2}(p), \dots, x^{\dim M}(p)))) \times A_{0}(x^{2}(p), \dots, x^{\dim M}(p)), \quad (9.31)$$

where  $p \in U$ , the matrix-valued function  $Y = Y(x^1(p), x^1(p_0); Z)$ , with continuous matrix-valued function Z on  $x^1(U)$ , is the unique solution of the initial-value problem (see Lemma II.3.2 on page 96 or Lemma II.4.1 on page 105)

$$\frac{\partial Y}{\partial x^1}\Big|_p = ZY, \qquad Y|_{x^1(p)=x^1(p_0)} = \mathbb{1}$$
(9.32)

and  $A_0$  is non-degenerate matrix-valued function of its arguments.

This ends the proof of the first part of the theorem. The second one is a consequence of Proposition 9.2 (for  $U = \{p_0\}$ ).

The Theorem 9.4 gives a complete local description of all frames normal on M. If X is non-zero everywhere on M or if  $D_X$  is the zero derivation at the points at which X vanishes, all frames globally normal on the whole manifold M for  $D_X$  can be constructed from the local normal frames provided by Theorem 9.4. For this end, a method similar to the one described in Subsection II.5.2 should be applied (see page 123).

<sup>&</sup>lt;sup>12</sup>This is a special case of the theorem for straightening (of integral paths); see [91, p. 121].

If (U, x) is the chart provided by Theorem 9.4, the coordinate functions  $z^i$  of every normal chart (V, z) with  $p \in V \subseteq U$  are solutions of (9.2') with  $x'^i = z^i$  and  $X^m = \delta_1^m$ :

$$\left(\frac{\partial^2 z^i}{\partial x^1 \partial x^k} - \frac{\partial z^i}{\partial x^j} \Gamma_X^{\ j}{}_k\right)\Big|_V = 0.$$
(9.33)

According to the theorems of the theory of (linear partial) differential equations [34], this equation always has (local) solutions with respect to  $z^i$ . The general solution of (9.33) can not be expressed in a closed form for arbitrary  $\Gamma_X$ . But if M and  $D_X$  are of class  $C^{\infty}$ , it can be found by expanding  $z^i = z^i(x^1, \ldots, x^{\dim M})$ and  $\Gamma_X{}^i{}_k$  in power series with respect to  $[x^i(p) - x^i(p_0)]$ ,

$$z^{i}(p) = a^{i} + \sum_{n=1}^{\infty} a^{i}_{i_{1}\dots i_{n}} [x^{i_{1}}(p) - x^{i_{1}}(p_{0})] \cdots [x^{i_{n}}(p) - x^{i_{n}}(p_{0})]$$
(9.34a)

$$\Gamma_X{}^j{}_k(p) = \gamma_k^j + \sum_{n=1}^{\infty} \gamma_{ki_1\dots i_n}^j [x^{i_1}(p) - x^{i_1}(p_0)] \cdots [x^{i_n}(p) - x^{i_n}(p_0)]$$
(9.34b)

where the *a*'s and  $\gamma$ 's are constant numbers symmetric in  $i_1 \dots i_n$  and det $[a_i^j] \neq 0$ . Indeed, substituting these expansions into (9.33), we get the following infinite system of recurrent equations

$$a_{1k}^{i} = a_{j}^{i} \gamma_{k}^{j}$$

$$a_{1ki_{2}}^{i} = a_{ji_{2}}^{i} \gamma_{k}^{j} + a_{j}^{i} \gamma_{ki_{2}}^{j}$$

$$a_{1ki_{2}i_{3}}^{i} = \left(a_{ji_{2}i_{3}}^{i} \gamma_{k}^{j} + a_{ji_{2}}^{i} \gamma_{ki_{3}}^{j} + a_{j}^{i} \gamma_{ki_{2}i_{3}}^{j}\right)_{(i_{2}i_{3})}$$

$$(9.35)$$

From these equalities and (9.34a), the general solution of (9.33) can be found. Since  $\frac{\partial z^i}{\partial x^j} = a_j^i$  and  $[a_j^i]$  is non-degenerate, the obtained series for  $z^i$  is convergent in some neighborhood  $V \subseteq U$  of  $p_0$  and the change  $\{x^i\} \to \{z^i\}$  is well-defined in V, i.e., it is with non-zero Jacobian in V. In this way all charts (V, z) normal for the initial derivation  $D_X$  can be found in a neighborhood of every point  $p_0 \in M$ at which  $X_{p_0} \neq 0$ . (Recall, if  $X_{p_0} = 0$  and  $D_X$  is the zero derivation at  $p_0$ , then every chart in a neighborhood of  $p_0$  is normal at  $p_0$  for  $D_X$ , but if  $X_{p_0} = 0$  and  $D_X$  is not the zero derivation at  $p_0$ , then charts (and frames) normal at  $p_0$  for  $D_X$ do not exist.)

## 9.4. Other cases

Following the methods of Subsections 9.1–9.3 and the ideas of Sections 4–8, one can try to attack the problems for existence and construction of frames and coordinates normal on submanifolds or along (locally injective) mappings for derivations along fixed vector field. The main obstacles on this way are the complicated systems of partial differential equations one gets, a typical example of which are (9.26) and (9.27).<sup>13</sup> For this reason, we are not going to study these situations here and want only to make some comments on them.

**Proposition 9.5.** Let M be  $C^1$  manifold, U be any subset of M, X be continuous vector field on M, and  $D_X$  be derivation along X. If  $p \in U$  and  $X_p \neq 0$ , then there is a subset  $U_p$  of U containing p on which frames normal for  $D_X$  exist. If  $p \in U$  and  $X_p = 0$ , then at p exist frames normal for  $D_X$  iff  $D_X$  is the zero derivation at p and if this is the case, every frame defined in a neighborhood of p is normal at p for  $D_X$ .

*Proof.* If  $p \in U \subset M$  and  $X_p \neq 0$ , by Theorem 9.4 exist frames normal on some neighborhood  $U_0$  of p in M (and all of them are described by this theorem). If  $\{E_i\}$  is such a frame, then, obviously, it is normal for  $D_X$  on  $U_p := U_0 \cap U \ni p$  as  $U_p \subseteq U$ , which completes the proof of the first part of the theorem. The second part of its assertion is a consequence of Proposition 9.2.

**Corollary 9.1.** If N is a submanifold of a  $C^1$  manifold M,  $X \in \mathfrak{X}^0(M)$ ,  $X \neq 0$ , and  $D_X$  is a derivation along X, then for every  $p \in N$  for which  $X_p \neq 0$  exists a neighborhood  $U_p^N$  of p in N on which frames normal for  $D_X$  exist.

*Proof.* See Proposition 9.5 or its proof for U = N and put  $U_p^N = U_p$ .

**Corollary 9.2.** Let  $f: N \to M$  be locally injective mapping between  $C^1$  manifolds N and  $M, X \in \mathfrak{X}^0(M), X \neq 0$ , and  $D_X$  be derivation along X. For every  $(\dim N)$ -dimensional submanifold N' of N on which f is injective and  $X|_{f(N')} \neq 0$  there exist frames normal along  $f|_{N'}$  for  $D_X$ .

*Proof.* See Proposition 9.5 and construct, if required, a frame globally normal along  $f_{N'}$  from the local frames normal on the constituents of an open cover of f(N') on each of which frames normal along f exist.

Applying some freedom of the language, we may paraphrase Corollaries 9.1 and 9.2 by saying that on any  $C^1$  submanifold N of a  $C^1$  manifold M or along a locally injective mapping f with range in M, frames locally normal on N or along f exist for every derivation along a fixed vector field X provided X does not vanish on N or on the range of f, respectively.

An important remark in connection with Proposition 9.5 and Corollaries 9.1 and 9.2 should be made. We proved the existence of frames (locally) normal on  $U \subseteq M$  by restricting to (subsets of) U the frames locally normal on the *whole* manifold M. All frames locally normal on M are described by Theorem 9.4 and their restrictions to (subsets of) U are, of course locally normal on U. The essential moment is that if  $U \neq M$ , then, generally, exist frames locally normal on U which are *not* (locally) normal on M. In other words, the set of frames locally normal on

<sup>&</sup>lt;sup>13</sup>In the general case, the analogues of (9.26) (resp. (9.27)) correspond to the cases when the fixed vector field X lies (resp. does not lie) in the tangent space of the corresponding submanifold considered as a manifold.

U is larger than the one consisting of locally normal frames obtained by frames (locally) normal on M via restriction to (subsets of) U.

An example of these more or less intuitively clear assertions is provided by a comparison of Theorems 9.4 and 9.1, the latter corresponding to  $U = \{p_0\}$ , with  $p_0 \in M$  and  $X_{p_0} \neq 0$  in the above notation. Since in the both theorems  $x^i$ are coordinates in which  $X = \frac{\partial}{\partial x^1}$  (on M or at  $p_0$  resp.), one can compare the matrices (9.31) and (9.8) by means of which is achieved the transition to normal frames (on M or at  $p_0$  resp.). Take, for simplicity,  $p_0$  in (9.31) and (9.8) to be one and the same point. Expanding the right-hand side of (9.31) with respect to  $x^k(p) - x^k(p_0)$  and taking into account (9.30), we get

$$\begin{aligned} A(p) &= \left\{ \mathbbm{1} - \mathbf{\Gamma}_X(p_0) [x^1(p) - x^1(p_0)] + B(x^1(p)) [x^1(p) - x^1(p_0)]^2 \right\} \\ &\times \left\{ B_0 + \sum_{k=2}^{\dim M} B_k [x^k(p) - x^k(p_0)] \\ &+ \sum_{k,l=2}^{\dim M} B_{kl}(x^2(p), \dots, x^{\dim M}(p)) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)] \right\} \\ &= B_0 - \mathbf{\Gamma}_X(p_0) B_0 [x^1(p) - x^1(p_0)] + \sum_{k=2}^{\dim M} B_k [x^k(p) - x^k(p_0)] \\ &+ B(x^1(p)) B_0 [x^1(p) - x^1(p_0)]^2 \\ &- \mathbf{\Gamma}_X(p_0) [x^1(p) - x^1(p_0)] \sum_{k=2}^{\dim M} B_k [x^k(p) - x^k(p_0)] \\ &+ \sum_{k,l=2}^{\dim M} B_{kl}(x^2(p), \dots, x^{\dim M}(p)) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)] + \cdots \end{aligned}$$

Here  $B_0$  and  $B_k$ ,  $k \ge 2$  are arbitrary constant matrices, det  $B_0 \ne 0$ ,  $B_{kl}$  with  $k, l \ge 2$  are arbitrary matrix-valued functions (in both cases the arbitrariness comes from the arbitrariness of  $A_0$  in (9.31)), and  $B(x^1(p))$  is a fixed matrix-valued function whose form depends entirely on  $\Gamma_X$  (*B* comes from the expansion of *Y* in (9.31)). Comparing the last expression with (9.8), we see that the both expressions coincide up to notation if in (9.8) we put

$$A_{kl}^{(x)}(p)[x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})] = B(x^{1}(p))B_{0}[x^{1}(p) - x^{1}(p_{0})]^{2} - \Gamma_{X}(p_{0})[x^{1}(p) - x^{1}(p_{0})] \sum_{k=2}^{\dim M} B_{k}[x^{k}(p) - x^{k}(p_{0})] + \sum_{k,l=2}^{\dim M} B_{kl}(x^{2}(p), \dots, x^{\dim M}(p))[x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})] + \cdots$$
(9.36)

Hence, up to third order terms, (9.8) corresponds to (9.31) with

$$A_{11}^{(x)}(p) = B(x^1(p))B_0, \quad A_{1k}^{(x)}(p) = -\mathbf{\Gamma}_X(p_0)B_k \text{ for } k \ge 2,$$
  
$$A_{kl}^{(x)}(p) = B_{kl}(x^2(p), \dots, x^{\dim M}(p)) \text{ for } k, l \ge 2.$$

**Exercise 9.1.** Find the exact expressions for  $A_{kl}^{(x)}$  for which (9.8) reduces to (9.31). (Hint: in the both sides of (9.36), take into account all quantities up to fifth order with respect to  $[x^k(p) - x^k(p_0)]$ .)

The general idea and conclusion from the above discussion is: on every  $U \subseteq M$  frames normal on U for a given derivation along some fixed vector field always exist but the 'smaller' U is, the 'larger' the variety of these normal frames is; this variety is 'maximal' at a single point and it is 'minimal' for the whole manifold M.

The aforesaid, concerning normal frames, can, evidently, *mutatis mutandis* be transferred to the description of frames normal on subsets of a manifold (or along locally injective mappings with range in it) for derivations along fixed vector field in it.

## 10. Normal frames and coordinates for derivations along paths

As it is well known (see p. 11), to every injective  $C^1$  path  $\gamma: J \to M$  corresponds a vector field  $\dot{\gamma}$  along  $\gamma$ , i.e., on  $\gamma(J)$ , called tangent to  $\gamma$ , assigning to every point  $\gamma(s), s \in J$ , the vector  $\dot{\gamma}(s)$  tangent to  $\gamma$  at  $\gamma(s)$  and defined via (I.2.3).<sup>1</sup> The vector field  $\dot{\gamma}$  can be extended to a vector field X on a neighborhood of  $\gamma(J)$ ,  $X_{\gamma(J)} = \dot{\gamma}$ , in infinitely many ways. Consider some derivation  $D_X$  along X of  $T^1(M)$ . Since outside  $\gamma(J)$  the field X is completely arbitrary, such is  $D_X$  too. One can get real profit of  $D_X$  if it can be made to depend only on  $\gamma$ . Since it is clear that outside  $\gamma(J)$  this is impossible without some additional very restrictive assumptions on X, let us see what happens with  $D_X$  when its action is restricted to  $\gamma(J)$ . Suppose K is a  $C^1$  tensor field of type (r,q) defined on a neighborhood of  $\gamma(J)$  (or on  $\gamma(J)$  if dim<sub>R</sub> M = 1 and  $\gamma(J) = M$ ). If  $\{E_i\}$  is a frame on this neighborhood, the components of  $D_X$  in the tensor frame induced by it are (2.5). Hereof, applying  $X_{\gamma(J)} = \dot{\gamma}$  and (I.2.3), we obtain

$$\left( (D_X(K))_{j_1\dots j_q}^{i_1\dots i_r} \right) \Big|_{\gamma(s)} = \frac{\mathrm{d} \left( K_{j_1\dots j_q}^{i_1\dots i_r}(\gamma(s)) \right)}{\mathrm{d}s} + \sum_{a=1}^r \Gamma_X{}^{i_a}{}_k(\gamma(s)) K_{j_1\dots j_q}^{i_1\dots i_{a-1}ki_{a+1}\dots i_r}(\gamma(s)) - \sum_{b=1}^q \Gamma_X{}^k{}_{j_b}(\gamma(s)) K_{j_1\dots j_{b-1}kj_{b+1}\dots j_q}^{i_1\dots i_r}(\gamma(s)).$$

<sup>&</sup>lt;sup>1</sup>If  $\gamma$  is not injective, the correspondence  $\gamma(s) \to \dot{\gamma}(s)$  is generally multiple-valued, while the mapping  $s \to \dot{\gamma}(s)$  is always single-valued.

From here an important conclusion follows: if  $D_X$  is a derivation along X and the restriction of its matrix  $\Gamma_X$  to  $\gamma(J)$  depends only on the path  $\gamma$ ,

$$\Gamma_X(\gamma(s))|_{X_{\gamma(s)}=\dot{\gamma}(s)} = \Gamma(s;\gamma),$$
  

$$s \in J,$$
(10.1)

then the restriction to  $\gamma(J)$  of the action of  $D_X$  on any tensor field is independent of the particular definition of X and depends only on  $\gamma$ . Indeed, combining (10.1) with the equation preceding it, we find

$$\left( (D_X(K))_{j_1\dots j_q}^{i_1\dots i_r} \right) \Big|_{\gamma(s)} = \frac{ d \left( K_{j_1\dots j_q}^{i_1\dots i_r}(\gamma(s)) \right)}{ds} + \sum_{a=1}^r \Gamma^{i_a}{}_k(s;\gamma) K_{j_1\dots j_q}^{i_1\dots i_{a-1}ki_{a+1}\dots i_r}(\gamma(s)) - \sum_{b=1}^q \Gamma^k{}_{j_b}(s;\gamma) K_{j_1\dots j_{b-1}kj_{b+1}\dots j_q}^{i_1\dots i_r}(\gamma(s))$$
(10.2)

for  $X|_{\gamma(J)} = \dot{\gamma}$ . One should have in mind the invariant character of the condition (10.1): if it holds (for some or every path  $\gamma$ ) in some frame  $\{E_i\}$ , then it is valid in any other frame  $\{E'_i = A^j_i E_j\}$  as (2.11) implies

$$\Gamma(s;\gamma) \mapsto \Gamma'(s;\gamma) = A^{-1}(\gamma(s)) \Big[ \Gamma(s;\gamma) A(\gamma(s)) + \frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} \Big]$$
(10.3)

due to  $X|_{\gamma(J)} = \dot{\gamma}$ .

**Example 10.1.** An important example of a derivation along vector fields possessing the property (10.1) is provided by the linear connections which, as we know from Section 2, are derivations linear at every point of the manifold. In fact, if  $\nabla$  is a linear connection with coefficients' matrices  $\Gamma_k = [\Gamma^i_{\ jk}], \Gamma^i_{\ jk}$  being its coefficients, then, by Proposition 2.2 (see also Definition 2.4), we have

$$\mathbf{\Gamma}_X(\gamma(s))|_{X_{\gamma(s)}=\dot{\gamma}(s)} = \Gamma_k(\gamma(s))\dot{\gamma}^k(s) \tag{10.4}$$

as  $X|_{\gamma(J)} = \dot{\gamma}$ . The mapping assigning to a  $C^1$  tensor field K defined in a neighborhood of  $\gamma(J)$  the tensor field  $(\nabla_X K)|_{\gamma(J)}$ , where  $X \in \mathfrak{X}(M)$  and  $X|_{\gamma(J)} = \dot{\gamma}$ , is called *covariant derivative along*  $\gamma$  generated by (associated to)  $\nabla$ . It is denoted by  $\nabla_{\dot{\gamma}}$  (and sometimes by  $\frac{D}{d_s}$  or  $\frac{D}{d_s}|_{\gamma}$ ).<sup>2</sup> According to (10.2) and (10.4), the com-

<sup>&</sup>lt;sup>2</sup>A far more logical is the covariant derivative along  $\gamma$  to be denoted by  $\nabla^{\gamma}$  (see below Definition 10.1) instead by  $\nabla_{\dot{\gamma}}$ , but we shall follow the established tradition.

ponents of  $\nabla_{\dot{\gamma}} K$  at  $\gamma(s)$  are

$$\left( \left( \nabla_{\dot{\gamma}}(K) \right)_{j_{1}\dots j_{q}}^{i_{1}\dots i_{r}} \right) \Big|_{\gamma(s)} = \frac{\mathrm{d} \left( K_{j_{1}\dots j_{q}}^{i_{1}\dots i_{r}}(\gamma(s)) \right)}{\mathrm{d}s} + \dot{\gamma}^{l}(s) \sum_{a=1}^{r} \Gamma^{i_{a}}{}_{kl}(\gamma(s)) K_{j_{1}\dots j_{q}}^{i_{1}\dots i_{a-1}ki_{a+1}\dots i_{r}}(\gamma(s)) - \dot{\gamma}^{l}(s) \sum_{b=1}^{q} \Gamma^{k}{}_{j_{b}l}(\gamma(s)) K_{j_{1}\dots j_{b-1}kj_{b+1}\dots j_{q}}^{i_{1}\dots i_{r}}(\gamma(s)).$$
 (10.5)

where  $\Gamma^i_{jk}(\gamma(s))$  are the coefficients of  $\nabla$  at  $\gamma(s)$ . It should be mentioned the different meaning of the symbols  $\nabla_X$  and  $\nabla_{\dot{\gamma}}$ : in the former X is a vector field defined on the manifold M while in the latter  $\dot{\gamma}$  is a vector field defined only along a path  $\gamma: J \to M$  and not outside the set  $\gamma(J)$ . That is why an expansion like  $\nabla_{\dot{\gamma}} = \dot{\gamma}^k \nabla_{E_k}$  for  $\dot{\gamma}(s) = \dot{\gamma}^k(s) E_k|_{\gamma(s)}$  is not quite correct; one should write  $\nabla_{\dot{\gamma}}|_{\mathbf{T}^1(\gamma(J))} = \dot{\gamma}^k(\nabla_{E_k}|_{\mathbf{T}^1(\gamma(J))})$  which, in more free terms, can be reduced to  $\nabla_{\dot{\gamma}} = \dot{\gamma}^k(\nabla_{E_k}|_{\gamma(J)})$  or to  $\nabla_{\dot{\gamma}}|_{\gamma(s)} = \dot{\gamma}^k(s)(\nabla_{E_k}|_{\gamma(s)}), s \in J$ .

The above considerations lead to the concept of a derivation along paths (of the tensor algebra over a manifold) assigning to every  $C^1$  injective path a derivation along it with properties like (10.1) and (10.2).

**Definition 10.1.** A derivation D along paths of the tensor algebra over a  $C^1$  manifold M, is a mapping assigning to every  $C^1$  path  $\gamma: J \to M$  without selfintersections a mapping  $\mathsf{D}^{\gamma}$ , called *derivation along*  $\gamma$  (of the algebra of  $C^1$  tensor fields along  $\gamma$ ), from the algebra of  $C^1$  tensor fields along  $\gamma$  into the one of  $C^0$ tensor fields,

$$\mathsf{D}^{\gamma} \colon \boldsymbol{T}^{1}(\gamma(J)) \to \boldsymbol{T}^{0}(\gamma(J)),$$

such that:

 $\begin{array}{ll} (\mathbf{i}) & \mathsf{D}^{\gamma}(K+L) = \mathsf{D}^{\gamma}K + \mathsf{D}^{\gamma}L, & K, L \in \mathfrak{T}^{r;1}_{s}(\gamma(J)); \\ (\mathbf{ii}) & \mathsf{D}^{\gamma}(K \otimes L) = (\mathsf{D}^{\gamma}K) \otimes L + K \otimes (\mathsf{D}^{\gamma}L), & K, L \in \mathbf{T}^{1}(\gamma(J)); \\ (\mathbf{iii} \mathbf{a}) & \mathsf{D}^{\gamma} \colon \mathfrak{T}^{1}_{0}(\gamma(J)) \to \mathfrak{T}^{0}_{0}(\gamma(J)); \\ (\mathbf{iii} \mathbf{b}) & \mathsf{D}^{\gamma} \colon \mathfrak{T}^{1;1}_{0}(\gamma(J)) \to \mathfrak{T}^{1;0}_{0}(\gamma(J)); \\ (\mathbf{iii} \mathbf{c}) & \mathsf{D}^{\gamma} \colon \mathfrak{T}^{0;1}_{1}(\gamma(J)) \to \mathfrak{T}^{0;0}_{1}(\gamma(J)); \\ (\mathbf{iv}) & \mathsf{D}^{\gamma}(g) = \dot{\gamma}(g), & g \in \mathfrak{F}^{1}(\gamma(J)); \\ (\mathbf{v}) & \mathsf{D}^{\gamma}(\omega(Z)) = C^{1}_{1}(\mathsf{D}^{\gamma}(\omega \otimes Z)), & \omega \in \mathfrak{T}^{0;1}_{1}(\gamma(J)), & Z \in \mathfrak{T}^{1;1}_{0}(\gamma(J)). \end{array}$ 

Comments 10.1. This definition is similar to the Definition I.3.1 of a linear connection: the fist two properties of the latter one are removed and all covariant derivatives along a vector field X are replaced with derivatives along  $\gamma$ . Hence the points (2)–(6) of comments I.3.1 can be repeated *mutatis mutandis*. Hereof we can say that  $D^{\gamma}$ : is K-linear, satisfies the Leibnitz rule (with respect to the tensor multiplication), preserves the types of the tensor fields, commutes with all contraction operators, and on scalar functions reduces to the vector field  $\dot{\gamma}$  tangent to  $\gamma$ .

*Remark* 10.1. In connection with the theory of derivations (along paths) in vector bundles (see Subsection IV.2.3), the so-defined derivations along paths should be called section- or tensor-derivations along paths as the tensor fields are, in fact, sections of the corresponding tensor bundles over a manifold (see Subsection IV.2.4). At the moment, this more complicated name is not required and, respectively, it will not be employed in the present chapter. (For details, see the definition of a section-derivation along paths on page 221 and Subsection IV.13.1.)

Suppose D is a derivation along paths,  $\gamma: J \to M$  is injective and  $C^1$ , and  $\{E_i\}$  is a  $C^1$  frame on  $\gamma(J)$  which may be defined and outside  $\gamma(J)$ . By condition (iii b) of Definition 10.1, there exist unique functions  $\Gamma_{\gamma j}^{i}$  for which the expansion

$$\mathsf{D}^{\gamma}(E_i) =: \Gamma_{\gamma i}^{j} E_j \tag{10.6}$$

holds. They are called *(local) components of*  $\mathsf{D}^{\gamma}$  *or of*  $\mathsf{D}$  *along*  $\gamma$ . We shall write  $\Gamma^{i}_{j}(s;\gamma)$  for the value of  $\Gamma_{\gamma}{}^{i}_{j}$  at  $\gamma(s), \Gamma_{\gamma}{}^{i}_{j}:\gamma(s) \mapsto \Gamma^{i}_{j}(s;\gamma)$  which is correct as  $\gamma$  is supposed to be injective. The components of  $\mathsf{D}^{\gamma}$  uniquely describe it in  $\{E_i\}$  as from Definition 10.1 follows that the components of  $\mathsf{D}^{\gamma}(K)$  for  $K \in \mathfrak{T}_{q}^{r;1}(\gamma(J))$  in the tensor frame induced by  $\{E_i\}$  are

$$\left( (\mathsf{D}^{\gamma}(K))_{j_{1}\dots j_{q}}^{i_{1}\dots i_{r}} \right) \Big|_{\gamma(s)} = \frac{\mathrm{d} \left( K_{j_{1}\dots j_{q}}^{i_{1}\dots i_{q}}(\gamma(s)) \right)}{\mathrm{d}s} + \sum_{a=1}^{r} \Gamma^{i_{a}}{}_{k}(s;\gamma) K_{j_{1}\dots j_{q}}^{i_{1}\dots i_{a-1}ki_{a+1}\dots i_{r}}(\gamma(s)) - \sum_{b=1}^{q} \Gamma^{k}{}_{j_{b}}(s;\gamma) K_{j_{1}\dots j_{b-1}kj_{b+1}\dots j_{q}}^{i_{1}\dots i_{r}}(\gamma(s)).$$
 (10.7)

which can be verified by direct calculation.

Comparing (10.7) with (10.2), we conclude that to every derivation D along vector fields, whose components have the property (10.1), there corresponds a derivation along paths D such that  $D: \gamma \mapsto D_X|_{T^1(\gamma(J))}$  for  $X \in \mathfrak{X}(M)$  with  $X|_{\gamma(J)} = \dot{\gamma}$ . A particular example is provided by the covariant derivative along paths associating to  $\gamma$  the covariant derivative  $\nabla_{\dot{\gamma}}$  along  $\gamma$  with  $\nabla$  being a linear connection.

If the frame  $\{E_i\}$  is change to  $\{E'_i = A^j_i E_j\}$  (along  $\gamma$ ) by means of a nondegenerate matrix  $A = [A^j_i]$ , equation (10.6) implies the change

$$\Gamma_{\gamma \ j}^{\ i} \mapsto \Gamma_{\gamma \ j}^{\prime \ i} = \left(A^{-1}\right)_{l}^{i} \left[\Gamma_{\gamma \ k}^{\ l} A_{j}^{k} + \dot{\gamma}(A_{j}^{l})\right].$$
(10.8)

Introducing the matrix  $\mathbf{\Gamma}_{\gamma} := [\Gamma_{\gamma}{}^{i}{}_{j}]_{i,j=1}^{\dim M}$  of the components of  $\mathsf{D}^{\gamma}$ , we can rewrite this as (cf. (2.11))

$$\Gamma_{\gamma} \mapsto \Gamma_{\gamma}' = A^{-1} [\Gamma_{\gamma} A + \dot{\gamma}(A)].$$
(10.9)

Notice, the main difference between the pairs of transformations (2.10) and (2.11), on one hand, and (10.8) and (10.9), on another hand, is that the former ones are defined on an arbitrary neighborhood of M, while the latter ones have a sense only along  $\gamma$ , i.e., on  $\gamma(J)$ . If we put  $\Gamma(s; \gamma) = \Gamma_{\gamma}(\gamma(s))$ ,  $s \in J$ , which is the matrix of  $\mathsf{D}^{\gamma}$  (in  $\{E_i\}$ ) at the point  $\gamma(s)$ , the change (10.9) takes the form (10.3), as one can expect. If one is interested in holonomic frames, like  $\{E_i = \frac{\partial}{\partial x^i}\}$  with  $\{x^i\}$  being coordinates in  $U \subseteq M$  and  $U \cap \gamma(J) \neq \emptyset$ , the coordinates change  $\{x^i\} \to \{x'^i\}$ will imply, e.g., the change (10.8) with  $A_j^i = \frac{\partial x^i}{\partial x'^j}$  and all quantities should be restricted on  $U \cap \gamma(J)$ .

**Example 10.2.** It is easily seen that, by virtue of (10.4), the component's matrix of a covariant derivative  $\nabla_{\dot{\gamma}}$  along  $\gamma$  is

$$\Gamma_{\gamma} = \Gamma_k \dot{\gamma}^k \tag{10.10}$$

where  $\Gamma_k := \left[\Gamma_{jk}^i\right]$  are the coefficients' matrices of a linear connection  $\nabla$ .

A derivation along paths is called *differentiable of class*  $C^k$  or simply of class  $C^k$ , if its matrix is of class  $C^k$  with respect to all coordinate frames or the ones obtainable from them by means of  $C^{k+1}$  transformations (cf. similar convention concerning derivations along vector fields on page 146).

Now the problem for the existence and properties of a special kind of frames, called normal, in which the components of a derivation along paths vanish can be posed. It has two sides: local, along particular path, and global, along every path. We shall start with the former case.

**Definition 10.2.** Let D be a derivation along paths and  $\gamma: J \to M$  be injective  $C^1$  path. A frame  $\{E'_i\}$  defined on  $\gamma(J)$  or on its subset is called normal for  $\mathsf{D}^{\gamma}$  (or for D along  $\gamma$ ) if in it the components of  $\mathsf{D}^{\gamma}$  (of D along  $\gamma$ ) vanish. A frame defined on a neighborhood (of part) of  $\gamma(J)$  is called normal for  $\mathsf{D}^{\gamma}$  if its restriction to  $\gamma(J)$  is normal for  $\mathsf{D}^{\gamma}$ .

Frames  $\{E'_i\}$  normal along particular path  $\gamma$  for a  $C^1$  derivation D along paths always exist and can easily be found on the base of the transformation law (10.9) of the components of D along  $\gamma$ .

**Theorem 10.1.** If D is a  $C^0$  derivation along paths in a  $C^2$  manifold M and  $\gamma: J \to M$  is injective and  $C^1$ , then all frames (globally) normal for D along  $\gamma$ , *i.e.*, for  $D^{\gamma}$ , are  $\{E'_i = A^j_i E_j\}$ , where  $\{E_i\}$  is arbitrary frame along  $\gamma$  and  $A = [A^j_j]$  is given by

$$A(\gamma(s)) = Y(s, s_0; -\Gamma_{\gamma} \circ \gamma) A_0, \qquad (10.11)$$

in which Y is the unique solution of (II.3.17),  $s \in J$ ,  $s_0 \in J$  is fixed,  $\Gamma_{\gamma}$  is the matrix of D in  $\{E_i\}$ , and  $A_0$  is constant non-degenerate matrix.

*Proof.* If  $\{E_i\}$  is an arbitrary frame on  $\gamma(J)$ , a frame  $\{E'_i = A^j_i E_j\}$  is normal along  $\gamma$ ,  $\Gamma'_{\gamma} = 0$ , iff the normal frame equation

$$\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} + \mathbf{\Gamma}(s;\gamma)A(\gamma(s)) = 0 \tag{10.12}$$

holds along  $\gamma$ , where  $A := [A_j^i]: \gamma(J) \to \operatorname{GL}(\dim M, \mathbb{K})$  is of class  $C^1$  and (10.3) was used. By Remark II.3.9 on page 96, the general solution of the last equation with respect to A is given by (10.11).

Since the provided by Theorem 10.1 frames normal for a derivation  $D^{\gamma}$  along a concrete path  $\gamma: J \to M$  are defined only on  $\gamma(J)$ , the problem for their holonomicity is ill-posed. The correct question is: can these frames be (locally) extended in a holonomic or anholonomic way on a neighborhood with non-empty intersection with  $\gamma(J)$ ? As the existence of anholonomic extensions is clear and beyond doubt, we shall show the existence of holonomic extensions on which base, if required, anholonomic ones can be constructed. Looking on M as on a real manifold (of dimension dim<sub> $\mathbb{R}$ </sub>  $M = 2 \dim M$  if M is complex) and admitting  $\gamma$  to be regular, the needed result is an evident corollary of the following lemma.

**Lemma 10.1.** Let  $\gamma: J \to M$  be injective  $C^1$  regular path in a  $C^1$  real manifold Mand  $\{E_i\}$  be a frame defined on  $\gamma(J)$ . For every  $s_0 \in J$ , there exists a chart (U, y)with  $U \ni \gamma(s_0)$  such that the coordinate frame  $\{\frac{\partial}{\partial y^i}\}$  reduces along  $\gamma$  to  $\{E_i\}$ ,  $\frac{\partial}{\partial y^i}|_{U\cap\gamma(J)} = E_i|_{U\cap\gamma(J)}$ , i.e., in a neighborhood of every point in  $\gamma(J)$  the initial frame  $\{E_i\}$  can be extended in a holonomic way.

Remark 10.2. Since now  $\gamma(J)$  is one-dimensional (real) submanifold of M, this lemma is a corollary of Lemma II.5.2 on page 116. Because of the importance of this result, we present here its independent proof following the one of [76, Lemma 4.1].

*Proof.* Since  $\gamma$  is  $C^1$ , regular, and without self-intersections, for every  $s_0 \in J$  the special chart  $(U_1, x)$ , with  $U_1 \ni \gamma(s_0)$  and described in Subsection II.3.2 on page 94, can be constructed. Recall, (see (II.3.12) and (II.3.13)), for  $p \in U_1$ , we have x(p) = (s, t) for some  $s \in J_1 := \{\sigma | \sigma \in J, \gamma(\sigma) \in U_1\} \subseteq J$  and  $t \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$  and  $x(\gamma(s)) = (s, t_0), s \in J_1$ , for fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} M - 1}$ . Let the non-degenerate matrix  $B = [B_j^i(s)], s \in J_1$ , be defined via the expansion

$$E_i|_{\gamma(s)} = B_i^j(s) \frac{\partial}{\partial x^j}\Big|_{\gamma(s)}.$$
(10.13)

Define the functions  $y^i \colon U_1 \to \mathbb{R}$  by

$$y^{i}(x^{-1}(s, \boldsymbol{t})) := b^{i} + \int_{s_{0}}^{s} \left(B^{-1}(\sigma)\right)_{1}^{i} d\sigma + \sum_{k=2}^{\dim_{\mathbb{R}}M} \left(B^{-1}(\sigma)\right)_{k}^{i}(t^{k} - t_{0}^{k}) + \sum_{k,l=2}^{\dim_{\mathbb{R}}M} b^{i}_{kl}(x^{-1}(s, \boldsymbol{t}))(t^{k} - t_{0}^{k})(t^{l} - t_{0}^{l})$$
(10.14)

where  $b^i \in \mathbb{R}$  and  $b^i_{kl} \colon U_1 \to \mathbb{R}$  are  $C^1$  and together with their derivatives are bounded on  $\gamma(J_1)$ . Since

$$\frac{\partial y^{i}}{\partial x^{j}}\Big|_{\gamma(s)} = \frac{\partial y^{i}}{\partial x^{j}}\Big|_{x^{-1}(s,t_{0})} = \left(B^{-1}(s)\right)_{j}^{i}$$

and B(s) is non-degenerate, the transformation  $\{x^i\} \to \{y^i\}$  is well-defined  $C^1$ diffeomorphism on  $U := x^{-1}(J_1, V) \subseteq U_1$  for some neighborhood V of  $\mathbf{t_0}$  in  $\mathbb{R}^{\dim_{\mathbb{R}} M-1}$ . Hence (U, y) is a chart of M with  $U \ni \gamma(s_0)$  and the coordinate frame generated by the associated with it coordinates  $\{y^i\}$  has the following basic vectors along  $\gamma$ , i.e., on  $\gamma(J_1) \subseteq \gamma(J)$ :

$$\frac{\partial}{\partial y^i}\Big|_{\gamma(s)} = \frac{\partial x^j}{\partial y^i}\Big|_{\gamma(s)} \frac{\partial}{\partial x^j}\Big|_{\gamma(s)} = \left(\left[\frac{\partial y^k}{\partial x^l}\Big|_{\gamma(s)}\right]^{-1}\right)_i^j \frac{\partial}{\partial x^j}\Big|_{\gamma(s)} = B_i^j(s)\frac{\partial}{\partial x^j}\Big|_{\gamma(s)}$$

Therefore  $\frac{\partial}{\partial y^i}|_{\gamma(s)} = E_i|_{\gamma(s)}$  and, consequently,  $\left\{\frac{\partial}{\partial y^i}\right\}$  is a holonomic extension of  $\{E_i\}$  in the neighborhood U of  $\gamma(s_0)$ .

The next proposition reveals the complete (local) consistency between the normal frames provided by Theorem 10.1 for the covariant derivatives along paths (generated by a linear connection  $\nabla$ ) and the ones given via Theorem II.3.1 on page 97.

**Proposition 10.1.** Let  $\gamma: J \to M$  be a  $C^1$  regular injective path in a  $C^1$  manifold M endowed with a  $C^0$  linear connection  $\nabla$ . The frames on  $\gamma(J)$  normal along  $\gamma$  for  $\nabla$  and the ones normal for the covariant derivative  $\nabla_{\dot{\gamma}}$  along  $\gamma$  are identical.

*Proof.* If a frame  $\{E'_i\}$  is normal along  $\gamma$  for  $\nabla$ , then, due to (10.10), the components' matrix of  $\nabla_{\dot{\gamma}}$  in  $\{E'_i|_{\gamma(J)}\}$  is  $\Gamma'_{\gamma} = \Gamma'_k \dot{\gamma}^k = \sum_k 0 \times \dot{\gamma}^k = 0$  as, by definition, in a frame normal for  $\nabla$  the coefficients' matrices  $\Gamma'_k$  of  $\nabla$  vanish.

To prove the converse assertion, we shall use the special chart  $(U_1, x)$ , in a neighborhood of some point  $\gamma(s_0) \in \gamma(J)$ , constructed at the beginning of Subsection II.3.2. Since in the frame  $\{E_i = \frac{\partial}{\partial x^i}|_{U_1}\}$  we have  $\dot{\gamma}^k = \delta_1^k$ , in it (10.10) gives the matrix  $\Gamma_{\gamma}$  of  $\nabla_{\dot{\gamma}}$  as  $\Gamma_{\gamma} = -\Gamma_1$ . Hereof, applying Theorem 10.1, we see that a frame  $\{E_i'\}$  on  $\gamma(J)$  normal for  $\nabla_{\dot{\gamma}}$  has on  $\gamma(J) \cap U_1$  basic vectors  $E_i'|_{\gamma(s)} = A_i^j(\gamma(s))\frac{\partial}{\partial x^j}|_{\gamma(s)}$  for  $s \in J_1 := \{\sigma | \sigma \in J, \ \gamma(\sigma) \in U_1\} \ni s_0$  with

$$A(\gamma(s)) = Y(s, s_0; -\Gamma_1 \circ \gamma) A_0 \tag{10.15}$$

which, up to notation, coincides with the matrix (II.3.22) on page 97 transforming  $\left\{\frac{\partial}{\partial x^i}\Big|_{\gamma(J)}\right\}$  into a frame on  $\gamma(J)$  normal for  $\nabla$ . So, by Theorem II.3.1, locally, i.e., on  $\gamma(J_1)$ , every frame  $\{E'_i\}$  normal for  $\nabla_{\dot{\gamma}}$  is a restriction along  $\gamma$  (more precisely, to  $\gamma(J_1)$ ) of a frame normal for  $\nabla$  along  $\gamma$ .

In short, admitting some abuse of the language, we can say that the frames normal along a path for a linear connection are normal for the generated by it covariant derivative along paths and *vice versa*.

One should have in mind a significant difference between the normal frames provided by Theorem 10.1 for  $\nabla_{\dot{\gamma}}$  and the ones provided by Theorem II.3.1 along  $\gamma$  for  $\nabla$ : the former normal frames are *global*, i.e., defined along the whole path  $\gamma$ , on the set  $\gamma(J)$ , while the latter ones are *local*, i.e., defined only on a neighborhood in  $\gamma(J)$  of each point in  $\gamma(J)$ . Now we turn our attention to the 'global' case.

**Definition 10.3.** A frame  $\{E_i\}$  defined on a subset  $U \subseteq M$  or on a larger set (if  $U \neq M$ ) is called normal on U for a derivation  $\mathsf{D}$  along paths if for every  $C^1$  injective path  $\gamma: J \to U$  lying entirely in U the restricted frame  $\{E_i|_{\gamma(J)}\}$  is normal along  $\gamma$  for the derivation  $\mathsf{D}^{\gamma}, \mathsf{D}: \gamma \to \mathsf{D}^{\gamma}$ . In particular, a frame normal on M for  $\mathsf{D}$  is called simply normal for  $\mathsf{D}$ .

**Proposition 10.2.** A derivation along paths admits frames normal on  $U \subseteq M$  if and only if on U it is a covariant derivative along paths induced by some linear connection for which frames normal on U exist.

*Proof.* Suppose, a derivation D along paths admits a frame  $\{E'_i\}$  normal on U. Let  $\{E_i\}$  be a frame on U and  $A = [A^j_i]: U \to \operatorname{GL}(\dim M, \mathbb{K})$  be defined via the expansion  $E'_i|_{\gamma(s)} = A^j_i(\gamma(s))E_j|_{\gamma(s)}$ . By Theorem 10.1, the matrix  $\Gamma(s; \gamma)$  of D along *arbitrary* path  $\gamma: J \to U$  in U and the matrix-valued function A are connected by (10.11) or, equivalently, satisfy (10.12). Hence, from (10.12), we get

$$\Gamma(s;\gamma) = -\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s}A^{-1}(\gamma(s)) = -E_k(A)|_{\gamma(s)}\dot{\gamma}^k(s)A^{-1}(\gamma(s))$$

where  $\dot{\gamma}^k$  are the components of  $\dot{\gamma}$  in  $\{E_k\}$ . Therefore the matrix  $\Gamma_{\gamma}$  of  $\mathsf{D}^{\gamma}$  in  $\{E_k|_{\gamma(J)}\}$  is (10.10) with

$$\Gamma_k : p \mapsto \Gamma_k(p) = -(E_k(A))|_p A^{-1}(p).$$
 (10.16)

Since a simple verification shows that these matrices  $\Gamma_k$  transform according to (I.5.3), they are matrices of the coefficients of a linear connection  $\nabla$  which, due to  $\Gamma(s;\gamma) = \Gamma_k(\gamma(s))\dot{\gamma}^k(s)$ , is such that  $D^{\gamma} = \nabla_{\dot{\gamma}}$ , i.e., the covariant derivative along paths generated by  $\nabla$  coincides with the initial derivation D along paths. As  $\{E'_i\}$  is normal for D, in it is fulfilled  $0 = \Gamma'(s;\gamma) = \Gamma'_k(\gamma(s))\dot{\gamma}'^k(s)$  for arbitrary  $\gamma: J \to U$ . Hereof  $\Gamma'_k(p)\dot{\gamma}'^k(s_0) = 0$  for every path  $\gamma: J \to U$  with  $\gamma(s_0) = p \in U$ for some  $s_0 \in J$  which implies  $\Gamma'_k|_U = 0$  and, consequently,  $\{E'_i\}$  is normal on Ufor  $\nabla$ .

Conversely, suppose  $\mathsf{D}^{\gamma} = \nabla_{\dot{\gamma}}$  for a linear connection  $\nabla$  for which a frame  $\{E'_i\}$  normal on U exist. Then the matrix  $\Gamma_{\gamma}$  of  $\mathsf{D}^{\gamma}$  is given by (10.10) for all frames  $\{E_i\}$  on U and all paths  $\gamma: J \to U$  in U. In particular, for  $E_i = E'_i|_{\gamma(J)}$ , we have  $\Gamma'_{\gamma}: \gamma(s) \mapsto \Gamma'(s; \gamma) = \Gamma'_k(\gamma(s))\dot{\gamma}^k(s) = \sum_k 0 \times \dot{\gamma}^k(s) = 0$  as  $\Gamma'_k = 0$  in a frame  $\{E'_i\}$  normal for  $\nabla$ . Therefore  $\Gamma_{\gamma} = 0$  in  $\{E'_i\}$  for every path  $\gamma$  lying entirely in U and, by Definition 10.3, the frame  $\{E'_i\}$  is normal for  $\mathsf{D}^{\gamma} = \nabla_{\dot{\gamma}}$ .

The meaning of Proposition 10.2 is that the condition on a derivation D along paths to admit frames normal on  $U \subseteq M$  is so strong that it reduces such a derivation to a covariant derivative along paths admitting frames normal on U. Since, by Proposition 10.1, such a covariant derivative along paths is always generated by a linear connection  $\nabla$  admitting frames normal on U, all problems concerning frames normal for D are reduced to similar problems for  $\nabla$ , which we have investigated earlier at length.

# 11. On frames simultaneously normal for two derivations

The problems concerning frames simultaneously normal for two derivations (along vector fields, fixed vector field, paths, fixed path) are new and not investigated. They arise naturally in some physical theories of gravity in which as a space-time model is taken a differentiable manifold endowed with two, generally or initially independent, linear connections. Such are the metric-affine theories [36, 92], the ones with metric and background affine connection [93,94] in which there are two connections, one of them being a Riemannian connection induced by some Riemannian metric, as well as the theories with covariant and contravariant affine connection [35, 95, 96] in which two different linear connections in the tangent and cotangent bundles are used.<sup>1</sup> The importance of the normal frames in these theories comes from the fact that the normal frames are the mathematical tool for description of the physical concept 'inertial frame' [97]. So, the equivalence principle selects only linear connections possessing normal frames for the description of the pure gravity. Hereof, if two linear connections admit common normal frame(s), they can serve together as a mathematical base for a gravitational theory, otherwise only one of them should be selected for the same purpose.

Below in this section, we shall present some quite simple results whose main moral is: frames simultaneously normal for two arbitrary and different derivations (along vector fields, fixed vector field, paths, fixed path), generally, do not exist; however, there are some exceptions. Physically, this can be interpreted as: in a gravity theory with two linear connections only one of them can describe the gravity directly, the other one may be connected with it but, generally, it should be primary related to an object different from the gravity. However, there exist exceptions from this conclusion which will be pointed.

The order of the rigorous results comes now.

**Proposition 11.1.** Let  $D^{(1)}$  and  $D^{(2)}$  be derivations along vector fields admitting frames normal on  $U \subseteq M$ . There exist frames normal on U for  $D^{(1)}$  and  $D^{(2)}$  simultaneously iff  $D^{(1)}|_U = D^{(2)}|_U$ .

Proof. Let  $\{E_i\}$  be a frame on (a neighborhood of) U and  $\Gamma_k^{(a)}$  be the coefficients' matrices of  $D^{(a)}$ , a = 1, 2. (Recall, by Proposition 3.1, the derivations are linear on U.) A frame  $\{E'_i = A^j_i E_j\}$  is normal for  $D^{(1)}$  and  $D^{(2)}$  iff (3.5) holds for  $\Gamma_k^{(a)}$  instead of  $\Gamma_k$ :  $(\Gamma_k^{(a)}A + E_k(A))|_U = 0$ ,  $a = 1, 2, A = [A^j_i]$  which is possible iff  $\Gamma_k^{(1)}|_U = \Gamma_k^{(2)}|_U$  as A is non-degenerate which, in its turn, means  $D^{(1)}|_U = D^{(2)}|_U$ .

<sup>&</sup>lt;sup>1</sup>In the last case, the two connections are equivalent to a single operator  $\nabla$  satisfying all conditions in Definition I.3.1 but the condition (vii); see the definition of a mixed linear connection on page 199.

**Proposition 11.2.** Let  $D_{X_{(1)}}^{(1)}$  and  $D_{X_{(2)}}^{(2)}$  be derivations along fixed vector fields  $X_{(1)}$ and  $X_{(2)}$ , respectively,  $U \subseteq M$ ,  $\{E_i\}$  be a frame on (a neighborhood of) U, and  $D_{X_{(1)}}^{(1)}$  and  $D_{X_{(2)}}^{(2)}$  admit frames normal on U. Then a frame  $\{E'_i = A^j_i E_j\}$  normal on U for  $D_{X_{(1)}}^{(1)}$  is also normal for  $D_{X_{(2)}}^{(2)}$  if and only if in  $\{E_i\}$  the components' matrix of  $D_{X_{(2)}}^{(2)}$  on U is

$$\Gamma_{X_{(2)}}^{(2)}|_U = -\left[ (X_{(2)}(A))A^{-1} \right] \Big|_U, \qquad A = [A_i^j].$$

*Proof.* The frame  $\{E'_i = A^j_i E_j\}$  is normal for  $D^{(2)}_{X_{(2)}}$  if its matrix  $\Gamma^{(2)}_{X_{(2)}}$  and A are connected via (9.1) with  $X_{(2)}$  and  $\Gamma^{(2)}_{X_{(2)}}$  for X and  $\Gamma_X$ , respectively, from where the assertion follows.

**Proposition 11.3.** Let  $U \subseteq M$ ,  $X \in \mathfrak{X}(M)$  be fixed,  $D^{(1)}$  and  $D_X^{(2)}$  be derivations along vector fields and X, respectively, for which frames normal on U exist. Frames normal on U for  $D^{(1)}$  and  $D_X^{(2)}$  simultaneously exist if and only if  $D_X^{(2)}|_U = D_X^{(1)}|_U$ , where  $D^{(1)}: X \mapsto D_X^{(1)}$ .

Proof. Let  $\{E_i\}$  be a frame on (a neighborhood of) U and  $\{E'_i = A^j_i E_j\}$  be a frame normal on U for  $D^{(1)}$ . Hereof  $A = [A^j_i]$  satisfies (3.5), with the coefficients' matrices  $\Gamma_l^{(1)}$  of  $D^{(1)}$  for  $\Gamma_l$ , which implies  $(\Gamma_l^{(1)}X^lA + X(A))|_U = 0$ ,  $X = X^k E_k$ . The frame  $\{E'_i\}$  is also normal for  $D^{(2)}_X$  iff (9.1) holds, with the matrix  $\Gamma^{(2)}_X$  of  $D^{(2)}_X$  for  $\Gamma_X$ , i.e.,  $(\Gamma^{(2)}_XA + X(A))|_U = 0$ , which is equivalent to  $\Gamma^{(2)}_X|_U = (\Gamma_l^{(1)}X^l)|_U$ , due to the previous equality, which, in its turn, expresses  $D^{(2)}_X|_U = D^{(1)}_X|_U$  in  $\{E_i\}$ .

**Proposition 11.4.** Let the derivations  ${}^{(1)}\mathsf{D}^{\gamma}$  and  ${}^{(2)}\mathsf{D}^{\gamma}$  along one and the same  $C^1$  injective path  $\gamma: J \to M$  admit normal frames (along  $\gamma$ ). There is a frame simultaneously normal for  ${}^{(1)}\mathsf{D}^{\gamma}$  and  ${}^{(2)}\mathsf{D}^{\gamma}$  if and only if  ${}^{(1)}\mathsf{D}^{\gamma} = {}^{(2)}\mathsf{D}^{\gamma}$ .

Proof. If  $\{E_i\}$  is a frame on  $\gamma(J)$  and the frame  $\{E'_i = A^j_i E_j\}$  is normal for  ${}^{(1)}\mathsf{D}^{\gamma}$ , i.e., (10.12) holds for the matrices  $\Gamma_{\gamma}^{(1)}$  of  ${}^{(1)}\mathsf{D}^{\gamma}$  in  $\{E_i\}$  for  $\Gamma_{\gamma}$ ,  $\Gamma_{\gamma} = \Gamma_{\gamma}^{(1)}$ , then  $\{E'_i\}$  is also normal for  ${}^{(2)}\mathsf{D}^{\gamma}$  iff  $\Gamma^{(2)}(s;\gamma)$  satisfies (10.12) with  $\Gamma_{\gamma} = \Gamma_{\gamma}^{(2)}$  which, due to the previous equality, is equivalent to  $\Gamma_{\gamma}^{(1)} = \Gamma_{\gamma}^{(2)}$ , i.e., to  ${}^{(1)}\mathsf{D}^{\gamma} = {}^{(2)}\mathsf{D}^{\gamma}$ .  $\Box$ 

**Proposition 11.5.** Let <sup>(1)</sup>D and <sup>(2)</sup>D be derivations along paths which admit frames normal on  $U \subseteq M$ . Then frames simultaneously normal for <sup>(1)</sup>D and <sup>(2)</sup>D exist iff <sup>(1)</sup>D|<sub>U</sub> = <sup>(2)</sup>D|<sub>U</sub>.

*Proof.* The result follows from Proposition 11.4 for arbitrary path  $\gamma: J \to M$ . The assertion is also a consequence of Propositions 10.2 and 11.1 as any linear connection is a (linear) derivation along vector fields.

**Proposition 11.6.** Frames simultaneously normal along  $\gamma: J \to M$  for derivations <sup>(1)</sup>D and <sup>(2)</sup>D<sup> $\gamma$ </sup> along paths and along  $\gamma$ , respectively, exist iff <sup>(2)</sup>D<sup> $\gamma$ </sup> = <sup>(1)</sup>D<sup> $\gamma$ </sup>, with <sup>(1)</sup>D:  $\gamma \mapsto {}^{(1)}D^{\gamma}$ .

Proof. See Proposition 11.4.

Now we want to say a few words on the physical theories with covariant and contravariant affine (linear) connections. As it was mentioned in Footnote 1 on page 197, their mathematical base are the so-called *linear connections of mixed type* [41, § 37], or, simply, *mixed linear connections* which are a generalization of the linear connections. A mapping  $\nabla$  is called *mixed linear connection (of type* (1, 1)) if it satisfies all conditions of Definition I.3.1 on page 21 except conditions (vii) in it. If  $\{E_i\}$  and  $\{E^i\}$  are frames in  $\mathfrak{T}_0^1(M)$  and  $\mathfrak{T}_1^0(M)$  respectively,<sup>2</sup> the *contravariant*  $\Gamma_{jk}^i$  and the *covariant*  $\Gamma_{jk}^i$  coefficients of  $\nabla$  in these frames are defined via the expansions

$$\nabla_{E_k} E_j =: \Gamma^i_{\ jk} E_i, \quad \nabla_{E_k} E^j =: \Gamma^j_{\ k} E^i.$$
(11.1)

If  $K = K_{j_1...j_s}^{i_1...i_r} E_{i_1} \otimes \cdots \otimes E_{i_r} \otimes E^{j_1} \otimes \cdots \otimes E^{j_s}$  and  $X = X^k E_k \in \mathfrak{X}(M)$ , then

$$\nabla_X K = X^k \nabla_{E_k} K \tag{11.2}$$

and the components of  $\nabla_{E_k} K$  are (cf. (I.3.3))

$$(\nabla_{E_k} K)_{j_1 \dots j_s}^{i_1 \dots i_r} := E_k \left( K_{j_1 \dots j_s}^{i_1 \dots i_r} \right) + \sum_{a=1}^{r} \Gamma_{k}^{i_a} K_{j_1 \dots j_s}^{i_1 \dots i_{a-1} l_{i_{a+1} \dots i_r}} + \sum_{b=1}^{s} \Gamma_{j_b}{}^l_k K_{j_1 \dots j_{b-1} l_{j_{b+1} \dots j_s}}.$$

$$(11.3)$$

If we change the frame and coframe by means of non-degenerate matrix-valued functions  $A = [A_i^j]$  and  $B = [B_i^j]$ ,

$$E_i \mapsto E'_i = A^j_i E_j, \quad E^i \mapsto E'^i = B^i_j E^j, \tag{11.4}$$

the definitions (11.1) imply the following transformations of the coefficients of  $\nabla$  (cf. (I.3.5)):

$$\Gamma^{i}_{jk} \mapsto \Gamma^{\prime i}_{\ jk} = \left(A^{-1}\right)^{i}_{l} A^{m}_{j} A^{n}_{k} \Gamma^{l}_{mn} + \left(A^{-1}\right)^{i}_{l} E^{\prime}_{k} \left(A^{l}_{j}\right)$$
(11.5)

$$\Gamma_{i\ k}^{\ j} \mapsto \Gamma_{i\ k}^{\prime j} = \left(B^{-1}\right)_{i}^{l} B_{m}^{j} A_{k}^{n} \Gamma_{l}^{\ m}{}_{n} + \left(B^{-1}\right)_{i}^{l} E_{k}^{\prime} \left(B_{l}^{j}\right)$$
(11.6)

which, if we introduce the *coefficients' matrices*  $\Gamma_k := [\Gamma_{jk}^i]_{i,j=1}^{\dim M}$  and  $\overline{\Gamma}_k := [\Gamma_{jk}^i]_{i,j=1}^{\dim M}$  of  $\nabla$ , in a matrix form read (see the convention on the matrix indices

<sup>&</sup>lt;sup>2</sup>The coframe  $\{E^i\}$  is not necessary to be dual to  $\{E_i\}$ .

on page xii)

$$\Gamma_k \mapsto \Gamma'_k = A^l_k A^{-1} \big[ \Gamma_l A + E_l(A) \big] \tag{11.5'}$$

$$\overline{\Gamma}_k \mapsto \overline{\Gamma}'_k = A_k^l \big[ B\overline{\Gamma}_l + E_l(B) \big] B^{-1} = A_k^l B \big[ \overline{\Gamma}_l B^{-1} - E_l(B^{-1}) \big].$$
(11.6')

Hence  $\Gamma^i_{jk}$  are coefficients of (ordinary) linear connection which we shall denote by  ${}^{(1)}\nabla$ .

If  $B = A^{-1}$ , which is equivalent to the requirement the action of the basic vectors of the coframe  $\{E^i\}$  on the ones of the frame  $\{E_i\}$  to be invariant under the change (11.4), i.e.,  $E^i(E_j) = E'^i(E'_j)$ , then (11.6') reduces to

$$\overline{\Gamma}_k \mapsto \overline{\Gamma}'_k = A_k^l A^{-1} \left[ \overline{\Gamma}_l A - E_l(A) \right]$$
(11.7)

which implies that

$${}^{(2)}\Gamma^{i}_{jk} := -\Gamma_{jk}^{i}, \quad {}^{(2)}\Gamma_{k} := -\overline{\Gamma}_{k}$$
(11.8)

are, respectively, coefficients and coefficients' matrices of a (usual) linear connection which we denote by  ${}^{(2)}\nabla^{.4}$ 

It is almost evident,  $\nabla$  is ordinary linear connection if and only if one, and hence any, of the following equivalent equations holds:

$${}^{(1)}\nabla = \nabla, \quad {}^{(1)}\nabla = {}^{(2)}\nabla, \quad {}^{(2)}\nabla = \nabla, \quad \Gamma_j{}^i{}^k{}_k + \Gamma^i{}_{jk} = 0, \quad (11.9)$$

the last of which is valid in, e.g., a pair of dual frame and coframe. Consequently, any one of these equalities is equivalent to the conditions (vii) in Definition I.3.1.

**Definition 11.1.** A pair  $({E_i}, {E^i})$  of a frame and coframe is called normal on  $U \subseteq M$  for a mixed linear connection  $\nabla$  if in it all of the coefficients, contravariant and covariant ones, of  $\nabla$  vanish on U.

**Proposition 11.7.** If only changes preserving the action of the coframe's basic vectors on the frame's ones are considered,<sup>5</sup> a mixed linear connection  $\nabla$  admits a pair of such frame and coframe normal on  $U \subseteq M$ , if and only if on  $U \nabla$  reduces to ordinary linear connection for which frames normal on U exist.

Proof. If  $(\{E_i\}, \{E^i\})$ , with  $E^i(E_j)$  constant under transformations of the frame and coframe, is normal on U for  $\nabla$ , then  $\{E_i\}$  is normal on U for  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$  simultaneously and, by Proposition 11.1,  ${}^{(1)}\nabla|_U = {}^{(1)}\nabla|_U$  which, as we said above, implies  $\nabla|_U = {}^{(1)}\nabla|_U$  and  $\{E_i\}$  is normal on U for  $\nabla|_U$ . Conversely, if  $\nabla$  is ordinary linear connection and  $\{E_i\}$  is a frame normal on U for  $\nabla$ , then  $\nabla = {}^{(1)}\nabla = {}^{(2)}\nabla$  and  $(\{E_i\}, \{E^i\})$ , with invariant  $E^i(E_j)$ , is normal on U for  $\nabla$ considered as a mixed linear connection.  $\Box$ 

 $<sup>^{3}\</sup>mathrm{In}$  particular such are all pairs of a frame and the coframe dual to it.

<sup>&</sup>lt;sup>4</sup>In some sense [95, 96] <sup>(1)</sup> $\nabla$  (resp. <sup>(2)</sup> $\nabla$ ) is the 'contravariant' (resp. covariant) component of  $\nabla$ .

<sup>&</sup>lt;sup>5</sup>In particular, such is the set of all dual frames and coframes.

#### 11. Frames normal for two derivations

Consider now the general case, when the frame  $\{E_i\}$  and coframe  $\{E^i\}$  are completely arbitrary,  $B \neq A^{-1}$ . By virtue of (11.5') and (11.6'), the pair ( $\{E'_i = A^j_i E_j\}, \{E'^i = B^i_j E^j\}$ ) is normal on  $U \subseteq M$  for a mixed linear connection  $\nabla$  iff the system of normal frame equations

$$(\Gamma_k A + E_k(A))|_U = 0 (11.10a)$$

$$(-\overline{\Gamma}_k B^{-1} + E_k (B^{-1}))|_U = 0$$
(11.10b)

holds. These equations are equivalent to the assertions that, respectively, the frame  $\{E'_i = A^j_i E_j\}$  is normal on U for the linear connection  ${}^{(1)}\nabla$  and the frame  $\{E''_i = (B^{-1})^j_i E_j\}$  is normal on U for the linear connection  ${}^{(2)}\nabla$ .<sup>6</sup> These very simple results imply the following theorem.

**Theorem 11.1.** A mixed linear connection  $\nabla$  admits frames normal on  $U \subseteq M$ if and only if the associated to it linear connections  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$  admit frames normal on U. Moreover, if frames normal on U for  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$  exist and  $\{E_i\}$ and  $\{E^i\}$  are respectively frame and coframe on (a neighborhood of) U, then all pairs of frames and coframes normal on U for  $\nabla$  are  $(\{E'_i = A^j_i E_j\}, \{E'^i = B^i_j E^j\})$  where  $\{E'_i\}$  is a frame normal on U for  ${}^{(1)}\nabla$  and  $\{E''_i = (B^{-1})^j_i E_j\}$  is a frame normal on U for  ${}^{(2)}\nabla$ .

Remark 11.1. For  ${}^{(1)}\nabla \neq {}^{(2)}\nabla$ , the second part of the theorem does not contradict to Propositions 11.7 and 11.1 because nowhere is imposed the additional condition  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$  to have a common frame normal on U, which could happen iff (11.10a) and (11.10b) have a common solution with respect to A and  $B^{-1}$ respectively.

From the above discussion the following conclusions can be made.

If pairs of frames  $\{E_i\}$  and coframes  $\{E^i\}$  for which the functions  $f_i^j := E^j(E_i)$  are invariant under the change (11.4) are considered, a mixed linear connection does not admit normal frames unless it is an ordinary linear connection for which normal frames exist; in particular, this is true if only pairs of dual frames and coframes, for which  $f_i^j = \delta_i^j$ , are taken into account.

If the considered frames and coframes are completely arbitrary, with no additional conditions on them imposed, then all problems concerning normal frames for a mixed linear connection  $\nabla$  are reduced to the same problems but for the linear connections  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$ .

Consequently, in the both cases, any problem concerning normal frames for mixed linear connections can equivalently be (re)formulated in terms of similar

<sup>&</sup>lt;sup>6</sup>From here immediately follows an independent proof of Proposition 11.7: If  $E'^{i}(E'_{j}) = E^{i}(E_{j})$ , then  $B^{-1} = A$  is a solution of (11.10b), so  $E''_{i} = E'_{i}$  and hence  $\{E'_{i}\}$  is a common normal frame for  ${}^{(1)}\nabla$  and  ${}^{(2)}\nabla$  which, by Proposition 11.1, implies  ${}^{(1)}\nabla = {}^{(2)}\nabla$ . The last result yields  $\nabla = {}^{(1)}\nabla = {}^{(2)}\nabla$ .

problem for ordinary linear connections. Since the last range of problems is explored at length in this book, we shall stop here with the investigation of the mixed linear connections from the view-point of normal frames.

**Exercise 11.1.** Reformulate *mutatis mutandis* the definitions and results of this section for the case of (normal) local coordinates or (normal) charts.

## 12. Normal frames for linear connections (review)

Since the linear connections find a far more numerous applications than the other derivations (along vector fields), in the present section are summarized the main ideas and are discussed the results concerning normal frames for linear connections which are spread over the preceding material in this chapter.

First of all, recall that a derivation along vector fields, which is linear on the whole manifold M, is a linear connection on M. More generally, a derivation D along vector fields which is linear on some subset(s) of M coincides on it (them) with a linear connection whose coefficients on this (these) set(s) coincide with the ones of D. This fact allows the results on normal frames for derivations along vector fields to be applied directly for 'local' linear connections, i.e., ones that are defined on a proper subset of the whole manifold M.

In Chapter II, a complete description of the frames and coordinates normal on submanifolds of M and along paths in M for linear connections defined on the whole manifold M was presented. In the present chapter (Sections 3–7) analogous results for derivations D along vector fields were proved. Frames normal on  $U \subseteq M$ for D exist iff  $D|_U = \nabla|_U$  where  $\nabla$  is linear connection on U or on a larger set (if  $U \neq M$ ) and  $\nabla$  admits frames normal on U; moreover, if such  $\nabla$  exists, all frames normal on U for D are normal on U for  $\nabla$  and *vice versa*. Reversing the situation, we have

Assertion 12.1. If  $\nabla$  is a linear connection defined on U, then all problems concerning the frames normal on  $U' \subseteq U$  for  $\nabla$ , if any, are equivalent to the same problems for a derivation D along vector fields such that  $D|_U = \nabla|_U$  and D is not linear everywhere on  $M \setminus U$ , i.e., outside U (if  $U \neq M$ ).

From here an essential conclusion follows.

**Conclusion 12.1.** The linear connections defined on M (resp. on subsets of M) are globally (resp. locally) the only derivations along vector fields for which frames normal on M (resp. on the corresponding subsets of M) could exist.

As we proved, normal frames always exist at a single point or along given (locally injective) path; on submanifolds of dimension at least two, normal frames exist as an exception iff some additional conditions, derived here, are satisfied.

In Section 8, we studied problems devoted to frames normal along (locally injective) mappings like  $f: N \to M, N$  and M being manifolds, for derivations

D along vector fields. The main results are expressed by Theorem 8.4, the first assertion of which can be restated as (see also Theorems 7.1 and II.5.2): A derivation D along vector fields admits frames normal along locally injective mapping  $f: N \to M$  if and only if for every (dim N)-dimensional submanifold N' of N such that  $f|_{N'}$  is injective, the restriction of D to f(N') (i.e., to  $T^1(f(N'))$ ) coincides with some linear connection (possibly restricted to f(N')) for which frames normal on f(N') exist. Therefore the exploration of frames normal along f for D can equivalently be reformulated in terms of linear connections. Reversing, ones again, the situation, we get

Assertion 12.2. If  $\nabla$  is a linear connection, all problems regarding frames normal along f for  $\nabla$  are equivalent to the same problems for a derivation D along vector fields such that  $D|_{f(N')} = \nabla|_{f(N')}$  and D is not linear on  $M \setminus f(N')$ .

Consequently a conclusion similar to the above one can be made.

**Conclusion 12.2.** Locally the linear connections are the only derivations along vector fields for which frames normal along locally injective mappings exist. If  $\dim_{\mathbb{R}} N = 0, 1$ , then such frames always exist, otherwise some additional conditions must be satisfied for their existence.

The only case, when the linear connections are not 'selected' from the viewpoint of normal frames or coordinates, is the one for derivations along fixed vector field (Section 9). This is not surprising, on the opposite such a result is quite natural as the linear connections are special type of derivations along (arbitrary) vector fields, not along fixed vector field. Of course, all results of Section 9 are completely valid for every covariant derivative  $\nabla_X$  along fixed vector field X.<sup>1</sup> We hope that they can be useful in theoretical physics where derivatives with respect to a fixed vector X find applications. The role of X can be played by a number of vectorial physical quantities, for instance it could be the velocity field of a fluid, some force field, the (four-)potential of an electromagnetic field, etc.

Since the derivations along a fixed path  $\gamma: J \to M$  can be considered as a suitable restriction of special kind of derivations along a fixed vector field X with  $X_{\gamma(J)} = \dot{\gamma}$ , the just said for such derivations is valid *mutatis mutandis* for derivations along a fixed path. The derivations along a fixed path always admit frames normal along that path (Theorem 10.1), in particular this is true for the covariant derivative  $\nabla_{\dot{\gamma}}$  along  $\gamma$  generated by some linear connection  $\nabla$ . An interesting result is the coincidence of the frames normal along  $\gamma$  for  $\nabla_{\dot{\gamma}}$  and for  $\nabla$  (Proposition 10.1). As derivatives like  $\nabla_{\dot{\gamma}}$  often arise in physics, where  $\dot{\gamma}$  is interpreted as a velocity of a (point) particle moving along  $\gamma$ , it is possible that the results of Section 10 could find suitable physical applications.

If derivations along arbitrary paths are considered, then, as one can expected, such derivations admit normal frames iff they are covariant derivatives generated

<sup>&</sup>lt;sup>1</sup>Following the terminology of Section 2 (see, in particular, Definition 2.3), one can call  $\nabla_X$  a linear connection along X: a linear connection  $\nabla$  is a derivation along vector fields which is linear at every point and  $\nabla_X$  is the value of  $\nabla$  at X (see Definition I.3.1).

by linear connections for which normal frames exist (Proposition 10.2). This reveals, once again, the privileged role of the linear connections between the other derivations along vector fields.

If one wants frames simultaneously normal for two derivations along vector fields (resp. along paths) to exist, then these derivations must (locally) coincide (according to Proposition 11.1 (resp. 11.5)) and, consequently, they are globally or locally identical linear connections (resp. identical covariant derivatives along paths) (see Proposition 10.2) induced by some linear connection (see also Proposition 10.1).

By Theorem 11.1, all problems regarding pairs of a frame and coframe normal for a mixed linear connection are equivalent to similar ones for a pair of linear connections (its 'contravariant' and 'covariant' components). If these linear connections admit normal frames, the pairs of a frame and coframe normal for the mixed linear connections exist. It is important to be noted, such pairs of a frame and coframe for a mixed linear connection, which is not (ordinary) linear connection, may exist if no additional conditions on the employed frames and coframes are imposed, which is a highly unusual situation. Practically everywhere, the considered frames  $\{E_i\}$  and coframes  $\{E^i\}$  are supposed to be dual,  $E^i(E_j) = \delta^i_j$ , or such that the functions  $f^i_j := E^i(E_j)$  to be invariant under changes of the frames and coframes. If this is the case, then pairs of a frame and coframe normal for a mixed linear connection may exist if it is an ordinary linear connection (Proposition 11.7).

## 13. Examples

In the previous sections, we have demonstrated that different problems concerning frames normal for derivations along (arbitrary) vector fields and paths are more or less reduced to similar problems for suitable linear connections. For that reason, the reader is referred to Sections I.7 and II.6 for instances of normal frames of this kind for linear connections.

Below will be presented several examples of frames normal for derivations along fixed vector field or fixed path. As it was mentioned earlier, these cases have no analogues in the theory of frames normal for linear connections (along arbitrary vector fields).

**Example 13.1 (Covariant derivatives).** If  $\nabla$  is a linear connection and X is a fixed vector field on a manifold M, then  $\nabla_X$  is a derivation along X whose coefficients in a frame  $\{E_i\}$  are  $\Gamma_X{}^i{}_j = \Gamma^i{}_{jk}X^k$  (see Proposition 2.2), where  $\Gamma^i{}_{jk}$  are the coefficients of  $\nabla$  in  $\{E_i\}$ .

Let M be of class  $C^1$  and  $p_0 \in M$ . If  $X_{p_0} = 0$ , then all frames/coordinates in a neighborhood of  $p_0$  are normal for  $\nabla_X$  at  $p_0$  as  $\Gamma_X{}^i{}_j(p_0) = 0$ . Suppose  $X_{p_0} \neq 0$ . Then, in the notation and suppositions of Theorems 9.1 or 9.2, all respectively frames  $\{E'_i = A^j_i E_j\}$  and coordinate systems  $\{z^i\}$  defined on a neighborhood of
$p_0$  and normal at  $p_0$  are such that (see (9.8) and (9.13))

$$\begin{aligned} A^{(x)}(p) &= A_0^{(x)} - \sum_k X_{p_0}^k \left[ \Gamma^i{}_{jk}(p_0) \right]_{i,j=1}^{\dim M} A_0^{(x)} [x^1(p) - x^1(p_0)] \\ &+ \sum_{k=2}^{\dim M} A_k^{(x)} [x^k(p) - x^k(p_0)] + A_{kl}^{(x)}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)] \quad (13.1) \\ z^i(p) &= a^i + a^i_j [x^j(p) - x^j(p_0)] + a^i_l X_{p_0}^j \Gamma^l{}_{kj}(p_0) [x^k(p) - x^k(p_0)] [x^1(p) - x^1(p_0)] \\ &+ \sum_{j,k=2}^{\dim M} a^i_{jk} [x^j(p) - x^j(p_0)] [x^k(p) - x^k(p_0)] \\ &+ a^i_{jkl}(p) [x^j(p) - x^j(p_0)] [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]. \quad (13.2) \end{aligned}$$

Similarly, applying Theorem 9.3 or Proposition 9.4, one can describe the frames or coordinates, respectively, (locally) normal for  $\nabla$  along a fixed path.

**Example 13.2 (Lie derivatives).** The Lie derivative  $\mathcal{L}$  (see Example 2.1 and Subsection IV.13.3 below) is a fundamental derivation along vector fields on any  $C^1$  manifold [11]. If  $X \in \mathfrak{X}^1(M)$ , the coefficients of the Lie derivative  $\mathcal{L}_X$  along (with respect to) X are defined by

$$\mathcal{L}_X E_i =: \Gamma_X^j{}_i E_j$$

in a frame  $\{E_i\}$  and, due to (2.1), are  $(X = X^i E_i)$ 

$$\Gamma_X{}^i{}_j = -E_j(X^i) + C^i_{kj}X^k.$$
(13.3)

Consequently,  $\mathcal{L}$  is nowhere linear derivation, precisely the only set  $U \subseteq M$  on which  $\Gamma_X{}^i{}_j = X^k \Gamma^i{}_{jk}$  for some functions  $\Gamma^i{}_{jk} \colon U \to \mathbb{K}$  and all X, is the empty set,  $U = \emptyset$ . Thus Proposition 3.1 implies the non-existence of frames normal for  $\mathcal{L}$  on any subset  $U \subseteq M$ .

Let now X be a fixed vector field. Suppose  $U \subseteq M$  and  $X|_U = 0$ . If  $E_j(X^i)|_U = 0$  (which is equivalent  $\mathcal{L}_X$  to be the zero derivation on U), then any frame or coordinate system defined on (a neighborhood of) U is normal for  $\mathcal{L}_X$  on U. Otherwise, i.e., if  $E_j(X^i)|_p \neq 0$  for some  $p \in U$ , frames/coordinates normal on U for  $\mathcal{L}_X$  do not exist.

Let  $p_0 \in U$  and  $X_{p_0} \neq 0$ . Then, in the notation and suppositions of Theorem 9.1 or 9.2, all respectively frames  $\{E'_i = A^j_i E_j\}$  and coordinate systems  $\{z^i\}$  defined on a neighborhood of  $p_0$  and normal at  $p_0$  are such that (see (9.8) and (9.13))

$$A^{(x)}(p) = A_0^{(x)} - \left[ (-E_j(X^i) + C_{mj}^i X^m)|_{p_0} \right]_{i,j=1}^{\dim M} A_0^{(x)} [x^1(p) - x^1(p_0)] + \sum_{k=2}^{\dim M} A_k^{(x)} [x^k(p) - x^k(p_0)] + A_{kl}^{(x)}(p) [x^k(p) - x^k(p_0)] [x^l(p) - x^l(p_0)]$$
(13.4)

$$z^{i}(p) = a^{i} + a^{i}_{j}[x^{j}(p) - x^{j}(p_{0})] + a^{i}_{l}(-E_{k}(X^{l}) + C^{l}_{mk}X^{m})[x^{k}(p) - x^{k}(p_{0})][x^{1}(p) - x^{1}(p_{0})] + \sum_{j,k=2}^{\dim M} a^{i}_{jk}[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})] + a^{i}_{jkl}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})][x^{l}(p) - x^{l}(p_{0})].$$
(13.5)

Similarly, applying Theorem 9.3 or Proposition 9.4, one can describe the frames or coordinates, respectively, (locally) normal for  $\mathcal{L}$  along a fixed path. For instance, if the condition (9.15) is not valid for a fixed (regular,  $C^1$ , and locally injective) path  $\gamma: J \to M$ , then all frames locally normal along  $\gamma$  are, in the notation of Theorem 9.3 and its proof,  $\{E'_i = A^j_i \frac{\partial}{\partial x^i}\}$ , where the matrix-valued function  $A = [A^i_j]$  is given by (9.22), which in the particular case reads (see (13.3))

$$A(x^{-1}(s,\sigma,t)) = A_0(x^{-1}(s,\sigma_0,t)) - \frac{\mathrm{d}\tau_{\gamma(s)}(\sigma)}{\mathrm{d}\sigma}\Big|_{\sigma=\sigma_0} \Big[ (-E_j(X^i) + C^i_{kj}X^k)|_{\gamma(s)} \Big]_{i,j=1}^{\dim M} A_0(\gamma(s))(\sigma-\sigma_0) + B(x^{-1}(s,\sigma,t))(\sigma-\sigma_0)^2.$$
(13.6)

**Example 13.3 (Tensor fields of type** (1, 1)). As we said in Example 2.1, a tensor field S of type (1, 1) on a manifold M can be considered as a derivation. Hence  $S: \mathfrak{X}(M) \ni X \mapsto S \in \mathfrak{T}_1^1(M)$  is a derivation along (relative to) vector fields; it can be called (generalized) Frenet-Serret derivative due to the considerations in Example 13.6 below. If X is a vector field and  $\{E_i\}$  is a frame, then (2.2) yields the following coefficients of S (considered as a derivation):

$$\Gamma_X{}^i{}_j = S^i_j. \tag{13.7}$$

Since the coefficients of S along X are independent of X, Proposition 3.1 implies a non-existence of frames/coordinates normal for S on any subset (unless S = 0).

However, if we consider X as an arbitrarily fixed vector field, frames or coordinates normal for S exist on any subset, according to the results of Section 9. For instance, Theorems 9.1 and 9.2 (resp. 9.3 and Proposition 9.4) with respectively  $\Gamma_X = [S_j^i]$  and  $\Gamma_X^{\ i}_{\ j} = S_j^i$  give the complete (local) description of all frames and coordinates, respectively, normal at a single point (resp. along a path).

If  $p_0 \in M$ , X is fixed and  $X_{p_0} \neq 0$ , then Theorem 9.4 says that there is a neighborhood U of  $p_0$  on which a frame  $\{E'_i = A^j_i \frac{\partial}{\partial x^j}\}$  is normal for S iff (see the notation in Theorem 9.4)

$$A(p) = Y(x^{1}(p), x^{1}(p_{0}); -[S_{j}^{i}(x^{-1}(\cdot, x^{2}(p), \dots, x^{\dim M}(p)))]_{i,j=1}^{\dim M}) \times A_{0}(x^{2}(p), \dots, x^{\dim M}(p)), \quad (13.8)$$

where Y is the solution of the initial-value problem

$$\frac{\partial Y}{\partial x^1}\Big|_p = -\left[S_j^i(x^{-1}(\cdot, x^2(p), \dots, x^{\dim M}(p)))\right]_{i,j=1}^{\dim M} Y, \qquad Y|_{x^1(p)=x^1(p_0)} = \mathbb{1}.$$
(13.9)

**Example 13.4 (General derivations and particular cases).** As we know from Definition 2.3, any derivation D along vector fields is a sum of the Lie derivative and a mapping  $S: \mathfrak{X}(M) \to \mathfrak{T}_1^1(M)$  assigning a tensor field  $S_X$  of type (1, 1) to a vector field X; the choice  $S_X = \Sigma_X$ , with  $\Sigma_X$  defined by (2.4), corresponds to a linear connection  $\nabla$ . According to (2.7), the coefficients of D and their matrix (at X) are

$$\Gamma_X{}^i{}_j = (S_X)^i_j - E_j(X^i) + C^i_{kj}X^k$$
(13.10)

$$\Gamma_X = \left[ (S_X)_j^i - E_j(X^i) + C_{kj}^i X^k \right]_{i,j=1}^{\dim M}.$$
(13.11)

Consequently, if X is fixed, frames and/or coordinates normal for D on some subset can be found by applying the results of Section 9, precisely by substituting in them the equations (13.10) and (13.11). We are not going to write the corresponding explicit expressions as the mentioned substitution is a trivial technical task.

Suppose on a manifold M is given a derivation  $D_X$  along a fixed vector field X and a linear connection  $\nabla$ . As a result of (2.3) and Example 2.2, we can write the representations

$$D_X = \mathcal{L}_X + S_X \tag{13.12a}$$

$$D_X = \nabla_X + S_X - \Sigma_X. \tag{13.12b}$$

The former equality (which is independent of the existence of  $\nabla$ ) is suitable for general mathematical considerations, while the latter one finds applications in some physical applications.

Here is a list of four derivations along fixed vector field X based on the decompositions (13.12):

$$D^{\text{F-W}} = \nabla_X - 2Q_X$$
 (Fermi-Walker derivative) (13.13a)

$$D^{\mathrm{F}} = \nabla_X^{\{\}} - 2\bar{Q}_X$$
 (Fermi derivative) (13.13b)

$$D^{\mathrm{T}} = \mathcal{L}_X + \theta \delta$$
 (Truesdell derivative) (13.13c)

$$D^{\mathrm{J}} = \nabla_X^{\{\}} - \omega$$
 (Jaumann derivative), (13.13d)

where  $\delta$  is the unit tensor field, the tensor fields  $Q_X$ ,  $\bar{Q}_X$  and  $\omega$  are type (1,1) and they, as well as the scalar function  $\theta$ , are defined below in Examples IV.13.3–IV.13.6 to which the reader is referred for further details.

**Exercise 13.1.** Calculate the coefficients of the derivations (13.13) and find all frames/coordinates normal for them, e.g., at a given point.

**Example 13.5 (Fermi derivative along a path).** Let  $D^{\rm F}$  be Fermi derivative, with a representation (13.13b), on a  $C^3$  Riemannian manifold M with metric g. Its restriction to a  $C^2$  path  $\gamma: J \to M$  results into the Fermi derivative  $D_{\gamma}^{\rm F}$  along  $\gamma$ ,

$$D_{\gamma}^{\mathrm{F}} = \nabla_{\dot{\gamma}}^{\{\}} - 2\bar{Q}_{\gamma}, \qquad (13.14)$$

where, according to Example IV.13.4 on page 297,

$$\begin{aligned} \left(\bar{Q}_{\gamma}\right)_{j}^{i} &= \dot{\gamma}^{i}(s)g_{jk}(\gamma(s))\left(\nabla_{\dot{\gamma}}^{\{\}}\dot{\gamma}\right)^{k}\Big|_{\gamma(s)} \\ &= \dot{\gamma}^{i}(s)g_{jk}(\gamma(s))\left(\frac{\mathrm{d}\dot{\gamma}^{k}(s)}{\mathrm{d}s} + \left\{\begin{matrix}k\\lm\end{matrix}\right\}\Big|_{\gamma(s)}\dot{\gamma}^{l}(s)\dot{\gamma}^{m}(s)\right) \end{aligned} \tag{13.15}$$

with the Christoffel symbols  $\binom{i}{ik}$  being defined by (I.4.14).

In particular, if  $\gamma$  is a geodesic, then  $\bar{Q}_{\gamma} = 0$ , by virtue of the equation of geodesics (I.4.17), and hence

$$D_{\gamma}^{\mathrm{F}} = \nabla_{\dot{\gamma}}^{\{\}}$$
 for a geodesic path  $\gamma$ . (13.16)

Combining this result with Proposition 10.1, we can assert that all frames and/or coordinates normal for the Fermi derivative along a (fixed) geodesic path  $\gamma$  coincide with the ones for the Riemannian connection  $\nabla^{\{\}}$  along the path  $\gamma$ . For further details on frames normal along a (geodesic) path for a linear connection, the reader is referred to Section II.3; for a concrete instance of this kind, see Example II.6.8.

In the general case, the components of the Fermi derivative (13.14) are (see (10.6))

$$\Gamma_{j}^{i}(s;\gamma) = \left\{ \frac{i}{jk} \right\} \Big|_{\gamma(s)} - 2\dot{\gamma}^{i}(s)g_{jk}(\gamma(s)) \Big( \frac{\mathrm{d}\dot{\gamma}^{k}(s)}{\mathrm{d}s} + \left\{ \frac{k}{lm} \right\} \Big|_{\gamma(s)} \dot{\gamma}^{l}(s)\dot{\gamma}^{m}(s) \Big).$$
(13.17)

Theorem 10.1, with  $\mathbf{\Gamma}_{\gamma} \circ \gamma = [\Gamma_{\gamma_j}^i \circ \gamma]$  defined via (13.17), gives a complete description of all frames on  $\gamma(J)$  which are normal for  $D_{\gamma}^{\mathrm{F}}$ . If one needs holonomic frames defined on a neighborhood of  $\gamma(J)$  and normal on  $\gamma(J)$ , the results of Lemma 10.1 should be applied.

**Example 13.6 (Generalized Frenet-Serret derivative along path).** The generalized Frenet-Serret derivative  $D_{\gamma}^{\text{F-S}}$  along a  $C^1$  path  $\gamma$  is a derivation along  $\gamma$  whose coefficients in a frame  $\{E_i\}$  (along  $\gamma$ ) are

$$\Gamma_j^i(s;\gamma) = S_j^i(\gamma(s)) \tag{13.18}$$

where  $S_j^i(\gamma(s))$  are the components at  $\gamma(s)$  in  $\{E_i\}$  of a tensor field S of type (1, 1).<sup>1</sup> Comparing (13.18) and (13.7), we see that the latter reduces to the former one for

<sup>&</sup>lt;sup>1</sup>The (usual) Frenet-Serret derivative is obtainable from (13.18) for a particular choice of  $S_{i}^{i}(\gamma(s))$  as a function of the curvature(s) and torsion(s) of the path  $\gamma$  – see [98, eq. (6)].

#### 14. Terminology 2: Normal and geodesic frames

 $X = \dot{\gamma}(s)$  at the point  $\gamma(s)$ . In this sense, the derivation determined by a tensor field of type (1, 1) can be called (generalized) Frenet-Serret derivation along vector fields; respectively, the restriction to a path of this derivative along the vector field tangent to the path results into the generalized Frenet-Serret derivative along that path.

According to Theorem 10.1, all frames  $\{E'_i = A^j_i E_j\}$  defined on  $\gamma(J)$  and normal for  $D^{\text{F-S}}_{\gamma}$  are such that

$$A(\gamma(s)) = Y(s, s_0; -[S_i^i \circ \gamma])A_0$$
(13.19)

with  $A_0$  being a constant non-degenerate matrix and Y being the solution of

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = -[S_j^i(\gamma(s))]Y \qquad Y|_{s=s_0} = 1;$$

in particular, if  $S_j^i(\gamma(s))$  is independent of s, then  $Y = \exp(S_j^i(\gamma(s))(s-s_0))$ . Holonomic frames  $\{\frac{\partial}{\partial y^i}\}$  defined on a neighborhood of  $\gamma(J)$  and normal on  $\gamma(J)$  for  $D_{\gamma}^{\text{F-S}}$  can be constructed as pointed in the proof of Lemma 10.1, viz. one should define  $B_j^i$  via (10.13) with  $E_i'$  for  $E_i$ , then the coordinates  $y^i$  are given by (10.14).

## 14. Terminology 2: Normal and geodesic frames

We have called normal some special kinds of local bases, frames, charts, and coordinates investigated in the present book. This needs some explanations which are given below.

For symmetric linear connections, the local coordinates in which their components vanish at a given point are called normal in [70, Chapter V, Section 3] or in [20, § 11.6]. In [11, Chapter III, § 8] and in [71, p. 278] the local coordinates normal at a point, introduced there via the exponential mapping (see Subsection II.2.3) for linear connection (symmetric or not), are defined as such for which the symmetric part of the connection's components vanish at that point (see Proposition II.2.3). Evidently, the latter definition includes the former one as a special case. Note that the both definitions originate from the consideration of the equation (I.3.22) of geodesic paths [11, 70, 71]. This is the primary reason for calling these local coordinates geodesic (or Riemannian, or normal Riemannian [20, § 11.5]) in the special case of a Riemannian manifold [51, § 42, p. 201], where they are (sometimes) equivalently introduced via the condition that in them the partial derivatives of the metric's components vanish at a given point [51, § 42] (cf. Proposition I.6.1).

The case of a symmetric linear connection is investigated in [19, Chapter III, § 7, pp. 156–158] (see the references therein too). There is made a distinction between geodesic and normal at a point local coordinates. Geodesic coordinates are called the ones in which the connection's coefficients vanish at that point

and normal coordinates are called the geodesic ones satisfying at the given point equation (7.23) of [19, Chapter III, § 7] which, in particular, implies the vanishment at that point of the connection's coefficients together with their symmetrized partial derivatives.<sup>1</sup> (Note that the possibility for the existence of the last type of coordinates is ensured by our (non-)uniqueness result expressed by Proposition I.5.2 with which is compatible the mentioned equation; see also equations (I.6.11) and (II.3.10) and Proposition II.2.3.) Analogous opinion is shared in [99, pp. 13–14].

It is known that the symmetric parts of the connection coefficients of arbitrary linear connection  $\nabla$  are directly linked with the equation (I.3.22) of geodesic lines (curves, paths) and uniquely determine them [11, 19]. By our opinion, this suggests the following convenient convention. Call normal or respectively geodesic on a set U a local coordinate system (chart, frame), defined in a neighborhood of U, in which the local components of  $\nabla$  or respectively their symmetric parts vanish on U. Thus, in the torsion free case, the concepts of normal and geodesic coordinate system coincide. Generally a normal frame is geodesic, the converse being not valid. In this sense, the normal coordinates described in [19, p. 158] are a special type of (our) normal coordinates, specified by the additional conditions described in this reference. These conditions are consistent with Proposition I.5.2. Note that the proposed definition is in accordance with the special one used in [100].

If one adopts the suggested convention, then the generalization from linear connections to arbitrary derivations D (along vector fields, fixed vector field, paths, fixed path) of the tensor algebra over a manifold is evident: only the concept of a normal frame is applicable because, generally, of some symmetry properties of the components of D cannot be spoken about.<sup>2</sup> This explains the terminology accepted in the present monograph.

Let us mention that the so-defined normal frames on U for D have a connection with a kind of generalized geodesic paths corresponding to D (cf. [101]) which will be discussed briefly in Section IV.15.

## 15. Conclusion

As the title of the present chapter indicates, it deals with different kinds of problems concerning frames (coordinates, charts) normal for derivations (along vector fields, fixed vector field, paths, fixed path) of the tensor algebra over (vector fields on) a differentiable manifold. Excluding the cases along fixed vector field or fixed path, the general conclusion is that a derivation admits frames normal on some set if and only if on it the derivation coincides with some linear connection (with the

 $<sup>^1\</sup>mathrm{Cf.}$  the derived here similar equations (I.6.11) and (II.3.10) for respectively the Riemannian coordinates and Fermi coordinates.

<sup>&</sup>lt;sup>2</sup>If D is derivation along arbitrary vector fields or paths and it admits frames normal on U, then one can speak of the symmetry properties of the coefficients of D on U as D is linear on U (Proposition 3.1).

same coefficients as the derivation on the set given) for which normal frames exist. This fact singles out the linear connections, defined locally or globally, as the only derivations which admit normal frames in the above sense. The last conclusion is important for the theoretical physics and explains to a great extent why the linear connections find so many applications in it. The reason is, from certain positions, quite simple: due to the equivalence principle, the 'normal frames' are the mathematical concept for describing the physical notion of 'inertial frame' [97].

Most of the material in this chapter is new and written for the present book. Exceptions are some results in Sections 4–7 which are taken from [76,80,83],<sup>1</sup> but the proofs are completely revised or new.

The main results obtained can be summarized as follows: Necessary and sufficient conditions for the existence of frames or coordinates normal at a single point, along locally injective paths or, more generally, mappings, and on submanifolds for derivations along either arbitrary or fixed vector field or path are found. Frames normal at a single point or along a path always exist. On other sets normal frames exist iff some additional conditions are satisfied, exception being the case of a derivation along a fixed vector field, when normal frames exist on every subset of the manifold or, in particular, on the whole manifold. When normal frames or coordinates exist, their complete constructive description is given.<sup>2</sup>

<sup>&</sup>lt;sup>1</sup>These papers are revised journal versions of the earlier works [84–86], respectively.

 $<sup>^{2}</sup>$ Exceptions are some cases of a derivation along fixed vector field, in which complicated systems of partial differential equations arise.

## Chapter IV

# Normal Frames in Vector Bundles

The theory of linear transports along paths in vector bungeneralizing the parallel transports gendles. erated by linear connections, is developed. The normal frames for them are defined as ones in which their matrices are the identity one. A number of results, including theorems of existence and uniqueness, concerning normal frames are derived. Special attention is paid to the case when the bundle's base is a manifold. The normal frames are defined and investigated also for derivations along paths and along tangent vector fields in the last case. Frames normal at a single point or along a given path always exist. On other subsets normal frames exist only in the curvature free case. The privileged role of the parallel transports is pointed out in this context.  $\heartsuit$ 

## 1. Introduction

The analysis of Corollary II.5.1 on page 123 reveals that the properties of the parallel transport assigned to a linear connection, not directly the ones of the connection itself, are responsible for the existence of frames normal on a submanifold for the connection.<sup>1</sup> This observation forms the groundwork of the idea the 'normal' frames to be defined directly for (parallel) transports without referring to the concept of a (linear) connection (or some other derivation along vector fields). The main obstacle for the realization of such an approach to 'normal frames' is that, ordinary, the concept of a parallel transport is a secondary one, it is introduced on the base of the concept of a (linear) connection. To the solution of the last problem and the development of the mentioned approach to normal frames (in finite-dimensional vector bundles) is devoted the present chapter of the book. As we shall demonstrate below, the consistent realization of the above idea leads to a completely new look on the 'normal frames', which is self-contained and incorporates as special cases all of the results of the preceding chapters.

The material in Sections 3–6 and 8 is based on the work [102] and the one after them is practically new and written especially for the present book.<sup>2</sup>

In the present chapter is studied a wide range of problems concerning frames normal for linear transports and derivations along paths in vector bundles and for derivations along tangent vector fields in the case when the bundle's base is a differentiable manifold. In the last case, when tangent bundles are concerned, the only general result, known to the author and regarding normal frames, is [23, p. 102, Theorem 2.106].

The structure of this chapter is as follows.

Section 2 introduces some basic concepts from the theory of (fibre) bundles, in particular of the one of vector bundles, required for the investigations in this chapter. After the concepts of bundle, section, and vector bundle are fixed, a special attention to the ones of liftings of paths and derivations along paths, which will play an important role further, is paid. The tensor bundles over a manifold are pointed as particular examples of vector bundles. Details on these and many other concepts regarding (fibre) bundles, the reader can fined in the monographs [7, 11, 106–110].

Section 3 is devoted to the general theory of linear transports along paths in vector fibre bundles which is a far reaching generalization of the theory of parallel transports generated by linear connections.<sup>3</sup> The general form and other properties of these transports are studied. A bijective correspondence between them and derivations along paths is established. In Section 4, the normal frames are de-

<sup>&</sup>lt;sup>1</sup>Here the situation is similar to the one described in the second paragraph of Section II.1 on page 74: the properties of the Christoffel symbols, not directly the ones of the Riemannian metric generating them, are fully responsible for the existence of coordinates normal at a single point in a Riemannian manifold.

<sup>&</sup>lt;sup>2</sup>Although, some initial ideas and results are borrowed from the papers [103–105].

<sup>&</sup>lt;sup>3</sup>This section is based on the early works [101, 105, 111-115] of the author. For some more general results, see Chapter V.

fined as ones in which the matrix of a linear transport along paths is the identity (unit) one or, equivalently, in which its coefficients, as defined in Section 3, vanish 'locally'. A number of properties of the normal frames are found. In Section 5 is explored the problem of existence of normal frames. Several necessary and sufficient conditions for such existence are proved and the explicit construction of normal frames, if any, is presented.

Section 6 concentrates on, possibly, the most important special case of frames normal for linear transports or derivations along smooth paths in vector bundles with a differentiable manifold as a base. A specific necessary and sufficient condition for existence of normal frames in that case is proved. In particular, normal frames may exist only for those linear transports or derivations along paths whose (2–index) coefficients linearly depend on the vector tangent to the path along which they act. Obviously, this is a generalization of the derivation along curves assigned to a linear connection. Section 7 examplifies the theory in a case of line bundles. Section 8 is devoted to problems concerning frames normal for derivations along tangent vector fields in a bundle with a manifold as a base. Necessary and sufficient conditions for the existence of these frames are derived. The conclusion is made that there is a one-to-one onto correspondence between the sets of linear transports along paths, derivations along paths, and derivations along tangent vector fields all of which admit normal frames.

In the first part of Section 9, based on [103], the concept of a curvature of a linear transport along paths is introduced and some its properties are explored. In its second part, relations between the curvature of a linear transports along paths and the frames normal for them are studied. The main result is that only the curvature free transports admit normal frames. The concept of a torsion of a linear transport along paths in the tangent bundle over a manifold is introduced in Section 10 (cf. the early paper [103]). Links between the torsion and holonomic normal frames are investigated. The vanishment of the torsion is pointed as a necessary and sufficient condition for existence of normal coordinates on submanifolds. If such coordinates exist, their complete description is given.

Section 11 deals with parallel transports in the tangent bundles over manifolds and frames normal for these transports. It is shown that the parallel transport assigned to a linear connection is a special kind of a linear transport in tangent bundles. As a side result, an axiomatic definition of a parallel transport is obtained, on the base of which a new definition of a linear connection, equivalent to the usual one, is given. The flat parallel transports are pointed as the only linear transports along paths in tangent bundles which transports admit normal frames. The coordinates normal for flat and torsionless parallel transports are explicitly presented.

Section 12 concerns a special type of normal frames in which the 3-index coefficients, if any, of a linear transport along paths vanish.

Section 13 is similar to Section 11, but it deals with the interrelations between different types of derivations along vector fields over a manifold and the linear

transports along paths in the tangent bundle over it. As examples, particular derivations or transports, such as Fermi-Walker, Jaumann, etc., are considered.

The aim of Section 14 is twofold. On one hand (Subsections 14.1–14.3), the rigorous relations between the theory of linear transports along paths in vector bundles and the one of parallel transports and connections in these bundles are investigated. On the base of the axiomatic approach to the theory of parallel transports, as presented in [23], we show how it (and hence the one of connections) is incorporated as a special case in the general theory of linear transports along paths. On another hand (Subsections 14.4 and 14.5), we demonstrate how the results concerning normal frames and derived for linear connections on manifolds and linear transports along paths are almost *in extenso* applicable to the theory of parallel transports and connections on vector bundles.

In Section 15 is introduced the notion of autoparallel paths in manifolds whose tangent bundle is endowed with a linear transport along paths. If this transport is a parallel one, it is proved that the autoparallels coincide with the geodesics of the linear connection generating the transport.

Section 16 gives an idea of the role of the linear transports along paths and normal frames in a fibre bundle formulation of quantum mechanics.

The chapter ends with some notes in Section 17.

All fibre bundles in this chapter are vectorial ones. The base and total bundle space of such bundles can be general topological spaces. However, if some kind of differentiation in one/both of these spaces is needed to be introduced (considered), it/they should possess a smooth structure; if this is the case, we require it/they to be smooth, of class  $C^1$ , differentiable manifold(s). Starting from Section 6, the base and total bundle space are supposed to be  $C^1$  manifolds. Sections 3–5 do not depend on the existence of a smoothness structure in the bundle's base. Smoothness of the bundle space is partially required in Sections 2–5.<sup>4</sup>

## 2. Vector bundles

The purpose of the present section is the introduction of some simple basic concepts of the theory of (fibre) bundles, in particular of vector bundles, on which this chapter rests.

The most general intuitive idea of a bundle is the one of an object consisting of a set (space) to each point of which is 'attached' some other set (space). Hence a bundle should be thought as a mapping whose values are some sets, the attached sets. On the sets forming a bundle may be imposed a large number of 'reasonable' conditions whose goal is particular bundles to be obtained for the exploration of more or less concrete problems; for instance, the sets can be topological spaces, in

<sup>&</sup>lt;sup>4</sup>The bundle space is required to be a  $C^1$  manifold in Section 2 (starting from Definition 2.1), in Definition 4.1', in Proposition 4.1–4.2, if (4.1c) and (4.1d) are taken into account, in Theorem 5.2, and in Proposition 5.6.

particular manifolds. Often the attached sets are vector (linear) spaces, in which case the bundle is called vector bundle. For quite a big class of problems, the notion of a 'smooth' bundle is required: the idea is when one 'moves' in the initial set, the corresponding attached sets (or selected points in them) to change smoothly.

A particular example of a smooth vector bundle is the tangent bundle to a manifold M: its basic set (space) is the manifold M itself and the set attached at  $p \in M$  is the space  $T_p(M)$  tangent to M at p. Similar structure have the tensor bundles of different types over M. If  $f: M \to \mathbb{K}$  is a smooth function, a completely trivial smooth bundle structure over M arises by assigning to each  $p \in M$  the value f(p) of f at p.

#### 2.1. Basic definitions

A bundle is a triple  $(E, \pi, B)$  of sets E and B, called (total) bundle space and base (space) respectively, and (generally) surjective mapping  $\pi: E \to B$ , called projection.<sup>1</sup> For every  $b \in B$ , the set  $\pi^{-1}(b)$  is called the fibre over b.

If  $X \subseteq B$ , the bundle  $(E, \pi, B)|_X := (\pi^{-1}(X), \pi|_{\pi^{-1}(X)}, X)$  is called the *restriction to* X of a bundle  $(E, \pi, B)$ .

A section of the bundle  $(E, \pi, B)$  is a mapping  $\sigma: B \to E$  such that  $\pi \circ \sigma = id_B$ , i.e.,  $\sigma: b \mapsto \sigma(b) \in \pi^{-1}(b)$ . The set of sections of  $(E, \pi, B)$  is denoted by  $Sec(E, \pi, B)$ .

If E and B are topological spaces, which is the most widely considered case, the bundle  $(E, \pi, B)$  is called *topological*. In this case in the definition of a bundle is included the *bundle property*: there exists a (topological) space  $\mathcal{E}$  such that, for each  $b \in B$ , there is an open set ('directory space')  $W \ni b$  in B and homeomorphism ('decomposition function')  $\phi_W : W \times \mathcal{E} \to \pi^{-1}(W)$  of  $W \times \mathcal{E}$  onto  $\pi^{-1}(W)$ satisfying the condition  $(\pi \circ \phi_W)(w, e) = w$  for  $w \in W$  and  $e \in \mathcal{E}$ . Besides, if the restriction  $\phi_W|_b : \{b\} \times \mathcal{E} \to \pi^{-1}(b), b \in B$ , is homeomorphism, the bundle property is called *local triviality*,  $\mathcal{E}$  is called (*typical, standard*) fibre of the bundle, and the fibre  $\pi^{-1}(b)$  is homeomorphic to  $\mathcal{E}$  for every  $b \in B$ .

A vector bundle is a locally trivial bundle  $(E, \pi, B)$  such that: (i) the fibres  $\pi^{-1}(b), b \in B$ , and the standard fibre  $\mathcal{E}$  are (linearly) isomorphic vector spaces and (ii) the decomposition mappings  $\phi_W$  and their restrictions  $\phi_W|_b$  are (linear) isomorphisms between vector spaces. The dimension of  $\mathcal{E}$ , dim  $\mathcal{E} = \dim \pi^{-1}(b)$  for all  $b \in B$ , is called the (fibre) dimension of the vector bundle, resp. it is called (dim  $\mathcal{E}$ )-dimensional. Here the vector spaces are considered over the field  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ .

All bundles in this chapter are supposed to be vectorial.

<sup>&</sup>lt;sup>1</sup>The notation  $(E, \pi, B)$  for a bundle comes from the one  $E \xrightarrow{\pi} B$  for a mapping  $\pi$  from E on B.

#### 2.2. Liftings of paths

A lifting<sup>2</sup> (in a vector bundle  $(E, \pi, B)$ ) of a mapping  $g: X \to B$  to E, X being a set, is a mapping  $\overline{g}: X \to E$  such that  $\pi \circ \overline{g} = g$ . The set of all liftings of a mapping  $g: X \to B$  will be denoted by  $\operatorname{Lift}_g(E, \pi, B) := \{\overline{g}: X \to E | \pi \circ \overline{g} = g\}$ . In particular, the liftings of the identity mapping  $\operatorname{id}_B$  of the base B are called sections and their set is  $\operatorname{Sec}(E, \pi, B) := \operatorname{Lift}_{\operatorname{id}_B}(E, \pi, B) = \{\sigma | \sigma : B \to E, \pi \circ \sigma = \operatorname{id}_B\}$ (see Subsection 2.1).

Let  $P(A) := \{\gamma | \gamma \colon J \to A\}$  and  $PLift(E, \pi, B) := \{\lambda | \lambda \colon P(B) \to P(E), (\pi \circ \lambda)(\gamma) = \gamma \text{ for } \gamma \in P(B)\}$  be respectively the set of paths in a set A and the set of liftings of paths from B to  $E^{.3}$  The set  $PLift(E, \pi, B)$  is: (i) A natural K-vector space if we put  $(a\lambda + b\mu) \colon \gamma \mapsto a\lambda_{\gamma} + b\mu_{\gamma}$  for  $a, b \in \mathbb{K}, \lambda, \mu \in PLift(E, \pi, B)$ , and  $\gamma \in P(B)$ , where, for brevity, we write  $\lambda_{\gamma}$  for  $\lambda(\gamma), \lambda \colon \gamma \mapsto \lambda_{\gamma}$ ; (ii) A natural left module with respect to K-valued functions on B: if  $f, g \colon B \to \mathbb{K}$ , we define  $(f\lambda + g\mu) \colon \gamma \mapsto (f\lambda)_{\gamma} + (g\mu)_{\gamma}$  with  $(f\lambda)_{\gamma}(s) \coloneqq f(\gamma(s))\lambda_{\gamma}(s)$  for  $\gamma \colon J \to B$  and  $s \in J$ ; (iii) A left module with respect to the set  $PF(B) \coloneqq \{\varphi | \varphi \colon \gamma \mapsto \varphi_{\gamma}, \gamma \colon J \to B, \varphi_{\gamma} \colon J \to \mathbb{K}\}$  of functions along paths in the base B: for  $\varphi, \psi \in PF(B)$ , we set  $(\varphi\lambda + \psi\mu) \colon \gamma \mapsto (\varphi\lambda)_{\gamma} + (\psi\mu)_{\gamma}$  where  $(\varphi\lambda)_{\gamma}(s) \coloneqq (\varphi_{\gamma}\lambda_{\gamma})(s) \coloneqq \varphi_{\gamma}(s)\lambda_{\gamma}(s)$ .

If we consider  $\text{PLift}(E, \pi, B)$  as a  $\mathbb{K}$ -vector space, its dimension is equal to infinity. If we regard  $\text{PLift}(E, \pi, B)$  as a left PF(B)-module, its rank is equal to the dimension of  $(E, \pi, B)$  (i.e., to the dimension of the fibre(s) of  $(E, \pi, B)$ ). In the last case, a basis in  $\text{PLift}(E, \pi, B)$  can be constructed as follows (cf. [116, Proposition 2.1.14] in a case of sections of a vector bundle).

For every path  $\gamma: J \to B$  and a point  $s \in J$ , choose a basis  $\{e_i(s;\gamma)|i = 1, \ldots, \dim \pi^{-1}(\gamma(s))\}$  in the fibre  $\pi^{-1}(\gamma(s))$ ; if the total space E is a  $C^r$  manifold, we suppose  $e_i(s;\gamma)$  to have a  $C^r$  dependence on s. Define liftings along paths  $\hat{e}_i \in \operatorname{PLift}(E, \pi, B)$  by  $\hat{e}_i: \gamma \mapsto \hat{e}_i|_{\gamma} := e_i(\cdot;\gamma)$ , i.e.,  $\hat{e}_i|_{\gamma}: s \mapsto \hat{e}_i|_{\gamma}(s) := e_i(s;\gamma)$ . The set  $\{\hat{e}_i\}$  is a basis in  $\operatorname{PLift}(E, \pi, B)$ , i.e., for every  $\lambda \in \operatorname{PLift}(E, \pi, B)$  there are  $\lambda^i \in \operatorname{PF}(B)$  such that  $\lambda = \sum_i \lambda^i \hat{e}_i$  and  $\{\hat{e}_i\}$  are  $\operatorname{PF}(B)$ -linearly independent. Actually, for any path  $\gamma: J \to B$  and number  $s \in J$ , we have  $\lambda_{\gamma}(s) \in \pi^{-1}(\gamma(s))$ , so there exists numbers  $\lambda^i_{\gamma}(s) \in \mathbb{K}$  such that  $\lambda_{\gamma}(s) = \sum_i \lambda^i_{\gamma}(s)e_i(s;\gamma)$ . Defining  $\lambda^i \in \operatorname{PF}(B)$  by  $\lambda^i: \gamma \mapsto \lambda^i_{\gamma}$  with  $\lambda^i_{\gamma}: s \mapsto \lambda^i_{\gamma}(s)$ , we get  $\lambda = \sum_i \lambda^i \hat{e}_i$ ; if  $e_i(\cdot;\gamma)$  is of class  $C^r$ , so are  $\lambda^i_{\gamma}$ . The  $\operatorname{PF}(B)$ -linear independence of  $\{\hat{e}_i(s;\gamma)\}$ . As we notice above, if E is  $C^r$  manifold, we choose  $\hat{e}_i$ , i.e.,  $\hat{e}_i|_{\gamma}$ , to be  $C^r$  and, consequently, the components  $\lambda^i$ , i.e.,  $\lambda^i_{\gamma}$ , are of class  $C^r$  too.

Let  $(E, \pi, B)$  be a vector bundle of dimension  $n \in \mathbb{N}$ ,  $n < \infty$ ,  $U \subseteq B$ , and  $g: Q \to B$  with  $Q \neq \emptyset$ . A frame on (over) U (resp. along g) is a set  $\{e_i | i = 1, ..., n\}$  of  $n \ F(U)$ -linearly (resp. F(g(Q))-linearly) independent sections of  $(E, \pi, B)|_U$  (resp. of  $(E, \pi, B)|_{g(Q)}$ ), where  $F(U) := \{f: f: U \to \mathbb{K}\}$  is the

<sup>&</sup>lt;sup>2</sup>For detail see, e.g., [108].

<sup>&</sup>lt;sup>3</sup>Every linear transport *L* along paths (vide infra Section 3, in particular Definition 3.3) provides a lifting of paths: for every  $\gamma: J \to B$  fix some  $s \in J$  and  $u \in \pi^{-1}(\gamma(s))$ , the mapping  $\gamma \mapsto \overline{\gamma}_{s;u}$  with  $\overline{\gamma}_{s;u}(t) := L_{s \to t}^{\gamma} u, t \in J$  is a lifting of paths from *B* to *E*.

set of functions on U. Said differently,  $\{e_i\}$  is a frame on U (resp. along g) if  $e_i \in \text{Sec}((E, \pi, B)|_U)$  (resp.  $e_i \in \text{Sec}((E, \pi, B)|_{g(Q)})$ ) and, for every  $p \in U$  (resp.  $p \in g(Q)$ ), the set of vectors  $\{e_i|_p\}$  is a basis in  $\pi^{-1}(p)$ .

#### 2.3. Derivations along paths

Let  $(E, \pi, B)$  be a vector bundle whose bundle space E is  $C^r$ ,  $r \in \mathbb{N}$ , manifold. A lifting  $\lambda \in \operatorname{PLift}(E, \pi, B)$  is said to be of class  $C^k$ ,  $k = 0, 1, \ldots, r$ , if in some (and hence in any)  $C^k$  frame in  $\operatorname{PLift}(E, \pi, B)$  its components are of class  $C^k$  along paths in some set of paths, i.e.,  $\lambda$  is of class  $C^k$  if  $\lambda_{\gamma}$  is a  $C^k$  path for a path  $\gamma$ in that set. Obviously, not every path in B has a  $C^k$  lifting in E; for instance, all liftings of a discontinuous path in B are discontinuous paths in E. The set of paths in B having  $C^k$  liftings in E is  $\pi \circ \operatorname{P}^k(E) := \{\pi \circ \overline{\gamma} | \overline{\gamma} \in \operatorname{P}^k(E)\}$ , with  $\operatorname{P}^k(E)$  being the set of  $C^k$  paths in E. Therefore, when talking of  $C^k$  liftings, we shall implicitly assume that they are acting on paths in  $\pi \circ \operatorname{P}^k(E) \subset \operatorname{P}(B)$ . The discontinuous paths in B are, of course, not in  $\pi \circ \operatorname{P}^k(E)$ , so that they are excluded from the considerations below.

Analogously,  $\varphi \in PF(B)$  is of class  $C^k$  if  $\varphi_{\gamma}$  is of class  $C^k$  for a path  $\gamma$  in some set of paths. Denote by  $PLift^k(E, \pi, B), k = 0, 1, ..., r$ , the set of  $C^k$  liftings of paths from B to E and by  $PF^k(B), k = 0, 1, ..., r$ , the set of  $C^k$  functions along paths in B.

If the base B is  $C^{r'}$ ,  $r' \in \mathbb{N}$ , manifold, we denote by  $\operatorname{Sec}^{k}(E, \pi, B)$ ,  $k = 0, 1, \ldots, r'$ , the set of  $C^{k}$  sections of the bundle  $(E, \pi, B)$ .

**Definition 2.1.** A derivation along paths in  $(E, \pi, B)$  or a derivation of liftings of paths in  $(E, \pi, B)$  is a mapping

$$D: \operatorname{PLift}^{1}(E, \pi, B) \to \operatorname{PLift}^{0}(E, \pi, B)$$
 (2.1a)

which is K-linear,

$$D(a\lambda + b\mu) = aD(\lambda) + bD(\mu)$$
(2.2a)

for  $a, b \in \mathbb{K}$  and  $\lambda, \mu \in \text{PLift}^1(E, \pi, B)$ , and the mapping

$$D_s^{\gamma}$$
: PLift<sup>1</sup>( $E, \pi, B$ )  $\rightarrow \pi^{-1}(\gamma(s)),$  (2.1b)

defined via  $D_s^{\gamma}(\lambda) := ((D(\lambda))(\gamma))(s) = (D\lambda)_{\gamma}(s)$  and called *derivation along*  $\gamma: J \to B$  at  $s \in J$ , satisfies the 'Leibnitz rule':

$$D_s^{\gamma}(f\lambda) = \frac{\mathrm{d}f_{\gamma}(s)}{\mathrm{d}s}\lambda_{\gamma}(s) + f_{\gamma}(s)D_s^{\gamma}(\lambda)$$
(2.2b)

for every  $f \in PF^1(B)$ . The mapping

$$D^{\gamma}$$
: PLift<sup>1</sup>( $E, \pi, B$ )  $\rightarrow P(\pi^{-1}(\gamma(J))),$  (2.1c)

defined by  $D^{\gamma}(\lambda) := (D(\lambda))|_{\gamma} = (D\lambda)_{\gamma}$ , is called a *derivation along*  $\gamma$ .

Let  $\{e_i\}$  be a frame along paths  $\gamma$ , i.e., for any  $\gamma: J \to B$  and  $s \in J$ ,  $e_i: (s; \gamma) \mapsto e_i(s; \gamma)$  and  $\{e_i(s; \gamma)\}$  is a basis in  $\pi^{-1}(\gamma(s))$  that can depend on s and  $\gamma$ . The components<sup>4</sup>  $\Gamma^i_{\ j}(s; \gamma)$  of a derivation D along paths (along  $\gamma$  at s in  $\{e_i\}$ ) are defined via the expansion

$$D_s^{\gamma} \hat{e}_j =: \Gamma^i_{\ j}(s;\gamma) e_i(s;\gamma), \tag{2.3}$$

where  $\hat{e}_i: \gamma \mapsto e_i(\cdot; \gamma)$  is a lifting of paths, generated by  $e_i$  according to (2.8) below, and the dimension of  $(E, \pi, B)$  is assumed to be finite. If  $\lambda \in \text{PLift}^1(E, \pi, B)$  has an expansion  $\lambda = \lambda^i \hat{e}_i$ , then the properties (2.2) imply the explicit local expansion

$$D_s^{\gamma}\lambda = \sum_i \left(\frac{\mathrm{d}\lambda_{\gamma}^i(s)}{\mathrm{d}s} + \Gamma^i_{\ j}(s;\gamma)\lambda_{\gamma}^j(s)\right)e_j(s;\gamma).$$
(2.4)

Hence the components of D in a given frame uniquely define it.

From the last equation, we get the explicit expansion for  $D\lambda$ :

$$D(\lambda) = (\dot{\lambda}^i + \Gamma^i_{\ j} \lambda^j) \hat{e}_i \tag{2.5}$$

where  $\dot{\lambda}^i, \Gamma^i_{\ j} \in \operatorname{PF}(B)$  and  $\hat{e}_i \in \operatorname{PLift}(E, \pi, B)$  are such that

$$\dot{\lambda}^{i} \colon \gamma \mapsto \dot{\lambda}^{i}_{\gamma} \colon s \mapsto \frac{\mathrm{d}\lambda^{i}_{\gamma}(s)}{\mathrm{d}s} \tag{2.6}$$

$$\Gamma^{i}_{j} \colon \gamma \mapsto \Gamma^{i}_{j}(\,\cdot\,;\gamma) \colon s \mapsto \Gamma^{i}_{j}(s;\gamma) \tag{2.7}$$

$$\hat{e}_i \colon \gamma \mapsto e_i(\,\cdot\,;\gamma) \colon s \mapsto e_i(s;\gamma). \tag{2.8}$$

Similarly, the derivation of  $\lambda$  along  $\gamma$  is

$$D^{\gamma}(\lambda) := (D\lambda)(\gamma) = (\dot{\lambda}^{i}_{\gamma} + \Gamma^{i}_{\ j}(\cdot;\gamma)\lambda^{j}_{\gamma})e_{i}(\cdot;\gamma).$$
(2.9)

If the frame  $\{e_i\}$  is changed by means of a non-degenerate matrix-valued function  $A = [A_i^j], \{e_i(s;\gamma)\} \mapsto \{e'_i(s;\gamma) = A_i^j(s;\gamma)e_j(s;\gamma)\}$ , from (2.3) and (2.2), we see that the matrix  $\mathbf{\Gamma}(s;\gamma) := [\Gamma_j^i(s;\gamma)]$  of D transform into

$$\Gamma'(s;\gamma) := [\Gamma'_{j}^{i}(s;\gamma)] = A^{-1}(s;\gamma)\Gamma(s;\gamma)A(s;\gamma) + A^{-1}(s;\gamma)\frac{\mathrm{d}A(s;\gamma)}{\mathrm{d}s}.$$
 (2.10)

Applying (2.3), one can easily prove that every system  $\Gamma^{i}_{j} \in PF(B)$  of functions along paths,  $\Gamma^{i}_{j}: \gamma \mapsto \Gamma^{i}_{j}(\cdot; \gamma): s \mapsto \Gamma^{i}_{j}(s; \gamma)$ , which transform according to (2.10), defines via (2.4) a derivation D along paths in  $(E, \pi, B)$ . Combining this result with the above one, we can assert the existence of a bijective correspondence

<sup>&</sup>lt;sup>4</sup>In connection with the theory of normal frames (see Section 4 and further), it is convenient to call  $\Gamma_{j}^{i}(s;\gamma)$  also (2-index) coefficients of  $D^{\gamma}$ . This agrees with the fact that  $\Gamma_{j}^{i}$  are coefficients of some linear transport along paths (see below).

between the set of functions  $\{\Gamma_{j}^{i}\}$  along paths with transformation law (2.10) and the set of derivations along paths of a vector bundle  $(E, \pi, B)$ .

The set  $\operatorname{PSec}(E, \pi, B)$  of sections along paths of a bundle  $(E, \pi, B)$  consists of mappings  $\boldsymbol{\sigma} \colon \gamma \mapsto \boldsymbol{\sigma}_{\gamma}$  assigning to every path  $\gamma \colon J \to B$  a section  $\boldsymbol{\sigma}_{\gamma} \in \operatorname{Sec}((E, \pi, B)|_{\gamma(J)})$  of the bundle restricted to  $\gamma(J)$ . Every (ordinary) section  $\sigma \in \operatorname{Sec}(E, \pi, B)$  generates a section  $\boldsymbol{\sigma}$  along paths via  $\boldsymbol{\sigma} \colon \gamma \mapsto \boldsymbol{\sigma}_{\gamma} \coloneqq \sigma|_{\gamma(J)}$ , i.e.,  $\boldsymbol{\sigma}_{\gamma}$  is simply the restriction of  $\sigma$  to  $\gamma(J)$ ; hence  $\boldsymbol{\sigma}_{\alpha} = \boldsymbol{\sigma}_{\gamma}$  for every path  $\alpha \colon J_{\alpha} \to B$  with  $\alpha(J_{\alpha}) = \gamma(J)$ . Every  $\boldsymbol{\sigma} \in \operatorname{PSec}(E, \pi, B)$  generates a lifting  $\hat{\boldsymbol{\sigma}} \in \operatorname{PLift}(E, \pi, B)$  by  $\hat{\boldsymbol{\sigma}} \colon \gamma \mapsto \hat{\boldsymbol{\sigma}}_{\gamma} \coloneqq \boldsymbol{\sigma}_{\gamma} \circ \gamma$ ; in particular, the lifting  $\hat{\sigma}$  associated to  $\sigma \in \operatorname{Sec}(E, \pi, B)$  is given via  $\hat{\sigma} \colon \gamma \mapsto \hat{\boldsymbol{\sigma}}_{\gamma} = \boldsymbol{\sigma}|_{\gamma(J)} \circ \gamma$ .

If B is a manifold, every derivation D along paths generates a mapping

$$\overline{D}$$
: PSec<sup>1</sup>( $E, \pi, B$ )  $\rightarrow$  PLift<sup>0</sup>( $E, \pi, B$ )

such that, if  $\boldsymbol{\sigma} \in \operatorname{PSec}^1(E, \pi, B)$ , then  $\overline{D}: \boldsymbol{\sigma} \mapsto \overline{D}\boldsymbol{\sigma} = \overline{D}(\boldsymbol{\sigma})$  where  $\overline{D}\boldsymbol{\sigma}: \gamma \mapsto \overline{D}^{\gamma}\boldsymbol{\sigma}$ is a lifting of paths defined by  $\overline{D}^{\gamma}\boldsymbol{\sigma}: s \mapsto (\overline{D}^{\gamma}\boldsymbol{\sigma})(s) := D_s^{\gamma}\hat{\boldsymbol{\sigma}}$  with  $\hat{\boldsymbol{\sigma}}$  being the lifting generated by  $\boldsymbol{\sigma}$ , i.e.,  $\gamma \mapsto \hat{\boldsymbol{\sigma}}_{\gamma} := \boldsymbol{\sigma}_{\gamma} \circ \gamma$ . The mapping  $\overline{D}$  may be called a *derivation of*  $C^1$  sections along paths. Notice, if  $\gamma: J \to B$  has intersection points and  $x_0 \in \gamma(J)$  is such a point, the mapping  $\gamma(J) \to \pi^{-1}(\gamma(J))$  given by  $x \mapsto \{D_s^{\gamma}(\hat{\boldsymbol{\sigma}}) | \gamma(s) = x, s \in J\}, x \in \gamma(J)$ , is generally multiple-valued at  $x_0$  and, consequently, is not a section of  $(E, \pi, B)|_{\gamma(J)}$ .

If B is a  $C^1$  manifold and for some  $\gamma: J \to B$  there exists a subinterval  $J' \subseteq J$  on which the restricted path  $\gamma|J: J' \to B$  is without self-intersections, i.e.,  $\gamma(s) \neq \gamma(t)$  for  $s, t \in J'$  and  $s \neq t$ , we can define the *derivation along*  $\gamma$  *of sections* over  $\gamma(J')$  as a mapping

$$\mathsf{D}^{\gamma} \colon \operatorname{Sec}^{1}\left((E, \pi, B)|_{\gamma(J')}\right) \to \operatorname{Sec}^{0}\left((E, \pi, B)|_{\gamma(J')}\right)$$
(2.11)

such that

$$(\mathsf{D}^{\gamma}\sigma)(x) := D_s^{\gamma}\hat{\sigma} \quad \text{for } x = \gamma(s)$$
 (2.12)

where  $s \in J'$  is unique for a given  $x \in \gamma(J')$  and  $\hat{\sigma} \in \text{PLift}((E, \pi, B)|_{\gamma(J')})$ is given by  $\hat{\sigma} = \sigma|_{\gamma(J')} \circ \gamma|_{J'}$ . Generally the mapping (2.11) defined by (2.12) is multiple-valued at the points of self-intersections of  $\gamma$ , if any, as  $(\mathsf{D}^{\gamma}\sigma)(x) := \{D_s^{\gamma}\hat{\sigma} : s \in J, \ \gamma(s) = x\}$ . The so-defined mapping  $\mathsf{D} : \gamma \mapsto \mathsf{D}^{\gamma}$  is called a *sectionderivation* along paths.<sup>5</sup> As we said, it is single-valued only along paths without self-intersections.

Let *D* be a derivation along paths in a vector bundle  $(E, \pi, B)$ ,  $\gamma: J \to B$ be injective, *D* be the generated by *D* section-derivation along paths, and  $\sigma \in$ Sec<sup>1</sup> $(E, \pi, B)|_{\gamma(J)}$ . As a consequence of (2.12) and (2.4), we have

$$(\mathsf{D}^{\gamma}\sigma)|_{\gamma(s)} = \sum_{i} \left[ \frac{\mathrm{d}\sigma^{i}(\gamma(s))}{\mathrm{d}s} + \Gamma^{i}{}_{j}(s;\gamma)\sigma^{j}(\gamma(s)) \right] e_{i}(s;\gamma)$$
(2.13)

 $<sup>^{5}</sup>$ A particular example of a section-derivation along injective  $C^{1}$  paths is the derivation along paths provided by Definition III.10.1. For details – see Remark III.10.1 and Subsection 13.1.

where  $\{e_i(s;\gamma)\}$  is a basis in  $\pi^{-1}(\gamma(s)), \sigma(\gamma(s)) =: \sigma^i(\gamma(s))e_i(s;\gamma)$ , and  $\Gamma^i_{\ j}(s;\gamma)$  are the components of D (or of D) in  $\{e_i\}$ .

Generally a section along paths or lifting of paths does not define a (single-valued) section of the bundle as well as to a lifting along paths there does not correspond some (single-valued) section along paths. The last case admits one important special exception, viz. if a lifting  $\lambda$  is such that the lifted path  $\lambda_{\gamma}$  is an 'exact topological copy' of the underlying path  $\gamma: J \to B$ , i.e., if there exist  $s, t \in J, s \neq t$  for which  $\gamma(s) = \gamma(t)$ , then  $\lambda_{\gamma}(s) = \lambda_{\gamma}(t)$ , which means that if  $\gamma$  has intersection points, then the lifting  $\lambda_{\gamma}$  also possesses such points and they are in the fibres over the corresponding intersection points of  $\gamma$ . Such a lifting  $\lambda$  generates a section  $\overline{\lambda} \in \operatorname{PSec}(E, \pi, B)$  along paths given by  $\overline{\lambda}: \gamma \mapsto \overline{\lambda}_{\gamma}$  with  $\overline{\lambda}: \gamma(s) \mapsto \lambda_{\gamma}(s)$ . In the general case, the mapping  $\gamma(s) \mapsto \lambda_{\gamma}(s)$  for a lifting  $\lambda$  of paths is multiple-valued at the points of self-intersection of  $\gamma: J \to B$ , if any; for injective path  $\gamma$ , this mapping is a section of  $(E, \pi, B)|_{\gamma(J)}$ . Such mappings will be called *multiple-valued sections along paths*.

#### 2.4. Tensor bundles

Over every differentiable manifold M exists a collection of natural bundles, viz. the different tensor bundles with M as a base, uniquely related to the differentiable structure of M.

The tangent bundle  $(T(M), \pi, M)$ , called also bundle tangent to M, is a well explored example of a tensor bundle over a manifold M. Its base is the manifold Mitself, the bundle space T(M) is the disjoint union of the spaces tangent to M (see Subsection I.2.3), i.e.,  $T(M) := \bigcup_{p \in M} T_p(M)$ , and the projection  $\pi : T(M) \to M$ is defined by  $\pi(X_p) := p$  if  $X_p \in T_p(M)$ . So the fibre of  $(T(M), \pi, M)$  over  $p \in M$ is exactly the space tangent to M at p,  $\pi^{-1}(p) = T_p(M)$ . It is a trivial exercise to be verified the equivalence

$$X \in \mathfrak{X}(M) \iff X \in \operatorname{Sec}(T(M), \pi, M),$$

i.e., the vector fields over a manifold are simply sections of its tangent bundle, and, consequently,

$$\mathfrak{X}(M) = \operatorname{Sec}(T(M), \pi, M).$$

Respectively, the sections along paths of the tangent bundle are mappings assigning to a path  $\gamma: J \to M$  a vector field over  $\gamma(J)$  etc.

Since each space  $T_p(M)$  is  $(\dim M)$ -dimensional vector (linear) space, the tangent bundle  $(T(M), \pi, M)$  is  $(\dim M)$ -dimensional vector bundle for which  $\mathbb{K}^{\dim M}$  can be taken as a (standard) fibre.

Similar to the tangent bundle is the cotangent bundle  $(T^*(M), \pi^*, M)$  whose bundle space is  $T^*(M) := \bigcup_{p \in M} T_p^*(M)$  and the projection  $\pi^*$  is such that  $\pi^*(\omega_p) := p$  if  $\omega_p \in T_p^*(M)$  where  $T_p^*(M)$  is the cotangent space to M at p(see Subsection I.2.4).

#### 2. Vector bundles

The tensor bundle  $(T_s^r(M), \pi_s^r, M)$  of type  $(r, s), r, s \in \mathbb{N} \cup \{0\}$  is a natural generalization of the tangent and cotangent bundles. Its bundle space is the disjoint union of the tensor spaces of type (r, s) over M,

$$T_s^r(M) := \bigcup_{p \in M} T_{p_s}^{\ r}(M)$$

with  $T_{p_s}^r(M)$  being the tensor space of type (r, s) at  $p \in M$  (see Subsection I.2.5). The projection  $\pi_s^r \colon T_s^r(M) \to M$  is defined by

$$\pi_s^r(K_p) := p \quad \text{for } K_p \in T_{p_s}^r(M).$$

Therefore the dimension of  $(T_s^r(M), \pi_s^r, M)$  is  $(\dim M)^{r+s} = \dim T_{p_s}^r(M)$  and  $\mathbb{K}^{(\dim M)^{r+s}}$  can be taken as its standard fibre. The choices (r, s) = (1, 0) and (r, s) = (0, 1) correspond to the tangent and cotangent bundles, respectively. Since  $T_{p_0}^0 \equiv \mathbb{K}$ , the tensor bundle of type (0, 0) is  $(M \times \mathbb{K}, \pi_0^0, M)$  with  $\pi_0^0(p, a) := p$  for  $(p, a) \in M \times \mathbb{K}$ .

Obviously, the tensor fields over M are sections of the corresponding tensor bundles and *vice versa*,

$$\mathfrak{T}_{s}^{r}(M) = \operatorname{Sec}(T_{s}^{r}(M), \pi_{s}^{r}, M).$$

The algebraic tensor bundle  $(\mathbf{T}(M), \boldsymbol{\pi}, M)$  on (over) a manifold M is also an example of a bundle structure over manifolds. By definition, its bundle space is the disjoint union of the tensor algebras over M,  $\mathbf{T} := \bigcup_{p \in M} \mathbf{T}_p(M)$ , and its projection is such that  $\boldsymbol{\pi}(K_p) := p$  if  $K_p \in \mathbf{T}_p(M)$  for  $p \in M$ , i.e., the fibre over pis the tensor algebra at  $p, \pi^{-1}(p) = \mathbf{T}_p(M)$ . All tensor fields are specific (of type (r, s)) sections of the algebraic tensor bundle.

One can expect the existence of a link between the derivations along vector fields of the tensor algebra over M (see Definition III.10.1 on page 191) and the derivations along paths of the tensor bundles over M. Since the former derivations are (uniquely) defined only along injective paths, the mentioned connection could be expressed via the section-derivations along paths of the tensor bundles over the manifold M (see (2.11) and (2.12)).

Let *D* be a derivation along paths of T(M),  $\gamma: J \to M$  be injective, and *K* be a  $C^1$  tensor field(=section) of type (r, q) along  $\gamma$ . Locally, in some frame along  $\gamma$ , the derivative  $D^{\gamma}(K)$  of *K* along  $\gamma$  is given by (III.10.7) in which  $\Gamma^i_{\ j}(s; \gamma)$  are the components of *D* along  $\gamma$  at  $s \in J$ .

Let now D be a section-derivation along paths, generated by a derivation along paths, in the tensor bundle  $(T_q^r(M), \pi_q^r, M)$ . If  $K \in \mathfrak{T}_q^r(M)$ , then, in the tensor frame (I.2.41) induced by a frame  $\{E_i\}$ , equation (2.13), with K for  $\sigma$  and this frame for  $\{e_i\}$ , reads<sup>6</sup>

$$(\mathsf{D}^{\gamma}K)(\gamma(s)) = \sum_{\substack{i_1,\dots,i_r\\j_1,\dots,j_q}} \left[ \frac{\mathrm{d}K^{i_1,\dots,i_r}_{j_1,\dots,j_q}(\gamma(s))}{\mathrm{d}s} + \Gamma^{i_1,\dots,i_r,k_1,\dots,k_q}_{j_1,\dots,j_r}(s;\gamma)K^{l_1,\dots,l_r}_{k_1,\dots,k_q}(\gamma(s)) \right] E^{j_1,\dots,j_q}_{i_1,\dots,i_r}|_{\gamma(s)}.$$
 (2.14)

This expression is identical for every K with the one corresponding to the righthand side of (III.10.7) if and only if the components of the section-derivation D have the form

$$\Gamma_{j_{1},\dots,i_{q},l_{1},\dots,l_{r}}^{i_{1},\dots,i_{r},k_{1},\dots,k_{q}}(s;\gamma) = \left(\sum_{a=1}^{r} \delta_{l_{1}}^{i_{1}} \cdots \delta_{l_{a-1}}^{i_{a-1}} \Gamma^{i_{a}}{}_{l_{a}}(s;\gamma) \delta_{l_{a+1}}^{i_{a+1}} \cdots \delta_{l_{r}}^{i_{r}}\right) \prod_{b=1}^{q} \delta_{j_{b}}^{k_{b}} - \left(\prod_{a=1}^{q} \delta_{l_{a}}^{i_{a}}\right) \sum_{b=1}^{q} \delta_{j_{1}}^{k_{1}} \cdots \delta_{j_{b-1}}^{k_{b-1}} \Gamma^{k_{b}}{}_{i_{b}}(s;\gamma) \delta_{j_{b+1}}^{k_{b+1}} \cdots \delta_{j_{q}}^{k_{q}} \quad (2.15)$$

where  $\Gamma_{l}^{i}$  are the components of a (section-)derivation along paths in the tangent bundle  $(T(M), \pi, M)$ .

Thus we have proved that a derivation along vector fields of the tensor algebra over a manifold M is equivalent to a set of (section-)derivations along paths, one in each tensor bundle  $(T_q^r(M), \pi_q^r, M)$  with  $q, r \in \mathbb{N} \cup \{0\}$ , whose components have in some (and hence in every) frame the form (2.15); the corresponding (section-) derivation along paths in  $(T_q^r(M), \pi_q^r, M)$  being equal to the restriction of the initial derivation along paths of T(M) to the bundle space  $T_q^r(M)$  of the tensor bundle of type (r, q).

**Example 2.1.** A particular example of the above derivations is provided by the one generated by a linear connection  $\nabla$  over M. As we said earlier in Section III.10, any linear connection  $\nabla$  generates along every  $C^1$  path  $\gamma$  a derivation  $\nabla_{\dot{\gamma}}$  along  $\gamma$  (of the tensor algebra along  $\gamma$ ), locally given via (III.10.5). Taking into account the afore-said, we can assert that in the tensor bundle of type (r, q) every connection  $\nabla$  generates a derivation along paths with local coefficients (2.15) in which

$$\Gamma^{i}_{\ j}(s;\gamma) = \Gamma^{i}_{\ jk}(\gamma(s))\dot{\gamma}^{k}(s) \tag{2.16}$$

where  $\Gamma^{i}_{\ ik}$  are the coefficients of  $\nabla$ .

<sup>&</sup>lt;sup>6</sup>Notice, for r = q = 0 equation (2.14) implies  $(\mathsf{D}^{\gamma}K)(\gamma(s)) = \frac{\mathrm{d}K(\gamma(s))}{\mathrm{d}s} = \dot{\gamma}(s)(K)$ , i.e., on scalar functions  $\mathsf{D}^{\gamma}$  acts as ordinary directional derivative, as it should be.

## 3. Linear transports along paths in vector bundles

In the majority of the mathematical literature, the concept of a connection (in locally trivial differentiable bundles of class  $C^1$ ) is taken as a basic one and the concept of a parallel transport is introduced by its means (see, e.g., [6, 10–13, 16, 28, 106, 117]). However, the opposite approach is also known [17, 23, 30–33]: in it the 'parallel transport' is axiomatically defined and on its base the concept of a 'connection' is introduced as a secondary one.<sup>1</sup> It seems that the earliest written accounts on this approach are the ones due to Ü.G. Lumiste [30, Section 2.2] and C. Teleman [17, Chapter IV, Section B.3] (both published in 1964), the next essential steps being made by P. Dombrowski [31, § 1] and W. Poor [23].<sup>2</sup> The comparison and the features of the both approaches to the theory of connections and parallel transports is not a subject of the present book,<sup>3</sup> but, for completeness, it is partially investigated in Section V.8.

In the present section, the concept of a 'linear transport along paths in vector bundles' is axiomatically defined and some its simple properties are derived. This concept is a straightforward generalization of the one of a parallel transport assigned to a linear connection (see Section 11) and some other transports defined on its base (see Section 13). As we shall see later (see Definitions 11.2 and 11.3 and the comments after them), on its base independent, but equivalent, definition of a parallel transport and linear connection can be given. Moreover, in Subsection 14.3 will be shown that the parallel transports in general vector bundles are a special type of linear transports along paths, as a consequence of which the whole theory of connections and covariant derivatives can be deduced from the theory of linear transports in these bundles.

#### **3.1.** Definition and general form

Let  $(E, \pi, B)$  be a K-vector bundle<sup>4</sup> with bundle (total) space E, base B, projection  $\pi: E \to B$ , and homeomorphic fibres  $\pi^{-1}(p), p \in B$ . Whenever some kind of differentiation in E is considered, the bundle space E will be required to be a  $C^1$  differentiable manifold. The base B is supposed to be a general topological space in Sections 3–5 and from Section 6 onwards is required to be a  $C^1$  differentiable manifold. By J and  $\gamma: J \to B$  are denoted a real interval and path in B,

 $<sup>^{1}</sup>$ A summary of the realization of this approach in [23] is presented in Subsection 14.1 below; see also [118].

<sup>&</sup>lt;sup>2</sup>The author of [31] states that his paper is based on unpublished lectures of prof. W. Rinow in 1949. See also [23, p. 46] where the author claims that the first axiomatical definition of a parallel transport in the tangent bundle case is given by prof. W. Rinow in his lectures at the Humboldt University in 1949. Some heuristic comments on the axiomatic approach to parallel transport theory can be found in [8, Section 2.1] too.

<sup>&</sup>lt;sup>3</sup>For some details, see [115, 118] and the references given therein.

 $<sup>^{4}</sup>$ Most of our results are valid for vector bundles over more general fields too but this is inessential for the following.

respectively. The paths considered are generally *not* supposed to be continuous or differentiable unless their differentiability class is stated explicitly.

**Definition 3.1.** A linear transport along paths in the bundle  $(E, \pi, B)$  is a mapping L assigning to every path  $\gamma$  a mapping  $L^{\gamma}$ , transport along  $\gamma$ , such that  $L^{\gamma} \colon (s, t) \mapsto L_{s \to t}^{\gamma}$  where the mapping

$$L_{s \to t}^{\gamma} \colon \pi^{-1}(\gamma(s)) \to \pi^{-1}(\gamma(t)) \qquad s, t \in J,$$
(3.1)

called transport along  $\gamma$  from s to t, has the properties:

$$L_{s \to t}^{\gamma} \circ L_{r \to s}^{\gamma} = L_{r \to t}^{\gamma}, \qquad r, s, t \in J, \qquad (3.2)$$

$$L_{s \to s}^{\gamma} = \operatorname{id}_{\pi^{-1}(\gamma(s))}, \qquad \qquad s \in J, \qquad (3.3)$$

$$L_{s \to t}^{\gamma}(\lambda u + \mu v) = \lambda L_{s \to t}^{\gamma} u + \mu L_{s \to t}^{\gamma} v, \qquad \lambda, \mu \in \mathbb{K}, \quad u, v \in \pi^{-1}(\gamma(s)).$$
(3.4)

Remark 3.1. Equations (3.2) and (3.3) mean that L is a transport along paths in the bundle  $(E, \pi, B)$  (see Definition V.8.1 below or [114, Definition 2.1]), which may be an arbitrary topological bundle, not only a vector one in the general case,<sup>5</sup> while (3.4) specifies that it is *linear* [114, equation (2.8)]. In the present book, with an exception of Section V.8, only linear transports in vector bundles will be explored.

Remark 3.2. Definition 3.1 is a generalization of the concept of 'linear connection' given, e.g., in [31, Section 1.2] (see especially [31, p. 138, axiom  $(L_1)$ ]) which practically defines the covariant derivative in terms of linear transports along paths (see (3.29) below which is equivalent to [31, p. 138, axiom  $(L_3)$ ]). Our definition is much weaker; e.g., we completely drop [31, p. 138, axiom  $(L_3)$ ] and use, if required, weaker smoothness conditions. An excellent introduction to the theory of vector bundles and the parallel transports in them can be found in the book [23]. In particular, in this reference is proved the equivalence of the concepts parallel transport, connection and covariant derivative operator in vector bundles (as defined there). Analogous results concerning linear transports along paths will be presented below. The detailed comparison of Definition 3.1 with analogous ones in the literature is not a subject of this work.

Further, in Subsection 13.3 (see also Definition 3.1, [105, Definition 2.1] and [105, Proposition 4.1]), it will be proved that special types of linear trans-

<sup>&</sup>lt;sup>5</sup>The definition of a connection in a topological bundle  $(E, \pi, B)$  in [17, Chapter IV, Section B.3] is, in fact, an axiomatic definition of a parallel transport. If we neglect the continuity condition in this definition, it defines a connection in  $(E, \pi, B)$  as a mapping  $C: (\gamma, q) \mapsto C(\gamma, q)$  assigning to any continuous path  $\gamma: [0, 1] \to B$  and a point  $q \in \pi^{-1}(\gamma(0))$  a path  $C(\gamma, q): [0, 1] \to E$  such that  $C(\gamma, q)|_0 = q$  and  $\pi \circ C(\gamma, q) = \gamma$ . If I is a transport along paths in  $(E, \pi, B)$ , then  $C: (\gamma, q) \mapsto C(\gamma, q): t \mapsto C(\gamma, q)|_t = I_{0 \to t}^{\gamma}(q)$  defines a connection C in  $(E, \pi, B)$  in the sense mentioned. Moreover, if this definition is broadened by replacing [0, 1] with an arbitrary and not fixed closed interval [a, b], with  $a, b \in \mathbb{R}$  and  $a \leq b$ , then the converse is also true, i.e.,  $C(\gamma, q)|_t = I_{a \to t}^{\gamma}(q)$ ,  $t \in [a, b]$ , for some transport I. However, the proof of this statement is not trivial; cf. similar considerations in Subsections 3.2 and 3.3 below, which deal with analogous problems in vector bundles (with other definition of a parallel transport).

ports along paths are: the parallel transport assigned to a linear connection (covariant derivative) of the tensor algebra of a manifold [11, 19],<sup>6</sup> Fermi-Walker transport [62, 119], Fermi transport [62], Truesdell transport [120, 121], Jaumann transport [122], Lie transport [19, 119], the modified Fermi-Walker and Frenet-Serret transports [98], etc. Consequently, Definition 3.1 is general enough to cover a list of important transports used in theoretical physics and mathematics. Thus studying the properties of the linear transports along paths, we can make corresponding conclusions for any one of the transports mentioned.<sup>7</sup>

From (3.2) and (3.3), we get that  $L_{s \to t}^{\gamma}$  are invertible mappings and

$$\left(L_{s\to t}^{\gamma}\right)^{-1} = L_{t\to s}^{\gamma}, \qquad s, t \in J.$$

$$(3.5)$$

Hence the linear transports along paths are in fact linear isomorphisms of the fibres over the path along which they act.

The following two propositions establish the general structure of linear transports along paths.<sup>8</sup>

**Proposition 3.1.** A mapping (3.1) is a linear transport along  $\gamma$  from s to t for every  $s, t \in J$  if and only if there exist a vector space V, isomorphic with  $\pi^{-1}(x)$  for all  $x \in B$ , and a family  $\{F(s; \gamma) : \pi^{-1}(\gamma(s)) \to V, s \in J\}$  of linear isomorphisms such that

$$L_{s \to t}^{\gamma} = F^{-1}(t;\gamma) \circ F(s;\gamma), \qquad s,t \in J.$$
(3.6)

Proof. If (3.1) is a linear transport along  $\gamma$  from s to t, then fixing some  $s_0 \in J$  and using (3.3) and (3.5), we get  $L_{s \to t}^{\gamma} = L_{s_0 \to t}^{\gamma} \circ L_{s \to s_0}^{\gamma} = \left(L_{t \to s_0}^{\gamma}\right)^{-1} \circ L_{s \to s_0}^{\gamma}$ . So (3.6) holds for  $V = \pi^{-1}(\gamma(s_0))$  and  $F(s; \gamma) = L_{s \to s_0}^{\gamma}$ . Conversely, if (3.6) is valid for some linear isomorphisms  $F(s; \gamma)$ , then a straightforward calculation shows that it converts (3.2) and (3.3) into identities and (3.4) holds due to the linearity of  $F(s; \gamma)$ .

**Proposition 3.2.** Let a representation (3.6) for some vector space V and linear isomorphisms  $F(s;\gamma): \pi^{-1}(\gamma(s)) \to V$ ,  $s \in J$  be given for a linear transport along paths in the bundle  $(E, \pi, B)$ . For a vector space \*V, there exist linear isomorphisms  $*F(s;\gamma): \pi^{-1}(\gamma(s)) \to *V$ ,  $s \in J$ , for which

$$L_{s \to t}^{\gamma} = {}^{\star}F^{-1}(t;\gamma) \circ {}^{\star}F(s;\gamma), \qquad s, t \in J,$$
(3.7)

 $<sup>^{6}</sup>$ For the proof, see Proposition 11.1 on page 282 below.

<sup>&</sup>lt;sup>7</sup>The concept of linear transport along paths in vector bundles can be generalized to the transports along paths in arbitrary bundles [114] and to transports along mappings in bundles [123]. An interesting consideration of the concept of (parallel) 'transport' (along closed paths) in connection with homotopy theory and the classification problem of bundles can be found in [124]. These generalizations are out of the scope of the present book.

<sup>&</sup>lt;sup>8</sup>Particular examples of Proposition 3.1 are known for parallel transports in vector bundles. For instance, Proposition 1 in [125, p. 240] realizes it for parallel transport in a bundle associated to a principal one and induced by a connection in the latter case; see also the proof of the lemma in the proof of Proposition 1.1 in [11, Chapter III, § 1], where a similar result is obtained implicitly.

iff there exists a linear isomorphism  $D(\gamma) \colon V \to {}^*V$  such that

$${}^{*}F(s;\gamma) = D(\gamma) \circ F(s;\gamma), \qquad s \in J.$$
(3.8)

*Proof.* If equation (3.8) holds, the substitution of  $F(s;\gamma) = D^{-1}(\gamma) \circ {}^*F(s;\gamma)$  into (3.6) yields (3.7). Vice versa, if (3.7) is valid, then from its comparison with (3.6) follows that  $D(\gamma) = {}^*F(t;\gamma) \circ (F(t;\gamma))^{-1} = {}^*F(s;\gamma) \circ (F(s;\gamma))^{-1}$  is the required (independent of  $s, t \in J$ ) isomorphism.

Starting from this point, we shall investigate further only the finite-dimensional case,  $\dim \pi^{-1}(x) = \dim \pi^{-1}(y) < \infty$  for all  $x, y \in B$ . In this way we shall avoid a great number of specific problems arising when the fibres have infinite dimension (see, e.g., [126] for details). A lot of our results are valid, possibly *mu*tatis mutandis, in the infinite-dimensional treatment too. One way for transferring results from finite to infinite-dimensional spaces is the direct limit from the first to the second ones. Then, for instance, if the bundle's dimension is countably or uncountably infinite, the corresponding sums must be replaced by series or integrals whose convergence, however, requires special exploration [126]. Linear transports along paths in infinite-dimensional vector bundles naturally arise, e.g., in the fibre bundle formulation of quantum mechanics [127–131]. Generally, there are many difficulties with the infinite-dimensional problem which deserves a separate investigation.

#### **3.2.** Representations in frames along paths

Let  $\{e_i(s;\gamma)\}$  be a basis in  $\pi^{-1}(\gamma(s)), s \in J$ .<sup>9</sup> So, along  $\gamma: J \to B$  we have a set  $\{e_i\}$  of bases on  $\pi^{-1}(\gamma(J))$ , i.e.,  $\{e_i\}$  is a frame along  $\gamma$ .<sup>10</sup> The dependence of  $e_i(s;\gamma)$  on s is inessential if we are interested only in the *algebraic* properties of the linear transports along paths. The mapping  $s \mapsto e_i(s;\gamma)$  will be required to be of class  $C^1$  if some kind of differentiation of liftings of paths will be considered.

The matrix  $\mathbf{L}(t,s;\gamma) := \left[L^{i}_{j}(t,s;\gamma)\right]$  (along  $\gamma$  at (s,t) in  $\{e_{i}\}$ ) of a linear transport L along  $\gamma$  from s to t is defined via the expansion<sup>11</sup>

$$L_{s \to t}^{\gamma} \left( e_i(s;\gamma) \right) =: L_i^j(t,s;\gamma) e_j(t;\gamma) \qquad s, t \in J.$$
(3.9)

We call  $L: (t, s; \gamma) \to L(t, s; \gamma)$  the matrix (function) of L; respectively  $L_i^j$  are its matrix elements or components in the given frame.

It is almost evident that

$$L^{j}_{i}(t,s;\gamma)e_{j}(t;\gamma) \otimes e^{i}(s;\gamma) \in \pi^{-1}(\gamma(t)) \otimes \left(\pi^{-1}(\gamma(s))\right)^{*}$$
(3.10)

<sup>&</sup>lt;sup>9</sup>Here and henceforth the Latin indices run from 1 to dim  $\pi^{-1}(p)$ ,  $p \in B$ .

<sup>&</sup>lt;sup>10</sup>Regardless of the fact that  $\{e_i(s;\gamma)\}$  is a basis in  $\pi^{-1}(\gamma(s))$ , we write  $e_i(s;\gamma)$ , not  $e_i(\gamma(s))$ , as we consider  $e_i: \gamma \mapsto e_i(\cdot;\gamma)$  as a lifting of paths; hence  $e_i(s;\gamma)$  generally depends on s and  $\gamma$ , not only on the combination  $\gamma(s)$ . If  $\gamma$  has self-intersections, then the mapping  $p \mapsto e_i(s;\gamma)$ , with  $p \in \gamma(J)$  and  $s \in J$  such that  $p = \gamma(s)$ , is, generally, multiple-valued at these points.

<sup>&</sup>lt;sup>11</sup>Notice the different positions of the arguments s and t in  $L_{s \to t}^{\gamma}$  and in  $L(t, s; \gamma)$ .

#### 3. Linear transports in vector bundles

where the asterisk (\*) denotes dual objects and  $e^i(s;\gamma) := (e_i(s;\gamma))^*$ . Hence the change of the bases  $\{e_i(s;\gamma)\} \mapsto \{e'_i(s;\gamma) := A^j_i(s;\gamma)e_j(s;\gamma)\}, s \in J$ , by means of a non-degenerate matrix  $A(s;\gamma) := [A^j_i(s;\gamma)]$ , implies

$$\boldsymbol{L}(t,s;\gamma) \mapsto \boldsymbol{L}'(t,s;\gamma) = A^{-1}(t;\gamma)\boldsymbol{L}(t,s;\gamma)A(s;\gamma)$$
(3.11)

or in component form

$${L'}^{j}_{i}(t,s;\gamma) = \left(A^{-1}(t;\gamma)\right)^{j}_{k} L^{k}_{\ l}(t,s;\gamma) A^{l}_{i}(s;\gamma).$$
(3.11')

Evidently, for  $u = u^i(s; \gamma) e_i(s; \gamma) \in \pi^{-1}(\gamma(s))$ , due to (3.4), we have

$$L_{s \to t}^{\gamma} u = \left( L_{i}^{j}(t,s;\gamma) u^{i}(s;\gamma) \right) e_{j}(t;\gamma).$$
(3.12)

In terms of the matrix L of L, the basic equations (3.2) and (3.3) read respectively

$$\boldsymbol{L}(t,s;\gamma)\boldsymbol{L}(s,r;\gamma) = \boldsymbol{L}(t,r;\gamma) \qquad r,s,t \in J,$$
(3.13)

$$\boldsymbol{L}(s,s;\gamma) = \mathbb{1} \qquad \qquad s \in J \qquad (3.14)$$

with  $\mathbb{1}$  being the identity (unit) matrix of corresponding size. From these equalities immediately follows that L is always non-degenerate.

**Proposition 3.3.** A linear mapping (3.1) is a linear transport along  $\gamma$  from s to t iff its matrix, defined via (3.9), satisfies (3.13) and (3.14).

*Proof.* The necessity was already proved. The sufficiency is trivial: a simple checking proves that (3.13) and (3.14) convert respectively (3.2) and (3.3) into identities.

**Proposition 3.4.** A non-degenerate matrix-valued function  $L: (t, s; \gamma) \mapsto L(t, s; \gamma)$ is a matrix of some linear transport along paths L (in a given field  $\{e_i\}$  of bases along  $\gamma$ ) iff

$$\boldsymbol{L}(t,s;\gamma) = \boldsymbol{F}^{-1}(t;\gamma)\boldsymbol{F}(s;\gamma)$$
(3.15)

where  $\mathbf{F}: (t; \gamma) \mapsto \mathbf{F}(t; \gamma)$  is a non-degenerate matrix-valued function.

*Proof.* This proposition is simply a matrix form of Proposition 3.1. If  $\{f_i\}$  is a basis in V and  $F(s; \gamma)e_i(s; \gamma) = F^j_i(s; \gamma)f_j$ , then (3.15) with  $F(s; \gamma) = [F^j_i(s; \gamma)]$  is equivalent to (3.6).

**Proposition 3.5.** If the matrix L of a linear transport L along paths has a representation

$$\boldsymbol{L}(t,s;\gamma) = {}^{*}\boldsymbol{F}^{-1}(t;\gamma) {}^{*}\boldsymbol{F}(s;\gamma)$$
(3.16)

for some matrix-valued function  ${}^{*}F(s;\gamma)$ , then all matrix-valued functions F representing L via (3.15) are given by

$$\boldsymbol{F}(s;\gamma) = \boldsymbol{D}^{-1}(\gamma) \,^{*} \boldsymbol{F}(s;\gamma) \tag{3.17}$$

where  $D(\gamma)$  is a non-degenerate matrix depending only on  $\gamma$ .

*Proof.* In fact, this propositions is a matrix variant of Proposition 3.2;  $D(\gamma)$  is simply the matrix of the mapping  $D(\gamma)$  in some bases.

If  $F(s; \gamma)$  and  $F'(s; \gamma)$  are two matrix-valued functions, representing the matrix of L via (3.15) in two bases  $\{e_i\}$  and  $\{e'_i\}$  respectively, then, as a consequence of (3.11), the relation

$$\mathbf{F}'(s;\gamma) = C(\gamma)\mathbf{F}(s;\gamma)A(s;\gamma) \tag{3.18}$$

holds for some non-degenerate matrix-valued function C of  $\gamma$ .

#### 3.3. Linear transports and derivations along paths

Below we want to consider some properties of the linear transports along paths connected with their 'differentiability'; in particular, we shall establish a bijective correspondence between them and the derivations along paths. For the purpose is required a smooth, of class at least  $C^1$ , transition from fibre to fibre when moving along a path in the base. Rigorously this is achieved by exploring transports in bundles whose bundle space is a  $C^1$  differentiable manifold which will be supposed from now on in the present chapter.

Let  $(E, \pi, B)$  be a vector bundle whose bundle space E is a  $C^r$ ,  $r \in \mathbb{N} \cup \{0, \infty, \omega\}$ , differentiable manifold. A linear transport  $L^{\gamma}$  along  $\gamma : J \to B$  is called differentiable of class  $C^k$ ,  $k = 0, 1, \ldots, r$ , or simply  $C^k$  transport, if for arbitrary  $s \in J$  and  $u \in \pi^{-1}(\gamma(s))$ , the path  $\overline{\gamma}_{s;u} : J \to E$  with  $\overline{\gamma}_{s;u}(t) := L_{s \to t}^{\gamma} u \in \pi^{-1}(\gamma(t)), t \in J$ , is a  $C^k$  mapping in the bundle space E. If a  $C^k$  linear transport has a representation (3.6), the mapping  $s \mapsto F(s; \gamma)$  is of class  $C^k$ . So, the transport  $L^{\gamma}$  is of class  $C^k$  iff  $L_{s \to t}^{\gamma}$  has  $C^k$  dependence on s and t simultaneously. If  $\{e_i(\cdot; \gamma)\}$  is a  $C^k$  frame along  $\gamma$ , i.e.,  $\{e_i(s; \gamma)\}$  is a basis in  $\pi^{-1}(\gamma(s))$  and the mapping  $s \mapsto e_i(s; \gamma)$  is of class  $C^k$  for all i, from (3.12) follows that  $L^{\gamma}$  is of class  $C^k$  iff its matrix  $\mathbf{L}(t, s; \gamma)$  has  $C^k$  dependence on s and t.

Let E be a  $C^1$  manifold and S a set of paths in  $B, S \subseteq P(B) = \{\gamma : J \to B\}$ . A transport L along paths in  $(E, \pi, B)$ , E being  $C^r$  manifold, is said to be of class  $C^k, k = 0, 1, \ldots, r$ , on S if the corresponding transport  $L^{\gamma}$  along  $\gamma$  is of class  $C^k$  for all  $\gamma \in S$ . A transport along paths may turn to be of class  $C^k$  on some set S of paths in B and not to be of class  $C^k$  on other set S' of paths in B. Below, through Section 6, the set S will not be specialized and written explicitly; correspondingly, we shall speak simply of  $C^k$  transports implicitly assuming that they are such on some set S to be the one of  $C^1$  paths in B. Further we consider only  $C^1$  linear transports along paths whose matrices will be referred to smooth frames along paths.

**Definition 3.2.** The derivation D along paths generated by a  $C^1$  linear transport L along paths in  $(E, \pi, B)$ , E being a  $C^1$  manifold, is a mapping of type (2.1a) such

#### 3. Linear transports in vector bundles

that for every path  $\gamma: J \to B$ , we have  $D^{\gamma}: \lambda \mapsto (D\lambda)_{\gamma}$  with  $D^{\gamma}\lambda: s \mapsto D_s^{\gamma}\lambda$ ,  $s \in J$ , where  $D_s^{\gamma}$  is a mapping (2.1b) given via

$$D_s^{\gamma}(\lambda) := \lim_{\varepsilon \to 0} \left\{ \frac{1}{\varepsilon} \left[ L_{s+\varepsilon \to s}^{\gamma} \lambda_{\gamma}(s+\varepsilon) - \lambda_{\gamma}(s) \right] \right\}$$
(3.19)

for every lifting  $\lambda \in \text{PLift}^1(E, \pi, B)$  with  $\lambda: \gamma \mapsto \lambda_{\gamma}$ . The mapping  $D^{\gamma}$  (resp.  $D_s^{\gamma}$ ) will be called a *derivation along*  $\gamma$  generated by L (resp. a *derivation along*  $\gamma$  at s assigned to L).

*Remark* 3.3. The operator  $D_s^{\gamma}$  is an analogue of the covariant derivative along paths assigned to a linear connection; cf., e.g., [31, p. 139, equation (12)].

Remark 3.4. Notice, if  $\gamma$  has self-intersections and  $x_0 \in \gamma(J)$  is such a point, the mapping  $x \mapsto \pi^{-1}(x), x \in \gamma(J)$ , given by  $x \mapsto \{D_s^{\gamma}(\lambda) | \gamma(s) = x, s \in J\}$  is, generally, multiple-valued at  $x_0$ .

Let L be a linear transport along paths in  $(E, \pi, B)$ . For every path  $\gamma: J \to B$ , choose some  $s_0 \in J$  and  $u_0 \in \pi^{-1}(\gamma(s_0))$ . The mapping

$$\overline{L}: \gamma \mapsto \overline{L}_{s_0, u_0}^{\gamma}, \quad \overline{L}_{s_0, u_0}^{\gamma}: J \to E, \quad \overline{L}_{s_0, u_0}^{\gamma}: t \mapsto \overline{L}_{s_0, u_0}^{\gamma}(t) := L_{s_0 \to t}^{\gamma} u_0$$
(3.20)

is, evidently, a lifting of paths.

**Definition 3.3.** The lifting of paths  $\overline{L}$  from B to E in  $(E, \pi, B)$ , defined via (3.20), is called *lifting (of paths) generated by the (linear) transport L*.

Equations (3.2) and (3.4), combined with (3.19), immediately imply

$$D_t^{\gamma}(\overline{L}) \equiv 0, \qquad t \in J, \tag{3.21}$$

$$D_s^{\gamma}(a\lambda + b\mu) = aD_s^{\gamma}\lambda + bD_s^{\gamma}\mu, \qquad a, b \in \mathbb{K}, \quad \lambda, \mu \in \mathrm{PLift}^1(E, \pi, B), \quad (3.22)$$

where  $s_0 \in J$  and  $u_0 \in \pi^{-1}(\gamma(s_0))$  are fixed. In other words, equation (3.21) means that the lifting  $\overline{L}$  is constant along every path  $\gamma$  with respect to D.

Let  $\{e_i(s;\gamma)\}$  be a smooth field of bases along  $\gamma: J \to B$ ,  $s \in J$ , i.e.,  $\{e_i\}$  to be a frame along  $\gamma$ . Combining (3.12) and (3.19), we find the explicit local action of  $D_s^{\gamma}$  as<sup>12</sup>

$$D_s^{\gamma} \lambda = \left[ \frac{\mathrm{d}\lambda_{\gamma}^i(s)}{\mathrm{d}s} + \Gamma_j^i(s;\gamma)\lambda_{\gamma}^j(s) \right] e_i(s;\gamma).$$
(3.23)

Here the (2-index) coefficients  $\Gamma^{i}_{i}$  of the linear transport L are defined by

$$\Gamma^{i}_{\ j}(s;\gamma) := \frac{\partial L^{i}_{\ j}(s,t;\gamma)}{\partial t} \bigg|_{t=s} = -\frac{\partial L^{i}_{\ j}(s,t;\gamma)}{\partial s} \bigg|_{t=s}$$
(3.24)

and, evidently, uniquely determine the derivation D generated by L.

<sup>&</sup>lt;sup>12</sup>The existence of derivatives like  $d\lambda_{\gamma}^{i}(s)/ds$ , viz. that  $\lambda_{\gamma}^{i}: J \to \mathbb{K}$  are  $C^{1}$  mappings, follows from  $\lambda \in \mathrm{PLift}^{1}(E, \pi, B)$ .

**Exercise 3.1.** Prove that the derivation along paths generated by a linear transport is actually a derivation along paths (see Definition 2.1). (Hint: use (3.22) and (3.23).)

Below, we shall prove that, freely speaking, a linear transport along path(s) can locally, in a given field of local bases, be described equivalently by the set of its local coefficients (with transformation law (3.26) written below).

If the transport's matrix L has a representation (3.15), from (3.24) we get

$$\boldsymbol{\Gamma}(s;\gamma) := \left[\Gamma^{i}_{\ j}(s;\gamma)\right] = \frac{\partial \boldsymbol{L}(s,t;\gamma)}{\partial t}\Big|_{t=s} = \boldsymbol{F}^{-1}(s;\gamma)\frac{\mathrm{d}\boldsymbol{F}(s;\gamma)}{\mathrm{d}s}.$$
(3.25)

From here, (3.11), and (3.14), we see that the change  $\{e_i\} \to \{e'_i = A_i^j e_i\}$  of the bases along  $\gamma$  with a non-degenerate  $C^1$  matrix-valued function  $A(s;\gamma) := [A_i^j(s;\gamma)]$  implies

$$\mathbf{\Gamma}(s;\gamma) = \left[ \Gamma^{i}{}_{j}(s;\gamma) \right] \mapsto \mathbf{\Gamma}'(s;\gamma) = \left[ {\Gamma'}^{i}{}_{j}(s;\gamma) \right]$$

with

$$\Gamma'(s;\gamma) = A^{-1}(s;\gamma)\Gamma(s;\gamma)A(s;\gamma) + A^{-1}(s;\gamma)\frac{\mathrm{d}A(s;\gamma)}{\mathrm{d}s}.$$
(3.26)

**Proposition 3.6.** Let along every (resp. given) path  $\gamma: J \to B$  be given a geometrical object  $\Gamma$  whose local components  $\Gamma^i{}_j$  in a frame  $\{e_i\}$  along  $\gamma$  change according to (3.26) with  $\Gamma(s;\gamma) = [\Gamma^i{}_j(s;\gamma)]$ . There exists a unique linear transport L along paths (resp. along  $\gamma$ ) the matrix of whose coefficients is exactly  $\Gamma(s;\gamma)$  in  $\{e_i\}$  along  $\gamma$ . Moreover, the matrix of the components of L in  $\{e_i\}$  is

$$\boldsymbol{L}(t,s;\gamma) = Y(t,s_0;-\boldsymbol{\Gamma}(\cdot;\gamma))Y^{-1}(s,s_0;-\boldsymbol{\Gamma}(\cdot;\gamma)), \qquad s,t \in J$$
(3.27)

where  $s_0 \in J$  is arbitrarily fixed and the matrix  $Y(s, s_0; Z)$ , for a  $C^0$  matrix-valued function  $Z: s \mapsto Z(s)$ , is the unique solution of the initial-valued problem

$$\frac{\mathrm{d}Y}{\mathrm{d}s} = Z(s)Y, \qquad Y = Y(s, s_0; Z), \quad s \in J, \tag{3.28a}$$

$$Y(s_0, s_0; Z) = \mathbb{1}.$$
 (3.28b)

*Proof.* At the beginning, we note that the existence and uniqueness of the solution of (3.28) is ensured by Lemma II.3.2 on page 96.

Given a linear transport L with a matrix (3.15). Suppose its components are exactly  $\Gamma_{j}^{i}(s;\gamma)$  in a frame  $\{e_{i}\}$ . Solving (3.25) with respect to  $d\mathbf{F}^{-1}/ds$ , we obtain  $d\mathbf{F}^{-1}(s;\gamma)/ds = -\Gamma(s;\gamma)\mathbf{F}^{-1}(s;\gamma)$  and, consequently,  $\mathbf{F}^{-1}(s;\gamma) =$  $Y(s, s_{0}; -\Gamma(\cdot;\gamma))\mathbf{F}^{-1}(s_{0};\gamma)$ . So, due to equation (3.15), the matrix of L is (3.27). Because of [34, Chapter IV, equation (1.10)], the expression

$$Y(t,s;Z) = Y(t,s_0;Z)Y(s_0,s;Z) = Y(t,s_0;Z)Y^{-1}(s,s_0;Z)$$

is independent of  $s_0$ . Besides, as a consequence of (3.26), the matrix (3.27) transforms according to (3.11) when the local bases are changed. Hence equation (3.3) holds and, due to (3.12), the linear mapping L with a matrix (3.27) in  $\{e_i\}$  is a linear transport along  $\gamma$ . In this way we have proved two things: On one hand, a linear mapping with a matrix (3.27) in  $\{e_i\}$  is a linear transport with local coefficients  $\Gamma^i_{\ i}(s;\gamma)$  in  $\{e_i\}$  along  $\gamma$  and, on the other hand, any linear transport with local coefficients  $\Gamma^i_{\ i}(s;\gamma)$  in  $\{e_i\}$  has a matrix (3.27) in  $\{e_i\}$ .

Now we are ready to prove a fundamental result: there exists a bijective mapping between the sets of  $C^1$ -linear transports along paths and derivations along paths. The explicit correspondence between linear transports along paths and derivations along paths is through the equality of their local coefficients and components, respectively, in a given field of bases. After the prove of this result, we shall illustrate it in a case of linear connections on a manifold.

**Proposition 3.7.** A mapping (2.1a) (resp. (2.1c)) is a derivation along paths (resp. along  $\gamma$ ) iff there exists a unique linear transport along paths (resp. along  $\gamma$ ) generating it via (3.19). Besides, the components and coefficients of corresponding in this way, respectively, derivation and linear transport along paths are equal.

*Proof.* Let  $\{e_i(s;\gamma)\}$  be a frame along  $\gamma$  and D (resp.  $D^{\gamma}$ ) be a derivation along paths (resp. along  $\gamma$ ). Define the components  $\Gamma^i_{\ j}(s;\gamma)$  of  $D^{\gamma}$  in  $\{e_i\}$  by the expansion (2.3) in which we consider  $\hat{e}_i : \gamma \mapsto e_i(\cdot;\gamma)$  as a lifting of paths. They uniquely define  $D^{\gamma}$  as (2.2) implies (3.23) (see the identical equality (2.4)). Besides, it is trivial to verify the transformation law (3.26) for them. So, by Proposition 3.6, there is a unique linear transport along paths (resp. along  $\gamma$ ) with the same local coefficients.

Conversely, as we already proved, to any linear transport L along paths (resp. along  $\gamma$ ) there corresponds a derivation  $D^{\gamma}$  along  $\gamma$  given via (3.19) whose components coincide with the coefficients of  $L^{\gamma}$  and transform according to (3.26).

**Example 3.1.** Let  $\nabla$  be a linear connection (see Definition I.3.1) on a  $C^1$  differentiable manifold M and  $\Gamma^i_{jk}(p)$ ,  $i, j, k = 1, \ldots, \dim M, p \in M$ , be its local coefficients in a field  $\{E_i(p)\}$  of bases in the tangent bundle over M, i.e.,  $\nabla_{E_i}E_j = \Gamma^k_{ji}E_k$ . If  $\gamma$  is a  $C^1$  path in M, the covariant derivative  $\nabla_{\dot{\gamma}} = \dot{\gamma}^i \nabla_{E_i}$  along  $\gamma$  (see Section III.10, in particular equations (III.10.4) and (III.10.5)),  $\dot{\gamma}$  being the vector field tangent to  $\gamma$ , is a derivation along  $\gamma$  in the tensor bundles over M and its local components are

$$\Gamma^{i}{}_{j}(s;\gamma) = \Gamma^{i}{}_{jk}(\gamma(s))\dot{\gamma}^{k}(s).$$
(3.29)

**Exercise 3.2.** Verify that the unique linear transport along paths, corresponding, in accordance with Proposition 3.7, to the derivation with local components given by (3.29), along  $\gamma: J \to M$  from s to t, with  $s, t \in J$  and  $s \leq t$ , is exactly the

parallel transport along  $\gamma|_{[s,t]}$  generated via the initial connection  $\nabla$ . (See Definition I.3.2 and, for some details, Section 11.)

**Exercise 3.3.** Prove that, given on the tangent bundle over M a derivation D along paths with local components (3.29), then  $\Gamma^i_{jk}$  are the coefficients of a linear connection  $\nabla$ , which, when restricted to the tangent bundle, is such that  $(\nabla_V U)|_p = D^{\gamma}_s(U \circ \gamma)$ , where  $\gamma: J \to M$  is any  $C^1$  path with  $\dot{\gamma}(s) = V_p$ ,  $\gamma(s) = p \in M$  for some  $s \in J$ , and U, V are vector fields on M.

**Example 3.2.** Consider a concrete kind of a linear transports L in the trivial line bundle  $(B \times \mathbb{R}, \operatorname{pr}_1, B)$ , where B is a topological space, which in particular can be a  $C^0$  manifold,  $\times$  is the Cartesian product sign, and  $\operatorname{pr}_1 \colon B \times \mathbb{R} \to B$  is the projection on B. An element of  $B \times \mathbb{R}$  is of the form u = (b, y) for some  $b \in B$  and  $y \in \mathbb{R}$  and the fibre over  $c \in B$  is  $\operatorname{pr}_1^{-1}(c) = \{c\} \times \mathbb{R} = \{(c, z) : z \in \mathbb{R}\}$ ; the linear structure of  $\operatorname{pr}_1^{-1}(c)$  is given by  $\lambda_1(c, z_1) + \lambda_2(c, z_2) = (c, \lambda_1 z_1 + \lambda_2 z_2)$  for  $\lambda_1, \lambda_2, z_1, z_2 \in \mathbb{R}$ . The bundle  $(B \times \mathbb{R}, \operatorname{pr}_1, B)$  admits a global frame field  $\{e_1\}$  consisting of a single section  $e_1 \in \operatorname{Sec}(B \times \mathbb{R}, \operatorname{pr}_1, B)$  such that  $e_1 \colon B \ni b \mapsto e_1(b) = (b, 1) \in \operatorname{pr}_1^{-1}(b)$ . For  $\gamma \colon J \to B$  and  $s, t \in J$ , define  $L \colon \gamma \mapsto L^{\gamma} \colon (s, t) \mapsto L_{s \to t}^{\gamma} \colon \operatorname{pr}_1^{-1}(\gamma(s)) \to$  $\operatorname{pr}_1^{-1}(\gamma(t))$  by

$$L_{s \to t}^{\gamma}(u) = \left(\gamma(t), \frac{f(\gamma(s))}{f(\gamma(t))}y\right) \quad \text{for } u = (\gamma(s), y) \in \mathrm{pr}_{1}^{-1}(\gamma(s)), \quad (3.30)$$

where  $f: \gamma(J) \to \mathbb{R} \setminus \{0\}$  is a non-vanishing function on  $\gamma(J)$ . The verification of (3.2)-(3.4) is trivial and hence L is a linear transport along paths. Its matrix in the frame  $\{e_1\}$  is  $\mathbf{L}(t, s; \gamma) = L_1^1(t, s; \gamma) = \frac{f(\gamma(s))}{f(\gamma(t))}$ , in conformity with (3.15). If  $f \circ \gamma: J \to \mathbb{R} \setminus \{0\}$  is of class  $C^1$ , the single coefficient of L is (see (3.24))  $\Gamma_1^1(s; \gamma) = \frac{d}{ds} \ln(f(\gamma(s)))$ ; however, this coefficient is a useful quantity if  $B \times \mathbb{R}$  (and hence B) is a  $C^1$  manifold – see (3.23). Going some pages ahead (see Proposition 5.2 and Definition 4.4 below), we see that the transport L satisfies equation (5.2) below and therefore admits normal frames; in particular the frame  $\{f_1\}$  such that (see (5.3) below)

$$f_1|_{\gamma(s)} = L_{s_0 \to s}^{\gamma} \left( e_1|_{\gamma(s_0)} \right) = \left( \gamma(s), \frac{f(\gamma(s_0))}{f(\gamma(s))} \right)$$

for a fixed  $s_0 \in J$  and any  $s \in J$  is normal along  $\gamma$ , i.e., the matrix of L in  $\{f_1\}$  is the identity matrix (the number one in the particular case).

## 4. Normal frames for linear transports

The parallel transport in a Euclidean space  $\mathbb{E}^n$  (or in  $\mathbb{R}^n$ ) has the property that, in Cartesian coordinates, it preserves the components of the vectors that are transported, changing only their initial points [3]. This evident observation, which can

be taken even as a definition for parallel transport in  $\mathbb{E}^n$ , is of fundamental importance when one tries to generalize the situation.

Let L be a linear transport along paths in a vector bundle  $(E, \pi, B), U \subseteq B$ be an arbitrary subset in B, and  $\gamma: J \to U$  be a path in U.

**Definition 4.1.** A frame (field of bases) in  $\pi^{-1}(\gamma(J))$  is called normal along  $\gamma$  for L if the matrix of L in it is the identity matrix along the given path  $\gamma$ .

**Definition 4.2.** A frame field (of bases) defined on U is called normal on U for L if it is normal along every path  $\gamma: J \to U$  in U. The frame is called normal for L if U = B.

Notice that 'normal' refers to a 'normal form' as opposed to orthogonal to tangential.

In the context of the present book, we pose the following problem. Given a linear transport along paths, is it possible to find a local basis or a field of bases (frame) in which its matrix is the identity one? Below we shall rigorously formulate and investigate this problem. If frames with this property exist, we call them *normal* (for the transport given). According to (3.12), the linear transports do not change vectors' components in such a frame and, conversely, a frame with the last property is normal. Hence the normal frames are a straightforward generalization of the Cartesian coordinates in Euclidean space.<sup>1</sup> Because of this and following the established terminology with respect to metrics [11, 12], we call *Euclidean* a linear transport admitting normal frame(s).

Since a frame field on an arbitrary set U is actually a basis in the set  $\operatorname{Sec}((E, \pi, B)|_U) = \operatorname{Sec}(\pi^{-1}(U), \pi|_U, U)$ , we call such a basis *normal* if the corresponding field of bases is normal on U.

Remark 4.1. It should be emphasized that a frame normal on U must be defined on U but outside U (if  $U \neq M$ ), i.e., at points in  $M \setminus U$ , it may not be defined. In this aspect the frames normal for linear transports differ from the ones for derivations along vector fields, in particular for linear connections, in which case they must be defined on an open subset of M containing or equal to U. This difference comes from the fact that in the definitions of the normal frames in the latter case are involved implicitly derivations with respect to some frames or local coordinates which derivations are well defined only on neighborhoods, while in the former case appear only derivations along paths.

**Definition 4.3.** A linear transport along paths (or along a path  $\gamma$ ) is called Euclidean along some (or the given) path  $\gamma$  if it admits a frame normal along  $\gamma$ .

**Definition 4.4.** A linear transport along paths is called Euclidean on U if it admits frame(s) normal on U. It is called Euclidean if U = B.

<sup>&</sup>lt;sup>1</sup>According to the argument presented, it is more natural to call Cartesian the special kind of local bases (or frames) we are talking about. But, in our opinion and for historical reasons, it is better to use the already established terminology for linear connections and derivations of the tensor algebra over a differentiable manifold (see below and [97, appendix A] or [83]).

We want to note that the name "Euclidean transport" is connected with the fact that if we put  $B = \mathbb{R}^n$  and  $\pi^{-1}(p) = T_p(\mathbb{R}^n)$  (the tangent space to  $\mathbb{R}^n$  at p) and identify  $T_p(\mathbb{R}^n)$  with  $\mathbb{R}^n$ , then in an orthonormal frame, i.e., in Cartesian coordinates, the Euclidean transport coincides with the standard parallel transport in  $\mathbb{R}^n$  (leaving the vectors' components unchanged).

**Example 4.1.** Euclidean transports exist always in a case of a trivial bundle  $(B \times V, \operatorname{pr}_1, B)$ , with V being a vector space and  $\operatorname{pr}_1 \colon B \times V \to B$  being the projection on B; cf. Example 3.2. For instance, the mapping  $L_{s \to t}^{\gamma}(\gamma(s), v) = (\gamma(t), v)$ , for  $v \in V$ , defines a Euclidean transport which is similar to the parallel one in  $\mathbb{R}^n$ . Indeed, if  $\{f_i : i = 1, \ldots, \dim V\}$  is a basis of V and  $v = v^i f_i$ , then  $e_i \colon p \mapsto e_i|_p \coloneqq (p, f_i)$ ,  $p \in B$ , is a (global) frame on B if we put  $v^i e_i|_p = (p, v^i f_i) = (p, v)$  and therefore  $L_{s \to t}^{\gamma}(e_i|_{\gamma(s)}) = e_i|_{\gamma(t)}$ , which means that  $L \colon \gamma \mapsto L^{\gamma} \colon (s, t) \mapsto L_{s \to t}^{\gamma}$  is a Euclidean transport and  $\{e_i\}$  is a normal frame for it (see Corollary 4.1 below).

Below we present some general results concerning normal frames leaving the problem of their existence for the next section.

The importance of normal frames is established by the following result.

**Proposition 4.1.** The following statements are equivalent in a given frame  $\{e_i\}$  over  $U \subseteq B$ :

 (i) The matrix of L is the identity (unit) matrix on U, i.e., along every path γ in U:

$$\boldsymbol{L}(t,s;\gamma) = \boldsymbol{\mathbb{1}}.\tag{4.1a}$$

(ii) The matrix of L along every γ: J → U depends only on γ, i.e., it is independent of the points at which it is calculated:

$$\boldsymbol{L}(t,s;\gamma) = C(\gamma) \tag{4.1b}$$

where C is a matrix-valued function of  $\gamma$ .

 (iii) If E is a C<sup>1</sup> manifold, the coefficients Γ<sup>i</sup><sub>j</sub>(s; γ) of L vanish on U, i.e., along every path γ in U

$$\Gamma(s;\gamma) = 0. \tag{4.1c}$$

(iv) The explicit local action of the derivation D along paths generated by L reduces on U to differentiation of the components of the liftings with respect to the path's parameter if the path lies entirely in U:

$$D_s^{\gamma} \lambda = \frac{\mathrm{d}\lambda_{\gamma}^i(s)}{\mathrm{d}s} e_i(s;\gamma) \tag{4.1d}$$

where  $\lambda = \lambda^i e_i \in \text{PLift}^1((E, \pi, B)|_U)$ , with E being a  $C^1$  manifold, and  $\lambda \colon \gamma \mapsto \lambda_{\gamma}$ .

#### 4. Normal frames for linear transports

(v) The transport L leaves the vectors' components unchanged along any path in U:

$$L_{s \to t}^{\gamma} \left( u^{i} e_{i}(s;\gamma) \right) = u^{i} e_{i}(t;\gamma)$$
(4.1e)

where  $u^i \in \mathbb{K}$ .

(vi) The basic vector fields are L-transported along any path  $\gamma: J \to U$ :

$$L_{s \to t}^{\gamma} \left( e_i(s;\gamma) \right) = e_i(t;\gamma). \tag{4.1f}$$

*Proof.* We have to prove the equivalences

$$\begin{split} \boldsymbol{L}(t,s;\gamma) &= C(\gamma) \iff \boldsymbol{L}(t,s;\gamma) = \mathbb{1} \iff \boldsymbol{\Gamma}(s;\gamma) = 0\\ \iff D_s^{\gamma}\lambda &= \frac{\mathrm{d}\lambda_{\gamma}^i(s)}{\mathrm{d}s}e_i(s;\gamma) \iff L_{s\to t}^{\gamma}\left(u^i e_i(s;\gamma)\right) = u^i e_i(t;\gamma)\\ \iff L_{s\to t}^{\gamma}\left(e_i(s;\gamma)\right) = e_i(t;\gamma). \end{split}$$

$$(4.2)$$

If  $\mathbf{L}(t,s;\gamma) = C(\gamma)$ , then, using the representation (3.15), we get  $\mathbf{F}(t;\gamma) = \mathbf{F}(s;\gamma)C(\gamma) = \mathbf{F}(s_0;\gamma)$  for some fixed  $s_0 \in J$  as the points s and t are arbitrary, so  $\mathbf{L}(t,s_0;\gamma) = \mathbf{F}^{-1}(s_0;\gamma)\mathbf{F}(s_0;\gamma) = \mathbf{1}$ . The inverse implication is trivial. The second equivalence is a consequence of (3.25) and (3.15) since  $\mathbf{\Gamma} = 0$  implies  $\mathbf{F}(s;\gamma) = \mathbf{F}(\gamma)$ , while the third one is a corollary of (3.23). The validity of the last but one equivalence is a consequence of  $\mathbf{L}(t,s;\gamma) = \mathbf{1} \iff L_{s\to t}^{\gamma}(u^i e_i(s;\gamma)) = u^i e_i(t;\gamma)$  which follows from (3.12). The last equivalence is a corollary of the linearity of L and the arbitrariness of  $u^i$ .

*Remark* 4.2. An evident corollary of the last proof is

$$\boldsymbol{L}(t,s;\gamma) = \mathbb{1} \iff \boldsymbol{F}(s;\gamma) = \boldsymbol{B}(\gamma) \tag{4.3}$$

with B being a matrix-valued function of the path  $\gamma$  only. According to Proposition 3.5, this dependence is inessential and, consequently, in a normal frame, we can always choose representation (3.15) with

$$\boldsymbol{F}(s;\gamma) = \mathbb{1}.\tag{4.4}$$

**Corollary 4.1.** Any one of the equalities (4.1a)–(4.1f) express a necessary and sufficient condition for a frame to be normal for L in U.

*Proof.* This result is a direct consequence of Definition 4.2 and Proposition 4.1.  $\Box$ 

**Proposition 4.2.** The equations (4.1a)–(4.1f) are equivalent in a given frame  $\{e_i\}$  along a (fixed) path  $\gamma: J \to B$ .

*Proof.* This proof is identical with the one of Proposition 4.1 for  $U = \gamma(J)$ .

**Corollary 4.2.** A frame is normal along  $\gamma$  for L if and only if in that frame one (and hence all) of the equalities (4.1a)–(4.1f) is (are) valid.

*Proof.* The result follows from Definition 4.1 and Proposition 4.2.

In particular, a frame is normal for L along  $\gamma$  iff in the frame the coefficients of L vanish along  $\gamma$ , i.e., iff (4.1c) holds.

A lifting of paths  $\lambda \in \text{PLift}(E, \pi, B)$  is called *L*-transported along  $\gamma \colon J \to B$ , if for every  $s, t \in J$  is fulfilled

$$\lambda_{\gamma}(t) = L_{s \to t}^{\gamma} \lambda_{\gamma}(s)$$

with  $\lambda: \gamma \mapsto \lambda_{\gamma}$ . Hence a frame  $\{e_i(s,\gamma)\}$  along  $\gamma$  is *L*-transported along  $\gamma$  if the liftings  $\hat{e}_1, \ldots, \hat{e}_{\dim B}$ , defined via (2.8), are *L*-transported along  $\gamma$ .

Therefore a frame is normal for L along  $\gamma$  iff it is L-transported along  $\gamma$ , i.e., if, by definition, its basic vectors  $e_i(s; \gamma)$  satisfy (4.1f). As we shall see below (see Proposition 4.4), this allows a convenient and useful way for constructing normal frames, if any.

For the above reasons, sometimes, it is convenient for the Definition 4.1 to be replaced, equivalently, by the next ones.

**Definition 4.1'.** If E is a  $C^1$  manifold, a *frame (or frame field)* over  $\gamma(J)$  is called *normal along*  $\gamma: J \to B$  for a linear transport L along paths if the coefficients of L along  $\gamma$  vanish in that frame.

**Definition 4.1".** A frame over  $\gamma(J)$  is called normal along  $\gamma: J \to B$  for a linear transport L along paths if it is L-transported along  $\gamma$ .

The last definition of a normal frame is, in a sense, the 'most invariant (basis-free)' one.

The next proposition describes the class of normal frames, if any, along a given path.

**Proposition 4.3.** All frames normal for some linear transport along paths which is Euclidean along a certain (fixed) path are connected by linear transformations whose matrices may depend only on the given path but not on the point at which the bases are defined.

*Proof.* Let  $\{e_i\}$  and  $\{e'_i := A_i^j e_j\}$  be frames normal along  $\gamma: J \to B$  for a linear transport L along paths and L and L' be the matrices of L in them respectively. As, by definition L = L' = 1, from (3.11), we get  $A(s; \gamma) = A(t; \gamma)$  for any  $s, t \in J$ , i.e.,  $A(s; \gamma)$  depends only on  $\gamma$  and not on s.

If E is a  $C^1$  manifold and  $\Gamma$  and  $\Gamma'$  are the matrices of the coefficients of L in  $\{e_i\}$  and  $\{e'_i\}$ , respectively, by Proposition 4.1 we have  $\Gamma = \Gamma' = 0$ , so the transformation law (3.26) implies  $dA(s;\gamma)/ds = 0$ ,  $A(s;\gamma) := [A^j_i(s;\gamma)]$ .

**Corollary 4.3.** All frames normal for a Euclidean transport along a given path are obtained from one of them via linear transformations whose matrices may depend only on the path given but not on the point at which the bases are defined.

*Proof.* See Proposition 4.3 or its proof.

The following two results describe the class of all frames normal on an arbitrary set U, if such frames exist.

**Corollary 4.4.** If a linear transport along paths admits frames normal on a set U, then all of them are connected via linear transformations with constant (on U) matrices.

*Proof.* Let  $\{e_i\}$  and  $\{e'_i := A^j_i e_i\}$  be frames normal on U and  $p \in U$ . By Proposition 4.3 (see also Definition 4.2) for any paths  $\beta$  and  $\gamma$  in U passing though p, we have  $A(p) := [A^j_i] = \mathbf{B}(\beta) = \mathbf{B}(\gamma)$  for some matrix-valued function  $\mathbf{B}$  on the set P(U) of the paths in U. Hence A(x) = const on U, due to the arbitrariness of  $\beta$  and  $\gamma$ .

**Corollary 4.5.** If a linear transport along paths admits a frame normal on a set U, then all such frames on U for it are obtained from that frame by linear transformations with constant (on U) coefficients.

*Proof.* The result immediately follows from Corollary 4.4  $\Box$ 

We end this section with a simple but important result which shows how the normal frames, if any, can be constructed along a given path.

**Proposition 4.4.** If L is Euclidean transport along  $\gamma: J \to B$  and  $\{e_i^0\}$  is a basis in  $\pi^{-1}(\gamma(s_0))$  for some  $s_0 \in J$ , then the frame  $\{e_i\}$  along  $\gamma$  defined by

$$e_i(s;\gamma) = L^{\gamma}_{s_0 \to s} \left( e^0_i \right), \qquad s \in J \tag{4.5}$$

is normal for L along  $\gamma$ .

*Proof.* Due to (3.2) and (4.5), the frame  $\{e_i\}$  satisfies (4.1f) along  $\gamma$ . Hence, by Corollary 4.2, it is normal for L along  $\gamma$ .

An analogous result on a set  $U \subseteq B$  will be presented in the next section (see Proposition 5.5 below).

## 5. On the existence of normal frames

In the previous section there were derived a number of properties of the normal frames, but the problem of their existence was neglected. This is the subject of the present section.

Following the ideas of the previous chapters, one may attack the problem of existence of frames normal for a linear transport L along paths as follows. Suppose  $\{e_i\}$  is a frame on  $U \subseteq B$  (resp. along  $\gamma$ ) and  $\Gamma(s; \gamma)$  is the matrix of the coefficients of L along  $\gamma: J \to U$  (resp. along the given path  $\gamma$ ). Then L admits a frame  $\{e'_i\}$  normal on U (resp. along  $\gamma$ ) iff there is a matrix-valued function A :=

 $[A_j^i]$ :  $(s, \gamma) \mapsto A(s; \gamma) = A(\gamma(s))$  (resp. A:  $(s, \gamma) \mapsto A(s; \gamma)$ ) transforming  $\{e_i\}$  into  $\{e_i'\}, e_i'(s; \gamma) = A_i^j(s; \gamma)e_j(s; \gamma)$ , and satisfying, by virtue of (4.1c) and (3.26), the equation

$$\frac{\mathrm{d}A(s;\gamma)}{\mathrm{d}s} + \mathbf{\Gamma}(s;\gamma)A(s;\gamma) = 0 \tag{5.1}$$

for every  $\gamma: J \to U$  and  $s \in J$  (resp. for the given path  $\gamma: J \to B$  and  $s \in J$ ). This assertion is also an obvious corollary of the transformation law (3.26) and Definition 4.1' (or Proposition 4.1, points (i) and (iii)). Therefore all problems concerning the properties and existence of frames normal on U (resp. along  $\gamma$ ) can be reduced to the investigation of the equation (5.1) with  $A(s;\gamma) = A(\gamma(s))$ ,  $\gamma: J \to U$  (resp. with A depending, generally, separately on s and  $\gamma$ ); for this, we call (5.1) the normal frame equation along  $\gamma$  for L. We shall comment on this approach in Subsection 13.4. Below we shall follow other, more direct, methods which are primary related to the ('global') general properties of the linear transports along paths.

At a given point  $p \in B$  the following result is valid.

**Proposition 5.1.** A linear transport  $L^{\gamma}$  along  $\gamma: J \to B$  such that  $\gamma(J) = \{p\}$  for a given point  $p \in B$  admits normal frame(s) iff it is the identity mapping of the fibre over p, i.e.,  $L_{s \to t}^{\gamma} = \operatorname{id}_{\pi^{-1}(p)}$  for every  $s, t \in J$ .

*Proof.* The sufficiency is trivial (see Definition 3.1). If  $\{e_i\}$  is normal for  $L^{\gamma}$  (at p), then  $L_{s \to t}^{\gamma}(u^i e_i|_p) = u^i L_{s \to t}^{\gamma} e_i|_p = u^i e_i|_p$ ,  $u^i \in \mathbb{K}$ , due to  $\gamma(s) = \gamma(t) = p$  and Proposition 4.1, point (iv). Therefore  $L_{s \to t}^{\gamma} = \operatorname{id}_{\pi^{-1}(p)}$ .

Thus, for a degenerate path  $\gamma: J \to \{p\} \subset B$  for some  $p \in B$ , the identity mapping of the fibre over p is the only realization of a Euclidean transport along paths. Evidently, for such a transport every basis of that fibre is a frame normal at p for it.

**Proposition 5.2.** A linear transport L along paths admits frame(s) normal along a given path  $\gamma: J \to B$  iff

 $L_{s \to t}^{\gamma} = \mathsf{id}_{\pi^{-1}(\gamma(s))} \qquad \text{for every } s, t \in J \text{ such that } \gamma(s) = \gamma(t), \tag{5.2}$ 

i.e., if  $\gamma$  contains loops, the L-transport along each of them reduces to the identity mapping of the fibre over the initial/final point of the transportation.

Remark 5.1. For s = t the equation (5.2) is identically satisfied due to (3.3). But for  $s \neq t$ , if such s and t exist, this is highly non-trivial restriction: it means that the result of L-transportation along  $\gamma$  of a vector  $u \in \pi^{-1}(x_0)$  for some  $x_0 \in \gamma(J)$  from  $x_0$  to a point  $x \in \gamma(J)$  is independent of how long the vector has 'travelled' along  $\gamma$  or, more precisely, if  $x_0, x \in \gamma(J)$  are fixed and, for each  $y \in \gamma(J), J_y := \{r \in J : \gamma(r) = y\}$ , then the vector  $L^{\gamma}_{s_0 \to s}(u)$  is independent of the choice of the points  $s_0 \in J_{x_0}$  and  $s \in J_x$  (if some of the sets  $J_{x_0}$  and/or  $J_x$  contain more than one point). This is trivial if  $\gamma$  is without self-intersections (see (3.2)). If  $\gamma$  has self-intersections, e.g., if  $\gamma$  intersects itself one time at  $\gamma(s)$ , i.e., if  $\gamma(s) = \gamma(t)$  for some  $s, t \in J$ ,  $s \neq t$ , then the result of *L*-transportation of  $u \in \pi^{-1}(\gamma(s_0))$  from  $p_0 = \gamma(s_0)$  to  $p = \gamma(s) = \gamma(t)$  along  $\gamma$  is  $u_s = L_{s_0 \to s}^{\gamma} u$ or  $u_t = L_{s_0 \to t}^{\gamma} u$ . We have  $u_s = u_t$  iff (5.2) holds. Rewording, if we fix some  $u_0 \in \pi^{-1}(\gamma(s_0))$ , the bundle-valued function  $u: \gamma(J) \to E$  given by  $u: \gamma(s) \to$  $u_s = L_{s_0 \to s}^{\gamma} u_0 \in \pi^{-1}(\gamma(s))$  for  $s \in J$  is single-valued iff (5.2) is valid.<sup>1</sup> Notice, since  $\pi \circ u_s \equiv \gamma(s)$  (see (3.1)), the mapping u is (a single-valued) lifting of  $\gamma$  in Ethrough  $u_0$  irrespectively of the validity of (5.2).

Prima facie the above may be reformulated in terms of the concept of holonomy in vector bundles [23, pp. 51–54]. But a rigorous analysis reveals that this is impossible in the general case without imposing further restrictions, like equation (5.6) below, on the transports involved. For instance, without requiring equation (5.6) below to be valid, one cannot introduce the concept of a holonomy group.

*Proof.* If L is Euclidean along  $\gamma$ , then (5.2) follows from equation (4.1e) as it holds for every  $u^i \in \mathbb{K}$  in some normal frame  $\{e_i\}$ . Conversely, let (5.2) be valid. Put

$$e_i|_{\gamma(s)} := L^{\gamma}_{s_0 \to s} \left( e^0_i \right) \tag{5.3}$$

where  $\{e_i^0\}$  is a fixed basis in  $\pi^{-1}(\gamma(s_0))$  for a fixed  $s_0 \in J$ . Due to the nondegeneracy of L,  $\{e_i\}$  is a basis at  $\gamma(s)$  for every s. According to (5.2), the so-defined field of bases  $\{e_i\}$  along  $\gamma$  is single-valued. By means of (3.2), we easily verify that (4.1f) holds for  $\{e_i\}$ . Hence  $\{e_i\}$  is normal for L along  $\gamma$ .

Remark 5.2. Regardless of the validity of (5.2), equation (5.3) defines a field of, generally multiple-valued, normal frames in the set of sections along  $\gamma$  of  $(E, \pi, B)$ . Such a multi-valued property can be avoid if  $\gamma$  is supposed to be injective ( $\Leftrightarrow$  without self-intersections). Prima facie one may think that this solves the multi-valued problem in the general case by decomposing  $\gamma$  into a union of injective paths. However, this is not the most general situation because a transport along a composition of paths does not generally equal to the composition of the transports along its constituent sub-paths (see equation (5.6) below); besides, since equation (5.10) below does not hold generally, the absents of a natural/canonical definition of composition (product) of paths introduces an additional indefiniteness.

**Corollary 5.1.** Every linear transport along paths is Euclidean along every fixed path without self-intersections.

*Proof.* For a path  $\gamma: J \to B$  without self-intersections, the equality  $\gamma(s) = \gamma(t)$ ,  $s, t \in J$  is equivalent to s = t. So, according to (3.3), the condition (5.2) is identically satisfied.

Now we shall establish an important necessary and sufficient condition for the existence of frames normal on an arbitrary subset  $U \subseteq B$ .

<sup>&</sup>lt;sup>1</sup>The so-defined mapping u is a multiple-valued section along  $\gamma$  (see Subsection 2.3).
**Theorem 5.1.** A linear transport along paths admits frames normal on some set (resp. along a given path) if and only if its action along every path in this set (resp. along the given path) depends only on the initial and final points of the transportation but not on the particular path connecting these points. In other words, a transport is Euclidean on  $U \subseteq B$  iff it is path-independent on U.

Proof. Let a linear transport L admit a frame  $\{e_i\}$  normal on  $U \subseteq B$ . By (3.12) and Definitions 4.1 and 4.2, this implies  $L_{s \to t}^{\gamma} u^i(\gamma(s))(e_i|_{\gamma(s)}) = u^i(\gamma(s))e_i|_{\gamma(t)}$ for  $\gamma: J \to U$  and  $u(p) \in \pi^{-1}(p)$ ,  $p \in B$ . Conversely, let  $L_{s \to t}^{\gamma} u(\gamma(s))$  depend only on  $\gamma(s)$  and  $\gamma(t)$  but not on  $\gamma$  and  $\{e_i\}$  be a field of bases on U (resp. on  $\gamma(J)$ ). Then, due to (3.12), the matrix L of L in  $\{e_i\}$  has the form  $L(t,s;\gamma) =$  $B(\gamma(t), \gamma(s))$  for some matrix-valued function B on  $U \times U$ . Combining this result with Propositions 3.4 and 3.5, we see that L admits a representation

$$\boldsymbol{L}(t,s;\gamma) = \boldsymbol{F}_0^{-1}(\gamma(t))\boldsymbol{F}_0(\gamma(s)), \qquad s,t \in J$$
(5.4)

for some non-degenerate matrix-valued function  $F_0$  on the set U. Putting  $e'_i|_p = (F_0^{-1}(p))^j_{i}e_j|_p, p \in U$ , we obtain from (3.11) that the matrix of L in  $\{e'_i\}$  is  $L'(t,s;\gamma) = 1$ , i.e., the frame  $\{e'_i\}$  is normal for L on U.

An evident corollary of Theorem 5.1 is the following assertion. Let a linear transport L be Euclidean on  $U \subseteq B$  and  $h_a: J \to U$ ,  $a \in [0, 1]$ , be a homotopy of paths passing through two fixed points  $p, q \in U$ , i.e.,  $h_a(s_0) = p$  and  $h_a(t_0) = q$  for some  $s_0, t_0 \in J$  and any  $a \in [0, 1]$ . Then  $L_{s_0 \to t_0}^{h_a}$  is independent of  $a \in [0, 1]$ . In particular, we have  $L_{s_0 \to t_0}^{h_a} = id_{\pi^{-1}(p)}$  owing to Proposition 5.2.

Equation (5.4) and the part of the proof of Theorem 5.1 after it are a hint for the formulation of the following result.

**Theorem 5.2.** A linear transport L along paths in a vector bundle, with  $C^1$  manifold as a bundle space, is Euclidean on U (resp. along  $\gamma$ ) iff for some, and hence for every, frame  $\{e_i\}$  on U (resp. on  $\gamma(J)$ ) there exists a matrix-valued function  $\mathbf{F}_0$ on U such that the matrix  $\mathbf{L}$  of L in  $\{e_i\}$  is given by (5.4) for every  $\gamma: J \to U$ (resp. for the given  $\gamma$ ) or, equivalently, iff the matrix  $\mathbf{\Gamma}$  of the coefficients of L in  $\{e_i\}$  is

$$\boldsymbol{\Gamma}(s;\gamma) = \boldsymbol{F}_0^{-1}(\gamma(s)) \frac{\mathrm{d}\boldsymbol{F}_0(\gamma(s))}{\mathrm{d}s}.$$
(5.4')

Proof. Suppose a linear transport L is Euclidean. There is a frame  $\{e_i^0\}$  normal for L on U (resp. along  $\gamma$ ). Define a matrix  $\mathbf{F}_0(p)$  via the expansion  $e_i|_p = (\mathbf{F}_0(p))^j{}_i e_j^0|_p$ ,  $p \in U$ . Since, by definition, the matrix of L in  $\{e_i^0\}$  is the unit (identity) matrix on U, the matrix of L in  $\{e_i\}$  is given via (5.4) due to (3.11). Conversely, if (5.4) holds in  $\{e_i\}$  on U, the frame  $\{e_i'|_p = (\mathbf{F}_0^{-1}(p))^j{}_i e_j^0|_p\}$  is normal for L on U (resp. along  $\gamma$ ), as we saw at the end of the proof of Theorem 5.1. The equivalence of (5.4') and (5.4) is a consequence of (3.24) (cf. (3.25), (3.26), and (4.2)).

### 5. On the existence of normal frames

*Remark* 5.3. An alternative proof of theorem follows directly from equation (5.1): we have  $A(s; \gamma) = A(\gamma(s))$  and (5.4') is valid for  $\mathbf{F}_0 = A^{-1}$ .

The proof of Theorem 5.2 suggest a way for generating Euclidean transports along paths by 'inverting' the definition of normal frames: take a given field of bases over  $U \subseteq B$  and define a linear transport by requiring its matrix to be the identity matrix in the given field of bases. More precisely, we have in mind the following. Let  $\{e_i\}$  be a fixed frame on U,  $\{e'_i = A^j_i e_j\}$ , with  $A = \begin{bmatrix} A^j_i \end{bmatrix}$  being non-degenerate, be arbitrary frame on U, and  $\gamma: J \to U$  be a path in U. Define a linear mapping (3.1) by its matrix in  $\{e'_i\}$  (see (3.12)):

$$\boldsymbol{L}(t,s;\gamma) = A(\gamma(t))A^{-1}(\gamma(s)), \qquad A(p) := \left[A_i^j(p)\right], \quad p \in U.$$
(5.5)

According to Proposition 3.4, the mapping  $L: \gamma \mapsto L^{\gamma}: (s,t) \mapsto L_{s \to t}^{\gamma}$  is a linear transport along paths. By Theorem 5.1, this transport is Euclidean. Moreover, from (3.11), we see that the matrix of L in  $\{e_i\}$  is unit on U, i.e.,  $\{e_i\}$  is a frame normal for L on U. We call this Euclidean transport generated by (or assigned to) the given initial frame, which is normal for it.

**Proposition 5.3.** All frames normal for a Euclidean linear transport along paths in U generate one and the same Euclidean transport along paths in U coinciding with the initial one.

*Proof.* The result is an almost evident consequence of (5.5) and Corollary 4.5.  $\Box$ 

**Proposition 5.4.** Two or more frames on U generate one and the same Euclidean transport along paths iff they are connected via linear transformations with constant (on U) coefficients.

*Proof.* If  $\{e_i\}$  and  $\{e'_i\}$  generate L, then they are normal for it (Proposition 5.3) and, by Corollary 4.4, they are connected in the way pointed. The converse is a trivial corollary of (5.5) for A(p) = const with  $p \in U$ .

In this way we have established a bijective correspondence between the set of Euclidean linear transports along paths in U and the class of sets of frames on U connected by linear transformations with constant coefficients.

The comparison of Proposition 5.2 with Theorem 5.1 suggests that a transport is Euclidean in  $U \subseteq B$  iff (5.2) holds for every  $\gamma: J \to U$ . But this is not exactly the case. The right result is the following one.

**Theorem 5.3.** A linear transport L along paths is Euclidean on some path-connected set<sup>2</sup>  $U \subseteq B$  iff the next three conditions are valid:

(i) Equation (5.2) holds for every continuous path  $\gamma: J \to U$ ;

 $<sup>^{2}</sup>$ A set is path-connected if every two its points can be connected by a continuous path lying entirely in it. Sometimes such sets are called linearly connected or arc-connected.

(ii) The transport along a product of paths is equal to the composition of the transports long the paths of the product, i.e.,

$$L^{\gamma_1\gamma_2} = L^{\gamma_2} \circ L^{\gamma_1} \tag{5.6}$$

where  $\gamma_1$  and  $\gamma_2$  are paths in U such that the end of  $\gamma_1$  coincides with the beginning of  $\gamma_2$  and  $\gamma_1\gamma_2$  is the product of these paths;

(iii) For any subinterval  $J' \subseteq J$  the locality condition

$$L_{s \to t}^{\gamma | J'} = L_{s \to t}^{\gamma}, \qquad s, t \in J' \subseteq J, \tag{5.7}$$

with  $\gamma | J'$  being the restriction of  $\gamma \colon J \to U$  to J', is valid.

Remark 5.4. Here and below we do not present and use a particular definition of the product of paths. There are slightly different versions of that definition; for details see [108, 132] or [114, Section 3]. Our results are independent of any concrete such definition because the transports, we are considering here, are independent of the particular path they are acting along (see Theorem 5.1).

*Proof.* If L is Euclidean, then, by Definition 4.4, it admits normal frame(s) along every  $\gamma: J \to U$  and, consequently, according to Proposition 5.2, the condition (5.2) is valid along every  $\gamma: J \to U$ . By Theorem 5.1, the transport  $L_{s \to t}^{\gamma}$ ,  $s, t \in J$  depends only on the points  $p = \gamma(s)$  and  $q = \gamma(t)$  but not on the particular path  $\gamma$  connecting  $p, q \in U$ . Equations (5.6) and (5.7) follow from here.

Conversely, let (5.2), (5.6), and (5.7) be true for all paths  $\gamma$ ,  $\gamma_1$ , and  $\gamma_2$  in U, the end of  $\gamma_1$  coinciding with the beginning of  $\gamma_2$ , and subinterval  $J' \subseteq J$ . Meanwhile, we notice the equality

$$L^{\gamma^{-1}} = (L^{\gamma})^{-1}, \tag{5.8}$$

 $\gamma^{-1}$  being the path inverse to  $\gamma^{3}$ , which is a consequence of (5.2) and (5.6).

Let  $p_0$  be arbitrarily chosen fixed point in U and  $\{e_i^0\}$  an arbitrarily fixed basis in the fibre  $\pi^{-1}(p_0)$  over it. In the fibre  $\pi^{-1}(p)$  over  $p \in U$ , we define a basis  $\{e_i|_p\}$  via (cf. (5.3))

$$e_i|_p := L_{s_0 \to s}^{\gamma_{p_0, p}} (e_i^0)$$
(5.9)

where  $\gamma_{p_0,p}: J \to U$  is an arbitrary continuous path through  $p_0$  and p, i.e., for some  $s_0, s \in J$ , we have  $\gamma_{p_0,p}(s_0) = p_0$  and  $\gamma_{p_0,p}(s) = p$ . Below we shall prove that the field  $\{e_i\}$  of bases over U is normal for L on U.

At first, we shall prove the independence of  $e_i|_p$  from the particular continuous path  $\gamma_{p_0,p}$ . Let  $\beta_a: J_a \to U$ , a = 1, 2 and  $\beta_a(s_a) = p_0$  and  $\beta_a(t_a) = p$  for some  $s_a, t_a \in J_a, a = 1, 2$ . For definiteness, we assume  $s_a \leq t_a$ . (The other combinations

<sup>&</sup>lt;sup>3</sup>Here we do not need a particular definition of  $\gamma^{-1}$  (cf. Remark 5.4). More precisely, if  $\gamma \colon [s,t] \to U$ , and  $\gamma^{-1} \colon [s',t'] \to U$ , for  $s,t,s',t' \in \mathbb{R}$ , s < t, s' < t', and  $\gamma^{-1}(s') = \gamma(t)$ , we shall apply (5.8) in the form  $L_{s' \to t'}^{\gamma^{-1}} = (L_{s \to t}^{\gamma})^{-1} = L_{t \to s}^{\gamma}$ .

of ordering between  $s_1$ ,  $t_1$ ,  $s_2$ , and  $t_2$  can be considered analogously.) Defining  $\beta'_a := \beta_a | [s_a, t_a], a = 1, 2$  and using (5.7), (5.8), (5.6), and (5.2), we get

$$L_{s_2 \to t_2}^{\beta_2} \circ L_{t_1 \to s_1}^{\beta_1} = L_{s_2 \to t_2}^{\beta_2'} \circ L_{t_1 \to s_1}^{\beta_1'} = L_{s_2 \to t_2}^{\beta_2'} \circ L_{s_1 \to t_1}^{(\beta_1')^{-1}} = L_{s_0 \to t_0}^{(\beta_1')^{-1}\beta_2'} = \mathsf{id}_{\pi^{-1}(p)},$$

where  $(\beta'_1)^{-1}\beta'_2$ :  $[s_0, t_0] \to U$  is the product of  $(\beta'_1)^{-1}$  and  $\beta'_2$  and we have used that, from the definition of  $(\beta'_1)^{-1}$  and  $\beta'_2$ , is clear that  $((\beta'_1)^{-1}\beta'_2)(s_0) =$  $((\beta'_1)^{-1}\beta'_2)(t_0) = p$ , i.e.,  $(\beta'_1)^{-1}\beta'_2$  is a closed path passing through p. Applying the last result, (3.2), and (3.3), we obtain:

$$L_{s_2 \to t_2}^{\beta_2} e_i^0 = \left( L_{s_2 \to t_2}^{\beta_2} \circ L_{t_1 \to s_1}^{\beta_1} \right) \circ \left( L_{s_1 \to t_1}^{\beta_1} \right) e_i^0 = L_{s_1 \to t_1}^{\beta_1} e_i^0.$$

Since  $\beta_1$  and  $\beta_2$  are arbitrary, from here we conclude that the frame  $\{e_i\}$ , defined via (5.9) on U, is independent from the particular path used in (5.9).

Now we shall prove that  $\{e_i\}$  is normal for L on U, which will complete this proof.

From the proof of Proposition 5.2 (compare (5.9) and (5.3)) follows that  $\{e_i\}$  is normal for L along any path in U passing through  $p_0$ . Let  $\gamma: J \to U$  be such a path,  $s_0 \in J$  be fixed, and  $\beta: [0,1] \to U$  be such that  $\beta(0) = p$  and  $\beta(1) = \gamma(s_0) =: p_0$ . Defining  $\gamma_{\pm} := \gamma | J_{\pm}$  for  $J_{\pm} := \{s \in J, \pm s \ge \pm s_0\}$ , we conclude that  $\{e_i\}$  is normal for L along  $\beta\gamma_+$  and  $\beta\gamma_-^{-1}$ . Take, for example, the path  $\beta\gamma_+$ . If for some  $s'_0, s', s^* \in \mathbb{R}$  is fulfilled  $(\beta\gamma_+)(s') = p, (\beta\gamma_+)(s') = \gamma(s)$ , and  $(\beta\gamma_+)(s^*) = p_0$ , then, applying (5.9), (5.6), and (5.7), we find for  $s \ge s_0$ :

$$e_{i}|_{\gamma(s)} = L_{s_{0}^{\prime} \to s^{\prime}}^{\beta\gamma_{+}} \left( e_{i}|_{p} \right) = L_{s_{0}^{\prime} \to s^{\prime}}^{\beta\gamma_{+}} \circ L_{s^{*} \to s_{0}^{\prime}}^{\beta\gamma_{+}} \left( e_{i}|_{p_{0}} \right) = L_{s^{*} \to s^{\prime}}^{\beta\gamma_{+}} \left( e_{i}|_{p_{0}} \right) \\ = L_{s_{0} \to s}^{\gamma_{+}} \circ L_{0 \to 1}^{\beta} \left( e_{i}|_{p_{0}} \right) = L_{s_{0} \to s}^{\gamma_{+}} \left( e_{i}|_{p} \right) = L_{s_{0} \to s}^{\gamma} \left( e_{i}|_{p} \right).$$

Analogously one can prove that  $e_i|_{\gamma(s)} = L_{s_0 \to s}^{\gamma}(e_i|_p)$  for  $s \leq s_0$  by using  $\beta \gamma_{-}^{-1}$  instead of  $\beta \gamma_{+}$ . So, due to (3.2), the frame  $\{e_i\}$  satisfies (4.1f) along  $\gamma$ . Consequently, by Corollary 4.1, the frame so-constructed is normal for L along  $\gamma$ .

*Remark* 5.5. According to [114, Proposition 3.4], the equality (5.6) is a consequence of (5.7) and the reparametrization condition

$$L_{s \to t}^{\gamma \circ \tau} = L_{\tau(s) \to \tau(t)}^{\gamma}, \qquad s, t \in J''$$
(5.10)

where J'' is an  $\mathbb{R}$ -interval and  $\tau: J'' \to J$  is bijection. Hence in the formulation of Theorem 5.3 we can replace the condition (5.6) with (5.10). So, we have:

**Theorem 5.3'.** A transport L is Euclidean on a path-connected set  $U \subseteq B$  if (5.2), (5.7), and (5.10) are valid for every continuous path  $\gamma: J \to U$ .

*Remark* 5.6. A transport along paths satisfying (5.7) and (5.10) is in fact a parallel transport (along paths). For details, see Section V.8.

The next result is analogous to Proposition 4.4. According to it, a frame normal for L on  $U \subseteq B$ , if any, can be obtained by L-transportation of a fixed basis over some point in U to the other points in U.

**Proposition 5.5.** If L is a Euclidean transport on a path-connected set  $U \subseteq B$  and  $\{e_i^0\}$  is a given basis in  $\pi^{-1}(p_0)$  for a fixed  $p_0 \in U$ , then the frame  $\{e_i\}$  over U defined via

$$e_i|_p = L_{s_0 \to s}^{\gamma} \left( e_i^0 \right), \tag{5.11}$$

where  $\gamma: J \to U$  is such that  $\gamma(s_0) = p_0$  and  $\gamma(s) = p$  for some  $s_0, s \in J$ , is normal for L on U.

*Proof.* By Theorem 5.1, the basis  $\{e_i|_p\}$  is independent of the particular path  $\gamma$  used in (5.11). According to Theorem 5.3, the conditions (5.2), (5.6), and (5.7) hold for L. Further, repeating step-by-step the last paragraph of the proof of Theorem 5.3, we verify that  $\{e_i\}$  is normal for L on U.

Alternatively, the assertion is a consequence of (3.21) and Proposition 5.6 presented a few lines below.

A simple way to check whether a given frame is normal along some path is provided by the following proposition.

**Proposition 5.6.** A frame  $\{e_i\}$  along  $\gamma: J \to B$  is normal for a linear transport L along paths in  $(E, \pi, B)$ , E being a  $C^1$  manifold, if and only if the liftings  $\hat{e}_i: \gamma \mapsto e_i(\cdot, \gamma)$  are constant (along  $\gamma$ ) with respect to the derivation D generated by L:

$$D^{\gamma}\hat{e}_i = 0. \tag{5.12}$$

*Proof.* If  $\{e_i\}$  is normal for L along  $\gamma$ , equation (4.1f) is valid (see Corollary 4.2), so (5.12) follows from (3.21). If (5.12) holds, by virtue of (3.21), its solution is<sup>4</sup>  $e_i|_{\gamma(s)} = L_{s_0 \to s}^{\gamma}(e_i|_{\gamma(s_0)})$  and consequently, by Proposition 4.4, the frame  $\{e_i\}$  is normal along  $\gamma$ .

Recall (see the remark preceding Definition 2.1), the path  $\gamma$  in Proposition 5.6 cannot be an arbitrary continuous path in *B* as it must be in the set  $\pi \circ P^k(E)$ , with  $P^k(E)$ , k = 0, 1, being the set of  $C^k$  paths in *E*. Notice, the derivative in (5.12) does not require *B* to be a manifold.

Of course, it is true that if (5.12) holds in a frame  $\{e_i\}$  along every path  $\gamma$  in U, the frame  $\{e_i\}$  is normal for L on U. But it is more natural to find a 'global' version of (5.12) concerning the whole set U, not the paths in it. The corresponding result is formulated below as Theorem 9.1 (see also Theorem 6.1).

<sup>&</sup>lt;sup>4</sup>Equation (5.12) is an ordinary differential equation of first order with respect to the local components of  $e_i$  (see (3.23)).

# 6. The case of a manifold as a base

Starting from this section, we consider some peculiarities of frames normal for linear transports along paths in a vector bundle  $(E, \pi, M)$  whose base M is a  $C^1$  differentiable manifold. Besides, the bundle space E will be required to be a  $C^1$  manifold too. This will allow links to be made with the general results of Chapter III concerning frames normal for derivations of the tensor algebra of the vector space of vector fields over a manifold which, in particular, can be linear connections.

The local coordinates of  $p \in M$  will be denoted by  $p^{\mu}$ ,  $p^{\mu} := x^{\mu}(p)$ . Here and below the Greek indices  $\alpha, \beta, \ldots, \mu, \nu, \ldots$  run from 1 to dim M and, as usual, a summation from 1 to dim M on such indices repeated on different levels will be assumed. The below-considered paths, like  $\gamma: J \to M$ , are supposed to be of class  $C^1$  and by  $\dot{\gamma}(s)$  is denoted the vector tangent to  $\gamma$  at  $\gamma(s), s \in J$ , (more precisely at s) i.e.,  $\dot{\gamma}$  is the vector field tangent to  $\gamma$  provided  $\gamma$  is injective. By  $\{E_{\mu}\}$  will be denoted a frame along  $\gamma$  in the bundle space tangent to M, i.e., for every  $s \in J$ the vectors  $E_1|_{\gamma(s)}, \ldots, E_{\dim M}|_{\gamma(s)}$  form a basis in the space  $T_{\gamma(s)}(M)$  tangent to M at  $\gamma(s)$ . In particular, the frame  $\{E_{\mu}\}$  can be a coordinate one,  $E_{\mu}|_{p} = \frac{\partial}{\partial x^{\mu}}|_{p}$ , in some neighborhood of  $p \in \gamma(J)$ . The transports along paths investigated below are supposed to be of class  $C^1$  on the set of  $C^1$  paths in M.

**Proposition 6.1.** Let *L* be a linear transport along paths in  $(E, \pi, M)$ , *E* and *M* being  $C^1$  manifolds, and *L* be Euclidean on  $U \subseteq M$  (resp. along a  $C^1$  path  $\gamma: J \to M$ ). Then the matrix  $\Gamma$  of its coefficients has the representation

$$\Gamma(s;\gamma) = \sum_{\mu=1}^{\dim M} \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s) \equiv \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s)$$
(6.1)

in any frame  $\{e_i\}$  along every (resp. the given)  $C^1$  path  $\gamma: J \to U$ , where  $\Gamma_{\mu} = [\Gamma^i_{\ j\mu}]^{\dim \pi^{-1}(p)}_{i,j=1}$  are some matrix-valued functions, defined on an open set V containing U (resp.  $\gamma(J)$ ) or equal to it, and  $\dot{\gamma}^{\mu}$  are the components of  $\dot{\gamma}$  in some frame  $\{E_{\mu}\}$  along  $\gamma$  in the bundle space T(M) tangent to  $M, \dot{\gamma} = \dot{\gamma}^{\mu} E_{\mu}$ .

*Proof.* By Theorem 5.2, the representation (5.4') is valid in  $\{e_i\}$  for some matrixvalued function  $\mathbf{F}_0$  on U. Hence, if U is a neighborhood, equation (6.1) holds for

$$\Gamma_{\mu}(p) = \mathbf{F}_{0}^{-1}(p) \left( E_{\mu}(\mathbf{F}_{0})|_{p} \right)$$
(6.2)

with  $p \in U$ . In the general case, e.g., if U is a submanifold of M of dimension less than the one of M, the terms  $E_{\mu}(\mathbf{F}_0)|_U$ ,  $\mu = 1, \ldots, \dim M$ , in the last equality may turn to be undefined as the matrix-valued function  $\mathbf{F}_0$  is defined only on U. To overcome this possible problem, let us take some  $C^1$  matrix-valued function  $\mathbf{F}$ , defined on an open set V containing U (resp.  $\gamma(J)$ ) or equal to it, such that  $\mathbf{F}|_U = \mathbf{F}_0$ . Since (5.4) and (5.4') depend only on the values of  $\mathbf{F}_0$ , i.e., on the

 $\Box$ 

ones of F on U, these equations hold also if we replace  $F_0$  in them with F. From the so-modified equality (5.4'), with F for  $F_0$ , we see that (6.1) is valid for

$$\Gamma_{\mu}(p) = \boldsymbol{F}^{-1}(p)(E_{\mu}(\boldsymbol{F}))|_{p}$$
(6.3)

with  $p \in V$ .

Consider now the transformation properties of the matrices  $\Gamma_{\mu}$  in (6.1). Let U be an open set, e.g., U = M. If we change the frame  $\{E_{\mu}\}$  in the bundle space tangent to M,  $\{E_{\mu}\} \mapsto \{E'_{\mu} = B^{\nu}_{\mu}E_{\nu}\}$  with  $B = [B^{\nu}_{\mu}]^{\dim M}_{\mu,\nu=1}$  being non-degenerate matrix-valued function, and simultaneously the bases in the fibres  $\pi^{-1}(p), p \in M$ ,  $\{e_i|_p\} \mapsto \{e'_i|_p = A^j_i(p)e_j|_p\}$ , then, from (3.26) and (6.1), we see that  $\Gamma_{\mu}$  transforms into  $\Gamma'_{\mu}$  such that

$$\Gamma'_{\mu} = B^{\nu}_{\mu} A^{-1} \Gamma_{\nu} A + A^{-1} E'_{\mu} (A) = B^{\nu}_{\mu} A^{-1} \big( \Gamma_{\nu} A + E_{\nu} (A) \big)$$
(6.4)

where  $A := \left[A_i^j\right]_{i,j=1}^{\dim \pi^{-1}(p)}$  is non-degenerate and of class  $C^1$ .

Note 6.1. While deriving (6.4), we supposed (6.1) to be valid on M, i.e., for U = M. If  $U \neq M$ , equation (6.1) holds only on U, i.e., for  $\gamma: J \to U$ . Therefore the result (6.4) is true only on U, but in this case the frames  $\{e_i\}$  and  $\{e'_i\}$  must be defined on an open subset of M containing or equal to U. This follows from (3.26) in which the derivative  $\frac{dA(s;\gamma)}{ds} = \frac{dA(\gamma(s))}{ds}$  enters. To derive (6.4), we have expressed  $\frac{dA(\gamma(s))}{ds}$  as  $(E_{\mu}(A))|_{\gamma(s)}\dot{\gamma}^{\mu}(s)$  which is meaningful iff A is defined on a neighborhood of each point in U. Consequently A, as well as  $\{e_i\}$  and  $\{e'_i\}$ , must be defined on an open set  $V \supseteq U$ . For this reason, below, when derivatives like  $E_{\mu}(A)$  appear, we admit the employed frames in the bundle space E to be defined always on some neighborhood in M containing or equal to the set U on which some normal frames are investigated.

The above considerations, when applied to parallel transports assigned to linear connections (see below Section 11) in the tangent bundle  $(T(M), \pi, M)$ , are the cause for which in the definition of frames normal for a linear connection (see Definition I.5.1 on page 37) is included the requirement they to be defined on a neighborhood, possibly coinciding with U if U is a neighborhood.

Denoting by  $\Gamma^{i}_{\ i\mu}$  the components of  $\Gamma_{\mu}$ , we can rewrite (6.4) as

$$\Gamma'^{i}{}_{j\mu} = \sum_{\nu=1}^{\dim M} \sum_{k,l=1}^{\dim \pi^{-1}(p)} B^{\nu}_{\mu} (A^{-1})^{i}{}_{k}^{l} A^{l}_{j} \Gamma^{k}{}_{l\nu} + \sum_{\nu=1}^{\dim M} \sum_{k=1}^{\dim \pi^{-1}(p)} B^{\nu}_{\mu} (A^{-1})^{i}{}_{k}^{l} E_{\nu} (A^{k}_{j}).$$
(6.5)

Thus, we observe that the functions  $\Gamma^{i}_{j\mu}$  are very similar to the coefficients of a linear connection.<sup>1</sup> Below, in Subsection 13.4, we shall see that this is not accidental (compare (6.1) with (3.29)). These functions are also called coefficients of

<sup>&</sup>lt;sup>1</sup>Compare (6.4) with (I.5.3) or (6.5) with (I.3.5). The equations in these pairs are identical for dim  $M = \dim \pi^{-1}(p), p \in M$ , and B = A.

the transport L. To make a distinction between  $\Gamma_{j}^{i}$  and  $\Gamma_{j\mu}^{i}$ , we call the former ones 2-index coefficients of L and the latter ones 3-index coefficients of L when there is a risk of ambiguities. It should be emphasized on the fact that the 2-index coefficients are defined with respect to a single frame  $\{e_i\}$  along  $\gamma$  in the vector bundle  $(E, \pi, M)$ , while the 3-index coefficients are defined with respect to a pair  $(\{e_i\}, \{E_{\mu}\})$  of frames,  $\{e_i\}$  in the bundle space E of  $(E, \pi, M)$  and  $\{E_{\mu}\}$  in the bundle space T(M) of the bundle  $(T(M), \pi, M)$  tangent to M. Besides, if (6.1) holds for every  $\gamma: J \to U$  for a transport L, then, in the general case, there are (infinitely) many such representations unless U an open set. For instance, if (6.1) is valid for some  $\Gamma_{\mu}$ , it is also true if we replace in it  $\Gamma_{\mu}$  with  $\Gamma_{\mu} + G_{\mu}$  where the matrix-valued functions  $G_{\mu}$  are such that  $G_{\mu}\dot{\gamma}^{\mu} = 0$  for every  $\gamma: J \to U$ . Hence, generally, the 3-index coefficients of L depend also on U and are not unique due to, e.g., the nonuniqueness of  $\mathbf{F}$  in (6.3).

Prima facie one may think that the converse of Proposition 6.1 is true, but this is not the general case, i.e., if in some frame equation (6.1) holds, then  $\{e_i\}$ is generally not normal unless some conditions are fulfilled. Thus (6.1) is only a necessary, but, generally, not sufficient condition for a frame to be normal.

**Theorem 6.1.** A  $C^2$  linear transport L along paths is Euclidean on a neighborhood  $U \subseteq M$  if and only if in every frame the matrix  $\Gamma$  of its coefficients has a representation (6.1) along every  $C^1$  path  $\gamma$  in U in which the matrix-valued functions  $\Gamma_{\mu}$ , defined on an open set containing U or equal to it, satisfy the equalities

$$\left(R_{\mu\nu}(-\Gamma_1,\ldots,-\Gamma_{\dim M})\right)(p)=0\tag{6.6}$$

where  $p \in U$  and

$$R_{\mu\nu}(-\Gamma_1,\ldots,-\Gamma_{\dim M}) := -E_{\nu}(\Gamma_{\mu}) + E_{\mu}(\Gamma_{\nu}) + \Gamma_{\mu}\Gamma_{\nu} - \Gamma_{\nu}\Gamma_{\mu} - C_{\mu\nu}^{\varkappa}\Gamma_{\varkappa} \quad (6.7)$$

in a frame  $\{E_{\mu}\}$  over U in the bundle space tangent to M, with  $C_{\mu\nu}^{\varkappa}$  being the structure functions of  $\{E_{\mu}\}, [E_{\mu}, E_{\nu}] = : C_{\mu\nu}^{\varkappa} E_{\varkappa}.$ 

*Remark* 6.1. This result is a direct analogue of Theorem III.6.1 on page 157 in the theory considered here.

Remark 6.2. If we change the frame  $\{e_i\}$ ,  $\{e_i\} \mapsto \{e'_i = A^j_i e_j\}$ , over U and simultaneously the frame  $\{E_{\mu}\}$ ,  $\{E_{\mu}\} \mapsto \{E'_{\mu} = B^{\nu}_{\mu}E_{\nu}\}$ , in the tangent bundle space over U, from (6.7), (I.8.4), and (6.4), we get that the matrices  $R_{\mu\nu}$  transform into

$$R'_{\mu\nu} = B^{\varkappa}_{\mu} B^{\lambda}_{\nu} A^{-1} R_{\varkappa\lambda} A. \tag{6.8}$$

Therefore the conditions (6.6) have an invariant character, i.e., they are independent of the particular choice of the frames (or coordinates) involved in them. The last result also shows that the quantities  $R_{\mu\nu}$  depend on the frames  $\{e_i\}$  and  $\{E_{\mu}\}$ on U and are completely independent of the their values outside U (if  $U \neq M$ ) in case they are defined there (see Note 6.1). *Remark* 6.3 (*Important!*). This theorem has a stronger version: the requirement the representation (6.1) to hold can be dropped as it is a consequence of (6.6). We skip this detail till Subsection 9.2 (see the proof of Theorem 9.1 and the comments after it on page 269). Corollary 9.1 on page 269 is a formulation of the mentioned stronger variant of Theorem 6.1 employing the concept of a 'curvature'. The same remarks are valid with respect to Theorem 6.2 on page 253 below.

*Proof.* NECESSITY. For a transport L Euclidean on U is valid (6.1) due to Proposition 6.1. Moreover, we know from the proof of this proposition that  $\Gamma_{\mu}$  admit representation (6.3) for some  $C^1$  non-degenerate matrix-valued function  $\mathbf{F}$ . The proof of the necessity is completed by the following lemma.

**Lemma 6.1.** A set of matrix-valued functions  $\{\Gamma_{\mu} : \mu = 1, ..., \dim M\}$ , of class  $C^1$  and defined on a neighborhood V, admits a representation (6.3) iff the conditions (6.6) are fulfilled for  $p \in V$ .

Proof of Lemma 6.1. A representation (6.3) exists iff it, considered as a matrix linear partial differential equation of first order, has a solution with respect to F. Rewriting (6.3) as

$$E_{\mu}(\boldsymbol{F}^{-1})|_{p} = -\Gamma_{\mu}(p)\boldsymbol{F}^{-1}(p), \qquad p \in V,$$

from Lemma II.4.1 on page 105 with N = M and  $E_{\mu}$  for  $e_{\mu}$ , we conclude that the solutions of this equation with respect to  $\mathbf{F}^{-1}$  exist iff (6.6) holds. In fact, fixing some initial value  $\mathbf{F}^{-1}(p_0) = f_0$ , we get (see Remark II.4.5)

$$\mathbf{F}(p) = f_0^{-1} Y^{-1}(p, p_0; -\Gamma_1, \dots, -\Gamma_{\dim M})$$
(6.9)

where  $Y(p, p_0; Z_1, \ldots, Z_{\dim M})$  is the solution of the initial-value problem

$$E_{\mu}(Y)|_{p} = Z_{\mu}(p)Y|_{p}, \qquad Y|_{p=p_{0}} = \mathbb{1}.$$
 (6.10)

Here  $Z_1, \ldots, Z_{\dim M}$  are continuous matrix-valued functions and  $\mathbb{1}$  is the identity (unit) matrix of the corresponding size. According to Lemma II.4.1, the problem (6.10) with  $Z_{\mu} = -\Gamma_{\mu}$  has (a unique) solution (of class  $C^2$ ) iff the (integrability) conditions (6.6) are valid.

SUFFICIENCY. Let (6.1) and (6.6) be valid. As a consequence of Lemma 6.1, there is a representation (6.3) for  $\Gamma$  with some F. Substituting (6.3) into (6.1), we get

$$\Gamma(s;\gamma) = \mathbf{F}^{-1}(\gamma(s))E_{\mu}(\mathbf{F})|_{\gamma(s)}\dot{\gamma}^{\mu}(s) = \mathbf{F}^{-1}(\gamma(s))\frac{\mathrm{d}\mathbf{F}(\gamma(s))}{\mathrm{d}s}.$$

So, by Theorem 5.2 (see (5.4') for  $\mathbf{F}_0 = \mathbf{F}|_U$ ), the considered transport L along paths is Euclidean.

The just-proved Theorem 6.1 expresses a very important practical necessary and sufficient condition for existence of frames normal on neighborhoods because the conditions (6.1) and (6.6) are easy to check for a given linear transport along paths in bundles with a differentiable manifold as a base.

Now, combining (4.1c) and (6.1), applying Corollaries 4.1 and 4.2, and using the arbitrariness of  $\gamma$ , we can formulate the following essential result.

**Proposition 6.2.** A necessary and sufficient condition for a frame to be normal on a neighborhood  $U \subseteq M$  for a Euclidean on U linear transport along paths in  $(E, \pi, M)$  is the vanishment of its 3-index coefficients, i.e.,

$$\Gamma_{\mu}(p) := \left[\Gamma_{j\mu}^{i}\right]_{i,j=1}^{\dim \pi^{-1}(p)} = 0$$
(6.11)

for every  $p \in U$ , where  $\Gamma_{\mu}(p)$  define the (2-index) coefficients of the transport via (6.1).

Remark 6.4. The assumption in this proposition for U to be a neighborhood is an essential one. From it and the arbitrariness of  $\gamma$  follows that  $\dot{\gamma}(s)$  is an arbitrary vector in  $T_{\gamma(s)}(M)$  which, together with (6.1) and  $\Gamma(s;\gamma) = 0$ , implies (6.11). If U is not a neighborhood, then, generally, such a conclusion can not be made. For instance, if U is a submanifold of M and dim  $U < \dim M$ , then  $\dot{\gamma}(s) \in T_{\gamma(s)}(U)$  but  $\dot{\gamma}(s) \notin T_{\gamma(s)}(M) \setminus T_{\gamma(s)}(U)$  as  $\gamma: J \to U$  is a path in U. Therefore, in this example, only dim U of the matrices  $\Gamma_1, \ldots, \Gamma_{\dim M}$  must vanish in a (suitable) frame normal on U, the remaining dim M – dim U of them need not to be zeros on U. Obviously, this assertion can be inverted, i.e., frames normal on a submanifold U exist iff in them dim U of the matrices of the transport's 3-index coefficients vanish on U(in suitable coordinates or frames). More details for normal frames in which the 3-index coefficients of a linear transport vanish are presented in Section 12.

Because of Propositions 6.1 and 6.2, as we said above, the functions  $\Gamma^i_{j\mu}$  are convenient to be called also coefficients of the transport, like  $\Gamma^i_j$ . If there is a risk of ambiguity whether we have in mind  $\Gamma^i_j$  or  $\Gamma^i_{j\mu}$ , when speaking about transport's (or corresponding derivation's) coefficients, we shall call them respectively 2-*index* and 3-*index* coefficients. Note that any linear transport has 2-index coefficients while 3-index ones exist only for some of them; in particular such are the Euclidean transports (see Proposition 6.1 and Theorem 6.2). Through the 3-index coefficients can be defined concrete classes of, possibly Euclidean on some sets, linear transports along paths in a given vector bundle with a manifold as a base. For the purpose one should define a transport by the matrix (6.1) of its 2-index coefficients in which  $\Gamma_{\mu}$  are fixed matrix-valued functions over the whole base M. In particular, if  $\Gamma_{\mu}$  are the matrices of the coefficients of a linear connection in the tangent bundle over M, we obtain in this way the class of parallel transports generated by such connections in this bundle (see (3.29) and the assertion after it).

It should be emphasized, the 3-index coefficients  $\Gamma^i_{j\mu}$  of a given linear transport L admitting them are defined *uniquely* on  $U \subseteq M$  by (6.3) or (6.2) if (and only if) U is an open subset of M, e.g., if U = M; the same is valid for the transformed coefficients (6.5). If U is not such, the 3-index coefficients of L contain an

arbitrariness connected with the one of the matrix-valued function  $\mathbf{F}$ , appearing in (6.3), which is subjected only on the condition  $\mathbf{F}|_U = \mathbf{F}_0$ . Besides, if (6.1) holds for some  $\Gamma_{\mu}$ , it remains true if we replace in its right-hand side  $\Gamma_{\mu}$  with  $\Gamma_{\mu} + G_{\mu}$ where the matrix-valued functions  $G_{\mu}$  are such that  $G_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s) = 0$  for all  $C^1$ paths  $\gamma: J \to U$  and  $s \in J$ . The mentioned arbitrariness will be described below when U is a submanifold of M.

Now we are going to find an analogue of Theorem 6.1 when the neighborhood  $U \subseteq M$  in it is replaced with a submanifold of the base M.

Let N be a submanifold of M and L a linear transport along paths in  $(E, \pi, M)$  which is Euclidean on N. Let the  $C^1$  matrix-valued function  $\mathbf{F}_0$  determines the coefficients' matrix of L via (5.4'). Suppose  $p_0 \in N$  and (V, x) is a chart of M such that  $V \ni p_0$  and the local coordinates of every  $p \in N \cap V$  are  $x(p) = (x^1(p), \ldots, x^{\dim N}(p), t_0^{\dim N+1}, \ldots, t_0^{\dim M})$ , where  $t_0^{\rho}$ ,  $\rho = \dim N + 1, \ldots, \dim M$ , are constant numbers (see the definition of a submanifold on page 7).

In the chart (V, x), we have  $\frac{\mathrm{d} F_0(\gamma(s))}{\mathrm{d} s} = \sum_{\alpha=1}^{\dim N} \frac{\partial F_0}{\partial x^{\alpha}} \Big|_{\gamma(s)} \dot{\gamma}^{\alpha}(s)$ , with  $\gamma^{\mu} := x^{\mu} \circ \gamma$ , for every  $C^1$  path  $\gamma \colon J \to N$  and  $s \in J$ . From here and (5.4'), it follows that (6.1) holds for

$$\Gamma_{\alpha}(p) = \boldsymbol{F}_{0}^{-1}(p) \frac{\partial \boldsymbol{F}_{0}}{\partial x^{\alpha}} \Big|_{p}, \qquad \alpha = 1, \dots, \dim N$$
(6.12)

and arbitrary  $\Gamma_{\dim N+1}, \ldots, \Gamma_{\dim M}$  since in the coordinates  $\{x^{\mu}\}$  is fulfilled  $\gamma^{\rho}(s) = t_0^{\rho} = \text{const}$  and hence

$$\dot{\gamma}^{\dim N+1} = \dots = \dot{\gamma}^{\dim M} \equiv 0. \tag{6.13}$$

Comparing (6.12) with (6.2) for  $E_{\mu} = \frac{\partial}{\partial x^{\mu}}$ , we conclude that  $\Gamma_{\alpha}$ , given via (6.12), are exactly the first dim N of the matrices  $\Gamma_{\mu} = [\Gamma^{i}{}_{j\mu}]$  of the 3-index coefficients of the transport L in the pair of frames  $(\{e_i\}, \{\frac{\partial}{\partial x^{\mu}}\})$ . As we said, the rest of the 3-index coefficients of L (on N) are completely arbitrary. In particular, one can choose them according to (6.3),

$$\Gamma_{\rho}(p) = \boldsymbol{F}^{-1}(p) \frac{\partial \boldsymbol{F}}{\partial x^{\rho}}, \qquad \rho = \dim N + 1, \dots, \dim M, \quad \boldsymbol{F}|_{N} = \boldsymbol{F}_{0}, \qquad (6.14)$$

which leads to the validity of (6.3) in every frame, or, if the representation (6.1) holds for every  $\gamma: J \to M$  (this does not mean that L is Euclidean on M!), the matrices  $\Gamma_{\rho}$  can be identified with the ones appearing in (6.1) in the frame  $\{\frac{\partial}{\partial x^{\mu}}\}$ .

If  $\{x'^{\mu}\}$  are other coordinates on V like  $\{x^{\mu}\}$ , i.e.,  $x'^{\rho}(p) = \text{const}$  for  $p \in N \cap V$  and  $\rho = \dim N + 1, \ldots, \dim M$ , the change  $\{x^{\mu}\} \mapsto \{x'^{\mu}\}$ , combined with  $\{e_i\} \mapsto \{e'_i = A^j_i e_j\}$  leads to

$$\Gamma_{\alpha} \mapsto \Gamma_{\alpha}' = B_{\alpha}^{\beta} A^{-1} \Gamma_{\beta} A + A^{-1} \frac{\partial A}{\partial x'^{\alpha}}, \quad B_{\alpha}^{\beta} := \frac{\partial x^{\beta}}{\partial x'^{\alpha}}, \quad \alpha, \beta = 1, \dots, \dim N$$
(6.15)

#### 6. The case of a manifold as a base

on  $N \cap V$ . So, equation (6.4) remains valid only for frames  $\{E_{\mu}\}$  normal on N. But using the arbitrariness of  $\Gamma_{\rho}$ , we can force (6.4) to hold on N for arbitrary frames defined on a neighborhood of N.

The above discussion implies that the condition (6.6) in Theorem 6.1, when applied on a submanifold N, imposes restrictions on the transport L as well as ones on the 'inessential' 3-index coefficients of L, like  $\Gamma_{\rho}$  above, or on the matrixvalued function  $\mathbf{F}$  entering in (6.3) or in (6.14). Since the restrictions of the last type are not connected with the transport L, below we shall 'repair' Theorem 6.1 on submanifolds in such a way as to exclude them from the final results.

**Theorem 6.2.** A linear transport L along paths is Euclidean on a submanifold N of M if and only if in every frame  $\{e_i\}$ , in the bundle space over N, the matrix of its coefficients has a representation (6.1) along every  $C^1$  path in N and, for every  $p_0 \in N$  and a chart (V, x) of M such that  $V \ni p_0$  and  $x(p) = (x^1(p), \ldots, x^{\dim N}(p), t_0^{\dim N+1}, \ldots, t_0^{\dim M})$  for every  $p \in N \cap V$  and constant numbers  $t_0^{\dim N+1}, \ldots, t_0^{\dim M}$ , the equalities

$$\left(R^{N}_{\alpha\beta}(-\Gamma_{1},\ldots,-\Gamma_{\dim N})\right)(p)=0,\qquad \alpha,\beta=1,\ldots,\dim N$$
(6.16)

hold for all  $p \in N \cap V$  and

$$R^{N}_{\alpha\beta}(-\Gamma_{1},\ldots,-\Gamma_{\dim N}) = -\frac{\partial\Gamma_{\alpha}}{\partial x^{\beta}} - \frac{\partial\Gamma_{\beta}}{\partial x^{\alpha}} + \Gamma_{\alpha}\Gamma_{\beta} - \Gamma_{\beta}\Gamma_{\alpha}.$$
 (6.17)

Here  $\Gamma_1, \ldots, \Gamma_{\dim N}$  are first dim N of the matrices of the 3-index coefficients of L in the coordinate frame  $\{\frac{\partial}{\partial x^{\mu}}\}$  in the tangent bundle space over  $N \cap V$ . They are uniquely defined via (6.12).

*Remark* 6.5. In the theory considered here, this result is a direct analogue of Theorem III.8.1, point (i), on page 165.

Remark 6.6. It is intuitively clear, generally not all of the equations (6.16) are independent. One can expect only  $(\dim N)[(\dim N)-1]/2$  of them to be independent because of  $R_{\mu\nu} = -R_{\nu\mu}$ , due to (6.7).

Remark 6.7. This theorem is, in fact, a special case of Theorem 6.1: if in the latter theorem we put U = N, restrict the transport L to the bundle  $(\pi^{-1}(N), \pi|_N, N)$ , replace M with N, and notice that  $\{x^1, \ldots, x^{\dim N}\}$  provide an internal coordinate system on N, we get the former one with  $E_{\alpha} = \partial/\partial x^{\alpha}$ . Because of the importance of the result obtained, we call it 'theorem' and present below its independent proof.

*Proof.* If L is Euclidean on N, equation (6.1) holds in every frame on N (Proposition 6.1); in particular it is valid in the frame  $\{\frac{\partial}{\partial x^{\mu}}\}$ , induced by the chart (V, x), in which, as was proved above, equation (6.12) is satisfied. The substitution of (6.12) into (6.17) results in (6.16). Conversely, let (6.1) for  $\gamma: J \to N$  and (6.16) be valid. By Lemma 6.1 with N for M, from (6.16) follows the existence of a representation (6.12) for some matrix-valued function  $F_0$  on N. Substituting (6.12)

into (6.1) and using that  $\gamma$  is a path in N and (6.13) is valid, in the frame  $\left\{\frac{\partial}{\partial x^{\mu}}\right\}$ , we obtain:

$$\begin{split} \mathbf{\Gamma}(s;\gamma) &= \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s) = \sum_{\alpha=1}^{\dim N} \Gamma_{\alpha}(\gamma(s))\dot{\gamma}^{\alpha}(s) \\ &= \mathbf{F}_{0}^{-1}(\gamma(s)) \sum_{\alpha=1}^{\dim N} \left. \frac{\partial \mathbf{F}_{0}}{\partial x^{\alpha}} \right|_{\gamma(s)} \dot{\gamma}^{\alpha}(s) = \mathbf{F}_{0}^{-1}(\gamma(s)) \frac{\partial \mathbf{F}}{\partial x^{\mu}} \right|_{\gamma(s)} \dot{\gamma}^{\mu}(s) \\ &= \mathbf{F}_{0}^{-1}(\gamma(s)) \frac{\mathrm{d}\mathbf{F}(\gamma(s))}{\mathrm{d}s} = \mathbf{F}_{0}^{-1}(\gamma(s)) \frac{\mathrm{d}\mathbf{F}_{0}(\gamma(s))}{\mathrm{d}s} \end{split}$$

where  $\mathbf{F}$  is a  $C^1$  matrix-valued function defined on an open set containing N or equal to it and such that  $\mathbf{F}|_N = \mathbf{F}_0$ . Thus, by Theorem 5.2, the transport L is Euclidean on N.

**Corollary 6.1.** Every linear transport along paths in a vector bundle whose base and bundle spaces are  $C^1$  manifolds, is Euclidean at every single point or along every path without self-intersections.

*Proof.* See Theorem 6.2 for dim N = 0, 1, in which cases  $R^N_{\alpha\beta} \equiv 0$ .

It should be noted, the last result agrees completely with Proposition 5.1 and Corollary 5.1.

# 7. Linear transports and normal frames in line bundles

The purpose of this section is to exemplify the above general considerations concerning linear transports and normal frames on *line vector bundles*, i.e., ones with one-dimensional fibres.

Let  $(E, \pi, M)$  be one-dimensional vector bundle over a  $C^1$  manifold M. Thus the (typical) fibre of  $(E, \pi, M)$  can be identified with  $\mathbb{K}$  and then the fibre  $\pi^{-1}(x)$ over  $x \in M$  will be an isomorphic image of  $\mathbb{K}$ . Let  $\gamma: J \to M$  be of class  $C^1$  and Lbe a linear transport along paths in  $(E, \pi, M)$ . A frame  $\{e\}$  along  $\gamma$  consists of a single non-zero vector field  $e: (s; \gamma) \to e(s; \gamma) \in \pi^{-1}(\gamma(s)) \setminus \{0\}, s \in J$ , and in it the matrix of  $L^{\gamma}$  at  $(t, s) \in J \times J$  is simply a number  $L(t, s; \gamma) \in \mathbb{K}$ ,  $L_{s \to t}^{\gamma}(ue(s; \gamma)) =$  $uL(t, s; \gamma)e(t; \gamma)$  for  $u \in \mathbb{K}$  and  $s, t \in J$ . By Proposition 3.4, the general form of L is

$$\boldsymbol{L}(t,s;\gamma) = \frac{f(s;\gamma)}{f(t;\gamma)}$$
(7.1)

where  $f: (s; \gamma) \mapsto f(s; \gamma) \in \mathbb{K} \setminus \{0\}$  is defined up to (left) multiplication with a function of  $\gamma$  (Proposition 3.5). Respectively, due to (3.25), the matrix of the coefficient(s) of L is

$$\Gamma(s;\gamma) = \frac{\partial \boldsymbol{L}(t,s;\gamma)}{\partial s}\Big|_{t=s} = \frac{1}{f(s;\gamma)}\frac{\mathrm{d}f(s;\gamma)}{\mathrm{d}s} = \frac{\mathrm{d}}{\mathrm{d}s}\big[\ln(f(s;\gamma))\big]$$
(7.2)

## 7. Transports and normal frames in line bundles

and (3.27) takes the form

$$\boldsymbol{L}(t,s;\gamma) = \exp\left(-\int_{s}^{t} \boldsymbol{\Gamma}(\sigma;\gamma) \,\mathrm{d}\sigma\right). \tag{7.3}$$

A change  $e(s;\gamma) \mapsto e'(s;\gamma) = a(s;\gamma)e(s;\gamma)$ , with  $a(s;\gamma) \in \mathbb{K} \setminus \{0\}$ , of the frame  $\{e\}$  implies (see (3.11) and (3.26))

$$\boldsymbol{L}(t,s;\gamma) \mapsto \boldsymbol{L}'(t,s;\gamma) = \frac{a(s;\gamma)}{a(t;\gamma)} \boldsymbol{L}(t,s;\gamma)$$
(7.4a)

$$\Gamma(s;\gamma) \mapsto \Gamma'(s;\gamma) = \Gamma(s;\gamma) + \frac{\mathrm{d}}{\mathrm{d}s} \left[ \ln(a(s;\gamma)) \right].$$
(7.4b)

The explicit local action of the derivation D along paths generated by L is

$$D_s^{\gamma} \lambda = \left(\frac{\mathrm{d}\lambda_{\gamma}(s)}{\mathrm{d}s} + \Gamma(s;\gamma)\lambda_{\gamma}(s)\right)e(s;\gamma)$$
(7.5)

where  $\lambda \in \text{PLift}^1(E, \pi, M)$  and (3.23) was used. Let us now look on the normal frames on one-dimensional vector bundles.

A frame  $\{e\}$  is normal for L along  $\gamma$  (resp. on U) iff in that frame (7.1) holds with

$$f(s;\gamma) = f_0(\gamma) \tag{7.6}$$

where  $\gamma: J \to M$  (resp.  $\gamma: J \to U$ ) and  $f_0: \gamma \mapsto f_0(\gamma) \in \mathbb{K} \setminus \{0\}$  (see Remark 4.2 and Proposition 4.1). Since, in a frame normal along  $\gamma$  (resp. on U), it is fulfilled

$$\boldsymbol{L}(t,s;\gamma) = \mathbb{1}, \quad \boldsymbol{\Gamma}(s;\gamma) = 0 \tag{7.7}$$

for the given path  $\gamma$  (resp. every path in U), in every frame  $\{e' = ae\}$ , we have

$$\boldsymbol{L}'(t,s;\gamma) = \frac{a(s;\gamma)}{a(t;\gamma)}, \quad \boldsymbol{\Gamma}'(s;\gamma) = \frac{\mathrm{d}}{\mathrm{d}s} \big[ \ln(a(s;\gamma)) \big].$$
(7.8)

In addition, for Euclidean on  $U \subseteq M$  transport L, the representation

$$\Gamma'(s;\gamma) = \Gamma'_{\mu}(\gamma(s))\dot{\gamma}'^{\mu}(s)$$
(7.9)

holds for every  $C^1$  path  $\gamma: J \to U$  and some  $\Gamma'_{\mu}: V \to \mathbb{K}$  with V being an open set such that  $V \supseteq U$  (Proposition 6.1). This means (see Theorems 5.1 and 5.2) that (7.8) holds for

$$a(s;\gamma) = a_0(\gamma(s)), \tag{7.10}$$

where  $a_0: U \to \mathbb{K} \setminus \{0\}$ , and, consequently, the equality (7.9) can be satisfied if we choose

$$\Gamma'_{\mu} = E_{\mu}(a) \tag{7.11}$$

with  $a: V \to \mathbb{K}$ ,  $a|_U = a_0$  and  $\{E_{\mu}\}$  being a frame in the bundle space tangent to M which, in particular, can be a coordinate one,  $E_{\mu} = \frac{\partial}{\partial x^{\mu}}$ . Of course, if U is not an open set, this choice of  $\Gamma'_{\mu}$  is not necessary (see Section 6); for example, the equality (7.9) will be preserved, if to the right-hand side of (7.11) is added a function  $G'_{\mu}$  such that  $G'_{\mu}\dot{\gamma}'^{\mu} = 0$ .

By virtue of (6.4), the functions  $\Gamma_{\mu}$  and  $\Gamma'_{\mu}$  in two arbitrary pairs of frames  $(\{e\}, \{E_{\mu}\})$  and  $(\{e' = ae\}, \{E'_{\mu} = B^{\nu}_{\mu}E_{\nu}\})$ , respectively, are connected via

$$\Gamma'_{\mu} = B^{\nu}_{\mu}\Gamma_{\nu} + \frac{1}{a}E'_{\mu}(a) = B^{\nu}_{\mu}\big(\Gamma_{\nu} + E_{\nu}(\ln a)\big)$$
(7.12)

and, consequently, with respect to changes of the frames in the tangent bundle space over M, when a = 1, they behave like the components of a covariant vector field (one-form). Therefore, on an open set U, e.g., U = M, the quantity

$$\omega = \Gamma_{\mu} E^{\mu}, \tag{7.13}$$

where  $\{E^{\mu}\}$  is the coframe dual to  $\{E_{\mu}\}$  (in local coordinates:  $E_{\mu} = \frac{\partial}{\partial s^{\mu}}$  and  $E^{\mu} = dx^{\mu}$ ), is a 1-form over M (with respect to changes of the local coordinates on M or of the frames in the (co)tangent bundle space over M). However, it depends on the choice of the frame  $\{e\}$  in the bundle space E and a change  $e \mapsto e' = ae$  implies

$$\omega \mapsto \omega' = \omega + (E_{\nu}(\ln a))E^{\nu} = \omega + (E'_{\nu}(\ln a))E'^{\nu}.$$
(7.14)

Using the 1-form (7.13), we see that

$$\Gamma(s;\gamma) = \omega|_{\gamma(s)}(\dot{\gamma}(s)) \tag{7.15}$$

and (7.3) can be rewritten as

$$\boldsymbol{L}(t,s;\gamma) = \exp\left(-\int_{\gamma(s)}^{\gamma(t)} \omega\right)$$
(7.16)

where the integration is along some path in U (on which the transport L is Euclidean). Hence L (or L) depends only on the points  $\gamma(s)$  and  $\gamma(t)$ , not on the particular path connecting them, as it should be (Theorem 5.1). The self-consistency of our results is confirmed by the equation

$$R_{\mu\nu}|_U = 0 \tag{7.17}$$

which is a consequence of (7.11) and (6.7) and which is a necessary and sufficient condition for the existence of frames normal on an open set U (Theorem 6.1).

We end this section with the remark that frames normal along injective paths always exist (Corollary 5.1), but on an arbitrary submanifold  $N \subseteq M$  they exist iff the functions  $\Gamma_{\mu}$  satisfy the conditions (6.16) with  $x \in N$  in the coordinates described in Theorem 6.2.

# 8. Normal frames for derivations in vector bundles with a manifold as a base

For a general bundle  $(E, \pi, B)$  whose bundle space E is  $C^1$  manifold, we call a frame  $\{e_i\}$  normal on  $U \subseteq B$  (resp. along  $\gamma: J \to M$ ) for a derivation D along paths (resp.  $D^{\gamma}$  along  $\gamma$ ) (see Definition 2.1 on page 219) if  $\{e_i\}$  is normal on U (resp. along  $\gamma$ ) for the linear transport L along paths generating it by (3.19) (see Proposition 3.7). We can also, equivalently, define a frame normal for D(resp.  $D^{\gamma}$ ) as one in which the components of D (resp.  $D^{\gamma}$ ) vanish (see the proof of Proposition 3.7, Proposition 4.2, and Corollary 4.2). A derivation admitting normal frame(s) is called *Euclidean*.

In connection with concrete physical applications, far more interesting case is the case of a bundle  $(E, \pi, M)$  with a differentiable manifold M as a base. The cause for this is the existence of natural structures over M, e.g., the different tensor bundles and the tensor algebra over it. Below we concentrate on this particular case.

**Definition 8.1.** A derivation over an open set  $V \subseteq M$  or in  $(E, \pi, M)|_V$  along tangent vector fields is a mapping  $\mathcal{D}$  assigning to every tangent vector field X over V a K-linear mapping

$$\mathcal{D}_X \colon \operatorname{Sec}^1((E,\pi,M)|_V) \to \operatorname{Sec}^0((E,\pi,M)|_V), \tag{8.1}$$

called a *derivation along* X, such that

$$\mathcal{D}_X(f \cdot \sigma) = X(f) \cdot \sigma + f \cdot \mathcal{D}_X(\sigma) \tag{8.2}$$

for every  $C^1$  section  $\sigma$  over V and every  $C^1$  function  $f: V \to \mathbb{K}$ .

**Example 8.1.** The mapping  $X \mapsto \nabla_X|_{\mathfrak{T}^{r;1}_s(M)}$ , where  $\nabla$  is a linear connection and  $\mathfrak{T}^{r;1}_s(M) = \operatorname{Sec}^1(T^r_s(M), \pi^r_s, M)$  is the set of  $C^1$  tensor fields of type (r, s) over M, is a derivation along tangent vector fields in the tensor bundle  $(T^r_s(M), \pi^r_s, M)$  of type (r, s).

Obviously (see Definition 2.1), if  $\gamma: J \to V$  is a  $C^1$  path, the mapping  $\overline{D}: \hat{\sigma} \mapsto \overline{D}\hat{\sigma}$ , with  $\overline{D}\hat{\sigma}: \gamma \mapsto \overline{D}^{\gamma}\hat{\sigma}$ , where  $\overline{D}^{\gamma}\hat{\sigma}: s \mapsto \overline{D}_s^{\gamma}\hat{\sigma}$  is defined via

$$\overline{D}_{s}^{\gamma}(\hat{\sigma}) = \left( (\mathcal{D}_{X}\sigma)|_{X=\dot{\gamma}} \right) (\gamma(s)), \qquad \hat{\sigma} \colon \gamma \mapsto \sigma \circ \gamma, \tag{8.3}$$

is a derivation along paths on the set of  $C^1$  liftings generated by sections of  $(E, \pi, M)|_V$ . From Section 2, we know that along paths without self-intersections every derivation along paths generates a derivation of the sections of  $(E, \pi, M)$  (see (2.11) and (2.12)). Thus to any derivation  $\mathcal{D}$  along (tangent) vector fields on V there corresponds, via (8.3), a natural derivation D along the paths in V on the set of liftings generated by sections. These facts are a hint for the possibility to introduce 'normal' frames for  $\mathcal{D}$ . This can be done as follows.

Let  $\{e_i\}$  be a  $C^1$  frame in  $\pi^{-1}(V)$ . We define the *components* or (2-index) coefficients  $\Gamma_X^{i}_{i}: V \to \mathbb{K}$  of  $\mathcal{D}_X$  by the expansion (cf. (2.3))

$$\mathcal{D}_X e_i = \Gamma_X{}^j{}_i e_j. \tag{8.4}$$

So  $\Gamma_X := [\Gamma_X{}^j{}_i]$  is the *matrix* of  $\mathcal{D}_X$  in  $\{e_i\}$ .

Applying (8.2) to  $\sigma = \sigma^i e_i$  and using the linearity of  $\mathcal{D}_X$ , we get the explicit expression (cf. (3.23))

$$\mathcal{D}_X(\sigma) = \left(X(\sigma^i) + \Gamma_X{}^i{}_j \,\sigma^j\right) e_i. \tag{8.5}$$

A simple verification proves that the change  $\{e_i\} \mapsto \{e'_i = A^j_i e_j\}$ , with a non-degenerate  $C^1$  matrix-valued function  $A = [A^j_i]$ , leads to (cf. (3.26))

$$\Gamma_X := \left[\Gamma_X{}^i{}_j\right] \mapsto \Gamma'_X := \left[\Gamma'_X{}^i{}_j\right] = A^{-1}\Gamma_X A + A^{-1}X(A), \tag{8.6}$$

where  $X(A) := [X(A_i^j)]$  and the frames  $\{e_i\}$  and  $\{e'_i\}$  (as well as A) are supposed to be defined on an open subset of M containing or equal to U (see Note 6.1). Conversely, if a geometrical object with components  $\Gamma_X{}^i{}_j$  is given in a frame  $\{e_i\}$ and a change  $\{e_i\} \mapsto \{e'_i = A_i^j e_j\}$  implies the transformation (8.6), then there exists a unique derivation along X, defined via (8.5), whose components in  $\{e_i\}$ are exactly  $\Gamma_X{}^i{}_j$  (cf. Proposition 3.6).

Below, for the sake of simplicity, we take V = M, i.e., the derivations are over the whole base M.

**Definition 8.2.** A frame  $\{e_i\}$ , defined on an open subset of M containing or equal to a set U, is called normal for a derivation  $\mathcal{D}$  along tangent vector fields (resp. for  $\mathcal{D}_X$  along a given tangent vector field X) on U if in  $\{e_i\}$  the components of  $\mathcal{D}$  (resp.  $\mathcal{D}_X$ ) vanish on U for every (resp. the given) tangent vector field X.

If  $\mathcal{D}$  (resp.  $\mathcal{D}_X$ ) admits frames normal on  $U \subseteq M$ , we call it *Euclidean* on U. A number of results, analogous to those of Sections 4–6, can be proved for such derivations. Here we shall mention only a few of them.

**Proposition 8.1** (cf. Theorem 5.2). A derivation  $\mathcal{D}$  along tangent vector fields admits frame(s) normal on  $U \subseteq M$  iff in every frame its matrix on U has the form

$$\Gamma_X|_U = \left(F^{-1}X(F)\right)|_U \tag{8.7}$$

where F is a  $C^1$  non-degenerate matrix-valued function defined on an open set containing U.

*Proof.* If  $\{e'_i\}$  is normal on U for  $\mathcal{D}$ , then (8.7) with  $F = A^{-1}$  follows from (8.6) with  $\Gamma'_X|_U = 0$ . Conversely, if (8.7) holds, then (8.6) with  $A = F^{-1}$  yields  $\Gamma'_X|_U = 0$ .

**Proposition 8.2** (cf. Corollary 4.5). The frames normal on  $U \subseteq M$  for a Euclidean derivation along vector fields (resp. given vector field X) are connected by linear transformations whose matrices A are constant (resp. X(A) = 0) on U.

*Proof.* The result is a consequence of (8.6) for  $\Gamma_X = \Gamma'_X = 0$ .

**Definition 8.3.** A derivation  $\mathcal{D}$  along (tangent) vector fields is called linear on U if in some (and hence in any) frame its components admit the representation

$$\Gamma_X{}^i{}_j(p) = \Gamma^i{}_{j\mu}(p)X^\mu(p) \qquad \text{or} \quad \Gamma_X = \Gamma_\mu X^\mu \tag{8.8}$$

where  $p \in U$ ,  $\Gamma_{\mu} = [\Gamma^{i}_{j\mu}(p)]_{i,j=1}^{\dim \pi^{-1}(p)}$  are matrix-valued functions on (a neighborhood of) U, and  $X^{\mu}$  are the local components of a vector field X in some frame  $\{E_{\mu}\}$  in the bundle space T(M) of the tangent bundle,  $X = X^{\mu}E_{\mu}$ .

Remark 8.1. The invariant definition of a derivation linear on U is via the equation

$$\mathcal{D}_{fX+gY} = f\mathcal{D}_X + g\mathcal{D}_Y \tag{8.9}$$

where  $f, g: U \to \mathbb{K}$  and X and Y are tangent vector fields over U. But for the purposes of this work the above definition is more suitable. Comparing Definitions 8.1 and 8.3 (see also (8.9)) with [23, p. 74, Definition 2.51] or with Definition 14.7 on page 305, we see that a derivation along tangent vector fields is linear iff it is a covariant derivative operator (covariant derivative) in  $(E, \pi, B)|_U$ . Therefore the concepts linear derivation along tangent vector fields and covariant derivative operator (covariant derivative) coincide.

We call  $\Gamma^i_{j\mu}$  3-index coefficients of  $\mathcal{D}$  or simply coefficients if there is no risk of misunderstanding. It is trivial to check that under changes of the frames they transform according to (6.5). It is easy to verify that to every linear derivation  $\mathcal{D}$ there corresponds a unique derivation along paths or linear transport along paths whose 2-index coefficients are given via (6.1) with  $\Gamma_{\mu} := [\Gamma^i_{j\mu}]$  being the matrices of the 3-index coefficients of  $\mathcal{D}$ .<sup>1</sup> Conversely, to any such transport or derivation along paths there corresponds a unique linear derivation along tangent vector fields with components ((2-index) coefficients) given by (8.8), i.e., with the same 3-index coefficients. So, there is a bijective correspondence between the sets of linear derivations along tangent vector fields and derivations (or linear transports) along paths whose (2-index) coefficients admit the representation (6.1). It should be emphasized, if the above discussion is restricted to a subset U, i.e., only for paths lying entirely in U, it remains valid iff U is an open set in M. Otherwise, if U is not an open set, the correspondence between derivations along tangent vector fields and derivations or linear transports along paths via their 3-index coefficients is surjective in

<sup>&</sup>lt;sup>1</sup>One can verify that the action of the derivation along paths induced by  $\mathcal{D}$  on the liftings generated by sections is given by (8.3).

the right direction.<sup>2</sup> In the opposite direction it is injective, if the 3-index coefficients of the derivations along paths are fixed, or/and to a single derivation along paths may correspond different derivations along tangent vector fields, if the arbitrariness of the 3-index coefficients of the former derivations is taken into account. This remark is important when the normal frames in the both cases are compared.

**Proposition 8.3.** A derivation along tangent vector fields is Euclidean on U iff it is linear on U and, in every frame  $\{e_i\}$  over U in the bundle space and every frame  $\{E_{\mu}\}$  over U in the tangent bundle space over U, the matrices  $\Gamma_{\mu}$  of its 2-index coefficients have the form (6.3) for some non-degenerate  $C^1$  matrix-valued function  $\mathbf{F}$  on U.

*Proof.* The result is a corollary from Proposition 8.1 as  $X = X^{\mu}E_{\mu}$  and (8.7) imply (8.8) with  $\Gamma_{\mu} := [\Gamma^{i}{}_{j\mu}] = F^{-1}E_{\mu}(F)$ .

**Theorem 8.1** (cf. Theorem 6.1). Frames normal on a neighborhood U for a derivation  $\mathcal{D}$  along vector fields exist iff it is linear on U and its 3-index coefficients satisfy the conditions (6.6) on U.

*Proof.* By Proposition 8.3, a derivation  $\mathcal{D}$  along vector fields is Euclidean iff (6.3) holds for some  $\mathbf{F}$  which, according to Lemma 6.1, is equivalent to (6.6).

**Proposition 8.4** (cf. Proposition 6.2). A frame is normal on  $U \subseteq M$  for some linear derivation along tangent vector fields iff the derivation's 3-index coefficients vanish on U.

*Proof.* This result is a corollary of Definition 8.2, equation (8.8) and the arbitrariness of X in it.  $\Box$ 

Remark 8.2. The arbitrariness of U in this Proposition does not contradict Remark 6.4 as the restriction of a derivation along tangent vector fields to (a neighborhood of) U is along vector fields X on U tangent to M, i.e.,  $X_p \in T_p(M)$  with  $p \in U$ .

In this way we have proved the existence of a bijective mappings between the sets of Euclidean derivations along paths and Euclidean linear transports along paths. It is given via the (local) coincidence of their 3-index coefficients in some (local) frames. Moreover, the normal frames for the corresponding objects of these sets coincide. What concerns the frames normal for Euclidean derivations along tangent vector fields, in them, by Proposition 8.4, vanish not only their 2-index coefficients, but also the 3-index ones. Hence the set of these frames is, generally, a subset of the one of frames normal for derivations or linear transports along paths. More details on frames in which the 3-index coefficients of a derivations or transport vanish will be presented in Section 12. These observations, combined with Remark 8.1, are quite important because they allow *in extenso* transferring

<sup>&</sup>lt;sup>2</sup>If  $\Gamma_{\mu}$  and  $\overline{\Gamma}_{\mu}$  are the matrices of the 3-index coefficients of  $\mathcal{D}$  and  $\overline{\mathcal{D}}$ , then they define a single derivation along paths in U iff  $(\overline{\Gamma}_{\mu} - \Gamma_{\mu})\dot{\gamma}^{\mu} = 0$  for all  $C^1$  paths  $\gamma: J \to U$ .

of the results obtained for linear transports and derivations (along paths or along tangent vector fields) to the theory of connections, parallel transports, and covariant derivatives in general vector bundles. We shall return to this range of problems in Section 14.

# 9. Curvature and normal frames

Until now the concept of a 'curvature', in connection with normal frames, appeared several times in similar contexts. For the first time, we met it in Section II.4: the vanishment of the curvature (I.3.11) of a linear connection is a necessary and sufficient condition for the existence of frames normal on a neighborhood for the linear connection on it (Theorem II.4.1 on page 104). Next, the curvature of a linear connection appeared in Theorem II.5.2, point (i), on page 120 for analogous result on submanifolds. Similar results for arbitrary derivations along vector fields on a manifold are expressed by Theorems III.6.1, III.7.1, III.8.1, III.8.3, and III.8.4 in which the curvature (III.2.12) of the derivation plays an essential role. One would probably ask for the origin of the condition for ('partial') vanishment of the curvature in all of these results. From algebraic view-point, this condition is nothing else but the integrability condition for a system of partial differential equations of first order, the rigorous result being expressed by Lemma II.4.1 on page 105 (see, in particular, equation (II.4.6) in it).<sup>1</sup> Alternatively, from geometrical point of view, this means the path-independence of the parallel transport, generated by a (suitable) linear connection, along paths in some set (see Lemma II.5.1 on page 112 and equations (II.5.1') on page 113 and (II.5.1'') on page 114).

All of the above-mentioned results have natural analogues in the theory of frames normal for linear transports along paths (or derivations along paths or along tangent vector fields). Practically they were derived, in algebraic terms, in Sections 5 and 6. To the reformulation of a part of the corresponding assertions, employing the concept of a 'curvature', is devoted the present section.

## 9.1. Curvature of linear transport or derivation along paths

Since there is a bijective correspondence between linear transports along paths and derivations along paths (see Section 3), two equivalent ways for the introduction of curvature can be pointed out. The algebraic approach is based on the consideration of a 'commutator' of two derivations along two different paths passing through one and the same point. Alternatively, in the geometrical approach, a vector is transported along a closed path and the result of this transportation is investigated when the path is contracted (tends) to a single point. Below we shall realize the former approach as it is somewhat more direct and concise.

 $<sup>^1{\</sup>rm From}$  here the adjective 'integrable' as a synonym for 'flat' in connection with vanishing curvature comes from.

The direct transferring of the concept of a 'curvature' of a derivation along vector fields (of the tensor algebra over a manifold), introduced via (III.2.12) (see also (I.3.11) in the particular case of linear connections), to derivations along paths is impossible as, due to (2.1c), the derivative along a path of a lifting is not a lifting. For the purpose is required a kind of derivation along paths which preserves the type of the objects on which it acts. Without going into details, we notice that the only derivation of this kind, introduce until now and having the property mentioned, is the section-derivation along paths generated by a derivation along paths and defined via (2.11) and (2.12). Alternatively, the 'commutator' of two derivations along paths should be suitable defined. Below we briefly sketch these approaches to the concept of a curvature of linear transports along paths.

Let D be a  $C^1$  derivation along paths in a vector bundle  $(E, \pi, B)$ , E being  $C^1$  manifold, and D be the section-derivation generated by it (see page 221). Let  $\eta: J \times J' \to B$  be an *injective* mapping, i.e., for every  $(s,t) \in J \times J'$  there does not exist  $(s',t') \in J \times J'$  such that  $(s',t') \neq (s,t)$  and  $\eta(s',t') = \eta(s,t)$ .

The (section-)curvature (operator) along  $\eta$  at  $(s,t) \in J \times J'$  of a  $C^1$  sectionderivation D along paths, generated by a  $C^1$  derivation D along paths, is a mapping

$$\mathbf{R}^{\eta}(s,t): \operatorname{Sec}^{2}((E,\pi,B)|_{\eta(J,J')}) \to \pi^{-1}(\eta(s,t))$$
(9.1)

defined as follows. For every section  $\sigma \in \text{Sec}^2((E, \pi, B)|_{\eta(J,J')})$ , define sections  $\sigma_1, \sigma_2 \in \text{Sec}^1((E, \pi, B)|_{\eta(J,J')})$  such that, for every  $(s, t) \in J \times J'$ ,

$$\begin{split} \sigma_1(\eta(s,t)) &:= \left(\mathsf{D}^{\eta(s,\,\cdot\,)}(\sigma|_{\eta(s,J')})\right)(\eta(s,t)) = D_t^{\eta(s,\,\cdot\,)}(\hat{\sigma}) \in \pi^{-1}(\eta(s,t))\\ \sigma_2(\eta(s,t)) &:= \left(\mathsf{D}^{\eta(\,\cdot\,,t)}(\sigma|_{\eta(J,t)})\right)(\eta(s,t)) = D_s^{\eta(\,\cdot\,,t)}(\hat{\sigma}) \in \pi^{-1}(\eta(s,t)) \end{split}$$

where the lifting  $\hat{\sigma} \in \text{PLift}(E, \pi, B)$  is such that  $\hat{\sigma} : \gamma \mapsto \hat{\sigma}_{\gamma} := \sigma \circ \gamma$  for every path  $\gamma$  in  $\eta(J, J')$ .<sup>2</sup>

By definition, the action of the mapping (9.1) on  $\sigma$  is

$$(\mathsf{R}^{\eta}(s,t))(\sigma) := (\mathsf{D}^{\eta(\cdot,t)}(\sigma_1|_{\eta(J,t)}))(\eta(s,t)) - (\mathsf{D}^{\eta(s,\cdot)}(\sigma_2|_{\eta(s,J')}))(\eta(s,t)) = D_s^{\eta(\cdot,t)}(\hat{\sigma}_1) - D_t^{\eta(s,\cdot)}(\hat{\sigma}_2)$$
(9.2)

where  $\hat{\sigma}_1, \hat{\sigma}_2 \in \text{PLift}(E, \pi, B)|_{\eta(J, J')}$  and  $\hat{\sigma}_a \colon \gamma \mapsto (\hat{\sigma}_a)_{\gamma} \coloneqq \sigma_a \circ \gamma, a = 1, 2$  for every path  $\gamma$  in  $\eta(J, J')$ .

Symbolically, by abuse of the notation, one may write

$$\mathsf{R}^{\eta}(s,t) = \mathsf{D}^{\eta(\,\cdot\,,t)} \circ \mathsf{D}^{\eta(s,\,\cdot\,)} - \mathsf{D}^{\eta(s,\,\cdot\,)} \circ \mathsf{D}^{\eta(\,\cdot\,,t)}$$

but, as a consequence of (2.11), compositions like  $\mathsf{D}^{\eta(\cdot,t)} \circ \mathsf{D}^{\eta(s,\cdot)}$  are not quite correctly defined since the action of this expression on  $\sigma$  must be defined as  $\mathsf{D}^{\eta(\cdot,t)}(\sigma_1|_{\eta(J,t)})$ .

<sup>&</sup>lt;sup>2</sup>The index 1 (resp. 2) in  $\sigma_1$  (resp.  $\sigma_1$ ) reflects that in the right-hand side of its definition the first (resp. second) argument, i.e., s (resp. t), is fixed, i.e., considered as a parameter.

### 9. Curvature and normal frames

**Exercise 9.1.** Suppose  $\{e_i(p)\}, p \in \eta(J, J')$ , is a basis in the fibre  $\pi^{-1}(p)$ . In the frame  $\{\hat{e}_i\}$ , with  $\hat{e}_i: \gamma \mapsto (\hat{e}_i)_{\gamma}: r \mapsto e_i(r; \gamma) = e_i(\gamma(r)), r \in J''$ , for a path  $\gamma: J'' \to \eta(J, J')$ , the expansion (2.4) with s = r holds for every lifting  $\lambda \in \text{PLift}^1(E, \pi, B)|_{\eta(J,J')}$ . Applying this formula, find explicit expressions for the derivatives  $D_t^{\eta(s, \cdot)}\hat{\sigma}, D_s^{\eta(\cdot, t)}\hat{\sigma}, D_t^{\eta(s, \cdot)}\hat{\sigma}_2$ , and  $D_s^{\eta(\cdot, t)}\hat{\sigma}_1$ , entering in (9.2). Substituting the so-obtained expressions into (9.2), prove that the (section-)curvature operator (9.1) is *linear*,

$$(\mathsf{R}^{\eta}(s,t))(\sigma) = \left((\mathsf{R}^{\eta}(s,t))\right)^{i}{}_{j} \sigma^{j}(\eta(s,t))e_{i}(\eta(s,t)), \tag{9.3}$$

and that its components in a frame  $\{e_i\}$  are

$$\left( \left( \mathsf{R}^{\eta}(s,t) \right) \right)_{j}^{i} = \frac{\partial \Gamma_{j}^{i}(t;\eta(s,\cdot))}{\partial s} - \frac{\partial \Gamma_{j}^{i}(s;\eta(\cdot,t))}{\partial t} + \Gamma_{k}^{i}(s;\eta(\cdot,t))\Gamma_{j}^{k}(t;\eta(s,\cdot)) - \Gamma_{k}^{i}(t;\eta(s,\cdot))\Gamma_{j}^{k}(s;\eta(\cdot,t))$$
(9.4)

where  $\Gamma_{j}^{i}(\cdots)$  are the components of the section-derivation D, or of the derivation D generating it, in the frame  $\{\hat{e}_i\}$  (see (2.3)).

Having in mind the above introduction of the curvature of a section-derivation, now we shall define the curvature (operator) of arbitrary derivation along paths.

Let D be a  $C^1$  derivation along paths in a vector bundle  $(E, \pi, B)$ , E being  $C^1$  manifold, and  $\eta: J \times J' \to B$  be *arbitrary*, injective or not, mapping. Let  $\bar{\eta}: J \times J' \to E$  be of class  $C^2$  and a lifting of  $\eta, \pi \circ \bar{\eta} := \eta$ .

The curvature (operator) along  $\eta$  at  $(s,t)\in J\times J'$  of the derivation D is a mapping

$$R^{\eta}(s,t) \colon \bar{\eta} \mapsto (R^{\eta}(s,t))(\bar{\eta}) \in \pi^{-1}(\eta(s,t))$$

$$(9.5)$$

defined in the following way.

Let  $\hat{\eta}_1, \hat{\eta}_2 \in \text{PLift}^2(E, \pi, B)$  and  $\check{\eta}_1, \check{\eta}_2: J \times J' \to E$  be such that

$$\hat{\eta}_1(\eta(s,\,\cdot\,)) := \bar{\eta}(s,\,\cdot\,), \quad \check{\eta}_1(s,t) := \left(D^{\eta(s,\,\cdot\,)}(\hat{\eta}_1)\right)(t) \\ \hat{\eta}_2(\eta(\,\cdot\,,t)) := \bar{\eta}(\,\cdot\,,t), \quad \check{\eta}_2(s,t) := \left(D^{\eta(\,\cdot\,,t)}(\hat{\eta}_2)\right)(s).$$

By definition, the action of the curvature  $R^{\eta}(s,t)$  (along  $\eta$  at (s,t) of D) on  $\bar{\eta}$  is

$$(R^{\eta}(s,t))(\bar{\eta}) := \left(D^{\eta(\cdot,t)}(\hat{\eta}_2)\right)(s) - \left(D^{\eta(s,\cdot)}(\hat{\eta}_1)\right)(t)$$
(9.6)

where the liftings  $\hat{\hat{\eta}}_1, \hat{\hat{\eta}}_2 \in \text{PLift}^1(E, \pi, B)$  are such that<sup>3</sup>

$$\hat{\eta}_1(\eta(s,\,\cdot\,)) := \check{\eta}_2(s,\,\cdot\,), \quad \hat{\eta}_2(\eta(\,\cdot\,,t)) := \check{\eta}_1(\,\cdot\,,t).$$

<sup>&</sup>lt;sup>3</sup>Notice the positions of the subscripts 1 and 2 in the next definitions.

The explicit 'local' expression for the curvature  $R^{\eta}(s,t)$  can be found by means of the expansion (2.4). Let  $\{\hat{e}_i(p)\}, p \in \eta(J, J')$ , be a basis in  $\pi^{-1}(p)$ . In the frame  $\{e_i\}$ , with  $\hat{e}_i \colon \gamma \mapsto (\hat{e}_i)_{\gamma} \colon r \mapsto e_i(r; \gamma) = e_i(\gamma(r))$  for a path  $\gamma \colon J'' \to \eta(J, J')$ and  $r \in J''$ , we have (see (2.9))

$$D^{\eta(s,\,\cdot\,)}(\hat{\eta}_1) = \left[\dot{\eta}_1^i(\eta(s,\,\cdot\,)) + \Gamma^i{}_j(\,\cdot\,;\eta(s,\,\cdot\,))\hat{\eta}_1^j(s,\,\cdot\,)\right] e_i(\,\cdot\,;\eta(s,\,\cdot\,))$$
  
$$D^{\eta(\,\cdot\,,t)}(\hat{\eta}_2) = \left[\dot{\eta}_2^i(\eta(\,\cdot\,,t)) + \Gamma^i{}_j(\,\cdot\,;\eta(\,\cdot\,,t))\hat{\eta}_2^j(\,\cdot\,,t)\right] e_i(\,\cdot\,;\eta(\,\cdot\,,t))$$

where  $\Gamma^{i}_{j}(s;\gamma)$  are the components of D in  $\{\hat{e}_{i}\}$  and (see (2.6))

$$\dot{\eta}_1^i(\eta(s,\,\cdot\,))\colon t\mapsto \frac{\partial \left(\hat{\eta}_1^i(\eta(s,\,\cdot\,))(t)\right)}{\partial t} = \frac{\partial \bar{\eta}^i(s,t)}{\partial t}, \quad \dot{\eta}_2^i(\eta(\,\cdot\,,t))\colon s\mapsto \frac{\partial \bar{\eta}^i(s,t)}{\partial s}.$$

Using these results, (2.4), and the definitions of  $\check{\eta}_1$  and  $\check{\eta}_2$ , we find:

$$\check{\eta}_1(s,t) = \left[\frac{\partial \bar{\eta}^i(s,t)}{\partial t} + \Gamma^i{}_j(t;\eta(s,\,\cdot\,))\bar{\eta}^j(s,t)\right] e_i(t;\eta(s,\,\cdot\,))$$
$$\check{\eta}_2(s,t) = \left[\frac{\partial \bar{\eta}^i(s,t)}{\partial s} + \Gamma^i{}_j(s;\eta(\,\cdot\,,t))\bar{\eta}^j(s,t)\right] e_i(s;\eta(\,\cdot\,,t)).$$

Finally, calculating the derivatives  $(D^{\eta(\cdot,t)}(\hat{\eta}_2))(s)$  and  $(D^{\eta(s,\cdot)}(\hat{\eta}_1)(t))$  by means of equation (2.4), using the relations (see (2.6))

$$\dot{\hat{\eta}}_1^i(\eta(s,\,\cdot\,))\colon t\mapsto \frac{\partial \left(\hat{\hat{\eta}}_1^i(\eta(s,\,\cdot\,))(t)\right)}{\partial t} = \frac{\partial \check{\eta}_2^i(s,t)}{\partial t}, \quad \dot{\hat{\eta}}_2^i(\eta(\,\cdot\,,t))\colon s\mapsto \frac{\partial \check{\eta}_1^i(s,t)}{\partial s},$$

and the formulae just-obtained, from (9.6), we get

$$(R^{\eta}(s,t))(\bar{\eta}) = \left(R^{\eta}(s,t)\right)^{i}{}_{j} \bar{\eta}^{j} e_{i}(\eta(s,t)),$$
(9.7)

where  $e_i(s; \eta(\cdot, t)) = e_i(t; \eta(s, \cdot)) = e_i(\eta(s, t))$  was taken into account, and

$$\left( R^{\eta}(s,t) \right)_{j}^{i} = \frac{\partial \Gamma_{j}^{i}(t;\eta(s,\cdot))}{\partial s} - \frac{\partial \Gamma_{j}^{i}(s;\eta(\cdot,t))}{\partial t} + \Gamma_{k}^{i}(s;\eta(\cdot,t))\Gamma_{j}^{k}(t;\eta(s,\cdot)) - \Gamma_{k}^{i}(t;\eta(s,\cdot))\Gamma_{j}^{k}(s;\eta(\cdot,t))$$
(9.8)

which in a matrix form reads

$$\boldsymbol{R}^{\eta}(s,t) := \left[ \left( R^{\eta}(s,t) \right)_{j}^{i} \right] = \frac{\partial \boldsymbol{\Gamma}(t;\eta(s,\cdot))}{\partial s} - \frac{\partial \boldsymbol{\Gamma}(s;\eta(\cdot,t))}{\partial t} + \boldsymbol{\Gamma}(s;\eta(\cdot,t))\boldsymbol{\Gamma}(t;\eta(s,\cdot)) - \boldsymbol{\Gamma}(t;\eta(s,\cdot))\boldsymbol{\Gamma}(s;\eta(\cdot,t)). \quad (9.9)$$

Consequently the curvature  $R^{\eta}(s,t)$  of a derivation D along paths is a *linear* operator whose *matrix elements* (components) are (9.8) in  $\{\hat{e}_i\}$  and which coincide,

for injective  $\eta$ , with the ones of the section-derivation D along paths generated by D (see (9.4)),

$$\left(R^{\eta}(s,t)\right)^{i}_{\ j} = \left(\mathsf{R}^{\eta}(s,t)\right)^{i}_{\ j}.$$
(9.10)

This equality is not accidental: if  $\eta$  is injective and one puts  $\bar{\eta} = \sigma \circ \eta$  for a  $C^1$  section  $\sigma$  over  $\eta(J, J')$ , then, making the corresponding changes, one can transform (9.7) into (9.3) with  $\left(\mathsf{R}^{\eta}(s,t)\right)_{i}^{i} = \left(R^{\eta}(s,t)\right)_{i}^{i}$ .

**Definition 9.1.** The curvature (operator) along  $\eta: J \times J' \to B$  at  $(s,t) \in J \times J'$  of a  $C^1$  linear transport along paths is called the curvature (operator) along  $\eta: J \times J' \to B$  at  $(s,t) \in J \times J'$  of the generated by it via (3.19) derivation along paths.

**Definition 9.2.** The curvature (operator) of linear transports or derivations along paths is a mapping R assigning to each transport or derivation and every  $\eta: J \times J' \to B$  and  $(s,t) \in J \times J'$  the corresponding curvature  $R^{\eta}(s,t)$  along  $\eta$  at (s,t).

The following result is important in connection with the theory of normal frames (see Proposition 6.1 on page 247), as well as for making a link with the concept of a curvature of the linear connections.

**Proposition 9.1.** Let  $(E, \pi, M)$  be a vector bundle whose bundle space E and base M are  $C^1$  manifolds,  $U \subseteq M$ , and the matrix of the 2-index coefficients of a linear transport L or derivations D along paths has on U the form

$$\Gamma(s;\gamma) = \sum_{\mu=1}^{\dim M} \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s)$$
(9.11)

in every frame  $\{e_i\}$  on U, where  $\Gamma_{\mu} = [\Gamma^i_{\ j\mu}]$  are some matrix-valued functions on an open set containing U or equal to it,  $\gamma: J'' \to U$  is of class  $C^1$ ,  $s \in J''$ , and  $\dot{\gamma}^{\mu}$  are the components of the tangent vector field  $\dot{\gamma}$  in some frame  $\{E_{\mu}\}$  over U in the tangent bundle space over M,  $\dot{\gamma} = \dot{\gamma}^{\mu}E_{\mu}$ . Then, for every  $C^1$  mapping  $\eta: J \times J' \to U$ , the components of the curvature of L or D in  $\{e_i\}$  along  $\eta$  at  $(s,t) \in J \times J'$  are

$$\left(R^{\eta}(s,t)\right)^{i}_{\ j} = R^{i}_{\ j\alpha\beta}(\eta(s,t)) \left(\eta'(s,t)\right)^{\alpha} \left(\eta''(s,t)\right)^{\beta},\tag{9.12}$$

where

$$R^{i}_{j\alpha\beta} := -\Gamma^{i}_{j\alpha,\beta} + \Gamma^{i}_{j\beta,\alpha} - \Gamma^{k}_{j\alpha}\Gamma^{i}_{k\beta} + \Gamma^{k}_{j\beta}\Gamma^{i}_{k\alpha} - \Gamma^{i}_{j\mu}C^{\mu}_{\alpha\beta} \colon U \to \mathbb{K}$$
(9.13)

with  $[E_{\alpha}, E_{\beta}]_{-} := C^{\mu}_{\alpha\beta}E_{\mu}$  and  $f_{,\mu} := E_{\mu}(f)$  for a  $C^{1}$  function f on U. Here  $\eta'(\cdot, t)$  and  $\eta''(s, \cdot)$  denote the tangent vectors to the paths  $\eta(\cdot, t)$  and  $\eta(s, \cdot)$ , respectively.

*Proof.* One should substituted (9.11) for the corresponding paths into (9.8), then to use the equalities  $\frac{\partial}{\partial s}|_{\eta(s,t)} = (\eta'(s,t))^{\alpha} E_{\alpha}|_{\eta(s,t)} = \eta'(s,t)$  and  $\frac{\partial}{\partial t}|_{\eta(s,t)} = (\eta''(s,t))^{\beta} E_{\beta}|_{\eta(s,t)} = \eta''(s,t)$ , and, at last, to apply (I.8.1).<sup>4</sup>

For future purposes, we shall rewrite (9.13) in a matrix form:

$$\boldsymbol{R}_{\alpha\beta} := \left[ R^{i}_{\ j\alpha\beta} \right]^{\dim \pi^{-1}(p)}_{i,j=1} = -E_{\beta}(\Gamma_{\alpha}) + E_{\alpha}(\Gamma_{\beta}) - \Gamma_{\beta}\Gamma_{\alpha} + \Gamma_{\alpha}\Gamma_{\beta} - C^{\mu}_{\alpha\beta}\Gamma_{\mu}$$
(9.14)

for some point p in M (recall,  $\dim \pi^{-1}(p)$  is simply the (fibre) dimension of the bundle).

We shall call the functions  $R^i_{j\alpha\beta}$ , given via (9.13), 4-index components of the curvature with respect to the pair of frames ( $\{e_i\}, \{E_\mu\}$ ). In this connection, the matrix elements  $(R^{\eta}(s,t)^i_{\ j})$  are natural to be called 2-index components of the curvature with respect to the frame  $\{e_i\}$ . Evidently, the curvature of every linear transport has 2-index components, while only the ones of transports with coefficients' matrix (9.11) have 4-index components. A straightforward calculation shows that a change  $\{e_i\} \mapsto \{e' = A^i_i e_j\}$  implies

$$\boldsymbol{R}^{\eta}(s,t) \mapsto \boldsymbol{R}^{\prime \, \eta}(s,t) = A^{-1}(\eta(s,t)\boldsymbol{R}^{\eta}(s,t)A(\eta(s,t),$$
(9.15)

while its combination with  $\{E_{\alpha}\} \mapsto \{E'_{\alpha} = B^{\beta}_{\alpha}E_{\beta}\}$  leads to

$$\boldsymbol{R}_{\alpha\beta} \mapsto \boldsymbol{R}_{\alpha\beta}' = B^{\mu}_{\alpha} B^{\nu}_{\beta} A^{-1} \boldsymbol{R}_{\mu\nu} A.$$
(9.16)

So, the 4-index components of the curvature of a transport/derivation for which (9.11) holds, of which kind are the Euclidean transports or derivations, are independent of the particular mapping  $\eta$  by means of which they are defined, while the 2-index components depend on the point at which they are calculated and on the tangent vectors to  $\eta$  at it.

**Proposition 9.2.** Let  $(E, \pi, M)$ , E and M being  $C^1$  manifolds, be a vector bundle and L be a linear transport along paths in it for which (9.11) holds for every path  $\gamma: J \to U, U \subseteq M$ . If dim M = 0, 1, then the curvature of L identically vanishes on every  $U \subseteq M$ .

*Proof.* See (9.13) for  $\alpha = \beta = 1$  (if dim M = 1) or remove the subscripts  $\alpha$  and  $\beta$  (and put  $C^{\mu} = 0$ ) in it (if dim M = 0).

**Example 9.1.** A simple look over equations (9.13) on the preceding page and (I.3.13) on page 26 reveals a striking similarity between them. Of course, this is not accidental: if we take the tangent bundle  $(T(M), \pi, M)$  for  $(E, \pi, M)$  and the covariant derivative along paths induced by some linear connection  $\nabla$  (see

<sup>&</sup>lt;sup>4</sup>The computation is somewhat simpler if one proves (9.13) for a coordinate frame  $\{E_{\alpha} = \frac{\partial}{\partial x^{\alpha}}\}$  and then transforms the result, by means of (6.5) with A = 1, to an arbitrary frame  $E_{\alpha} = B_{\alpha}^{\beta} \frac{\partial}{\partial x^{\beta}}$  and, at last, applies (I.8.3).

Section III.10) for D, then (9.11) holds for the coefficients' matrices  $\Gamma_{\mu}$  of  $\nabla$  and, consequently, the right hand sides of (9.13) and (I.3.13) coincide as now all indices run from 1 to dim M and  $\Gamma^{i}_{jk}$  are the coefficients of  $\nabla$ . We can express this result by saying that the components of the curvature tensor of a linear connection  $\nabla$  are identical with the (4-index) components of the curvature of the parallel transport or covariant derivative along paths generated by  $\nabla$ .

**Example 9.2.** The result of Example 9.1 can be proved in a basis-free way as follows. Let  $\nabla$  be a linear connection on a manifold M, D be a derivation along paths in  $(T(M), \pi, M)$  with matrix (9.11) in some (every) frame in which  $\Gamma_{\mu}$  are the matrices of the coefficients of  $\nabla$ , and  $A, B \in \mathfrak{X}^2(M)$ . Suppose  $p \in M$  and a  $C^2$  mapping  $\eta: J \times J' \to M$  be defined by the equations  $\eta(s_0, t_0) = p, \eta'(s_0, t_0) = A_p$ , and  $\eta''(s_0, t_0) = B_p$  for a fixed  $(s_0, t_0) \in J \times J'$ . By means of (2.12), one can easily prove that the section-derivation D generated by D is identical with the covariant derivative along paths induced by  $\nabla$ , D:  $\gamma \mapsto \mathsf{D}^{\gamma} = \nabla_{\dot{\gamma}}$  for every injective path  $\gamma$ . Applying this result and (9.2), we find

$$(\mathsf{R}^{\eta}(s_0, t_0))(X) = [(R^{\nabla}(A, B))(X)]|_p, \qquad X \in \mathfrak{X}(M)$$
(9.17)

where  $R^{\nabla}$  is the curvature (operator) of  $\nabla$ , given by the right-hand side of (I.3.11) (see also (III.2.12) for  $D = \nabla$ ).

**Exercise 9.2.** Prove that, if **D** is a *linear* derivation along vector fields of  $T^1(M)$  (see Definition III.2.2 on page 145) and D is a derivation along paths in the tangent bundle  $(T(M), \pi, M)$  for which (9.11) is valid with  $\Gamma_{\mu}$  given via (III.2.9) (with  $k = \mu$ ), then D:  $\gamma \mapsto D\gamma = \mathbf{D}_{\dot{\gamma}}$  and

$$(\mathsf{R}^{\eta}(s_0, t_0))(X) = [(R^{\mathbf{D}}(A, B))(X)]|_p, \qquad X \in \mathfrak{X}(M), \ p = \eta(s_0, t_0), \qquad (9.18)$$

where  $R^{\mathbf{D}}$  is the curvature (operator) of  $\mathbf{D}$  defined via (III.2.12) and whose local components are given by (III.2.14).<sup>5</sup>

**Definition 9.3.** A linear transport or derivation along paths is called *flat*, or *curva*ture free, or *integrable* (resp. on U) if its curvature vanishes on M (resp. on a set  $U \subseteq M$ ), i.e.,  $R^{\eta}(s, t) = 0$  for every  $\eta: J \times J' \to M$  (resp. for every  $\eta: J \times J' \to U$ ).

Remark 9.1. It should be emphasized on a principal difference between the flatness on  $U \subseteq B$  of a linear transport along paths in a bundle  $(E, \pi, B)$  and of a linear connection or derivation along tangent vector fields on a manifold M. In the former case, the curvature is zero on U along mappings  $\eta$  in U, i.e.,  $\eta: J \times J' \to U$ , while in the latter one it vanishes only along vectors tangent to U, i.e., from the tangent space  $T_p(M), p \in U$ . If  $(E, \pi, B) = (T(M), \pi, M)$  and U is a submanifold of M, the first case means that  $\eta'(s, t), \eta''(s, t) \in T_{\eta(s,t)}(U) \subseteq T_{\eta(s,t)}(M)$ , while in the second one, we have  $\eta'(s, t), \eta''(s, t) \in T_{\eta(s,t)}(M)$ , i.e.,  $\eta'$  and  $\eta''$  are not generally in the tangent bundle space over U of  $(T(U), \pi, U)$ .

<sup>&</sup>lt;sup>5</sup>For further generalization of these results, see Proposition 14.15.

## 9.2. On the curvature of Euclidean transports along paths

The main result expressing the relation between curvature of and the frames normal for a linear transport along paths is the following one.

**Theorem 9.1.** Let  $(E, \pi, B)$ , E being  $C^1$  manifold, be a vector bundle, L be a linear transport along paths in it, and  $U \subseteq B$ . The transport L is Euclidean on U if and only if it is flat on U, i.e., frames normal on U for L exist iff the curvature of L along mappings in U is identically zero.

*Remark* 9.2. In some sense, this theorem is the 'peak' or 'summit' of the series of similar Theorems II.5.2, III.7.1, III.8.1, and 6.1 which are its special cases.

*Proof.* NECESSITY. If L is Euclidean on U, by Definition 4.1' or Proposition 4.1, its coefficients vanish in every frame normal on U,  $\Gamma^i{}_j(r;\gamma) \equiv 0$  for every  $\gamma: J'' \to U$  and  $r \in J''$ . Substituting this into equation (9.8) with  $\eta: J \times J'' \to U$ , we get  $R^{\eta}(s,t) \equiv 0$ .

SUFFICIENCY. According to the remark at the beginning of Section 5 (see the paragraph containing equation (5.1) on page 240), we have to prove the existence of solution of the normal frame equation

$$\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} + \mathbf{\Gamma}(s;\gamma)A(\gamma(s)) = 0 \tag{9.19}$$

in a frame  $\{e_i\}$  on U with respect to  $A: U \to \operatorname{GL}(\dim \pi^{-1}(p), \mathbb{K}), p \in B$ , for a flat on U linear transport along every path  $\gamma: J \to U$ .

Let  $V := J \times \cdots \times J$  (dim  $\pi^{-1}(p)$ -times),  $\chi : V \to U, y \in V$ , and the mappings  $\chi_{ij}^y : J \times J \to U$  be defined by  $\chi_{ij}^y(s,t) := \chi(y)|_{y^i = s, y^j = t}$ ,  $s, t \in J$ , for  $j \neq i$  and  $\chi_{ij}^y(s,t) := \chi(y)|_{y^i = s}$  for j = i.

The auxiliary system of partial differential equations

$$\frac{\partial A(\chi(y))}{\partial y^j} = -\Gamma(y^j; \chi^y_{jl}(\cdot, y^l)) A(\chi(y))$$
(9.19\*)

(the concrete value of the index l is insignificant:  $\chi_{jk}^{y}(\cdot, y^{k}) = \chi_{jl}^{y}(\cdot, y^{l})$ ), obtained from (9.19) for  $\gamma = \chi_{jk}^{y}(\cdot, y^{k})$  and  $s = y^{j}$ , always has (non-zero) solutions due to the flatness of the transport L on U. Indeed, the integrability conditions for it [34] are identically satisfied:

$$2\frac{\partial^2}{\partial y^{[j}\partial y^{k]}}A(\chi(y))$$
  
=  $\frac{\partial}{\partial y^j}[-\Gamma(y^k;\chi^y_{kj}(\cdot,y^j))A(\chi(y))] - \frac{\partial}{\partial y^k}[-\Gamma(y^j;\chi^y_{jk}(\cdot,y^k))A(\chi(y))]$ 

### 9. Curvature and normal frames

$$\begin{split} &= \Big[ -\frac{\partial \mathbf{\Gamma}(y^k; \chi^y_{kj}(\cdot, y^j))}{\partial y^j} + \mathbf{\Gamma}(y^k; \chi^y_{kj}(\cdot, y^j)) \mathbf{\Gamma}(y^j; \chi^y_{jk}(\cdot, y^k)) \\ &+ \frac{\partial \mathbf{\Gamma}(y^j; \chi^y_{jk}(\cdot, y^k))}{\partial y^k} - \mathbf{\Gamma}(y^j; \chi^y_{jk}(\cdot, y^k)) \mathbf{\Gamma}(y^k; \chi^y_{kj}(\cdot, y^j)) \Big] A(\chi(y)) \\ &= -R^{\chi^y_{jk}}(y^j, y^k) A(\chi(y)) \equiv 0 \end{split}$$

where we have used the initial system (9.19\*) (for different l) several times,  $\chi^y_{kj}(\cdot, y^j) = \chi^y_{jk}(\cdot, y^k)$ , (9.9) for  $\eta = \chi^y_{jk}$ ,  $s = y^j$ , and  $t = y^k$ , and the flatness of L on U.

Let now A be a solution of the system (9.19\*) for y,  $\chi$ , and  $\chi_{jk}^y$  such that  $y^{i_0} = s \in J$  for some fixed  $i_0$ ,  $\chi(y_0) = \gamma(s)$  for some  $y_0 \in V$  with  $y_0^{i_0} = s$ , and  $\chi_{i_0l}^{y_0}(\cdot, y_0^l) = \gamma$ .<sup>6</sup> Then the restriction of (9.19\*) at  $y_0$  for  $j = i_0$  is identical with (9.19) and, consequently, the frame  $\{e'_i\}$  with  $e'_i|_{\gamma(s)} := A_i^j(\gamma(s))e_j|_{\gamma(s)}$  is normal along  $\gamma: J \to U$ .

Now we have at our disposal the machinery required to look on Theorems 6.1 on page 249 and 6.2 on page 253 from the view-point of transports' curvature. Let  $(E, \pi, M)$  be a vector bundle whose bundle space and base are  $C^1$  manifolds and  $U \subseteq M$ . By Proposition 9.1, if the coefficients' matrix of a linear transport along paths has the form (9.11), the matrices of the 4-index components of the curvature are (9.14) and consequently

$$R_{\mu\nu}(-\Gamma_1,\ldots,-\Gamma_{\dim M}) = \mathbf{R}_{\mu\nu} \tag{9.20}$$

where  $R_{\mu\nu}(\dots)$  are defined by (6.7).<sup>7</sup> From this equality, the conclusion can be made that in Theorem 6.1 the condition (6.6) can, equivalently, be replace by the flatness of the transport on U. Moreover, in this theorem the validity of the representation (6.1) can be dropped: it follows either from the Euclidness of the transport (Proposition 6.1) or from the flatness (as by Theorem 9.1 the flatness implies the Euclidness). This completes the proof of the following assertion which is a stronger version of Theorem 6.1.

**Corollary 9.1.** A linear transport along paths in vector bundle  $(E, \pi, M)$ , E and M being  $C^1$  manifolds, is Euclidean on  $U \subseteq M$  iff it is flat on U or, equivalently, iff the conditions (6.6) hold for  $p \in U$ .

**Corollary 9.2.** Let  $(E, \pi, M)$ , E and M being  $C^1$  manifolds, be a vector bundle and L be a linear transport along paths in it for which (9.11) holds for every path  $\gamma: J \to U, U \subseteq M$ . If dim M = 0, 1, then L is Euclidean on every set  $U \subseteq M$ .

Proof. See Proposition 9.2 and Theorem 9.1.

 $\square$ 

<sup>&</sup>lt;sup>6</sup>Such a choice of  $\chi_{i_0l}^{y_0}$  is always (locally) possible; for the case when B is a  $C^1$  manifold, see the paragraph containing equations (II.3.12) and (II.3.13).

<sup>&</sup>lt;sup>7</sup>In the equality (9.20), we implicitly suppose that the matrix-valued functions  $\Gamma_{\mu}$ , entering in (9.11) and defining  $\mathbf{R}_{\mu\nu}$  via (9.14), are identical with the matrices of the 3-index coefficients of the transport which appear in (6.1). If U is not a (dim M)-dimensional submanifold of M, in this way we introduce in the definition of  $\mathbf{R}_{\mu\nu}$  an arbitrariness which is a consequence of the one in the definition of  $\Gamma_{\mu}$ ; for some details, see the discussion after Proposition 6.2.

The combination of the above results with the ones of Section 6 leads to a necessary and sufficient condition for the flatness on  $U \subseteq M$  of a linear transport whose coefficient matrix has a representation (9.11) on U. At first sight, one may suppose L is flat on U iff

$$R_{\mu\nu} = 0 \tag{9.21}$$

on U. But, as it is clear from (9.12), these equalities imply the flatness on U but the opposite is generally not true unless U is a neighborhood, e.g., if U = M, or if on  $\Gamma_{\mu}$  are imposed some additional conditions. Besides, as we know from Section 6, the equations (9.21) do not concern solely the transport under consideration, but they also imply some restrictions on the arbitrariness in the definition of the transport's 3-index coefficients. The elimination of the last restrictions on a submanifold of M from (9.21) leads to the following result.

**Proposition 9.3.** A linear transport L along paths in  $(E, \pi, M)$  is flat on a submanifold N of M if and only if for every  $p_0 \in N$  and a chart (V, x) of M such that  $V \ni p_0$  and  $x(p) = (x^1(p), \ldots, x^{\dim N}(p), t_0^{\dim N+1}, \ldots, t_0^{\dim M})$  for every  $p \in N \cap V$ and constant numbers  $t_0^{\dim N+1}, \ldots, t_0^{\dim M}$ , the equalities (6.16) hold for  $p \in N \cap V$ with  $R_{\alpha\beta}^N$  given via (6.17) in which the matrices  $\Gamma_1, \ldots, \Gamma_{\dim N}$ , given via (6.12), are the first dim N of the matrices of the 3-index coefficients of L in the coordinate frame  $\{\frac{\partial}{\partial x^{\mu}}\}$  in the tangent bundle space over  $N \cap V$ .

*Proof.* See Theorem 6.2 on page 253 and Theorem 9.1 or Corollary 9.1.  $\Box$ 

The essence of Proposition 9.3 is that it provides a criterion for a flatness on a submanifold of a linear transport along paths in terms of the 4-index components of its curvature, if they exist.

An important consequence of Theorem 9.1 is that the combination of it and any one of the necessary and sufficient conditions or assertions for the existence of normal frames derived in Sections 5 and 6 results in a corresponding criterion or assertion, respectively, for flatness of a linear transport or derivation along paths. For example, a transport is curvature free iff its action depends only on the initial and final points of the transportation, not on the particular path connecting them (cf. Theorem 5.1); along every fixed path every transport is flat (see Corollary 5.1), etc.

A simple corollary of Theorems 9.1 and 5.1 is the following important criterion for path-independence of a linear transport along paths.

**Corollary 9.3.** Let L be a linear transport along paths in a vector bundle  $(E, \pi, B)$  and  $U \subseteq B$ . The transport L is path-independent over U, i.e., along paths in U, if and only if L is flat on U.

*Proof.* If L is path-independent over U, it is Euclidean on U(Theorem 5.1) and so it is flat on U (Theorem 9.1). Conversely, if L is flat on U, it is Euclidean on U (Theorem 9.1) and, consequently, it is path-independent over U (Theorem 5.1).  $\Box$ 

In short, we can summarize the content of this section by stating that the sets of Euclidean, flat, and path-independent linear transports along paths coincide.

# 10. Torsion and normal coordinates

In the preceding chapters, we have had a number of chances to see that, if frames normal for a linear connection or derivation of the tensor algebra on a manifold exist, then normal coordinates exist iff the torsion of the connection or derivation, respectively, vanishes. *Prima facie* one can expect similar results for linear transports or derivations along paths in vector bundles. But a simple look on the corresponding definitions (I.3.12) and (III.2.13) of the torsion reveals a peculiarity that can not be generalized to arbitrary vector bundles: they include the commutator of two tangent vector fields and the action of a derivation along one of them on the another one. These operations are primary related to the bundle  $(T(M), \pi, M)$ tangent to a manifold M and at present are not known suitable generalizations of them for arbitrary vector bundles.<sup>1</sup> Since it seems that the tangent bundles are the only sufficiently general and frequently used bundles for which the concept of a 'torsion' can be introduced, they will be considered below in the present section.

## 10.1. Torsion of linear transport or derivation along paths in the tangent bundle over a manifold

Let M be a  $C^1$  manifold and  $(T(M), \pi, M)$  the bundle tangent to it. Suppose  $\eta: J \times J' \to M$  is a  $C^2$  injective mapping. Denote by  $\eta'(\cdot, t)$  and  $\eta''(s, \cdot)$  for  $(s, t) \in J \times J'$  the vector fields tangent to the paths  $\eta(\cdot, t): J \to M$  and  $\eta(s, \cdot): J' \to M$ , respectively. We consider  $\eta'$  and  $\eta''$  as vector fields on  $\eta(J, J'), \eta', \eta'' \in \mathfrak{X}(\eta(J, J')) = \operatorname{Sec}((T(M), \pi, M)|_{\eta(J,J')})$ , such that

$$\eta'\colon \eta(s,t)\mapsto \eta'_{\eta(s,t)}:=\eta'(s,t) \quad \text{and} \quad \eta''\colon \eta(s,t)\mapsto \eta''_{\eta(s,t)}:=\eta''(s,t)$$

for every  $(s,t) \in J \times J'$ .<sup>2</sup>

At last, if  $X \in \mathfrak{X}(\eta(J,J'))$ , we denote with  $\hat{X}$  a lifting of paths  $\hat{X} \in \text{PLift}((T(M), \pi, M)|_{\eta(J,J')})$  such that for every path  $\gamma \colon J'' \to \eta(J, J')$ , we have  $\hat{X} \colon \gamma \mapsto \hat{X}_{\gamma} \in P(\pi^{-1}(\gamma(J'')))$  with  $\hat{X}_{\gamma} \colon r \mapsto \hat{X}_{\gamma}(r) := X|_{\gamma(r)}, r \in J''$ . In particular, the equalities  $\hat{\eta}'_{\eta(s,\cdot)}(t) = \eta'(s,t)$  and  $\hat{\eta}''_{\eta(\cdot,t)}(s) = \eta''(s,t)$  are valid.

The torsion (vector) along  $\eta: J \times J' \to M$  at  $(s,t) \in J \times J'$  of a derivation D along paths in the tangent bundle  $(T(M), \pi, M)$  is called the vector

$$T^{\eta}(s,t) := \left(D^{\eta(\cdot,t)}(\hat{\eta}'')\right)(s) - \left(D^{\eta(s,\cdot)}(\hat{\eta}')\right)(t) \in \pi^{-1}(\eta(s,t)).$$
(10.1)

<sup>&</sup>lt;sup>1</sup>The analogue of a derivation along vector fields in a general vector bundle is the concept of a derivation along paths but it acts on liftings of paths, (version: on sections (possibly along paths)), not on paths or on the elements of the total bundle space. If the elements of the bundle space are mappings (operators), under certain conditions, their commutator can be defined, but this is an exceptional case.

<sup>&</sup>lt;sup>2</sup>This is possible as  $\eta$  is supposed injective.

Using the operator  $D_s^{\gamma}$  (see Definition 2.1) and the one of the section-derivation D generated by D (see (2.12)), we can write

$$T^{\eta}(s,t) = D_{s}^{\eta(\cdot,t)}(\hat{\eta}'') - D_{t}^{\eta(s,\cdot)}(\hat{\eta}') = \left(\mathsf{D}^{\eta(\cdot,t)}(\eta'')\right)(\eta(s,t)) - \left(\mathsf{D}^{\eta(s,\cdot)}(\eta')\right)(\eta(s,t)).$$
(10.2)

Introducing a frame  $\{E_i\}$  on  $\eta(J, J')$ , applying (2.4), and using the relation

$$\frac{\partial \eta'^{i}(s,t)}{\partial t} - \frac{\partial \eta''^{i}(s,t)}{\partial s} = C^{i}_{kl}(\eta(s,t))\eta'^{k}(s,t)\eta''^{l}(s,t),$$

which is a consequence of the definitions of  $\eta'$  and  $\eta''$  and the  $C^2$  differentiability of  $\eta$ ,<sup>3</sup> from (10.2), we find the local expression<sup>4</sup>

$$T^{\eta}(s,t) = \left(T^{\eta}(s,t)\right)^{i} E_{i}|_{\eta(s,t)}$$
(10.3)

where the *components* of the torsion in  $\{E_i\}$  are

$$(T^{\eta}(s,t))^{i} := \Gamma^{i}{}_{j}(s;\eta(\cdot,t))\eta^{\prime\prime j}(s,t) - \Gamma^{i}{}_{j}(t;\eta(s,\cdot))\eta^{\prime j}(s,t) - C^{i}_{jk}(\eta(s,t))\eta^{\prime j}(s,t)\eta^{\prime\prime k}(s,t), \quad (10.4)$$

with  $\Gamma^i_{j}$  being the components of D in  $\{E_i\}$  and  $[E_j, E_k]_- :: C^i_{jk} E_i$ .

**Definition 10.1.** The torsion (vector) along  $\eta: J \times J' \to B$  at  $(s,t) \in J \times J'$  of a linear transport along paths in the tangent bundle over a manifold is called the torsion (vector) along  $\eta: J \times J' \to B$  at  $(s,t) \in J \times J'$  of the generated by it via (3.19) derivation along paths.

**Definition 10.2.** The torsion (operator) of linear transports or derivations along paths is a mapping T assigning to each transport or derivation and every  $\eta: J \times J' \to B$  and  $(s,t) \in J \times J'$  the corresponding torsion  $T^{\eta}(s,t)$  along  $\eta$  at (s,t).

**Proposition 10.1.** If the coefficients of a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  have on  $U \subseteq M$  the representation (9.11) along every path  $\gamma$ in U in some (every) frame  $\{E_i\}$  on U, then the components of its torsion vector along  $\eta: J \times J' \to U$  at  $(s, t) \in J \times J'$  are

$$\left(T^{\eta}(s,t)\right)^{i} = T^{i}_{jk}(\eta(s,t))\eta'^{j}(s,t)\eta''^{k}(s,t)$$
(10.5)

with

$$T_{jk}^{i} := -\Gamma_{jk}^{i} + \Gamma_{kj}^{i} - C_{jk}^{i}.$$
(10.6)

 $\Box$ 

*Proof.* Substitute (9.11) for  $\gamma = \eta(\cdot, t), \eta(s, \cdot)$  into (10.4).

<sup>&</sup>lt;sup>3</sup>To derive this equality, calculate the commutator  $[\eta', \eta'']_{-}|_{\eta(s,t)}$  in  $\{E_i\}$  and use  $[\eta', \eta'']_{-} = 0$ as  $(\eta'(s,t))(f) = \frac{\partial f(\eta(s,t))}{\partial s}$  and  $(\eta''(s,t))(f) = \frac{\partial f(\eta(s,t))}{\partial t}$  for a  $C^1$  function f on  $\eta(J, J')$ . <sup>4</sup>One can at first derive (10.4) in a coordinate frame and then, applying the corresponding

<sup>&</sup>lt;sup>4</sup>One can at first derive (10.4) in a coordinate frame and then, applying the corresponding transformation laws, the result can be transformed into an arbitrary frame.

**Lemma 10.1.** The representation (9.11) on page 265 holds on  $U \subseteq M$ , i.e., for paths lying entirely in U, for some linear transport L along paths in the tangent bundle  $(T(M), \pi, M)$  if and only if L along paths in U coincides with the parallel transport generated by a linear connection.

Proof. If (9.11) holds, the matrices  $\Gamma_{\mu}$  transform according to (6.4) with B = A (as now  $(E, \pi, M) = (T(M), \pi, M)$ ) and, consequently (see (I.5.3)), they are coefficients' matrices of a linear connection  $\nabla$ . Therefore the derivation D along paths generated by L coincides with the covariant derivative along paths generated by  $\nabla$ . Since the linear transport along paths generated by the last derivation is the parallel transport assigned to  $\nabla$  (see Definition I.3.2 on page 27 and Proposition 3.7), the last transport coincides with L.

Conversely, from definitions (I.3.18) and (I.3.17) follows that, for every parallel transport generated by a linear connection  $\nabla$ , the representation (9.11) is valid if in it  $\Gamma_{\mu}$  are the coefficients' matrices of  $\nabla$ .

Lemma 10.1 explains the reason why in equation (10.5) appear the components  $T^i_{jk}$  of the torsion tensor of some linear connection with local coefficients  $\Gamma^i_{ik}$  (see (I.3.14)).

**Example 10.1.** The torsion tensor  $T^{\nabla}$  of a linear connection  $\nabla$  can be introduced by means of the torsion operator T of the assigned to it parallel transport or covariant derivative along paths. Indeed, defining the mapping  $\eta: J \times J' \to M$  as we did before equation (9.17), after a simple calculation, we find

$$T^{\eta_p}(s_0, t_0) = (T^{\nabla}(A, B))\Big|_n$$
 (10.7)

where  $T^{\nabla}$  is exactly the torsion tensor of  $\nabla$  defined by (I.3.12) and with local components (I.3.14), i.e., (10.6).

**Example 10.2.** If **D** is a linear derivation along vector fields (see Definition III.2.2) and D is a derivation along paths in the bundle  $(T(M), \pi, M)$  tangent to M for which (9.11) holds for the coefficients' matrices  $\Gamma_{\mu}$  of **D**, then the torsion of D along  $\eta_p$  at  $(s_0, t_0)$  is

$$T^{\eta_p}(s_0, t_0) = \left(T^{\mathbf{D}}(A, B)\right)\Big|_n$$
 (10.8)

with  $T^{\mathbf{D}}$  being the torsion of  $\mathbf{D}$  given via (III.2.13) and with local components given by (III.2.15).

**Definition 10.3.** A linear transport or derivation along paths in  $(T(M), \pi, M)$  is called *torsionless* or *torsion free* (resp. on  $U \subseteq M$ ) if its torsion (operator) vanishes on M (resp. on U).

Remark 10.1 (Cf. Remark 9.1 on page 267). It should be emphasized on a principal difference between the torsionless on  $U \subseteq B$  of a linear transport along paths in the bundle  $(T(M), \pi, M)$  and of a linear connection or derivation along tangent vector fields on a manifold M. In the former case, the torsion is zero on U along mappings

 $\eta$  in U, i.e.,  $\eta: J \times J' \to U$ , while in the latter one it vanishes only along vectors tangent to U, i.e., from the tangent space  $T_p(M)$ ,  $p \in U$ . If U is a submanifold of M, the first case means that  $\eta'(s,t), \eta''(s,t) \in T_{\eta(s,t)}(U) \subseteq T_{\eta(s,t)}(M)$ , while in the second one, we have  $\eta'(s,t), \eta''(s,t) \in T_{\eta(s,t)}(M)$ , i.e.,  $\eta'$  and  $\eta''$  are not generally in the tangent bundle space over U in  $(T(U), \pi, U)$ .

**Proposition 10.2.** A linear transport (resp. section-derivation) along paths in the tangent bundle  $(T(M), \pi, M)$  is torsionless over a submanifold N of M, i.e., along paths in N, if and only if its restriction to the bundle space  $T(N) := \bigcup_{p \in N} T_p(N)$  (resp. to Sec<sup>1</sup> $(T(N), \pi|_N, N)$ ) coincides with the restriction to the same set of the parallel transport (resp. covariant derivative) along paths in N generated by a linear connection and which transport (resp. covariant derivative) is torsion free on the mentioned set.

Proof. If D (resp. L) is a torsion free covariant derivative (resp. parallel transport) along paths generated by a linear connection  $\nabla$ , then, due to Lemma 10.1 (resp. Definition I.3.2 and (I.3.17)), its components (resp. coefficients) have locally the representation (9.11) in which  $\Gamma_{\mu}$  are the coefficients' matrices of  $\nabla$ . Therefore the torsion of D (resp. L) is given via (10.5) which vanishes on N, by virtue of (10.6), (10.7), and the torsionless of D (resp. L).

Conversely, let D (resp. L) be a section-derivation (resp. linear transport) along paths which is torsion free on the tangent bundle  $(T(N), \pi, N)$  over N.

At first, we consider the case  $\dim N = \dim M$ .

Let  $\eta: J \times J' \to N$ ,  $(s,t) \in J \times J'$ ,  $\{E_i\}$  be a frame on N, and  $\Gamma_j^i$  be the components of D (resp. the coefficients of L) in  $\{E_i\}$ . From (10.3) and (10.4) follows that the torsionless is equivalent to

$$\Gamma^{i}{}_{j}(s;\eta(\,\cdot\,,t))\eta^{\prime\prime\,j}(s,t) = \Gamma^{i}{}_{j}(t;\eta(s,\,\cdot\,))\eta^{\prime\,j}(s,t) + C^{i}_{jk}(\eta(s,t))\eta^{\prime\,j}(s,t)\eta^{\prime\prime\,k}(s,t).$$

Since  $\eta$  is arbitrary, so are  $\eta'$  and  $\eta''$  and, consequently, the last equation implies the expansion  $^{5,\ 6}$ 

$$\Gamma^{i}_{\ j}(r;\gamma)=\Gamma^{i}_{\ jk}(r;\gamma)\dot{\gamma}^{k}(r),\qquad r=s,t\quad \text{and resp. }\gamma=\eta(\,\cdot\,,t),\eta(s,\,\cdot\,)$$

for some functions  $\Gamma^i_{jk}$  (generally non-symmetric in j and k) which do not depend on  $\dot{\gamma}$ . Inserting this representation for  $\Gamma^i_{j}$  into the last equation, we get

$$\Gamma^{i}_{kj}(s;\eta(\,\cdot\,,t)) = \Gamma^{i}_{jk}(t;\eta(s,\,\cdot\,)) + C^{i}_{jk}(\eta(s,t))$$

as  $\eta'$  and  $\eta''$  are arbitrary. Using, once again, the arbitrariness of  $\eta$ , hereof follows that  $\Gamma^i_{\ jk}(r;\gamma)$  must depend on the combination  $\gamma(r)$ , not on r and  $\gamma$  separately,

<sup>&</sup>lt;sup>5</sup>From the differentiation of the last equality with respect to  $\eta'^{i}(s,t)$  or  $\eta''^{i}(s,t)$  follows the below-written expansion for  $\Gamma^{i}_{j}$  and the second derivative of the this equality with respect to  $\eta'^{i}(s,t)$  and  $\eta''^{i}(s,t)$  results in the equation presented after the expansion.

<sup>&</sup>lt;sup>6</sup>Here and below the assumption that N is a neighborhood is essentially used as otherwise, generally, from equality like  $A_i(t; \eta(s, \cdot))\eta'^i(s, t) = 0$  does not follow  $A_i(t; \eta(s, \cdot)) = 0$ .

i.e.,  $\Gamma^{i}_{\ jk}(r;\gamma) = \Gamma^{i}_{\ jk}(\gamma(r))$ . Therefore, we have:

$$\Gamma^{i}_{\ j}(r;\gamma) = \Gamma^{i}_{\ jk}(\gamma(r))\dot{\gamma}^{k}(r)$$
  
$$\Gamma^{i}_{\ kj}(\gamma(r)) = \Gamma^{i}_{\ jk}(\gamma(r)) + C^{i}_{\ jk}(\gamma(r))$$

for r = s, t and respectively  $\gamma = \eta(\cdot, t), \eta(s, \cdot)$ . Since  $\eta: J \times J' \to N$  is arbitrary, the last two equations must hold for *every* path  $\gamma: J'' \to N$ . The former of these equations, combined with Lemma 10.1, implies that D (resp. L) is on N a covariant derivative (resp. parallel transport) along paths assigned to a linear connection  $\nabla$ (with local coefficients  $\Gamma^i_{jk}$ ) on N. From the latter equation, when substituted into (10.6) or (I.3.14), follows that  $\nabla$  and the assigned to it parallel transport are torsionless on N.

Let now N be a submanifold of M of dimension  $\dim N < \dim M$ .

Note 10.1. The proof of the theorem can be completed by noticing that N is a manifold by itself (as it is a submanifold of M – see page 7) and, consequently, to this case is valid the already proved theorem for dim  $N = \dim M$  with N for M. The reformulation of this observation in terms of submanifolds yields the required result. The below-presented alternative and independent proof is considerably longer but it reveals some features of the methods employed and parts of it will be used later. If, at this point, this is not interesting for the reader, he/she can skip the text after the present note till the end of this proof.

Suppose  $\{E_i = \frac{\partial}{\partial x^i}\}$  is the coordinate frame associated to a local coordinate system  $\{x^i\}$  in a neighborhood of  $\eta(s,t)$  in M such that  $x^i(q) = a^i = \text{const} \in \mathbb{K}$ for every  $i > \dim N$  and q in a neighborhood of  $\eta(s,t)$  in N (see page 7). Then in the above proof of the case with dim  $N = \dim M$  one should put  $\eta'^i = \eta''^i = 0$  for  $i > \dim N$  as the range of  $\eta$  is in N. Taking this into account and repeating *mutatis mutandis* the above lines, concerning the case dim  $N = \dim M$ , we conclude that, for every path  $\gamma: J'' \to N$ , is fulfilled

$$\begin{split} \Gamma^{i}{}_{j}(r;\gamma) &= \Gamma^{i}{}_{jk}(\gamma(r))\dot{\gamma}^{k}(r), \qquad j \leq \dim N, \, i=1,\ldots, \dim M \\ \Gamma^{i}{}_{kj}(p) &= \Gamma^{i}{}_{jk}(p) + C^{i}_{jk}(p), \qquad j,k \leq \dim N, \, i=1,\ldots, \dim M, \, \, p \in N \end{split}$$

and for  $j, k > \dim N$  the functions  $\Gamma^{i}_{jk} \colon N \to \mathbb{K}$  are completely arbitrary. (Notice, since  $\gamma$  is a path in N, we have  $\dot{\gamma}^{k} = 0$  for  $k > \dim N$ ; besides,  $C^{i}_{jk} = 0$  as  $E_{i} = \frac{\partial}{\partial x^{i}}$ .)

Let  $\nabla$  be a linear connection on (a neighborhood of) N whose coefficients on N in  $\{E_i\}$  are equal to the above functions  $\Gamma^i_{jk}$  which for  $j, k > \dim N$  are arbitrarily fixed. Consider the restriction to  $\operatorname{Sec}^1(T(N), \pi|_N, N)$  of the covariant derivative (resp. to T(N) of the parallel transport) along paths in N assigned to  $\nabla$ . According to (I.3.17) (resp. Definition I.3.2), it depends only on  $\Gamma^i_j(r; \gamma) =$  $\Gamma^i_j(\gamma(r))$  with  $j \leq \dim N$  when applied on tensor fields (resp. tensors) over N as a manifold, i.e., sections of the tangent bundle  $(T(N), \pi|_N, N)$  over N (resp. in the bundle space T(N)). Hence, by construction, this restriction coincides with the one to the same set of the initial section-derivation D (resp. linear transport L) along paths. As a consequence of the last result, it is torsionless.

Proposition 10.2 has an equivalent version which is more suitable for the most applications.

**Proposition 10.2'.** Let L (resp. D) be a linear transport (resp. section-derivation) along paths in the tangent bundle  $(T(M), \pi, M)$  over a  $C^1$  manifold M and N be a submanifold of M. Then L (resp. D) is torsionless on N if and only if it coincides on N, i.e., along paths in N, with the torsionless on N parallel transport (resp. the torsionless on N covariant derivative) along paths assigned to some linear connection (on (a neighborhood of) N).

*Proof.* This proof is identical with the one of Proposition 10.2 till, but *not* including, its last paragraph. It should be replace by the following text.

Let  $\nabla$  be a linear connection on (a neighborhood of) N whose coefficients  $\overline{\Gamma}^{i}_{jk}$  on N in  $\{E_i\}$  are such that  $\overline{\Gamma}^{i}_{jk} = \Gamma^{i}_{jk}$  for  $j, k \leq \dim N$  ( $\Gamma^{i}_{jk}$  are the same as above) and  $\overline{\Gamma}^{i}_{jk} = 0$  for  $j, k > \dim N$ .<sup>7</sup> Then the parallel transport (resp. covariant derivative) along paths in N depends only on  $\overline{\Gamma}^{i}_{jk}$  with  $j, k \leq \dim N$  (see Definition I.3.2 (resp. (I.3.17))) and, by construction, it coincides with the initial transport L (resp. section-derivation D), due to which it is torsion free on N.  $\Box$ 

Remark 10.2. Having in mind Remarks 9.1 and 10.1, we want to emphasize on the fact that the vanishment on  $U \subseteq M$  of the curvature/torsion of the parallel transport assigned to a linear connection  $\nabla$ , generally, does not imply that  $\nabla$ is flat/torsionless. An exception is the case when U is a  $(\dim M)$ -dimensional submanifold of M.<sup>8</sup>

## **10.2.** Holonomic normal frames in the tangent bundle

The main purpose of this subsection is the proof of the below-presented theorem whose meaning is that, in the tangent bundle over a manifold, the flat torsionless parallel transports (resp. covariant derivatives) along paths generated by linear connections are the only linear transports (resp. section-derivations) along paths admitting normal coordinates, i.e., holonomic normal frames. Notice, here the flatness is required for the existence of the normal frames (Theorem 9.1).

**Theorem 10.1.** Let L be an arbitrary linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  over a  $C^3$  manifold M and L be flat over a neighborhood

<sup>&</sup>lt;sup>7</sup>The condition  $\overline{\Gamma}^{i}_{\ jk} = 0$  for  $k > \dim N$  is not necessary as  $\overline{\Gamma}^{i}_{\ jk}$  appear only in the combination  $\overline{\Gamma}^{i}_{\ jk}\dot{\gamma}^{k}$  with  $\gamma$  in U, i.e.,  $\dot{\gamma}^{k} = 0$  for  $k > \dim N$  in the coordinates used.

<sup>&</sup>lt;sup>3</sup><sup>8</sup>These assertions are clearly seen from Corollary 11.1 and Remark 11.3 below in a case U is a submanifold of M.

 $U \subseteq M$ . Holonomic on U frames normal for L on U exist if and only if L is torsionless over U or, equivalently, iff L coincides over U with the (flat and) torsionless parallel transport assigned to a (flat and torsionless) linear connection on U.

*Proof.* (i) It follows from Proposition 10.2' and (9.17) that L is torsionless and flat over (a submanifold, in particular neighborhood) U iff over U it coincides with the parallel transport generated by a linear connection and which transport is flat and torsion free on U.

(ii) If L admits a frame  $\{E_i\}$  which is holonomic and normal on U, i.e., such that  $C_{jk}^i = 0$  on (a neighborhood of) U and  $\Gamma_j^i = 0$  on U, equation (10.4) yields  $T^{\eta}(s,t) = 0$  for every  $\eta: J \times J' \to U$ . Hence, by definition, L is torsionless on U.

(iii a) At last, let on T(U) the transport L coincides with the flat torsionless parallel transport generated by a linear connection  $\nabla$  (on U).<sup>9</sup> Then (see Lemma 10.1) the coefficients of L are  $\Gamma^i{}_j(s;\gamma) = \Gamma^i{}_{jk}(\gamma(s))\dot{\gamma}^k(s), s \in J$ , for every path  $\gamma: J \to U$  with  $\Gamma^i{}_{jk}$  being the coefficients of the mentioned linear connection. Therefore the curvature (resp. torsion) of L is given via (9.12) and (9.13) (resp. (10.5) and (10.6)) with  $i, j, \alpha, \beta = 1, \ldots, \dim M$  and the values of  $\eta$  in U.

(iii b) Since L is flat on U, it is Euclidean on U (Theorem 9.1) and, by Theorem 5.2, its coefficients' matrix on U in a frame  $\{E_i\}$  on U has the form  $\Gamma(s;\gamma) = F_0^{-1}(\gamma(s)) \frac{\mathrm{d}F_0(\gamma(s))}{\mathrm{d}s}$  for some non-degenerate matrix-valued function  $F_0: U \to \mathrm{GL}(\mathbb{K}, \dim M)$  of class  $C^1$  and every path  $\gamma: J \to U$ . On U, we define a frame  $\{E'_i\}$  by  $E'_i = (F_0^{-1})^j_i E_j$ .

Our intention is to prove that the frame  $\{E'_i\}$  is holonomic and normal on U, i.e.,  $[E'_j, E'_k] = : C'_{jk} E'_i$  with  $C'_{jk} = 0$  and  $\Gamma'_{j} = 0$  on U. The functions  $C'_{jk}$  are expressible via the analogous ones  $C^i_{jk}$  for  $\{E_i\}$ ,  $[E_j, E_k] = : C^i_{jk} E_i$ , through equation (I.8.4) on page 69 with  $A = \mathbf{F}_0^{-1}$ :

$$C_{jk}^{\prime i} = (\mathbf{F}_0)_l^i \Big[ \big( \mathbf{F}_0^{-1} \big)_j^m \big( \mathbf{F}_0^{-1} \big)_k^n C_{mn}^l + 2E_{[j}^{\prime} \big( \mathbf{F}_0^{-1} \big)_{k]}^l \Big].$$

After some simple algebraic manipulations, from here we get

$$C_{jk}^{\prime i} = (\mathbf{F}_0)_l^i (\mathbf{F}_0^{-1})_j^m (\mathbf{F}_0^{-1})_k^n T_{mn}^l = -T_{jk}^{\prime i}$$
(10.9)

where  $T_{jk}^i$  are the components in  $\{E_i\}$  of the torsion tensor of  $\nabla$  given by (10.6) and  $T'_{jk}^i$  are the same ones in  $\{E'_i\}$ .<sup>10</sup> Since the parallel transport generated by the

<sup>&</sup>lt;sup>9</sup>Since U is a neighborhood, from here follows (see (10.7), (9.17), and Definition I.3.2 on page 27) that  $\nabla$  is also flat and torsion free on U. Therefore, by Theorem II.4.3 on page 108,  $\nabla$  admits holonomic frames normal on U. As the 3-index coefficients of  $\nabla$  and of the parallel transport assigned to it coincide, this transport is also Euclidean on U and the (holonomic) frames normal for it and  $\nabla$  are identical. This completes the proof of the theorem. If the reader is not interested from alternative proof and some other details, he/she can skip the rest of the proof of Theorem 10.1, parts (iii b)–(iii d) below.

<sup>&</sup>lt;sup>10</sup>From (10.6) follows that (10.9) is valid in a frame  $\{E''_i\}$  iff  ${\Gamma''}^i{}_{jk} = {\Gamma''}^i{}_{kj}$ , in particular this is true in every frame normal for  $\nabla$  on U.
linear connection  $\nabla$  is supposed torsionless on U, the equality

$$C_{jk}^{\prime i}(\eta(s,t))(\eta^{\prime}(s,t))^{\prime j}(\eta^{\prime\prime}(s,t))^{\prime k} = 0$$
(10.10)

is valid for every  $\eta: J \times J' \to U$  as a consequence of (10.9), (10.5), and  $T^{\eta}(s,t) = 0$  on U. Here  $(\eta')^{i}$  and  $(\eta'')^{i}$  are the components of  $\eta'$  and  $\eta''$  in  $\{E'_i\}$ .

(iii c) Taking into account that U is a neighborhood in M (and hence it is  $(\dim M)$ -dimensional submanifold of M) and the arbitrariness of  $\eta$  (which means that  $\eta'(s,t)$  and  $\eta''(s,t)$  are arbitrary vectors in  $T_{\eta(s,t)}(M)$ ), from equation (10.10), we obtain  $C_{ik}^{\prime i}|_{U} = 0$ , i.e.,  $\{E_{i}^{\prime}\}$  is holonomic on U.

(iii d) The proof is completed by the observation that the frame  $\{E'_i = (\mathbf{F}_0^{-1})_i^j E_j\}$  is normal on U. In fact, inserting the representation (5.4') into the transformation law (3.26) with  $A = \mathbf{F}_0^{-1}$ , we find  $\Gamma'(s; \gamma) = 0$  for every  $\gamma: J \to U$ .

In the above proof of Theorem 10.1, we intentionally have singled out its several parts, (i)-(iii d), and the intermediate equations (10.9) and (10.10). A superficial look on them reveals that parts (ii)–(iii b) and (iii d) are valid for arbitrary subset  $U \subseteq M$  as the initial assumption on U to be a neighborhood in M was not used in them. On the contrary, the conclusions in part (iii c) rely entirely on the supposition that U is a neighborhood in M as otherwise from (10.10) one can not, generally, infer the holonomicity of the frame  $\{E'_i\}, C'_{ik} = 0$  on some open set containing or equal to U: if U is not a neighborhood in M, then, generally,  $\eta'(s,t)$ and  $\eta''(s,t)$  are in some proper subset of  $T_{\eta(s,t)}(M)$  (see, e.g., Proposition 11.3 below). The origin of this situation comes from the fact that the torsionlessness on  $U \subseteq M$  of the parallel transport generated by a linear connection  $\nabla$  is, generally, weaker condition than the torsionlessness of  $\nabla$ : the latter one implies the former but the former, generally, does not lead to the latter one. This conclusion is completely similar to an analogous result concerning the curvature on a submanifold N of M of linear connections which is expressed by Lemma II.5.1 on page 112: the path-independence on N in it is equivalent to the flatness of the transport on U, so it is equivalent to (II.5.1), not to the flatness of the connection on N (cf. Remark II.5.1). At the end, the conclusions in part (i) rest on Proposition 10.2'which is valid if U is a submanifold of M, not only a neighborhood in it, but it may not hold on an arbitrary its subset.

On the base of the above comments, we can formulate the next two propositions which are stronger variants of different parts of Theorem 10.1.

**Proposition 10.3.** A linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  is flat and torsionless on a submanifold N of M if and only if it coincides over N with the parallel transport generated by a linear connection on (a neighborhood of) N which transport is flat and torsionless over N.

**Exercise 10.1.** Prove, independently of (10.9), that the frame  $\{E'_i = (F_0^{-1})_i^j E_j\}$  is normal for  $\nabla$  on U. (Notice, here is important the supposition U to be a neighborhood.)

*Proof.* See Proposition 10.2' and (9.17)

**Proposition 10.4.** If a linear transport along paths in  $(T(M), \pi, M)$  admits a frame which is normal and holonomic on a set  $U \subseteq M$ , then it is flat and torsion free on U.

*Proof.* See part (ii) of the proof of Theorem 10.1.

**Proposition 10.5.** Let L be a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$ . Let L be flat and torsionless over a neighborhood U in the manifold M and the  $C^1$  non-degenerate matrix-valued function  $\mathbf{F}_0: U \to \operatorname{GL}(\dim M, \mathbb{K})$  generates the coefficients of L via (5.4') in a frame  $\{E_i\}$  on U. Then:

(i) The frame  $\{E'_i\}$  with

$$E'_{i} = \left(\boldsymbol{F}_{0}^{-1}\right)_{i}^{j} E_{j} \tag{10.11}$$

is holonomic and normal for L on U.

- (ii) All frames normal for L on U are holonomic.
- (iii) All frames which are holonomic and normal for L on U are obtainable from  $\{E'_i\}$  by linear transformations with coefficients constant on U.

*Proof.* (i) See parts (iii b)–(iii d) of the proof of Theorem 10.1

(ii) According to Corollary 4.5 on page 239, all frames normal for L on U are  $\{E''_i = B^j_i E'_j\}$  with  $B^i_j|_U = \text{const.}$  Hence these frames are holonomic on U,  $[E''_j, E''_k]_{-} = 0$ , as the frame  $\{E'_i\}$  is such,  $[E'_j, E'_k]_{-} = 0$ .

(iii) This part of the assertion follows from part (ii) and Corollary 4.5.  $\Box$ 

Remark 10.3. The non-degenerate matrix-valued function  $F_0$ , appearing in equation (10.11), is not arbitrary: since the transport L is assumed torsionless on Uand U is a neighborhood, from (6.3) (see Theorem 5.2) and (10.6)) follows that the matrix elements  $(F_0)_i^i$  of  $F_0$  must be solutions of the system of equations

$$-(\boldsymbol{F}_{0}^{-1})_{m}^{i}E_{k}(\boldsymbol{F}_{0})_{j}^{m} + (\boldsymbol{F}_{0}^{-1})_{m}^{i}E_{j}(\boldsymbol{F}_{0})_{k}^{m} - C_{jk}^{\prime i} = 0$$
(10.12)

on U, where  $C'_{jk}^{i}$  are given by (I.8.4) with  $A = \mathbf{F}_0^{-1}$ . The constant solutions of (10.12) correspond to frames holonomic and normal for the linear connection generating the parallel transport with which L coincides on U.

**Definition 10.4.** Let L (resp. D) be a linear transport (resp. derivation) along paths in the tangent bundle  $(T(M), \pi, M)$  over a  $C^1$  manifold M. A chart (V, x) of M and the associated to it coordinate system  $\{x^i\}$  are called normal (for L (resp. D)) on a set  $W \subseteq V$  if the coordinate frame  $\{\frac{\partial}{\partial x^i}\}$  is normal for L (resp. D) on W.

**Proposition 10.6.** Let L be a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$ . Let L be flat and torsion free on a neighborhood U in M and  $F_0: U \to \operatorname{GL}(\dim, \mathbb{K})$  be of class  $C^1$  and generates on V the coefficients of L

via (5.4') in the frame  $\left\{\frac{\partial}{\partial y^i}\right\}$  associate to a local chart (V, y) of M such that  $V \subseteq U$ . Then all coordinates  $\{x^i\}$  normal for L on V are given by

$$x^{i}(p) = x_{0}^{i} + \int_{p_{0}}^{p} (\boldsymbol{F}_{0}B^{-1})_{j}^{i} \,\mathrm{d}y^{j} = x_{0}^{i} + (B^{-1})_{j}^{k} \int_{p_{0}}^{p} (\boldsymbol{F}_{0})_{k}^{i} \,\mathrm{d}y^{j}$$
(10.13)

where  $p \in V$ ,  $p_0 \in V$  is fixed,  $x_0^i \in \mathbb{K}$  are constants, and  $B = [B_j^i]$  is constant non-degenerate matrix.

Proof. From the proof of Proposition 10.5 follows that  $\{x^i\}$  is normal iff  $\frac{\partial}{\partial x^i} = B_i^j (\mathbf{F}_0^{-1})_j^k \frac{\partial}{\partial y^k}$  for non-degenerate  $B = [B_i^j] \in \operatorname{GL}(\dim M, \mathbb{K})$ , which is equivalent to (10.13).

**Exercise 10.2.** Prove that the path-independence of the integral in (10.13) is a consequence of the torsionless of L on  $U \supseteq V$ . (Cf. the proof of the path-independence of the integrals in (II.5.14), or (II.5.3), or (11.13). For a rigorous proof, see (11.13) below and the proof after it for  $A = BF_0^{-1}$ .)

**Proposition 10.7.** If L is torsionless linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  over a  $C^1$  real manifold M, then along every fixed injective path  $\gamma: J \to M$  exist holonomic frames (local coordinates) normal for L along  $\gamma$ .

*Proof.* By Corollary 5.1 and Proposition 5.5, the transport L admits frames defined solely along  $\gamma$  and normal along  $\gamma$ . All of them, by Lemma III.10.1 on page 194, can be extended in a holonomic way outside  $\gamma(J)$ .

Here we shall stop the investigation of the holonomic frames normal for linear transports along paths in the tangent bundle over a differentiable manifold. The reason is that these problems, as well as any ones concerning such transports on submanifolds, can be reduced to similar problems for parallel transports (assigned to linear connections) or to linear connections over the same manifold (see Proposition 10.3). Besides, the problems for existence and description of the holonomic frames are solved by Proposition 10.5. The explicit description of the coordinates normal on a neighborhood is provided by Proposition 10.6. A similar result on arbitrary submanifolds will be derived in Subsection 11.2.

### 11. Parallel transports in tangent bundles

The (flat and/or torsionless) parallel transports assigned to a linear connections appeared several times in Section 10, when the theory of normal frames was concerned. In the present section, we shall investigate the parallel transports (assigned to linear connections) from the view-point of the general theory of linear transports along paths. In particular, we shall show that the parallel transports are uniquely selected among the other transports along paths when normal frames are involved.

#### 11.1. The parallel transport as a transport along paths

The concept of a 'parallel transport', assigned to a linear connection over a manifold M, was introduced by Definition I.3.2 on page 27 in Subsection I.3.3 where some essential its properties were mentioned. They are enough to be proved that every parallel transport, when restricted to tensors of a fixed type, is a special kind of a linear transport along paths in the corresponding tensor bundle over M. To save some writing, below we shall investigate only the case of the tangent bundle  $(T(M), \pi, M)$  over a  $C^1$  manifold M but, mutatis mutandis (by adding additional indices, e.g., replacing T(M) with  $T_q^r(M)$ ), the below-presented material can be transferred on a tensor bundle  $(T_q^r(M), \pi_q^r, M)$  of arbitrary type (r, q) over M.

Let M be a  $C^1$  manifold,  $(T(M), \pi, M)$  be the tangent bundle over it,  $\gamma: J \to M$  be a  $C^1$  path, and P be the parallel transport assigned to a  $C^0$ linear connection  $\nabla$  on M. As we know (see Definition I.3.2), if  $a, b \in J$  and  $a \leq b$ , then  $\mathsf{P}^{\gamma|[a,b]}: T_{\gamma(a)}(M) \to T_{\gamma(b)}(M)$  and  $\mathsf{P}^{\gamma|[a,b]}(X_0) = X_{\gamma(b)}$  for every  $X_0 \in T_{\gamma(a)}(M)$  and a vector field X (over  $\gamma(J)$ ) which is the solution of  $\nabla_{\dot{\gamma}} X = 0$ with  $X_{\gamma(a)} = X_0$ . A simple analysis reveals that the only place, where the ordering of the real numbers a and  $b, a \leq b$ , is used, is in the interval [a, b] in which, by definition, its left end must be less than or equal to its right one. This restriction in the definition of a parallel transport is, by our opinion, completely unnecessary and can be removed by a suitable generalization or redefinition of the concept of a 'parallel transport'.

**Definition 11.1.** A parallel transport (along paths), assigned to a linear connection  $\nabla$ , in the tangent bundle  $(T(M), \pi, M)$  is a mapping  $P: \gamma \mapsto P^{\gamma}, \gamma: J \to M$  being a  $C^1$  path, where  $P^{\gamma}$ , called *parallel transport along*  $\gamma$ , maps every pair  $(a, b) \in J \times J$  into a mapping

$$P_{a \to b}^{\gamma} \colon T_{\gamma(a)}(M) \to T_{\gamma(b)}(M), \tag{11.1}$$

called parallel transport along  $\gamma$  from a to b, such that if  $X_0 \in T_{\gamma(a)}(M)$ , then

$$P_{a \to b}^{\gamma}(X_0) = X_{\gamma(b)} \tag{11.2}$$

where X is a vector field over  $\gamma(J)$  defined as the unique solution of the initialvalue problem

$$\nabla_{\dot{\gamma}} X = 0 \tag{11.3a}$$

$$X_{\gamma(a)} = X_0.$$
 (11.3b)

Remark 11.1. Recall (see Remark I.3.3 on page 28), the vector field X defined via (11.3) is generally multiple-valued at the points of self-intersection of  $\gamma$ , if any. To restore the single-valueness, one should consider X as a lifting of  $\gamma$  in  $(T(M), \pi, M)$ , i.e.,  $X: J \to T(M)$  with  $\pi \circ X = \gamma$ ; in this case it is better to write  $X_{\gamma}$  for X and  $X_{\gamma}(s)$  for  $X_{\gamma(s)}$ ,  $s \in J$  and the action of  $\nabla_{\dot{\gamma}}$  on  $X_{\gamma}$  is given by (I.3.19) which is in a full agreement with (2.4) and (2.16) (see also (III.10.5) and Section 13). Comparing Definitions I.3.2 and 11.1, we see that the parallel transports P and P, introduced by them, are equivalent in a sense that each of them can be expressed through the other one:

$$P_{a \to b}^{\gamma} = \begin{cases} \mathsf{P}^{\gamma | [a,b]} & \text{for } a \le b \\ \left(\mathsf{P}^{\gamma | [b,a]}\right)^{-1} & \text{for } a \ge b \end{cases}$$
(11.4a)

$$\mathsf{P}^{\beta} = P^{\beta}_{a \to b} \qquad \text{for } \beta \colon [a, b] \to M, \ a \le b. \tag{11.4b}$$

**Proposition 11.1.** The parallel transport P, as defined by Definition 11.1, is a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$ .

*Proof.* Due to Definition 3.1 and (11.1), we have to check (3.2)–(3.4) with P for L, which is almost trivial: the equality (3.2) is a simple consequence of the uniqueness of the solutions of the initial-value problem (11.3) along  $\gamma$  written in terms of P, equation (3.3) follows from (11.2) with b = a and (11.3b) as  $X_0$  is arbitrary, and the linearity condition (3.4) is a result of the linearity of equation (11.3a) as  $\nabla_{\dot{\gamma}}$  is a linear operator (see Definition I.3.1, points (ii) and (iii), or (I.3.17)).

This quite simple result has two consequences which are important for us. On one hand, the parallel transport (along paths and assigned to a linear connection) provides a concrete and essential example of a linear transport along paths and, on another hand, all results derived until now in the present chapter are valid for the parallel transports (in the tensor bundles over a manifold).

Below, when talking about parallel transports, we shall have in mind the ones introduced by Definition 11.1, *not* by Definition I.3.2. If there is a risk of misunderstanding, we shall call such transports *parallel transports along paths.*<sup>1</sup>

It is well known, a parallel transport on a neighborhood U, i.e., along paths in U, is path-independent iff the generating it linear connection is flat on U. Analogous result on arbitrary submanifold N of M is expressed by Lemma II.5.1 on page 112. Proposition 11.1 ensures the formulation of this important assertion in internal terms, i.e., ones involving only parallel transports.

**Proposition 11.2.** Let N be a submanifold of a  $C^3$  manifold M. A  $C^1$  parallel transport is path-independent over N, i.e., along  $C^1$  paths in N, if and only if it (considered as a linear transport along paths in  $(T(M), \pi, M)$ ) is flat on N.

*Proof.* The result follows from Lemma II.5.1, Remark II.5.3 after its proof, and equations (9.20) and (9.12).

A little below (see Corollary 11.2), it will be proved the validity of Proposition 11.2 for arbitrary set  $U \subseteq M$ , not only for submanifolds of M.

**Proposition 11.3.** The parallel transports in the tangent bundle  $(T(M), \pi, M)$  are the only linear transports along  $C^1$  paths in this bundle whose coefficients have the

<sup>&</sup>lt;sup>1</sup>For some general relations between parallel transports, connections, and transports along paths in differentiable bundles, the reader is referred to Section V.8.

11. Parallel transports in tangent bundles

representation

$$\Gamma(s;\gamma) = \Gamma_k(\gamma(s))\dot{\gamma}^k(s) \tag{11.5}$$

in any frame.

*Proof.* This is a reformulation of Lemma 10.1 on page 273.

**Proposition 11.4.** If P is a parallel transport along paths in  $(T(M), \pi, M)$  assigned to a linear connection  $\nabla$ , then the generated by it derivation along  $C^1$  paths D, defined via (3.19), is such that

$$\left(D^{\gamma}(\hat{X})\right)(s) = (\nabla_{\dot{\gamma}}X)|_{\gamma(s)} \tag{11.6}$$

where  $\gamma: J \to M$  is a  $C^1$  path,  $X \in \mathfrak{X}(M)$ , and  $\hat{X} \in \text{PLift}^1(T(M), \pi, M)$  is given by  $\hat{X}: \gamma \mapsto \hat{X}_{\gamma}: s \mapsto \hat{X}_{\gamma}(s) := X_{\gamma(s)}, s \in J$ . The section-derivation D corresponding to D coincides with the covariant derivative along injective  $C^1$  paths generated by  $\nabla$ :

$$\mathsf{D}^{\gamma} = \nabla_{\dot{\gamma}} \tag{11.7}$$

for every  $C^1$  path  $\gamma$  without self-intersections.

Proof. (Cf. the first part of the proof of Lemma 10.1.) Since P is a parallel transport, the equality (11.5) holds in a frame  $\{E_i\}$  (Proposition 11.3) and in any other frame  $\{E'_i = A^j_i E_j\}$  the matrices  $\Gamma'_k$  are given by (6.4) with  $B = A := [A^j_i]$  and  $\mu, \nu = 1, \ldots, \dim M$  as now  $(E, \pi, B) = (T(M), \pi, M)$ . Hence  $\Gamma_k$  transform according to (I.5.3), or (I.3.5) in components, which means that they are coefficients' matrices of a linear connection  $\nabla$ . Since the covariant derivative assigned to  $\nabla$  acts according to (I.3.17), in which exactly the components of (11.5) (see (11.5') below) appear, and the explicit action of  $D^{\gamma}_s$  is given by (3.23) with  $\{E_i\}$  for  $\{e_i\}$ , we derive that  $D^{\gamma}_s(\hat{X}) = (\nabla_{\gamma} X)|_{\gamma(s)}$ , i.e., (11.6) holds.

The equality (11.7) is a consequence of (11.6) and (2.12) for injective path  $\gamma$ .

At this point, we want to mention a corollary from Propositions 11.3 and 11.4 which is quite important in the theory of (linear) connections and parallel transports: on the base of these propositions, the concept of a parallel transport can be defined axiomatically, independently of the connection theory, and by its means the notion of a (linear) connection can be introduced. For the purpose, Definitions 11.1 (or I.3.2) and I.3.1 should be replaced, respectively, by the following ones.

**Definition 11.2.** A parallel transport P along paths in  $(T(M), \pi, M)$  for a  $C^1$  manifold M is a linear transport along paths in  $(T(M), \pi, M)$  the coefficients' matrix of which has the representation (11.5) in some (and hence in any) frame.

**Definition 11.3.** A linear connection (resp. assigned to a parallel transport P) on a manifold M (or in  $(T(M), \pi, M)$ ) is a mapping  $\nabla$  assigning to every  $X \in \mathfrak{X}(M)$ a mapping  $\nabla_X : \mathfrak{X}^1(M) \to \mathfrak{X}^0(M)$  such that

$$(\nabla_X(Y))|_p := D_s^{\gamma}(\hat{Y}), \qquad Y \in \mathfrak{X}^1(M), \quad p \in M$$
(11.8)

 $\Box$ 

where D is the derivation along paths corresponding via (3.19) to some parallel transport in  $(T(M), \pi, M)$  (resp. to P),  $\gamma: J \to M$  is the integral path of X through  $p, \gamma(s) = p$  for some  $s \in J$ , and  $\hat{Y}: \gamma \mapsto \hat{Y}_{\gamma} := Y \circ \gamma$ .

**Exercise 11.1.** Prove that the mapping  $\nabla$ , introduced via Definition 11.3, is in fact a linear connection (restricted to the algebra of vector fields over M) according to Definition I.3.1.

**Exercise 11.2.** Prove that the parallel transport assigned to a connection  $\nabla$  via Definition 11.1 (or I.3.2) coincides with the one generating  $\nabla$  via (11.8).

Therefore we conclude that a linear connection (resp. parallel transport) according to Definition I.3.1 (resp. I.3.2) is such with respect to Definition 11.3 (resp. 11.2) and *vice versa*, i.e., both definitions are equivalent. Thus we have at our disposal two equivalent approaches to the theory of linear connections and parallel transports in which one of these objects is taken (and axiomatically defined) as a primitive one, while the other one is introduced on its base.

**Proposition 11.5.** In the representation (11.5) for the coefficient matrix of a parallel transport, the matrices  $\Gamma_k$  are the coefficients' matrices of the generating it linear connection  $\nabla$ , i.e., in components, we have

$$\Gamma^{i}_{\ i}(s;\gamma) = \Gamma^{i}_{\ ik}(\gamma(s))\dot{\gamma}^{k}(s) \tag{11.5'}$$

with  $\Gamma^{i}_{\ ik}$  being the local coefficients of  $\nabla$ .

*Proof.* In the proof of Proposition 11.4, we have demonstrated that  $\Gamma_k$  are coefficients' matrices of a linear connection  $\nabla$ . According to (3.23) and (2.3), the coefficients of the parallel transport P assigned to  $\nabla$  coincide with the components of the derivation D along paths generated by P. Combining this result with (11.6) for P and (I.3.17), we see that the coefficients of P coincide with the ones given by (11.5') for the initial parallel transport. Hence both transports are identical.  $\Box$ 

Remark 11.2. This statement does not contradict to the arbitrariness in the 3index coefficients of a linear transport along paths discussed in Section 6 because the 2-index coefficients of a parallel transport have the form (11.5') on any subset  $U \subseteq M$ , i.e., for  $\gamma: J \to U$ , with  $\Gamma^i_{\ ik}$  being the coefficients of  $\nabla$ .

**Proposition 11.6.** In an arbitrary frame  $\{E_i\}$ , the curvature and torsion of a parallel transport along paths are

$$\mathsf{R}^{\eta}(s,t) = [(R(\eta',\eta'')E_j) \otimes E^j]|_{\eta(s,t)}$$
(11.9)

$$T^{\eta}(s,t) = [T(\eta',\eta'')]|_{\eta(s,t)}$$
(11.10)

where  $\eta: J \times J' \to M$ ,  $(s,t) \in J \times J'$ ,  $E^j := (E_j)^*$  is the dual of  $E_j$ , and R and T, given via (I.3.11) and (I.3.12), are respectively the curvature and torsion of the linear connection generating the parallel transport.

*Proof.* See (9.17) and (10.7), or Propositions 9.1 and 10.1 and use Proposition 11.5, (I.3.13) and (I.3.14).  $\Box$ 

**Corollary 11.1.** A parallel transport is flat or torsionless on a submanifold N of M iff, respectively,

$$R(X,Y) = 0 \qquad \text{for } X, Y \in \text{Sec}(T(N), \pi|_N, N)$$
(11.11)

$$T(X,Y) = 0$$
 for  $X, Y \in Sec(T(N), \pi|_N, N)$  (11.12)

with R and T being respectively the curvature and torsion of the linear connection generating the initial parallel transport.

*Proof.* See Proposition 11.6 for  $\eta: J \times J' \to N$  and use the arbitrariness of  $\eta$ .  $\Box$ 

Remark 11.3 (Cf. Remarks 9.1 and 10.1). If the dimension of the submanifold N is less than the one of M, dim  $N < \dim M$ , then from R(X,Y) = 0 or T(X,Y) = 0one can not conclude that R = 0 or T = 0, respectively, as, for vector fields A, B in  $[\operatorname{Sec}(T(M), \pi, M)] \setminus [\operatorname{Sec}(T(N), \pi|_N, N)] \neq \emptyset$ , the equality R(A, B) = 0 or T(A, B) = 0 may not hold. Therefore one should make a clear distinction between flatness/torsionless of a parallel transport and of the linear connection generating it: the latter implies the former one, but the opposite is, generally, not true.

### 11.2. Normal frames for parallel transports along paths

First of all, applying Theorems 9.1 and 5.1 for a parallel transport in the tangent bundle  $(T(M), \pi, M)$  over a  $C^1$  manifold M, we obtain, respectively, the following two propositions.

**Proposition 11.7.** A parallel transport is Euclidean on a set  $U \subseteq M$ , i.e., admits frames normal along  $C^1$  paths in U, if and only if it is flat on U.

**Proposition 11.8.** A parallel transport is Euclidean on a set  $U \subseteq M$  iff it is pathindependent on U, i.e., its action along paths in U depends only on the initial and final points of the transportation, not on the particular  $C^1$  path in U connecting them.

Combining these assertions, we derive the following generalization of Proposition 11.2.

**Corollary 11.2.** A parallel transport is path-independent over a set  $U \subseteq M$  iff it is flat on U.

Alternatively, this result follows from Corollary 9.3 with  $(T(M), \pi, M)$  for  $(E, \pi, B)$  and a parallel transport for L.

**Proposition 11.9.** A parallel transport P is Euclidean on a submanifold N of a  $C^1$  manifold M if and only if the generating it linear connection  $\nabla$  is Euclidean on N.

Remark 11.4. This is untrivial result as (11.5) for  $\Gamma(s;\gamma) = 0$  with  $\gamma: J \to N$  does not imply  $\Gamma_k = 0$  if dim  $N < \dim M$ .

*Proof.* The transport P is Euclidean on N iff it is path-independent on N (Proposition 11.8) a criterion for which is the fulfillment of (II.5.1) with R being the curvature of  $\nabla$  (Lemma II.5.1 on page 112 or Corollary 11.1). By Theorem II.5.2, point (i), on page 120, the last condition is a necessary and sufficient one for the existence of frames normal for  $\nabla$  on N.

**Proposition 11.10.** If a linear connection  $\nabla$  admits frames normal on a set U, then all frames normal on U for  $\nabla$  are also normal on U for the parallel transport P assigned to  $\nabla$ . If U is a neighborhood, the opposite statement is true too, i.e., a frame normal on a neighborhood U for P is normal on U for  $\nabla$ , and hence on a neighborhood the frames normal for P and  $\nabla$  are identical.

*Proof.* The first part of the assertion follows from Propositions 11.3 and 11.4 and Definitions 4.1, 4.2 and I.5.1. The second part is a consequence of the same propositions and definitions combined with the remark that, if U is a neighborhood, in (11.5) the vector  $\dot{\gamma}(s)$  is an arbitrary vector in  $T_{\gamma(s)}(M)$  and hence  $\Gamma|_U = 0$  is equivalent to  $\Gamma_k|_U = 0$ .

The statement opposite to the first part of Proposition 11.10 is, generally not true, i.e., in the general case not all, but only some, of the frames normal for a parallel transport are also normal for the linear connection generating it. This is clearly seen from (11.5) as well as from the (un)uniqueness Proposition I.5.2 and Corollary 4.5. In fact, from these results follows that, if  $\{E_i\}$  and  $\{E'_i\}$  are frames normal on  $U \subseteq M$  for  $\nabla$  or P resp., then  $E'_i = A^j_i E_j$  where the matrix-valued function  $A := [A^j_i]$ , defined on an open set containing U or equal to it or on Uresp., is such that  $A|_U = \text{const}$  or  $(E_i(A))|_U = 0$  resp. Generally, the restriction  $A|_U = \text{const}$  is stronger than  $(E_i(A))|_U = 0$  and, consequently, the set of the solutions of the equation  $A|_U = \text{const}$  with respect to A is not greater than the one of the  $(E_i(A))|_U = 0$ .

**Proposition 11.11.** A linear transport along paths in  $(T(M), \pi, M)$  for a  $C^1$  manifold M admits frames normal for it on  $U \subseteq M$  if and only if it coincides on Uwith some parallel transport which is flat on U.

*Proof.* The sufficiency of the assertion is an obvious corollary of Proposition 11.7.

If a transport L in  $(T(M), \pi, M)$  is Euclidean on U, it is flat on U (Theorem 9.1) and the matrix of its coefficients has the form (9.11) on U in every frame over U (Proposition 6.1). Therefore, by virtue of Lemma 10.1, the transport L coincides over U with some parallel transport which is flat on U as L is such.  $\Box$ 

The last proposition demonstrates the 'privileged' place of the parallel transports amongst all linear transports along paths in the tangent bundle over a manifold: they are the only ones which may admit normal frames, a necessary and sufficient condition for this being their flatness. Before going ahead with the normal frames, we want to prove an auxiliary statement which is a generalization of Lemma II.5.2 on page 116 for transports along path in the tangent bundle  $(T(M), \pi, M)$ , which, in fact, are parallel transports in it.

**Lemma 11.1.** Let L be a linear transport along paths in  $(T(M), \pi, M)$  for a  $C^3$ manifold M, N be a submanifold of M, and L be flat and torsionless on N. Every frame  $\{E_i\}$  defined on (resp. a neighborhood of) N and L-transported on N, i.e.,  $E_i|_{\gamma(t)} = L_{s \to t}^{\gamma}(E_i|_{\gamma(s)})$  for  $\gamma: J \to N$  and  $s, t \in J$ , can be expanded (resp. redefined) outside of N in a holonomic way, i.e., for every  $p_0 \in N$  there is a chart (V, z) of M with  $V \ni p_0$  such that  $E_i|_{\overline{V}} = \frac{\partial}{\partial z^i}|_{\overline{V}}$ ,  $\overline{V} := V \cap N$ . If dim N = 0 or if N is real and dim N = 1, the condition  $\{E_i\}$  to be L-transported is not necessary.

Remark 11.5. According to Proposition 4.1, points (i) and (vi), any frame L-transported on a set U is normal on U for L and vice versa (see also Definition 4.1'').

*Proof.* Since the proof of the assertion is practically identical with the one of Lemma II.5.2 on page 116, we shall point below only the new place requiring additional study in order to be obtained the needed proof from the one of Lemma II.5.2. The notation below is the same as in the proof of Lemma II.5.2.

To prove the lemma, define  $A := [A_i^j] : U_N \to \operatorname{GL}(\dim M, \mathbb{K})$  by the expansion  $E_i|_q := A_i^j(q) \frac{\partial}{\partial x^j}|_q$ ,  $q \in U_N$ , and simply repeat the proof of Lemma II.5.2 with A for D.

The only new problem, arising in this way, is to be proved the path-independence of the integral appearing in the definition of the looked for coordinate functions  $z^i$ , i.e., the integral in (see (II.5.4))

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \left(A^{-1}(p)\right)_{j}^{i} dp^{j} + \left(A^{-1}(p_{0})\right)_{j}^{i} [x^{j}(p) - x^{j}(p_{0})] + f_{jk}^{i}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})]. \quad (11.13)$$

The conditions for this independence are  $\frac{\partial^2 z^i}{\partial x^{[a} \partial x^{b]}}\Big|_{U_N} = 0$  with a and b running from 1 to dim N as the integration is performed along some path in N. To verify these equalities, we shall proceed as follows.

Since  $\{E_i\}$  is normal on N for L (Remark 11.5), every vector field  $E_i$ , considered as a lifting of paths according to (2.8), i.e.,  $\hat{E}_i : \gamma \mapsto (\hat{E}_i)_{\gamma} : s \mapsto E_i|_{\gamma(s)}$ , satisfies the equation  $D^{\gamma}\hat{E}_i = 0$  with D being the derivation along paths generated by L (Proposition 5.6). Writing this equation in the coordinate frame  $\{\widehat{\frac{\partial}{\partial x^i}}\}$ , where  $\widehat{\frac{\partial}{\partial x^i}}$  are liftings of paths, analogous to  $\hat{E}_i$  above, and using (3.23), we get

$$\frac{\mathrm{d}A(\gamma(s))}{\mathrm{d}s} = -\Gamma(s;\gamma)A(\gamma(s))$$

as  $E_i = A_i^j \frac{\partial}{\partial x^j}$ .<sup>2</sup> Here  $\Gamma$  is the matrix of (2-index) coefficients of L in  $\{\frac{\partial}{\partial x^i}\}$ . Since  $\gamma$  is a path in N, in  $\{\frac{\partial}{\partial x^i}\}$  we have  $\dot{\gamma}^k = 0$  for  $k > \dim N$ . Therefore in  $\{\frac{\partial}{\partial x^i}\}$  is fulfilled  $\frac{dA(\gamma(s))}{ds} = \frac{\partial A}{\partial x^k}|_{\gamma(s)}\dot{\gamma}^k(s) = \frac{\partial A}{\partial x^a}|_{\gamma(s)}\dot{\gamma}^a(s)$  and  $\Gamma(s;\gamma) = \Gamma_k(\gamma(s))\dot{\gamma}^k(s) = \Gamma_a(\gamma(s))\dot{\gamma}^a(s)$ , where  $\Gamma_k$  are the matrices of the 3-index coefficients of L in  $\{\frac{\partial}{\partial x^i}\}$  (Proposition 6.1). Due to these equalities, the last displayed equation is equivalent to

$$\frac{\partial A}{\partial x^a} = -\Gamma_a A, \qquad a = 1, \dots, \dim N.$$

Now, repeating the calculation after (II.5.3) and using the last equation, we find

$$\frac{\partial^2 y^i}{\partial x^{[a} \partial x^{b]}} = \frac{\partial}{\partial x^{[a}} \left(A^{-1}\right)^i_{b]} = \dots = \left(A^{-1} \Gamma_{[a]}\right)^i_{b]} = \left(A^{-1}\right)^i_k \Gamma^k_{[ba]}$$

The proof is completed by the observation that  $\Gamma^{k}_{[ba]} \equiv 0$  is equivalent to the torsionless of L in the coordinate frame  $\left\{\frac{\partial}{\partial x^{i}}\right\}$ . In fact, by Corollary 11.1, the torsionless is equivalent to (11.12), which in  $\left\{\frac{\partial}{\partial x^{i}}\right\}$  reduces to  $T^{i}_{ab} = 0$  as  $X^{k} = Y^{k} = 0$  for  $k > \dim N$  and, consequently, (see (10.6))  $-\Gamma^{i}_{ab} + \Gamma^{i}_{ba} = 0$  since the frame  $\left\{\frac{\partial}{\partial x^{i}}\right\}$  is holonomic,  $C^{i}_{jk} = 0$ .

**Proposition 11.12.** Let  $U \subseteq M$ , L be a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$ , and L be flat on U. Then:

- (i) If some frame normal for L on U is holonomic, then the transport L is torsionless on U and coincides over U with some parallel transport which is flat and torsionless on U.
- (ii) If U is a submanifold of M and L is torsion free on U, then all frames normal on U for L are holonomic or can be redefined outside U (if U ≠ M) in such a way that the redefined frames turn to be holonomic on U.

*Proof.* (i) The result follows from Propositions 10.4 and 11.11.

(ii) Since all frames normal on U for L are L-transported on U (Remark 11.5), by Lemma 11.1 they are holonomic on U or can be redefined outside U so that the redefined frames are holonomic on U.

**Corollary 11.3.** A linear transport along paths in  $(T(M), \pi, M)$  admits holonomic frame(s) normal on a submanifold N of M if and only if it coincides over N with some parallel transport which is flat and torsionless on N.

*Proof.* See Proposition 11.12.

Thus the moral is: the flat and torsion free parallel transports are the only linear transports along paths in  $(T(M), \pi, M)$  admitting holonomic normal frames.

<sup>&</sup>lt;sup>2</sup>If one does not want to use liftings of paths, the same result follows from  $\mathsf{D}^{\gamma} E_i = 0$  with  $E_i$  considered as a basic vector field and  $\mathsf{D}^{\gamma}$  being the section-derivation corresponding to D along injective path  $\gamma$ .

#### 12. Strong normal frames

Besides, all frames normal for such transports are holonomic or can be redefined to be such.

Since all frames normal for a given flat linear transport are described via Propositions 4.3 and Corollary 4.4, in the torsionless case, the explicit description of all local coordinates normal on a submanifold is given via (see (11.13) and the proof of Lemma 11.1)

$$z^{i}(p) = a^{i} + \int_{q_{0}}^{p_{0}} \left(B^{-1}A^{-1}(p)\right)_{j}^{i} dp^{j} + \left(B^{-1}A^{-1}(p_{0})\right)_{j}^{i} \left[x^{j}(p) - x^{j}(p_{0})\right] + f_{jk}^{i}(p)[x^{j}(p) - x^{j}(p_{0})][x^{k}(p) - x^{k}(p_{0})]. \quad (11.14)$$

Here  $B \in \operatorname{GL}(\dim M, \mathbb{K})$  is a constant matrix,  $\{E_i = A_i^j \frac{\partial}{\partial x^j}\}$  is a frame normal on the submanifold, and the other notation being explained in the proofs of lemmas II.5.2 and 11.1.

## 12. Strong normal frames

Let M be a manifold and  $(T(M), \pi, M)$  the tangent bundle over it. Let  $\nabla$  and P be, respectively, a linear connection on M and the parallel transport along paths in  $(T(M), \pi, M)$  generated by  $\nabla$  (see (3.29) and the statement after it). Suppose  $\nabla$  and P admit frames normal on a set  $U \subseteq M$ . Here a natural question arises: what are the links between both types of normal frames, the ones normal for  $\nabla$  on U and the ones normal for P on U?

Recall (see Definitions I.5.1 and 4.1' and equation (11.5')), if  $\Gamma^i_{jk}$  are the coefficients of  $\nabla$  in a frame  $\{E_i\}$ , the frame  $\{E_i\}$  is normal on  $U \subseteq M$  for  $\nabla$  or P iff respectively

$$\Gamma^i_{\ ik}(p) = 0 \tag{12.1}$$

$$\Gamma^{i}_{\ j}(s;\gamma) = \Gamma^{i}_{\ jk}(\gamma(s))\dot{\gamma}^{k}(s) = 0$$
(12.2)

for all  $p \in U$ ,  $\gamma \colon J \to U$ , and  $s \in J$ .

Two simple but quite important conclusions can be made from these equalities: (i) The frames normal for  $\nabla$  are normal for P, the converse being generally not valid, which is the main contents of Proposition 11.10, and (ii) in a frame normal for  $\nabla$  vanish the 2-index as well as the 3-index coefficients of P.

**Definition 12.1.** Let P be a parallel transport in  $(T(M), \pi, M)$  and  $U \subseteq M$ . A frame  $\{E_i\}$ , defined on an open set containing U, is called strong normal on U for P if the 3-index coefficients of P in  $\{E_i\}$  vanish on U. Respectively,  $\{E_i\}$  is strong normal along  $g: Q \to M$  if it is strong normal on g(Q).

Obviously, the set of frames strong normal on U for a parallel transport P coincides with the set of frames normal for the linear connection  $\nabla$  generating P.

The above considerations can be generalized directly to linear transports for which 3-index coefficients exist and are fixed.

**Definition 12.2.** Let E and M be  $C^1$  manifolds,  $U \subseteq M$ , and  $(E, \pi, M)$  be a vector bundle over M. Let L (resp. D) be a linear transport (resp. derivation) along paths in  $(E, \pi, M)$  admitting 3-index coefficients on U which are supposed to be fixed, i.e., its coefficient matrix is of the form

$$\Gamma(s;\gamma) = \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s) \tag{12.3}$$

in every pair of frames  $\{e_i\}$  in E and  $\{E_\mu\}$  in T(M) defined on an open set containing U or equal to it, where  $\gamma: J \to U$  is of class  $C^1$  and  $\Gamma_\mu := [\Gamma^i_{\ j\mu}]$  are the (fixed) matrices of the 3-index coefficients of L. A frame  $\{e_i\}$ , defined on an open set containing U or equal to it, is called strong normal on U for L (resp. D), if in the pair ( $\{e_i\}, \{E_\mu\}$ ) for some (and hence any)  $\{E_\mu\}$  the 3-index coefficients of L vanish on U. Respectively,  $\{e_i\}$  is strong normal along  $g: Q \to M$  if it is strong normal on g(Q).

So, a frame  $\{e_i\}$  is strong normal or normal on U if (cf. (12.1) and (12.2)) respectively

$$\Gamma_{\mu}(p) = 0 \tag{12.4}$$

$$\Gamma(s;\gamma) = \Gamma_{\mu}(\gamma(s))\dot{\gamma}^{\mu}(s) = 0$$
(12.5)

for all  $p \in U$ ,  $\gamma: J \to U$ , and  $s \in J$ . From these equations, is evident, a strong normal frame is a normal one, the opposite being valid as an exception, e.g., if U is a neighborhood. This situation is identical with the one for parallel transports in  $(T(M), \pi, M)$  which is a consequence of the fact that Definition 12.2 incorporates Definition 12.1 as its obvious special case.

The main difference between the cases of parallel transports and arbitrary linear transports along paths is that for the former the condition (12.3) holds globally, i.e., for every path  $\gamma: J \to M$ , for some uniquely fixed  $\Gamma_{\mu}$ , while for the latter (12.3) is valid, generally, locally, i.e., for  $\gamma: J \to U$  with  $U \subseteq M$ , and in it  $\Gamma_{\mu}$  are fixed but are not uniquely defined by the transport and may depend on U (see Sections 6 and 11). The cause for this is that for a parallel transport, equation (12.3) on M with uniquely defined  $\Gamma_{\mu}$  follows from its definition, while if for a given linear transport L this equation holds on U for some  $\Gamma_{\mu}$ , it is also true if we replace  $\Gamma_{\mu}$  with  $\Gamma_{\mu} + G_{\mu}$  where the matrix-valued functions  $G_{\mu}$  are subjected to the condition  $G_{\mu}\dot{\gamma}^{\mu} = 0$  for every path  $\gamma$  in U. If U is an open set, then  $\dot{\gamma}(s)$ is an arbitrary vector in  $T_{\gamma(s)}(M)$ , which implies  $G_{\mu}|_{U} = 0$ , i.e., in this case the 3-index coefficients of L are unique; just this is the case with a parallel transport when U = M and its 3-index coefficients are fixed and, by definition, are equal to the coefficients of the linear connection generating it.

If in Definition 12.2 one replaces D with a derivation  $\mathcal{D}$  along tangent vector fields and (12.4) with (8.8), the definition of a frame strong normal on U for  $\mathcal{D}$ 

will be obtained. But, by Proposition 8.4, every frame normal on U for  $\mathcal{D}$  is strong normal on U for  $\mathcal{D}$  and vice versa. Therefore the concepts of a 'normal frame' and 'strong normal frame', when applied to derivations along tangent vector fields, are identical. Returning to the considerations in Section 8, we see that frames (strong) normal for a derivation along tangent vector fields are strong normal for some derivation or linear transport along paths and vice versa. For this reason, below only strong normal frames for the latter objects will be investigated.

To make the situation easier and clearer, below the following problem will be studied. Let  $(E, \pi, M)$  be a vector bundle over a  $C^1$  manifold  $M, V \subseteq M$  be an *open* subset,  $U \subseteq V$ , and L be a linear transport along paths in  $(E, \pi, M)$  whose coefficient matrix has the form (12.3) on V, i.e., for every  $C^1$  path  $\gamma: J \to V$ .<sup>1</sup> The problem to be investigated frames strong normal on U for L.

Let  $\{e_i\}$  be a frame over V in E and  $\{E_\mu\}$  a frame over V in T(M). A frame  $\{e'_i = A^j_i e_j\}$  over V in E is strong normal on  $U \subseteq V$  if for some frame  $\{E'_\mu\}$  over V in T(M) is fulfilled  $\Gamma'_{\mu|U} = 0$  with  $\Gamma'_{\mu}$  given by (6.4). Hence  $\{e'_i\}$  is strong normal on U iff the matrix-valued function  $A = [A^j_i]$  satisfies the (strong) normal frame equation

$$(\Gamma_{\mu}A + E_{\mu}(A))|_{U} = 0 \tag{12.6}$$

where  $\Gamma_{\mu}$  are the 3-index coefficients' matrices of L in  $(\{e_i\}, \{E_{\mu}\})$ . This equation, describing the matrix  $A = [A_i^j]$  which provides a transition from an arbitrary to a strong normal frame(s), is also a consequence of (5.1), (12.3), and the arbitrariness of  $\gamma: J \to V \supseteq U$ .

If on U exists a frame  $\{e_i\}$  strong normal for L, then all frames  $\{e'_i = A^j_i e_j\}$ which are normal or strong normal on U can easily be described: for the normal frames, the matrix  $A = [A^j_i]$  must be constant on U (Corollary 4.5),  $A|_U = 0$ , while for the strong normal frames it must be such that  $E_{\mu}(A)|_U = 0$  for some (every) frame  $\{E_{\mu}\}$  over U in T(M) (see Corollary III.3.1 or (12.6) with  $\Gamma_{\mu}|_U = 0$ ).<sup>2</sup>

Comparing equations (12.6) and (III.3.5) or (I.5.4), we see that they are identical with the only difference that the size of the square matrices  $\Gamma_1, \ldots, \Gamma_{\dim M}$ , and A in (III.3.5) is dim  $M \times \dim M$  while in (12.6) it is  $v \times v$ , where v is the dimension of the vector bundle  $(E, \pi, M)$ , i.e.,  $v = \dim \pi^{-1}(p)$ ,  $p \in M$ , which is generally not equal to dim M. But this difference is completely insignificant from the view-point of solving these equations (in a matrix form) or with respect to the integrability conditions for them. Therefore all of the results of Chapter III (or II) concerning the solution of the matrix differential equation (12.6) are (*mutatis mutandis*) applicable to the investigation of the frames strong normal on a set  $U \subseteq M$ .

The transferring of results from Chapter III is so trivial that their explicit reformulations makes a sense if one really needs the corresponding rigorous as-

<sup>&</sup>lt;sup>1</sup>From here follows the existence of unique 3-index coefficients of L on V which, under a change of frames, transform into (6.5). We suppose the 3-index coefficients of L on U to be fixed and equal to the ones on V when restricted to U.

<sup>&</sup>lt;sup>2</sup>This conclusions agree with the discussion after the proof of Proposition 11.10.

sertions for some concrete purpose. For this reason, we describe below briefly the general situation and one its corollary.

The only peculiarity one must have in mind, when such transferring is carried out, consist in the observation that in this way can be obtained, generally, only *part* of the frames normal for some linear transport, viz. the frames strong normal for it. But such a state of affairs is not a trouble as we need a single normal frame to construct all of them by means of Corollary 4.5.

Let  $n \in \mathbb{N}$ ,  $J^n$  be a neighborhood in  $\mathbb{R}^n$ , and  $\gamma_n \colon J^n \to M$  be a  $C^1$  locally injective mapping. Then, from Theorem III.8.1 and Proposition III.8.1, we derive the following theorem.

**Theorem 12.1.** A necessary and sufficient condition for the existence of frame(s) strong normal on  $\gamma_n(J^n)$  for some linear transport along paths or derivation along paths or along vector fields tangent to M, is its (3-index) coefficients to satisfy, in some neighborhood (in  $\mathbb{R}^n$ ) of every  $s \in J^n$ , the equations

$$\left(R_{\mu\nu}(-\Gamma_1 \circ \gamma_n, \dots, -\Gamma_{\dim M} \circ \gamma_n)\right)(s) = 0, \qquad \mu, \nu = 1, \dots, n$$
(12.7)

where  $R_{\mu\nu}$  are given via (6.7) for  $x^{\mu} = s^{\mu}$ ,  $\mu, \nu = 1, ..., n$  with  $\{s^{\mu}\}$  being Cartesian coordinates in  $\mathbb{R}^{n}$ .

It is almost evident, in the coordinates used, equation (12.7) is identical with (6.16) for  $N = \gamma_n(J^n)$  and  $p = \gamma_n(s)$ . Thus, on a submanifold or along locally injective mappings, the existence of normal frames (for linear transports of the considered type) implies the existence of strong normal frames.

From (12.7), an immediate observation follows: strong normal frames always exist at every point (n = 0) or/and along every locally injective path (n = 1). Besides, these are the *only cases* when strong normal frames *always exist* because for them (12.7) is identically valid. On submanifolds with dimension greater than or equal to two normal frames exist only as an exception if (and only if) (12.7) holds. Notice, equations (12.7) express the flatness of the corresponding linear transports (see Section 9, equation (9.14), or Corollary 9.1) or, if  $n = \dim M$ , derivations (see Section III.2).

# 13. Linear transports assigned to derivations in tangent bundles

Since the linear connections are a special kind of derivations along vector fields (see Section III.2), one can expect a possible existence of linear transports along paths whose relation to the derivations is similar to the one between parallel transports and linear connections (see Subsection 11.1). Below we shall see that such an expectation has a firm background. The aim of the present section is to be clarified the links between the different derivations of the algebra of vector fields over a manifold M and the linear transports along paths in the tangent bundle

 $(T(M), \pi, M)$  over M. The next ideas and results have a natural generalization in the case of arbitrary tensor bundles (or the algebraic tensor bundle) over Mwhich we leave to the reader.

#### 13.1. Derivations along paths

Since the Definition 11.1 on page 281 of a parallel transport employs only the covariant derivative along paths assigned to a linear connection, not directly the connection, it can *mutatis mutandis* be transferred to arbitrary derivations along paths. On this base most of the material in Section 11 could be suitably generalized. Here are some details of this procedure.

Let M be a  $C^1$  manifold, D be a derivation along (injective  $C^1$ ) paths of the tensor algebra over M (see Definition III.10.1 on page 191), and  $\gamma: J \to M$  be injective and of class  $C^1$ . Define a mapping L on the set  $P^1(M)$  of  $C^1$  paths in  $M, L: \gamma \mapsto L^{\gamma}$ , such that

$$L^{\gamma} \colon (s,t) \mapsto L^{\gamma}_{s \to t} \colon T_{\gamma(s)}(M) \to T_{\gamma(t)}(M)$$
(13.1)

for  $s, t \in J$  and

$$L_{s \to t}^{\gamma}(X_0) = X_{\gamma(t)}, \qquad X_0 \in T_{\gamma(s)}(M)$$
 (13.2)

where the vector field X over  $\gamma(J)$  is the unique solution of the initial-value problem

$$\mathsf{D}^{\gamma}X = 0 \tag{13.3a}$$

$$X_{\gamma(s)} = X_0. \tag{13.3b}$$

**Proposition 13.1.** The mapping  $L: \gamma \to L^{\gamma}$ , defined via (13.1)–(13.3), is a linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  over the manifold M.

*Proof.* Take into account (III.10.7) and repeat the proof of Proposition 11.1 with L for P and  $\mathsf{D}^{\gamma}$  for  $\nabla_{\dot{\gamma}}$ .

**Definition 13.1.** The linear transport L along paths in  $(T(M), \pi, M)$ , defined via equations (13.1)–(13.3), will be called *transport assigned to or generated by the derivation* D *along paths* (of  $T^{1}(M)$  or of T(M)).

On the base of (13.1)–(13.3), one can prove a number of properties of the transport L assigned to D, e.g., Proposition 11.4 holds with L for P, D for the linear connection  $\nabla$ , and  $D^{\gamma}$  for  $\nabla_{\dot{\gamma}}$ . But such efforts, for deriving the properties of L using only (13.1)–(13.3), are practically completely needless and the result of them will be only new ways for derivation or proving the properties of the general linear transports in  $(T(M), \pi, M)$ . The cause for this situation is that the afore-presented definitions of a linear transport (along injective  $C^1$  paths) assigned to a derivation along paths is simply an equivalent way for defining what a linear transport (along injective  $C^1$  paths) in the tangent bundle is.

The above assertion can be proved, most easily, in some local frame  $\{E_i\}$  on (a neighborhood of)  $\gamma(J)$  for an arbitrary injective  $C^1$  path  $\gamma: J \to M$ .

Comparing the local expansions (III.10.7) on page 192 for (r,q) = (1,0)and (2.13) on page 221 for  $e_i(s;\gamma) = E_i|_{\gamma(s)}$  and  $\sigma = K$ , we see that they are identical for every vector field K defined on  $\gamma(J)$  (or on a larger set). Therefore a derivation along paths according to Definition III.10.1, when restricted to vector fields, is in fact a section-derivation along paths in the tangent bundle, according to the definition of the latter one on page 221, assigned to a derivation D along paths in  $(T(M), \pi, M)$  (see Definition 2.1) whose local components  $\Gamma_j^i$  along  $\gamma$  coincide with the functions  $\Gamma_j^i(s;\gamma)$  appearing in (III.10.7). For this reason, further the derivations along paths of the algebra of  $C^1$  vector fields over M will be called vector-derivations along paths. Thus we have proved the following result.

**Proposition 13.2.** For every vector-derivation D along paths of the algebra of vector fields over a manifold M, there exists a unique derivation D along paths in the tangent bundle  $(T(M), \pi, M)$  over M which generates it via (2.12). In a local frame, D is explicitly defined via its local components which coincide with the ones of D.

Remark 13.1. The components of D are defined along injective paths, but this is insignificant for D which is defined along arbitrary, with or without self-intersections, paths.

Denoting, in some frame  $\{E_i\}$ , by  $L(t, s; \gamma)$  the matrix of the transport L assigned to a vector-derivation D along paths, from equations (13.2), (13.3), and (III.10.7), we derive that it is the unique solution of the initial-value problem

$$\frac{\partial \boldsymbol{L}(t,s;\gamma)}{\partial t} = -\boldsymbol{\Gamma}(t;\gamma)\boldsymbol{L}(t,s;\gamma)$$
(13.4a)

$$\boldsymbol{L}(s,s;\gamma) = \mathbb{1} \tag{13.4b}$$

where  $s, t \in J$  and  $\Gamma := [\Gamma_j^i]$  is the matrix of the components of D (or of D) in  $\{E_i\}$ . So the explicit form of L is

$$\boldsymbol{L}(t,s;\gamma) = Y(t,s_0;-\boldsymbol{\Gamma}(\cdot;\gamma))Y^{-1}(s,s_0;-\boldsymbol{\Gamma}(\cdot;\gamma))$$
(13.5)

where Y is defined via (3.28) and  $s_0 \in J$  is fixed.

At last, noticing that (13.5) is identical with (3.27), we conclude from Proposition 3.6 that L is identical with the linear transport along paths generating Dvia (3.19) and whose coefficients coincide with the components of D (or of D). This result completes the proof of our statement that a linear transport along paths assigned to a vector-derivation is simply a linear transport along injective  $C^1$  paths in the tangent bundle and *vice versa*.

#### **13.2.** Derivations along vector fields

From Section III.10, we know that to every derivation along vector fields whose matrix in some (and hence in every – see (III.2.11)) frame  $\{E_i\}$  satisfies the condition

$$\Gamma_X(\gamma(s))|_{X_{\gamma(s)}=\dot{\gamma}(s)} = \Gamma(s;\gamma)$$
(13.6)

for every (injective)  $C^1$  path  $\gamma: J \to M$  and  $s \in J$ , there corresponds a vectorderivation D along paths in  $(T(M), \pi, M)$  such that (see (III.10.2))

$$(\mathsf{D}^{\gamma}V)(\gamma(s)) = \left(\frac{\mathrm{d}V^{i}(\gamma(s))}{\mathrm{d}s} + \Gamma^{i}{}_{j}(s;\gamma)V^{j}_{\gamma(s)}\right)E_{i}|_{\gamma(s)}$$
(13.7)

for  $V \in \mathfrak{X}^1(M)$ .

**Example 13.1.** A kind of vector-derivation is the covariant derivative assigned to a linear connection  $\nabla$  for which

$$\Gamma_X(p) = \Gamma_k(p) X_p^k, \tag{13.8}$$

where  $p \in M$  and  $\Gamma_k$  are the coefficients' matrices of  $\nabla$ . In this case, obviously, equation (13.6) holds with

$$\Gamma(s;\gamma) = \Gamma_k(\gamma(s))\dot{\gamma}^k(s).$$
(13.9)

Since to a vector-derivation along paths in the tangent bundle  $(T(M), \pi, M)$ there corresponds unique linear transport along paths in the same bundle (see Subsection 13.1), to every derivation D along vector fields of T(M), whose components' matrix satisfies the condition (13.6), there corresponds a linear transport along paths generating the assigned to D vector-derivation D via (3.19) and (2.12).

So, the general scheme for generating a linear transport from a derivation along vector fields is

derivation along vector fields satisfying (13.6)

 $\rightarrow$  vector-derivation along paths

 $\rightarrow$  linear transport along paths. (13.10)

A typical example, realizing this procedure, is the parallel transport assigned to a linear connection. A modification of the scheme (13.10) for derivations along fixed vector fields will be developed in the next subsection.

#### 13.3. Derivations along fixed vector field

In the present subsection, we want to apply (possibly a modification of) the scheme (13.10) to a derivation  $D_X$  along a *fixed* vector field X of the tensor algebra over a manifold M. Of course, such a derivation can not define a linear

transport along arbitrary  $(C^1)$  paths in  $(T(M), \pi, M)$  since X is fixed, not arbitrary, vector field and hence (13.6) cannot hold for arbitrary path  $\gamma$ .<sup>1</sup> As one could expect, the transports corresponding to  $D_X$  are along the integral paths of X. Except this peculiarity, the other general considerations in Subsection 13.2 preceding (13.10) remain unchanged. Here are some details of the procedure assigning a linear transport to a derivation along a fixed vector field.

Let  $X \in \mathfrak{X}^1(M)$  be fixed and non-singular,  $X \neq 0$ . Let  $\gamma_p: J \to M$  be the integral path of X through  $p \in M$ , i.e.,  $\gamma_p(s_0) = p$  for some  $s_0 \in J$  and  $\dot{\gamma}_p(s) = X_{\gamma_p(s)}$  for all  $s \in J$ . Suppose  $D_X$  is a derivation along X such that the matrix of its components  $\Gamma_X$  (in some and hence in any frame) satisfies the condition (13.6) with  $\gamma = \gamma_p$  for every  $p \in M$ . Then, for every integral path  $\gamma$ of X, it is defined a unique derivation  $D^{\gamma}$  along  $\gamma$ , explicitly given via (13.7). (Note,  $D^{\gamma}$  is defined only along integral paths  $\gamma$  of X!) Repeating the proof of Proposition 13.1 with  $\gamma$  being an integral path of X, we see that (13.1)–(13.3) define a unique linear transport L along the integral paths of X in  $(T(M), \pi, M)$ . In a local frame, the matrix of L satisfies the initial-value problem (13.4) and its explicit form is (13.5).

Thus the conclusion is: to every derivation along a fixed vector field X (which is  $C^1$  and non-singular) of the tensor algebra over a manifold M, there corresponds a unique linear transport along the integral paths of X in the tangent bundle over M (provided (13.6) holds with an integral path of X for  $\gamma$ ). Therefore, we have the following modification of (13.10):

derivation along fixed vector field X

 $\rightarrow$  vector-derivation along the integral paths of X

 $\rightarrow$  linear transport along the integral paths of X. (13.11)

In this scheme, we intentionally dropped the condition (13.6). The reason is that it works independently of (13.6): if (13.6) is not valid, the resulting derivation and linear transport along the integral paths of X could depend also on the vector field X in a neighborhood of the integral paths of X, not only on the values of X on them. In other words, if (13.6) does not hold, the corresponding derivation and linear transport along a particular integral path may depend not only on it but also on a congruence of the integral paths of X in a neighborhood of the one along which the derivation and transport act. This situation is not an exceptional one, on the opposite, it is often met in concrete applications, first of all in the theoretical physics.<sup>2</sup> Below we shall present a list of important examples of this kind.

<sup>&</sup>lt;sup>1</sup>For a fixed X, the equality  $X_{\gamma(s)} = \dot{\gamma}(s)$  cannot hold for arbitrary  $\gamma$  as it defines the integral paths of X – see (I.2.19).

<sup>&</sup>lt;sup>2</sup>In the physically-oriented literature, the situation is 'inverse' with respect to X: there is given a congruence of (injective,  $C^1$ , and regular) paths and X is defined as the vector field tangent to the paths of that congruence.

Recall (see Definition III.2.3), for every vector field X and derivation D along vector fields, the decomposition  $D_X = \mathcal{L}_X + S_X$ , with  $D: X \to D_X$  and  $\mathcal{L}_X$  being the Lie derivative along X, holds for some tensor field  $S_X$  of type (1, 1). Since for a linear connection  $\nabla$  this equality holds for

$$S_X = \Sigma_X, \quad \Sigma_X(Y) := \nabla_X Y - [X, Y]_{-} = \nabla_Y X + T(X, Y), \quad Y \in \mathfrak{X}(M),$$
(13.12)

with T being the torsion of  $\nabla$ , for every derivation D along vector fields is fulfilled

$$D_X = \nabla_X + S_X - \Sigma_X \tag{13.13}$$

for (arbitrarily chosen) linear connection  $\nabla$  and some  $S_X \in \mathfrak{T}_1^1(M)$ . Applying the last decomposition and the explicit local expression (III.2.2), one can prove the following statements.

**Example 13.2.** Let M be a manifold endowed with a linear connection  $\nabla$ . To  $\nabla$  corresponds a decomposition (13.13) with

$$S_X = \Sigma_X \tag{13.14a}$$

where X can be fixed, as well as arbitrary, and the condition (13.6) is valid.

**Example 13.3.** Let M be Einstein-Cartan manifold, i.e., one endowed with (generally said independent) Riemannian metric g and linear connection  $\nabla$ . The Fermi–Walker derivative [62, 119] is a one for which (13.13) holds with

$$S_X = \Sigma_X - 2Q_X. \tag{13.14b}$$

Here X is fixed, (time-like,)  $C^1$ , and unit vector field<sup>3</sup> and  $Q_X \in \mathfrak{T}_1^1(M)$  with

$$g_{im}(Q_X)_j^m := \left\{ h_i^k h_j^l(g(\,\cdot\,,\nabla_{E_k}X))_l + h_i^m g_{ml} T_{jk}^l X^k + (g(\,\cdot\,,\nabla_{E_j}X))_i \right\}_{[ij]}$$

where  $h_i^k := \delta_i^k - X^k(g(\cdot, X))_i/g(X, X)$  and  $T_{jk}^i$  are the components of the torsion tensor of  $\nabla$ .

**Example 13.4.** On a Riemannian manifold, the Fermi derivative assigned to its metric g [62] corresponds to the choice

$$S_X = \Sigma_X^{\{\}} - 2\overline{Q}_X \tag{13.14c}$$

where X is a fixed, (time-like,)  $C^1$  and unit vector field, the index {} means that the connection is with respect to the Christoffel symbols (I.4.14) formed from g,  $\overline{Q}_X \in \mathfrak{T}_1^1(M)$ , and  $(\overline{Q}_X)_j^i := X^i(g(\cdot, \nabla_X^{\{\}}X))_j$ .

<sup>&</sup>lt;sup>3</sup>The choice g(X, X) = +1 or g(X, X) = -1 depends on the accepted signature of g: the former (resp. latter) one corresponds to a positive (resp. negative) 'time' eigenvalue of g in a diagonal form.

**Example 13.5.** On a Riemannian manifold, the Truesdell derivative assigned to its metric g [120, 121] is obtained from (13.13) for<sup>4</sup>

$$S_X = \theta \cdot \delta. \tag{13.14d}$$

Here X is a fixed,  $C^1$  and null or unit vector field (which could be time-like or space-like),  $\theta := \sum_i (\nabla_{E_i} X)^i$  is the expansion of X, and  $\delta$  is the unit tensor with local components  $\delta_i^j$  (in any frame).

**Example 13.6.** On a Riemannian manifold, the Jaumann derivative assigned to its metric g [122] is described via (13.13) with

$$S_X = \Sigma_X^{\{\}} - \omega. \tag{13.14e}$$

Here X is a fixed  $C^1$  and null or unit vector field (which could be time-like or space-like),  $\omega \in \mathfrak{T}_1^1(M)$  and<sup>5</sup>

$$g_{ik}\omega_{j}^{k} := \left\{ (g(\,\cdot\,,\nabla_{E_{j}}^{\{\}}X))_{i} - (g(\,\cdot\,,\nabla_{\nabla_{E_{j}}^{\{\}}X}^{\{\}}X))_{i} \right\}_{[ij]}$$

The above list of particular derivations (along, possibly, fixed vector fields) obtained via (13.13) can easily be completed by the Lie derivative [19, 89, 119], modified Fermi-Walker and Frenet-Serret derivatives [98], etc.

Excluding the case of linear connections, all of the above-mentioned derivatives are along a fixed vector field and, generally, the condition (13.6) is not satisfied for them. Consequently, the derivations and linear transports along the integral paths of X corresponding to them depend, generally, on a congruence of integral paths, not only on the particular integral path along which they act. Regardless of this, these derivations and the corresponding transports find a number of important applications.

**Exercise 13.1.** Following the implications in (13.10) (or in (13.11)), prove that from (13.13) and (13.14a)-(13.14e) can be obtained respectively: the parallel transport assigned to a linear connection (covariant derivative) of the tensor algebra of a manifold [11, 19], Fermi-Walker transport [62, 119], Fermi transport [62], Truesdell transport [120, 121], and Jaumann transport [122]. This list of transports can be completed with the Lie transport [19, 119], modified Fermi-Walker and Frenet-Serret transports [98], etc.

#### **13.4.** Normal frames

In Subsections 13.1–13.3, it was demonstrated how to different derivations of the tensor algebra over a manifold can be assigned linear transports along paths in

<sup>&</sup>lt;sup>4</sup>The choice of  $\nabla$  is insignificant as  $\nabla_X - \Sigma_X = \mathcal{L}_X$ .

<sup>&</sup>lt;sup>5</sup>The sign before the second term in the right-hand side of the next equality must be opposite to the one of the eigenvalue of the 'time' term of g in a diagonal form.

the tangent bundle over it. Hence to every such derivation can be assigned two kinds of normal frames: the ones corresponding to it as a derivation and the ones corresponding to the linear transports generated by the derivations. These frames were introduced and investigated respectively in Chapter III and in the present one. Since a simple observation reveals that the derivations' components coincide with the coefficients of the linear transports generated by the derivations, one can expect the existence of some links between both sorts of normal frames. More precisely, relying on the considerations in Section 12, we can assert that the frames, normal for derivations along paths or (arbitrary) vector fields, are strong normal for the linear transports generated by these derivations. What concerns the frames normal for derivations along a fixed vector field, it is easy to be proved that they are also normal for the linear transports (along the integral paths of the vector field) generated by this kind of derivations and vice versa. The opposite link is partially valid in a sense that if a linear transport generated by a derivation admits normal frames, then some (or all) of them, viz. the strong normal ones, are normal for the corresponding derivation generating the initial transport.

In this way, we have came to the following important

**Conclusion 13.1.** All of the results, concerning normal frames and obtained in Chapter III for different derivations of the tensor algebra over a manifold, are *mutatis mutandis* valid for the frames (strong) normal for the (corresponding) linear transports along paths in the tangent bundle over the same manifold.

The afore-presented conclusions can also be confirmed by the next considerations some of which are valid for general vector bundles with a manifold as a base, not only for the tangent bundle over a manifold.

From Proposition 6.1 and Theorem 6.2, we know that only linear transports/derivations along paths with (2-index) coefficients given by (6.1) admit normal frames. Besides, from equations (6.1) and (6.4), it follows that frames normal on a subset  $U \subseteq M$  for such transports/derivations along paths exist if and only if the matrix differential equation (5.1), or, equivalently,

$$\left[\dot{\gamma}^{\mu}(\Gamma_{\mu}A + E_{\mu}(A))\right]\Big|_{U} = 0, \qquad (13.15)$$

has a solution for every  $\gamma: J \to U$  with respect to  $A.^6$  In fact, the equations (6.16) are the integrability conditions for (13.15).<sup>7</sup> Evidently, the same is the situation with derivations along tangent vector fields (see Section 8) when, due to (8.6), such a derivation admits frames normal on U iff the equation

$$\left(\mathbf{\Gamma}_X A + X(A)\right)\Big|_U = 0, \tag{13.16}$$

<sup>&</sup>lt;sup>6</sup>If such A exist in a frame  $\{e_i\}$ , then the frame  $\{e'_i = A^j_i e_j\}$  is normal on U and vice versa; see (6.4).

<sup>&</sup>lt;sup>7</sup>If (6.6) hold and U is a neighborhood, then  $A = Y(p, p_0; -\Gamma_1, \ldots, -\Gamma_{\dim M})A_0$ ,  $A_0$  being non-degenerate matrix.

 $\Gamma_X$  being the derivation's matrix along a vector field X, has a solution with respect to A. As we proved in Section 8, if X is arbitrary and tangent to the paths in U, this equation is equivalent to (13.15) with  $\Gamma_{\mu}$  being the matrices of the 3-index coefficients of the derivation; if X is completely arbitrary, (13.16) is equivalent to equation (13.17) below.

Now it is time to recall that, from mathematical view-point, the material of Chapter III is actually devoted precisely to the solution of the normal frame equation

$$(\Gamma_{\mu}A + E_{\mu}(A))|_{U} = 0$$
 (13.17)

which is equivalent to (13.15) if U is a neighborhood. The fact that in Chapter III are studied frames normal for derivations of the tensor algebra over a manifold M is inessential because the equations describing the matrices by means of which is performed the transformation from an arbitrary frame to a (strong) normal one are the same in this chapter and in Chapter III. The only difference is what objects are transformed by means of the matrices satisfying (13.16): in the present chapter these are the frames in the restricted bundle space  $\pi^{-1}(U) \subseteq E$ , while in Chapter III they are the tensor bases over U, in particular the ones in the bundle tangent to M. In Chapter III, the only explicit use of the derivations of the tensor algebra over M was to define their components (2-index coefficients) and the transformation law for the latter. Since this law (see (III.2.11) and (III.10.9)) is identical with (8.6),<sup>8</sup> all results concerning the 2- and 3-index coefficients of derivations of the tensor algebra over M and the ones of derivations along tangent vectors in vector bundle  $(E, \pi, M)$  coincide.

Thus, we have came to the following very important conclusion.

**Conclusion 13.2.** All of the results of Chapter III, concerning derivations along vector fields, their components, and frames normal for them, are *mutatis mutandis* valid (as investigated in the present chapter) for linear transports along paths, derivations along paths or along tangent vector fields, their coefficients (or components), and the frames (strong) normal for them in vector bundles with a differentiable manifold as a base.

The only change, if required, to transfer the results is to replace the term 'derivation along vector fields' with 'derivation along tangent vector fields', or 'derivation along paths', or 'linear transport along paths' and, possibly, the term 'normal frame' with 'strong normal frame'.

Because of the widespread usage of covariant derivatives (linear connections), we want to mention them separately regardless of the fact that this case was completely covered in Chapters II and III. As a consequence of (3.29), the covariant derivatives are derivations linear on the whole base M (as well as on any its subset). Thus for them the condition (6.1) is identically satisfied. Therefore, by

 $<sup>^{8}</sup>$ The transformation laws (3.26) and (6.4) can be considered, under certain conditions, as special cases of (8.6).

Theorem 6.2, a covariant derivative (or the corresponding parallel transport) admits normal frames on a submanifold  $U \subseteq M$  iff (6.16) holds on U. Consequently, every covariant derivative admits normal frames at every point or along given (smooth locally injective) path. However, only the flat covariant derivatives on Uadmit frames normal on U if U is a neighborhood (dim  $U = \dim M$ ).

In the theoretical physics, we find applications of a number of linear transports along paths: parallel [11, 19], Fermi-Walker [62, 119], Fermi [62], Truesdell [120, 121], Jaumann [122], Lie [19, 119], modified Fermi-Walker and Frenet-Serret [98], etc. Our results are fully applicable to all of them, in particular for all of them there exist frames normal at a given point or/and along (smooth locally injective) paths.

# 14. Links with the theory of connections and parallel transports

The goal of this section is twofold: on one hand, we would like to clarify the relations between the linear transports along paths and parallel transports in general vector bundles and, on another hand, to transfer the results obtained for connections and parallel transports in vector bundles, thus generalizing the material from Sections 11–13 concerning mainly the tangent bundle over a manifold. A part of these problems will be generalized further in Chapter V from the view-point of general connection theory on differentiable bundles.

The book [23] will be used consistently for reference purposes below. This is due to the fact that in it is elaborated an axiomatical approach to the concept of a parallel transport in vector bundles (whose bundle and base spaces are  $C^{\infty}$ manifolds) and on this base the connection theory is studied. Such an exposition perfectly matches our aims, as pointed at the beginning of Section 3.

In the present section  $(E, \pi, M)$  will denote a vector bundle whose bundle and base spaces, E and M, are  $C^{\infty}$  manifolds, N stands for a  $C^{\infty}$  manifold, and  $g: N \to M$  is supposed to be of class  $C^{\infty}$ . In general, for the consistency with [23], all manifolds and mappings between them are supposed to be of class  $C^{\infty}$  in this section.<sup>1</sup>

# 14.1. Parallelism structures, connections and covariant derivatives

In this subsection the (axiomatical) definition of a parallel transport and its links with the concepts of connection and covariant derivative from [23] are reproduced

<sup>&</sup>lt;sup>1</sup>Such a supposition is too strong. For the most of the material that follows, smoothness of class  $C^2$  (and sometimes  $C^1$ ) is sufficient. (Smoothness of class  $C^3$  is required if normal coordinates, if any, are concerned.)

and briefly reviewed. We have chosen the pointed definition from [23] for two reasons: (i) in the sense of [23, Theorems 2.28 and 2.33] it provides an equivalent description of the notion of a connection on vector bundles, and (ii) in the available to the author literature, the book [23] provides the most advanced and well-developed axiomatical approach to the notion of a parallel transport (in vector bundles).

**Definition 14.1 (cf.** [23, Definition 2.7]). Let  $(E, \pi, M)$ , E and M being  $C^{\infty}$  manifolds, be a vector bundle and  $\beta \colon [a, b] \to M$ ,  $a, b \in \mathbb{R}$  with  $a \leq b$ , be a  $C^{\infty}$  path in M. A mapping  $\mathbb{P}$  associating to each pair  $(u, \beta)$ ,  $u \in \pi^{-1}(\beta(a))$ , a (unique)  $C^{\infty}$  lifting of  $\beta$  from M to E is called a *parallelism structure* (or a *system of parallel transport*) in  $(E, \pi, M)$  if the following five conditions are satisfied:

(i) Existence. The value  $\mathbb{P}_{u}^{\beta}$  of  $\mathbb{P}$  at  $(u,\beta)$ ,  $\mathbb{P}$ :  $(u,\beta) \mapsto \mathbb{P}_{u}^{\beta}$ , is such that

$$\mathbb{P}_u^\beta(a) = u \tag{14.1}$$

(and, by definition,  $\mathbb{P}_u: \beta \mapsto \mathbb{P}_u^{\beta}: [a, b] \to E$  with  $\pi \circ \mathbb{P}_u^{\beta} = \beta$ ), i.e.,  $\mathbb{P}_u^{\beta}$  is a lifting of  $\beta$  through u starting exactly from the point  $u \in \pi^{-1}(\beta(a))$ .

(ii) Linearity and invertability. The mapping

$$\mathsf{P}^{\beta} \colon \pi^{-1}(\beta(a)) \to \pi^{-1}(\beta(b)) \tag{14.2}$$

defined by

$$\mathsf{P}^{\beta}(u) := \mathbb{P}^{\beta}_{u}(b) \tag{14.3}$$

for every  $u \in \pi^{-1}(\beta(a))$  is a vector space isomorphism. It is called *parallel* transport along  $\beta \colon [a,b] \to M$  (assigned to  $\mathbb{P}$ ). The parallel transport along the inverse path  $\beta^- \colon [a,b] \to M$ ,  $\beta^- \colon s \mapsto \beta^-(s) \coloneqq \beta(a+b-s)$  for each  $s \in [a,b]$ , is

$$\mathsf{P}^{\beta^{-}} = \left(\mathsf{P}^{\beta}\right)^{-1}.\tag{14.4}$$

(iii) Parameterization independence. Let  $\varphi \colon [c,d] \to [a,b]$  be  $C^{\infty}$  function such that  $\varphi(c) = a$  and  $\varphi(d) = b$ . Then

$$\mathbb{P}_{u}^{\beta \circ \varphi} = \mathbb{P}_{u}^{\beta} \circ \varphi. \tag{14.5}$$

(iv)  $C^{\infty}$  dependence on initial conditions. For every open set  $U \subseteq M$  and each  $C^{\infty}$  mapping  $f: T(U) \to M$  such that  $f(0_p) = p, p \in U$ , with  $0_p$  being the zero vector in  $T_p(U) = T_p(M)$ , the mapping

$$\tilde{f}: T(U) \times \pi^{-1}(U) \to E, \quad \tilde{f}: (X, u) \mapsto \mathbb{P}_u^{\alpha}(1), \qquad \alpha(s) := f(sX)$$
(14.6)

where  $(X, u) \in T(U) \times \pi^{-1}(U)$  and  $s \in [0, 1]$ , is of class  $C^{\infty}$ .

(v) Initial uniqueness. If  $p \in M$  and the  $C^{\infty}$  paths  $\alpha, \beta \colon [0,1] \to M$  are such that  $\alpha(0) = \beta(0) = p$  and  $\dot{\alpha}(0) = \dot{\beta}(0)$ , then for each  $u \in \pi^{-1}(p)$  the tangent vectors to the paths  $\mathbb{P}_{u}^{\alpha}, \mathbb{P}_{u}^{\beta} \colon [0,1] \to E$  at the point s = 0 coincide:

$$\dot{\widehat{\mathbb{P}_u^{\alpha}}}(0) = \dot{\widehat{\mathbb{P}_u^{\beta}}}(0).$$
(14.7)

<sup>&</sup>lt;sup>2</sup>The tangent vectors  $\dot{\alpha}(0)$  and  $\dot{\beta}(0)$  are considered as defined by one-sided derivative operators – see (I.2.3).

The parallelism structures are an equivalent way for describing general connections on vector bundles. To formulate this statement in a rigorous way (see Theorem 14.1 below), we have to present some preliminary material.

**Definition 14.2.** A lifting (or lift)  $\bar{\beta}: [a, b] \to E$  of a  $C^{\infty}$  path  $\beta: [a, b] \to M$ ,  $\pi \circ \bar{\beta} = \beta$ , is said to be parallel with respect to a parallelism structure  $\mathbb{P}$  in a vector bundle  $(E, \pi, M)$  if  $\bar{\beta} = \mathbb{P}^{\beta}_{\bar{\beta}(a)}$ ; respectively, the paths  $\mathbb{P}^{\beta}_{u}$ , with  $u \in \pi^{-1}(\beta(a))$  and  $\beta: [a, b] \to M$ , in E are called parallel (with respect to  $\mathbb{P}$ ). A section  $\sigma \in \text{Sec}(E, \pi, M)$  is parallel along  $\beta$  (with respect to  $\mathbb{P}$ ) if the lifting  $\hat{\beta} = \sigma \circ \beta$  is parallel (with respect to  $\mathbb{P}$ ).

**Definition 14.3** (cf. [23, Definition 2.26]). A connection on vector bundle  $(E, \pi, M)$  is a  $(\dim M)$ -dimensional distribution  $T^h(E): E \to T(E)$  such that:

(i) For each  $u \in E$  the value  $T_u^h(E)$  of  $T^h(E)$  at  $u, T^h(E): u \mapsto T_u^h(E)$ , is in the tangent space  $T_u(E)$  and is (direct) complement to its vertical component  $T_u^v(E) := T_u(\pi^{-1}(\pi(u)))$ , i.e., to the space tangent to the fibre  $\pi^{-1}(\pi(u))$  through u:

$$T_u(E) =: T_u^h(E) \oplus T_u^v(E). \tag{14.8}$$

(ii) The spaces  $T_u^h(E)$  are homogeneous in a sense that

$$\mu_{c*}(T_u^h(E)) = T_{\mu_c(u)}^h(E) = T_{cu}^h(E), \qquad c \in \mathbb{K}$$
(14.9)

where  $\mu_c \colon E \to E$  with  $\mu_c \colon u \mapsto cu$  and  $\mu_{c*} \coloneqq (\mu_c)_*$  is the differential of  $\mu_c$ .

Let us set  $T^v(E) := \bigcup_{u \in E} T^v_u(E)$ . By abuse of the notation, we put  $T^h(E)$  to denote also the union  $\bigcup_{u \in E} T^h_u(E)$ . Thus, we have the direct decomposition

$$T(E) = T^{h}(E) \oplus T^{v}(E).$$
(14.10)

Remark 14.1. A connection on the tangent bundle  $(T(M), \pi_T, M)$  over M is called (linear) connection on M. For some details, see below Proposition 14.5 and Remark 14.3.

**Definition 14.4.** The spaces  $T_u^h(E)$  and  $T_u^v(E)$  are called *horizontal* and *vertical*, respectively, space (tangent to E at u); the vectors of these spaces are called respectively *horizontal and vertical vectors*.

**Definition 14.5** (cf. [23, Definition 2.27]). Let  $T^h(E)$  be a connection on  $(E, \pi, M)$ . A lifting  $\overline{f} \colon N \to E$  of a  $C^{\infty}$  mapping  $f \colon N \to M$ , N being a manifold, is horizontal with respect to  $T^h(E)$  (or is  $T^h(E)$ -horizontal) if  $\overline{f}_*(T_p(N)) \subseteq T^h_{\overline{f}(p)}(E)$ for every  $p \in N$ , i.e., the induced tangent mapping  $\overline{f}_*$  of  $\overline{f}$  sends the vectors tangent to N at p to horizontal vectors (tangent) at  $\overline{f}(p)$ . In particular, a path  $\overline{\beta}$  in E, which is a lifting of the path  $\pi \circ \overline{\beta}$  in M, is horizontal if its tangent vector field is a path in  $T^h(E)$ . **Definition 14.6.** If  $T^{h}(E)$  is a connection on  $(E, \pi, M)$ , the  $(T^{h}(E)$ -)horizontal lifting of a vector field  $X \in \mathfrak{X}(M) = \operatorname{Sec}(T(M), \pi_T, M)$  is the unique vector field  $\overline{X} \in \mathfrak{X}(E) = \operatorname{Sec}(E, \pi, M)$  such that  $\overline{X}_u \in T^{h}_u(E)$  and  $\overline{X}$  and X are  $\pi$ -related, i.e.,  $\pi_* \circ \overline{X} = X \circ \pi$ .

**Proposition 14.1.** Let  $T^{h}(E)$  be a connection on some vector bundle  $(E, \pi, M)$  and  $\beta : [a, b] \to M$  be a  $C^{\infty}$  path in M. For every  $t_{0} \in [a, b]$  and  $u \in \pi^{-1}(\beta(t_{0}))$ , there exists a unique  $T^{h}(E)$ -horizontal lifting  $\overline{\beta}$  of  $\beta$  to E such that  $\overline{\beta}(t_{0}) = u$ .

Proof. See [23, Proposition 2.32 on p. 59].

**Theorem 14.1** (cf. [23, Theorems 2.28 and 2.33]). Each given parallelism structure  $\mathbb{P}$  in a vector bundle  $(E, \pi, M)$  determines a connection  $T^{h}(E)$  on  $(E, \pi, M)$  such that a path  $\overline{\beta}$  in E is a parallel lifting of the path  $\beta = \pi \circ \overline{\beta}$  in M with respect to  $\mathbb{P}$  if and only if  $\overline{\beta}$  is horizontal with respect to  $T^{h}(E)$ , i.e., iff  $\overline{\beta}$  is a path in  $T^{h}(E)$ . Conversely, if  $T^{h}(E)$  is a connection on a vector bundle  $(E, \pi, M)$ , the system of  $T^{h}(E)$ -horizontal liftings to E of the  $C^{\infty}$  paths in M is a parallelism structure in  $(E, \pi, M)$  and, besides, the connection on  $(E, \pi, M)$  determined by it according to the previous assertion is just  $T^{h}(E)$ .

Proof. See [23, pp. 55-61].

The last theorem is a rigorous expression of the assertion that a parallelism structure in a vector bundle is equivalent to a connection on the same bundle. By its means one can easily prove some important properties of the parallelism structures.

**Proposition 14.2.** Let  $\mathbb{P}$  be a parallelism structure in vector bundle  $(E, \pi, M)$  and  $\beta: [a, b] \to M$  be a  $C^{\infty}$  path. For each  $c \in [a, b]$  and  $u \in \pi^{-1}(\beta(a))$ ,

$$\mathbb{P}_{u}^{\beta}(\lambda) = \mathbb{P}_{u}^{\beta|[a,c]}(\lambda), \qquad \lambda \in [a,c].$$
(14.11)

*Proof.* Let  $T^{h}(E)$  be the connection on  $(E, \pi, M)$  determined by  $\mathbb{P}$  as in Theorem 14.1. Since  $\mathbb{P}^{\beta}_{u}$  and  $\mathbb{P}^{\beta|[a,c]}_{u}$  are parallel liftings of respectively  $\beta$  and  $\beta|_{[a,c]}$  through u, they are the unique  $T^{h}(E)$ -horizontal liftings of respectively  $\beta$  and  $\beta|_{[a,c]}$  through u. Consider the restricted lifting  $\mathbb{P}^{\beta}_{u}|_{[a,c]}$ . As  $\pi \circ (\mathbb{P}^{\beta}_{u}|_{[a,c]}) = \beta|_{[a,c]}$  and  $\mathbb{P}^{\beta}_{u}|_{[a,c]}(a) = \mathbb{P}^{\beta}_{u}(a) = u$ , it is a  $T^{h}(E)$ -horizontal lifting through u and, due to the uniqueness of such liftings (see Proposition 14.1), we have

$$\mathbb{P}_u^\beta|_{[a,c]} = \mathbb{P}_u^{\beta|[a,c]} \tag{14.11'}$$

which is equivalent to (14.11).

**Proposition 14.3.** Let  $\mathbb{P}$  be a parallelism structure in  $(E, \pi, M)$ , the path  $\beta \colon [a, b] \to M$  be  $C^{\infty}$ , and  $u \in \pi^{-1}(\beta(a))$ . For each  $c \in [a, b]$ ,

$$\mathbb{P}_{u}^{\beta}(\lambda) = \mathbb{P}_{\mathbb{P}_{u}^{\beta|[a,c]}(c)}^{\beta|[c,b]}(\lambda), \qquad \lambda \in [c,b].$$
(14.12)

$$\square$$

Proof. Let  $T^{h}(E)$  be the connection on  $(E, \pi, M)$  determined by  $\mathbb{P}$  according to Theorem 14.1. Since  $\mathbb{P}_{u}^{\beta}$  and  $\mathbb{P}_{\mathbb{P}_{u}^{\beta|[a,c]}(c)}^{\beta|[c,b]}$  are parallel liftings of  $\beta$  through u and of  $\beta|[c,b]$  through  $\mathbb{P}_{u}^{\beta|[a,c]}(c)$ , respectively, they are the unique  $T^{h}(E)$ -horizontal liftings of  $\beta$  through u and of  $\beta|[c,b]$  through  $\mathbb{P}_{u}^{\beta|[a,c]}(c)$ , respectively. Evidently  $\pi \circ (\mathbb{P}_{u}^{\beta}|[c,b]) = \beta|[c,b]$  and  $\mathbb{P}_{u}^{\beta}(c) = \mathbb{P}_{u}^{\beta|[a,c]}(c)$  (see (14.11) with  $\lambda = c$ ). Hence  $\mathbb{P}_{u}^{\beta}|[c,b]$  is a  $T^{h}(E)$ -horizontal lifting of  $\beta|_{[c,b]}$  through  $\mathbb{P}_{u}^{\beta|[a,c]}(c)$  and, by Proposition 14.1, it coincides with  $\mathbb{P}_{u}^{\beta|[a,c]}(c)$ , i.e.,

$$\mathbb{P}_{u}^{\beta}\Big|_{[c,b]} = \mathbb{P}_{\mathbb{P}_{u}^{\beta|[c,c]}(c)}^{\beta|[c,b]}$$
(14.12')

which is tantamount to (14.12).

Now we shall consider briefly the description of parallelism structures (and hence connections) on vector bundles in terms of covariant derivatives.

**Definition 14.7.** Let  $(E, \pi, M)$ , E and M being  $C^{\infty}$  manifolds, be a K-vector bundle and  $g: N \to M$ , N being  $C^{\infty}$  manifold, be of class  $C^{\infty}$ . A mapping  $\nabla: \mathfrak{X}(N) \times \text{Lift}_g(E, \pi, M) \to \text{Lift}_g(E, \pi, M)$  is called a *covariant derivative* (or *covariant derivative operator*) in  $(E, \pi, M)$  along g if for every  $U, V \in \mathfrak{X}(N)$  and  $X, Y \in \text{Lift}_g(E, \pi, M)$  the mapping  $\nabla: (U, X) \mapsto \nabla_U X$  has the properties:

(i)  $\nabla_{U+V}X = \nabla_UX + \nabla_VX;$ 

(ii) 
$$\nabla_{fU}X = f\nabla_U X, \quad f \in \mathfrak{F}(N);$$

(iii) 
$$\nabla_U(X+Y) = \nabla_U X + \nabla_U Y$$

(iv)  $\nabla_U(hX) = U(h)X + h\nabla_U X, \quad h \in \mathfrak{F}^1(N).$ 

The mapping  $\nabla X \colon \mathfrak{X}(N) \to \operatorname{Lift}_g(E, \pi, M)$ , associated to a covariant derivative  $\nabla$  and such that  $\nabla X \colon U \mapsto (\nabla X)(U) := \nabla_U X$ , is called covariant differential of X (along g) and the lifting  $\nabla_U X$  of g is called covariant derivative (along g) of X with respect (or along) U. If  $u \in T_p(N)$ ,  $p \in N$ , we set  $\nabla_u X := (\nabla_U X)_p$  with  $U \in \mathfrak{X}(N)$  such that  $U_p = u$ , which is well-defined due to (i) and (ii).

Two important special cases are worth singling out. If N = M and  $g = id_M$ , then  $\nabla$  is called covariant derivative in  $(E, \pi, M)$ . Besides, if in addition E = T(M), i.e., if  $\nabla$  is a covariant derivative in the tangent bundle over M, then  $\nabla$  is called covariant derivative on M.

To formulate the second main result of this subsection, Theorem 14.2 below, we shall briefly present some preliminary material.<sup>3</sup>

Put  $\pi^*E := \pi^*(E) := \{(\zeta, \xi) | \zeta, \xi \in E, \ \pi(\zeta) = \pi(\xi)\}$ .<sup>4</sup> Evidently,  $\pi^*E$  is a vector space. We assert that  $\pi^*E$  is isomorphic to the vertical component  $T^v(E) := \bigcup_{u \in E} T_u^v(E) = \bigcup_{u \in E} T_u(\pi^{-1}(\pi(u)))$  of the bundle space T(E). To prove this, define

 $<sup>^3 \</sup>rm For$  details, see [23]. A reader familiar with [23] or equivalent material can skip the text from this point to Theorem 14.2.

<sup>&</sup>lt;sup>4</sup>The set  $\pi^*(E)$  is the bundle space of the pullback of  $(E, \pi, M)$  along  $\pi$  [23, p. 8].

 $\mathfrak{g}: \pi^*E \to T^v(E)$  by  $\mathfrak{g}: (\zeta, \xi) \mapsto \mathfrak{g}_{\zeta}\xi := \dot{\alpha}(0)$  where  $(\zeta, \xi) \in \pi^*$  and  $\alpha$  is a path in E given via  $\alpha: t \mapsto \alpha(t) := \zeta + t\xi, t \in \mathbb{R}$ . The mapping  $\mathfrak{g}$  is a vector space isomorphism and its inverse is  $\mathfrak{g}^{-1}: w \mapsto (\zeta, \xi), w \in T^v(E) \subset T(E)$ , where  $\zeta := \pi_{T(E)}(w) \in E, \pi_{T(E)}$  being the projection of the tangent bundle  $(T(E), \pi_{T(E)}, E)$  over E, and  $\xi$  is a point in  $\pi^{-1}(\pi(\zeta))$ , considered as a manifold, such that its coordinates in some fixed local coordinates  $\{x^i\}$  are equal to the local coordinates of w (as an element of  $T^v_{\zeta}(E) \subset T_{\zeta}(E)$ ) in the associated frame  $\{\frac{\partial}{\partial x^i}\}$  at  $\zeta$ .

Denote by  $\operatorname{pr}_2: T^v(E) \to E$  a mapping assigning to each  $w \in T^v(E)$  the "second component" of the element of  $\pi^*(E)$  corresponding to w via  $\mathfrak{g}$ , i.e.,  $\operatorname{pr}_2(w) := \xi$  if  $w = \mathfrak{g}_{\zeta}\xi$  for some (unique)  $(\zeta, \xi) \in \pi^*(E)$ .

If  $T^h(E)$  is a connection on a vector bundle  $(E, \pi, M)$ , every vector  $w \in T(E)$ has, according to (14.8), a unique decomposition  $w = w^h + w^v$ , where, if  $w \in T_u(E)$ for some  $u \in E$ ,  $w^h \in T_u^h(E)$  and  $w^v \in T_u^v(E)$  are, respectively, the horizontal and vertical components of w. The connection mapping  $\varkappa: T(E) \to E$  of  $T^h(E)$ is defined by  $\varkappa: w \mapsto \varkappa(w) := \operatorname{pr}_2(w^v) = \operatorname{pr}_2(w - w^h)$  for each  $w \in T(E)$ .

**Theorem 14.2.** Let  $T^h(E)$  be a connection on some vector bundle  $(E, \pi, M)$ , the mapping  $\varkappa: T(E) \to E$  be its connection mapping, and  $g: N \to M$ , where Nis  $C^{\infty}$  manifold, be of class  $C^{\infty}$ . The mapping  $\nabla: \mathfrak{X}(N) \times \operatorname{Lift}_g(E, \pi, M) \to$  $\operatorname{Lift}_g(E, \pi, M)$  given by  $\nabla: (U, X) \mapsto \nabla_U X := \varkappa(X_*(U))$  for every  $(U, X) \in$  $\mathfrak{X}(N) \times \operatorname{Lift}_g(E, \pi, M)$  is a covariant derivative in  $(E, \pi, M)$  along g. Besides, a lifting  $X \in \operatorname{Lift}_g(E, \pi, M)$  is parallel (with respect to the parallelism structure assigned to  $T^h(E)$  via 14.1) if and only if  $\nabla_U X = 0$  for all  $U \in \mathfrak{X}(N)$  and if  $h: L \to N$  for a  $C^{\infty}$  manifold L, then  $\nabla(X \circ h) = (\nabla X) \circ h_*$  where the first (resp. second)  $\nabla$  is along  $g \circ h$  (resp. g). Conversely, if  $\nabla$  is a covariant derivative in  $(E, \pi, M)$  (along  $\operatorname{id}_M$ ), then there exists a connection  $T^h(E)$  on  $(E, \pi, M)$  such that  $X_*(v) \in T^h(E)$  for given  $X \in \operatorname{Sec}(E, \pi, M)$  and  $v \in T(M)$  if and only if  $\nabla_v X = 0$ .

*Proof.* See [23, pp. 74–77].

**Corollary 14.1** (cf. [23, Corollary 2.59]). A covariant derivative  $\nabla$  in a vector bundle  $(E, \pi, M)$  determines a parallelism structure  $\mathbb{P}$  in it such that a lifting  $X \in \text{Lift}_{\beta}(E, \pi, M)$  of a  $C^{\infty}$  locally injective path  $\beta \colon [a, b] \to M$  is parallel (along  $\beta$  with respect to  $\mathbb{P}$ ) iff

$$\nabla_{\dot{\beta}}X = 0. \tag{14.13}$$

Remark 14.2. Regardless of the appearance of equation (14.13) in the book [23, Corollary 2.59], its left-hand side is not well defined without special explanations which are not given in [23], where implicitly is assumed  $\beta$  to be injective, as in our text until now. The problem is that the mapping  $\nabla_V$ ,  $V \in \mathfrak{X}(M)$ , in particular  $\nabla_v$  with  $v \in T(M)$  or  $\nabla_{\dot{\beta}(s)}$ ,  $s \in [a, b]$ , acts on sections of  $(E, \pi, M)$ , which are liftings of  $g = \operatorname{id}_M$  to E, not on liftings of paths or on a lifting of a particular path such as  $X \in \operatorname{Lift}_{\beta}(E, \pi, M)$ . Besides, the symbol  $\nabla_{\dot{\beta}}$  is not defined as  $\dot{\beta}$  is a lifting of  $\beta$  to T(M) and, if  $\beta$  has self-intersections, it cannot be considered as a vector

#### 14. Connections and parallel transports

field on  $\beta([a,b])$ . For these reasons, the symbol  $\nabla_{\dot{\beta}}X$  is incorrect. For injective path  $\beta$ , the symbol  $\nabla_{\dot{\beta}}X$  may be defined as a section of  $(E, \pi, M)|_{\beta([a,b])}$  such that  $\nabla_{\dot{\beta}}X:\beta(s) \mapsto \nabla_{\dot{\beta}(s)}X := (\nabla_V \check{X})_{\beta(s)}, s \in [a,b]$ , where  $V \in \mathfrak{X}(M)$  satisfies  $V_{\beta(s)} := \dot{\beta}(s)$  and  $\check{X} \in \text{Sec}(E, \pi, M)$  is such that  $\check{X}:\beta(s) \mapsto \check{X}_{\beta(s)} := X(s)$ . If  $\beta$  is not injective, i.e., if it has self-intersections, then  $\nabla_{\dot{\beta}}X$ , with  $X \in \text{Lift}_{\beta}(E, \pi, M)$ , should be regarded as a lifting of  $\beta$  to  $E, \nabla_{\dot{\beta}}X \in \text{Lift}_{\beta}(E, \pi, M)$ , such that

$$(\nabla_{\dot{\beta}}X)\colon s\mapsto (\nabla_{\dot{\beta}}X)(s):= \left(\frac{\mathrm{d}X^{i}(s)}{\mathrm{d}s} + \Gamma^{i}{}_{j\mu}(\beta(s))X^{j}(s)\dot{\beta}^{\mu}(s)\right)e_{i}(\beta(s)) \quad (14.14)$$

where the notation is explained in the paragraphs containing equations (14.18)–(14.21) below. In a case of injective path  $\beta$  both definitions agree in a sense that  $\nabla_{\dot{\beta}(s)}\check{X} = (\nabla_{\dot{\beta}}X)(s)$ , i.e., the value of  $\nabla_{\dot{\beta}}X$ , as a section over  $\beta([a, b])$ , at  $\beta(s)$  is equal to the value of  $\nabla_{\dot{\beta}}X$ , as a lifting of  $\beta$ , at  $s \in [a, b]$ .

Proof. The operator  $\nabla$ , by the second part of Theorem 14.2, determines a connection  $T^h(E)$  in  $(E, \pi, M)$  which in turn, by Theorem 14.1, determines a parallelism structure  $\mathbb{P}$  in  $(E, \pi, M)$  (via the  $T^h(E)$ -horizontal liftings of the  $C^{\infty}$  paths in M). Since a path in E is parallel with respect to  $\mathbb{P}$  iff it is  $T^h(E)$ -horizontal, X is parallel iff  $\dot{X}_{X(s)} \in T^h_{X(s)}(E)$  for each  $s \in [a, b]$ . Denoting by  $D := \frac{d}{dt} \in \mathfrak{X}(\mathbb{R})$  the tangent vector field to  $\mathbb{R}$ , we can write [7, Section 1.23 (e)]  $\dot{X}_{X(s)} = X_*|_s(D|_s)$ . So, X is parallel iff

$$\dot{X}_{X(s)} = X_*|_s(D|_s) \in T^h_{X(s)}(E).$$
 (14.15)

Suppose  $\beta$  is injective and denote by  $\beta^{-1}: \beta([a, b]) \to [a, b]$  the mapping inverse to  $\beta$  on the image  $\beta([a, b])$ . Since  $\dot{\beta}(s) = \beta_*|_{\beta(s)}(D|_s)$ , we have  $D|_s = (\beta_*|_{\beta(s)})^{-1}(\dot{\beta}(s)) = ((\beta^{-1})_*|_{\beta(s)})(\dot{\beta}(s))$ . Substituting this in  $\dot{X}_{X(s)} = X_*|_s(D|_s)$ , we get

$$\dot{X}_{X(s)} = [(X \circ \beta^{-1})_*|_{\beta(s)}](\dot{\beta}(s))$$
(14.16)

for every injective path  $\beta: [a,b] \to M$  and  $s \in [a,b]$ . Let  $\check{X} \in \text{Sec}(E,\pi,M)$  be such that  $\check{X}|_{\beta([a,b])} := X \circ \beta^{-1}$ .<sup>6</sup> Now the condition (14.15) is tantamount to  $\check{X}_*|_{\beta(s)}(\dot{\beta}(s)) \in T^h_{X(s)}(E)$  which, by the last assertion of Theorem 14.2, is equivalent to

$$\nabla_{\dot{\beta}(s)} \check{X} := (\nabla_V \check{X})|_{\beta(s)} = 0 \tag{14.17}$$

where  $V \in \mathfrak{X}(M)$  and  $V|_{\beta(s)} := \dot{\beta}(s)$ . At the end, if we identify  $\check{X}$  along  $\beta$  with X, as  $\check{X} : \beta(s) \mapsto (X \circ \beta^{-1})(\beta(s)) \equiv X(s)$ , the last equality can be rewritten as (14.13).

Let now  $\beta \colon [a, b] \to M$  be locally injective and, for each  $s \in [a, b]$ , the symbol  $J_s$  denotes the maximal subinterval  $J_s \subseteq [a, b]$  containing  $s, J_s \ni s$ , such that

<sup>&</sup>lt;sup>5</sup>Here and below we apply the formula  $(\varphi \circ \psi)_*|_p = \varphi_*|_{\psi(p)} \circ \psi_*|_p$ , where  $\psi: L \to M$ ,  $\varphi: M \to N, L, M$ , and N are  $C^1$  manifolds and  $p \in L$ ; see [7, Section 1.25 (c)].

<sup>&</sup>lt;sup>6</sup>The existence of a  $C^{\infty}$  section  $\check{X}$  with the property mentioned can be easily proved by applying [15, Lemma 5 on p. 28] to the components of  $\check{X}$  in some frame over  $\beta([a, b])$ .

 $\begin{array}{l} \beta_s := \beta|_{J_s} \text{ is injective. Repeating the above procedure, we see that } X_s := X|_{J_s} \\ \text{is parallel along } \beta_s \text{ iff } \nabla_{\dot{\beta}_s(t)} \check{X}_s = 0, \ t \in J_s, \text{ with } \check{X}_s \in \operatorname{Sec}(E, \pi, M) \text{ such that } \\ \check{X}_s|_{\beta_s(J_s)} = \check{X}_s|_{\beta(J_s)} := X \circ \beta_s^{-1}. \text{ From the derived further local expression (14.20)} \\ \text{follows that } \nabla_{\dot{\beta}_{s_1}(t)} \check{X}_{s_1} = \nabla_{\dot{\beta}_{s_2}(t)} \check{X}_{s_2} \text{ for } s_1, s_2 \in [a, b], \ J_{s_1} \cap J_{s_2} \neq \emptyset, \text{ and } t \in J_{s_1} \cap J_{s_2} \text{ and, moreover, } \nabla_{\dot{\beta}_s(t)} \check{X}_s = \hat{X}(t), \text{ where the lifting } \hat{X} \in \operatorname{Lift}_{\beta}(E, \pi, M) \text{ is locally given by the right-hand side of (14.14). At the end, if, as above, we identify \\ \check{X}_s \text{ with } X|_{J_s}, \text{ we can set } \nabla_{\dot{\beta}_s(t)} X_s = \hat{X}|_{J_s} \text{ which is a local version of (14.14) and, consequently, (14.13) is a global variant of } \nabla_{\dot{\beta}_s} \check{X}_s = 0, \ s \in [a, b]. \end{array}$ 

Theorems 14.1 and 14.2 and Corollary 14.1 imply

**Conclusion 14.1.** The concepts of parallelism structure, connection, and covariant derivative respectively in, on, and in a vector bundle are equivalent: given any one of them, the other two can appropriately be assigned to it.

By abuse of the language but without risk of ambiguity, the covariant derivatives are called connections (as in Definition I.3.1 on page 21), the reason being the last conclusion.

Before closing this long introductory subsection, we would like to write some local expressions and decompositions which will essentially be used further.

Let  $\nabla^g$  be a covariant derivative in a vector bundle  $(E, \pi, M)$  along a  $C^{\infty}$ mapping  $g \colon N \to M$ , where E, M, and N are  $C^{\infty}$  manifolds. Let U be an open subset in N. Suppose  $\{E_{\mu} | \mu = 1, \ldots, \dim N\}$  is a frame in T(N) over U and  $\{e_i | i = 1, \ldots, \pi^{-1}(p), p \in M\}$  is a frame, possibly depending on g, along  $g|_U$  in E, i.e., for each  $q \in N, \{E_{\mu} | q\}$  is a basis in  $T_q(N)$  and  $\{e_i(q; g)\}$  is a basis in  $\pi^{-1}(g(q))$ (along g). Notice,  $E_{\mu}$  are vector fields over U in T(N), while  $e_i(\cdot; g) \colon q \mapsto e_i(q; g)$ are liftings of g.

For every  $V \in \mathfrak{X}(U)$  and  $X_g \in \text{Lift}_g((E, \pi, M)|_U)$ , we write the expansions  $(p \in M)$ 

$$V = V^{\mu} E_{\mu} \equiv \sum_{\mu=1}^{\dim N} V^{\mu} E_{\mu}, \quad X_g = X_g^i e_i(\,\cdot\,;g) \equiv \sum_{i=1}^{\dim \pi^{-1}(p)} X_g^i e_i(\,\cdot\,;g)$$

for some  $C^{\infty}$  functions  $V^{\mu}, X_g^i: U \to \mathbb{K}$ . Applying the properties (i)–(iv) of Definition 14.7 and using that  $\nabla_V^g X_g \in \text{Lift}_g((E, \pi, M)|_U)$  (here and below g is not a summation index!), we get

$$\nabla_{V}^{g}(X_{g}) = V^{\mu} \left[ E_{\mu}(X_{g}^{i}) + \Gamma_{j\mu}^{i}(\cdot;g) X_{g}^{j} \right] e_{i}(\cdot;g)$$
(14.18)

where the summation over  $\mu(,\nu,...)$  is from 1 to dim N and over i(,j,...) is from 1 to dim  $\pi^{-1}(p), p \in M$ . Here the functions  $\Gamma^{i}_{j\mu}(\cdot;g): U \to \mathbb{K}$  are defined by (cf. (I.3.1))

$$\nabla_{E_{\mu}}(e_j(\,\cdot\,;g)) =: \Gamma^i{}_{j\mu}(\,\cdot\,;g)e_i(\,\cdot\,;g) \tag{14.19}$$

and are called *local coefficients* of the covariant derivative  $\nabla^g$  with respect to the pair of frames  $(\{E_\mu\}, \{e_i\})$ .

If  $\nabla$  is a covariant derivative in  $(E, \pi, M)$ ,  $\nabla = \nabla^{\mathrm{id}_M}$ , then the argument  $g(=\mathrm{id}_M)$  above will be omitted, i.e.,

$$\nabla_{V}(X) = V^{\mu}(E_{\mu}(X^{i}) + \Gamma^{i}_{\ j\mu}X^{j})e_{i}$$
(14.20)

where  $X \in \text{Sec}((E, \pi, M)|_U)$ ,  $e_i$  are considered as vector fields, and the functions  $\Gamma^i{}_{i\mu}: U \to \mathbb{K}, U \subseteq M$ , are given via (cf. (I.3.1))

$$\nabla_{E_{\mu}}(e_j) =: \Gamma^i{}_{j\mu}e_i. \tag{14.21}$$

**Proposition 14.4.** Every derivation  $D^{\gamma}$  in  $(E, \pi, M)$  along a fixed  $C^1$  path  $\gamma: J \to M$  with coefficients (cf. (2.9) and (6.1))

$$\Gamma^{i}{}_{j}(\,\cdot\,;\gamma) = \Gamma^{i}{}_{j\mu}(\,\cdot\,;\gamma)\dot{\gamma}^{\mu},\tag{14.22}$$

where  $\Gamma^i_{j\mu}(\cdot;\gamma): J \to \mathbb{K}$ , is a covariant derivative in  $(E,\pi,M)$  along  $\gamma$  with respect to  $\dot{\gamma}$ .

*Proof.* Compare (14.18) with (2.9) or check the properties described in Definition 14.7 with  $\gamma$  and  $D^{\gamma}$  for g and  $\nabla$ , respectively, invoking the definition of  $D^{\gamma}$  (see Definition 2.1 on page 219).

**Proposition 14.5.** A linear connection  $\nabla$  in the tangent bundle over a manifold M is a covariant derivative in this bundle (along the identity mapping id<sub>M</sub>).

*Proof.* Compare (14.18) with (I.3.2) and (I.3.3) for r = s = 1 or restrict Definition I.3.1 on page 21 to the tangent bundle over M and check the properties described in Definition 14.7.

*Remark* 14.3. If one starts with the general theory of connections on vector bundles, Proposition 14.5 should be converted into a definition, i.e., in such an approach a linear connection on a manifold is defined as a connection on its tangent bundle and, by abuse of the language, the associated covariant derivative is also called linear connection on the same manifold. For details, see [23, p. 93 ff].

Suppose the frames  $\{E_{\mu}\}$  and  $\{e_i\}$  are subjected simultaneously to the following changes

$$E_{\mu}|_{q} \mapsto E'_{\mu}|_{q} = B^{\nu}_{\mu}(q)E_{\nu}|_{q}, \quad e_{i}(q;g) \mapsto e'_{i}(q;g) = A^{j}_{i}(q;g)e_{j}(q;g)$$
(14.23)

where  $q \in N$ , and  $B := [B_{\mu}^{\nu}]$  and  $A(\cdot; g) := [A_i^j(\cdot; g)]$  are non-degenerate matrixvalued functions of class  $C^{\infty}$ . (In what follows, it is sufficient  $A(\cdot; g)$  to be of class  $C^1$  or  $C^2$  and B to be arbitrary of class at most  $C^2$ .) A straightforward calculation, based on (14.19), reveals that the change (14.23) leads to the transformation  ${\Gamma^i}_{j\mu} \mapsto {\Gamma'}^i{}_{j\mu}$  with

$$\Gamma_{j\mu}^{\prime i}(\cdot;g) = \sum_{\nu=1}^{\dim N} \sum_{k,l=1}^{\dim \pi^{-1}(p)} B_{\mu}^{\nu} (A^{-1}(\cdot;g))_{k}^{i} A_{j}^{l}(\cdot;g) \Gamma_{l\nu}^{k}(\cdot;g) + \sum_{\nu=1}^{\dim N} \sum_{k=1}^{\dim \pi^{-1}(p)} B_{\mu}^{\nu} (A^{-1}(\cdot;g))_{k}^{i} E_{\nu}(A_{j}^{k}(\cdot;g)). \quad (14.24)$$

This means that the functions  $\Gamma_{j\mu}^{\prime i}(\cdot;g)$  are the coefficients of  $\nabla^{g}$  in the new pair of frames  $(\{E'_{\mu}\}, \{e'_{i}\}), \nabla^{g}_{E'_{\mu}}(e'_{j}(\cdot;g)) = \Gamma_{j\mu}^{\prime i}(\cdot;g)e'_{i}(\cdot;g)$ . Introducing the matrices of the coefficients of  $\nabla^{g}$  (the coefficients' matrices of  $\nabla^{g}$ )  $\Gamma_{\mu}(\cdot;g) := [\Gamma_{j\mu}^{i}(\cdot;g)]_{i,j=1}^{\dim \pi^{-1}(p)}, p \in M$ , we can rewrite (14.24) in a more compact form:

$$\Gamma'_{\mu}(\cdot;g) = B^{\nu}_{\mu}[A^{-1}(\cdot;g)\Gamma_{\nu}(\cdot;g)A(\cdot;g) + A^{-1}(\cdot;g)E_{\nu}(A(\cdot;g)].$$
(14.25)

The reader may have notice that, up to notation and meaning of the symbols, (14.24) and (14.25) are identical with (6.5) and (6.4), respectively. We shall comment on this fact later in subsection 14.3.

If one is dealing with a connection in a bundle  $(E, \pi, M)$  (along  $id_M$ ), the argument  $g(= d_M)$  should be deleted in equations (14.23)–(14.25) in accordance with (14.20) and (14.21).

Now we shall derive the parallel transport equation in a given vector bundle  $(E, \pi, M)$  (cf. [23, p. 77, Exercise 1]).

Let  $\nabla$  be a covariant derivative in  $(E, \pi, M)$  and  $\mathbb{P}$  be the determined by it parallelism structure according to Corollary 14.1. By the same corollary, a section  $X \in \text{Sec}(E, \pi, M)$  is parallel along  $\beta: [a, b] \to M$  iff  $\nabla_{\beta} X = 0$ . Writing this equation in a pair of frames  $(\{E_{\mu}\}, \{e_i\}), \{E_{\mu}\}$  in T(M) and  $\{e_i\}$  in E, and applying (14.20), we get

$$\dot{\beta}(X^i) + \Gamma^i_{\ j\mu} X^j \dot{\beta}^\mu = 0 \tag{14.26}$$

along  $\beta$ , or

$$\frac{\mathrm{d}X^i(\beta(s))}{\mathrm{d}s} + \Gamma^i{}_{j\mu}(\beta(s))X^j(\beta(s))\dot{\beta}^\mu(s) = 0 \qquad (14.26')$$

with  $s \in [a, b]$ ,  $\dot{\beta}(s) = \dot{\beta}^{\mu}(s)E_{\mu}|_{\beta(s)} \in T_{\beta(s)}(M)$ , and  $X = X^{i}e_{i}$ .

Relying on the results obtained, we can partially generalize Definition 14.2 as follows.<sup>7</sup>

**Definition 14.8.** Let  $\nabla^g$  be a covariant derivative in a vector bundle  $(E, \pi, M)$  along a mapping  $g: N \to M$ . A lifting  $X \in \text{Lift}_g(E, \pi, M)$  is called parallel (with respect to  $\nabla^g$ ) if  $\nabla^g_V X = 0$  for every  $V \in \mathfrak{X}(N)$ . A lifting X of g to E is called parallel (with respect to  $\nabla^g$ ) along a path  $\beta: [a, b] \to N$  in N (not in M!) if  $\nabla^g_{\beta} X = 0$ .

 $<sup>^7\</sup>mathrm{For}$  the rest of the generalization of Definition 14.2, vide infra Definition 14.9 on the facing page.

Exercise 14.1. Verify the consistency of Definitions 14.8 and 14.2.

**Exercise 14.2.** By repeating *mutatis mutandis* the procedure leading to equations (14.26) and (14.26'), prove that a lifting  $X_g \in \text{Lift}_g(E, \pi, M)$  is parallel along  $\beta$  iff in any pair of frames  $(\{E_\mu\}, \{e_i\})$ , with  $\{E_\mu\}$  in T(N) (not in T(M)) and  $\{e_i\}$  in E, is valid

$$\dot{\beta}(X_g^i) + \Gamma^i{}_{j\mu}(\,\cdot\,;g)X_g^j\dot{\beta}^\mu = 0 \tag{14.27}$$

along  $\beta \colon [a, b] \to N$ , or

$$\frac{\mathrm{d}X_g^i(\beta(s))}{\mathrm{d}s} + \Gamma^i_{\ j\mu}(\beta(s);g)X_g^j(\beta(s))\dot{\beta}^\mu(s) = 0 \tag{14.27'}$$

were  $s \in [a, b], \dot{\beta}(s) = \dot{\beta}^{\mu}(s)E_{\mu}|_{\beta(s)} \in T_{\beta(s)}(N)$ , and  $X_g = X_g^i e_i(\cdot; g)$ .

### 14.2. Parallel transports in vector bundles

This subsection is devoted to some problems connected with the axiomatical definition of parallel transports in vector bundles.

**Definition 14.9.** The parallel transport  $\mathsf{P}$  in a vector bundle  $(E, \pi, M)$  corresponding (assigned) to a parallelism structure  $\mathbb{P}$  in  $(E, \pi, M)$  is a mapping  $\mathsf{P} \colon \beta \mapsto \mathsf{P}^{\beta}$ assigning to every  $C^{\infty}$  path  $\beta \colon [a, b] \to M$  in M the parallel transport along  $\beta$ defined by  $\mathbb{P}$  via (14.2) and (14.3).

**Proposition 14.6.** The parallel transport is reparameterization invariant in a sense that, if  $\beta \colon [a,b] \to M$  and  $\varphi \colon [c,d] \to [a,b]$  is  $C^{\infty}$  and such that  $\varphi(c) = a$  and  $\varphi(d) = b$ , then

$$\mathsf{P}^{\beta \circ \varphi} = \mathsf{P}^{\beta} \tag{14.28}$$

*Proof.* See (14.3) and (14.5).

Define the product (or the product path  $\alpha\beta$ ) of two paths  $\alpha: [a, b] \to M$  and  $\beta: [b, c] \to M$  with  $\alpha(b) = \beta(b)$  by<sup>8</sup>

$$\alpha \cdot \beta := \alpha \beta \colon [a, c] \to M, \quad \alpha \beta(s) := \begin{cases} \alpha(s) & \text{for } s \in [a, b] \\ \beta(s) & \text{for } s \in [b, c] \end{cases}.$$
(14.29)

The product of paths is associative but not commutative.

**Definition 14.10.** Suppose  $\alpha: [a, b] \to M$  and  $\beta: [b, c] \to M$  are of class  $C^{\infty}$  and  $\alpha(b) = \beta(b)$ . If the product path  $\alpha\beta$  is not  $C^{\infty}$  at the point s = b, the *parallel transport along*  $\alpha\beta$  is defined to be composition of the parallel transports along the constituent paths:

$$\mathsf{P}^{\alpha\beta} = \mathsf{P}^{\beta} \circ \mathsf{P}^{\alpha}. \tag{14.30}$$

<sup>&</sup>lt;sup>8</sup>In [23, p. 51] the product  $\alpha\beta$  is denoted by  $\beta * \alpha$ ; this is a matter of convention. See also the comments in [23, Section 2.16].

More generally, if  $\alpha_i: [a_i, a_{i+1}] \to M$ , i = 1, ..., k with  $k \in \mathbb{N} \setminus \{1\}$ , are  $C^{\infty}$  paths,  $\alpha_i(a_{i+1}) = \alpha_{i+1}(a_{i+1})$  for i = 1, ..., k - 1, and the product path  $\alpha_1 \cdots \alpha_k$  is not  $C^{\infty}$  at the points  $a_2, \ldots, a_{k-1}$ , then by definition

$$\mathsf{P}^{\alpha_1 \cdots \alpha_k} = \mathsf{P}^{\alpha_k} \circ \cdots \circ \mathsf{P}^{\alpha_1}. \tag{14.31}$$

Remark 14.4. In (14.30) and (14.31) we use the equality sign (=) instead of the equal by definition one (:=) because they are valid also in the case when, respectively,  $\alpha\beta$  and  $\alpha_1 \cdots \alpha_k$  are  $C^{\infty}$  paths (not only  $C^{\infty}$  piecewise as assumed above) which will be proved later; but the validity of, e.g., (14.30) for a  $C^{\infty}$  path  $\alpha\beta$  is a theorem, not a definition (see Proposition 14.8).

Remark 14.5. Suppose  $\alpha: [a, b] \to M$  is (finite)  $C^{\infty}$  piecewise path, i.e., there are  $k \in \mathbb{N} \setminus \{1\}$  and numbers  $a_1, \ldots, a_{k+1} \in [a, b]$  such that  $a = a_1 < a_2 < \cdots < a_{k+1} = b$ , the paths  $\alpha_i := \alpha|_{[a_i, a_{i+1}]}, i = 1, \ldots, k$  are  $C^{\infty}$ , and  $\alpha$  is not  $C^{\infty}$  at the points  $a_2, \ldots, a_{k-1}$ . Since

$$\beta = (\beta|_{[b_1, b_2]}) \cdots (\beta|_{[b_k, b_{k+1}]}) \tag{14.32}$$

for every  $\beta : [c,d] \to M$  and  $b_1, \ldots, b_{k+1} \in [c,d]$  such that  $c = b_1 < b_2 < \cdots < b_{k+1} = d$ , we have  $\alpha = \alpha_1 \cdots \alpha_k$  and, by (14.31),

$$\mathsf{P}^{\alpha} = \mathsf{P}^{\alpha_k} \circ \dots \circ \mathsf{P}^{\alpha_1}. \tag{14.33}$$

Therefore the parallel transport along a piecewise  $C^{\infty}$  path is equal to the composition (in the corresponding order) of the parallel transports along the  $C^{\infty}$  parts of the initial path.

**Proposition 14.7.** If  $\mathsf{P}$  is a parallel transport assigned to a parallelism structure  $\mathbb{P}$  in a vector bundle,  $\beta: [a, b] \to M$  is  $C^{\infty}$ , and  $c \in [a, b]$ , then

$$\mathsf{P}^{\beta} = \mathsf{P}^{\beta|[c,b]} \circ \mathsf{P}^{\beta[a,c]} \tag{14.34}$$

*Proof.* The equality (14.34) is a consequence of (14.12) with  $\lambda = b$  and equation (14.3).

**Proposition 14.8.** The equalities (14.30) and (14.31) are valid for  $C^{\infty}$  (and, by definition, for piecewise  $C^{\infty}$ ) paths  $\alpha\beta$  and  $\alpha_1 \cdots \alpha_k$  respectively.

*Proof.* Apply (14.34) with  $\alpha\beta$  for  $\beta$  and use (14.29).

Let us recall, in the approach of [23], which was followed in the previous subsection and until now in the present one, the concept of a parallel transport is a secondary one as it was defined on the base of the one of a parallelism structure (or, equivalently, connection or covariant derivative). Now we want to revert the situation by showing that the parallel transport can be defined in an independent axiomatic way as a consequence of which the parallelism structures (and, hence, connections and covariant derivatives) can be introduced on its ground.<sup>9</sup> Before giving the corresponding definition, we would like to mention that there are three groups of axioms when one tries to define a parallel transport axiomatically. First, there are axioms of a "functional type" describing the parallel transport as a mapping on the set of paths, which we assume to be of class  $C^{\infty}$  for the consistency with [23] but, generally, this is a too strong restriction. These axioms are well known and appear in practically identical form in different works. Below they are written as (14.35)-(14.38). The second group describes the "smoothness properties" of the parallel transport. The concrete axioms in it vary significantly in the different works depending on their particular aims. Here we choose them in a way consistent with the properties of the parallelism structures (see (14.39) and (14.40) below). The last group includes axioms describing the compatibility of the parallel transport with other structures, if any, such as metric, (almost) complex structure, etc. This group is not a subject of this investigation, so we admit it to be empty. As an example of such a type of axiom, we point the condition (I.4.2)on page 35 for a linear connection which means that the parallel transport does not change the scalar products of the vectors when they are transported by its means.

**Definition 14.11.** Let  $(E, \pi, M)$  be a vector bundle. A *parallel transport in* the bundle  $(E, \pi, M)$  is a mapping P assigning to each path  $\beta : [a, b] \to M$  a mapping

$$\mathsf{P}^{\beta} \colon \pi^{-1}(\beta(a)) \to \pi^{-1}(\beta(b)) \tag{14.35}$$

such that:

- (i)  $\mathsf{P}^{\beta}$  is a vector space isomorphism;
- (ii) if  $\varphi \colon [c,d] \to [a,b]$  is  $C^{\infty}$  and  $\varphi(c) = a$  and  $\varphi(d) = b$ , then

$$\mathsf{P}^{\beta \circ \varphi} = \mathsf{P}^{\beta}; \tag{14.36}$$

(iii) if  $\beta^-$  is the path inverse to  $\beta$ ,

$$\mathsf{P}^{\beta^{-}} = \left(\mathsf{P}^{\beta}\right)^{-1}; \tag{14.37}$$

(iv) if  $\alpha : [c, a] \to M$ ,  $\alpha(a) = \beta(a)$  and  $\alpha\beta$  is the product of  $\alpha$  and  $\beta$ ,

$$\mathsf{P}^{\alpha\beta} = \mathsf{P}^{\beta} \circ \mathsf{P}^{\alpha}; \tag{14.38}$$

(v) under the hypotheses preceding equation (14.6) in condition (iv) of Definition 14.1, the mapping

$$\tilde{f}: T(U) \times \pi^{-1}(U) \to E, \quad \tilde{f}: (X, u) \mapsto \mathsf{P}^{\alpha}(u), \qquad \alpha(s) := f(sX), \quad (14.39)$$
  
where  $(X, u) \in T(U) \times \pi^{-1}(U)$  and  $s \in [0, 1]$ , is of class  $C^{\infty}$ ;

 $<sup>^{9}</sup>$ A reader interested in the earliest essential published account on this problem should read the paper [31].
(vi) if  $\alpha, \beta \colon [0,1] \to M$ ,  $\alpha(0) = \beta(0)$ ,  $\dot{\alpha}(0) = \dot{\beta}(0)$ ,  $\bar{\alpha}_u(t) := \mathsf{P}^{\alpha|[0,t]}(u)$ , and  $\bar{\beta}_u(t) := \mathsf{P}^{\beta|[0,t]}(u)$  for  $u \in \pi^{-1}(\alpha(0))$  and  $t \in [0,1]$ , then

$$\dot{\bar{\alpha}}(0) = \dot{\bar{\beta}}(0).$$
 (14.40)

The following theorem says that, when talking of a parallel transport, it is insignificant whether we have in mind Definition 14.11 or Definition 14.9.

**Theorem 14.3.** The parallel transport P corresponding to a parallelism structure  $\mathbb{P}$  according to Definition 14.9 is a parallel transport in a sense of Definition 14.11. Conversely, if P is a parallel transport according to Definition 14.11, then the mapping  $\mathbb{P}: (u, \beta) \mapsto \mathbb{P}_u^{\beta}, \beta: [a, b] \to M, \ u \in \pi^{-1}(\beta(a)), \ where the mapping \mathbb{P}_u^{\beta}: [a, b] \to E \ is \ given \ by$ 

$$\mathbb{P}_{u}^{\beta}(t) = \mathsf{P}^{\beta|[a,t]},\tag{14.41}$$

is a parallelism structure such that the assigned to it parallel transport according to Definition 14.9 coincides with P, i.e., given P, there is  $\mathbb{P}$  generating it via (14.3).

*Proof.* Suppose  $\mathbb{P}$  is a parallelism structure and  $\mathsf{P}$  is the assigned to it parallel transport. The properties (14.36) and (14.38) ware already established in Propositions 14.6 and 14.8, while (14.35), (14.37), and point (i) of Definition 14.11 are valid due to Definition 14.1, condition (ii). The conditions (v) and (vi) of Definition 14.11 are simple reformulation of, respectively, conditions (iv) and (v) of Definition 14.1.

Conversely, it is almost completely trivial to be verified that the mapping  $\mathbb{P}$  defined via (14.41), in which P is a parallel transport according to Definition 14.11, satisfies all of the conditions mentioned in Definition 14.1. For instance, (14.5) follows from

$$\mathbb{P}_u^{\beta \circ \varphi}(s) = \mathsf{P}^{(\beta \circ \varphi)|[a,s]}(u) = \mathsf{P}^{\beta|[\varphi(a),\varphi(s)]} = \mathbb{P}_u^{\beta}(\varphi(s)), \qquad s \in [a,b],$$

while  $\mathbb{P}_{u}^{\beta}(a) = \mathsf{P}^{\beta|[a,a]}(u) = u$  is a result of

$$\mathsf{P}^{\beta_p} = \mathsf{id}_{\pi^{-1}(p)}, \qquad \beta_p \colon [a, a] = \{a\} \to \{p\} \in M \tag{14.42}$$

which is a consequence of  $\mathsf{P}^{\beta_p} = \mathsf{P}^{\beta_p \beta_p^-} = \mathsf{P}^{\beta_p^-} \circ \mathsf{P}^{\beta_p} = (\mathsf{P}^{\beta_p})^{-1} \circ \mathsf{P}^{\beta_p} = \mathsf{id}_{\pi^{-1}(p)}$ as  $\beta_p \beta_p^- = \beta_p = \beta_p^-$ . We leave to the reader to check the other properties. At last, by (14.3), the parallel transport  $\bar{\mathsf{P}}$  corresponding to  $\mathbb{P}$  is given via  $\bar{\mathsf{P}}^{\beta}(u) := \mathbb{P}^{\beta}_u(b) = \mathsf{P}^{\beta[[a,b]]}(u) = \mathsf{P}^{\beta}(u)$  for every  $u \in \pi^{-1}(\beta(a))$  and  $\beta \colon [a,b] \to M$ , which means  $\bar{\mathsf{P}} = \mathsf{P}$ .

The fundamental meaning of the last theorem is that the concept of a parallel transport in a vector bundle is equivalent to the one of a parallelism structure in it and, consequently, to the ones of connection on it or a covariant derivative in it.

In the present, as well as in the previous, subsection all mappings (manifolds, and other structures) were supposed to be of classes  $C^{\infty}$ . This is sufficient, but not

necessary for proving the results. Most of them remain valid when  $C^1$  or  $C^2$  (and rarely  $C^3$ ) smoothness is supposed. To simplify the formulation of some results, as well as to keep in touch with [23], further in this section we also suppose  $C^{\infty}$  smoothness. But a careful reader can reformulate the material by counting the necessary class of smoothness really required.

#### 14.3. Parallel transports and linear transports along paths

The main aim of this subsection is to be shown that the parallel transports can be considered as linear transports along paths. Once this is proved, one can freely apply the results, concerning normal frames for linear transports along paths, to arbitrary parallel transports and, as a consequence of the results obtained, to parallelism structures, connections and covariant derivatives in/on vector bundles.

**Proposition 14.9.** Let  $(E, \pi, M)$  be a vector bundle and  $\mathsf{P}$  be a parallel transport in it. If  $\gamma: J \to M$ , with J being  $\mathbb{R}$ -interval, the mapping  $P: \gamma \mapsto P^{\gamma}: (s,t) \mapsto P_{s \to t}^{\gamma}$ ,  $s, t \in J$ , such that

$$P_{s \to t}^{\gamma} = \begin{cases} \mathsf{P}^{\gamma|[s,t]} & \text{for } s \le t \\ \left(\mathsf{P}^{\gamma|[t,s]}\right)^{-1} & \text{for } s \ge t \end{cases},$$
(14.43)

is a linear transport along paths in  $(E, \pi, M)$ .

*Note* 14.1. Compare this result with Proposition 11.1, as well as equation (14.43) with (11.4a).

*Proof.* We have to check the conditions in Definition 3.1 with P for L: equation (3.1) follows from (14.35), (3.3) is a consequence of (14.42), while (3.4) is true as  $\mathsf{P}^{\gamma|[a,b]}$ ,  $a, b \in J$ ,  $a \leq b$ , is a vector space isomorphism between  $\pi^{-1}(\gamma(a))$  and  $\pi^{-1}(\gamma(b))$ . At last, (3.2) is a corollary of (14.38) as, for every path  $\gamma: J \to M$ ,  $\gamma|[a,c] = (\gamma|[a,c])|_{[a,b]}(\gamma|[a,c])|_{[b,c]} = \gamma|_{[a,b]}\gamma|_{[b,c]}$  for all  $a, b, c \in J$  such that  $a \leq b \leq c$  (see (14.32)).

**Exercise 14.3.** Prove that (14.43) is equivalent to

$$P_{s \to t}^{\gamma} = F^{-1}(t, \gamma) \circ F(s, \gamma) \tag{14.43'}$$

where

$$F(r,\gamma) := \begin{cases} \mathsf{P}^{\gamma|[r,a]} & \text{for } r \leq a \\ \left(\mathsf{P}^{\gamma|[a,r]}\right)^{-1} & \text{for } r \geq a \end{cases}, \qquad r = s, t$$

for some fixed  $a \in [\min(s, t), \max(s, t)]$ . (Hint: use the equality (14.32) and invoke equation (14.38).) Then Proposition 14.9 follows from Proposition 3.1.

It is almost evident, (14.43) can be inverted, i.e., P can be expressed via P (cf. (11.4b)):

$$\mathsf{P}^{\beta} = P^{\beta}_{a \to b} \qquad \text{for } \beta \colon [a, b] \to M, \, a \le b.$$
(14.44)

Therefore the mapping P provides an equivalent description of the parallel transport P, the only difference being that P is defined along arbitrary paths  $\gamma: J \to M$ . We shall call P parallel transport along paths if we want to emphasize its difference from P but, if it is not important and there is not a risk of ambiguities, P will simply be called parallel transport as P.

At this point a natural problem arises: when a linear transport along paths is a parallel transport (along paths)? Since this problem is aside from our main topic and its solution will not be used further in the present work, we shall consider it quite briefly.<sup>10</sup> For the purpose one should, using (14.43) or (14.44), express all of the properties of a parallel transport P mentioned in Definition 14.11 through the parallel transport P along paths corresponding to P.

**Lemma 14.1.** If P is a linear transport along paths in a vector bundle  $(E, \pi, M)$ and it satisfies the conditions

$$P_{s \to t}^{\gamma \circ \varphi} = P_{\varphi(s) \to \varphi(t)}^{\gamma}, \qquad s, t \in J'', \tag{14.45}$$

where  $\gamma: J \to M$  and  $\varphi: J'' \to J$  is orientation preserving diffeomorphism, and

$$P_{s \to t}^{\gamma|J'} = P_{s \to t}^{\gamma}, \qquad s, t \in J', \tag{14.46}$$

where  $J' \subseteq J$  is a subinterval, then the mapping  $\mathsf{P} \colon \beta \mapsto \mathsf{P}^{\beta}, \beta \colon [a, b] \to M$ , given via (14.44) satisfies the conditions (14.35) and (i)–(iv) of Definition 14.11. Conversely, given a mapping  $\mathsf{P}$  satisfying the last mentions conditions, the mapping  $P \colon \gamma \mapsto P^{\gamma} \colon (s, t) \mapsto P_{s \to t}^{\gamma}, \gamma \colon J \to M, s, t \in J$ , defined via (14.43) is a linear transport along paths possessing the properties (14.45) and (14.46).

*Proof.* NECESSITY: (14.35) follows from (3.1), (i) is a consequence of equations (3.4) and (3.5), (14.36) is a corollary of (14.45), (14.37) is a result of (14.45) with  $\varphi(t) = a + b - t, t \in [a, b]$  and (3.5), and, at last, in the proof of (14.38) the conditions (14.46), (14.45) and (3.4) are involved:  $\mathsf{P}^{\alpha\beta} = P^{\alpha\beta}_{c\to b} = P^{\alpha\beta}_{a\to b} \circ P^{\alpha\beta}_{c\to a} = P^{(\alpha\beta)|[a,b]}_{a\to b} \circ P^{(\alpha\beta)|[c,a]}_{c\to a} = P^{\beta}_{a\to b} \circ \mathsf{P}^{\alpha}$ .

SUFFICIENCY: The fact that P is a linear transport along paths was established in Proposition 14.9 in which proof the conditions (v) and (vi) of Definition 14.11 were not used. Taking for definiteness  $s \leq t$ , the proof of (14.45) and (14.46) is respectively:

$$\begin{split} P_{s \to t}^{\gamma \circ \varphi} &= \mathsf{P}^{(\gamma \circ \varphi)|[s,t]} = \mathsf{P}^{\gamma | [\varphi(s),\varphi(t)]} = P_{\varphi(s) \to \varphi(t)}^{\gamma} \\ P_{s \to t}^{\gamma | J'}, &= \mathsf{P}^{(\gamma | J')|[s,t]} = \mathsf{P}^{\gamma | [s,t]} = P_{s \to t}^{\gamma}. \end{split}$$

The case  $s \ge t$  can be proved similarly.

 $<sup>^{10}\</sup>mathrm{For}$  some details and a more general discussion of the situation, see [115] and Section V.8 below.

**Theorem 14.4.** If P is a parallel transport, then the mapping  $P: \gamma \mapsto P^{\gamma}: (s,t) \mapsto P_{s \to t}^{\gamma}, \gamma: J \to M, s, t \in J$ , defined by (14.43) is a linear transport along paths satisfying (14.45), (14.46) and the conditions:

(a) under the hypotheses preceding (14.6) in condition (iv) of Definition 14.1, the mapping

$$\tilde{f}: T(U) \times \pi^{-1}(U) \to E, \quad \tilde{f}: (X, u) \mapsto P^{\alpha}_{0 \to 1}(u), \qquad \alpha(s) := f(sX),$$
(14.47)

where  $(X, u) \in T(U) \times \pi^{-1}(U)$  and  $s \in [0, 1]$ , is of class  $C^{\infty}$ ;

(b) if  $\alpha, \beta \colon [0,1] \to M$ ,  $\alpha(0) = \beta(0)$ ,  $\dot{\alpha}(0) = \dot{\beta}(0)$ ,  $\bar{\alpha}_u(t) := P_{0 \to t}^{\alpha}(u)$ , and  $\bar{\beta}_u(t) := P_{0 \to t}^{\beta}(u)$  for  $u \in \pi^{-1}(\alpha(0))$  and  $t \in [0,1]$ , then

$$\dot{\bar{\alpha}}(0) = \dot{\bar{\beta}}(0). \tag{14.48}$$

Besides, the corresponding to P via (14.44) mapping is a parallel transport coinciding with P. Conversely, if P is a linear transport along paths satisfying the conditions (14.45)–(14.48), the mapping  $P: \beta \mapsto P^{\beta}$  given via (14.43) is a parallel transport along paths; moreover, the corresponding to P via (14.43) linear transport along paths coincides with P.

*Proof.* See Lemma 14.1 and reformulate conditions (v) and (vi) of Definition 14.11 in terms of the mapping P.

Thus we can make two important conclusions:

**Conclusion 14.2.** The parallel transports along paths are the only linear transports along paths having the special properties (14.45)-(14.48).

Therefore, we can give the following definition of a parallel transport along paths which, by Theorem 14.4 is equivalent to the one presented above.

**Definition 14.12.** A parallel transport along paths in a vector bundle  $(E, \pi, M)$  is a linear transport along paths in it satisfying the conditions (14.45)-(14.48).

**Conclusion 14.3.** The parallel transports along paths provide an equivalent description of the parallel transports and, consequently, of the parallelism structures, connections, and covariant derivatives. Therefore any result concerning linear transports along paths, in general, or parallel transports along pats, in particular, can *mutatis mutandis* be formulated equivalently in terms of parallel transports, parallelism structures, connections, or covariant derivatives.

Let  $\nabla^g$  be a covariant derivative in a vector bundle  $(E, \pi, M)$  along  $g: N \to M$  and  $\Gamma_{\mu}(\cdot; g) = [\Gamma^i{}_{j\mu}(\cdot; g)]$  be the matrix of its coefficients in a pair of frames  $(\{E_{\mu}\}, \{e_i\})$ , over a set  $U \times g(U) \subseteq N \times M$ ,  $\{E_{\mu}\}$  being a frame in T(N) over  $U \subseteq N$  and  $\{e_i\}$  being a frame over g(U) in E.

Comparing the transformation equations (14.25) on page 310 and (6.4) on page 248, or equivalently (14.24) and (6.5), one can notice from the first sight a

striking similarity between them: the only formal difference being that in equation (14.25) the indices  $\mu, \nu, \ldots$  run from 1 to dim N and the matrix-valued function  $A(\cdot;g): U \to \operatorname{GL}(\dim(\pi^{-1}(p)), \mathbb{K}), U \subseteq N, p \in M$ , stands for  $A: W \to$  $\operatorname{GL}(\dim(\pi^{-1}(p)), \mathbb{K}), W \subseteq M$ . This analogy is a hint to be considered the quantities (cf. (6.1))

$$\boldsymbol{\Gamma}(s;\gamma,g) := \left[\Gamma^{i}{}_{j\mu}(\gamma(s);g)\dot{\gamma}^{\mu}(s)\right]_{i,j=1}^{\dim \pi^{-1}(p)} = \boldsymbol{\Gamma}(\gamma(s);g)$$
$$:= \sum_{\mu=1}^{\dim N} \Gamma_{\mu}(\gamma(s);g)\dot{\gamma}^{\mu}(s) \equiv \Gamma_{\mu}(\gamma(s);g)\dot{\gamma}^{\mu}(s), \quad (14.49)$$

where  $\gamma: J \to N$  is a  $C^1$  path and  $s \in J$ , which under a change  $(\{E_\mu\}, \{e_i\}) \mapsto (\{E'_\mu = B^\nu_\mu E_\nu\}, \{e'_i = A^j_i e_j\})$  transform into (see (14.25) and cf. (3.26))

$$\Gamma'(s;\gamma,g) = A^{-1}(\gamma(s);g)\Gamma(\gamma(s);g)A(\gamma(s);g) + A^{-1}(\gamma(s);g)\frac{\partial A(\gamma(s);g)}{\partial s}.$$
 (14.50)

Prima facie one may think, relying on the Proposition 3.6 on page 232, that there is a (unique) linear transport along paths in some bundle over N whose 2index coefficient matrix is  $\Gamma(s; \gamma, g)$ . To clarify the situation, it must be emphasized on the fact that the path  $\gamma$  is in N (not in M) while the 'transport' should act between some of the fibres of  $(E, \pi, M)$ , viz. the ones over g(N). Hence the bundle in which the hypothetical transport should act must have N as a base while its bundle space must be a subspace of E, viz.  $\pi^{-1}(g(N))$ . Besides, the fibre over  $q \in N$  must be  $\pi^{-1}(g(q))$ . This is admissible iff g is injective,  $g(q_1) \neq g(q_2)$  for each  $q_1, q_2 \in N$  such that  $q_1 \neq q_2$ , in which case g is bijective on the image g(N)and vice versa, i.e., if  $g: N \to M$  is  $C^{\infty}$  and  $g|_N: N \to g(N)$  is bijection.

The results obtained in the above analysis can be summarized in the following assertion.

**Proposition 14.10.** Let  $\nabla^g$  be a covariant derivative in a vector bundle  $(E, \pi, M)$ along an injective mapping  $g: N \to M$  and  $g^{-1}: g(N) \to N$  be the inverse of g on the image  $g(N) \subseteq M$ . If  $\Gamma_{\mu}(\cdot, g)$  are the coefficients' matrices of  $\nabla^g$  in a pair of frames  $(\{E_{\mu}\}, \{e_i\})$ , then there exists a unique linear transport  ${}^{g}P$  along paths in the vector bundle  $(\pi^{-1}(g(N)), \pi_N, N)$  with  $\pi_N := g^{-1} \circ (\pi|_{\pi^{-1}(g(N))})$  such that in  $\{e_i\}$  the matrix of its (2-index) coefficients is given by (14.49) for every  $C^1$  path  $\gamma: J \to N$  or, equivalently, such that in  $(\{E_{\mu}\}, \{e_i\})$  its 3-index coefficients coincide with the coefficients of  $\nabla^g$ . Conversely, to every linear transport along paths in  $(\pi^{-1}(g(N)), \pi_N, N)$  with given 3-index coefficients  $\Gamma^i_{j\mu}(\cdot; g)$ , there corresponds a unique covariant derivative  $\nabla^g$  in  $(E, \pi, M)$  with coefficients  $\Gamma^i_{j\mu}(\cdot; g)$ .

*Proof.* See the above discussion, Proposition 3.6 on page 232, and the definition of the 3-index coefficients of a linear transport along paths in Section 6.  $\Box$ 

Remark 14.6. Notice, in the case N = M and  $g = id_M$ , i.e., when a covariant derivative in  $(E, \pi, M)$  is considered, the bundle  $(\pi^{-1}(g(N)), \pi_N, N)$ , in which the transport  ${}^{g}P$  determined by Proposition 14.9 acts, coincides with the initial bundle  $(E, \pi, M)$ .

**Proposition 14.11** (cf. Proposition 11.4). Let <sup>g</sup>D be the derivation along paths corresponding via (3.19) to the linear transport <sup>g</sup>P along paths described in Proposition 14.10. If  $X_g$  is a lifting of g to E and  $\hat{X}_g$  is a lifting of the paths in N to  $\pi^{-1}(N)$  in the bundle  $(\pi^{-1}(g(N)), \pi_N, N)$  such that  $\hat{X}_g : \gamma \mapsto X_g \circ \gamma$  for every  $C^1$  path  $\gamma : J \to N$ , then, for each  $s \in J$ ,

$$\left({}^{g}D^{\gamma}(\hat{X}_{g})\right)(s) := {}^{g}D_{s}^{\gamma}(\hat{X}_{g}) = \left(\nabla_{\dot{\gamma}}^{g}(X_{g})\right)(\gamma(s)).$$
 (14.51)

*Proof.* Invoke (3.23), (14.18), and (14.49).

Remark 14.7. In a case of a covariant derivative  $\nabla$  in  $(E, \pi, M)$ , i.e., for N = M and  $g = \operatorname{id}_M$ , equation (14.51) reads  $D_s^{\gamma}(\hat{X}) = (\nabla_{\dot{\gamma}}(X))|_{\gamma(s)}$  for every  $X \in \operatorname{Sec}(E, \pi, M)$  and  $\hat{X} \in \operatorname{PLift}(E, \pi, M)$  such that  $\hat{X} \colon \gamma \mapsto \hat{X}_{\gamma} \colon s \mapsto X_{\gamma(s)}$ . In the tangent bundle case,  $(E, \pi, M) = (T(M), \pi, M)$ , the last equality is identical with (11.6).

The above result, combined with Proposition 11.4 on page 283 is a tip for the existence of some connection between  ${}^{g}P$  and the parallel transport determined by  $\nabla^{g}$ .

**Definition 14.13** (see Definitions 14.2 and 14.8). Suppose the symbols  $(E, \pi, M)$ ,  $N, g, \nabla^g$ , and  ${}^gP$  have the same meaning as in Proposition 14.10. A path  $\bar{\gamma}: J \to \pi^{-1}(g(N)) \subseteq E$  is called parallel (along  $\gamma := \pi_N \circ \bar{\gamma}: J \to N$ ) with respect to  $\nabla^g$  if

$$\bar{\gamma}(t) = {}^{g}P^{\gamma}_{s \to t}\bar{\gamma}(s) \tag{14.52}$$

for some (and hence for any)  $s \in J$  and all  $t \in J$ ; respectively, in this case  $\bar{\gamma}$ is called a parallel (along  $\gamma$  with respect to  $\nabla^g$ ) lifting of  $\gamma$ . A *lifting*  $\lambda$  of the paths in N to paths in  $\pi^{-1}(g(N))$  is called *parallel with respect to*  $\nabla^g$  if for any  $\gamma: J \to N$ , the path  $\lambda_{\gamma}, \lambda: \gamma \mapsto \lambda_{\gamma}$ , is parallel along  $\gamma$  with respect to  $\nabla^g$ . A section  $\sigma \in \text{Sec}(\pi^{-1}(g(N)), \pi_N, N)$  is parallel along  $\gamma: J \to N$  (resp. on  $U \subseteq N$ ) with respect to  $\nabla^g$ , if the lifting  $\hat{\sigma}: \gamma \mapsto \hat{\sigma}_{\gamma} = \sigma \circ \gamma$  is parallel along the given path  $\gamma$  (resp. along every path  $\gamma: J \to U$ ) with respect to  $\nabla^g$ .

**Proposition 14.12.** Let  $\nabla^g$  and  ${}^gP$  be as in Proposition 14.10. A lifting  $X_g \in \text{Lift}_g(E, \pi, M)$  is parallel (resp. along  $\gamma: J \to N$ ) with respect to  $\nabla^g$  if and only if the lifting of paths  $\hat{X}_g \in \text{PLift}(\pi^{-1}(g(N)), \pi_N, N)$  with  $\hat{X}_g: \gamma \mapsto X_g \circ \gamma$ ,  $\gamma: J \to N$ , is parallel on N (resp. along the given path  $\gamma$ ) with respect to  $\nabla^g$ .

*Proof.* See Definitions 14.8 and 14.13 and use (3.21), on page 231, and (14.51).

So, allowing some freedom of the language, we can say that equation (14.52) describes, besides the parallel paths in  $\pi^{-1}(g(N))$ , also the parallel sections of  $(\pi^{-1}(g(N)), \pi_N, N)$ , and the parallel liftings of g to  $E \supseteq \pi^{-1}(g(N))$ .

**Proposition 14.13.** Let  $\nabla$  be a covariant derivative in a vector bundle  $(E, \pi, M)$ and  $\mathbb{P}$  be the determined by it parallelism stricture according to Corollary 14.1 on page 306. The linear transport  ${}^{\mathsf{id}_M}P$  along paths in  $(E, \pi, M)$  determined by  $\nabla$ according to Proposition 14.10 (with N = M and  $g = \mathsf{id}_M$ ) coincides with the parallel transport P along paths corresponding to  $\mathbb{P}$  via (14.3) and (14.43), i.e.,  $P = {}^{\mathsf{id}_M}P$ .

*Proof.* Let  $\gamma: J \to M$  be a  $C^1$  path,  $s_0 \in J$ , and  $u_0 \in \pi^{-1}(\gamma(s_0))$ . We have to show that the liftings  $\bar{\gamma}$  and  $\bar{\gamma}': J \to E$  of  $\gamma$ , given via

$$\bar{\gamma}(t) := P_{s_0 \to t}^{\gamma}(u_0), \quad \bar{\gamma}'(t) := {}^{\operatorname{id}_M} P_{s_0 \to t}^{\gamma}(u_0), \qquad t \in J$$

are identical,  $\bar{\gamma} = \bar{\gamma}'$ . It is trivial to see, the path  $\bar{\gamma}$  is parallel with respect to  $\mathbb{P}$  (see Definition 14.2). Therefore, by Corollary 14.1,  $\nabla_{\dot{\gamma}}\bar{\gamma} = 0$  or, in component form (see (14.14) or (14.26'))

$$\frac{\mathrm{d}\bar{\gamma}^{i}(t)}{\mathrm{d}t} + \Gamma^{i}_{\ j\mu}(\gamma(t))\dot{\gamma}^{\mu}(t) = 0 \qquad (14.53)$$

where  $\Gamma^i_{\ j\mu}$  are simultaneously 3-index coefficients of P and coefficients of  $\nabla$  (see Proposition 14.10). According to (3.20) and (3.21), the path  $\bar{\gamma}'$  satisfies the equation  $D^{\gamma}_t \bar{\gamma}' = 0$ , D being the derivation along paths generated by  ${}^{\mathrm{id}_M} P$  (see Definition 3.2 on page 230), which, due to (3.23) and (6.1), coincides with (14.53) with  $\bar{\gamma}'$  for  $\bar{\gamma}$ . Hence (the components of)  $\bar{\gamma}$  and  $\bar{\gamma}'$  are solutions of identical first order ordinary differential equations passing through one and the same point  $u_0 = \bar{\gamma}(s_0) = \bar{\gamma}'(s_0)$ . Consequently, by the uniqueness of the solutions of such equations,  $\bar{\gamma} = \bar{\gamma}'$ , which, due to the arbitrariness of  $\gamma$ ,  $s_0$ , and  $u_0$ , implies  $P = {}^{\mathrm{id}_M} P$ .

## 14.4. Normal frames for parallel transports, connections and covariant derivatives

The conclusions made in the previous subsection enable all of the general results and definitions related to linear transports along paths in vector bundles to be transferred, with appropriate changes, to the theory of connections, parallelism structures, parallel transports, and covariant derivatives in/on these bundles. In particular, such a procedure can be realized with respect to the theory of normal frames (and, possibly, coordinates) for linear transports along paths. Below we turn our attention to its brief description. To save some space and writing, we shall perform it with respect to the parallel transports (along paths) and covariant derivatives, which are most suitable for our purposes, but, applying Theorems 14.1, 14.2, 14.3, and 14.4, one can reformulate the material in terms of parallelism structures or connections.

**Definition 14.14** (see Definitions 4.1–4.4). Let P be a parallel transport along paths in a vector bundle  $(E, \pi, M)$ . A *frame* normal for P along  $\gamma: J \to M$ 

(resp. on  $U \subseteq M$ ) is called normal along  $\gamma$  (resp. on U) for the parallel transport P, parallelism structure P, or/and connection  $T^{h}(E)$  determined by it. If such frames exist, P, P, P, and/ or  $T^{h}(E)$  are called Euclidean along  $\gamma$  (resp. on U). A frame strong normal for P (on an open set containing or equal to U) is called strong normal for P, P, and/or  $T^{h}(E)$ .

**Definition 14.15 (cf.** Definition I.5.1). Let  $(E, \pi, M)$  be a vector bundle,  $g: N \to M$ ,  $\nabla^g$  be a covariant derivative in  $(E, \pi, M)$  along g, and  $U \subseteq N$ , where E, M, and N are manifolds. A frame  $\{e_i\}$  in E, defined over an open set containing or equal to g(U), is called normal for  $\nabla^g$  on U, if for some (and hence for any – see (14.25) or Conclusion 14.5 below) frame  $\{E_\mu\}$  in T(N) over U, the coefficients of  $\nabla^g$  in the pair of frames  $(\{E_\mu\}, \{e_i\})$  vanish everywhere on U. Respectively, a frame  $\{e_i\}$  is normal for  $\nabla^g$  along  $\varphi: Q \to N, Q \neq \emptyset$ , if it is normal on  $\varphi(Q) \subseteq N$ .

Obviously, it is valid

**Conclusion 14.4.** Definition 14.14 transfers all problems concerning frames normal for parallelism structures, parallel transports, and connections to similar problems for a special type of linear transports along paths, viz. to the parallel transports along paths, to which are applicable all of the results from Sections 3–9.

As a result of this conclusion, we are not going to consider the mentioned in it type of problems here. Of course, the additional conditions (14.45)-(14.48), singling out the parallel transports along paths from the other linear transports along paths, lead to some specific properties of the corresponding normal frames. But their concrete description is out of the main subject of the present book. However, below some of these special properties will be described (implicitly) when frames normal for covariant derivatives will be explored.

Let us now turn our attention to the frames normal for covariant derivatives. We distinguish these frames from the ones described by Definition 14.14 because the class of these frames is, generally, larger than the one of the letter. As we shall see below, a frame normal for a covariant derivative  $\nabla$  in a vector bundle is strongly normal for the corresponding to  $\nabla$  (or determining  $\nabla$ ) parallel transport P along paths. The relations between the frames normal for  $\nabla$  and P is similar to the ones described in Section 12 in the case of linear connections on a manifold.

Suppose  $(E, \pi, M)$  is a vector bundle,  $g: N \to M$  is of class  $C^{\infty}, \nabla^g$  is a covariant derivative in  $(E, \pi, M)$  along  $g, U \subseteq N$ ,  $\{E_{\mu} | \mu = 1, \ldots, \dim N\}$  is a frame over U in T(N), and  $\{e_i | i = 1, \ldots, \dim \pi^{-1}(p), p \in M\}$  is a frame over an open set V containing or equal to  $g(U) \subseteq M$  in E. Let  $B = [B^{\nu}_{\mu}]_{\mu,\nu=1}^{\dim N}$  and  $A = [A^j_i]_{i,j=1}^{\dim \pi^{-1}(p)}$ , be non-degenerate matrix-valued functions on V and g(V), respectively, and A be of class  $C^1$ . The frame  $\{e'_i = A^j_i e_j\}$  over  $V \supseteq g(U)$  in E is normal for  $\nabla^g$  on U iff there can be found A and B such that in the pair of frames ( $\{E_{\mu}\}, \{e_i\}$ ) the coefficients  ${\Gamma'}^i{}_{j\mu}$  of  $\nabla^g$  vanish on U. According to the transformation law (14.25), the frame  $\{e'_i\}$ 

frame equation (cf. (I.5.4) on page 39 or (12.6) on page 291)

$$(\Gamma_{\mu}(\,\cdot\,;g)A + E_{\mu}(A))|_{U} = 0 \tag{14.54}$$

holds. From here two quite fundamental conclusions can be made:

**Conclusion 14.5.** The choice of the frame  $\{E_{\mu}\}$  over U in T(N) involved in the definition of a frame normal for a covariant derivative along a mapping is completely insignificant (see (14.54) and (14.25)): a frame normal for one such choice is normal for any other choice. From general positions, this choice may be important only when one tries to solve equation (14.54) with respect to A as it may lead to a simplification as well as to a complication of the concrete form of the equation. So, it is a matter of convenience to select a particular frame  $\{E_{\mu}\}$  when exploring frames normal for covariant derivatives along mappings.

**Conclusion 14.6.** The more important result is that (14.54) is identical with equation (12.6) (with  $\mu = 1, \ldots, \dim N$  while in Section 12 is supposed  $\mu = 1, \ldots, \dim M$ ) describing the frames strongly normal on U for linear transports along paths with fixed 3-index coefficients. Therefore the discussion after (12.6) in Section 12 can be repeated word by word if we replace in it the base manifold M with the manifold N (which is the domain of the mapping  $g: N \to M$  along which is the covariant derivative  $\nabla^g$ ) and the phrase "frame(s) (strong) normal on  $\ldots$  for  $\ldots$ " with "frame(s) normal for  $\nabla^{g}$ ". In short, this results in the conclusion that all of the results of Chapter III (or II) concerning the solution of the matrix equation (14.54) are *mutatis mutandis* valid with respect to frames normal for covariant derivatives along mappings.

In particular, we can assert the existence of frames normal at any fixed point  $q \in N$ , i.e., for  $U = \{q\}$ , or along any  $C^1$  locally injective path  $\gamma: J \to N$ , i.e., for  $U = \gamma(J)$ ; also the frames normal on U, if any, are connected via linear transformations such that their matrices A satisfy the equation  $E_{\mu}(A)|_{U} = 0$  for some (and hence any) frame  $\{E_{\mu}\}$  in T(N) over an open set containing or equal to U, etc.

Thus the general, and possibly unexpected, inference is encoded in

**Conclusion 14.7.** The methods developed for the investigation of frames normal for linear connections on manifolds and the ones for linear transports along paths in vector bundles are completely applicable for the exploration of frames normal for covariant derivatives along mappings and frames strongly normal for linear transports along paths; moreover, all kinds of problems concerning the last two types of frames can be solved by the methods mentioned.

This conclusion is in full agreement with the ones at the end of the last paragraph of Section 8 and with Remark 8.1 on page 259.

Now it is time to formulate and solve the problem: what are the links between frames normal for a covariant derivative and the determined by (or determining it) parallel transport?

323

**Proposition 14.14.** Let  $\nabla^g$  and  ${}^gP$  be as in Proposition 14.9 on page 315. The frames normal on  $U \subseteq N$  for  $\nabla^g$  are strong normal on U for  ${}^gP$  and vice versa.

*Proof.* See Definitions 14.15 and 12.2 and invoke Proposition 14.9.  $\hfill \square$ 

## 14.5. On the role of the curvature

This subsection shows that the curvature of a connection (or parallel transport or covariant derivative) plays with respect to the normal frames the same role as the curvature of a linear connection (transport) with respect to this kind of frames.

The curvature tensor field or simply curvature of a connection  $T^{h}(E)$  on a vector bundle  $(E, \pi, M)$  is a mapping [23, Section 2.43]

$$R: \mathfrak{X}(M) \times \mathfrak{X}(M) \times E \to E$$

such that, given  $U, V \in \mathfrak{X}(M)$  and  $\xi \in E$ ,

$$R\colon (U, V, \xi) \mapsto R(U, V)\xi := -\mathrm{pr}_2\big(([\overline{U}, \overline{V}]_{-})^v_{\xi}\big). \tag{14.55}$$

Here:  $\operatorname{pr}_2: T^v(E) \to E$  was defined on page 306,  $\overline{U}$  and  $\overline{V}$  are the horizontal liftings to E of U and V respectively (see Definition 14.6),  $([\overline{U}, \overline{V}]_{-})_{\xi}$  is the commutator of  $\overline{U}$  and  $\overline{V}$  at  $\xi$ , and the superscript v denotes the vertical component of a vector in T(E) with respect to  $T^h(E)$ . If  $X \in \operatorname{Sec}(E, \pi, M)$ , we set  $R(U, V)X \in \operatorname{Sec}(E, \pi, M)$  with  $(R(U, V)X)(p) := R(U, V)(X(p)), p \in M$ .

For us will be important that R is tensorial in all its three arguments, it is skewsymmetric in the first two arguments, and, if  $\nabla$  is a covariant derivative in  $(E, \pi, M)$  which determines  $T^h(E)$ , then<sup>11</sup>

$$R(U,V)X = \left(R^{\nabla}(U,V)\right)(X) \equiv R^{\nabla}(U,V)X$$
(14.56)

where  $X \in \text{Sec}(E, \pi, M)$  and  $R^{\nabla}$ :  $\text{Sec}(E, \pi, M) \to \text{Sec}(E, \pi, M)$  is a mapping given by

$$R^{\nabla}(U,V) := [\nabla_U, \nabla_V]_{-} - \nabla_{[U,V]_{-}} = \nabla_U \circ \nabla_V - \nabla_V \circ \nabla_U - \nabla_{[U,V]_{-}}$$
(14.57)

and it is called *curvature (operator) of the covariant derivative*  $\nabla$ . Notice, in the tangent bundle case,  $(E, \pi, M) = (T(M), \pi, M)$ , the last formula reduces to the Definition (I.3.11) on page 25, as one can expect.

Let  $\{E_{\alpha}|\alpha = 1, \ldots, \dim M\}$  be a frame in T(M) over a set  $W \subseteq M$  and  $\{e_i|i = 1, \ldots, \dim \pi^{-1}(p), p \in M\}$  be a frame in E over W. The components  $R^i_{j\alpha\beta}$  and  $(R^{\nabla})^i_{j\alpha\beta}$  of the curvatures R and  $R^{\nabla}$ , respectively, are defined by

$$R(E_{\alpha}, E_{\beta})e_j =: R^i_{\ j\alpha\beta}e_i, \qquad R^{\nabla}(E_{\alpha}, E_{\beta})e_j =: (R^{\nabla})^i_{\ j\alpha\beta}e_i.$$

Obviously, due to (14.56), if  $\nabla$  induces  $T^h(E)$ , then  $R^i_{\ j\alpha\beta} = (R^{\nabla})^i_{\ j\alpha\beta}$ 

 $<sup>^{11}</sup>$ For the proof of these assertions, as well as for other properties of the curvature, see [23, p. 67 ff and p. 82 ff].

**Exercise 14.4.** Let  $R^{\nabla}$  be the curvature of a covariant derivative in a vector bundle  $(E, \pi, M)$ . Applying (14.56), (14.20), and (14.21), prove that

$$(R^{\nabla})^{i}_{\ j\alpha\beta} := -\Gamma^{i}_{\ j\alpha,\beta} + \Gamma^{i}_{\ j\beta,\alpha} - \Gamma^{k}_{\ j\alpha}\Gamma^{i}_{\ k\beta} + \Gamma^{k}_{\ j\beta}\Gamma^{i}_{\ k\alpha} - \Gamma^{i}_{\ j\mu}C^{\mu}_{\alpha\beta} \colon W \to \mathbb{K}$$
(14.58)

where  $[E_{\alpha}, E_{\beta}]_{-} := C^{\mu}_{\alpha\beta}E_{\mu}$  and  $f_{,\mu} := E_{\mu}(f)$  for a  $C^{1}$  function f on U.

**Proposition 14.15.** Let  $\nabla$  be a covariant derivative in a vector bundle  $(E, \pi, M)$ and P be the corresponding to it unique linear transport along paths according to Propositions 14.10 (which coincides with the parallel transport assigned to the parallelism structure  $\mathbb{P}$  described in Corollary 14.1). The components  $(R^{\nabla})^{i}_{j\alpha\beta}$  of the curvature  $R^{\nabla}$  of  $\nabla$  coincide with the 4-index components  $R^{i}_{j\alpha\beta}$  of the curvature of P (defined by (9.13) in Proposition 9.1 on page 265):

$$\left(R^{\nabla}\right)^{i}{}_{j\alpha\beta} = R^{i}{}_{j\alpha\beta}.$$
(14.59)

Moreover, if  ${}^{s}R$  is the curvature of the section-derivation corresponding to the derivation along paths generated by P (see Subsection 9.1), then

$$\left[ (R^{\nabla}(U,V))(X) \right] \Big|_{p} = ({}^{s} \mathsf{R}^{\eta}(s_{0},t_{0}))(X), \qquad (14.60)$$

where:  $U, V \in \mathfrak{X}(M), X \in \text{Sec}(E, \pi, M), p \in M, \eta: J \times J' \to M$  is of class  $C^2, \eta(s_0, t_0) = p$  for some  $(s_0, t_0) \in J \times J', \eta'(s_0, t_0) = U_p$  and  $\eta''(s_0, t_0) = V_p$  with  $\eta'(\cdot, t_0)$  and  $\eta''(s_0, \cdot)$  being the tangent vectors to  $\eta(\cdot, t_0)$  and  $\eta(s_0, \cdot)$ , respectively.

*Proof.* Equation (14.59) follows from (14.58), (9.13), and the coincidence of the coefficients of  $\nabla$  and the 3-index coefficients of P (by Proposition 14.10). To verify the relation (14.60), one should simply repeat the derivation of equality (9.17) on page 267 with the only difference that now  $\nabla$  is a covariant derivative in  $(E, \pi, M)$  and  $(E, \pi, M)$  should stand for the tangent bundle  $(T(M), \pi, M)$ .

Call a connection, covariant derivative, or parallel transport (along paths) flat (on  $W \subseteq M$ ) if their curvatures vanish (on W). If a connection, covariant derivative, and parallel transport (along paths) are connected as above supposed, then the results obtained demonstrate that the flatness of one of these three quantities implies the flatness of the remaining two of them.

Of course, all general results of Section 9, in particular the ones connecting curvature and normal frames, are completely applicable to the parallel transports along paths in vector bundles as they are a special type of linear transports along paths (Theorem 14.4 on page 317). But Proposition 14.15, combined with the results of Subsection 14.4, permits something more:

**Conclusion 14.8.** The material of Section 11, excluding the one concerning torsion and normal coordinates, can *mutatis mutandis*, practically *in extenso*, be generalized to the case of parallel transports (along paths) in general vector bundles. To this end one should replace the tangent bundle  $(T(M), \pi, M)$  with a vector bundle  $(E, \pi, M)$ , the linear connection  $\nabla$  on M with a connection  $\nabla$  in  $(E, \pi, M)$ , and to make some other, not so important changes.

This transferring of results is so trivial that the explicit formulation of the corresponding assertions makes sense if they are really required for some purpose.

**Exercise 14.5.** Reformulate some important inferences of Section 11, such as the analogues of Propositions 11.1–11.4 and 11.9–11.11, for parallel transports (along paths) in general vector bundles.

A partial generalization to arbitrary vector bundles admits the material of Section 13, but this is almost out of the subject of this book.

## 15. Autoparallel paths

The geodesics in a  $C^2$  manifold M endowed with a  $C^0$  linear connection  $\nabla$  are  $C^1$  paths whose tangent vector undergoes parallel transport along them by means of the parallel transport P assigned to  $\nabla$  (see Definition I.3.5 on page 30). If P is the parallel transport along paths corresponding to P via (11.4a), then a  $C^1$  path  $\gamma: J \to M$  is geodesic iff

$$\dot{\gamma}(s) = P^{\gamma}_{s_0 \to s} \dot{\gamma}(s_0) \tag{15.1}$$

for some (and hence any)  $s_0 \in J$  and all  $s \in J$ , by virtue of (I.3.20). Obviously, this equality can be taken as a base for a new, but equivalent, definition of a geodesic path.

Denote by  $\mathcal{P} \in \text{PLift}(T(M), \pi, M)$  the lifting of paths generated by P via equation (3.20) with  $u_0 = \dot{\gamma}(s_0)$  and P for L, viz., for some  $s_0 \in J$ ,

$$\mathcal{P}\colon \gamma \to \mathcal{P}_{\gamma}\colon s \to \mathcal{P}_{\gamma}(s) := P_{s_0 \to s}^{\gamma} \dot{\gamma}(s_0). \tag{15.2}$$

Evidently, the equality (15.1) is equivalent to

$$\mathcal{P}_{\gamma} = \dot{\gamma},\tag{15.3}$$

where  $\dot{\gamma} \colon s \mapsto \dot{\gamma}(s)$  is considered as a lifting of  $\gamma$  to T(M).

If  $\tau \in \text{PLift}(T(M), \pi, M)$  denotes the lifting of paths

$$\tau \colon \gamma \mapsto \tau_{\gamma} := \dot{\gamma}, \tag{15.4}$$

which may be called *tangent lifting of paths*, the equality (15.3) is tantamount to

$$\nabla D^{\gamma} \tau = 0 \tag{15.5}$$

as a result of the equality (3.21) on page 231. Here  $\nabla D$  is the derivation along paths in  $(T(M), \pi, M)$  generated by P via (3.19). Its components  $\Gamma_i^i$  coincide

with the coefficients of P (Proposition 3.7 on page 233) and along  $\gamma: J \to M$  in a frame  $\{E_i\}$  in T(M) over  $\gamma(J)$  are

$$\Gamma^{i}_{\ j}(s;\gamma) = \Gamma^{i}_{\ jk}(\gamma(s))\dot{\gamma}^{k}(s), \qquad s \in J$$
(15.6)

where  $\Gamma^{i}_{ik}$  are the coefficients of  $\nabla$  in  $\{E_i\}$  (see, e.g., Propositions 11.3 and 11.4).

The above conclusions can be summarized in the following assertion.

**Proposition 15.1.** A  $C^1$  path in a  $C^2$  manifold endowed with  $C^0$  linear connection is geodesic if and only if it satisfies any one (and hence all) of the equivalent equations (15.1), (15.3) and (15.5).

Remark 15.1. Applying equation (3.23) on page 231 one can prove that, due to (15.4) and (15.6), the equation (15.5) is an equivalent invariant form of the equation (I.3.22) of geodesic paths (see also equations (15.11) and (15.12) below). Thus (15.1), (15.3) and (15.5) are equivalent versions of the geodesic equation. Also the equation (15.5) is a rigorous version of (I.3.21) on page 30 for arbitrary, with or without self-intersections, paths (see Remark I.3.8 on page 31): if  $\gamma$  is not injective, one should define the symbol  $\nabla_{\dot{\gamma}}\dot{\gamma}$  as  $\nabla D^{\gamma}\tau$ ,  $\nabla_{\dot{\gamma}}\dot{\gamma} := \nabla D^{\gamma}\tau$ , otherwise, for an injective path  $\gamma$ , the lifting  $\dot{\gamma}$  of  $\gamma$  is equivalent to a vector field  $\gamma(s) \mapsto \dot{\gamma}(s)$ on  $\gamma(J)$  and  $\nabla_{\dot{\gamma}}\dot{\gamma}$  has its usual meaning if we identify this field with  $\dot{\gamma}$ .

One should observe, in (15.1), (15.3), and (15.5) are involved only the parallel transport P along paths and the generated by it derivation along paths  $\nabla D$ , not directly the linear connection  $\nabla$ . This observation hints to the existence of a generalization of the concept of geodesic paths to more general ones when P is replace with arbitrary linear transport along paths in the tangent bundle  $(T(M), \pi, M)$  over M. The following definition is a rigorous realization of such an idea.

**Definition 15.1.** Let M be a  $C^2$  manifold and L be a  $C^0$  linear transport along paths in its tangent bundle  $(T(M), \pi, M)$ . Call a  $C^1$  path  $\gamma: J \to M$  autoparallel with respect to L if

$$\dot{\gamma}(s) = L^{\gamma}_{s_o \to s}(\dot{\gamma}(s_0)) \tag{15.7}$$

for some (and hence any)  $s_0 \in J$  and all  $s \in J$ .

*Remark* 15.2. For other meaning of the term 'autoparallel path', see Remark I.3.7 on page 30.

**Proposition 15.2.** The autoparallel paths with respect to a parallel transport along paths generated by a linear connection coincide with the geodesics of this connection.

*Proof.* Compare (15.1) with (15.7) and invoke Proposition 15.1.

Similarly to the above considerations, we can rewrite (15.7) as

$$\mathcal{L}_{\gamma} = \dot{\gamma} \tag{15.8}$$

or

$$D^{\gamma}\tau = 0 \tag{15.9}$$

#### 15. Autoparallel paths

where

$$\mathcal{L}: \gamma \mapsto \mathcal{L}_{\gamma}: s \mapsto \mathcal{L}_{\gamma}(s) := L_{s_0 \to s}^{\gamma} \dot{\gamma}(s_0), \tag{15.10}$$

 $\tau$  is the tangent lifting of paths given via (15.4), and D is the derivation along paths generated by L via (3.19). The derivation of these equations is in fact the proof of the following proposition (cf. Proposition 15.1).

**Proposition 15.3.** A  $C^1$  path  $\gamma$  in  $C^2$  manifold is autoparallel with respect to a  $C^0$  linear transport along paths in the tangent bundle over that manifold if and only if for it any one (and hence both) of the equivalent conditions (15.8) and (15.9) is (are) valid.

From practical view-point, the equation of autoparallels (15.9) is quite suitable for the study of the (local) properties of the autoparallels. In particular this concerns its representation in a frame  $\{E_i\}$  along  $\gamma$  in T(M). As a consequence of (3.23) on page 231 with  $\lambda = \tau$  and (15.4), in  $\{E_i\}$ , the equation (15.9) is equivalent to

$$\frac{\mathrm{d}\dot{\gamma}^{i}(s)}{\mathrm{d}s} + \Gamma^{i}{}_{j}(s;\gamma)\dot{\gamma}^{j}(s) = 0$$
(15.11)

where  $\Gamma_{j}^{i}$  are the (local) coefficients of the transport L in  $\{E_{i}\}$ . In the special case when  $\{E_{i}\}$  is the frame induced by some local coordinates  $\{x^{i}\}, E_{i} = \frac{\partial}{\partial x^{i}}$ , the last equation transforms into

$$\frac{\mathrm{d}^2\gamma^i(s)}{\mathrm{d}s^2} + \Gamma^i{}_j(s;\gamma)\frac{\mathrm{d}\gamma^j(s)}{\mathrm{d}s} = 0$$
(15.12)

where  $s \in J$  must be such that  $\gamma(s)$  is in the domain of  $\{x^i\}$  and  $\gamma^i := x^i \circ \gamma$ .

In the case of a parallel transport, L = P when (15.6) holds, equations (15.11) and (15.12) reduce, respectively, to the equations (I.3.22) and (I.3.23) of the geodesics. Therefore they, as well as (15.9), should be called equation(s) of the autoparallel paths. It is clear, the equation of the autoparallels is a system of ordinary differential equations with respect to the local coordinates of the paths (in some local coordinates). The concrete order and type of this system depends on the transport L through its coefficients  $\Gamma^i_{\ j}$ . The investigation of the autoparallels' equation is out of the subject of our work. Below we present only one (almost trivial) result that generalizes the well know theorem that the geodesics are described via linear equations in normal coordinates (see the paragraph containing equation (II.2.15) on page 83 or, e.g., [11, Chapter III, Proposition 8.3]).

**Proposition 15.4.** Let M be  $C^2$  manifold and L be a  $C^0$  linear transport along paths in its tangent bundle  $(T(M), \pi, M)$ . Suppose L admits an autoparallel path  $\gamma: J \to M$  such that  $L_{s \to t}^{\gamma} = \operatorname{id}_{\pi^{-1}(\gamma(s))}$  for all  $s, t \in J$  with  $\gamma(s) = \gamma(t)$ , i.e., the mapping  $\gamma(s) \mapsto \mathcal{L}_{\gamma}(s) = \dot{\gamma}(s)$  (see (15.10) or (15.9)) is a single-valued vector field over  $\gamma(J)$ . Then, in any frame  $\{E_i\}$  normal along  $\gamma$  for L, the components  $\dot{\gamma}^i(s)$ of  $\dot{\gamma}(s) \in T_{\gamma(s)}(M)$  are constant, i.e., we have

$$\dot{\gamma}(s) = a^i E_i \tag{15.13}$$

for some numbers  $a^i \in \mathbb{K}$ . Besides, if L is torsionless,  $\gamma$  is injective and (V, x) is a chart normal along  $\gamma$  for L, then

$$x^{i}(\gamma(s)) = a^{i}s + b^{i}, \quad \dot{\gamma}(s) = a^{i}\frac{\partial}{\partial x^{i}}$$
(15.14)

where  $\{x^i\}$  is the associated to (V, x) normal coordinate system,  $s \in J$  is such that  $\gamma(s) \in V$ , and  $a^i, b^i \in \mathbb{K}$  are constant numbers.

Remark 15.3. The condition  $L_{s\to t}^{\gamma} = \mathrm{id}_{\pi^{-1}(\gamma(s))}$  for all  $s, t \in J$  with  $\gamma(s) = \gamma(t)$ , by Proposition 5.2 on page 240, is a criterion for the existence of frames normal along  $\gamma$  while, by Proposition 10.7, the torsionless of L and the injectiveness of  $\gamma$ ensure the existence of coordinates normal along  $\gamma$ .

*Proof.* Recall, by definition, the coefficients  $\Gamma^i{}_j$  of L vanish in a normal frame or in normal coordinates. So, evidently, (15.13) (resp. (15.14)) is the general solution of the equations of autoparallels (15.11) (resp. (15.12)) in a normal frame (resp. normal coordinate system  $\{x^i\}$ ).

Consequently, in the sense of Proposition 15.4, the autoparallel paths in a manifold with a transport in its tangent bundle are the analogues of the straight lines (resp. geodesics) in a Euclidean space  $\mathbb{E}^n$  (resp. manifold with a linear connection).

## 16. On a fibre bundle view at quantum mechanics

The purpose of this section is to be given an idea of the fibre bundle formulation of nonrelativistic quantum mechanics which essentially employs the theory of linear transports along paths in vector bundles; the reader can find in [127–131] a detailed exposition of this approach to quantum mechanics. In particular, it will be demonstrated below that the bundle Schrödinger and Heisenberg pictures of motion are (locally) identical in a suitable normal frame

The reader should be aware that some of the considerations below will not be quite rigorous from mathematical view-point. This is a result of the fact that Hilbert bundles of generally infinite fibre dimension will be used. Correspondingly, we will be dealing with infinite matrices applying the rules known from the theory of finite matrices, which is a common practice in quantum mechanics [68,133–135] but such an approach requires additional investigation and even may turn to be wrong sometimes [126] (see also [136] and [135, Chapter 7, Section 18]).

Before going on, we shall make a technical remark. The (fibre) indices below can take discrete and/or continuous values depending on the spectrum of the corresponding operators (the Hamiltonian in the particular case). For that reason, sums like  $\lambda^i e_i(x)$  must be understood as  $\sum_{i \in \Lambda_d} \lambda^i e_i(x) + \int_{i \in \Lambda_c} \lambda^i e_i(x) di$ , where  $\Lambda_d$  (resp.  $\Lambda_c$ ) is the set of discrete (resp. continuous) values the index *i* can take. For more details on this item, see [128, p. 4921].

#### 16. On a fibre bundle view at quantum mechanics

In the bundle approach to quantum mechanics, to a quantum system is assigned a unique  $C^1$  Hilbert bundle  $(F, \pi, M)$ , M being a  $C^1$  manifold, whose (standard) fibre is identified with the usual Hilbert space of states  $\mathcal{F}$  of the quantum system. The system state is described via a lifting of paths  $\Psi \in \text{PLift}(F, \pi, M)$ such that, if  $\gamma: J \to M$  is a  $C^1$  path,

$$\Psi\colon \gamma\mapsto \Psi_{\gamma}\colon t\in J\mapsto \Psi_{\gamma}(t)=l_{\gamma(t)}^{-1}(\psi(t))\in \pi^{-1}(\gamma(t)),$$
(16.1)

where  $l_x^{-1}: \mathcal{F} \to \pi^{-1}(x), x \in M$ , are fixed isomorphisms and  $\psi(t) \in \mathcal{F}$  is the conventional state vector of the system at a moment  $t \in J$ .

The time evolution of the system (or of its state lifting  $\Psi$ ) is governed by the equation

$$\Psi_{\eta}(t) = U_{\gamma}(t,s)(\Psi_{\gamma}(s)) \qquad s, t \in J,$$
(16.2)

where the evolution transport

$$U: \gamma \mapsto U_{\gamma}: (s,t) \mapsto U_{\gamma}(t,s): \pi^{-1}(\gamma(s)) \to \pi^{-1}(\gamma(t))$$

is a (Hermitian and isometric) linear transport along paths in  $(F, \pi, M)$  such that

$$U_{\gamma}(t,s) = l_{\gamma(t)}^{-1} \circ \mathcal{U}(t,s) \circ l_{\gamma(s)} \colon \pi^{-1}(\gamma(s)) \to \pi^{-1}(\gamma(t))$$
(16.3)

with  $\mathcal{U}(t,s)$  being the system's evolution operator in  $\mathcal{F}$ . It is a remarkable fact that the matrix  $\Gamma(s;\gamma)$  of the coefficients of U is

$$\boldsymbol{\Gamma}(s;\gamma) = -\frac{1}{\mathrm{i}\hbar} \boldsymbol{H}_{\gamma}^{\mathrm{m}}(t)$$
(16.4)

where  $i = +\sqrt{-1}$  is the imaginary unit,  $\hbar$  is the Planck constant (deviled by  $2\pi$ ) and the matrix-bundle Hamiltonian  $H^m_{\gamma}(t)$  is

$$\boldsymbol{H}_{\gamma}^{\mathrm{m}}(t) = \boldsymbol{l}_{\gamma(t)}^{-1}(t)\boldsymbol{\mathcal{H}}(t)\boldsymbol{l}_{\gamma(t)}(t) - \mathrm{i}\hbar\boldsymbol{l}_{\gamma(t)}^{-1}(t) \Big(\frac{\mathrm{d}\boldsymbol{l}_{\gamma(t)}(t)}{\mathrm{d}t} + \boldsymbol{E}(t)\boldsymbol{l}_{\gamma(t)}(t)\Big).$$
(16.5)

Here  $\mathcal{H}(t)$  is the matrix of the system's Hamiltonian in a basis  $\{f_i(t)\}$  of  $\mathcal{F}, \mathbf{l}_{\gamma(t)}(t)$  is the matrix of  $l_{\gamma(t)}$  in this basis and a frame  $\{e_i\}$  over M in F, and the matrix  $\mathbf{E}(t) := [E_i^j(t)]$  is defined via the expansion  $\frac{\mathrm{d}f_i(t)}{\mathrm{d}t} = E_i^j(t)f_j(t)$ .

If D is the derivation along paths generated by U (see Definition 3.2 on page 230), then the state lifting  $\Psi$  is a solution of the bundle Schrödinger equation

$$D(\Psi) = 0 \tag{16.6}$$

satisfying some initial condition, like  $\Psi_{\gamma}(t_0) = \Psi_{\gamma}^0$  for fixed  $t_0 \in J$  and  $\Psi_{\gamma}^0 \in \pi^{-1}(\gamma(t_0))$ .

Let  $\mathcal{A}(t): \mathcal{F} \to \mathcal{F}$  be the linear Hermitian (self-adjoint) operator corresponding to a dynamical variable  $\mathbb{A}$  in the standard quantum mechanics. In the bundle approach, to A corresponds a lifting A of paths in the bundle of restricted M-morphisms of  $(F, \pi, M)$  such that

$$A_{\gamma}(t) = l_{\gamma(t)}^{-1} \circ \mathcal{A}(t) \circ l_{\gamma(t)} \colon \pi^{-1}(\gamma(t)) \to \pi^{-1}(\gamma(t)).$$
(16.7)

The description of a quantum system via liftings like  $\Psi$  and A is known as the bundle Schrödinger picture of motion. The shift to the bundle Heisenberg picture of motion is via the changes

$$\Psi \mapsto \Psi^{\mathrm{H}} \colon (t, t_0; \gamma) \mapsto \Psi^{\mathrm{H}}_{\gamma, t}(t_0) \qquad A \mapsto A^{\mathrm{H}} \colon (t, t_0; \gamma) \mapsto A^{\mathrm{H}}_{\gamma, t}(t_0), \qquad (16.8)$$

where  $t, t_0 \in J$  and

$$\Psi_{\gamma,t}^{\rm H}(t_0) = \Psi_{\gamma}(t_0) = l_{\gamma(t_0)}^{-1}(\psi(t_0)) = U_{\gamma}(t_0,t)\Psi_{\gamma}(t)$$
(16.9a)

$$A_{\gamma,t}^{\rm H}(t_0) = U_{\gamma}(t,t_0)^{-1} \circ A_{\gamma}(t) \circ U_{\gamma}(t,t_0).$$
(16.9b)

(The mapping  $A \mapsto A^{\text{H}}$  defines a linear transport in the bundle of point restricted morphisms – see [127, p. 4903] and [129, p. 4941].) Thus, in the bundle Heisenberg picture, the time dependence is entirely transferred to the observables liftings of paths. The Schrödinger and Heisenberg pictures of motion are completely equivalents from physical point of view as the mean values of the dynamical variables in them are identical.

Let  $\{e_i\}$  be a frame in F (locally) normal along  $\gamma$  for the evolution transport U.<sup>1</sup> By Definition 4.1 this means that the matrix U of U in  $\{e_i\}$  is the (corresponding infinite) identity matrix,  $U(t, t_0) = \mathbb{1}$ . Consequently equations (16.9) take the following form in the normal frame  $\{e_i\}$ :

$$\boldsymbol{\Psi}_{\gamma,t}^{\mathrm{H}}(t_0) = \boldsymbol{\Psi}_{\gamma}(t_0) = \boldsymbol{\Psi}_{\gamma}(t)$$
(16.10a)

$$\boldsymbol{A}_{\gamma,t}^{\mathrm{H}}(t_0) = \boldsymbol{A}_{\gamma}(t). \tag{16.10b}$$

In this way we have proved the following result.

**Proposition 16.1.** The bundle Schrödinger and Heisenberg pictures of motion are identical in a frame (locally) normal along the reference path  $\gamma$ .

## 17. Conclusion

The following three main topics have found their place in this chapter. First, the grounds of the theory of linear transports along paths in vector bundles are set and its relations with the one of parallel transports and connections in these bundles are revealed. Second, the theory of normal frames for such transports was developed. And third, it was demonstrated that the obtained formalism naturally

<sup>&</sup>lt;sup>1</sup>Here we suppose that the condition (5.2) with U for L in Proposition 5.1 is fulfilled.

agrees with the already existing theory of frames (and coordinates) normal for (linear) connections and other derivations of the tensor algebra over a differentiable manifold.

It should be noticed, from the view-point of linear transports, a lot of the earlier-obtained results concerning linear connections (or other derivations) look considerably simpler and more natural. Here are two typical examples. (i) A linear transport admits frames normal on some set iff it is flat on it; a linear connection admits frames normal on a neighborhood iff it is flat on it. (ii) If a linear transport admits frames normal on some set, such a frame can be obtained via transportation by means of the transport of a basis over a fixed point in this set to the other points in it along paths lying entirely in that set; previously this result was known only on neighborhoods for linear connections and the parallel transports generated by them: a frame normal on a neighborhood for a flat linear connection can be obtained by parallelly transporting a basis over a fixed point in this neighborhood to the other points in it along paths lying in the neighborhoods.

Separately should be mentioned the case of linear transports along paths in the tangent bundle over a manifold. In it, the flat parallel transports generated by linear connections are the only ones admitting normal frames. This conclusion is similar to the one derived in Chapter III concerning frames normal for derivations along vector fields and linear connections. Besides, the torsionless flat parallel transports are the only ones admitting normal coordinates.

At the end, we would like to express the hope that the material in the present chapter could serve as a groundwork for a systematic exploration of frames (and, possibly, coordinates) normal for different kinds of derivations or transports along paths in infinitely dimensional vector bundles, in particular for linear connections in infinitely dimensional manifolds. Partially or *mutatis mutandis* some of our results remain valid in the infinite-dimensional case, but, rigorously speaking, this case is not studied until now and it is open for further research.

## Chapter V

# Normal Frames for Connections on Differentiable Fibre Bundles

The general connection theory on differentiable fibre bundles, with emphasis on the vector ones, is partially considered. The theory of frames normal for general connections on these bundles is developed. Links with the theory of frames normal for linear connections in vector bundles are revealed. Existence of buncoordinates dle normal at a given point and/or along injective horizontal path is proved and a necessary and sufficient condition of existence of bundle coordinates normal along injective horizontal mappings is proved. The concept of a transport along paths in differentiable bundles is introduced. Different links between connections, parallel transports (along paths) and transports along paths are investigated.

## 1. Introduction

All connections considered until now, on manifolds and on vector bundles, were linear. It is well known that there exist non-linear connections on vector bundles as well as on non-vector ones. Can normal frames (and/or coordinates) be introduced for such more general connections? The positive solution of that problem is the main goal of the present chapter of this book. For the purpose and for a comparison with the definitions and results already obtained is required some preliminary material on general connection theory on differentiable bundles, which is collected in Sections 2–5. On its base, the normal frames for connections on such bundles are studied in Sections 6 and 7.

Sections 2–5 follow the work [137],<sup>1</sup> Sections 6 and 7 are a slightly revised version of [139], and Section 8 reproduces in a modified form the paper [118]

The work is organized as follows.

In Section 2 is collected some introductory material, like the notion of Lie derivatives and distributions on manifolds, needed for our exposition. Here some of our notation is fixed too.

Section 3 is devoted to the general connection theory on bundles whose base and bundles spaces are differentiable manifolds. From different view-points, this theory can be found in many works, like [6, 7, 10–13, 16, 28, 60, 98, 106, 107, 117, 138, 140–146]. In Subsection 3.1 are reviewed some coordinates and frames/bases on the bundle space which are compatible with the fibre structure of a bundle. Subsection 3.2 deals with the general connection theory. A connection on a bundle is defined as a distribution on its bundle space which is complimentary to the vertical distribution on it. The notions of parallel transport generated by connection and of specialized frame are introduced. The fibre coefficients and fibre components of the curvature of a connection are defined via part of the components of the anholonomicity object of a specialized frame. Frames adapted to local bundle coordinates are introduced and the local (2-index) coefficients in them of a connection are defined; their transformation law is derived and it is proved that a geometrical object with such transformation law uniquely defines a connection.

In Section 4, the general connection theory from Section 3 is specified on vector bundles. The most important structures in/on them are the ones that are consistent/compatible with the vector space structure of their fibres. The vertical lifts of sections of a vector bundle and the horizontal lifts of vector fields on its base are investigated in more details in Subsection 4.1. Subsection 4.2 is devoted to linear connections on vector bundles, i.e., connections such that the assigned to them parallel transport is a linear mapping. It is proved that the 2-index coefficients of a linear connection are linear in the fibre coordinates, which leads to the introduction of the (3-index) coefficients of the connection; the latter coefficients

<sup>&</sup>lt;sup>1</sup>The presentation of the material in Sections 2-4 is according to some of the main ideas of [138, Chapters 1 and 2], but their realization here is quite different and follows the modern trends in differential geometry.

being defined on the base space. The transformations of different objects under changes of vector bundle coordinates are explored. The covariant derivatives are introduced and investigated in Subsection 4.3. They are defined via the Lie derivatives [138] and a mapping realizing an isomorphism between the vertical vector fields on the bundle space and the sections of the bundle. The equivalence of that definition with the widespread one, defining them as mappings on the module of sections of the bundle with suitable properties, is proved. In Subsection 4.4, the affine connections on vector bundles are considered briefly.

In Section 5, some of the results of the previous sections are generalized when frames more general than the ones generated by local coordinates on the bundle space are employed. The most general such frames, compatible with the fibre structure, and the frames adapted to them are investigated. The main differentialgeometric objects, introduced in the previous sections, are considered in such general frames. Particular attention is paid on the case of a vector bundle. In vector bundles, a bijective correspondence between the mentioned general frames and pairs of bases, in the vector fields over the base and in the sections of the bundle, is proved. The (3-index) coefficients of a connection in such pairs of frames and their transformation laws are considered. The covariant derivatives are also mentioned on that context.

The theory of normal frames for connections on bundles is considered in Section 6. Subsection 6.1 deals with the general case. Loosely said, an adapted frame is called normal if the 2-index coefficients of a connection vanish in it (on some set). It happens that a frame is normal if and only if it coincides with the frame it is adapted to. The set of these frames is completely described in the most general case. The problems of existence, uniqueness, etc. of normal frames adapted to holonomic frames, i.e., adapted to local coordinates, are discussed in Subsection 6.2. If such frames exist, their general form is described. The existence of frames normal at a given point and/or along an injective horizontal path is proved. The flatness of a connection on an open set is pointed as a necessary condition of existence of (locally) holonomic frames normal on that set. Some links between the general theory of normal frames and the one of normal frames in vector bundles, presented in Chapter IV, are given in Subsection 6.3. It is proved that a frame is normal on a vector bundle with linear connection if and only if in it vanish the 3-index coefficients of the connection. The equivalence of the both theories on vector bundles is established.

In Section 7 is formulated and proved a necessary and sufficient condition for existence of coordinates normal along injective mappings with non-vanishing horizontal component, in particular along injective horizontal mappings.

Section 8 is devoted to some aspects of the axiomatical approach to parallel transport theory [17,23,30–33,91,147–150] and its relations to connection theory; it is based on the paper [118]. It starts with a definition of a transport along paths in a bundle and a result stating that, under some assumptions, it defines a connection. The most important properties of the parallel transports generated

by connections are used to be (axiomatically) defined the concept of a parallel transport (irrespectively to some connection on a bundle). In a series of results are constructed bijective mappings between the sets of transports along paths satisfying some additional conditions, connections, and parallel transports. In this way, two different, but equivalent, systems of axioms defining the concept "parallel transport" will be established.

The chapter ends with some remarks and conclusions in Section 9.

## 2. Preliminaries

This section contains an introductory material, notation etc. that will be needed for our exposition. The reader is referred for details to Chapter I, Section III.2, and standard books on differential geometry, like [7, 23, 151].

A differentiable finite-dimensional manifold over a field  $\mathbb{K}$  will be denoted typically by M. Here  $\mathbb{K}$  stands for the field  $\mathbb{R}$  of real or the field  $\mathbb{C}$  of complex numbers,  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ . The manifolds we consider are supposed to be smooth of class  $C^{2,1}$  The set of vector fields, realized as first-order differential operators, and of differential k-forms,  $k \in \mathbb{N}$ , over M will be denoted by  $\mathcal{X}(M)$  and  $\Lambda^k(M)$ , respectively. The space tangent (resp. cotangent) to M at  $p \in M$  is  $T_p(M)$  (resp.  $T_p^*(M)$ ) and  $(T(M), \pi_T, M)$  (resp.  $(T^*(M), \pi_{T^*}, M)$ ) will stand for the tangent (resp. cotangent) bundle over M. The value of  $X \in \mathcal{X}(M)$  at  $p \in M$  is  $X_p \in T_p(M)$ and the action of X on a  $C^1$  function  $\varphi \colon M \to \mathbb{K}$  is a function  $X(\varphi) \colon M \to \mathbb{K}$ such that  $X(\varphi)|_p \coloneqq X_p(\varphi) \in \mathbb{K}$ .

If M and  $\overline{M}$  are manifolds and  $f: \overline{M} \to M$  is a  $C^1$  mapping, then  $f_* := df: T(\overline{M}) \to T(M)$  denotes the induced tangent mapping (or differential) of f such that, for  $p \in \overline{M}$ ,  $f_*|_p := df|_p := T_p(f): T_p(\overline{M}) \to T_{f(p)}(M)$  and, for a  $C^1$  function g on M,  $(f_*(X))(g) := X(g \circ f): p \mapsto f_*|_p(g) = X_p(g \circ f)$ , with  $\circ$  being the composition of mappings sign. Respectively, the induced cotangent mapping is  $f^*: T^*(M) \to T^*(\overline{M})$ . If  $h: N \to \overline{M}$ , N being a manifold, we have the chain rule  $d(f \circ h) = df \circ dh$ , which is an abbreviation for  $d(f \circ h)_q = (df)_{f(q)} \circ (dh)_q$  for  $q \in N$ .

By  $J \subseteq \mathbb{R}$  will be denoted an arbitrary real interval that can be open or closed at one or both its ends. The notation  $\gamma: J \to M$  represents an arbitrary path in M. For a  $C^1$  path  $\gamma: J \to M$ , the vector tangent to  $\gamma$  at  $s \in J$  will be denoted by  $\dot{\gamma}(s) := \frac{d}{dt}\Big|_{t=s} (\gamma(t)) = \gamma_* (\frac{d}{dr}|_s) \in T_{\gamma(s)}(M)$ , where r in  $\frac{d}{dr}|_s$  is the standard coordinate function on  $\mathbb{R}$ , i.e.,  $r: \mathbb{R} \to \mathbb{R}$  with r(s) := s for all  $s \in \mathbb{R}$  and hence  $r = \mathrm{id}_{\mathbb{R}}$  is the identity mapping of  $\mathbb{R}$ . If  $s_0 \in J$  is an end point of J and J

<sup>&</sup>lt;sup>1</sup>Some of our definitions or/and results are valid also for  $C^1$  or even  $C^0$  manifolds, but we do not want to overload the material with continuous counting of the required degree of differentiability of the manifolds involved. Some parts of the text admit generalizations on more general spaces, like the topological ones, but this is out of the subject of the present work.

#### 2. Preliminaries

is closed at  $s_0$ , the derivative in the definition of  $\dot{\gamma}(s_0)$  is regarded as a one-sided derivative at  $s_0$ .

The Lie derivative relative to  $X \in \mathcal{X}(M)$  will be denoted by  $\mathcal{L}_X$ . It is defined on arbitrary geometrical objects on M [89], but below we shall be interested in its action on tensor fields [11, Chapter I, § 2] (see also [152]). If f, Y, and  $\theta$  are  $C^1$ respectively function, vector field and 1-form on M, then

$$\mathcal{L}_X(f) = X(f) \tag{2.1a}$$

$$\mathcal{L}_X(Y) = [X, Y]$$
(2.1b)

$$(\mathcal{L}_X(\theta))(Y) = X(\theta(Y)) - \theta([X,Y]) = (\mathrm{d}\theta)(X,Y) + Y(\theta(X)), \qquad (2.1c)$$

where  $[A, B]_{-} = A \circ B - B \circ A$  is the commutator of operators A and B (with common domain) and d denotes the exterior derivative operator.

Since  $\mathcal{L}_X$  is a derivation of the tensor algebra over the vector fields on M, for a tensor field  $S: \Lambda^1(M) \times \cdots \times \Lambda^1(M) \times \mathcal{X}(M) \times \cdots \times \mathcal{X}(M) \to \mathfrak{F}(M)$ , we have

$$(\mathcal{L}_X S)(\theta, \dots; Y, \dots) = X(S(\theta, \dots; Y, \dots)) - S(\mathcal{L}_X \theta, \dots; Y, \dots) - \dots - S(\theta, \dots; \mathcal{L}_X Y, \dots)) - \dots,$$
(2.2)

which defines  $\mathcal{L}_X S$  explicitly, due to (2.1).

Let the Greek indices  $\lambda, \mu, \nu, \ldots$  run over the range  $1, \ldots, \dim M$  and  $\{E_{\mu}\}$ be a  $C^{1}$  frame in T(M), i.e.,  $E_{\mu} \in \mathcal{X}(M)$  be of class  $C^{1}$  and, for each  $p \in M$ , the set  $\{E_{\mu}|_{p}\}$  be a basis of the vector space  $T_{p}(M)$ .<sup>2</sup> Let  $\{E^{\mu}\}$  be the coframe dual to  $\{E_{\mu}\}$ , i.e.,  $E^{\mu} \in \Lambda^{1}(M)$ ,  $\{E^{\mu}|_{p}\}$  be a basis in  $T_{p}^{*}(M)$ , and  $E^{\mu}(E_{\nu}) = \delta_{\nu}^{\mu}$  with  $\delta_{\mu}^{\nu}$  being the Kronecker deltas  $(\delta_{\mu}^{\nu} = 1 \text{ for } \mu = \nu \text{ and } \delta_{\mu}^{\nu} = 0 \text{ for } \mu \neq \nu)$ . Assuming the Einstein's summation convention (summation on indices repeated on different levels over the whole range of their values), we define the *components*  $(\Gamma_{X})^{\mu}{}_{\nu}$  of  $\mathcal{L}_{X}$  in (relative to)  $\{E_{\mu}\}$  via the expansion

$$\mathcal{L}_X E_\mu =: (\Gamma_X)^\nu_{\ \mu} E_\nu \tag{2.3}$$

which is equivalent to

$$\mathcal{L}_X E^\mu = -(\Gamma_X)^\mu{}_\nu E^\nu \tag{2.3'}$$

by virtue of  $E^{\mu}(E_{\nu}) = \delta^{\mu}_{\nu}$  and the commutativity of the Lie derivatives and contraction operators.<sup>3</sup> From (2.3) and (2.1b), we get

$$(\Gamma_X)^{\nu}{}_{\mu} = -E_{\mu}(X^{\nu}) - C^{\nu}_{\mu\lambda}X^{\lambda}, \qquad (2.4)$$

<sup>&</sup>lt;sup>2</sup>There are manifolds, like the even-dimensional spheres  $\mathbb{S}^{2k}$ ,  $k \in \mathbb{N}$ , which do not admit global, continuous (and moreover  $C^k$  for  $k \geq 1$ ), and nowhere vanishing vector fields [153]. If this is the case, the considerations must be localized over an open subset of M on which such fields exist. We shall not overload our exposition with such details.

<sup>&</sup>lt;sup>3</sup>The sign before  $(\Gamma_X)^{\mu}{}_{\nu}$  in (2.3) or (2.3') is conventional and we have chosen it in a way similar to the accepted convention for the components of a covariant derivative (or, equivalently, the coefficients of a linear connection – see Section 4).

in  $\{E_{\mu}\}$ , where  $X = X^{\mu}E_{\mu}$  and the functions  $C^{\nu}_{\mu\lambda}$ , known as the components of the anholonomicity object of  $\{E_{\mu}\}$ , are defined by

$$[E_{\mu}, E_{\nu}]_{-} =: C^{\lambda}_{\mu\nu} E_{\lambda} \tag{2.5}$$

or, equivalently, by its dual (see (2.1c))

$$dE^{\lambda} = -\frac{1}{2}C^{\lambda}_{\mu\nu}E^{\mu} \wedge E^{\nu}, \qquad (2.5')$$

with  $\wedge$  being the exterior (wedge) product sign.<sup>4</sup> Equation (2.4) is a special case of (III.2.7) for  $S_X = 0$ . The explicit local action of  $\mathcal{L}_X$  on a general  $C^1$  tensor field is given by (III.2.1).

A frame  $\{E_{\mu}\}$  or its dual coframe  $\{E^{\mu}\}$  is called *holonomic* (anholonomic) if  $C_{\mu\nu}^{\lambda} = 0$  ( $C_{\mu\nu}^{\lambda} \neq 0$ ) for all (some) values of the indices  $\mu$ ,  $\nu$ , and  $\lambda$ . For a holonomic frame there always exist local coordinates  $\{x^{\mu}\}$  on M such that *locally*  $E_{\mu} = \frac{\partial}{\partial x^{\mu}}$  and  $E^{\mu} = dx^{\mu}$ . Conversely, if  $\{x^{\mu}\}$  are local coordinates on M, then the local frame  $\{\frac{\partial}{\partial x^{\mu}}\}$  and local coframe  $\{dx^{\mu}\}$  are well defined and holonomic on the domain of  $\{x^{\mu}\}$ . For more information concerning (an)holonomic frames, see Section I.8.

A straightforward calculation by means of (2.5) reveals that a change

$$\{E_{\mu}\} \to \{\bar{E}_{\mu} = B^{\nu}_{\mu}E_{\nu}\}$$
 (2.6)

of the frame  $\{E_{\mu}\}$ , where  $B = [B^{\nu}_{\mu}]$  is a non-degenerate matrix-valued function, entails the transformation

$$C^{\lambda}_{\mu\nu} \mapsto \bar{C}^{\lambda}_{\mu\nu} = (B^{-1})^{\lambda}_{\varrho} \left( B^{\sigma}_{\mu} E_{\sigma}(B^{\varrho}_{\nu}) - B^{\sigma}_{\nu} E_{\sigma}(E^{\varrho}_{\mu}) + B^{\sigma}_{\mu} B^{\tau}_{\nu} C^{\varrho}_{\sigma\tau} \right).$$
(2.7)

Besides, from (2.4) and (2.7), we see that the quantities  $(\Gamma_X)^{\nu}{}_{\mu}$  undergo the change (cf. equation (III.2.10))

$$(\Gamma_X)^{\nu}{}_{\mu} \mapsto (\bar{\Gamma}_X)^{\nu}{}_{\mu} = (B^{-1})^{\mu}_{\varrho} \big( (\Gamma_X)^{\varrho}{}_{\sigma} B^{\sigma}_{\nu} + X(B^{\sigma}_{\nu}) \big)$$
(2.8)

when (2.6) takes place. Setting  $\Gamma_X := [(\Gamma_X)^{\nu}{}_{\mu}]$  and  $\overline{\Gamma}_X := [(\overline{\Gamma}_X)^{\nu}{}_{\mu}]$ , we can rewrite (2.8) in a more compact matrix form as (cf. (III.2.11))

$$\Gamma_X \mapsto \bar{\Gamma}_X = B^{-1} \cdot (\Gamma_X \cdot B + X(B)).$$
(2.9)

If  $n \in \mathbb{N}$  and  $n \leq \dim M$ , an *n*-dimensional distribution  $\Delta$  on M is defined as a mapping  $\Delta \colon p \mapsto \Delta_p$  assigning to each  $p \in M$  an *n*-dimensional subspace  $\Delta_p$ of the tangent space  $T_p(M)$  of M at  $p, \Delta_p \subseteq T_p(M)$ . A distribution is *integrable* if there is a submersion  $\psi \colon M \to N$  such that Ker  $\psi_* = \Delta$ ; a necessary and locally

<sup>&</sup>lt;sup>4</sup>If M is a Lie group and  $\{E_{\mu}\}$  is a basis of its Lie algebra (defined as the set of left invariant vector fields in on M), then  $C^{\lambda}_{\mu\nu}$  are constants, called structure constants of M, and (2.5) and (2.5') are known as the structure equations of M.

sufficient condition for the integrability of  $\Delta$  is the commutator of every two vector fields in  $\Delta$  to be in  $\Delta$ . We say that a vector field  $X \in \mathcal{X}(M)$  is in  $\Delta$  and write  $X \in \Delta$ , if  $X_p \in \Delta_p$  for all  $p \in M$ . A basis on  $U \subseteq M$  for  $\Delta$  is a set  $\{X_1, \ldots, X_n\}$ of *n* linearly independent (relative to functions  $U \to \mathbb{K}$ ) vector fields in  $\Delta|_U$ , i.e.,  $\{X_1|_p, \ldots, X_n|_p\}$  is a basis for  $\Delta_p$  for all  $p \in U$ .

A distribution is convenient to be described in terms of (global) frames or/and coframes over M. In fact, if  $p \in M$  and  $\varrho = 1, \ldots, n$ , in each  $\Delta_p \subseteq T_p(M)$ , we can choose a basis  $\{X_{\varrho}|_p\}$  and hence a frame  $\{X_{\varrho}\}, X_{\varrho}: p \mapsto X_{\varrho}|_p$ , in  $\{\Delta_p: p \in M\} \subseteq T(M)$ ; we say that  $\{X_{\rho}\}$  is a basis for/in  $\Delta$ . Conversely, any collection of n linearly independent (relative to functions  $M \to \mathbb{K}$ ) vector fields  $X_{\varrho}$  on Mdefines a distribution  $p \mapsto \{\sum_{\varrho=1}^n f^\varrho X_{\varrho}|_p: f^\varrho \in \mathbb{K}\}$ . Consequently, a frame in T(M) can be formed by adding to a basis for  $\Delta$  a set of (dim M - n) new linearly independent vector fields (forming a frame in  $T(M) \setminus \{\Delta_p: p \in M\}$ ) and v.v., by selecting n linearly independent vector fields on M, we can define a distribution  $\Delta$  on M. Equivalently, one can use dim M - n linearly independent 1-forms  $\omega^a$ ,  $a = n + 1, \ldots, \dim M$ , which are annihilators for it,  $\omega^a|_{\Delta_p} = 0$  for all  $p \in M$ . For instance, if  $\{X_{\mu}: \mu = 1, \ldots, \dim M\}$  is a frame in T(M) and  $\{X_{\varrho}: \varrho = 1, \ldots, n\}$ is a basis for  $\Delta$ , then one can define  $\omega^a$  to be elements in the coframe  $\{\omega^{\mu}\}$  dual to  $\{X_{\mu}\}$ . We call  $\{\omega^a\}$  a *cobasis* for  $\Delta$ .

## 3. Connections on bundles

Before presenting the general connection theory in Subsection 3.2, we shall fix first some notation and concepts concerning fibre bundles in Subsection 3.1. The reader is referred for more details to Subsection IV.2.1 and to the literature cited in Section IV.1.

### **3.1.** Frames and coframes on the bundle space

Let  $(E, \pi, M)$  be a bundle with bundle space E, projection  $\pi \colon E \to M$ , and base space M. Suppose that the spaces M and E are manifolds of finite dimensions  $n \in \mathbb{N}$  and n+r, for some  $r \in \mathbb{N}$ , respectively; so the dimension of the fibre  $\pi^{-1}(x)$ , with  $x \in M$ , i.e., the fibre dimension of  $(E, \pi, M)$ , is r. Besides, let these manifolds be  $C^2$  differentiable, if the opposite is not stated explicitly.<sup>1</sup>

Let the Greek indices  $\lambda, \mu, \nu, \dots$  run from 1 to  $n = \dim M$ , the Latin indices  $a, b, c, \dots$  take the values from n + 1 to  $n + r = \dim E$ , and the uppercase Latin

<sup>&</sup>lt;sup>1</sup>Most of our considerations are valid also if  $C^1$  differentiability is assumed and even some of them hold on  $C^0$  manifolds. By assuming  $C^2$  differentiability, we skip the problem of counting the required differentiability class of the whole material that follows. Sometimes, the  $C^2$ differentiability is required explicitly, which is a hint that a statement or definition is not valid otherwise. If we want to emphasize that some text is valid under a  $C^1$  differentiability assumption, we indicate that fact explicitly. However, the proofs of Proposition 6.5 and all assertions in Section 7 require  $C^3$  differentiability, which will be indicated explicitly.

indices  $I, J, K, \ldots$  take values in the whole set  $\{1, \ldots, n+r\}$ . One may call these types of indices respectively base, fibre, and bundle indices.

Suppose  $\{u^I\} = \{u^{\mu}, u^a\} = \{u^1, \dots, u^{n+r}\}$  are local bundle coordinates on an open set  $W \subseteq E$ , i.e., on the set  $\pi(W) \subseteq M$  there are local coordinates  $\{x^{\mu}\}$ such that  $u^{\mu} = x^{\mu} \circ \pi$ ;<sup>2</sup> the coordinates  $\{u^{\mu}\}$  (resp.  $\{u^a\}$ ) are called *basic* (resp. fibre) coordinates [23].<sup>3</sup>

Further only coordinate changes

$$\{u^{\mu}, u^{a}\} \mapsto \{\tilde{u}^{\mu}, \tilde{u}^{a}\}$$
(3.1a)

on E between bundle coordinates will be considered. This means that

$$\tilde{u}^{\mu}(p) = f^{\mu}(u^{1}(p), \dots, u^{n}(p))$$
  

$$\tilde{u}^{a}(p) = f^{a}(u^{1}(p), \dots, u^{n}(p), u^{n+1}(p), \dots, u^{n+r}(p))$$
(3.1b)

for  $p \in U \cap \tilde{U}$ , with U being the domain of the coordinates  $\{\tilde{u}^I\}$  and some functions  $f^I$ . The bundle coordinates  $\{u^{\mu}, u^a\}$  induce the (local) frame  $\{\partial_{\mu} := \frac{\partial}{\partial u^{\mu}}, \partial_a := \frac{\partial}{\partial u^a}\}$  and coframe  $\{du^{\mu}, du^a\}$  over W in respectively the tangent T(E) and cotangent  $T^*(E)$  bundle spaces of the tangent and cotangent bundles over the bundle space E. Since a change (3.1) of the coordinates on E implies  $\partial_I \mapsto \tilde{\partial}_I := \frac{\partial}{\partial \tilde{u}^I} = \frac{\partial u^J}{\partial \tilde{u}^I} \partial_J$  and  $du^I \mapsto d\tilde{u}^I = \frac{\partial \tilde{u}^I}{\partial u^J} du^J$ , the transformation (3.1) leads to

$$(\partial_{\mu}, \partial_{a}) \mapsto (\tilde{\partial}_{\mu}, \tilde{\partial}_{a}) = (\partial_{\nu}, \partial_{b}) \cdot A$$
(3.2a)

$$(\mathrm{d}u^{\mu},\mathrm{d}u^{a})^{\top} \mapsto (\mathrm{d}\tilde{u}^{\mu},\mathrm{d}\tilde{u}^{a})^{\top} = A^{-1} \cdot (\mathrm{d}u^{\nu},\mathrm{d}u^{b})^{\top}.$$
 (3.2b)

Here and below expressions like  $(\partial_{\mu}, \partial_{a})$  are shortcuts for ordered (n + r)-tuples like  $(\partial_{1}, \ldots, \partial_{n+r}) = ([\partial_{\mu}]_{\mu=1}^{n}, [\partial_{a}]_{a=n+1}^{n+r}), \top$  is the matrix transpositions sign, the centered dot  $\cdot$  stands for the matrix multiplication, and the transformation matrix A is

$$A := \left[\frac{\partial u^{I}}{\partial \tilde{u}^{J}}\right]_{I,J=1}^{n+r} = \begin{pmatrix} \left[\frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}}\right] & 0_{n \times r} \\ \left[\frac{\partial u^{b}}{\partial \tilde{u}^{\mu}}\right] & \left[\frac{\partial u^{b}}{\partial \tilde{u}^{a}}\right] \end{pmatrix} =: \begin{bmatrix} \frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}} & 0 \\ \frac{\partial u^{b}}{\partial \tilde{u}^{\mu}} & \frac{\partial u^{b}}{\partial \tilde{u}^{a}} \end{bmatrix} , \qquad (3.3)$$

where  $0_{n \times r}$  is the  $n \times r$  zero matrix. Besides, in expressions of the form  $\partial_I a^I$ , like the one in the right-hand side of (3.2a), the summation excludes differentiation, i.e.,  $\partial_I a^I := a^I \partial_I = \sum_I a^I \partial_I$ ; if a differentiation really takes place, we write  $\partial_I (a^I) := \sum_I \partial_I (a^I)$ . This rule allows a lot of formulae to be written in a compact matrix form, like (3.2a). The explicit form of the matrix inverse to (3.3) is  $A^{-1} = \left[\frac{\partial \tilde{u}^I}{\partial u^J}\right] = \dots$  and it is obtained from (3.3) via the change  $u \leftrightarrow \tilde{u}$ .

 $<sup>^2 \</sup>mathrm{On}$  a bundle or fibred manifold, these coordinates are known also as adapted coordinates [116, Definition 1.1.5].

<sup>&</sup>lt;sup>3</sup>If (W, v) is a bundle chart, with  $v: W \to \mathbb{K}^n \times \mathbb{K}^r$  and  $e^a: \mathbb{K}^r \to \mathbb{K}$  are such that  $e^a(c_1, \ldots, c_r) = c_a \in \mathbb{K}$ , then one can put  $u^a = e^a \circ \operatorname{pr}_2 \circ v$ , where  $\operatorname{pr}_2: \mathbb{K}^n \times \mathbb{K}^r \to \mathbb{K}^r$  is the projection on the second multiplier  $\mathbb{K}^r$ .

#### 3. Connections on bundles

The formula (3.2a) can be generalized for arbitrary frames  $\{e_I\} = \{e_{\mu}, e_a\}$ and  $\{\tilde{e}_I\} = \{\tilde{e}_{\mu}, \tilde{e}_a\}$  in T(E) and their dual coframes  $\{e^I\} = \{e^{\mu}, e^a\}$  and  $\{\tilde{e}^I\} = \{\tilde{e}^{\mu}, \tilde{e}^a\}$  in  $T^*(E)$  whose *admissible changes* are given by

$$(e_I) = (e_{\mu}, e_a) \mapsto (\tilde{e}_I) = (\tilde{e}_{\mu}, \tilde{e}_a) = (e_{\nu}, e_b) \cdot A$$
(3.4a)

$$\begin{pmatrix} e^{\mu} \\ e^{a} \end{pmatrix} \mapsto \begin{pmatrix} \tilde{e}^{\mu} \\ \tilde{e}^{a} \end{pmatrix} = A^{-1} \cdot \begin{pmatrix} e^{\nu} \\ e^{b} \end{pmatrix}.$$
 (3.4b)

Here  $A = [A_J^I]$  is a nondegenerate matrix-valued function with a block structure similar to (3.3), viz.

$$A = \begin{pmatrix} [A_{\mu}^{\nu}]_{\mu,\nu=1}^{n} & 0_{n \times r} \\ [A_{\mu}^{b}]_{\substack{\mu=1,\dots,n\\b=n+1,\dots,n+r}} & [A_{a}^{b}]_{a,b=n+1}^{n+r} \end{pmatrix} =: \begin{bmatrix} A_{\mu}^{\nu} & 0 \\ A_{\mu}^{b} & A_{a}^{b} \end{bmatrix}$$
(3.5a)

with inverse matrix

$$A^{-1} = \begin{pmatrix} [A^{\nu}_{\mu}]^{-1} & 0\\ -[A^{a}_{b}]^{-1} \cdot [A^{a}_{\mu}] \cdot [A^{\nu}_{\mu}]^{-1} & [A^{a}_{b}]^{-1} \end{pmatrix} .$$
(3.5b)

Here  $A^a_{\mu} \colon W \to \mathbb{K}$  and  $[A^{\nu}_{\mu}]$  and  $[A^a_b]$  are non-degenerate matrix-valued functions on W such that  $[A^{\nu}_{\mu}]$  is constant on the fibres of E, i.e., for  $p \in U$ ,  $A^{\nu}_{\mu}(p)$  depends only on  $\pi(p) \in M$ , which is equivalent to any one of the equations

$$A^{\nu}_{\mu} = B^{\nu}_{\mu} \circ \pi \qquad \frac{\partial A^{\nu}_{\mu}}{\partial u^a} = 0, \qquad (3.6)$$

with  $[B^{\nu}_{\mu}]$  being a nondegenerate matrix-valued function on  $\pi(W) \subseteq M$ . Obviously, (3.2) corresponds to (3.4) with  $e_I = \frac{\partial}{\partial u^I}$ ,  $\tilde{e}_I = \frac{\partial}{\partial \tilde{u}^I}$ , and  $A^J_I = \frac{\partial u^J}{\partial \tilde{u}^I}$ .

All frames  $\{\tilde{e}_I\}$  on E connected via (3.4)–(3.5), which are (locally) obtainable from holonomic ones  $\{e_I\}$ , induced by bundle coordinates, via admissible changes, will be referred as *bundle frames*. Only such frames will be employed in the present chapter.

If we deal with a vector bundle  $(E, \pi, M)$  endowed with vector bundle coordinates  $\{u^I\}$  [23], then the new fibre coordinates  $\{\tilde{u}^a\}$  in (3.1) must be *linear and homogeneous* in the old ones  $\{u^a\}$ , i.e.,

$$\tilde{u}^a = (B^a_b \circ \pi) \cdot u^b \text{ and } u^a = ((B^{-1})^a_b \circ \pi) \cdot \tilde{u}^b,$$
(3.7)

with  $B = [B_b^a]$  being a non-degenerate matrix-valued function on  $\pi(W) \subseteq M$ . In that case, the matrix (3.3) and its inverse take the form

$$A = \begin{bmatrix} \frac{\partial u^{\mu}}{\partial \tilde{u}^{\nu}} & 0\\ \left(\frac{\partial (B^{-1})_{a}^{b}}{\partial \tilde{x}^{\nu}} \circ \pi\right) \cdot \tilde{u}^{b} & (B^{-1})_{a}^{b} \circ \pi \end{bmatrix} \qquad A^{-1} = \begin{bmatrix} \frac{\partial \tilde{u}^{\nu}}{\partial u^{\mu}} & 0\\ \left(\frac{\partial B_{a}^{b}}{\partial x^{\mu}} \circ \pi\right) \cdot u^{a} & B_{a}^{b} \circ \pi \end{bmatrix}.$$
(3.8)

More generally, in the vector bundle case, admissible are transformations (3.4) with matrices like

$$A = \begin{pmatrix} [A_{\nu}^{\mu}] & 0\\ [A_{c\mu}^{b}\tilde{u}^{c}] & [A_{b}^{a}] \end{pmatrix} \qquad A^{-1} = \begin{pmatrix} [A_{\nu}^{\mu}]^{-1} & 0\\ -[A_{b}^{a}]^{-1} \cdot [A_{c\mu}^{b}\tilde{u}^{c}] \cdot [A_{\nu}^{\mu}]^{-1} & [A_{b}^{a}]^{-1} \end{pmatrix}$$
(3.9)

with  $A^a_{b\mu} \colon W \to \mathbb{K}$  being functions on W which are constant on the fibres of E,

$$A^a_{b\mu} = B^a_{b\mu} \circ \pi \qquad \frac{\partial A^a_{b\mu}}{\partial u^c} = 0 \tag{3.10}$$

for some functions  $B^a_{b\mu}$ :  $\pi(W) \to \mathbb{K}$ . Obviously, (3.9) corresponds to (3.5) with  $A^b_{\mu} = A^b_{c\mu}\tilde{u}^c$  and the setting  $A^J_I = \frac{\partial u^J}{\partial \tilde{u}^I}$  reduces (3.9) to (3.8) due to (3.7).

## **3.2.** Connection theory

From a number of equivalent definitions of a connection on differentiable manifold [146, Sections 2.1 and 2.2], we shall use the following one.

**Definition 3.1 (cf.** Definition IV.14.3 on page 303). A connection on differentiable bundle  $(E, \pi, M)$  is an  $n = \dim M$ -dimensional distribution  $\Delta^h$  on E such that, for each  $p \in E$  and the vertical distribution  $\Delta^v$  defined by

$$\Delta^{v} \colon p \mapsto \Delta^{v}_{p} := T_{i(p)} \big( \pi^{-1}(\pi(p)) \big) \cong T_{p} \big( \pi^{-1}(\pi(p)) \big), \tag{3.11}$$

with  $\iota: \pi^{-1}(\pi(p)) \to E$  being the inclusion mapping, is fulfilled

$$\Delta_p^v \oplus \Delta_p^h = T_p(E), \tag{3.12}$$

where  $\Delta^h : p \mapsto \Delta_p^h \subseteq T_p(E)$  and  $\oplus$  is the direct sum sign. The distribution  $\Delta^h$  is called *horizontal* and symbolically we write  $\Delta^v \oplus \Delta^h = T(E)$ .

A vector at a point  $p \in E$  (resp. a vector field on E) is said to be vertical or horizontal if it (resp. its value at p) belongs to  $\Delta_p^v$  or  $\Delta_p^h$ , respectively, for the given (resp. any) point p. A vector  $Y_p \in T_p(E)$  (resp. vector field  $Y \in \mathcal{X}(E)$ ) is called a horizontal lift of a vector  $X_{\pi(p)} \in T_{\pi(p)}(M)$  (resp. vector field  $X \in$  $\mathcal{X}(M)$  on  $M = \pi(E)$ ) if  $\pi_*(Y_p) = X_{\pi(p)}$  for the given (resp. any) point  $p \in E$ . Since  $\pi_*|_{\Delta_p^h} \colon \Delta_p^h \to T_{\pi(p)}(M)$  is a vector space isomorphism for all  $p \in E$  [23, Section 1.24], any vector in  $T_{\pi(p)}(M)$  (resp. vector field in  $\mathcal{X}(M)$ ) has a unique horizontal lift in  $T_p(E)$  (resp.  $\mathcal{X}(E)$ ).

As a result of (3.12), any vector  $Y_p \in T_p(E)$  (resp. vector field  $Y \in \mathcal{X}(E)$ ) admits a unique representation  $Y_p = Y_p^v \oplus Y_p^h$  (resp.  $Y = Y^v \oplus Y^h$ ) with  $Y_p^v \in \Delta_p^v$ and  $Y_p^h \in \Delta_p^h$  (resp.  $Y^v \in \Delta^v$  and  $Y^h \in \Delta^h$ ). If the distribution  $p \mapsto \Delta_p^h$  is differentiable of class  $C^m$ ,  $m \in \mathbb{N} \cup \{0, \infty, \omega\}$ , it is said that the *connection*  $\Delta^h$ is (differentiable) of class  $C^m$ . A connection  $\Delta^h$  is of class  $C^m$  if and only if, for every  $C^m$  vector field Y on E, the vertical  $Y^v$  and horizontal  $Y^h$  vector fields are of class  $C^m$ .

A  $C^1$  path  $\beta: J \to E$  is called *horizontal (vertical)* if its tangent vector  $\dot{\beta}$  is horizontal (vertical) vector along  $\beta$ , i.e.,  $\dot{\beta}(s) \in \Delta^h_{\beta(s)}$  ( $\dot{\beta}(s) \in \Delta^v_{\beta(s)}$ ) for all  $s \in J$ . A lifting  $\bar{\gamma}: J \to E$  of a path  $\gamma: J \to M$ , i.e.,  $\pi \circ \bar{\gamma} = \gamma$ , is called *horizontal* if  $\bar{\gamma}$  is a horizontal path, i.e., when the vector field  $\dot{\bar{\gamma}}$  tangent to  $\bar{\gamma}$  is horizontal or, equivalently, if  $\dot{\bar{\gamma}}$  is a horizontal lift of  $\dot{\gamma}$ . Since  $\pi^{-1}(\gamma(J))$  is an (r+1)-dimensional submanifold of E, the distribution  $p \mapsto \Delta^h_p \cap T_p(\pi^{-1}(\gamma(J)))$  is one-dimensional and, consequently, is integrable for arbitrary  $C^1$  path  $\gamma$ . The integral paths of that distribution are horizontal lifts of  $\gamma$  and, for each  $p \in \pi^{-1}(\gamma(J))$ , there is (locally) a unique horizontal lift  $\bar{\gamma}_p$  of  $\gamma$  passing through  $p.^4$ 

**Definition 3.2.** Let  $\gamma \colon [\sigma, \tau] \to M$ , with  $\sigma, \tau \in \mathbb{R}$  and  $\sigma \leq \tau$ , and  $\bar{\gamma}_p$  be the unique horizontal lift of  $\gamma$  in E passing through  $p \in \pi^{-1}(\gamma([\sigma, \tau]))$ . The *parallel transport* (translation, displacement) generated by (assigned to, defined by) a connection  $\Delta^h$  is a mapping  $\mathsf{P} \colon \gamma \mapsto \mathsf{P}^{\gamma}$ , assigning to the path  $\gamma$  a mapping

$$\mathsf{P}^{\gamma} \colon \pi^{-1}(\gamma(\sigma)) \to \pi^{-1}(\gamma(\tau)) \qquad \gamma \colon [\sigma, \tau] \to M$$
(3.13)

such that, for each  $p \in \pi^{-1}(\gamma(\sigma))$ ,

$$\mathsf{P}^{\gamma}(p) := \bar{\gamma}_p(\tau). \tag{3.14}$$

In vector bundles are important the *linear* connections for which is required the parallel transport assigned to them to be linear in a sense that the mapping (3.13) is linear for every path  $\gamma$  (see Subsection 4.2 below).

Let us now look on the connections  $\Delta^h$  on a bundle  $(E, \pi, M)$  from a view point of (local) frames and their dual coframes on E. Let  $\{e_\mu\}$  be a basis for  $\Delta^h$ , i.e.,  $e_\mu \in \Delta^h$  and  $\{e_\mu|_p\}$  is a basis for  $\Delta^h_p$  for all  $p \in E$ , and  $\{e^a\}$  be the coframe for  $\Delta^h$ , i.e., a collection of 1-forms  $e^a$ ,  $a = n + 1, \ldots, n + r$ , which are linearly independent (relative to functions  $E \to \mathbb{K}$ ) and such that  $e^a(X) = 0$  if  $X \in \Delta^h$ .

**Definition 3.3.** A frame  $\{e_I\}$  in T(E) over E is called *specialized* for a connection  $\Delta^h$  if the first  $n = \dim M$  of its vector fields form a basis  $\{e_\mu\}$  for the horizontal distribution  $\Delta^h$  and its last  $r = \dim \pi^{-1}(x), x \in M$ , vector fields form a basis  $\{e_a\}$  for the vertical distribution  $\Delta^v$ . Respectively, a coframe  $\{e^I\}$  on E is called *specialized* if  $\{e^a\}$  is a cobasis for  $\Delta^h$  and  $\{e^\mu\}$  is a cobasis for  $\Delta^v$ .

The horizontal lifts of vector fields and 1-forms can easily be described in specialized (co)frames. Indeed, let  $\{e_I\}$  and  $\{e^I\}$  be respectively a specialized frame and its dual coframe. Define a frame  $\{E_{\mu}\}$  and its dual coframe  $\{E^{\mu}\}$  on M which are  $\pi$ -related to  $\{e_I\}$  and  $\{e^I\}$ , i.e.,  $E_{\mu} := \pi_*(e_{\mu})$  and  $e^{\mu} := \pi^*(E^{\mu}) = E^{\mu} \circ \pi_*.^5$ 

<sup>&</sup>lt;sup>4</sup>In this sense, a connection  $\Delta^h$  is an Ehresmann connection [16, p. 314] and vice versa [116, pp. 85–89].

<sup>&</sup>lt;sup>5</sup>Recall,  $\pi_*|_{\Delta_p^h} \colon \Delta_p^h \to T_{\pi(p)}(M)$  is a vector space isomorphism.

If  $Y = Y^{\mu}E_{\mu} \in \mathcal{X}(M)$  and  $\phi = \phi_{\mu}e^{\mu} \in \Lambda^{1}(M)$ , then their horizontal lifts (from M to E) respectively are

$$\bar{Y} = (Y^{\mu} \circ \pi)e_{\mu} \qquad \bar{\phi} = (\phi_{\mu} \circ \pi)e^{\mu}. \tag{3.15}$$

The specialized (co)frames transform into each other according to the general rules (3.4) in which the transformation matrix and its inverse have the following block structure:

$$A = \begin{pmatrix} [A_{\mu}^{\nu}] & 0_{n \times r} \\ 0_{r \times n} & [A_{a}^{b}] \end{pmatrix} \qquad A^{-1} = \begin{pmatrix} [A_{\mu}^{\nu}]^{-1} & 0_{n \times r} \\ 0_{r \times n} & [A_{a}^{b}]^{-1} \end{pmatrix},$$
(3.16)

where  $A^{\nu}_{\mu}, A^{b}_{a}: E \to \mathbb{K}$  and the functions  $A^{\nu}_{\mu}$  are constant on the fibres of the bundle  $(E, \pi, M)$ , that is, we have

$$A^{\nu}_{\mu} = B^{\nu}_{\mu} \circ \pi \quad \text{or} \quad \frac{\partial A^{\nu}_{\mu}}{\partial u^{a}} = 0 \tag{3.17}$$

for some nondegenerate matrix-valued function  $[B^{\nu}_{\mu}]$  on M. Besides, in a case of vector bundle, the functions  $A^a_b$  are also constant on the fibres of the bundle  $(E, \pi, M)$ , i.e.,

$$A_a^b = B_a^b \circ \pi \quad \text{or} \quad \frac{\partial A_a^b}{\partial u^a} = 0 \tag{3.18}$$

for some nondegenerate matrix-valued function  $B = [B_a^b]$  on M. Changes like (3.4), with A given by (3.16), respect the fibre as well as the connection structure of the bundle.

Let E be a  $C^2$  manifold and  $\Delta^h$  a  $C^1$  connection on  $(E, \pi, M)$ . The components  $C_{IJ}^K$  of the anholonomicity object of a *specialized* frame  $\{e_I\}$  are (local) functions on E defined by (see (2.5))

$$[e_I, e_J]\_ =: C_{IJ}^K e_K \tag{3.19}$$

and are naturally divided into the following six groups (cf. [138, p. 21]):

$$\{C^{\lambda}_{\mu\nu}\}, \{C^{a}_{\mu\nu}\}, \{C^{\lambda}_{\mu b}=0\}, \{C^{\lambda}_{ab}=0\}, \{C^{c}_{\mu b}\}, \{C^{c}_{ab}\}.$$
 (3.20)

The functions  $C^{\lambda}_{\mu\nu}$  are constant on the fibres of  $(E, \pi, M)$ , precisely  $C^{\lambda}_{\mu\nu} = f^{\lambda}_{\mu\nu} \circ \pi$ where  $f^{\lambda}_{\mu\nu}$  are the components of the anholonomicity object for the  $\pi$ -related frame  $\{\pi_*(e_\mu)\}$  on M, as the commutators of  $\pi$ -related vector fields are  $\pi$ -related [7, Section 1.55]. Besides, since the vertical distribution  $\Delta^v$  is integrable (the space  $\Delta^v_p$  is the space tangent to the fibre through  $p \in E$  at p), we have

$$[e_a, e_b]_{-} = C^c_{ab} e_c \tag{3.21}$$

(so that  $C_{ab}^{\lambda} = 0$ ), due to which  $C_{ab}^{c}$  are called components of the vertical anholonomicity object. To prove that  $C_{\mu b}^{\lambda} = 0$ , one should expand  $\{e_{I}\}$  along  $\{\partial_{I} = \frac{\partial}{\partial u^{I}}\}$ ,

#### 3. Connections on bundles

with  $\{u^I\}$  being some bundle coordinates, viz.  $e_{\mu} = e^{\nu}_{\mu}\partial_{\nu} + e^b_{\mu}\partial_b$  and  $e_a = e^b_a\partial_b$ , with some functions  $e^{\nu}_{\mu}$ ,  $e^b_{\mu}$  and  $e^b_a$ , and to notice that  $e^{\nu}_{\mu}$  are constant on the fibres, i.e.,  $\partial_a(e^{\nu}_{\mu}) = 0$ .

The non-trivial mixed "vertical-horizontal" components between (3.20), viz.  $C^a_{\mu\nu}$  and  $C^a_{\mu b}$ , are important characteristics of the connection  $\Delta^h$ . The functions

$$^{\circ}\Gamma^{a}_{b\mu} := +C^{a}_{b\mu} = -C^{a}_{\mu b} \tag{3.22a}$$

$$R^a_{\mu\nu} := +C^a_{\mu\nu} = -C^a_{\nu\mu}, \qquad (3.22b)$$

which enter into the commutators

$$\mathcal{L}_{e_{\mu}}e_{b} = [e_{\mu}, e_{b}]_{-} = {}^{\circ}\Gamma^{a}_{b\mu}e_{a} \tag{3.23a}$$

$$\mathcal{L}_{e_{\mu}}e_{\nu} = [e_{\mu}, e_{\nu}]_{-} = R^{a}_{\mu\nu}e_{a} + C^{\lambda}_{\mu\nu}e_{\lambda}, \qquad (3.23b)$$

are called respectively the fibre coefficients of  $\Delta^h$  (or components of the connection object of  $\Delta^h$ ) and fibre components of the curvature of  $\Delta^h$  (or components of the curvature (object) of  $\Delta^h$ ) in  $\{e_I\}$ . Under a change (3.4), with a matrix (3.16), of the specialized frame, the functions (3.22) transform into respectively

$${}^{\circ}\tilde{\Gamma}^{a}_{b\mu} = A^{\nu}_{\mu} \left( [A^{f}_{e}]^{-1} \right)^{a}_{d} \left( {}^{\circ}\Gamma^{d}_{c\nu}A^{c}_{b} + e_{\nu}(A^{d}_{b}) \right)$$
(3.24a)

$$\tilde{R}^a_{\mu\nu} = \left( [A^f_e]^{-1} \right)^a_b A^\lambda_\mu A^\varrho_\nu R^b_{\lambda\varrho}, \qquad (3.24b)$$

which formulae are direct consequences of (3.23). If we put  $\bar{A} := [A_a^b]$ ,  ${}^{\circ}\Gamma_{\nu} := [{}^{\circ}\Gamma_{c\nu}^d]$ , and  ${}^{\circ}\tilde{\Gamma}_{\nu} := [{}^{\circ}\tilde{\Gamma}_{c\nu}^d]$ , then (3.24a) is tantamount to

$${}^{2}\tilde{\Gamma}_{\mu} = A^{\nu}_{\mu}\bar{A}^{-1} \cdot (\,{}^{\circ}\Gamma_{\nu} \cdot \bar{A} + e_{\nu}(\bar{A})) = A^{\nu}_{\mu}(\bar{A}^{-1} \cdot {}^{\circ}\Gamma_{\nu} - e_{\nu}(\bar{A}^{-1})) \cdot \bar{A}.$$
(3.25)

Up to a meaning of the matrices  $[A^{\nu}_{\mu}]$  and  $\bar{A}$  and the size of the matrices  ${}^{\circ}\Gamma_{\nu}$  and  $\bar{A}$ , the last equation is identical with the one expressing the transformed matrices of the coefficients of a linear connection (covariant derivative operator) in a vector bundle (see (IV.14.25) for N = M and  $g = id_M$  or [87, eq. (3.5)]) on which we shall return later in this chapter (see Section 4, in particular equation (4.23') in it). Equation (3.24b) indicates that  $R^a_{\mu\nu}$  are components of a tensor, viz.

$$\Omega := \frac{1}{2} R^a_{\mu\nu} e_a \otimes e^\mu \wedge e^\nu, \qquad (3.26)$$

called *curvature tensor* of the connection  $\Delta^h$ . By (3.23a), the horizontal distribution  $\Delta^h$  is (locally) integrable iff its curvature tensor vanishes,  $\Omega = 0$ .

**Definition 3.4.** A connection with vanishing curvature tensor is called *flat*, or *integrable*, or *curvature free*.

**Proposition 3.1.** The flat connections are the only ones that may admit holonomic specialized frames.

*Proof.* See Definition 3.4 and (3.23b).

The above considerations of specialized (co)frames for a connection  $\Delta^h$  on a bundle  $(E, \pi, M)$  were global as we supposed that these (co)frames are defined on the whole bundle space E, which is always possible if no smoothness conditions on  $\Delta^h$  are imposed. Below we shall show how *local* specialized (co)frames can be defined via local bundle coordinates on E.

Let  $\{u^I\}$  be local bundle coordinates on an open set  $W \subseteq E$ . They define on  $T(W) \subseteq T(E)$  the local basis  $\{\partial_I := \frac{\partial}{\partial u^I}\}$ , so that any vector can be expended along its vectors. In particular, we can do so with any basic vector field  $e_I^W$  of a *specialized* frame  $\{e_I\}$  restricted to W,  $e_I^W := e_I|_W$ . Since  $\{\partial_a|_p\}$ , with  $p \in W$ , is a basis for  $\Delta_p^v$ , we can write

$$(e^W_\mu, e^W_a) = (A^\nu_\mu \partial_\mu + A^a_\mu \partial_a, A^b_a \partial_b) = (\partial_\nu, \partial_b) \cdot \begin{pmatrix} [A^\nu_\mu] & 0\\ [A^b_\mu] & [A^b_a] \end{pmatrix} , \qquad (3.27)$$

where  $[A^{\nu}_{\mu}]$  and  $[A^{b}_{a}]$  are non-degenerate matrix-valued functions on  $W^{.6}$ 

**Definition 3.5.** A frame  $\{X_I\}$  over W in T(W) is called *adapted (to the coordinates*  $\{u^I\}$  in W) for a connection  $\Delta^h$  if it is the specialized frame obtained from (3.27) via admissible transformation (3.4) with the matrix  $A = \begin{pmatrix} [A^{\nu}_{\mu}]^{-1} & 0 \\ 0 & [A^{b}_{a}]^{-1} \end{pmatrix}$ .

**Exercise 3.1.** An arbitrary specialized frame  $\{e_I^W\}$  in T(E) over W enters in the definition of a frame  $\{X_I\}$  adapted to bundle coordinates  $\{u^I\}$  on W. Prove that  $\{X_I\}$  is independent of the particular choice of the frame  $\{e_I^W\}$ . (Hint: apply Definition 3.5 and (3.4a) with A given by (3.16).) The below-derived equality (3.34) is an indirect proof of that fact too.

According to (3.4) and Definition 3.5, the adapted frame  $\{X_I\}$  and the corresponding to it *adapted coframe*  $\{\omega^I\}$  are given by

$$X_{\mu} = \partial_{\mu} + \Gamma^{a}_{\mu} \partial_{a} \qquad X_{a} = \partial_{a} \tag{3.28a}$$

$$\omega^{\mu} = \mathrm{d}u^{\mu} \qquad \qquad \omega^{a} = \mathrm{d}u^{a} - \Gamma^{a}_{\mu}\mathrm{d}u^{\mu}. \qquad (3.28\mathrm{b})$$

Here the functions  $\Gamma^a_{\mu} \colon W \to \mathbb{K}$  are defined via

$$[\Gamma^a_{\mu}] = + [A^a_{\nu}] \cdot [A^{\nu}_{\mu}]^{-1}$$
(3.29)

and are called (2-index) coefficients of  $\Delta^h$ . In a matrix form, the equations (3.28) can be written as

$$(X_{\mu}, X_{a}) = (\partial_{\nu}, \partial_{b}) \cdot \begin{bmatrix} \delta^{\nu}_{\mu} & 0\\ +\Gamma^{b}_{\mu} & \delta^{b}_{a} \end{bmatrix} \quad \begin{pmatrix} \omega^{\mu}\\ \omega^{a} \end{pmatrix} = \begin{bmatrix} \delta^{\mu}_{\nu} & 0\\ -\Gamma^{a}_{\nu} & \delta^{a}_{b} \end{bmatrix} \cdot \begin{pmatrix} \mathrm{d}u^{\nu}\\ \mathrm{d}u^{b} \end{pmatrix} .$$
(3.30)

<sup>6</sup>The non-degeneracy of  $[A^{\nu}_{\mu}]$  follows from the fact that the vector fields  $\pi_*|_{\Delta^h}(e^W_{\mu}) = A^{\nu}_{\mu}\pi_*(\frac{\partial}{\partial u^{\mu}})$  form a basis for  $\mathcal{X}(\pi(W)) \subseteq \mathcal{X}(M)$ .

 $\Box$ 

#### 3. Connections on bundles

The operators  $X_{\mu} = \partial_{\mu} + \Gamma^{a}_{\mu}\partial_{a}$  are known as *covariant derivatives on* T(W) and the plus sing in (3.28a) before  $\Gamma^{a}_{\mu}$  (hence in the right-hand side of (3.29)) is conventional.

If  $\{u^I\}$  and  $\{\tilde{u}^I\}$  are local coordinates on open sets  $W \subseteq E$  and  $\tilde{W} \subseteq E$ , respectively, and  $W \cap \tilde{W} \neq \emptyset$ , then, on the overlapping set  $W \cap \tilde{W}$ , a problem arises: how are connected the adapted frames corresponding to these coordinates? Let us mark with a tilde all quantities that refer to the coordinates  $\{\tilde{u}^I\}$ . Since the adapted frames are, by definitions, specialized ones, we can write (see (3.4))

$$(\tilde{X}_{\mu}, \tilde{X}_{a}) = (X_{\nu}, X_{b}) \cdot A \qquad \begin{pmatrix} \tilde{\omega}^{\mu} \\ \tilde{\omega}^{a} \end{pmatrix} = A^{-1} \cdot \begin{pmatrix} \omega^{\nu} \\ \omega^{b} \end{pmatrix},$$
 (3.31a)

where the transformation matrix A and its inverse have the form (3.16). Recalling (3.2) and (3.3), from these equalities, we get

$$A = \operatorname{diag}\left(\left[\frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}}\right], \left[\frac{\partial u^{b}}{\partial \tilde{u}^{a}}\right]\right) = \begin{pmatrix} \left[\frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}}\right] & 0\\ 0 & \left[\frac{\partial u^{b}}{\partial \tilde{u}^{a}}\right] \end{pmatrix} .$$
(3.31b)

Combining (3.29) and (3.31), one can easily prove

**Proposition 3.2.** A change  $\{u^I\} \mapsto \{\tilde{u}^I\}$  of the local bundle coordinates implies the following transformation of the 2-index coefficients of the connection:

$$\Gamma^a_{\mu} \mapsto \tilde{\Gamma}^a_{\mu} = \left(\frac{\partial \tilde{u}^a}{\partial u^b} \Gamma^b_{\nu} + \frac{\partial \tilde{u}^a}{\partial u^{\nu}}\right) \frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}}.$$
(3.32)

It is obvious, a connection  $\Delta^h$  is of class  $C^m$ ,  $m \in \mathbb{N} \cup \{0\}$ , if and only if its coefficients  $\Gamma^a_\mu$  are  $C^m$  functions on W, provided  $\partial_I$  are  $C^m$  vector fields on W(which is the case when E is a  $C^{m+1}$  manifold). By virtue of (3.32), the  $C^{m+1}$ changes of the local bundle coordinates preserve the  $C^m$  differentiability of  $\Gamma^a_\mu$ . Thus the  $C^{m+1}$  differentiability of the base M and bundle E spaces is a necessary condition for existence of  $C^m$  connections on  $(E, \pi, M)$ ; as we assumed m = 1 in this chapter, the connections considered here can be at most of differentiability class  $C^1$ .

The next proposition states that a connection on a bundle is locally equivalent to a geometric object whose components transform like (3.32).

**Proposition 3.3.** To any connection  $\Delta^h$  in a bundle  $(E, \pi, M)$  can be assigned, according to the procedure described above, a geometrical object on E whose components  $\Gamma^a_\mu$  in bundle coordinates  $\{u^I\}$  on E transform according to (3.32) under a change  $\{u^I\} \mapsto \{\tilde{u}^I\}$  of the bundle coordinates on the intersection of the domains of  $\{u^I\}$  and  $\{\tilde{u}^I\}$ . Conversely, given a geometrical object on E with local transformation law (3.32), there is a unique connection  $\Delta^h$  in  $(E, \pi, M)$  which generates the components of that object as described above. Proof. The first part of the statement was proved above, when we constructed the adapted frame (3.28a) and derived (3.32). To prove the second part, choose a point  $p \in E$  and some local coordinates  $\{u^I\}$  on an open set W in E containing p in which the geometrical object mentioned has local components  $\Gamma^a_{\mu}$ . Define a local frame  $\{X_I\} = \{X_{\mu}, X_a\}$  on W by (3.28a). The required connection is then  $\Delta^h: q \mapsto \Delta^h_q := \{r^{\mu}X_{\mu}|_q: r^{\mu} \in \mathbb{K}\}$  for any  $q \in W$ , which means that  $\Delta^h_q$  is the linear cover of  $\{X_{\mu}|_q\}$ . The transformation law (3.32) insures the independence of  $\Delta^h$  from the local coordinates employed in its definition.  $\Box$ 

From the construction of an adapted frame  $\{X_I\}$ , as well as from the proof of Proposition 3.3, follows that the set of vectors  $\{X_\mu\}$  is a basis for the horizontal distribution  $\Delta^h$  and the set  $\{X_a\}$  is a basis for the vertical distribution  $\Delta^v$ . The matrix of the restricted tangent projection  $\pi_*|_{\Delta^h}$  in bundle coordinate system  $\{u^{\mu} = x^{\mu} \circ \pi, u^a\}$  on E, where  $x^{\mu}$  are local coordinates on M, is the identity matrix as  $(\pi_*|_{\Delta^h_p})^{\nu}_{\mu} = \frac{\partial(x^{\mu} \circ \pi)}{\partial u^{\mu}}\Big|_p = \delta^{\nu}_{\mu}$  for any point p in the domain of  $\{u^I\}$ . Hereof we get

$$\pi_*|_{\Delta^h}(X_\mu) = \frac{\partial}{\partial x^\mu} \qquad \Big( \iff \pi_*|_{\Delta^h_p}(X_\mu|_p) = \frac{\partial}{\partial x^\mu}\Big|_{\pi(p)} \Big). \tag{3.33}$$

In particular, from here follows that  $\pi_*|_{\Delta_p^h} \colon \Delta_p^h \to T_{\pi(p)}(M)$  is a vector space isomorphism. The inverse to equation (3.33), viz.

$$X_{\mu} = (\pi_*|_{\Delta^h})^{-1} \left(\frac{\partial}{\partial x^{\mu}}\right) = (\pi_*|_{\Delta^h})^{-1} \circ \pi_* \left(\frac{\partial}{\partial u^{\mu}}\right), \tag{3.34}$$

can be used in an equivalent definition of a frame  $\{X_I\}$  adapted to local coordinates  $\{u^I\}$ , namely, this is the frame  $\{(\pi_*|_{\Delta^h})^{-1} \circ \pi_*(\frac{\partial}{\partial u^u}), \frac{\partial}{\partial u^a}\}$ . If one accepts such a definition of an adapted frame for  $\Delta^h$ , the (2-index) coefficients of  $\Delta^h$  have to be defined via the expansion (3.28a); the only changes this may entail are in the proofs of some results, like (3.31) and (3.32).

It is useful to be recorded also the simple fact that, by construction, we have

$$\pi_*(X_a) = 0. \tag{3.35}$$

Let E be a  $C^2$  manifold and  $\Delta^h$  be a  $C^1$  connection. The *adapted frames are* generally anholonomic as the commutators between the basic vector fields of the adapted frame (3.28a) are (cf. (3.20) and (3.22))

$$[X_{\mu}, X_{\nu}] = R^{a}_{\mu\nu} X_{a} \quad [X_{\mu}, X_{b}] = {}^{\circ} \Gamma^{a}_{b\mu} X_{a} \quad [X_{a}, X_{b}] = 0,$$
(3.36)

with

$$R^{a}_{\mu\nu} = \partial_{\mu}(\Gamma^{a}_{\nu}) - \partial_{\nu}(\Gamma^{a}_{\mu}) + \Gamma^{b}_{\mu}\partial_{b}(\Gamma^{a}_{\nu}) - \Gamma^{b}_{\nu}\partial_{b}(\Gamma^{a}_{\mu}) = X_{\mu}(\Gamma^{a}_{\nu}) - X_{\nu}(\Gamma^{a}_{\mu}) \quad (3.37a)$$

$${}^{\mathcal{D}}\Gamma^{a}_{b\mu} = -\partial_{b}(\Gamma^{a}_{\mu}) = -X_{b}(\Gamma^{a}_{\mu})$$
(3.37b)

being the fibre components of the curvature and fibre coefficients of the connection.

An obvious corollary from (3.36) is

Proposition 3.4. An adapted frame is holonomic if and only if

$$R^a_{\mu\nu} = 0 \quad (\iff \Omega = 0) \quad ^\circ\Gamma^a_{b\mu} = 0. \tag{3.38}$$

Therefore only the flat (integrable)  $C^1$  connections, for which  $\Omega = 0$ , may admit holonomic adapted frames (cf. Proposition 3.1). Besides, as a consequence of (3.37b) and (3.38), such connections admit holonomic adapted frames on  $W \subseteq E$ if and only if there are local coordinates on W in which the coefficients  $\Gamma^a_{\mu}$  are constant on the fibres passing through W, i.e., iff  $\Gamma^a_{\mu} = G^a_{\mu} \circ \pi$  for some functions  $G^a_{\mu} : \pi(W) \to \mathbb{K}$ , which is equivalent to  $\partial_b(\Gamma^a_{\mu}) = 0$ .

**Example 3.1 (horizontal lifting of a path).** Recall, the horizontal lift of a  $C^1$  path  $\gamma: J \to M$  passing through a point  $p \in \pi^{-1}(\gamma(t_0))$  for some  $t_0 \in J$  is the unique path  $\bar{\gamma}_p: J \to E$  such that  $\pi \circ \bar{\gamma}_p = \gamma$ ,  $\bar{\gamma}_p(t_0) = p$ , and  $\dot{\bar{\gamma}}_p(t) \in \Delta^h_{\bar{\gamma}_p(t)}$  for all  $t \in J$ . As in a specialized frame  $\{e_I\}$  the relation  $X_p \in \Delta^h_p$  is equivalent to  $e^a(X) = 0$  for any  $X \in \mathcal{X}(M)$ , in an adapted coframe, given by (3.28b), the horizontal lift  $\bar{\gamma}_p$  of  $\gamma$  is the unique solution of the initial value problem

$$\omega^a(\dot{\bar{\gamma}}_p) = 0 \tag{3.39a}$$

$$\bar{\gamma}_p(t_0) = p \tag{3.39b}$$

which is tantamount to any one of the initial-value problems  $(t \in J)$ 

$$\dot{\gamma}^{a}_{p}(t) - \Gamma^{a}_{\mu}(\bar{\gamma}_{p}(t))\dot{\gamma}^{\mu}_{p}(t) = 0$$
 (3.39'a)

$$\bar{\gamma}_p^I(t_0) = p^I := u^I(p)$$
 (3.39'b)

$$\frac{\mathrm{d}(u^a \circ \bar{\gamma}_p(t))}{\mathrm{d}t} - \Gamma^a_\mu(\bar{\gamma}_p(t)) \frac{\mathrm{d}(x^\mu \circ \gamma(t))}{\mathrm{d}t} = 0$$
(3.39"a)

$$u^{I}(\bar{\gamma}_{p}(t_{0})) = u^{I}(p),$$
 (3.39"b)

where  $x^{\mu}$  are the local coordinates in the base space that induce the basic coordinates  $u^{\mu}$  on the bundle space,  $u^{\mu} = x^{\mu} \circ \pi$ . (Note that the quantities  $\frac{d(x^{\mu} \circ \gamma(t))}{dt}$ , entering into (3.39"a), are the components of the vector  $\dot{\gamma}$  tangent to  $\gamma$  at parameter value t.) One may call (3.39a), or any one of its versions (3.39'a) or (3.39"a), the *parallel transport equation* in an adapted frame as it uniquely determines the parallel transport along the restriction of  $\gamma$  to any closed subinterval in J (see Definition 3.2).

**Example 3.2 (the equation of geodesic paths).** A connection  $\Delta^h$  on the tangent bundle  $(T(M), \pi_T, M)$  of a manifold M is called a *connection on* M. In this case, equation (3.39) defines also the geodesics (relative to  $\Delta^h$ ) in M. A  $C^2$  path  $\gamma: J \to M$  in a  $C^2$  manifold M is called a *geodesic path* if its tangent vector field  $\dot{\gamma}$  undergoes parallel transport along the same path  $\gamma$ , i.e.,  $\mathsf{P}^{\gamma|[\sigma,\tau]}(\dot{\gamma}(\sigma)) = \dot{\gamma}(\tau)$  for all  $\sigma, \tau \in J$ , which means that the lifted path  $\dot{\gamma} \colon J \to T(M)$  is a horizontal lift of  $\gamma$  (relative to  $\Delta^h$ ). So, if  $x^{\mu}$  are local coordinates on  $\pi(W) \in M$  and the bundle coordinates on  $W \subseteq E$  are such that [7, Section 1.25]  $u^{\mu} = x^{\mu} \circ \pi$  and  $u^{n+\mu} = dx^{\mu}$  $(\mu, \nu, \dots = 1, \dots, n = \dim M)$ , then (3.39"a) transforms into the geodesic equation (on M)

$$\frac{\mathrm{d}^2(x^\mu \circ \gamma(t))}{\mathrm{d}t^2} - \Gamma_\nu^{n+\mu}(\dot{\gamma}(t))\frac{\mathrm{d}(x^\nu \circ \gamma(t))}{\mathrm{d}t} = 0 \qquad t \in J,$$
(3.40)

which (locally) defines all geodesics in M. Of course, a particular geodesic is specified by fixing some initial values for  $\gamma(t_0)$  and  $\dot{\gamma}(t_0)$  for some  $t_o \in J$ . Notice, equation (3.40) is an equation for a path  $\gamma$  in M, while (3.39"a) is an equation for the lifted path  $\bar{\gamma}$  in T(M) provided the path  $\gamma$  in M is known; for a geodesic path, evidently, we have  $\bar{\gamma} = \dot{\gamma}$ . With obvious renumbering of the indices, one usually writes  $\Gamma^{\mu}_{\nu}$  for  $\Gamma^{n+\mu}_{\nu}$ , so then (3.40) coincides with the classical geodesic equation (I.3.23); for other point of view on this equation, see Section IV.15.

## 4. Connections on vector bundles

In this section, by  $(E, \pi, M)$  we shall denote an arbitrary vector bundle (see Section IV.2 or, e.g., [23]). A peculiarity of such bundles is that their fibres are isomorphic vector spaces, which leads to a natural description of the vertical distribution  $\Delta^{v}$  on their fibre spaces, as well as to existence of a kind of differentiation of their sections (known as covariant differentiation – see Subsection IV.14.1).

In the vector bundles are used, as we shall do in this section, the so-called vector bundle coordinates whose fibre coordinates are linear on their fibres and are constructed as follows (cf. [116, p. 30].

Let  $\{e_a\}$  be a frame in E over a subset  $W_M \subseteq M$ , i.e.,  $\{e_a(x)\}$  to be a basis in  $\pi^{-1}(x)$  for all  $x \in W_M$ . Then, for each  $p \in \pi^{-1}(W_M)$ , we have a unique expansion  $p = p^a e_a(\pi(p))$  for some numbers  $p^a \in \mathbb{K}$ . The vector fibre coordinates  $\{u^a\}$  on  $\pi^{-1}(W_M)$  induced (generated) by the frame  $\{e_a\}$  are defined via  $u^a(p) := p^a$  and hence can be identified with the elements of the coframe  $\{e^a\}$  dual to  $\{e_a\}$ , i.e.,  $u^a = e^a$ . Conversely, if  $\{u^I\}$  are coordinates on  $\pi^{-1}(W_M)$  for some  $W_M \subseteq M$  which are linear on the fibres over  $W_M$ , then there is a unique frame  $\{e_a\}$  in  $\pi^{-1}(W_M)$  which generates  $\{u^a\}$  as just described; indeed, considering  $u^{n+1}, \ldots, u^{n+r}$  as 1-forms on  $\pi^{-1}(W_M)$ , one should define the frame  $\{e_a\}$  required as a one whose dual is  $\{u^a\}$ , i.e., via the conditions  $u^a(e_b) = \delta_b^a$ .

A collection  $\{u^I\}$  of basic coordinates  $u^{\mu}$  and vector fibre coordinates  $u^a$ on  $\pi^{-1}(W_M)$  is called *vector bundle coordinate system* on  $\pi^{-1}(W_M)$ . Only such coordinates on E will be employed in this section.

## 4.1. Vertical lifts

The idea of describing the vertical distribution  $\Delta^v$  on a vector bundle is that, if L is a vector space, then to any  $Y \in L$  there corresponds a 'vertical' vector field
## 4. Connections on vector bundles

 $Y^{v} \in \mathcal{X}(L) = \operatorname{Sec}(T(L), \pi_{T}, L)$  whose value at  $X \in L$  is the vector tangent to the path  $t \mapsto X + tY \in L$ , with  $t \in \mathbb{R}$ , at t = 0, i.e.,  $Y^{v}|_{X} := \frac{\mathrm{d}}{\mathrm{d}t}|_{t=0}(X + tY)$ . Here, as usual, with  $\operatorname{Sec}(E, \pi, M)$  (resp.  $\operatorname{Sec}^{m}(E, \pi, M)$  with  $m \in \mathbb{N} \cup \{0\}$ ) we denote the module of sections (resp.  $C^{m}$  sections) of a bundle  $(E, \pi, M)$  (resp. of a  $C^{m+1}$  bundle  $(E, \pi, M)$ ).

Let  $(E, \pi, M)$  be a vector bundle and  $\Delta^v$  the vertical distribution on it, viz., for each  $p \in E$ ,  $\Delta^v \colon p \mapsto \Delta_p^v \coloneqq T_p(\pi^{-1}(\pi(p)))$ . To every  $Y \in \text{Sec}(E, \pi, M)$ , we assign a *vertical* vector field  $Y^v \in \Delta^v$  on E such that, for  $p \in E$ ,

$$Y_p^v := Y^v|_p := \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=0} (p + tY|_{\pi(p)}).$$
(4.1)

(The mapping  $(p, Y_{\pi(p)}) \mapsto Y_p^v$  defines an isomorphism from the pullback bundle  $\pi^* E$  into the vertical bundle  $\mathcal{V}(E)$  – see [23, Sections 1.27 and 1.28] and also [116, p. 41, Exercises 2.2.1 and 2.2.2].)

**Lemma 4.1.** Let  $\{u^a\}$  be vector fibre coordinates generated by a frame  $\{e_a\}$  on M. If  $Y \in \text{Sec}(E, \pi, M)$  and  $Y = Y^a e_a$ , then

$$Y^{v} = (Y^{a} \circ \pi) \frac{\partial}{\partial u^{a}}.$$
(4.2)

*Proof.* Using the Definition (4.1), we get for  $p \in E$ :

$$Y^{v}|_{p} = \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=0}(p+tY|_{\pi(p)}) = \frac{\mathrm{d}(u^{a}(p+tY|_{\pi(p)}))}{\mathrm{d}t}\Big|_{t=0}\frac{\partial}{\partial u^{a}}\Big|_{p}$$
$$= \frac{\mathrm{d}(p^{a}+tY^{a}(\pi(p)))}{\mathrm{d}t}\Big|_{t=0}\frac{\partial}{\partial u^{a}}\Big|_{p} = Y^{a}(\pi(p))\frac{\partial}{\partial u^{a}}\Big|_{p} = \left((Y^{a}\circ\pi)\cdot\frac{\partial}{\partial u^{a}}\right)\Big|_{p},$$

where equation (I.2.7) was applied.

If  $Y \in \text{Sec}(E, \pi, M)$ , the vector field  $Y^v := v(Y) \in \Delta^v$ , defined via (4.1), is called the *vertical lift of the section* Y. It is (locally) given by (4.2) in vector bundle coordinates.

**Proposition 4.1.** The mapping

$$v: \operatorname{Sec}(E, \pi, M) \to \{ \operatorname{vector fields in} \Delta^v \}$$
$$v: Y \mapsto Y^v: p \mapsto Y^v|_p := \frac{\mathrm{d}}{\mathrm{d}t} \Big|_{t=0} (p + tY_{\pi(p)})$$
(4.3)

is a bijection and it and its inverse are linear mappings.

*Proof.* The linearity and injectivity of v follow directly from (4.1). Now we shall prove that, for each  $Z \in \Delta^v$ , there is a  $Y \in \text{Sec}(E, \pi, M)$  such that  $Y^v = Z$ , i.e., v is also surjective. Let  $Z = Z^a \frac{\partial}{\partial u^a}$ , with  $\{u^I\}$  being (local) vector bundle coordinates on E and the functions  $Z^a$  being constant on the fibres of E, that is  $Z^I = z^I \circ \pi$ 

for some functions  $z^I$  on M. Define  $Y = z^a e_a$  with  $\{e_a\}$  being the frame in E over M generating  $\{u^a\}$ . By Lemma 4.1, we have  $Y^v = (z^a \circ \pi)\frac{\partial}{\partial u^a} = Z^a\frac{\partial}{\partial u^a} = Z$ . The linearity of  $v^{-1}$  follows from here too.

Consider a section  $\omega$  of the bundle dual to  $(E, \pi, M)$  [23]. Its vertical lift  $\omega_v$ is a 1-form on  $\Delta^v$  such that, for  $Z \in \Delta^v$  and  $p \in E$ ,  $\omega_v(Z)|_p = \omega(Y)|_{\pi(p)}$  for the unique section  $Y \in \text{Sec}(E, \pi, M)$  with  $Y^v = Z$  (see Proposition 4.1), i.e., we have  $\omega_v(\cdot)|_p = (\omega \circ v^{-1}(\cdot))|_{\pi(p)}$  which means that

$$\omega_{v}(Z) = (\omega \circ v^{-1}(Z)) \circ \pi \quad \text{or} \quad \omega_{v}(Y^{v})|_{p} = \omega(Y)|_{\pi(p)} \quad (=\omega_{\pi(p)}(Y_{\pi(p)})).$$
(4.4)

If  $\{e^a = u^a\}$  is the coframe dual to  $\{e_a\}$ , and  $\omega = \omega_a e^a$ , then in the coframe  $\{du^a\}$  dual to  $\{\frac{\partial}{\partial u^a}\}$ , we can write (cf. (4.2))

$$\omega_v = (\omega_a \circ \pi) \mathrm{d} u^a. \tag{4.5}$$

It should be mentioned, 'vertical' lifts of vector fields or 1-forms over the base space M are generally not defined unless E = T(M) or  $E = T^*(M)$ , respectively.<sup>1</sup>

A section Y of  $(E, \pi, M)$  and section  $\omega$  of the bundle dual to  $(E, \pi, M)$  can be lifted vertically via the mappings

$$v \colon Y \mapsto Y^v \in \Delta^v \tag{4.6a}$$

$$\omega \mapsto \omega_v \tag{4.6b}$$

respectively given by (4.3) and (4.4) (see also (4.2) and (4.5)). These mappings do not require a connection and arise only from the fibre structure of the bundle space induced from the projection  $\pi: E \to M$ .

If a connection  $\Delta^h$  on  $(E, \pi, M)$  is given, it generates horizontal lifts of the vector fields on the base space M and of the one-forms on the same base space M into respectively vector fields in  $\Delta^h$  and linear mappings on the vector fields in  $\Delta^h$ . Precisely, if  $F \in \mathcal{X}(M)$  and  $\phi \in \Lambda^1(M)$ , their horizontal lifts are defined by the mappings<sup>2</sup>

$$F \mapsto F^h \in \Delta^h$$
 with  $F^h \colon p \mapsto F^h_p := (\pi_*|_{\Delta^h_p})^{-1}(F_{\pi(p)}) \qquad p \in E$  (4.7a)

$$\phi \mapsto \phi_h \quad \text{with} \quad \phi_h := \phi \circ \pi_*|_{\Delta^h} : p \mapsto \phi_h|_p = \phi|_{\pi(p)} \circ (\pi_*|_{\Delta^h_p}). \tag{4.7b}$$

The horizontal lift  $\phi_h$  of  $\phi$  can also be defined alternatively via

$$\phi_h(F^h)|_p = \phi(F)|_{\pi(p)}$$
(4.8)

which equation is tantamount to (4.7b).

<sup>&</sup>lt;sup>1</sup>Since  $\pi_*(\Delta_p^v) = 0_{\pi(p)} \in T_{\pi(p)}(M)$ ,  $p \in E$ , we can say that only the zero vector field over M has vertical lifts relative to  $\pi$  and any vector field in  $\Delta^v$  is its vertical lift. This conclusion is independent of the existence of a connection on  $(E, \pi, M)$  and depends only on the fibre structure of E induced by  $\pi$ .

<sup>&</sup>lt;sup>2</sup>Alternatively, one may define  $\phi'_h = \phi \circ \pi_* = \pi^*(\phi)$ , which expands the domain of  $\phi_h$ , defined by (4.7b), on the whole space  $\mathcal{X}(E)$ . Obviously,  $\phi'_h(Z) = \phi_h(Z)$  for  $Z \in \Delta^h \subseteq \mathcal{X}(E)$  and  $\phi'_h(Z) = 0$  for  $Z \in \mathcal{X}(E) \setminus \{X \in \Delta^h\}$ .

## 4. Connections on vector bundles

Let  $\{u^{\mu} = x^{\mu} \circ \pi, u^a\}$  be vector bundle coordinate system and  $\{X_I\}$  (resp.  $\{\omega^I\}$ ) be the adapted to it frame (resp. coframe) constructed from it according to (3.28). If  $Y = Y^a e_a$ ,  $\omega = \omega_a e^a$ ,  $F = F^{\mu} \frac{\partial}{\partial x^{\mu}} \in \mathcal{X}(M)$ , and  $\phi = \phi_{\mu} dx^{\mu} \in \Lambda^1(M)$ , the equations (4.2) and (4.5) imply

$$Y^{v} = (Y^{a} \circ \pi)X_{a} \quad \omega_{v} = (\omega_{a} \circ \pi)\omega^{a}, \qquad (4.9)$$

while from (4.7) and (3.33), one gets

$$F^{h} = (F^{\mu} \circ \pi) X_{\mu} \quad \phi_{h} = (\phi_{\mu} \circ \pi) \omega^{\mu}, \qquad (4.10)$$

which agree with (3.15).

**Exercise 4.1.** Consider the vertical and horizontal lifts of vector fields and 1-forms in a case of a connection  $\Delta^h$  on the tangent bundle  $(T(M), \pi_T, M)$ .

**Exercise 4.2.** In a case of the tangent bundle  $(T(M), \pi_T, M)$  (resp. the cotangent bundle  $(T^*(M), \pi_T^*, M)$ ) over a manifold M, any coordinate system  $\{x^{\mu}\}$  on an open set  $W_M \subseteq M$  induces natural vector bundle coordinates in the bundle space [7, Section 1.25] (see also [116, pp. 8, 43]). For the purpose, we put  $e_{\mu} = \frac{\partial}{\partial x^{\mu}}$ , so that  $e^{\mu} = dx^{\mu}$  and we get  $(\lambda, \mu, \ldots = 1, \ldots, \dim M$  and  $a, b = \dim M + 1, \ldots, 2 \dim M$ )

$$\{u^{I}\} = \{x^{\mu} \circ \pi_{T}, \mathrm{d}x^{\nu}\} \quad \text{i.e.,} \quad u^{\mu} = x^{\mu} \circ \pi_{T} \quad u^{a} = \mathrm{d}x^{a-\mathrm{dim}\,M}$$
(4.11a)

on  $\pi_T^{-1}(W_M)$ , in the tangent bundle case, and

$$\{u^{I}\} = \left\{x^{\mu} \circ \pi_{T^{*}}, (\cdot)\left(\frac{\partial}{\partial x^{\nu}}\right)\right\} \quad \text{i.e.,} \quad u^{\mu} = x^{\mu} \circ \pi_{T^{*}} \quad u^{\dim M + \nu} \colon \xi \mapsto \xi\left(\frac{\partial}{\partial x^{\nu}}\right)$$
(4.11b)

on  $\pi_{T^*}^{-1}(W_M) \ni \xi$ , in the cotangent bundle case. In connection with the higher order (co)tangent bundles, it is convenient the vector fibre coordinates to be denoted also as  $u_1^{\mu} := \dot{x}^{\mu} := dx^{\mu}$  in T(M) and by  $u_{\mu}^1(\cdot) = (\cdot) \left(\frac{\partial}{\partial x^{\mu}}\right)$  in  $T^*(M)$ . Find in the coordinates (4.11) the adapted (co)vector fields and the vertical/horizontal lifts of vector fields or 1-forms in the (co)tangent bundle case.

## 4.2. Linear connections on vector bundles

The most valued structures in/on vector bundles are the ones which are compatible/consistent with the linear structure of the fibres of these bundles. Since a distribution  $\Delta: p \mapsto \Delta_p \subseteq T_p(E), p \in E$ , on the bundle space E of a (vector) bundle  $(E, \pi, M)$  cannot be considered as a linear mapping without additional hypotheses, the concept of a linear connection arises from the one of the parallel transport assigned to a connection (see Definition 3.2). (For an alternative approach, see [146, p. 42].) **Definition 4.1.** A connection on a vector bundle is called *linear* if the assigned to it parallel transport is a linear mapping along every path in the base space, i.e., if the mapping (3.13) is linear for all paths  $\gamma: [\sigma, \tau] \to M$  in the base, viz.

$$\mathsf{P}^{\gamma}(\rho X) = \rho \mathsf{P}^{\gamma}(X) \tag{4.12a}$$

$$\mathsf{P}^{\gamma}(X+Y) = \mathsf{P}^{\gamma}(X) + \mathsf{P}^{\gamma}(Y), \qquad (4.12b)$$

where  $\rho \in \mathbb{K}$  and  $X, Y \in \pi^{-1}(\gamma(\sigma))$ .

Remark 4.1. The equivalence between Definitions 4.1 and IV.14.3 will be established below in Subsection 4.3.

The restriction on a connection to be linear is quite severe and is described locally by

**Theorem 4.1** (cf. [138, Section 5.2]). Let  $(E, \pi, M)$  be a vector bundle,  $\{u^I\}$  be vector bundle coordinate system on an open set  $W \subseteq E$ , and  $\Delta^h$  be a connection on it described in the frame  $\{X_I\}$ , adapted to  $\{u^I\}$ , by its 2-index coefficients  $\Gamma^a_\mu$  (see (3.27)–(3.29)). The connection  $\Delta^h$  is linear if and only if, for each  $p \in W$ ,

$$\Gamma^{a}_{\mu}(p) = -\Gamma^{a}_{b\mu}(\pi(p))u^{b}(p) = -\left(\left(\Gamma^{a}_{b\mu} \circ \pi\right) \cdot u^{b}\right)(p), \tag{4.13}$$

where  $\Gamma^a_{b\mu}: \pi(W) \to \mathbb{K}$  are some functions on the set  $\pi(W) \subseteq M$  and the minus sign before  $\Gamma^a_{b\mu}$  in (4.13) is conventional.

*Proof.* Let  $\gamma: [\sigma, \tau] \to \pi(W)$  be a  $C^1$  path. Consider the parallel transport equation (3.39"a), viz.

$$\frac{\mathrm{d}\bar{\gamma}_p^a(t)}{\mathrm{d}t} = \Gamma_\mu^a(\bar{\gamma}_p(t))\dot{\gamma}^\mu(t), \qquad (4.14)$$

where  $\bar{\gamma}_p \colon [\sigma, \tau] \to W$  is the horizontal lift of  $\gamma$  through  $p \in \pi^{-1}(\gamma(\sigma)), \bar{\gamma}^a := u^a \circ \bar{\gamma}$ , and  $\dot{\gamma}^{\mu}(t) = \frac{\mathrm{d}(x^{\mu} \circ \gamma(t))}{\mathrm{d}t} = \frac{\mathrm{d}(u^{\mu} \circ \bar{\gamma}(t))}{\mathrm{d}t}$  as  $u^{\mu} = x^{\mu} \circ \pi$  for some coordinate system  $\{x^{\mu}\}$ on  $\pi(W)$ .

SUFFICIENCY. If (4.13) holds, (4.14) is transformed into

$$\frac{\mathrm{d}\bar{\gamma}_p^a(t)}{\mathrm{d}t} = -\Gamma_{b\mu}^a(\gamma(t))\bar{\gamma}_p^b(t)\dot{\gamma}^\mu(t), \qquad (4.15)$$

which is a system of r linear first order ordinary differential equations for the r functions  $\bar{\gamma}_p^{n+1}, \ldots, \bar{\gamma}_p^{n+r}$ . According to the general theorems of existence and uniqueness of the solutions of such systems [34], it has a unique solution

$$\bar{\gamma}_p^a(t) = Y_b^a(t)p^b \tag{4.16}$$

satisfying the initial condition  $\bar{\gamma}_p^a(\sigma) = u^a(p) =: p^a$ , where  $Y = [Y_b^a]$  is the fundamental solution of (4.15), i.e.,

$$\frac{\mathrm{d}Y(t)}{\mathrm{d}t} = -[\Gamma^{a}_{b\mu}(\gamma(t))\dot{\gamma}^{\mu}(t)]^{n+r}_{a,b=n+1} \cdot Y(t) \qquad Y(\sigma) = \mathbb{1}_{r \times r} = [\delta^{a}_{b}].$$
(4.17)

The linearity of (3.14) with respect to p follows from (4.16) for  $t = \tau$ .

## 4. Connections on vector bundles

NECESSITY. Suppose (3.13) is linear in p for all paths  $\gamma \colon [\sigma, \tau] \to \pi(W)$ . Then  $\bar{\gamma}_p(t) := \mathsf{P}^{\gamma \mid [\sigma,t]}(p)$  is the horizontal lift of  $\gamma \mid [\sigma,t]$  through p and (cf. (4.16))  $\bar{\gamma}_p^a(t) = A_b^a(\gamma(t))p^b$  for some  $C^1$  functions  $A_b^a \colon \pi(W) \to \mathbb{K}$ . The substitution of this equation in (4.14) results into

$$\frac{\partial A_b^a(x)}{\partial x^{\mu}}\Big|_{x=\gamma(t)=\pi(\bar{\gamma}_p(t))}\cdot\dot{\gamma}^{\mu}(t)p^b=\Gamma^a_{\mu}(\bar{\gamma}_p(t))\dot{\gamma}^{\mu}(t).$$

Since  $\gamma: [\sigma, \tau] \to M$ , we get equation (4.13) from here, for  $t = \sigma$ , with  $\Gamma^a_{b\mu}(x) = -\frac{\partial A^a_b(x)}{\partial x^{\mu}}$  for  $x \in \pi(U)$ .

The functions  $\Gamma_{b\mu}^a: \pi(W) \to \mathbb{K}$  will be referred as the (*local*) 3-index coefficients of the linear connection  $\Delta^h$  in the adapted frame  $\{X_I\}$ . If there is no risk to confuse them with the 2-index coefficients  $\Gamma_{\mu}^a: W \to \mathbb{K}$ , they will be called simply coefficients of  $\Delta^h$ . Note, the 2-index coefficients of a linear connections are defined on (a subset of) the bundle space E, while the 3-index ones are define on (a subset of) the base space M. The equation (4.15) is simply the parallel transport equation for the linear connection considered.

**Example 4.1.** Since  $u^a$  is replaced by  $u_1^{\mu} = dx^{\mu}$  in the tangent bundle case (see Exercise 4.4), the linear connections in  $(T(M), \pi_T, M)$  have 2-index coefficients of the form

$$\Gamma^{\nu}_{\mu} = -(\Gamma^{\nu}_{\lambda\mu} \circ \pi_T) \cdot u_1^{\lambda} = -(\Gamma^{\nu}_{\lambda\mu} \circ \pi_T) \cdot \mathrm{d}x^{\lambda}$$
(4.18)

and, consequently, they can be regarded as 1-forms on M.

Consider a linear connection  $\Delta^h$  on a vector bundle  $(E, \pi, M)$ . Let  $\Gamma^a_{\mu}$  and  $\Gamma^a_{b\mu}$  be its 2- and 3-index coefficients, respectively, in a frame  $\{X_I\}$  adapted to vector bundle coordinates  $\{u^I\}$ .

**Corollary 4.1.** The 3-index coefficients  $\Gamma^a_{b\mu}$  of a linear connection  $\Delta^h$  uniquely define the fibre coefficients of  $\Delta^h$  in  $\{X_I\}$  by

$$^{\circ}\Gamma^{a}_{b\mu} = \Gamma^{a}_{b\mu} \circ \pi = \pi^{*}(\Gamma^{a}_{b\mu}), \qquad (4.19)$$

that is the fibre coefficients of a linear connection are equal to the 3-index ones lifted by the projection  $\pi$ .

*Proof.* Since (3.28a) and (4.13) imply

$$[X_{\mu}, X_b] = (\Gamma^a_{b\mu} \circ \pi) X_a, \qquad (4.20)$$

the equation (4.19) follows from (3.22a) and (3.23a) or (3.37b) and (4.13).  $\Box$ 

As the vector bundle coordinates  $u^I$  are, by definition, linear on the fibres of the bundle, the general change of such coordinates is

$$\{u^{\mu}, u^{a}\} \mapsto \{\tilde{u}^{\mu} = \tilde{x}^{\mu} \circ \pi, \tilde{u}^{a} = (B^{a}_{b} \circ \pi) \cdot u^{b}\},$$
(4.21)

with  $B = [B_b^a]$  being a non-degenerate matrix-valued function on  $\pi(W)$ . The change (4.21) entails the following transformation of the corresponding adapted frames

$$\{X_{\mu}, X_a\} \mapsto \{\tilde{X}_{\mu} = (B^{\nu}_{\mu} \circ \pi) \cdot X_{\nu}, \tilde{X}_a = (B^b_a \circ \pi) \cdot X_b\},$$
(4.22)

where  $[B^{\nu}_{\mu}] = \left[\frac{\partial x^{\nu}}{\partial \tilde{x}^{\mu}}\right]$  is a non-degenerate matrix-valued function on the intersection of the domains of  $\{x^{\mu}\}$  and  $\{\tilde{x}^{\mu}\}$ . (In (4.22) we have used that  $\frac{\partial u^{\nu}}{\partial \tilde{u}^{\mu}}\Big|_{p} = \frac{\partial (x^{\nu} \circ \pi)}{\partial (\tilde{x}^{\mu} \circ \pi)}\Big|_{p} = \frac{\partial x^{\nu}}{\partial \tilde{x}^{\mu}}\Big|_{\pi(p)}$ .)

**Proposition 4.2.** The change (4.21) implies the following transformations of the 3-index coefficients of the linear connection:

$$\Gamma^a_{b\mu} \mapsto \tilde{\Gamma}^a_{b\mu} = B^{\nu}_{\mu} \Big( B^a_d \Gamma^d_{c\nu} - \frac{\partial B^a_c}{\partial x^{\nu}} \Big) (B^{-1})^c_b, \tag{4.23}$$

*Proof.* Apply (4.22), (3.32) and (4.13). Alternatively, the same transformation law follows also from equations (3.24a) and (4.19).

If we introduce the matrix-valued functions  $\Gamma_{\mu} := [\Gamma_{b\mu}^a]$  and  $\tilde{\Gamma}_{\mu} := [\tilde{\Gamma}_{b\mu}^a]$  on M, we can rewrite (4.23) as (cf. (IV.14.25) with N = M,  $g = \operatorname{id}_M$ , and  $A = B^{-1}$ )

$$\Gamma_{\mu} \mapsto \tilde{\Gamma}_{\mu} = B^{\nu}_{\mu} \left( B \cdot \Gamma_{\nu} - \frac{\partial B}{\partial x^{\nu}} \right) \cdot B^{-1}$$

$$= B^{\nu}_{\mu} B \cdot \left( \Gamma_{\nu} \cdot B^{-1} + \frac{\partial B^{-1}}{\partial x^{\nu}} \right).$$
(4.23')

This relation corresponds to (3.25) with  $[A_b^a] = B^{-1} \circ \pi$  (see also (4.19)) as the frame  $\{e_a \colon M \to E\}$ , relative to which the vector fibre coordinate system  $\{u^a\}$  is defined  $(E \ni p \mapsto u^a(p)$  with  $p = u^a(p)e_a(\pi(p)))$ , transforms via the matrix inverse to  $B \circ \pi$ .

Let E be a  $C^2$  manifold and  $\Delta^h$  a  $C^1$  connection on  $(E, \pi, M)$ . Substituting (4.13) into (3.37a), we get the fibre components of the curvature of a linear connection as

$$R^a_{\mu\nu} = -(R^a_{b\mu\nu} \circ \pi) \cdot u^b \tag{4.24}$$

where

$$R^{a}_{b\mu\nu} := \frac{\partial}{\partial x^{\mu}} (\Gamma^{a}_{b\nu}) - \frac{\partial}{\partial x^{\nu}} (\Gamma^{a}_{b\mu}) - \Gamma^{c}_{b\mu} \Gamma^{a}_{c\nu} + \Gamma^{c}_{b\nu} \Gamma^{a}_{c\mu}, \qquad (4.25)$$

or in a matrix form

$$R_{\mu\nu} := [R^a_{b\mu\nu}] = \frac{\partial\Gamma_\nu}{\partial x^\mu} - \frac{\partial\Gamma_\mu}{\partial x^\nu} - \Gamma_\nu \cdot \Gamma_\mu + \Gamma_\mu \cdot \Gamma_\nu, \qquad (4.25')$$

#### 4. Connections on vector bundles

are the components of the curvature operator (see below (4.42)). As a result of (3.22b) and (4.24), the transformation (4.21) entails the change

$$R^a_{b\mu\nu} \mapsto \tilde{R}^a_{b\mu\nu} = B^\lambda_\mu B^\varrho_\nu (B^{-1})^a_c B^d_b R^c_{d\lambda\varrho}, \qquad (4.26)$$

or in a matrix form

$$R_{\mu\nu} \mapsto \tilde{R}_{\mu\nu} = B^{\lambda}_{\mu} B^{\varrho}_{\nu} B^{-1} \cdot R_{\lambda\varrho} \cdot B, \qquad (4.26')$$

which corresponds to (3.24b) with  $A = B^{-1} \circ \pi$  (see also (4.24)).

# 4.3. Covariant derivatives in vector bundles

A possibility for introduction of differentiation in vector bundles, endowed with connection, comes from the vector space structure of their fibres. This operation can be defined in many independent ways, leading to identical results. In one of them is involved the parallel transport induced by the connection: the idea is the values of sections to be parallel transported (along paths in the base) into a single fibre (over the paths), where one can work with the 'transported' sections as with functions with values in a vector space. Other method uses the existence of natural vertical lifts of sections of the bundle and horizontal lifts of the vector fields on the base space; since the both lifts are vector fields on the bundle space, their commutator (or Lie derivative relative to each other) is well defined and can be used as a prototype of some sort of differentiation. We shall realize below the second method mentioned, which seems is first introduced in a rudimentary form in [138, p. 31].<sup>3,4</sup> The first way, as well as the axiomatic approach, for introduction of covariant derivatives will be obtained as theorems in what follows.

Let  $(E, \pi, M)$  be a vector bundle on which a *linear* connection  $\Delta^h$  is defined. Suppose  $\{E_a\}$  is a frame in E to which vector fibre coordinates  $u^a$  are associated and  $\{u^I\}$  is the corresponding vector bundle coordinate system. The frame adapted to  $\{u^I\}$  will be denoted by  $\{X_I\}$  and  $\{\omega^I\}$  will be its dual coframe, both defined by (3.28) through the (2-index) coefficients  $\Gamma^a_\mu$  of  $\Delta^h$ .

Let  $\hat{Z} = \hat{Z}^a X_a \in \Delta^v$  and  $\bar{Z} = \bar{Z}^\mu X_\mu \in \Delta^h$  be respectively vertical and horizontal vector fields on E. Define a mapping  $\hat{\nabla} \colon \Delta^v \oplus \Delta^h = T(E) \to \mathcal{X}(E)$ such that<sup>5</sup>

$$\hat{\nabla} \colon (\hat{Z}, \bar{Z}) \mapsto \hat{\nabla}_{\bar{Z}}(\hat{Z}) := \Pi(\mathcal{L}_{\bar{Z}}\hat{Z}) \in \mathcal{X}(E), \tag{4.27}$$

<sup>3</sup>In [138, p. 31] is proved that, for  $F = \frac{\partial}{\partial x^{\mu}}$  and in our notation, the *a*th component of the right hand sides of (4.35) and of (4.36) coincide in a frame  $\{E_a\}$  in E.

<sup>&</sup>lt;sup>4</sup>An equivalent alternative approach is realized in [23, Sections 2.49–2.52].

<sup>&</sup>lt;sup>5</sup>The idea of the construction (4.27) is to drag the vertical vector field  $\hat{Z}$  along the horizontal one  $\bar{Z}$ , which will give a vector field in  $\mathcal{X}(E)$ , and then to project the result onto the *vertical* distribution  $\Delta^v$  by means of the invariant projection operator  $\Pi = X_a \otimes \omega^a : \mathcal{X}(E) \to \mathcal{X}(E)$ . Evidently  $\Pi^2 = \Pi \circ \Pi = \Pi$  and  $\Pi$  is the unit (identity) tensor in the tensor product of vector fields and 1-forms on E.

where the (1,1) tensor field

$$\Pi := \sum_{a} X_a \otimes \omega^a \tag{4.28}$$

is considered as a operator on the set of vector fields on E. Since (see (2.1b) and (III.2.1))

$$\mathcal{L}_{\bar{Z}}\hat{Z} = \bar{Z}(\hat{Z}^a)X_a + \bar{Z}^{\mu}\hat{Z}^a[X_{\mu}, X_a]$$

and  $\omega^{a}(X_{\mu}) = \delta^{a}_{\mu} = 0$ , from (3.36), (3.37b) and (4.27), we obtain

$$\hat{\nabla}_{\bar{Z}}\hat{Z} = \bar{Z}^{\mu} \{ X_{\mu}(\hat{Z}^{a}) - \hat{Z}^{b} \partial_{b}(\Gamma^{a}_{\mu}) \} X_{a}, \qquad (4.29)$$

from where one can prove, via direct calculation, the independence of  $\hat{\nabla}_{\bar{Z}}\hat{Z}$  of the particular (co)frame used. For any particular point  $p \in E$ , the value of the vector field (4.29) at p is a vertical vector,  $(\hat{\nabla}_{\bar{Z}}\hat{Z})|_p \in \Delta_p^v$ , but generally  $\hat{\nabla}_{\bar{Z}}\hat{Z}$ is not a vertical vector field. The reason is that a vertical vector field on E is a mapping  $V: p \mapsto V_p \in \Delta_p^v := T_p(\pi^{-1}(\pi(p)) := T_{i(p)}(\pi^{-1}(\pi(p)) = (\pi_*|_p)^{-1}(0_{\pi(p)}),$ with  $i: \pi^{-1}(p) \to E$  being the inclusion mapping and  $0_{\pi(p)} \in T_{\pi(p)}(M)$  being the zero vector, due to which  $V_p$ , and hence its components, must depend only on  $\pi(p) \in M$ . Therefore, we have

$$\hat{\nabla}_{\bar{Z}}\hat{Z} \in \Delta^v \iff \partial_b(\Gamma^a_\mu) = -\Gamma^a_{b\mu} \circ \pi \iff \Gamma^a_\mu = -(\Gamma^a_{b\mu} \circ \pi) \cdot u^b + G^a_\mu \circ \pi, \quad (4.30)$$

for some functions  $\Gamma^a_{b\mu}, G^a_{\mu} \colon M \to \mathbb{K}$ . Thus  $\hat{\nabla}_{\bar{Z}} \hat{Z}$  is a vertical vector field if and only if the 2-index coefficients  $\Gamma^a_{\mu}$  in  $\{X_I\}$  of the connection  $\Delta^h$  are of the form

$$\Gamma^a_\mu = -(\Gamma^a_{b\mu} \circ \pi) \cdot u^b + G^a_\mu \circ \pi.$$
(4.31)

This equality selects the set of *affine connections* among all connections (see Subsection 4.4 below);<sup>6</sup> in particular, of this type are the linear connections for which  $G^a_{\mu} = 0$  and  $\Gamma^a_{b\mu}$  are their 3-index coefficients (see (4.13)). For connections with 2-index coefficients (4.31), equation (4.29) reduces to

$$\hat{\nabla}_{\bar{Z}}\hat{Z} = \bar{Z}^{\mu} \{ X_{\mu}(\hat{Z}^a) + \hat{Z}^b(\Gamma^a_{b\mu} \circ \pi) \} X_a \in \Delta^v.$$

$$(4.32)$$

Now the idea of introduction of a covariant derivative of a section  $Y \in$ Sec $(E, \pi, M)$  along a vector field  $F \in \mathcal{X}(M)$  is to 'lower' the operator  $\hat{\nabla}$  from T(E) to T(M).

**Definition 4.2.** A covariant derivative or covariant derivative operator, associated to a linear (or affine) connection  $\Delta^h$  on a vector bundle  $(E, \pi, M)$ , is a mapping

$$\nabla \colon \mathcal{X}(M) \times \operatorname{Sec}^{1}(E, \pi, M) \to \operatorname{Sec}^{0}(E, \pi, M)$$
$$\nabla \colon (F, Y) \mapsto \nabla_{F} Y$$
(4.33)

<sup>6</sup>Usually the affine connections are defined on affine bundles [11, 146].

such that, for  $F \in \mathcal{X}(M)$  and  $Y \in \text{Sec}^1(E, \pi, M)$ ,  $\nabla_F Y$  is the unique section of  $(E, \pi, M)$  whose vertical lift is  $\hat{\nabla}_{F^h} Y^v$ , with  $\hat{\nabla}$  defined by (4.27) (or (4.32)), viz.

$$(\nabla_F Y)^v := \hat{\nabla}_{F^h} Y^v \tag{4.34}$$

or

$$\nabla_F Y = v^{-1} \circ \hat{\nabla}_{(\pi_*|_{\Delta^h})^{-1}(F)}(v(Y)) = (v^{-1} \circ \hat{\nabla}_{(\pi_*|_{\Delta^h})^{-1}(F)} \circ v)(Y), \qquad (4.35)$$

where  $F^h \in \Delta^h$  and  $Y^v \in \Delta^v$  are respectively the horizontal and vertical lifts of F and Y.

*Remark* 4.2. Definition 4.2 and the rest of this subsection are valid also for affine connections for which (4.31) holds, not only for the linear ones. For some details, see Subsection 4.4.

**Proposition 4.3.** Let  $\{E_a\}$  be a frame in E and  $\{x^{\mu}\}$  local coordinate system on M. If  $Y = Y^a E_a \in \text{Sec}^1(E, \pi, M)$  and  $F = F^{\mu} \frac{\partial}{\partial x^{\mu}} \in \mathcal{X}(M)$ , then we have the explicit local expression

$$\nabla_F Y = F^{\mu} \left( \frac{\partial Y^a}{\partial x^{\mu}} + \Gamma^a_{b\mu} Y^b \right) E_a.$$
(4.36)

*Proof.* Apply (4.34), (4.9), (4.10), (4.32), and (4.2).

**Proposition 4.4.** Let  $\Delta^h$  be a linear connection on  $(E, \pi, M)$  and P be the generated by it parallel transport. Let  $x \in M$ ,  $\gamma \colon [\sigma, \tau] \to M$ ,  $\gamma(t_0) = x$  for some  $t_0 \in [\sigma, \tau]$ , and  $\dot{\gamma}(t_0) = F_x$ , i.e.,  $\gamma$  to be the integral path of  $F \in \mathfrak{X}(M)$  through x. Then

$$(\nabla_F Y)|_x = \lim_{s \to t_0} \frac{P_{s \to t_0}^{\gamma}(Y_{\gamma(s)}) - Y_{\gamma(t_0)}}{s - t_0} = \lim_{\varepsilon \to 0} \frac{P_{t_0 + \varepsilon \to t_0}^{\gamma}(Y_{\gamma(t_0 + \varepsilon)}) - Y_{\gamma(t_0)}}{\varepsilon}, \quad (4.37)$$

where  $Y \in \text{Sec}^1(E, \pi, M)$  and

$$P_{s \to t}^{\gamma} := \begin{cases} \mathsf{P}^{\gamma | [s,t]} & \text{for } s \le t \\ \left(\mathsf{P}^{\gamma | [t,s]}\right)^{-1} & \text{for } s \ge t. \end{cases}$$
(4.38)

*Proof.* Use Definition 3.2 and apply the parallel transport equation (4.15) with initial value  $\bar{\gamma}_{Y_{\gamma(s)}}(s) = Y_{\gamma(s)}$  at the point  $t = s \in [\sigma, \tau]$ .

Remark 4.3. The mapping  $P: \gamma \mapsto P^{\gamma}: (s,t) \mapsto P_{s \to t}^{\gamma}$  is a (parallel) transport along paths – see Proposition 8.6 on page 395.

By Proposition 4.4, the equation (4.37) can be used as an equivalent definition of a covariant derivative associated with a linear connection.

**Proposition 4.5.** Let  $F, G \in \mathcal{X}(M)$ ,  $Y, Z \in Sec^{1}(E, \pi, M)$ , and  $f: M \to \mathbb{K}$  be a  $C^{1}$  function. Then:

$$\nabla_{F+G}Y = \nabla_F Y + \nabla_G Y \tag{4.39a}$$

$$\nabla_{fF}Y = f\nabla_F Y \tag{4.39b}$$

$$\nabla_F(Y+Z) = \nabla_F Y + \nabla_F Z \tag{4.39c}$$

$$\nabla_F(fY) = F(f) \cdot Y + f \cdot \nabla_F Y. \tag{4.39d}$$

*Proof.* Apply (4.36).

**Proposition 4.6.** If a mapping (4.33) satisfies (4.39), there exists a unique linear connection  $\Delta^h$ , the assigned to which covariant derivative is exactly  $\nabla$ .

*Proof.* Define local functions  $\Gamma^a_{b\mu}$  on M, called *components* of  $\nabla$ , by the decomposition

$$\nabla_{\frac{\partial}{\partial x^{\mu}}} E_b =: \Gamma^a_{b\mu} E_a. \tag{4.40}$$

A simple verification proves that they transform according to (4.23) and hence the quantities (4.13) transform by (3.32). Proposition 3.3 ensures the existence of a unique linear connection whose 2-index (3-index) coefficients are  $\Gamma^a_{\mu}$  ( $\Gamma^a_{b\mu}$ ). Thus the covariant derivative of  $Y \in \text{Sec}(E, \pi, M)$  relative to  $F \in \mathcal{X}(M)$  is given by the right-hand side of (4.36). On another hand, (4.39) entail (4.36), with  $\Gamma^a_{b\mu}$  defined by (4.40), so that  $\nabla$  is exactly the covariant derivative operator assigned to the connection with 3-index coefficients  $\Gamma^a_{b\mu}$ .

Consequently, equations (4.39) and (4.40) provide a third equivalent definition of a covariant derivative (covariant derivative operator). Moreover, since Proposition 4.6 establishes a bijective correspondence between linear connections and operators (4.33) satisfying (4.39), quite often such operators are called linear connections.<sup>7</sup> As it is clear from the proof of Proposition 4.6, the bijection between linear connections and covariant derivative operators is locally given by the coincidence of their (3-index) coefficients and components, respectively.

The four equations (4.39) are identical with conditions (i)–(iv) in Definition IV.14.7 for N = M and  $g = id_M$ . This fact means that a covariant derivative according to Definition 4.2 is a covariant derivative in  $(E, \pi, M)$  according to Definition IV.14.7 and vice versa, so that the both definitions are equivalent (for N = M and  $g = id_M$  in the latter one). Combining that result with Conclusion IV.14.1, we see that the Definitions 4.1 and IV.14.3 of a linear connection on a vector bundle are equivalent (see also Definition 3.1).

**Exercise 4.3.** A  $C^1$  section  $\omega = \omega_a E^a$  of the bundle dual to  $(E, \pi, M)$  can be differentiated covariantly similarly as the sections of  $(E, \pi, M)$ . Show that the corresponding operator, say  $\nabla^*$ , can equivalently be defined by (the 'Leibnitz rule')

$$(\nabla_F^*\omega)(Y) = F(\omega(Y)) - \omega(\nabla_F Y)$$
(4.41)

<sup>&</sup>lt;sup>7</sup>See also [23, Sections 2.15 and 2.52].

## 4. Connections on vector bundles

and locally is valid the equation

$$\nabla_F^* \omega = F^{\mu} \Big( \frac{\partial \omega_a}{\partial x^{\mu}} - \Gamma_{a\mu}^b \omega_b \Big) E^a.$$

Equipped with the covariant derivative  $\nabla$  assigned to a  $C^1$  linear connection  $\Delta^h$ , we define the *curvature operator* of  $\Delta^h$  (or  $\nabla$ ) by

$$R: \mathcal{X}(M) \times \mathcal{X}(M) \to \operatorname{End}(\operatorname{Sec}(E, \pi, M))$$
  

$$R: (F, G) \mapsto R(F, G) := \nabla_F \circ \nabla_G - \nabla_G \circ \nabla_F - \nabla_{[F, G]_{\bullet}},$$
(4.42)

with End(...) denoting the set of endomorphisms of (...).

Exercise 4.4. Prove that locally

$$(R(F,G))(Y) = (R^{a}_{b\mu\nu}Y^{b}F^{\mu}G^{\nu})E_{a}, \qquad (4.43)$$

where the functions  $R^a_{b\mu\nu}: M \to \mathbb{K}$ , called the *components* of the curvature operator R in the pair of frames  $\left(\left\{\frac{\partial}{\partial x^{\mu}}\right\}, \left\{E_a\right\}\right)$ , are defined by

$$R\left(\frac{\partial}{\partial x^{\mu}}, \frac{\partial}{\partial x^{\nu}}\right)(E_b) =: R^a_{b\mu\nu}E_a \tag{4.44}$$

and are explicitly expressed through the coefficients of  $\nabla$  (= 3-index coefficients of  $\Delta^h$ ) via (4.25).

A linear connection or covariant derivative operator is called *flat* or *curvature free* if

$$R = 0 \qquad (\iff R^a_{b\mu\nu} = 0). \tag{4.45}$$

Obviously, the flatness of  $\Delta^h$  or  $\nabla$  is a necessary and sufficient condition for the (local) integrability of the horizontal distribution  $\Delta^h \colon p \mapsto \Delta^h_p \subseteq T_p(E), p \in E$  (see (3.23b) and (4.24)).

## 4.4. Affine connections

In Subsection 4.3, we met a class of connections on a vector bundle whose local 2-index coefficients have the form (see (4.31))

$$\Gamma^a_\mu = -(\Gamma^a_{b\mu} \circ \pi) \cdot u^b + G^a_\mu \circ \pi \tag{4.46}$$

in the frame  $\{X_I\}$  adapted to a vector bundle coordinate system  $\{u^I\}$ . From  $\partial_b \Gamma^a_{\mu} = -\Gamma^a_{b\mu}$  and (3.32), one derives that the functions  $\Gamma^a_{b\mu}$  in (4.46) transform according to (4.23), viz.

$$\Gamma^a_{b\mu} \mapsto \tilde{\Gamma}^a_{b\mu} = B^{\nu}_{\mu} \Big( B^a_d \Gamma^d_{c\nu} - \frac{\partial B^a_c}{\partial x^{\nu}} \Big) (B^{-1})^c_b \tag{4.47}$$

when the vector bundle coordinates or adapted frames undergo the change (4.21) or (4.22), respectively. Thus, combining (3.32), (4.47) and (4.46), we see that (4.21) or (4.22) implies the transition

$$G^a_\mu \mapsto \tilde{G}^a_\mu = B^a_b G^b_\nu B^\nu_\mu. \tag{4.48}$$

Consequently, the functions  $\Gamma^a_{b\mu}$  in (4.46) are 3-index coefficients of a linear connection, while  $G^a_{\mu}$  in it are the components of a linear mapping  $G: \mathcal{X}(M) \to \text{End}(\text{Sec}((E, \pi, M)^*))$  such that  $G: F \mapsto G(F): \omega \mapsto (G(F))(\omega)$ , for  $F \in \mathcal{X}(M)$  and a section  $\omega$  of the bundle  $(E, \pi, M)^*$  dual to  $(E, \pi, M)$ , and  $(G(\frac{\partial}{\partial x^{\mu}}))(E^a) = G^a_{\mu}$ . The invariant description of the connections with local 2-index coefficients of the type (4.46) is as follows.

**Definition 4.3.** A connection on a vector bundle is termed *affine connection* if the assigned to it parallel transport P is an affine mapping along all paths  $\gamma \colon [\sigma, \tau] \to M$  in the base space, i.e.,

$$\mathsf{P}^{\gamma}(\rho X) = \rho \mathsf{P}^{\gamma}(X) + (1 - \rho) \mathsf{P}^{\gamma}(\mathbf{0}) \tag{4.49a}$$

$$\mathsf{P}^{\gamma}(X+Y) = \mathsf{P}^{\gamma}(X) + \mathsf{P}^{\gamma}(Y) - \mathsf{P}^{\gamma}(\mathbf{0}), \qquad (4.49b)$$

where  $\rho \in \mathbb{K}$ ,  $X, Y \in \pi^{-1}(\gamma(\sigma))$ , and **0** is the zero vector in the fibre  $\pi^{-1}(\gamma(\sigma))$ , which is a K-vector space.

An affine connection for which  $\mathsf{P}^{\gamma}(\mathbf{0})$  is the zero vector in  $\pi^{-1}(\gamma(\tau))$  is a linear connection and *vice versa* – see Definitions 4.1 and 4.3.

**Theorem 4.2.** Let  $(E, \pi, M)$  be a vector bundle,  $\{u^I\}$  be vector bundle coordinate system over an open set  $U \subseteq E$ , and  $\Delta^h$  be a connection on it with 2-index coefficients  $\Gamma^a_\mu$  in the frame  $\{X_I\}$  adapted to  $\{u^I\}$ . The connection  $\Delta^h$  is an affine connection if and only if equation (4.46) holds for some functions  $\Gamma^a_{b\mu}, G^a_\mu: \pi(U) \to \mathbb{K}$ .

Proof (cf. the proof of Theorem 4.1). Take a  $C^1$  path  $\gamma: [\sigma, \tau] \to \pi(U)$  and consider the parallel transport equation (3.39"a), viz.

$$\frac{\mathrm{d}\bar{\gamma}_{p}^{a}(t)}{\mathrm{d}t} = \Gamma_{\mu}^{a}(\bar{\gamma}_{p}(t))\dot{\gamma}^{\mu}(t), \qquad (4.50)$$

where  $\bar{\gamma}_p \colon [\sigma, \tau] \to U$  is the horizontal lift of  $\gamma$  through  $p \in \pi^{-1}(\gamma(\sigma)), \bar{\gamma}^a := u^a \circ \bar{\gamma}$ , and  $\dot{\gamma}^{\mu}(t) = \frac{\mathrm{d}(x^{\mu} \circ \gamma(t))}{\mathrm{d}t} = \frac{\mathrm{d}(u^{\mu} \circ \bar{\gamma}(t))}{\mathrm{d}t}$  as  $u^{\mu} = x^{\mu} \circ \pi$  for some coordinates  $\{x^{\mu}\}$ on  $\pi(U)$ .

SUFFICIENCY. If (4.46) holds, equation (4.50) can be transformed into

$$\frac{\mathrm{d}\bar{\gamma}_{p}^{a}(t)}{\mathrm{d}t} = -\Gamma_{b\mu}^{a}(\gamma(t))\bar{\gamma}_{p}^{b}(t)\dot{\gamma}^{\mu}(t) + G_{\mu}^{a}(\gamma(t))\dot{\gamma}^{\mu}(t), \qquad (4.51)$$

which is a system of r linear inhomogeneous first order ordinary differential equations for the r functions  $\bar{\gamma}_p^{n+1}, \ldots, \bar{\gamma}_p^{n+r}$ . According to the general theorems of

## 4. Connections on vector bundles

existence and uniqueness of the solutions of such systems [34], it has a unique solution

$$\bar{\gamma}_p^a(t) = Y_b^a(t)p^b + y^a(t)$$
 (4.52)

satisfying the initial condition  $\bar{\gamma}_p^a(\sigma) = u^a(p) =: p^a$ , where  $Y = [Y_b^a]$  is the fundamental solution of (4.15) (see (4.17)) and  $y^a(t)$  is the solution of (4.51) with  $y^a(t)$  for  $\bar{\gamma}_p^a(t)$  satisfying the initial condition  $y^a(\sigma) = 0$ . The affinity of (3.13) in p, i.e., (4.49), follows from (4.52) for  $t = \tau$ .

NECESSITY. Suppose (3.13) is affine in p for all paths  $\gamma: [\sigma, \tau] \to \pi(U)$ . Then  $\bar{\gamma}_p(t) := \mathsf{P}^{\gamma|[\sigma,t]}(p)$  is the horizontal lift of  $\gamma|[\sigma,t]$  through p and (cf. (4.52))  $\bar{\gamma}_p^a(t) = A_b^a(\gamma(t))p^b + A^a(\gamma(t))$  for some  $C^1$  functions  $A_b^a, A^a: \pi(U) \to \mathbb{K}$ . The substitution of this equation in (4.50) results into

$$\frac{\partial A_b^a(x)}{\partial x^{\mu}}\Big|_{x=\gamma(t)=\pi(\bar{\gamma}_p(t))}\cdot\dot{\gamma}^{\mu}(t)p^b + \frac{\partial A^a(x)}{\partial x^{\mu}}\Big|_{x=\gamma(t)=\pi(\bar{\gamma}_p(t))}\cdot\dot{\gamma}^{\mu}(t) = \Gamma^a_{\mu}(\bar{\gamma}_p(t))\dot{\gamma}^{\mu}(t).$$

Since  $\gamma: [\sigma, \tau] \to M$ , we get equation (4.46) from here, for  $t = \sigma$ , with  $\Gamma^a_{b\mu}(x) = -\frac{\partial A^a_b(x)}{\partial x^{\mu}}$  and  $G^a_{\mu}(x) = \frac{\partial A^a(x)}{\partial x^{\mu}}$  for  $x \in \pi(U)$ .

**Proposition 4.7.** There is a bijective mapping  $\alpha$  between the sets of affine connections and of pairs  $(\nabla, G)$  of a linear connection  $\nabla$  and a linear mapping  $G: \mathcal{X}(M) \to \operatorname{End}(\operatorname{Sec}((E, \pi, M)^*)).$ 

Proof. If  ${}^{A}\Delta^{h}$  is an affine connection with 2-index coefficients given by (4.46) (see Theorem 4.2), then (see the discussion after equation (4.46)) to it corresponds the pair  $\alpha({}^{A}\Delta^{h}) := ({}^{L}\Delta^{h}, G)$  of a linear connection, with 3-index coefficients  $\Gamma^{a}_{b\mu}$  and linear mapping  $G: \mathcal{X}(M) \to \operatorname{End}(\operatorname{Sec}((E, \pi, M)^{*})))$ , with components  $G^{a}_{\mu}$ . Conversely, to a pair ( ${}^{L}\Delta^{h}, G$ ), locally described via the 3-index coefficients  $\Gamma^{a}_{b\mu}$  of  ${}^{L}\Delta^{h}$  and components  $G^{a}_{\mu}$  of G, there corresponds an affine connection  ${}^{A}\Delta^{h} = \alpha^{-1}({}^{L}\Delta^{h}, G)$  with 2-index coefficients given by (4.46).

In Subsection 4.3, it was demonstrated that covariant derivatives can be introduced for affine connections, not only for linear ones.

**Proposition 4.8.** The covariant derivative for an affine connection  ${}^{A}\Delta^{h}$  coincides with the one for the linear connection  ${}^{L}\Delta^{h}$  given via  $\alpha({}^{A}\Delta^{h}) = ({}^{L}\Delta^{h}, G)$  with  $\alpha$  defined in the proof of Proposition 4.7.

*Proof.* Apply (4.29)–(4.36).

If a linear connection  ${}^{L}\Delta^{h}$  and an affine one  ${}^{A}\Delta^{h}$  are connected by  $\alpha({}^{A}\Delta^{h}) = ({}^{L}\Delta^{h}, G)$  for some G, then some of their characteristics coincide; e.g., such are their fibre coefficients (see (3.37b), (4.46) and (4.13)) and all quantities expressed via the corresponding to them (identical) covariant derivatives. However, quantities, containing (depending on) partial derivatives relative to the basic coordinates  $u^{\mu}$ , are generally different for those connections. For instance, if  ${}^{A}R^{a}_{\mu\nu}$  and  ${}^{L}R^{a}_{\mu\nu}$ 

are the fibre components of the curvatures of  ${}^{A}\Delta^{h}$  and  ${}^{L}\Delta^{h}$ , respectively, then, by (3.37a) and (4.46), we have

$${}^{A}R^{a}_{\mu\nu} = -({}^{L}R^{a}_{b\mu\nu}\circ\pi)\cdot u^{b} - T^{a}_{\mu\nu}\circ\pi$$

$$(4.53)$$

$${}^{L}R^{a}_{\mu\nu} = -({}^{L}R^{a}_{b\mu\nu}\circ\pi)\cdot u^{b}$$

$$(4.54)$$

where (see (4.25))

$${}^{L}R^{a}_{b\mu\nu} := \frac{\partial}{\partial x^{\mu}} (\Gamma^{a}_{b\nu}) - \frac{\partial}{\partial x^{\nu}} (\Gamma^{a}_{b\mu}) - \Gamma^{c}_{b\mu} \Gamma^{a}_{c\nu} + \Gamma^{c}_{b\nu} \Gamma^{a}_{c\mu}, \qquad (4.55)$$

$$T^a_{\mu\nu} := -\frac{\partial}{\partial x^{\mu}} (G^a_{\nu}) + \frac{\partial}{\partial x^{\nu}} (G^a_{\mu}) + \Gamma^a_{c\nu} G^c_{\mu} - \Gamma^a_{c\mu} G^c_{\nu}$$
(4.56)

and the functions  $T^a_{\mu\nu}$  have a sense of components of the torsion of  ${}^L\Delta^h$  relative to G [146, pp. 42, 46].

Thus, in general, the affine connections and linear connections are essentially different. However, they imply identical theories of covariant derivatives.

If, for some reason, the linear mapping G is fixed, then the set of linear connections  $\{{}^{L}\Delta{}^{h}\}$  can be identified with the subset  $\{\alpha^{-1}({}^{L}\Delta{}^{h},G)\}$  of the set of affine connections  $\{{}^{A}\Delta{}^{h}\}$ .

**Example 4.2.** We shall exemplify the above material in a case of the tangent bundle  $(T(M), \pi_T, M)$  over a manifold M. Using the base indices  $\mu, \nu, \ldots$  for the fibre ones  $a, b, \ldots$  according to the rule  $a \mapsto \mu = a - \dim M$ , we rewrite (4.46) as

$$\Gamma^{\mu}_{\nu} = -(\Gamma^{\mu}_{\lambda\nu} \circ \pi_T) \cdot u_1^{\lambda} + G^{\mu}_{\nu} \circ \pi_T.$$
(4.57)

Now the affine connections on  $(T(M), \pi_T, M)$  are the generalized affine connections on M [11, Chapter III, § 3]. The choice of G via

$$G^{\mu}_{\nu} \colon M \to \delta^{\mu}_{\nu},$$
 (4.58)

which corresponds to the identical transformation of the spaces tangent to M, singles out the set of affine connections on M- see [11, Chapter III, § 3] or [23, pp. 103–105] – (known also as Cartan connections on M [146, p. 46]) whose 2-index coefficients have the form (see (4.57), (4.11a) and (4.58))

$$\Gamma^{\mu}_{\nu} = -(\Gamma^{\mu}_{\lambda\nu} \circ \pi_T) \cdot \mathrm{d}x^{\lambda} + \delta^{\mu}_{\nu}. \tag{4.59}$$

Combining this example with Proposition 4.7, we derive

**Proposition 4.9** (cf. [11, Chapter III, § 3, Theorem 3.3]). There is a bijective correspondence between the sets of linear connections and of affine ones on a manifold.

Often the terms "linear connection" and "affine connection" on a manifold are used as synonyms, due to the last result.

# 5. General (co)frames

Until now two special kinds of local (co)frames in the (co)tangent bundle to the bundle space of a bundle were employed, viz. the natural holonomic ones, induced by some local coordinates, and the adapted (co)frames determined by local coordinates and a connection on the bundle. The present section is devoted to (re)formulation of some important results and formulae in arbitrary (co)frames, which in particular can be natural or adapted (if a connection is presented) ones.

Let  $(E, \pi, M)$  be a  $C^2$  bundle and  $\{e_I\}$  a (local) frame in T(E). The components  $C_{IJ}^K$  of the anholonomicity object of  $\{e_I\}$  are defined by (3.19) and a change

$$\{e_I\} \mapsto \{\bar{e}_I = B_I^J e_J\} \tag{5.1}$$

with a non-degenerate matrix-valued function  $B = [B_I^J]_{I,J=1}^{n+r}$  entails (see (2.7))

$$C_{IJ}^{K} \mapsto \bar{C}_{IJ}^{K} = (B^{-1})_{L}^{K} \left( B_{I}^{M} e_{M}(B_{J}^{L}) - B_{J}^{M} e_{M}(B_{I}^{L}) + B_{I}^{M} B_{J}^{N} C_{MN}^{L} \right).$$
(5.2)

Let a connection  $\Delta^h$  on  $(E, \pi, M)$  be given. If  $\{e_I\}$  is a specialized frame for  $\Delta^h$  (see Subsection 3.2), then the set  $\{C_{IJ}^K\}$  is naturally divided into the six groups (3.20). The value of that division is in its invariance with respect to the class of specialized frames, which means that, if  $\{\bar{e}_I\}$  is also a specialized frame, then the transformed components of the elements of each group are functions only in the elements of the non-transformed components of the same group – see (3.24), (3.21), and (2.7).

**Exercise 5.1.** By means of (5.1), prove that, if a division like (3.20) holds in a frame  $\{e_I\}$ , then it holds in  $\{\bar{e}_I\}$  if and only if the matrix-valued function B is of the form (3.16).

In particular, we cannot talk about fibre coefficients of  $\Delta^h$  and of fibre components of the curvature of  $\Delta^h$  in frames more general than the specialized ones as in that case the transformation (5.1), with  $\{e_I\}$  (resp.  $\{\bar{e}_I\}$ ) being a specialized (resp. non-specialized) frame, will mix, for instance, the fibre coefficients and the curvature's fibre components of  $\Delta^h$  in  $\{\bar{e}_I\}$  – see (5.2).

It is a simple, but important, fact that the specialized frames are (up to renumbering) the most general ones which respect the splitting of T(E) into vertical and horizontal components. Suppose  $\{e_I\}$  is a specialized frame. Then the general element of the set of all specialized frames is (see (3.4a) and (3.16))

$$(\bar{e}_{\mu}, \bar{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} A^{\nu}_{\mu} & 0\\ 0 & A^{b}_{a} \end{bmatrix} = (A^{\nu}_{\mu}e_{\nu}, A^{b}_{a}e_{b}),$$
(5.3a)

where  $[A^{\nu}_{\mu}]^{n}_{\mu,\nu=1}$  and  $[A^{b}_{a}]^{n+r}_{a,b=n+1}$  are non-degenerate matrix-valued functions on E, which are constant on the fibres of  $(E, \pi, M)$ , i.e., we can set  $A^{\nu}_{\mu} = B^{\nu}_{\mu} \circ \pi$  and  $A^{b}_{a} = B^{b}_{a} \circ \pi$  for some non-degenerate matrix-valued functions  $[B^{\nu}_{\mu}]$  and  $[B^{b}_{a}]$ 

on *M*. Respectively, the general specialized coframe dual to  $\{\bar{e}_I\}$  is (see (3.4b) and (3.16))

$$\begin{pmatrix} \bar{e}^{\mu} \\ \bar{e}^{a} \end{pmatrix} = \begin{bmatrix} [A^{\lambda}_{\rho}]^{-1} & 0 \\ 0 & [A^{c}_{d}]^{-1} \end{bmatrix} \cdot \begin{pmatrix} e^{\nu} \\ e^{b} \end{pmatrix} = \begin{bmatrix} \left( [A^{\lambda}_{\rho}]^{-1} \right)^{\mu}_{\nu} e^{\nu} \\ \left( [A^{c}_{d}]^{-1} \right)^{a}_{b} e^{b} \end{bmatrix} , \quad (5.3b)$$

where  $\{e^I\}$  is the specialized coframe dual to  $\{e_I\}, e^I(e_J) = \delta^I_J$ .

Since  $\pi_*|_{\Delta^h} \colon \{X \in \Delta^h\} \to \mathcal{X}(M)$  is an isomorphism, any basis  $\{\varepsilon_\mu\}$  for  $\Delta^h$  defines a basis  $\{E_\mu\}$  of  $\mathcal{X}(M)$  such that

$$E_{\mu} = \pi_*|_{\Delta^h}(\varepsilon_{\mu}) \tag{5.4}$$

and v.v., a basis  $\{E_{\mu}\}$  for  $\mathcal{X}(M)$  induces a basis  $\{\varepsilon_{\mu}\}$  for  $\Delta^{h}$  via

$$\varepsilon_{\mu} = (\pi_*|_{\Delta^h})^{-1}(E_{\mu}).$$
(5.5)

Similarly, there is a bijection  $\{\varepsilon^{\mu}\} \mapsto \{E^{\mu}\}$  between the 'horizontal' coframes  $\{\varepsilon^{\mu}\}$ and the coframes  $\{E^{\mu}\}$  dual to the frames in T(M)  $(E^{\mu} \in \Lambda^{1}(M), E^{\mu}(E_{\nu}) = \delta^{\mu}_{\nu})$ . Thus a 'horizontal' change

$$\varepsilon_{\mu} \mapsto \bar{\varepsilon}_{\mu} = (B^{\nu}_{\mu} \circ \pi) \varepsilon_{\nu}, \qquad (5.6)$$

which is independent of a 'vertical' one given by

$$\varepsilon_a \mapsto \bar{\varepsilon}_a = (B_a^b \circ \pi) \varepsilon_b \tag{5.7}$$

with  $\{\varepsilon_a\}$  being a basis for  $\Delta^v$ , is equivalent to the transformation

$$E_{\mu} \mapsto \bar{E}_{\mu} = B^{\nu}_{\mu} E_{\nu} \tag{5.8}$$

of the basis  $\{E_{\mu}\}$  for  $\mathcal{X}(M)$ , related via (5.4) to the basis  $\{\varepsilon_{\mu}\}$  for  $\Delta^{h}$ . Here  $[B_{\mu}^{\nu}]$ and  $[B_{a}^{b}]$  are non-degenerate matrix-valued functions on M.

As  $\pi_*(\varepsilon_a) = 0 \in \mathcal{X}(M)$ , the 'vertical' transformations (5.7) do not admit interpretation analogous to the 'horizontal' ones (5.6). However, in a case of a *vector* bundle  $(E, \pi, M)$ , they are tantamount to changes of frames in the bundle space E, i.e., of the bases for  $\text{Sec}(E, \pi, M)$ . Indeed, if v is the mapping defined by (4.3), the sections

$$E_a = v^{-1}(\varepsilon_a) \tag{5.9}$$

form a basis for  $\text{Sec}(E, \pi, M)$  as the vertical vector fields  $\varepsilon_a$  form a basis for  $\Delta^v$ . Conversely, any basis  $\{E_a\}$  for the sections of  $(E, \pi, M)$  induces a basis  $\{\varepsilon_a\}$  for  $\Delta^v$  such that

$$\varepsilon_a = v(E_a). \tag{5.10}$$

As v and  $v^{-1}$  are linear, the change (5.7) is equivalent to the transformation

$$E_a \mapsto \bar{E}_a = B_a^b E_b \tag{5.11}$$

of the frame  $\{E_a\}$  in E related to  $\{\varepsilon_a\}$  via (5.9). In this way, we have proved the following result.

**Proposition 5.1.** There is a bijective correspondence between the set of specialized frames  $\{\varepsilon_I\} = \{\varepsilon_{\mu}, \varepsilon_a\}$  on a vector bundle  $(E, \pi, M)$  and the set of pairs  $(\{E_{\mu}\}, \{E_a\})$  of frames  $\{E_{\mu}\}$  on T(M) over M and  $\{E_a\}$  on E over M.<sup>1</sup>

Since conceptually the frames in T(M) and E are easier to be understood and in some cases have a direct physical interpretation, one often works with the pair ( $\{E_{\mu} = \pi_* | \Delta^h(\varepsilon_{\mu})\}, \{E_a = v^{-1}(\varepsilon_a)\}$ ) of frames instead with a specialized frame  $\{\varepsilon_I\} = \{\varepsilon_{\mu}, \varepsilon_a\}$ ; for instance  $\{E_{\mu}\}$  and  $\{E_a\}$  can be completely arbitrary frames in T(M) and E, respectively, while the specialized frames represent only a particular class of frames in T(E).

One can *mutatis mutandis* localize the above considerations when M is replaced with an open subset  $U_M$  in M and E is replaced with  $W = \pi^{-1}(U_M)$ . Such a localization is important when the bases/frames considered are connected with some local coordinates or when they should be smooth.<sup>2</sup>

Let us turn now our attention to frames adapted to local coordinate system  $\{u^I\}$  on an open set  $W \subseteq E$  for a given connection  $\Delta^h$  on a general  $C^1$  bundle  $(E, \pi, M)$  (see (3.27)–(3.30)). Since in their definition the local coordinates  $u^I$  enter only via the vector fields  $\partial_I := \frac{\partial}{\partial u^I} \in \mathcal{X}(E)$ , we can generalize this definition by replacing  $\{\partial_I\}$  with an arbitrary frame  $\{e_I\}$  defined in T(E) over an open set  $W \subseteq E$  and such that  $\{e_a|_p\}$  is a basis for the space  $T_p(\pi^{-1}(\pi(p)))$  tangent to the fibre through  $p \in W$ . So, using  $\{e_I\}$  for  $\{\partial_I\}$ , we have

$$(e^{W}_{\mu}, e^{W}_{a}) = (D^{\nu}_{\mu}e_{\nu} + D^{a}_{\mu}e_{a}, D^{b}_{a}e_{b}) = (e_{\nu}, e_{b}) \cdot \begin{pmatrix} [D^{\nu}_{\mu}] & 0\\ [D^{b}_{\mu}] & [D^{b}_{a}] \end{pmatrix} , \qquad (5.12)$$

where  $\{e_I^W\}$  is a *specialized* frame in T(W),  $[D^{\nu}_{\mu}]$  and  $[D^b_a]$  are non-degenerate matrix-valued functions on W, and  $D^a_{\mu} \colon W \to \mathbb{K}$ .

**Definition 5.1.** The specialized frame  $\{X_I\}$  over W in T(W), obtained from (5.12) via an admissible transformation (3.4a) with matrix  $A = \begin{pmatrix} [D_{\nu}^{\mu}]^{-1} & 0 \\ 0 & [D_{b}^{a}]^{-1} \end{pmatrix}$ , is called adapted to the frame  $\{e_I\}$  for  $\Delta^h$ .<sup>3</sup>

**Exercise 5.2.** Using (3.4) and (3.16), verify that the adapted frame  $\{X_I\}$  and coframe  $\{\omega^I\}$  dual to it are independent of the particular specialized frame  $\{e_I^W\}$  entering into their definitions via (5.12). The equalities (5.13a) and (5.21) derived below are indirect proof of that fact too.

<sup>&</sup>lt;sup>1</sup>It should be mentioned the evident fact that a frame  $\{E_{\mu}\}$  in T(M) over M is also a basis for the module  $\mathfrak{X}(M)$  of vector fields over M and hence is a basis for the set  $Sec(T(M), \pi_T, M)$ of section of the bundle tangent to M, due to  $\mathfrak{X}(M) = Sec(T(M), \pi_T, M)$ . Similarly, a frame  $\{E_a\}$  on E over M is a basis for the set  $Sec(E, \pi, M)$  of sections of the vector bundle  $(E, \pi, M)$ .

<sup>&</sup>lt;sup>2</sup>Recall, not every manifold admits a *global* nowhere vanishing  $C^m$ ,  $m \ge 0$ , vector field (see [153] or [1, Section 4.24]); e.g., such are the even-dimensional spheres  $\mathbb{S}^{2k}$ ,  $k \in \mathbb{N}$ , in Euclidean space.

<sup>&</sup>lt;sup>3</sup>Recall, here and below the adapted frames are defined only with respect to frames  $\{e_I\} = \{e_{\mu}, e_a\}$  such that  $\{e_a\}$  is a basis for the vertical distribution  $\Delta^v$  over W, i.e.,  $\{e_a|_p\}$  is a basis for  $\Delta^v_p$  for all  $p \in W$ . Since  $\Delta^v$  is integrable, the relation  $e_a \in \Delta^v$  for all  $a = n + 1, \ldots, n + r$  implies  $[e_a, e_b] \in \Delta^v$  for all  $a, b = n + 1, \ldots, n + r$ .

According to (3.4), the adapted frame  $\{X_I\} = \{X_\mu, X_a\}$  and the coframe  $\{\omega^I\} = \{\omega^\mu, \omega^a\}$  dual to it are given by the equations

$$(X_{\mu}, X_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} \delta^{\nu}_{\mu} & 0\\ +\Gamma^{b}_{\mu} & \delta^{b}_{a} \end{bmatrix} = (e_{\mu} + \Gamma^{b}_{\mu} e_{b}, e_{a})$$
(5.13a)

$$\begin{pmatrix} \omega^{\mu} \\ \omega^{a} \end{pmatrix} = \begin{bmatrix} \delta^{\mu}_{\nu} & 0 \\ -\Gamma^{a}_{\nu} & \delta^{a}_{b} \end{bmatrix} \cdot \begin{pmatrix} e^{\nu} \\ e^{b} \end{pmatrix} = \begin{pmatrix} e^{\mu} \\ e^{a} - \Gamma^{a}_{\nu} e^{\nu} \end{pmatrix} , \qquad (5.13b)$$

where  $\{e^I\}$  is the coframe dual to  $\{e_I\}$ ,  $e^I(e_J) = \delta^I_J$ , and the functions  $\Gamma^a_{\mu} \colon W \to \mathbb{K}$ , called (2-*index*) coefficients of  $\Delta^h$  in  $\{X_I\}$ , are defined by

$$[\Gamma^a_{\mu}] := + [D^a_{\nu}] \cdot [D^{\nu}_{\mu}]^{-1}.$$
(5.14)

**Proposition 5.2.** A change  $\{e_I\} \mapsto \{\tilde{e}_I\}$  with

$$(\tilde{e}_{\mu}, \tilde{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{pmatrix} [A_{\mu}^{\nu}] & 0\\ [A_{\mu}^{b}] & [A_{a}^{b}] \end{pmatrix} = (A_{\mu}^{\nu}e_{\nu} + A_{\mu}^{b}e_{b}, A_{a}^{b}e_{b}),$$
(5.15)

where  $[A^{\nu}_{\mu}]$  and  $[A^{b}_{a}]$  are non-degenerate matrix-valued functions on W, which are constant on the fibres of  $(E, \pi, M)$ , and  $A^{b}_{\mu} \colon W \to \mathbb{K}$ , entails the transformations

$$(X_{\mu}, X_{a}) \mapsto (\tilde{X}_{\mu}, \tilde{X}_{a}) = (\tilde{e}_{\mu} + \tilde{\Gamma}^{b}_{\mu} \tilde{e}_{b}, \tilde{e}_{a}) = (A^{\nu}_{\mu} X_{\nu}, A^{b}_{a} X_{b}) = (X_{\nu}, X_{b}) \cdot \begin{bmatrix} A^{\nu}_{\mu} & 0\\ 0 & A^{b}_{a} \end{bmatrix}$$
(5.16)

$$\Gamma^{a}_{\mu} \mapsto \tilde{\Gamma}^{a}_{\mu} = \left( [A^{c}_{d}]^{-1} \right)^{a}_{b} (\Gamma^{b}_{\nu} A^{\nu}_{\mu} - A^{b}_{\mu})$$
(5.17)

of the frame  $\{X_I\}$  adapted to  $\{e_I\}$  and of the coefficients  $\Gamma^a_\mu$  of  $\Delta^h$  in  $\{X_I\}$ , i.e.,  $\{\tilde{X}_I\}$  is the frame adapted to  $\{\tilde{e}_I\}$  and  $\tilde{\Gamma}^a_\mu$  are the coefficients of  $\Delta^h$  in  $\{\tilde{X}_I\}$ .

*Proof.* Apply 
$$(5.12)$$
– $(5.14)$ .

Note 5.1. If  $\{e_I\}$  and  $\{\tilde{e}_I\}$  are adapted, then  $A^b_{\mu} = 0$ . If  $\{Y_I\}$  is a specialized frame, it is adapted to any frame  $\{e_{\mu} = A^{\nu}_{\mu}Y_{\nu}, e_a = A^b_aY_b\}$  and hence any specialized frame can be considered as an adapted one; in particular, any specialized frame is a frame adapted to itself. Obviously (see (5.14)), the coefficients of a connection identically vanish in a given specialized frame considered as an adapted one. This leads to the concept of a *normal frame* to which is devoted Section 6 below. Besides, from the above observation follows that the set of adapted frames coincides with the one of specialized frames. **Exercise 5.3.** Verify that the formulae dual to (5.15) and (5.16) are (see (3.4b) and (3.5b))

$$\begin{pmatrix} \tilde{e}^{\mu} \\ \tilde{e}^{a} \end{pmatrix} = \begin{pmatrix} [A^{\varrho}_{\tau}]^{-1} & 0 \\ -[A^{c}_{d}]^{-1}[A^{c}_{\tau}][A^{\varrho}_{\tau}]^{-1} & [A^{c}_{d}]^{-1} \end{pmatrix} \cdot \begin{pmatrix} e^{\nu} \\ e^{b} \end{pmatrix}$$

$$= \begin{pmatrix} ([A^{\varrho}_{\tau}]^{-1})^{\mu}_{\nu} e^{\nu} \\ ([A^{c}_{d}]^{-1})^{a}_{b} e^{b} - ([A^{c}_{\tau}]^{-1}[A^{c}_{\tau}][A^{\varrho}_{\tau}]^{-1})^{a}_{\nu} e^{\nu} \end{pmatrix}$$

$$\begin{pmatrix} \omega^{\mu} \\ \omega^{a} \end{pmatrix} \mapsto \begin{pmatrix} \tilde{\omega}^{\mu} \\ \tilde{\omega}^{a} \end{pmatrix} = \begin{pmatrix} ([A^{\varrho}_{\tau}]^{-1}))^{\mu}_{\nu} e^{\nu} \\ ([A^{c}_{d}]^{-1})^{a}_{b} e^{b} \end{pmatrix}.$$

$$(5.19)$$

**Example 5.1.** If  $\{e_I\}$  and  $\{\tilde{e}_I\}$  are the frames generated by local coordinates  $\{u^I\}$  and  $\{\tilde{u}^I\}$ , viz.  $e_I = \frac{\partial}{\partial u^I}$  and  $\tilde{e}_I = \frac{\partial}{\partial \tilde{u}^I}$ , the changes (5.16) and (5.17) reduce to (3.31) and (3.32), respectively. The choice  $e_I = \frac{\partial}{\partial u^I}$  also reduces Definition 5.1 to Definition 3.5.

A result similar to Proposition 3.3 is valid too provided in its formulation equation (3.32) is replaced with (5.17).

If  $e_{\mu}$  has an expansion  $e_{\mu} = e_{\mu}^{\nu} \frac{\partial}{\partial u^{\nu}} + e_{\mu}^{b} \frac{\partial}{\partial u^{b}}$  in the domain W of  $\{u^{I}\} = \{u^{\mu} = x^{\mu} \circ \pi, u^{a}\}$ , where  $e_{\mu}^{b} \colon W \to \mathbb{K}$  and  $e_{\mu}^{\nu} = x_{\mu}^{\nu} \circ \pi$  for some  $x_{\mu}^{\nu} \colon \pi(W) \to \mathbb{K}$  such that  $\det[x_{\mu}^{\nu}] \neq 0, \infty$ , and we define a frame  $\{x_{\mu}\}$  in  $T(\pi(W)) \subseteq T(M)$  by  $\{x_{\mu} \coloneqq x_{\mu}^{\nu} \frac{\partial}{\partial x^{\nu}}\}$ , then

$$\pi_*(X_\mu) = x_\mu, \tag{5.20}$$

by virtue of (3.33) and (3.35). Thus, we have (cf. (3.34))

$$X_{\mu} = (\pi_*|_{\Delta^h})^{-1}(x_{\mu}) = (\pi_*|_{\Delta^h})^{-1} \circ \pi_*(e_{\mu})$$
(5.21)

which can be used in an equivalent definition of a frame  $\{X_I\}$  adapted to  $\{e_I\}$ (with  $\{e_a\}$  being a basis for  $\Delta^v$ ):  $X_\mu$  should be defined by (5.21) and  $X_a = e_a$ . If one accepts such a definition of an adapted frame, the 2-index coefficients of a connection should be defined via the equation (5.13a), not by (5.14), and the proofs of some results, like (5.16) and (5.17), should be modified.

**Proposition 5.3.** If  $\{X_I\}$  is a frame adapted to a frame  $\{e_I\}$ , with  $\{e_a\}$  being a basis for  $\Delta^v$ , for a  $C^1$  connection  $\Delta^h$ , then (cf. (3.36))

$$[X_{\mu}, X_{\nu}]_{-} = R^{a}_{\mu\nu} X_{a} + S^{\lambda}_{\mu\nu} X_{\lambda}$$
 (5.22a)

$$[X_{\mu}, X_b]_{-} = {}^{\circ}\Gamma^a_{b\mu}X_a + C^{\lambda}_{\mu b}X_{\lambda}$$
(5.22b)

$$[X_a, X_b] = C^d_{ab} X_d, \tag{5.22c}$$

`

where (cf. (3.37))

$$R^{a}_{\mu\nu} := X_{\mu}(\Gamma^{a}_{\nu}) - X_{\nu}(\Gamma^{a}_{\mu}) - C^{a}_{\mu\nu} - \Gamma^{b}_{\mu}C^{a}_{\nu b} + \Gamma^{b}_{\nu}C^{a}_{\mu b} + \Gamma^{a}_{\lambda}(-C^{\lambda}_{\mu\nu} + \Gamma^{b}_{\mu}C^{\lambda}_{\nu b} - \Gamma^{b}_{\nu}C^{\lambda}_{\mu b}) + \Gamma^{b}_{\mu}\Gamma^{d}_{\nu}C^{a}_{bd} S^{\lambda}_{\mu\nu} := C^{\lambda}_{\mu\nu} + \Gamma^{b}_{\mu}C^{\lambda}_{\nu b} - \Gamma^{b}_{\nu}C^{\lambda}_{\mu b}$$

$$(5.23a)$$

$$^{\circ}\Gamma^{a}_{b\mu} := -X_{b}(\Gamma^{a}_{\mu}) - C^{a}_{\mu b} + \Gamma^{d}_{\mu}C^{a}_{db} - \Gamma^{a}_{\lambda}C^{\lambda}_{\mu b}$$
(5.23b)

$$[e_I, e_J]\_ =: C_{IJ}^K e_K = C_{IJ}^a e_a + C_{IJ}^\lambda e_\lambda.$$
(5.23c)

*Proof.* Insert equation (5.13a) into the corresponding commutators, use the definition (5.23c) of the components of the anholonomicity object of  $\{e_I\}$ , and apply (5.13a). Notice, as  $\{e_a\}$  is a basis for the integrable distribution  $\Delta^v$ , we have  $[e_a, e_b]_{-} \in \Delta^v$  and consequently  $C_{ab}^{\lambda} \equiv 0$ .

The functions  $R^a_{\mu\nu}$  are the fibre components of the curvature of  $\Delta^h$  and  ${}^\circ\Gamma^a_{b\mu}$ are the fibre coefficients of  $\Delta^h$  in the adapted frame  $\{X_I\}$ ; if  $e_I = \frac{\partial}{\partial u^I}$  for some bundle coordinates  $\{u^I\}$  on E, they reduce to (3.37a) and (3.37b), respectively. From (5.22), we immediately derive

**Corollary 5.1.** A connection  $\Delta^h$  is integrable iff in some (and hence any) adapted frame

$$R^a_{\mu\nu} = 0.$$
 (5.24)

**Corollary 5.2.** An adapted frame  $\{X_I\}$  is (locally) holonomic iff in it

$$R^{a}_{\mu\nu} = {}^{\circ}\Gamma^{a}_{b\mu} = S^{\lambda}_{\mu\nu} = C^{d}_{ab} = C^{\lambda}_{\mu b} = 0.$$
 (5.25)

If the initial frame  $\{e_I\}$  is changed into (5.15), then the transformation laws of the quantities (5.23) follow from (5.22) and (5.16); in particular, the curvature components transform according to the tensor equation (3.24b).

Let us now pay attention to the case when  $(E, \pi, M)$  is a vector bundle endowed with a connection  $\Delta^h$ .

According to the above-said in this section, any *adapted* frame  $\{X_I\} = \{X_\mu, X_a\}$  in T(E) is equivalent to a pair of frames in T(M) and E according to

$$\{X_{\mu}, X_{a}\} \leftrightarrow (\{E_{\mu} = \pi_{*}|_{\Delta^{h}}(X_{\mu})\}, \{E_{a} = v^{-1}(X_{a})\}).$$
(5.26)

Therefore the vertical and horizontal lifts are given by (cf. Lemma 4.1, (4.7a) and (4.10))

$$\operatorname{Sec}(E,\pi,M) \ni Y = Y^{a}E_{a} \xrightarrow{v} v(Y) := Y^{v} = (Y^{a} \circ \pi)X_{a} \in \Delta^{v}$$
(5.27a)

$$\mathcal{X}(M) \ni F = F^{\mu} E_{\mu} \xrightarrow{h} h(F) := F^{h} = (F^{\mu} \circ \pi) X_{\mu} \in \Delta^{h}.$$
(5.27b)

## 5. General (co)frames

Thus, we have the linear isomorphism

$$(h,v)\colon \mathcal{X}(M) \times \operatorname{Sec}(E,\pi,M) \to \mathcal{X}(E)$$
  
$$(h,v)\colon (F,Y) \mapsto (F^{h},Y^{v})$$
(5.28)

which explains why the covariant derivatives (see Definition 4.2) represent an equivalent description of the linear connections in vector bundles. Since any vector field  $\xi = (\xi^I \circ \pi) X_I \in \mathcal{X}(E)$  has a unique decomposition  $\xi = \xi^h \oplus \xi^v$ , with  $\xi^h = (\xi^\mu \circ \pi) X_\mu$  and  $\xi^v = (\xi^a \circ \pi) X_a$ , we have

$$(h, v)^{-1}(\xi) = (\pi_*|_{\Delta^h}(\xi^h), v^{-1}(\xi^v)) = (\xi^\mu E_\mu, \xi^a E_a).$$
(5.29)

Suppose  $\{X_I\}$  and  $\{\tilde{X}_I\}$  are two adapted frames. Then they are connected via (cf. (5.3a) and (5.16))

$$\tilde{X}_{\mu} = (B^{\nu}_{\mu} \circ \pi) X_{\nu} \quad \tilde{X}_{a} = (B^{b}_{a} \circ \pi) X_{b},$$
(5.30)

where  $[B^{\nu}_{\mu}]$  and  $[B^{b}_{a}]$  are some non-degenerate matrix-valued functions on M. The pairs of frames corresponding to them, in accordance with (5.26), are related via

$$\tilde{E}_{\mu} = B^{\nu}_{\mu} E_{\nu} \quad \tilde{E}_a = B^b_a E_b \tag{5.31}$$

and vice versa.

**Proposition 5.4.** Let  $\Delta^h$  be a linear connection on a vector bundle  $(E, \pi, M)$  and  $\{X_I\}$  be the frame adapted for  $\Delta^h$  to a frame  $\{e_I\}$  such that  $\{e_a\}$  is a basis for  $\Delta^v$ . Let  $\{u^I\} = \{u^{\mu}, u^a\}$  be vector bundle coordinate system on  $U \subseteq E$ . Suppose that the frame  $\{e_I\}$ , to which  $\{X_I\}$  is adapted to, is such that

$$(e_{\mu}, e_{a})|_{U} = (\partial_{\nu}, \partial_{b}) \cdot \begin{bmatrix} B^{\nu}_{\mu} \circ \pi & 0\\ (B^{b}_{c\mu} \circ \pi) \cdot u^{c} & B^{b}_{a} \circ \pi \end{bmatrix}$$
  
$$= \left( (B^{\nu}_{\mu} \circ \pi) \partial_{\nu} + ((B^{b}_{c\mu} \circ \pi) \cdot u^{c}) \partial_{b}, (B^{b}_{a} \circ \pi) \partial_{b} \right),$$
  
(5.32)

where  $\partial_I := \frac{\partial}{\partial u^I}$ ,  $[B^{\nu}_{\mu}]$  and  $[B^b_a]$  are non-degenerate matrix-valued functions on  $\pi(U)$ , and  $B^b_{c\mu} : \pi(U) \to \mathbb{K}$ . Then the 2-index coefficients  $\Gamma^a_{\mu}$  of  $\Delta^h$  in  $\{X_I\}$  have the representation

$$\Gamma^a_\mu = -(\Gamma^a_{b\mu} \circ \pi) \cdot u^b \tag{5.33}$$

on U for some functions  $\Gamma^a_{b\mu}$ :  $\pi(U) \to \mathbb{K}$ , called 3-index coefficients of  $\Delta^h$  in  $\{X_I\}$ .

Remark 5.1. The representation (5.33) is not valid for frames more general than the ones given by (5.32). Precisely, equation (5.33) is valid if and only if (5.32) holds for some local coordinates  $\{u^I\}$  on W – see (5.17).

*Remark* 5.2. Since the vector fibre coordinates  $u^a$  are 1-forms on U, the 2-index coefficients (5.33) of a linear connection are also 1-forms on the bundle space.

*Proof.* Writing (5.17) for the transformation  $\{\partial_I\} \mapsto \{e_I\}$ , with the frame  $\{e_I\}$  given by (5.32), we get (5.33) with

$$\Gamma^{a}_{b\mu} = ([B^{e}_{d}]^{-1})^{a}_{c} (\,{}^{\partial}\Gamma^{c}_{b\nu}B^{\nu}_{\mu} + B^{c}_{b\mu}),$$

where  ${}^{\partial}\Gamma^{a}_{b\nu}$  are the 3-index coefficients of  $\Delta^{h}$  in the frame adapted to the coordinates  $\{u^{I}\}$  (see (4.13)).

Let  $\{X_I\}$  and  $\{\tilde{X}_I\}$  be frames adapted to  $\{e_I\}$  and  $\{\tilde{e}_I\}$ , respectively, such that (cf. (5.32))

$$(\tilde{e}_{\mu}, \tilde{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} B^{\nu}_{\mu} \circ \pi & 0\\ (B^{b}_{c\mu} \circ \pi) \cdot u^{c} & B^{b}_{a} \circ \pi \end{bmatrix} , \qquad (5.34)$$

and  $\Delta^h$  admits 3-index coefficients in  $\{X_I\}$  and  $\{\tilde{X}_I\}$ , which means that  $\{e_I\}$ and  $\{\tilde{e}_I\}$  are obtainable from the frames  $\{\frac{\partial}{\partial u^I}\}$  and  $\{\frac{\partial}{\partial \tilde{u}^I}\}$ , respectively, for some bundle coordinate systems  $\{u^I\}$  and  $\{\tilde{u}^I\}$  via equations like (5.32) (with  $\tilde{e}_I$  for  $e_I$  and  $\tilde{\partial}_I$  for  $\partial_I$  in the letter case) in which the *B*'s need not be the same as in (5.34).<sup>4</sup> Then, due to (5.17) and (5.33), the 3-index coefficients  $\Gamma^a_{b\mu}$  and  $\tilde{\Gamma}^a_{b\mu}$ of  $\Delta^h$  in respectively  $\{X_I\}$  and  $\{\tilde{X}_I\}$  are connected by (see also footnote 4 and cf. (4.23))

$$\tilde{\Gamma}^{a}_{b\mu} = \left( [B^{e}_{f}]^{-1} \right)^{a}_{c} (\Gamma^{c}_{d\nu} B^{\nu}_{\mu} + B^{c}_{d\mu}) B^{d}_{b}.$$
(5.35)

**Exercise 5.4.** Prove that the transformation  $\{e_I\} \mapsto \{\tilde{e}_I\}$ , with the frame  $\{\tilde{e}_I\}$  given by (5.34), is the most general one that preserves the existence of 3-index coefficients of  $\Delta^h$  provided they exist in  $\{e_I\}$  in a sense that, if  $\{e_I\}$  is given by (5.32) (which leads to (5.33)) and  $\{\tilde{e}^I\}$  is given by (5.34), then there exist vector bundle coordinates  $\{\tilde{u}^I\}$  which generate  $\{\tilde{e}^I\}$  according to (5.32) with  $\tilde{e}_I$  for  $e_I$ ,  $\tilde{\partial}_I$  for  $\partial_I$  and some *B*'s, which leads to (5.33) with  $\tilde{\Gamma}$  for  $\Gamma$  and  $\tilde{u}$  for *u*.

Introducing the matrices  $\Gamma_{\mu} := [\Gamma_{b\mu}^a]_{a,b=n+1}^{n+r}$ ,  $\tilde{\Gamma}_{\mu} := [\tilde{\Gamma}_{b\mu}^a]_{a,b=n+1}^{n+r}$ ,  $B := [B_b^a]$ , and  $B_{\mu} := [B_{b\mu}^a]$ , we rewrite (5.35) as (cf. (4.23'))

$$\tilde{\Gamma}_{\mu} = B^{-1} \cdot (\Gamma_{\nu} B^{\nu}_{\mu} + B_{\mu}) \cdot B.$$
(5.35')

A little below (see the text after equation (5.37)), we shall prove that the compatibility of the developed formalism with the theory of covariant derivatives requires further restrictions on the general transformed frames (5.15) to the ones given by (5.34) with

$$B_{\mu} = \tilde{E}_{\mu}(B) \cdot B^{-1} = B^{\nu}_{\mu} E_{\nu}(B) \cdot B^{-1}, \qquad (5.36)$$

where  $\tilde{E}_{\mu} := \pi_*|_{\Delta^h}(\tilde{X}_{\mu}) = \pi_*|_{\Delta^h}((B^{\nu}_{\mu} \circ \pi)X_{\nu}) = B^{\nu}_{\mu}E_{\nu}$ . In this case, (5.35') reduces to (cf. (4.23'))

$$\tilde{\Gamma}_{\mu} = B^{\nu}_{\mu} B^{-1} \cdot (\Gamma_{\nu} \cdot B + E_{\nu}(B)) = B^{\nu}_{\mu} (B^{-1} \cdot \Gamma_{\nu} - E_{\nu}(B^{-1})) \cdot B.$$
(5.37)

<sup>&</sup>lt;sup>4</sup>Notice, from (5.34) follows that the vector fibre coordinate systems  $\{u^a\}$  and  $\{\tilde{u}^a\}$  are connected via  $u^a = (B_b^a \circ \pi) \cdot \tilde{u}^b$ .

## 5. General (co)frames

At last, a few words on the covariant derivative operators  $\nabla$  are in order. Without lost of generality, we define such an operator (4.33) via the equations (4.39). Suppose  $\{E_{\mu}\}$  is a basis for  $\mathcal{X}(M)$  and  $\{E_a\}$  is a one for  $\operatorname{Sec}^1(E, \pi, M)$ . Define the *components*  $\Gamma^a_{b\mu} \colon M \to \mathbb{K}$  of  $\nabla$  in the pair of frames ( $\{E_{\mu}\}, \{E_a\}$ ) by (cf. (4.40))

$$\nabla_{E_{\mu}}(E_b) = \Gamma^a_{b\mu} E_a. \tag{5.38}$$

Then (4.39) imply (cf. (4.36))

$$\nabla_F Y = F^{\mu} (E_{\mu} (Y^a) + \Gamma^a_{b\mu} Y^b) E_a$$

for all  $F = F^{\mu}E_{\mu} \in \mathcal{X}(M)$  and  $Y = Y^{a}E_{a} \in \text{Sec}^{1}(E, \pi, M)$ . A frame change  $(\{E_{\mu}\}, \{E_{a}\}) \mapsto (\{\tilde{E}_{\mu}\}, \{\tilde{E}_{a}\})$ , given via (5.31), entails

$$\Gamma^{a}_{b\mu} \mapsto \tilde{\Gamma}^{a}_{b\mu} = B^{\nu}_{\mu} ([B^{e}_{f}]^{-1})^{a}_{c} (\Gamma^{c}_{d\nu} B^{d}_{b} + E_{\nu} (B^{c}_{b})), \qquad (5.39)$$

as a result of (5.38). In a more compact matrix form, the last result reads

$$\tilde{\Gamma}_{\mu} = B^{\nu}_{\mu} B^{-1} \cdot (\Gamma_{\nu} \cdot B + E_{\nu}(B))$$
(5.39')

with  $\Gamma_{\mu} := [\Gamma^a_{b\mu}], \ \tilde{\Gamma}_{\mu} := [\tilde{\Gamma}^a_{b\mu}], \text{ and } B := [B^a_b].$ 

Thus, if we identify the 3-index coefficients of  $\Delta^h$ , defined by (5.33), with the components of  $\nabla$ , defined by (5.38),<sup>5</sup> then the quantities (5.35') and (5.39') must coincide, which immediately leads to the equality (5.36). Therefore

$$(e_{\mu}, e_{a}) \mapsto (\tilde{e}_{\mu}, \tilde{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} B_{\mu}^{\nu} \circ \pi & 0\\ ((B_{\mu}^{\nu} E_{\nu} (B_{d}^{b}) (B^{-1})_{c}^{d}) \circ \pi) u^{c} & B_{a}^{b} \circ \pi \end{bmatrix} \Big|_{B = [B_{a}^{b}]}$$
(5.40)

is the most general transformation between frames in T(E) such that the frames adapted to them are compatible with the linear connection and the covariant derivative corresponding to it. In particular, such are all frames  $\{\frac{\partial}{\partial u^I}\}$  in T(E)induced by vector bundle coordinates  $\{u^I\}$  on E – see (4.21) and (3.1)–(3.3); the rest members of the class of frames mentioned are obtained from them via (5.40) with  $e_I = \frac{\partial}{\partial u^I}$  and non-degenerate matrix-valued functions  $[B^{\nu}_{\mu}]$  and B.

If  $\{X_I\}$  (resp.  $\{\tilde{X}_I\}$ ) is the frame adapted to a frame  $\{e_I\}$  (resp.  $\{\tilde{e}_I\}$ ), then the change  $\{e_I\} \mapsto \{\tilde{e}_I\}$ , given by (5.40), entails  $\{X_I\} \mapsto \{\tilde{X}_I\}$  with  $\{\tilde{X}_I\}$  given by (5.30) (see (5.15) and (5.16)). Since the last transformation is tantamount to the change

$$(\{E_{\mu}\}, \{E_{a}\}) \mapsto (\{\tilde{E}_{\mu}\}, \{\tilde{E}_{a}\})$$
 (5.41)

of the basis of  $\mathcal{X}(M) \times \text{Sec}(E, \pi, M)$  corresponding to  $\{X_I\}$  via the isomorphism (5.28) (see (5.26), (5.30), and (5.31)), we can say that the transition (5.41)

<sup>&</sup>lt;sup>5</sup>Such an identification is justified by the definition of  $\nabla$  via the parallel transport assigned to  $\Delta^h$  (see Proposition 4.4) or via a projection, generated by  $\Delta^h$ , of a suitable Lie derivative on  $\mathfrak{X}(E)$  (see Definition 4.2).

induces the change (5.39) of the 3-index coefficients of the connection  $\Delta^h$ . Exactly the same is the situation one meets in the literature [7, 11, 23] when covariant derivatives are considered (and identified with connections).

Regardless that the change (5.40) of the frames in T(E) looks quite special, it is the most general one that, through (5.16) and (5.26), is equivalent to an arbitrary change (5.41) of a basis in  $\mathcal{X}(M) \times \text{Sec}(E, \pi, M)$ , i.e., of a pair of frames in T(M) and E.

We would like to remark that, in the general case, equation (4.43) also holds with  $F = F^{\mu}E_{\mu}$ ,  $G = G^{\mu}E_{\mu}$ , and

$$(R(E_{\mu}, E_{\nu}))(E_b) = R^a_{b\mu\nu} E_a, \tag{5.42}$$

so that

$$R^a_{b\mu\nu} = E_\mu(\Gamma^a_{b\nu}) - E_\nu(\Gamma^a_{b\mu}) - \Gamma^c_{b\mu}\Gamma^a_{c\nu} + \Gamma^c_{b\nu}\Gamma^a_{c\mu} - \Gamma^a_{b\lambda}C^\lambda_{\mu\nu}, \qquad (5.43)$$

where the functions  $C^{\lambda}_{\mu\nu}$  define the anholonomicity object of  $\{E_{\mu}\}$  via  $[E_{\mu}, E_{\nu}]_{-} =: C^{\lambda}_{\mu\nu}E_{\lambda}$ .

The above results, concerning linear connections on vector bundles, can be generalized for affine connections on vector bundles. For instance, the analogue of Propositions 5.4 reads.

**Proposition 5.5.** Let  $\Delta^h$  be an affine connection on a vector bundle  $(E, \pi, M)$  and  $\{X_\mu\}$  be the frame adapted for  $\Delta^h$  to a frame  $\{e_I\}$  such that  $\{e_a\}$  is a basis for  $\Delta^v$  and

$$(e_{\mu}, e_{a})|_{U} = (\partial_{\nu}, \partial_{b}) \cdot \begin{bmatrix} B^{\nu}_{\mu} \circ \pi & 0\\ (B^{b}_{c\mu} \circ \pi) \cdot u^{c} & B^{b}_{a} \circ \pi \end{bmatrix}$$
  
$$= \left( (B^{\nu}_{\mu} \circ \pi) \partial_{\nu} + ((B^{b}_{c\mu} \circ \pi) \cdot u^{c}) \partial_{b}, (B^{b}_{a} \circ \pi) \partial_{b} \right),$$
  
(5.44)

where  $\partial_I := \frac{\partial}{\partial u^I}$  for some local bundle coordinate system  $\{u^I\} = \{u^{\mu} = x^{\mu} \circ \pi, u^b = E^b\}$  on  $U \subseteq E$ ,  $[B^{\nu}_{\mu}]$  and  $[B^b_a]$  are non-degenerate matrix-valued functions on U, and  $B^b_{c\mu} : U \to \mathbb{K}$ . Then the 2-index coefficients  $\Gamma^a_{\mu}$  of  $\Delta^h$  in  $\{X_I\}$  have the representation (cf. (4.46))

$$\Gamma^a_\mu = -(\Gamma^a_{b\mu} \circ \pi) \cdot u^b + G^a_\mu \circ \pi \tag{5.45}$$

on U for some functions  $\Gamma^a_{b\mu}, G^a_{\mu} \colon U \to \mathbb{K}$ .

Remark 5.3. The representation (5.45) is not valid for frames more general than the ones given by (5.44). Precisely, equation (5.45) is valid if and only if (5.44) holds for some local coordinate system  $\{u^I\}$  on U – see (5.17).

*Proof.* Writing the equation (5.17) for the transformation  $\{\partial_I\} \mapsto \{e_I\}$ , with  $\{e_I\}$  given by (5.44), we get (5.45) with

$$\Gamma^{a}_{b\mu} = ([B^{e}_{d}]^{-1})^{a}_{c} (\,^{\partial}\Gamma^{c}_{b\nu}B^{\nu}_{\mu} + B^{c}_{b\mu}) \quad G^{a}_{\mu} = ([B^{e}_{d}]^{-1})^{a}_{b} \,^{\partial}G^{b}_{\nu}B^{\nu}_{\mu}$$

where  ${}^{\partial}\Gamma^{a}_{b\nu}$  and  ${}^{\partial}G^{b}_{\nu}$  are defined via the 2-index coefficients  ${}^{\partial}\Gamma^{a}_{\mu}$  of  $\Delta^{h}$  in the frame adapted to the coordinates  $\{u^{I}\}$  via  ${}^{\partial}\Gamma^{a}_{\mu} = -({}^{\partial}\Gamma^{a}_{b\mu}\circ\pi)\cdot E^{b} + {}^{\partial}G^{a}_{\mu}\circ\pi$  (see Theorem 4.2).

Let  $\{X_I\}$  and  $\{\tilde{X}_I\}$  be frames adapted to  $\{e_I\}$  and  $\{\tilde{e}_I\}$ , respectively, with (cf. (5.44))

$$(\tilde{e}_{\mu}, \tilde{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} B^{\nu}_{\mu} \circ \pi & 0\\ (B^{b}_{c\mu} \circ \pi) \cdot u^{c} & B^{b}_{a} \circ \pi \end{bmatrix} , \qquad (5.46)$$

in which (5.45) holds for  $\Delta^h$ . Then, due to (5.17) and (5.45), the pairs  $(\Gamma^a_{b\mu}, G^a_{\mu})$ and  $(\tilde{\Gamma}^a_{b\mu}, \tilde{G}^a_{\mu})$  for  $\Delta^h$  in respectively  $\{X_I\}$  and  $\{\tilde{X}_I\}$  are connected by (cf. (4.47) and (4.48))

$$\tilde{\Gamma}^{a}_{b\mu} = \left( [B^{e}_{f}]^{-1} \right)^{a}_{c} (\Gamma^{c}_{d\nu} B^{\nu}_{\mu} + B^{c}_{d\mu}) B^{d}_{b}$$
(5.47a)

$$\tilde{G}^{a}_{\mu} = \left( [B^{e}_{f}]^{-1} \right)^{a}_{b} G^{b}_{\nu} B^{\nu}_{\mu}.$$
(5.47b)

**Exercise 5.5.** Prove that the transformation  $\{e_I\} \mapsto \{\tilde{e}_I\}$ , with the frame  $\{\tilde{e}_I\}$  given by (5.46), is the most general one that preserves the existence of the relation (5.45) for  $\Delta^h$  provided it holds in  $\{e_I\}$ .

Further one can repeat *mutatis mutandis* the text after Exercise 5.4 to the paragraph containing equation (5.41) including.

# 6. Normal frames

In the theory of linear connections on a manifold, the normal frames are defined as frames in the tangent bundle space in which the connections' (3-index) coefficients vanish on some subset of the manifold (see Definition I.5.1 on page 37). The definition of normal frames for a connection on a vector bundle is practically the same, the only difference being that these frames are in the bundle space, not in the tangent bundle space over the base space (see Sections IV.8 and IV.14.4). The present section is devoted to the introduction of normal frames for general connections on fibre bundles and some their properties.

# 6.1. The general case

To save some space and for brevity, in what follows we shall not indicate explicitly that the frames  $\{e_I\} = \{e_\mu, e_a\}$ , with respect to which the adapted frames are defined, are such that  $\{e_a\}$  is a (local) basis for the vertical distribution  $\Delta^v$  on the bundle considered.

**Definition 6.1.** Given a connection  $\Delta^h$  on a bundle  $(E, \pi, M)$  and a subset  $W \subseteq E$ . A frame  $\{X_I\}$  in T(E) adapted to a frame  $\{e_I\}$  in T(E) and defined over an open subset V of E containing or equal to  $W, V \supseteq W$ , is called normal for  $\Delta^h$  over/on W (relative to  $\{e_I\}$ ) if all (2-index) coefficients  $\Gamma^a_{\mu}$  of  $\Delta^h$  vanish in it everywhere on W. Respectively,  $\{X_I\}$  is normal for  $\Delta^h$  along a mapping  $g: Q \to E, Q \neq \emptyset$ , if  $\{X_I\}$  is normal for  $\Delta^h$  over the set g(Q).

Let  $\{X_I\}$  be the frame in T(E) adapted to a frame  $\{e_I\}$  in T(E) over an open subset  $V \subseteq E$ . Then the frame  $\{\tilde{X}_I\}$  in T(E) adapted to a frame  $\{\tilde{e}_I\}$ , given by (5.15), in T(E) over the same subset V is normal for  $\Delta^h$  over  $W \subseteq V$  if and only if

$$(A^{\nu}_{\mu}\Gamma^{b}_{\nu} - A^{b}_{\mu})|_{W} = 0, \qquad (6.1)$$

due to (5.16) and (5.17). Since  $\Gamma^b_{\mu}$  depend only on  $\Delta^h$  and  $\{e_I\}$ , the existence of solutions of (6.1), relative to  $A^{\nu}_{\mu}$  and  $A^b_{\mu}$ , and their properties are completely responsible for the existence and the properties of frames normal for  $\Delta^h$  over W. For that reason, we call (6.1) the (system of) equation(s) of the normal frames for  $\Delta^h$  over W or simply the normal frame (system of) equation(s) (for  $\Delta^h$  over W).

In the most general case, when no additional restrictions on the frames considered are imposed, the normal frame equation (6.1) is a system of *nr linear algebraic equations* for  $nr + n^2$  variables and, consequently, it has a solution depending on  $n^2$  independent parameters. In particular, if we choose the functions  $A^{\nu}_{\mu}: W \to \mathbb{K}$  (with det $[A^{\nu}_{\mu}] \neq 0, \infty$ ) as such parameters, we can write the general solution of (6.1) as

$$(\{A^{\nu}_{\mu}\},\{A^{b}_{\mu}\})|_{W} = (\{A^{\nu}_{\mu}\},\{\Gamma^{b}_{\nu}A^{\nu}_{\mu}\})|_{W}.$$
(6.2)

It should be noted, equation (6.1) or its general solution (6.2) defines the frame  $\{\tilde{e}_I\}$  and the frame  $\{\tilde{X}_I\}$  adapted to  $\{\tilde{e}_I\}$  only on W and leaves them completely arbitrary on  $V \setminus W$ , if it is not empty.

**Proposition 6.1.** Let  $\{X_I\}$  be the frame adapted to a frame  $\{e_I\}$  in  $T(V) \subseteq T(E)$ defined over an open set  $V \subseteq E$  and  $\Gamma^a_\mu$  be the coefficients of a connection  $\Delta^h$ in  $\{X_I\}$ . Then all frames  $\{\tilde{X}_I\}$ , normal on  $W \subseteq V$  for the connection  $\Delta^h$ , are adapted to frames  $\{\tilde{e}_I\}$  given on W by

$$\tilde{e}_{\mu}|_{W} = (A^{\nu}_{\mu}(e_{\nu} + \Gamma^{b}_{\nu}e_{b}))|_{W} \quad \tilde{e}_{a}|_{W} = (A^{b}_{a}e_{b})|_{W}.$$
(6.3)

where  $[A^{\nu}_{\mu}]$  and  $[A^{b}_{a}]$  are non-degenerate matrix-valued functions on V which are constant on the fibres of  $(E, \pi, M)$ . Moreover, the frame  $\{\tilde{X}_{I}\}$  adapted on V to  $\{\tilde{e}_{I}\}$ , given by (6.3) (and hence normal on W), is such that

$$\tilde{X}_{\mu}|_{W} = (A^{\nu}_{\mu}X_{\nu})|_{W} = \tilde{e}_{\mu}|_{W} \quad \tilde{X}_{a}|_{W} = (A^{b}_{a}X_{b})|_{W} = \tilde{e}_{a}|_{W}, \tag{6.4}$$

*Proof.* Apply (5.16), (5.15), and (5.13a) for the choice (6.2).

The equations (6.4) are not accidental as it is stated by the following assertion.

**Proposition 6.2.** The frame  $\{\tilde{X}_I\}$  in T(E) adapted to a frame  $\{\tilde{e}_I\}$  in T(E) and defined over an open set  $V \subseteq E$  is normal on  $W \subseteq V$  if and only if on W is fulfilled

$$\tilde{X}_I|_W = \tilde{e}_I|_W. \tag{6.5}$$

*Proof.* Apply (5.13a) or (5.16) and Definition 6.1.

Thus one can equivalently define the normal frames as adapted frames that coincide on some set with the frames they are adapted to or as frames (in the tangent bundles space over the bundle space) that coincide on some set with the frames adapted to them.

Since any specialized frame is adapted to itself (see Definition 5.1 and (5.12), with  $D_I^J = \delta_I^J$ ), the sets of normal, specialized, and adapted frames are identical.

As we see from Proposition 6.1, which gives a complete description of the normal frames, the theory of normal frames in the most general setting is trivial. It becomes more interesting and richer if the class of frames  $\{e_I\}$ , with respect to which are defined the adapted frames, is restricted in one or other way. To the theory of normal frames, adapted to such restricted classes of frames in T(E), is devoted the rest of this section.

## 6.2. Normal frames adapted to holonomic frames

The class of holonomic frames induced by local coordinates on E (see Subsection 3.2) is the most natural class of frames in T(E) relative to which the adapted, in particular normal, frames are defined. To specify the consideration of the previous subsection to normal frames adapted to local coordinates on E, we set  $e_I = \frac{\partial}{\partial u^I}$ and  $\tilde{e}_I = \frac{\partial}{\partial \tilde{u}^I}$ , where  $\{u^I\}$  and  $\{\tilde{u}^I\}$  are local coordinate systems on E whose domains have a non-empty intersection V and  $W \subseteq V$ . Then the matrix  $[A_I^J]$ in (6.1) is given by (3.3) (as  $\{e_I\} \mapsto \{\tilde{e}_I\}$  reduces to (3.2a)), so that the normal frame equation (6.1) reduces to the normal coordinates equation (see also (3.32))

$$\left(\frac{\partial \tilde{u}^a}{\partial u^b}\Gamma^b_\mu + \frac{\partial \tilde{u}^a}{\partial u^\mu}\right)\Big|_W = 0, \tag{6.6}$$

due to (3.1), which is a first-order system of *nr linear partial differential equations* on W relative to the r unknown functions  $\{\tilde{u}^{n+1}, \ldots, \tilde{u}^{n+r}\}$ .

Since the connection  $\Delta^h$  is supposed given and fixed, such are its coefficients  $\Gamma^b_{\mu}$  in  $\{\frac{\partial}{\partial u^I}\}$ . Therefore the existence, uniqueness and other properties of the solutions of (6.6) strongly depend on the set W (which is in the intersection of the domains of the local coordinates  $\{u^I\}$  and  $\{\tilde{u}^I\}$  on E).

**Proposition 6.3.** If the normal frame equation (6.6) has solutions, then all frames  $\{\tilde{X}_I\}$  normal on  $W \subseteq E$  and adapted to local coordinates, defined on an open set  $V \subseteq E$  such that  $V \supseteq W$ , are described by

$$\tilde{X}_{\mu}|_{W} = (A^{\nu}_{\mu}X_{\nu})|_{W} = \frac{\partial}{\partial \tilde{u}^{\mu}}\Big|_{W} \quad \tilde{X}_{a}|_{W} = (A^{b}_{a}X_{b})|_{W} = \frac{\partial}{\partial \tilde{u}^{a}}\Big|_{W}, \tag{6.7}$$

where  $\{X_I\}$  is the frame adapted to some arbitrarily fixed local coordinates  $\{u^I\}$ , defined on an open set containing or equal to V,  $\{\tilde{u}^I\}$  are local coordinates with domain V and such that  $\tilde{u}^a$  are solutions of (6.6), and  $A_I^J = \frac{\partial u^J}{\partial \tilde{u}^I}$  on the intersection of the domains of  $\{u^I\}$  and  $\{\tilde{u}^I\}$ .

*Proof.* Apply Proposition 6.1 for  $e_I = \frac{\partial}{\partial u^I}$  and  $\tilde{e}_I = \frac{\partial}{\partial \tilde{u}^I}$  and then use (3.2a) and (3.3).

This simple result gives a complete description of all normal frames, if any, adapted to (local) holonomic frames. It should be understood clearly, normal on W is the frame  $\{\tilde{X}_I\}$ , adapted to  $\{\frac{\partial}{\partial \tilde{u}^I}\}$  and coinciding with it on W, but not the frame  $\{\frac{\partial}{\partial \tilde{u}^I}\}$ ; in particular, the frame  $\{\frac{\partial}{\partial \tilde{u}^I}\}$  is holonomic while the frame  $\{\tilde{X}_I\}$  need not to be holonomic, even on W, if the connection considered does not satisfies some additional conditions, like the vanishment of its curvature on W.

Consider now briefly the existence problem for the solutions of (6.6). To begin with, we emphasize that in (6.6) enter only the fibre coordinates  $\tilde{u}^a$ , so that it leaves the basic ones  $\tilde{u}^{\mu}$  completely arbitrary.

**Proposition 6.4.** If E is of class  $C^2$ ,  $p \in E$  is fixed and  $W = \{p\}$ , then the general solution of (6.6) is

$$\tilde{u}^{a}(q) = g^{a} + g^{a}_{b} \{ -\Gamma^{b}_{\mu}(p)(q^{\mu} - p^{\mu}) + (q^{b} - p^{b}) \} + f^{a}_{IJ}(q)(q^{I} - p^{I})(q^{J} - p^{J}),$$
(6.8)

where  $g^a$  and  $g^a_b$  are constants in  $\mathbb{K} = \mathbb{R}, \mathbb{C}$ ,  $\det[g^a_b] \neq 0, \infty$ , the point q is in the domain V of  $\{u^I\}$ ,  $q^I := u^I(q)$ ,  $p^I := u^I(p)$ , and  $f^a_{IJ}$  are  $C^2$  functions on V such that they and their first partial derivatives are bounded when  $q^I \to p^I$ .

Proof. Expand  $\tilde{u}^a(q) = f^a(u^1(q), \dots, u^n(q), \dots, u^{n+r}(q)) = f^a(q^1, \dots, q^{n+r})$  into a Taylor first order polynomial with remainder term quadratic in  $(q^I - p^I)$  and insert the result into (6.6). In this way one gets (6.8) with  $g^a = \tilde{u}^a(p)$  and  $g^a_b = \frac{\partial \tilde{u}^a}{\partial u^b}\Big|_p$ .

Now we would like to investigate the existence of solutions of (6.6) along paths  $\beta: J \to E$ , i.e., for  $W = \beta(J)$ . The main result is formulated below as Proposition 6.5 and for its proof we shall essentially use Lemma II.3.1.

**Proposition 6.5.** Let  $\Delta^h$  be a  $C^1$  connection on a real  $C^3$  bundle  $(E, \pi, M)$ ,  $n = \dim M \ge 1$ , and  $r = \dim \pi^{-1}(x) \ge 1$  for  $x \in M$ . Let  $\beta : J \to E$  be an injective regular  $C^1$  path such that its tangent vector  $\dot{\beta}(s)$  at s is not a vertical vector for all  $s \in J$ ,  $\dot{\beta}(s) \notin \Delta^v_{\beta(s)}$ ; in particular, the path  $\beta$  can be horizontal, i.e.,  $\dot{\beta}(s) \in \Delta^h_{\beta(s)}$  for all  $s \in J$ , but generally the vector  $\dot{\beta}(s)$  can have also and a vertical component for some or all  $s \in J$ . Then, for every  $s_0 \in J$ , there exist a neighborhood  $W_1$  of the point  $\beta(s_0)$  in E and bundle coordinates  $\tilde{u}^I$  on  $W_1$  which are solutions of (6.6) for  $W = W_1 \cap \beta(J) = \beta(J_1)$ , with  $J_1 := \{s \in J : \beta(s) \in W_1\}$ , i.e., along the restricted path  $\beta|_{J_1}$ . All such bundle coordinates  $\tilde{u}^I$  are given via the equation (6.9) below.

*Proof.* Consider the chart  $(W_1, u)$  with  $W_1 \ni \beta(s_0)$  provided by Lemma II.3.1 for E and  $\beta$  instead of M and  $\gamma$ , respectively. For any  $p \in W_1$ , there is a unique  $(s, \mathbf{t}) \in J_1 \times \mathbb{R}^{\dim_{\mathbb{R}} E-1}$  such that  $p = u^{-1}(s, \mathbf{t})$ , i.e., in the coordinate system  $\{u^I\}$  associated to u, the coordinates of p are  $u^1(p) = s$  and  $u^I(p) = t^I \in \mathbb{R}$  for  $I \ge 2$ . Besides, we have  $u(\beta(s)) = (s, \mathbf{t}_0)$  for all  $s \in J_1$  and some fixed  $\mathbf{t}_0 \in \mathbb{R}^{\dim_{\mathbb{R}} E-1}$ .

Since  $\dot{\beta}(s)$  is not a vertical vector for all  $s \in J$ , the coordinate system  $\{u^I\}$  can be chosen to be *bundle* coordinate system. For the purpose, in the proof of Lemma II.3.1 one must choose  $\{y^I\}$  as bundle coordinate system and to take for  $\dot{\beta}_y^1(s_0)$  any non-vanishing component between  $\dot{\beta}_y^1(s_0), \ldots, \dot{\beta}_y^n(s_0)$ , viz. if  $\dot{\beta}_y^1(s_0) \neq 0$  the proof goes as it is written and, if  $\dot{\beta}_y^1(s_0) = 0$ , choose some  $\mu_0$  such that  $\dot{\beta}_y^{\mu_0}(s_0) \neq 0$  and make, e.g., the change  $\dot{\beta}_y^1(s_0) \leftrightarrow \dot{\beta}_y^{\mu_0}(s_0)$ . This, together with (II.3.12), with  $u^I$  for  $x^k$ , ensures that  $\{y^I\} \mapsto \{u^I\}$  is an admissible change, so that  $\{u^I\}$  are bundle coordinates if the initial coordinates  $\{y^I\}$  are such ones.

Let  $\{u^I\}$  be so constructed bundle coordinates and  $\eta := u^{-1}$ , so that  $\beta(s) = \eta(s, t_0)$ . Expanding  $\tilde{u}^a(\eta(s, t))$  into a first-order Taylor polynomial at the point  $t_0 \in K$ , we find the general solution of (6.6), with  $W = \beta(J_1) = W_1 \cap \beta(J)$ , in the form

$$\begin{split} \tilde{u}^{a}(\eta(s, t)) &= B^{a}(s) \\ &+ B^{a}_{b}(s) \{ -\Gamma^{b}_{\mu}(\beta(s)) [u^{\mu}(\eta(s, t)) - u^{\mu}(\beta(s))] + [u^{b}(\eta(s, t)) - u^{b}(\beta(s))] \} \\ &+ B^{a}_{IJ}(s, t; \eta) [u^{I}(\eta(s, t)) - u^{I}(\beta(s))] [u^{J}(\eta(s, t)) - u^{J}(\beta(s))], \end{split}$$
(6.9)

where  $B^a, B^a_b: J_1 \to \mathbb{K} = \mathbb{R}$ ,  $\det[B^a_b] \neq 0, \infty$ , and the  $C^1$  functions  $B^a_{IJ}$  and their first partial derivatives are bounded when  $\mathbf{t} \to \mathbf{t}_0$ . (Notice, the terms with  $\mu = 1$ and/or I = 1 and/or J = 1 do not contribute in (6.9) as  $u^1(\eta(s, \mathbf{t})) \equiv s$  and, besides, the functions  $B^a_{IJ}$  can be taken symmetric in I and  $J, B^a_{IJ} = B^a_{JI}$ .)  $\Box$ 

Remark 6.1. If there is  $s_0 \in J$  for which  $\dot{\beta}(s_0)$  is a vertical vector,  $\dot{\beta}(s_0) \in \Delta_{\beta(s_0)}^v$ , then Proposition 6.5 remains true with the only correction that the coordinates  $\tilde{u}^I$ will not be bundle coordinates. If this is the case, the constructed coordinates  $\tilde{u}^I$ will be solutions of (6.6), but we cannot assert that they are bundle coordinates which are (locally) normal along  $\beta$  in a neighborhood of the point  $\beta(s_0)$ .

Proposition 6.5 can be generalized by requiring  $\beta$  to be locally injective instead of injective, i.e., for each  $s \in J$  to exist a subinterval  $J_s \subseteq J$  such that  $J_s \ni s$  and the restricted path  $\beta|_{J_s}$  to be injective. Besides, if one needs a version of the above results for complex bundles, they should be considered as real ones (with doubled dimension of the manifolds) for which are applicable the above considerations.

**Corollary 6.1.** At any arbitrarily fixed point in E and/or along a given injective regular  $C^1$  path in E, whose tangent vector is not vertical, there exist (possibly local, in the latter case) normal frames.

*Proof.* See Definition 6.1, Propositions 6.4 and 6.5, and equation (6.6). If the path is not contained in a single coordinate neighborhood, one should cover its image in the bundle space with such neighborhoods and, then, to apply Proposition 6.5; in the intersection of the coordinate domains, the uniqueness (and, possibly, continuity or differentiability) of the normal frames may be lost.  $\Box$ 

**Definition 6.2.** Local bundle coordinates  $\{\tilde{u}^I\}$ , defined on an open set  $V \subseteq E$ , will be called *normal* on  $W \subseteq V$  for a connection  $\Delta^h$  if the frame  $\{\tilde{X}_I\}$  in T(E) adapted to  $\{\frac{\partial}{\partial \tilde{u}^I}\}$  over V is normal for  $\Delta^h$  on W.

Corollary 6.1 implies the existence of coordinates normal at a given point or (locally) along a given injective path whose tangent vector is not vertical; in particular, there exist coordinates normal along an injective horizontal path. However, normal coordinates generally do not exist on more general subsets of the bundle space E. A criterion for existence of coordinates normal on sufficiently general subsets  $W \subseteq E$ , e.g on 'horizontal' submanifolds, is given by Theorem 7.1 in Section 7 below. In particular, we have the following result.

**Proposition 6.6.** If  $\Delta^h$  is a  $C^1$  connection, W is an open set in E, and normal frames for  $\Delta^h$  on W exist, then there are holonomic such frames if  $\Delta^h$  is flat on W. Said otherwise, the system of equations (6.6) may admit solutions on an open set W if

$$R^a_{\mu\nu}|_W = 0 \tag{6.10}$$

where  $R^a_{\mu\nu}$  are the fibre components of the curvature of  $\Delta^h$  in some frame on E, defined by (3.37a) in the frame  $\{X_I\}$  adapted to a holonomic one.

*Proof.* Since the normal frames are also adapted ones (see Definition 6.1) and  $\Gamma^a_{\mu}|_W = 0$  in a frame normal on W, the statement is a corollary from Proposition 3.4.

Remark 6.2. However, in the general case the flatness of a connection on an open set is only a necessary, but not sufficient, condition for the existence of coordinates normal on that set – see Theorem 7.1 in Section 7 below. Exceptions are the linear connections on vector bundles – see Remark 7.1 in Section 7.

**Exercise 6.1.** Show that part of the integrability conditions for (6.6) for an open set W are

$$0 = \frac{\partial^2 \tilde{u}^a}{\partial u^\nu \partial u^\mu} - \frac{\partial^2 \tilde{u}^a}{\partial u^\mu \partial u^\nu} \equiv \frac{\partial \tilde{u}^a}{\partial u^b} R^b_{\mu\nu}$$
(6.11)

from where Proposition 6.6 immediately follows. However, the flatness of the connection on W generally does not imply the rest of the integrability conditions, viz.  $\frac{\partial^2 \tilde{u}^a}{\partial u^b \partial u^\mu} - \frac{\partial^2 \tilde{u}^a}{\partial u^\mu \partial u^b} = 0 \text{ and } \frac{\partial^2 \tilde{u}^a}{\partial u^b \partial u^c} - \frac{\partial^2 \tilde{u}^a}{\partial u^c \partial u^b} = 0.$ 

The combination of Propositions 6.6 and 6.2 implies the non-existence of coordinates normal on an open set for non-flat (non-integrable) connections.

# 6.3. Normal frames on vector bundles

The normal frames in vector bundles for covariant derivative operators (linear connections), other derivations, and linear transports along paths were investigated in Chapter IV. The goal of the present subsection is to be made a link between them and the general theory of Subsection 6.1.

Consider a linear connection  $\Delta^h$  on a vector bundle  $(E, \pi, M)$  (see Definition 4.1). Let the frame  $\{e_I\}$  in T(E) be given by (5.32) and  $\{X_I\}$  be the frame adapted to  $\{e_I\}$  for  $\Delta^h$ . Then, by Proposition 5.4, the 2- and 3-index coefficients of  $\Delta^h$  are connected via (5.33) in which  $\{u^a\}$  is vector fibre coordinate system.

**Proposition 6.7.** A frame  $\{X_I\}$  is normal on  $W \subseteq E$  for a linear connection  $\Delta^h$  if and only if in it vanish the 3-index coefficients of  $\Delta^h$  on  $\pi(W) \subseteq M$ ,

$$\Gamma^a_\mu|_W = 0 \iff \Gamma^a_{b\mu}|_{\pi(W)} = 0. \tag{6.12}$$

*Proof.* Since  $u^{n+1}, \ldots, u^{n+r}$  are 1-forms which are linearly independent for all  $p \in W$ , the assertion follows from equation (5.33).

Combining Proposition 6.7 with (5.35), we see that the normal frame equation (6.1) in vector bundle is equivalent to

$$(B^{\nu}_{\mu}\Gamma^{a}_{b\nu} + B^{a}_{b\mu})|_{\pi(W)} = 0 \tag{6.13}$$

or to its matrix variant (see also (5.35');  $\Gamma_{\nu} := [\Gamma^a_{b\nu}], B_{\mu} := [B^a_{b\mu}])$ 

$$(B^{\nu}_{\mu}\Gamma_{\nu} + B_{\mu})|_{\pi(W)} = 0. \tag{6.13'}$$

Taking into account (6.13) and (5.34), we can assert that the frame  $\{\tilde{X}_I\}$  adapted to the frame

$$(\tilde{e}_{\mu}, \tilde{e}_{a}) = (e_{\nu}, e_{b}) \cdot \begin{bmatrix} B^{\nu}_{\mu} \circ \pi & 0\\ -((B^{\lambda}_{\mu} \Gamma^{b}_{c\lambda}) \circ \pi) \cdot u^{c} & B^{b}_{a} \circ \pi \end{bmatrix} , \qquad (6.14)$$

where  $[B^{\nu}_{\mu}]$  and  $[B^{a}_{b}]$  are non-degenerate matrix-valued functions, is normal on W for  $\Delta^{h}$  and hence  $\tilde{X}_{I} = \tilde{e}_{I}$ , by virtue of Proposition 6.2. Recall (see (5.15), (5.16), and (5.31)), the change  $\{e_{I}\} \mapsto \{\tilde{e}_{I}\}$ , given by (6.14), entails  $\{X_{I}\} \mapsto \{\tilde{X}_{I}\}$ , where

$$\tilde{X}_{\mu} = (B^{\nu}_{\mu} \circ \pi) X_{\nu} \quad \tilde{X}_{a} = (B^{b}_{a} \circ \pi) X_{b}, \qquad (6.15)$$

which is equivalent to  $\{E_I\} \mapsto \{\tilde{E}_I\}$  with

$$\tilde{E}_{\mu} = B^{\nu}_{\mu} E_{\nu} \quad \tilde{E}_a = B^b_a E_b. \tag{6.16}$$

Here (see (5.26))  $\{E_{\mu} = \pi_*|_{\Delta^h}(X_{\mu})\}$  is a frame in T(M) and  $\{E_a = v^{-1}(X_a)\}$  is a frame in E.

Thus, if additional restriction are not imposed, the theory of normal frames in vector bundles is rather trivial, which reflects a similar situation in general bundles, considered in Subsection 6.1. However, the really interesting and sensible case is when one considers frames compatible with the covariant derivatives (see Section 5, the paragraphs including equations (5.38)–(5.40)). As we know (see (5.36)), it corresponds to arbitrary non-degenerate matrix-valued functions  $[B^{\nu}_{\mu}]$  and  $B = [B^{\nu}_{b\mu}]$  and a matrix-valued functions  $B_{\mu} = [B^{\nu}_{b\mu}]$  given by

$$B_{\mu} = \tilde{E}_{\mu}(B) \cdot B^{-1} = B^{\nu}_{\mu} E_{\nu}(B) \cdot B^{-1}.$$
(6.17)

In particular, such are all holonomic frames in T(E), locally induced by local coordinates on E, as discussed in Section 4. Now the normal frames equation (6.13) (or (6.1)) reduces to

$$(\Gamma_{\mu} \cdot B + E_{\mu}(B))|_{\pi(W)} = 0 \tag{6.18}$$

which is exactly the equation (IV.14.54) for N = M,  $g = id_M$  and  $U = \pi(W)$ , i.e., the normal frame equation for a linear connection in a vector bundle. This equation leaves the frame  $\{\tilde{E}_{\mu} = \pi_*|_{\Delta^h}(X_{\mu})\}$  in T(M) completely arbitrary and imposes restriction on the frame  $\{\tilde{E}_a = v^{-1}(X_a) = B_a^b E_b\}$  in E. This conclusion justifies the following definition.

**Definition 6.3.** Given a *linear* connection  $\Delta^h$  on a *vector* bundle  $(E, \pi, M)$  and a subset  $W_M \subseteq M$ . A frame  $\{E_a\}$  in E, defined over an open set  $V_M$  containing  $W_M$  or equal to it,  $V_M \supseteq W_M$ , is called normal for  $\Delta^h$  over/on  $W_M$  if their is a frame  $\{X_I\}$  in T(E), defined over an open set  $V_E \subseteq E$ , which is normal for  $\Delta^h$  over a subset  $W_E \subseteq E$  and such that  $\pi(W_E) = W_M$ ,  $\pi(V_E) = V_M$ , and  $E_a = v^{-1}(X_a)$ , with the mapping v defined by (4.3). Respectively,  $\{E_a\}$  is normal for  $\Delta^h$  over  $g(Q_M)$ .

Taking into account Definition 6.1, we see that the so-defined normal frames in the bundle space E are just the ones used in the theory of frames normal for linear connections in vector bundles – see Definition IV.14.15 for N = M and  $g = id_M$ .

It is quite clear, to any frame  $\{X_I\}$  in T(E) normal over  $W \subseteq E$ , there corresponds a unique frame  $\{E_a = v^{-1}(X_a)\}$  in E normal over  $\pi(W) \subseteq M$ . But, to a frame  $\{E_a\}$  in E normal over  $\pi(W)$ , there correspond infinitely many frames  $\{X_I\} = \{(\pi_*|_{\Delta^h})^{-1}(E_\mu), v(E_a)\}$  in T(E) normal over W, where  $\{E_\mu\}$ is an arbitrary frame in T(M) over  $\pi(W)$ . Thus the problems of existence and (un)uniqueness of normal frames in T(E) is completely reduced to the same problems for normal frames in E. The last kind of problems, as we noted at the beginning of the present section, are known and were investigated in Chapter IV.

We emphasize that a normal frame  $\{E_a\}$  in E, as well as the basis  $\{v(E_a)\}$ for  $\Delta^v$ , can be holonomic as well as anholonomic (see Chapter IV); at the same time, a normal frame  $\{X_I\}$  in T(E) is anholonomic unless the conditions (5.25) hold (Corollary 5.2), a necessary condition being the flatness (integrability) of the horizontal distribution  $\Delta^h$ .

## 7. Coordinates normal along injective mappings

Ending this section, let us say some words regarding frames normal for affine connections on vector bundles.

**Proposition 6.8** (cf. Proposition 6.7). A frame  $\{X_I\}$  is normal on  $U \subseteq E$  for an affine connection  $\Delta^h$ , with 2-index coefficients (5.45) on U, if and only if in it is fulfilled

$$\Gamma^a_{b\mu}|_{\pi(U)} = 0 \tag{6.19a}$$

$$G^a_\mu|_{\pi(U)} = 0.$$
 (6.19b)

*Proof.* The assertion follows from Definition 6.1, equation (5.45), and the linear independence of the vector fibre coordinates  $u^{n+1}, \ldots, u^{n+r}$ , considered as 1-forms.

**Corollary 6.2.** A necessary condition for existence of frames normal on  $U \subseteq E$  for an affine connection is

$$G^a_{\mu}|_{\pi(U)} = 0 \tag{6.20}$$

in all adapted frames on U; in particular, (6.20) is equivalent to  $G|_{(\pi(U))} = 0$  if U is an open set.

*Proof.* Use (6.19b) and (5.47b).

**Corollary 6.3.** A necessary condition for existence of frames normal on  $U \subseteq E$  for an affine connection is

$$\Gamma^a_\mu|_U = -\{(\Gamma^a_{b\mu} \circ \pi) \cdot u^b\}|_U \tag{6.21}$$

in all adapted frames on U; in particular, if U is an open set, then (6.21) means that the restriction of the connection considered on  $(U, \pi|_U, \pi(U))$  is a linear connection.

Proof. Apply Proposition 6.8 and Corollary 6.2.

**Corollary 6.4.** If an affine connection admits frames normal on  $U \subseteq E$ , then all of them are normal on U for the linear connection, corresponding to it via the mapping  $\alpha$  constructed in the proof of Proposition 4.7, and vice versa.

*Proof.* Use Corollary 6.3 and Proposition 6.7.

Thus, if the condition (6.20) is satisfied, the above results completely reduce the problems of existence, (un)uniqueness and the properties of frames normal for affine connections to the same problems for linear connections (that correspond to them).

# 7. Coordinates normal along injective mappings with non-vanishing horizontal component

The purpose of this section is a multi-dimensional generalization of Proposition 6.5 in the real case,  $\mathbb{K} = \mathbb{R}$ . It is formulated below as Theorem 7.1. For its proof we shall essentially need Lemma III.8.1 on page 163.

Let  $(E, \pi, M)$  be a  $C^3$  bundle endowed with  $C^1$  connection  $\Delta^h$ . Let  $k \in \mathbb{N}$ ,  $k \leq \dim M$ , and  $J^k$  be an open set in  $\mathbb{R}^k$ . Consider a  $C^2$  regular injective mapping  $\beta \colon J^k \to E$  such that the vector fields  $\dot{\beta}_{\alpha} \colon s \mapsto \dot{\beta}_{\alpha}(s) \coloneqq \frac{\partial \beta^I(s)}{\partial s^{\alpha}} \frac{\partial}{\partial u^I}|_{\beta(s)}$ , with  $s \coloneqq (s^1, \ldots, s^k) \in J^k$  and  $\alpha = 1, \ldots, k$ , do not belong to the vertical distribution  $\Delta^v, \dot{\beta}_{\alpha}(s) \notin \Delta^v_{\beta(s)}$  for all  $s \in J^k$ ; in particular, the mapping  $\beta$  can be a horizontal mapping in a sense that  $\dot{\beta}_{\alpha}(s) \in \Delta^h_{\beta(s)}$  for all  $s \in J^k$ , but generally these vectors can have a vertical component too. Our aim is to find the integrability conditions for the normal frame/coordinates equation (6.6) and its solutions, if any, when  $U = \beta(J_1^k)$  for some subset  $J_1^k \subseteq J^k$ .

Let us take some  $s_0 \in J^k$  and construct the chart  $(U_1, u)$  with  $U_1 \ni \beta(s_0)$ provided by Lemma III.8.1 with E for M and  $\beta$  for  $\gamma$ . If  $J_1^k := \{s \in J^k : \beta(s) \in U_1\}$ and  $p \in U_1$ , then there is a unique  $(s, t) \in J_1^k \times \mathbb{R}^{\dim_{\mathbb{R}} E-k}$  such that  $p = \eta(s, t)$  with  $\eta := u^{-1}$ , i.e.,  $u^I(p) = s^I$  for  $I = 1, \ldots, k$  and  $u^I(p) = t^I$  for  $I = k + 1, \ldots, n + r$ . Besides, we have  $u(\beta(s)) = (s, t_0)$  for all  $s \in J_1^k$  and some fixed  $t_0 \in \mathbb{R}^{\dim_{\mathbb{R}} E-k}$ . Since the vector fields  $\beta_{\alpha}$ ,  $\alpha = 1, \ldots, k$ , are not vertical, we can construct the coordinate system  $\{u^I\}$ , associated to the chart  $(U_1, u)$ , so that they it be *bundle* coordinate system on  $U_1$  (see the proof of Lemma III.8.1). Thus on  $U_1$  we have bundle coordinates  $u^I$  such that

$$(u^1(\eta(s,t)), \dots, u^{n+r}(\eta(s,t))) := (s,t) \in \mathbb{R}^{n+r}$$
  

$$s = (s^1, \dots, s^k) \in J_1^k \quad t = (t^{k+1}, \dots, t^{n+r}) \in \mathbb{R}^{n+r-k}.$$
(7.1)

Let the indices  $\alpha$  and  $\beta$  run from 1 to k and the indices  $\sigma$  and  $\tau$  take the values form k + 1 to n; we set  $\sigma = \tau = \emptyset$  if k = n. Thus, we have  $u^{\alpha}(\eta(s, t)) = s^{\alpha}$ ,  $u^{\sigma}(\eta(s, t)) = t^{\sigma}$ , and  $u^{a}(\eta(s, t)) = t^{a}$ .

**Proposition 7.1.** Under the hypotheses made above, the normal frame/coordinates equation (6.6) with  $U = \beta(J_1^k) = \beta(J^k) \cap U_1$  has solutions if and only if the system of equations

$$\left(\frac{\partial\Gamma_{\alpha}^{b}}{\partial u^{\beta}} - \frac{\partial\Gamma_{\beta}^{b}}{\partial u^{\alpha}}\right)\Big|_{\beta(s)}B_{b}^{a}(s) + \Gamma_{\alpha}^{b}(\beta(s))\frac{\partial B_{b}^{a}(s)}{\partial s^{\beta}} - \Gamma_{\beta}^{b}(\beta(s))\frac{\partial B_{b}^{a}(s)}{\partial s^{\alpha}} = 0, \quad (7.2)$$

where  $\Gamma^a_{\mu}$  are the 2-index coefficients of  $\Delta^h$  in  $\{u^I\}$ , has solutions  $B^a_b: J^k_1 \to \mathbb{R}$ with det $[B^a_b] \neq 0, \infty$ . Besides, if such solutions exist, then all solutions of (6.6) are given on  $U_1$  by the formula

$$\begin{split} \tilde{u}^{a}(\eta(s, t)) &= -\int_{s_{1}}^{s} B^{a}_{b}(s) \Gamma^{b}_{\alpha}(\beta(s)) \,\mathrm{d}s^{\alpha} \\ &- B^{a}_{b}(s) \Gamma^{b}_{\mu}(\beta(s)) [u^{\mu}(\eta(s, t)) - u^{\mu}(\beta(s))] + B^{a}_{b}(s) [u^{b}(\eta(s, t)) - u^{b}(\beta(s))] \\ &+ f^{a}_{\mu\nu}(s; t; \eta) [u^{\mu}(\eta(s, t)) - u^{\mu}(\beta(s))] [u^{\nu}(\eta(s, t)) - u^{\nu}(\beta(s))], \end{split}$$
(7.3)

where  $s_1 \in J_1^k$  is arbitrarily fixed,  $B_b^a$ , with det $[B_b^a] \neq 0, \infty$ , are solutions of (7.2), and the functions  $f_{\mu\nu}^a$  and their first partial derivatives are bounded when  $\mathbf{t} \to \mathbf{t}_0$ .

Remark 7.1. As  $u^{\alpha}(\eta(s, t)) = u^{\alpha}(\beta(s)) \equiv s^{\alpha}$  for all  $\alpha = 1, \ldots, k$ , the terms with  $\mu, \nu = 1, \ldots, k$  in (7.3) have vanishing contribution.

*Remark* 7.2. For k = 1, we have  $\alpha = \beta = 1$ , due to which the equations (7.2) are identically valid and Proposition 7.1 reduces to Proposition 6.5.

*Proof.* To begin with, we rewrite (6.6) as

$$\frac{\partial \tilde{u}^a}{\partial s^\alpha}\Big|_{\beta(s)} = -\frac{\partial \tilde{u}^a}{\partial t^b}\Big|_{\beta(s)}\Gamma^b_\alpha(\beta(s)) \quad \frac{\partial \tilde{u}^a}{\partial t^\sigma}\Big|_{\beta(s)} = -\frac{\partial \tilde{u}^a}{\partial t^b}\Big|_{\beta(s)}\Gamma^b_\sigma(\beta(s)).$$

Introducing a non-degenerate matrix-valued function  $[B_b^a]$  on  $J_1^k$  by

$$B_b^a(s) = \frac{\partial \tilde{u}^a}{\partial t^b}\Big|_{\beta(s)} = \frac{\partial \tilde{u}^a(s, t)}{\partial t^b}\Big|_{t=t_0},$$
(7.4)

we see that (6.6) is equivalent to

$$\frac{\partial \tilde{u}^a}{\partial s^\alpha}\Big|_{\beta(s)} = -B^a_b(s)\Gamma^b_\alpha(\beta(s)) \quad \alpha = 1, \dots, k$$
(7.5a)

$$\frac{\partial \tilde{u}^a}{\partial t^\sigma}\Big|_{\beta(s)} = -B^a_b(s)\Gamma^b_\sigma(\beta(s)) \quad \sigma = k+1,\dots,n.$$
(7.5b)

Expanding  $\tilde{u}^a(\eta(s, t))$  into a Taylor polynomial up to second-order terms relative to  $(t - t_0)$  about the point  $t_0$  and using (7.4) and (7.5), we get:

$$\begin{split} \tilde{u}^{a}(\eta(s,\boldsymbol{t})) &= f^{a}(s) - B^{a}_{b}(s)\Gamma^{b}_{\sigma}(\beta(s))[t^{\sigma} - t^{\sigma}_{0}] + B^{a}_{b}(s)[t^{b} - t^{b}_{0}] \\ &+ f^{a}_{\sigma\tau}(s;\boldsymbol{t};\eta)[t^{\sigma} - t^{\sigma}_{0}][t^{\tau} - t^{\sigma}_{0}] \\ &= f^{a}(s) - B^{a}_{b}(s)\Gamma^{b}_{\mu}(\beta(s))[u^{\mu}(\eta(s,\boldsymbol{t})) - u^{\mu}(\beta(s))] + B^{a}_{b}(s)[u^{b}(\eta(s,\boldsymbol{t})) - u^{b}(\beta(s))] \\ &+ f^{a}_{\mu\nu}(s;\boldsymbol{t};\eta)[u^{\mu}(\eta(s,\boldsymbol{t})) - u^{\mu}(\beta(s))][u^{\nu}(\eta(s,\boldsymbol{t})) - u^{\nu}(\beta(s))], \quad (7.6) \end{split}$$

where  $f^a$  and  $f^a_{\mu\nu}$  are  $C^1$  functions and  $f^a_{\mu\nu}$  and their first partial derivatives are bounded when  $\mathbf{t} \to \mathbf{t}_0$ . The equation (7.5a) is the only condition that puts some restrictions on  $f^a$  and  $B^a_b$  (besides det $[B^a_b] \neq 0, \infty$ ). Inserting (7.6) into (7.5a) and using that  $\beta(s) = \eta(s, t_0)$ , we obtain

$$\frac{\partial f^a(s)}{\partial s^{\alpha}} = -B^a_b(s)\Gamma^b_{\alpha}(\beta(s)).$$
(7.7)

Thus the initial normal coordinates equation (6.6), with  $U = \beta(J_b^k)$ , has solutions if and only if there exist solutions of (7.7) relative to  $f^a$  and/or  $B_b^a$ . The integrability conditions for (7.7) are [34]

$$0 = \frac{\partial^2 f^a}{\partial s^\beta \partial s^\alpha} - \frac{\partial^2 f^a}{\partial s^\alpha \partial s^\beta} = -\frac{\partial}{\partial s^\beta} \left( B^A_b(s) \Gamma^b_\alpha(s) \right) + \frac{\partial}{\partial s^\alpha} \left( B^A_b(s) \Gamma^b_\beta(s) \right) = \cdots$$

and coincide with (7.2), by virtue of  $u^{\alpha}(\beta(s)) = s^{\alpha}$ . This result concludes the proof of the first part of the proposition.

If (7.2) admits solutions  $B_b^a$  with  $\det[B_b^a] \neq 0, \infty$ , then the general solution of (7.7) is  $f^a(s) = -\int_{s_1}^s B_b^a(s) \Gamma_{\alpha}^b(\beta(s)) \, \mathrm{d}s^{\alpha}$  for some  $s_1 \in J_1^k$  and this solution is independent of the integration path in  $J_1^k$ , due to (7.2).

**Lemma 7.1.** Let  $(E, \pi, M)$  be a  $C^3$  bundle endowed with  $C^2$  connection with coefficients  $\Gamma^a_{\mu}$  in the frame adapted to local coordinate system  $\{u^i\}$ , defined before Proposition 7.1. There exist solutions  $B^a_b$  with det $[B^a_b] \neq 0, \infty$  of the system of equations (7.2) if and only if the coefficients  $\Gamma^a_{\mu}$  satisfy the equations

$$R^a_{\alpha\beta}(\beta(s)) = 0 \qquad s \in J^k_1 \tag{7.8a}$$

$$\left(\Gamma^{d}_{\alpha}\frac{\partial^{2}\Gamma^{c}_{\beta}}{\partial u^{b}\partial u^{d}} - \Gamma^{d}_{\beta}\frac{\partial^{2}\Gamma^{c}_{\alpha}}{\partial u^{b}\partial u^{d}}\right)\Big|_{\beta(s)} = 0 \qquad s \in J_{1}^{k}$$
(7.8b)

in which  $R^a_{\mu\nu}$  are the (fibre) components in  $\{u^I\}$  of the curvature of  $\Delta^h$ , defined by (3.37a). If the conditions (7.8) are valid, the set of the solutions of (7.2) coincides with the set of solutions of the system

$$\frac{\partial B_b^a(s)}{\partial s^\alpha} = -B_c^a(s) \frac{\partial \Gamma_\alpha^c}{\partial u^b}\Big|_{\beta(s)} + \frac{\partial D_b^a(s)}{\partial s^\alpha}$$
(7.9)

relative to  $B_b^a$ , where  $D_b^a$  are solutions of

$$\left(\Gamma^{b}_{\alpha}(\beta(s))\frac{\partial}{\partial s^{\beta}} - \Gamma^{b}_{\beta}(\beta(s))\frac{\partial}{\partial s^{\alpha}}\right)D^{a}_{b}(s) = 0.$$
(7.10)

*Proof.* Consider the integrability condition (7.2) for (6.6) in more details. Define functions  $D^a_{b\alpha}: J^k_1 \to \mathbb{K} = \mathbb{R}$  via the equation

$$\frac{\partial B_b^a(s)}{\partial s^\alpha} = -B_c^a(s) \frac{\partial \Gamma_\alpha^c}{\partial u^b}\Big|_{\beta(s)} + D_{b\alpha}^a(s).$$
(7.11)
#### 7. Coordinates normal along injective mappings

The substitution of this equality into (7.2) results in

$$R^b_{\beta\alpha}(\beta(s))B^a_b(s) - \Gamma^b_{\alpha}(\beta(s))D^a_{b\beta}(s) + \Gamma^b_{\beta}(\beta(s))D^a_{b\alpha}(s) = 0,$$

where the functions  $R^a_{\alpha\beta}$  are the (fibre) components in  $\{u^I\}$  of the curvature of  $\Delta^h$ , defined by (3.37a). The simple observation that  $\tilde{u}^{\alpha}$  and  $\tilde{u}^a$ , if they exist as solutions of (6.6), are normal coordinates on the whole bundle space of the restricted bundle  $(U, \pi|_U, \pi(U))$  with  $U = \beta(J_1^k)$  leads to

$$R^a_{\alpha\beta}(\beta(s)) = 0 \qquad s \in J^k_1, \tag{7.12}$$

by virtue of Proposition 6.6. Therefore the previous equation reduces to

$$\Gamma^b_{\alpha}(\beta(s))D^a_{b\beta}(s) - \Gamma^b_{\beta}(\beta(s))D^a_{b\alpha}(s) = 0.$$
(7.13a)

It is clear that (7.11)–(7.13a) are equivalent to (7.2). Consequently, the quantities  $D^a_{b\alpha}$  must be solutions of (7.13a) while the  $C^1$  functions  $B^a_b$  have to be solutions of (7.11). The integrability conditions  $\left(\frac{\partial^2}{\partial s^\beta \partial s^\alpha} - \frac{\partial^2}{\partial s^\alpha \partial s^\beta}\right) B^a_b(s) = 0$  for (7.11) can be written as<sup>1</sup>

$$\begin{split} \left( -\frac{\partial^2 \Gamma_{\alpha}^c}{\partial u^{\beta} \partial u^b} + \frac{\partial^2 \Gamma_{\beta}^c}{\partial u^{\alpha} \partial u^b} + \frac{\partial \Gamma_{\alpha}^d}{\partial u^b} \frac{\partial \Gamma_{\beta}^c}{\partial u^d} - \frac{\partial \Gamma_{\beta}^d}{\partial u^b} \frac{\partial \Gamma_{\alpha}^c}{\partial u^d} \right) \Big|_{\beta(s)} B_c^a(s) \\ + \frac{\partial D_{b\alpha}^a(s)}{\partial s^{\beta}} - \frac{\partial D_{b\beta}^a(s)}{\partial s^{\alpha}} = 0 \end{split}$$

which conditions split into

$$0 = \frac{\partial D^a_{b\alpha}(s)}{\partial s^{\beta}} - \frac{\partial D^a_{b\beta}(s)}{\partial s^{\alpha}}$$
(7.13b)

$$0 = \left( -\frac{\partial^2 \Gamma_{\alpha}^c}{\partial u^{\beta} \partial u^b} + \frac{\partial^2 \Gamma_{\beta}^c}{\partial u^{\alpha} \partial u^b} + \frac{\partial \Gamma_{\alpha}^d}{\partial u^b} \frac{\partial \Gamma_{\beta}^c}{\partial u^d} - \frac{\partial \Gamma_{\beta}^d}{\partial u^b} \frac{\partial \Gamma_{\alpha}^c}{\partial u^d} \right) \Big|_{\beta(s)} = \left( -\Gamma_{\alpha}^d \frac{\partial^2 \Gamma_{\beta}^c}{\partial u^b \partial u^d} + \Gamma_{\beta}^d \frac{\partial^2 \Gamma_{\alpha}^c}{\partial u^b \partial u^d} \right) \Big|_{\beta(s)}, \quad (7.14)$$

where (7.12) and (3.37a) were applied in the derivation of the second equality in (7.14).

Since the system of equations (7.13) always has solutions, e.g.,  $D_{b\alpha}^b(s) = 0$ , we can assert that (7.12) and (7.14) are the integrability conditions for (7.2) and, if (7.12) and (7.14) hold, every solution of (7.11), with  $D_{b\alpha}^a$  satisfying (7.13), is a solution of (7.2) and vice versa.

At the end, the only unproved assertion is that  $D^a_{b\alpha}$  in (7.11) equals to  $\partial_{\alpha}(D^a_b)$ with  $D^a_b$  satisfying (7.10). Indeed, since  $J^k_1$  is an open set and hence is contractible

<sup>&</sup>lt;sup>1</sup>At this point one should require  $\Delta^h$  to be of class  $C^2$  which is possible if the manifolds E and M are of class  $C^3$ .

one, the Poincaré's lemma (see [6, Section 6.3] or [5, pp. 21, 106]) implies the existence of functions  $D_b^a$  on  $J_1^k$  such that  $D_{b\alpha}^b(s) = \partial_{\alpha}(D_b^a)(s)$ , due to (7.13b); inserting this result into (7.13a), we get (7.10).

Remark 7.3. Regardless that the conditions (7.8b) look quite special, they are identically valid for connections with

$$\Gamma^a_{\alpha} = -(\Gamma^a_{b\alpha} \circ \pi) \cdot u^b + G^a_{\alpha} \circ \pi, \qquad (7.15)$$

where  $\Gamma_{b\alpha}^a$  and  $G_{\alpha}^a$  are  $C^2$  functions on  $\pi(\beta(J^k))$ , i.e., for affine connections (see Subsection 4.4). In particular, of this kind are the linear connections on vector bundles – see Proposition 5.4 and 5.5.

At last, we shall formulate the main result of the above considerations as a combination of Proposition 7.1 and Lemma 7.1.

**Theorem 7.1.** Let  $(E, \pi, M)$  be a  $C^3$  bundle endowed with a  $C^2$  connection. Under the hypotheses made and notation introduced before Proposition 7.1, there exist solutions of the normal frame/coordinates equation (6.6) if and only if the connection's coefficients satisfy the equations (7.8). If these equations hold, all coordinates normal on  $\beta(J_1^k)$  are given on  $U_1$  by (7.3), where  $B_b^a$  are solutions of (7.9), with  $D_b^a$  being solutions of (7.10).

Remark 7.4. If there are  $s_0 \in J^k$  and  $\alpha \in \{1, \ldots, k\}$  such that the vector  $\dot{\beta}_{\alpha}(s_0)$  is a vertical vector,  $\dot{\beta}_{\alpha}(s_0) \in \Delta^v_{\beta(s_0)}$ , then Theorem 7.1 remains true with the only correction that the coordinate system  $\{u^I\}$  will not be bundle coordinate system. If this is the case, the constructed coordinates  $\tilde{u}^I$  will be solutions of (6.6), but we cannot assert that they are bundle coordinates which are (locally) normal along  $\beta$  in a neighborhood of the point  $\beta(s_0)$ .

Theorem 7.1 provides a necessary and sufficient condition for the existence of local coordinates in a neighborhood of  $\beta(s_0)$  for any  $s_0 \in J^k$  which are locally normal along  $\beta$ , i.e., on  $\beta(J_1^k)$  for some open subset  $J_1^k \subseteq J^k$  containing  $s_0$ . Moreover, if this condition is valid, the theorem describes locally all coordinates normal along  $\beta$ .

**Exercise 7.1.** Prove that Theorem 7.1 remains valid by requiring  $\beta$  to be locally injective instead of injective, i.e., for each  $s \in J$  to exist subset  $J_s^k \subseteq J^k$  such that  $J_s^k \ni s$  and the restricted mapping  $\beta|_{J_s^k}$  to be injective.

If one needs a version of the above results for complex bundles, they should be considered as real ones (with doubled dimension of the manifolds) for which are applicable the above considerations.

### 8. Links between connections and transports along paths in fibre bundles

As the title of this section indicates, its content is outside of the main topic of the present book. It generalizes part of Section IV.14 and investigates relations between some axiomatic approaches to the general theories of connections, parallel transports, and transports along paths. We hope that the material below will clarify some problems that may have arisen in Chapter IV and will be useful for readers interested in the axiomatization of the concept of a 'parallel transport'.

The widespread approach to the concept of a "parallel transport" is it to be considered as a secondary one and defined on the basis of the connection theory [6,7,10–13,16,28,60,98,106,107,117,141–145]. However, the opposite approach, in which the parallel transport is axiomatically defined and from it the connection theory is constructed, is also known [17,23,30–33,91,147–150] and goes back to 1949<sup>1</sup>; e.g., it is systematically realized in [23], where the connection theory on vector bundles is investigated. In [114] the concept of a "parallel transport" was generalize to the one of "transport along paths". The relations between both concepts were analyzed in [115]; in particular, Theorem 3.1 of [115, p. 13] (see Theorem 8.1 below) contains a necessary and sufficient condition for a transport along paths to be (axiomatically defined) parallel transport. The aim of the present section is to be investigated some links between general connections on fibre bundles and transport along paths in them. Recall that similar problems, but in the linear case in vector bundles, were explored in Section IV.14.

The bundle and base spaces of the bundles in this section are supposed to be of differentiable of class  $C^1$ ; however, some parts of the text below, like Definition 8.1, are valid in more general situations.

**Definition 8.1.** A transport along paths in bundle  $(E, \pi, B)$  is a mapping I assigning to every path  $\gamma: J \to M$  a mapping  $I^{\gamma}$ , termed transport along  $\gamma$ , such that  $I^{\gamma}: (s,t) \mapsto I_{s \to t}^{\gamma}$  where the mapping

$$I_{s \to t}^{\gamma} \colon \pi^{-1}(\gamma(s)) \to \pi^{-1}(\gamma(t)) \qquad s, t \in J,$$

$$(8.1)$$

called transport along  $\gamma$  from s to t, has the properties:

$$I_{s \to t}^{\gamma} \circ I_{r \to s}^{\gamma} = I_{r \to t}^{\gamma} \qquad r, s, t \in J$$

$$(8.2)$$

$$I_{s \to s}^{\gamma} = \operatorname{id}_{\pi^{-1}(\gamma(s))} \qquad s \in J, \tag{8.3}$$

<sup>&</sup>lt;sup>1</sup>It seems that the earliest written accounts on this approach are the ones due to Ü.G. Lumiste [30, Section 2.2] and C. Teleman [17, Chapter IV, Section B.3] (both published in 1964), the next essential steps being made by P. Dombrowski [31, § 1] and W. Poor [23]. Besides, the author of [31] states that his paper is based on unpublished lectures of prof. Willi Rinow (1907– 1979) in 1949; see also [23, p. 46] where the author claims that the first axiomatical definition of a parallel transport in the tangent bundle case is given by prof. W. Rinow in his lectures at the Humboldt University in 1949. Some heuristic comments on the axiomatic approach to parallel transport theory can be found in [8, Section 2.1] too.

where  $\circ$  denotes composition of mappings and  $\mathsf{id}_X$  is the identity mapping of a set X.

*Remark* 8.1. If  $(E, \pi, M)$  is a vector bundle and the mappings (8.1) are linear, Definition 8.1 reduces to Definition IV.3.1.

An analysis and various comments on this definition can be found in [87,102, 114,115]; see also Section IV.3.

As we shall see below, an important special class of transports along paths is selected by the conditions

. \_/

$$I_{s \to t}^{\gamma|J'} = I_{s \to t}^{\gamma} \qquad \qquad s, t \in J'$$

$$(8.4)$$

$$I_{s \to t}^{\gamma \circ \chi} = I_{\chi(s) \to \chi(t)}^{\gamma} \qquad s, t \in J'', \tag{8.5}$$

where  $J' \subseteq J$  is a subinterval,  $\gamma | J'$  is the restriction of  $\gamma$  to J', and  $\chi : J'' \to J$  is a bijection of a real interval J'' onto J.

Putting r = t in (8.2) and using (8.3), we see that the mappings (8.1) are invertible and

$$(I_{s \to t}^{\gamma})^{-1} = I_{t \to s}^{\gamma}.$$
(8.6)

The following result describes how a transport along paths generates a connection.

**Proposition 8.1.** Let I be a transport along paths in a bundle  $(E, \pi, M)$ . Let  $\gamma: J \to M$  be a path and, for any  $s_0 \in J$  and  $p \in \pi^{-1}(\gamma(s_0))$ , the lift  $\bar{\gamma}_{s_0,p}: J \to E$  of  $\gamma$  be defined by

$$\bar{\gamma}_{s_0,p}(t) = I_{s_0 \to t}^{\gamma}(p) \qquad t \in J.$$
(8.7)

Suppose that I is such that  $\bar{\gamma}_{s_0,p}$  is a  $C^1$  path for every  $s_0$  and p and the conditions (8.4) and (8.5) hold for all paths  $\gamma: J \to M$ . Then the distribution

$$\Delta^{I} : p \mapsto \Delta^{I}_{p} := \left\{ \frac{\mathrm{d}}{\mathrm{d}t} \Big|_{t=s_{0}} \left( \bar{\gamma}_{s_{0},p}(t) \right) \\ : \gamma : J \to M \text{ is } C^{1} \text{ and injective, } s_{0} \in J, \ \gamma(s_{0}) = \pi(p) \right\} \subseteq T_{p}(E), \quad (8.8)$$

with  $p \in E$ , is a connection on  $(E, \pi, M)$ , i.e.,

$$\Delta_p^v \oplus \Delta_p^I = T_p(E) \qquad p \in E, \tag{8.9}$$

with  $\Delta^v$  being the vertical distribution on E,  $\Delta^v_p = T_p(\pi^{-1}(\pi(p)))$ .

*Proof.* Let  $\{u^{\mu} = x^{\mu} \circ \pi, u^a\}$  be bundle coordinate system on E, p be a point in its domain, and  $\bar{\gamma} \colon J \to E$  be a lift of  $\gamma \colon J \to M$ ,  $\pi \circ \bar{\gamma} = \gamma$ . Since  $\pi_* \left(\frac{\partial}{\partial u^I}\Big|_p\right) = \frac{\partial (x^{\mu} \circ \pi)}{\partial u^I}\Big|_p \frac{\partial}{\partial x^{\mu}}\Big|_{\pi(p)}$  and  $\dot{\bar{\gamma}}^{\mu}(t) = \frac{d(u^{\mu} \circ \bar{\gamma}(t))}{dt} = \frac{d(x^{\mu} \circ \gamma(t))}{dt} = \dot{\gamma}^{\mu}(t)$  for all  $t \in J$ , we have

$$\pi_*(\dot{\bar{\gamma}}(t)) = \dot{\gamma}(t) \tag{8.10}$$

for any lift  $\bar{\gamma}$  in E of a path  $\gamma$  in M. Therefore, from (8.8) and (8.10), we get

$$\pi_*(\Delta_p^I) = \left\{ \dot{\gamma}(s_0) : \gamma \colon J \to M \text{ is } C^1 \text{ and injective, } s_0 \in J, \ \gamma(s_0) = \pi(p) \right\}$$
$$= T_{\pi(p)}(M)$$

as  $\dot{\gamma}(s_0)$  is an arbitrary vector in  $T_{\pi(p)}(M) = T_{\gamma(s_0)}(M)$ . Thus the mapping  $\pi_*|_{\Delta_p^I} \colon \Delta_p^I \to T_{\pi(p)}(M)$  is surjective. It is also linear, due to the definition of a tangent mapping [7, Section 1.22]. At last, we shall prove that  $\pi_*|_{\Delta_p^I}$  is injective, from where it follows that  $\pi_*|_{\Delta_p^I}$  is a vector space isomorphism for every  $p \in E$  which, in its turn, implies (8.9) as  $\pi_*(\Delta_p^v) = 0_{\pi(p)} \in T_{\pi(p)}(M)$ .

If  $G_i \in \Delta_p^I$ , i = 1, 2, then there exist paths  $\gamma_i \colon J_i \to M$  such that  $\gamma_i(s_i) = \pi(p)$  for some  $s_i \in J_i$  and  $G_i = \dot{\gamma}_{i;s_i,p}$ , with i = 1, 2 and the lifted paths in the right-hand side being given by (8.7). Then  $\pi_*(G_i) = \dot{\gamma}_i(s_i)$ , due to (8.10). Suppose that  $\dot{\gamma}_1(s_1) = \dot{\gamma}_2(s_2)$ . Since  $\gamma_1(s_1) = \gamma_2(s_2) = \pi(p)$ , the last equality entails  $s_1 = s_2$ ,  $J_1 \cap J_2 \neq \emptyset$ , and the existence of an interval  $J' \subseteq J_1 \cap J_2$  such that  $s_1 = s_2 \in J'$  and  $\gamma_1 | J' = \gamma_2 | J'.^2$  Combining this result with (8.4), we get  $G_1 = G_2$ , which means that  $\pi_* | \Delta_p^I \colon \Delta_p^I \to T_{\pi(p)}(M)$  is injective.  $\Box$ 

Remark 8.2. The condition (8.5) was not used explicitly in the proof of Proposition 8.1, but it is important in the definition (8.8) of the connection  $\Delta^I$ . Namely, it ensures that, if  $X \in \Delta_p^I$ , then  $\varkappa X \in \Delta_p^I$  for all  $\varkappa \in \mathbb{K}$ . Indeed, if  $X = \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=s_0}(\bar{\gamma}_{s_0,p}(t))$  for some path  $\gamma: J \to M$  and  $\chi: J' \to J$  is a  $C^1$  diffeomorphism, the vector tangent to the path  $\bar{\beta}_{s'_0,p}(t')$  at  $s'_0 = \chi^{-1}(s_0)$ , with  $\beta = \gamma \circ \chi$ , is  $\chi(s'_0)X$ , by virtue of (8.7) and (8.5). So that  $\chi(s'_0)X \in \Delta_p^I$  and the arbitrariness of  $\chi$  leads to  $\varkappa X \in \Delta_p^I$  for all  $\varkappa \in \mathbb{K}$ .

Remark 8.3. The role of the transport I along paths in Proposition 8.1 is on its base to be constructed a lifting of the paths in M to paths in E with appropriate properties. Namely, such a lifting should assign to a path  $\gamma: J \to M$  a unique path  $\bar{\gamma}_{s_0,p}: J \to E$  passing through a given point  $p \in \pi^{-1}(\gamma(s_0))$ , for some  $s_0 \in J$ , and such that  $\pi \circ \bar{\gamma}_{s_0,p} = \gamma$  and, if  $q \in \bar{\gamma}_{s_0,p}(t_0)$  for some  $t_0 \in J$ , then  $\bar{\gamma}_{t_0,q} = \bar{\gamma}_{s_0,p}$ . On this ground one can generalize Proposition 8.1 as well as some of the next considerations and results.

**Definition 8.2.** The connection  $\Delta^{I}$ , defined in Proposition 8.1, will be called *assigned to (defined by, generated by)* the transport *I* along paths.

For the further exploration of the relations between transports along paths and connections (or parallel transports generated by them), we shall need the notion of an inverse path and of a product of paths. There are not 'natural' definitions of these concepts, but this is not important for us as the (parallel) transports we shall consider below are parametrization invariant in some sense, like (8.5).

 $<sup>^2\</sup>mathrm{Here}$  we use the local existence of a unique  $C^1$  path passing trough a given point and having a fixed tangent vector at it.

For that reason, the concepts mentioned will be defined only for canonical paths  $[0,1] \to M$ , whose domain is the real interval  $[0,1] := \{r \in \mathbb{R} : 0 \le r \le 1\}$ . The path inverse to  $\gamma : [0,1] \to M$  is  $\gamma_{-} := \gamma \circ \tau_{-} : [0,1] \to M$ , with  $\tau_{-} : [0,1] \to [0,1]$  being given by  $\tau_{-}(t) := 1 - t$  for  $t \in [0,1]$ . If  $\gamma_{1}, \gamma_{2} : [0,1] \to M$  and  $\gamma_{1}(1) = \gamma_{2}(0)$ , the product  $\gamma_{1}\gamma_{2}$  of  $\gamma_{1}$  and  $\gamma_{2}$  is a canonical path  $\gamma_{1}\gamma_{2} : [0,1] \to M$  such that  $(\gamma_{1}\gamma_{2})(t) := \gamma_{1}(2t)$  for  $t \in [0,1/2]$  and  $(\gamma_{1}\gamma_{2})(t) := \gamma_{2}(2t-1)$  for  $t \in [1/2,1]$ . For more details on this item, see [108,132].

Recall now the basic properties of the parallel transports generated by connections.

#### Proposition 8.2. Let

$$\mathsf{P} \colon \gamma \mapsto \mathsf{P}^{\gamma} \colon \pi^{-1}(\gamma(\sigma)) \to \pi^{-1}(\gamma(\tau)) \qquad \gamma \colon [\sigma, \tau] \to M$$
(8.11)

be the parallel transport generated by a connection on some bundle  $(E, \pi, M)$ . The mapping P has the following properties:

(i) The parallel transport P is invariant under orientation preserving changes of the paths' parameters. Precisely, if γ: [σ, τ] → M and χ: [σ', τ'] → [σ, τ] is an orientation preserving C<sup>1</sup> diffeomorphism, then

$$\mathsf{P}^{\gamma \circ \chi} = \mathsf{P}^{\gamma}. \tag{8.12}$$

(ii) If  $\gamma: [0,1] \to M$  and  $\gamma: [0,1] \to M$  is its canonical inverse,  $\gamma(t) = \gamma(1-t)$ for  $t \in [0,1]$ , then

$$\mathsf{P}^{\gamma_{-}} = \left(\mathsf{P}^{\gamma}\right)^{-1}.\tag{8.13}$$

(iii) If  $\gamma_1, \gamma_2: [0,1] \to M$ ,  $\gamma_1(1) = \gamma_2(0)$ , and  $\gamma_1\gamma_2: [0,1] \to M$  is their canonical product, then

$$\mathsf{P}^{\gamma_1\gamma_2} = \mathsf{P}^{\gamma_2} \circ \mathsf{P}^{\gamma_1}. \tag{8.14}$$

(iv) If  $\gamma_{r,x}$ :  $\{r\} = [r,r] \to \{x\}$  for some given  $r \in \mathbb{R}$  and  $x \in M$ , then

$$\mathsf{P}^{\gamma_{r,x}} = \mathsf{id}_{\pi^{-1}(x)}.\tag{8.15}$$

Remark 8.4. As a result of (8.12), some properties of the parallel transports generated by connections, like (8.13) and (8.14), are sufficient to be formulated/proved only for canonical paths  $[0, 1] \rightarrow M$ .

*Proof.* The proofs of (8.12)–(8.15) can be found in a number of works, for example in [3, 4, 6, 11, 30, 32, 33, 148, 149, 154]. Alternatively, the reader can prove them by applying the definitions given in this book. (See also Subsections IV.14.1 and IV.14.2 in a case of a vector bundle.)

**Definition 8.3** (cf. Definition IV.14.11 on page 313). A mapping (8.11) satisfying the equalities (8.12)–(8.15) will be called (*axiomatically defined*) parallel transport.

**Proposition 8.3.** Let  $\mathsf{P}$  be the parallel transport assigned to a  $\mathbb{C}^m$ , with  $m \in \mathbb{N} \cup \{0\}$ , connection on a smooth, of class  $\mathbb{C}^{m+1}$ , bundle  $(E, \pi, M)$ . Then  $\mathsf{P}$  is smooth, of class  $\mathbb{C}^m$ , in a sense that, if  $\gamma \colon [\sigma, \tau] \to M$  is a  $\mathbb{C}^1$  path, then  $\mathsf{P}^{\gamma}$  is in the set of  $\mathbb{C}^m$  diffeomorphisms between the fibres  $\pi^{-1}(\gamma(\sigma))$  and  $\pi^{-1}(\gamma(\tau))$ ,

$$\mathsf{P} \colon \gamma \mapsto \mathsf{P}^{\gamma} \in \mathrm{Diff}^{m} \left( \pi^{-1}(\gamma(\sigma)), \pi^{-1}(\gamma(\tau)) \right) \qquad \gamma \colon [\sigma, \tau] \to M.$$
(8.16)

*Proof.* See [16, 106, 107].

The axiomatic approach to parallel transport was developed mainly on the ground on the properties (8.12)–(8.16) of the parallel transports assigned to connections. However, this topic is out of the range of the present monograph and the reader is referred to the literature cited at the beginning of the present section.

Ending with the results we take for granted, we reproduce below a slightly modified version of [115, p. 13, Theorem 3.1].

**Theorem 8.1.** Let I be a transport along paths in bundle  $(E, \pi, M)$  and  $\gamma: [\sigma, \tau] \rightarrow M$ . If I satisfies the conditions (8.4) and (8.5), then the mapping

$$I: \gamma \mapsto I^{\gamma} := I^{\gamma}_{\sigma \to \tau}: \pi^{-1}(\gamma(\sigma)) \to \pi^{-1}(\gamma(\tau)) \qquad \gamma: [\sigma, \tau] \to M$$
(8.17)

is a parallel transport, i.e., it possess the properties (8.12)–(8.15), with I for P. Besides, if I is smooth in a sense that

$$I_{s \to t}^{\beta} \in \operatorname{Diff}^{m}\left(\pi^{-1}(\beta(s)), \pi^{-1}(\beta(t))\right) \qquad \beta \colon J \to M \quad s, t \in J$$
(8.18)

for some  $m \in \mathbb{N} \cup \{0\}$ , then the mapping (8.17) satisfies (8.16), with I for P.

Conversely, suppose the mapping

$$\mathsf{P} \colon \gamma \mapsto \mathsf{P}^{\gamma} \colon \pi^{-1}(\gamma(\sigma)) \to \pi^{-1}(\gamma(\tau)) \qquad \gamma \colon [\sigma, \tau] \to M$$
(8.19)

is a parallel transport, i.e., satisfies (8.12)–(8.15), and define the mapping

$$P \colon \beta \mapsto P^{\beta} \colon (s,t) \mapsto P_{s \to t}^{\beta} = \mathsf{P}^{(\beta|[\sigma,\tau]) \circ \chi_{t}^{[\sigma,\tau]}} \circ \left(\mathsf{P}^{(\beta|[\sigma,\tau]) \circ \chi_{s}^{[\sigma,\tau]}}\right)^{-1} \qquad \beta \colon J \to M,$$

$$(8.20)$$

where  $s, t \in J$ ,  $\sigma, \tau \in J$  are such that  $\sigma \leq \tau$  and  $[\sigma, \tau] \ni s, t, {}^{3}$  and  $\chi_{s}^{[\sigma,\tau]} : [\sigma, \tau] \rightarrow [\sigma, s]$  are for  $s > \sigma$  arbitrary orientation preserving  $C^{1}$  diffeomorphisms (depending on  $\beta$  via the interval  $[\sigma, \tau]$ ). Then the mapping (8.20) is a transport along paths in  $(E, \pi, M)$ , which transport satisfies the conditions (8.4) and (8.5), with P for I. Besides, under the same assumptions, the condition (8.16) for  $\mathsf{P}$  implies (8.18), with P for I, where P is given by (8.20).

 $\Box$ 

<sup>&</sup>lt;sup>3</sup>In particular, one can set  $\sigma = \min(s, t)$  and  $\tau = \max(s, t)$  or, if J is a closed interval, define  $\sigma$  and  $\tau$  as the end points of J, i.e.,  $J = [\sigma, \tau]$ .

*Remark* 8.5. Instead by (8.20), the transport P along paths generated by a parallel transport  $\mathsf{P}$  can be defined equivalently as follows. For a path  $\gamma \colon [\sigma, \tau] \to M$  and  $s, t \in [\sigma, \tau]$ , we put.

$$P: \gamma \mapsto P^{\gamma}: (s,t) \mapsto P_{s \to t}^{\gamma} = \mathsf{P}^{\gamma \circ \chi_{t}^{[\sigma,\tau]}} \circ \left(\mathsf{P}^{\gamma \circ \chi_{s}^{[\sigma,\tau]}}\right)^{-1} = \mathsf{P}^{\gamma \mid [\sigma,t]} \circ \left(\mathsf{P}^{\gamma \mid [\sigma,s]}\right)^{-1}$$
$$\gamma: [\sigma,\tau] \to M$$
(8.21a)

Now, for an arbitrary path  $\beta \colon J \to M$ , with J being closed or open at one or both its ends, we set

$$P: \beta \mapsto P^{\beta}: (s,t) \mapsto P_{s \to t}^{\beta} = \begin{cases} P_{s \to t}^{\beta \mid [s,t]} & \text{for } s \le t \\ \left(P_{t \to s}^{\beta \mid [t,s]}\right)^{-1} & \text{for } s \ge t \end{cases} \qquad \beta: J \to M \quad s, t \in J.$$

$$(8.21b)$$

It can easily be verified that (8.21) are tantamount to

$$P: \beta \mapsto P^{\beta}: (s,t) \mapsto P_{s \to t}^{\beta} = \begin{cases} \mathsf{P}^{\beta \mid [s,t]} & \text{for } s \le t \\ \left(\mathsf{P}^{\beta \mid [t,s]}\right)^{-1} & \text{for } s \ge t \end{cases} \qquad \beta: J \to M \quad s, t \in J.$$

$$(8.22)$$

**Definition 8.4.** A transport I along paths which has the properties (8.4) and (8.5) will be called *parallel* transport along paths.

Theorem 8.1 simply says that there is a bijective correspondence between the parallel transports along paths and the parallel transports.

**Definition 8.5.** If I is a parallel transport along paths, then we say that the parallel transport (8.17) is generated by (defined by, assigned to) I. Respectively, if P is a parallel transport, then we say that the (parallel) transport along paths (8.20) is generated by (defined by, assigned to) P.

Let us now return to the connection  $\Delta^I$  generated by a transport I along paths, introduced in Proposition 8.1.

**Proposition 8.4.** If  $\gamma: J \to M$  is an injective path and  $p \in \pi^{-1}(\gamma(s_0))$  for some  $s_0 \in J$ , then there is a unique  $\Delta^I$ -horizontal lift of  $\gamma$  (relative to  $\Delta^I$ ) in E through p and it is exactly the path  $\bar{\gamma}_{s_0,p}$  defined by (8.7).

*Proof.* Simply apply the definitions (8.7) and (8.8) and use the properties (8.1) and (8.2) of the transports along paths  $(t_0 \in J)$ :

$$\dot{\bar{\gamma}}_{s_0,p}(t_0) = \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=t_0} \left(I_{s_0 \to t}^{\gamma}(p)\right) = \frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=t_0} \left(I_{t_0 \to t}^{\gamma}(I_{s_0 \to t_0}^{\gamma}(p))\right) \in \Delta^{I}_{\bar{\gamma}(t_0)}. \qquad \Box$$

Remark 8.6. If  $\gamma$  is not injective and  $\gamma(s_0) = \gamma(t_0)$  for some  $s_0, t_0 \in J$  such that  $s_0 \neq t_0$ , then the paths  $\bar{\gamma}_{s_0,p}, \bar{\gamma}_{t_0,p}: J \to E$  need not to coincide as  $\bar{\gamma}_{s_0,p} =$ 

 $I_{t_0 \to s_0} \circ \bar{\gamma}_{t_0,p}$ , due to (8.7) and (8.2). Therefore the  $\Delta^I$ -horizontal lift of a noninjective path through a point in E lying in a fibre over a self-intersection point of the path, if any, may not be unique. Similar is the situation for an arbitrary connection. (This range of problems is connected with the so-called holonomy groups.)

**Proposition 8.5.** The parallel transport I generated by the connection, defined by a transport along paths I, is such that

$$I^{\gamma} = I^{\gamma}_{\sigma \to \tau} \quad for \ \gamma \colon [\sigma, \tau] \to M.$$
 (8.23)

*Proof.* According to Definition 3.2 and (8.7), we have:

$$\begin{split} \mathsf{I} \colon \gamma \mapsto \mathsf{I}^{\gamma} \colon \pi^{-1}(\gamma(\sigma)) \to \pi^{-1}(\gamma(\tau)) & \gamma \colon [\sigma, \tau] \to M \\ \mathsf{I}^{\gamma} \colon p \mapsto \mathsf{I}^{\gamma}(p) = \bar{\gamma}_{\sigma, p}(\tau) = I^{\gamma}_{\sigma \to \tau}(p) & p \in \pi^{-1}(\gamma(\sigma)). \end{split}$$

**Corollary 8.1.** The parallel transport I generated by the connection  $\Delta^I$ , assigned to a transport I along paths, is a parallel transport, i.e., it satisfies (8.12)–(8.15) with I for P.

*Proof.* This result is a particular case of Proposition 8.2. An alternative proof can be carried out by using (8.1)–(8.7), (8.23), and the definitions of inverse path and product of paths. The assertion is also a consequence of (8.23) and Theorem 8.1.

Until this point, we have studied how a transport along paths generates a connection (Proposition 8.1) and parallel transport (Proposition 8.5). Besides, Theorem 8.1 establishes a bijective correspondence between particular class of transports along paths and mappings having (some of) the main properties of the parallel transports generated by connections. Below we shall pay attention, in a sense, to the opposite links, starting from a connection on a bundle.

**Proposition 8.6.** Let  $\mathsf{P}$  be the parallel transport assigned to a connection  $\Delta^h$  on a bundle  $(E, \pi, M)$ . The mapping

$$P: \gamma \mapsto P^{\gamma}: (s,t) \mapsto P_{s \to t}^{\gamma} \qquad \gamma: J \to M$$
(8.24a)

defined by

$$P_{s \to t}^{\gamma} = \begin{cases} \mathsf{P}^{\gamma|[s,t]} & \text{for } s \le t \\ \left(\mathsf{P}^{\gamma|[t,s]}\right)^{-1} & \text{for } s \ge t \end{cases}$$
(8.24b)

is a transport along paths in  $(E, \pi, M)$ . Moreover, P is parallel transport along paths, i.e., it satisfies the equations (8.4) and (8.5) with P for I.

*Proof.* One should check the conditions (8.1)–(8.5) with P for I. The relations (8.1) and (8.2) follow directly from Definition 3.2 of a parallel transport generated by a connection. The rest conditions are consequences of (8.24) and a simple,

but tedious, application of the properties (8.12)–(8.14) of the parallel transports. Alternatively, this proposition is a consequence of the second part of Theorem 8.1 and Remark 8.5.

Remark 8.7. Applying (8.12)–(8.14), the reader can verify that

$$P_{s \to t}^{\gamma} = F^{-1}(t;\gamma) \circ F(s;\gamma) \tag{8.25}$$

with

$$F(r;\gamma) = \begin{cases} \mathsf{P}^{\gamma|[r,w]} & \text{for } r \le w\\ \left(\mathsf{P}^{\gamma|[w,r]}\right)^{-1} & \text{for } r \ge w \end{cases} \qquad r = s,t \tag{8.26}$$

for any (arbitrarily) fixed  $w \in J$ . This result is a special case of the general structure of the transports along paths [114, Theorem 3.1].

**Definition 8.6.** The parallel transport along paths, defined by a connection  $\Delta^h$  on a bundle through Proposition 8.6, will be called parallel transport along paths assigned to (defined by, generated by) the connection  $\Delta^h$ .

**Corollary 8.2.** Let  $\Delta^I$  be the connection generated by a parallel transport I along paths according to Proposition 8.1. If I is the parallel transport assigned to  $\Delta^I$ , then the transport along paths assigned to I (or  $\Delta^I$ ), as described in Proposition 8.6, coincides with the initial transport I along paths.

*Proof.* Substitute (8.23) into (8.24), with I for P and I for P.  $\Box$ 

**Corollary 8.3.** Let P be the transport along paths assigned to a connection  $\Delta^h$  (via its parallel transport P) according to Proposition 8.6. The connection  $\Delta^P$  generated by P, as described in Proposition 8.1, coincides with the initial connection  $\Delta^h$ ,  $\Delta^P = \Delta^h$ .

*Proof.* On one hand, if  $p \in E$ , the space  $\Delta_p^P$  consists of the vectors tangent at  $s_0$  to the paths  $\bar{\gamma}_{s_0,p} : t \mapsto P_{s_0 \to t}^{\gamma}(p)$ , with  $\gamma : J \to M$ ,  $s_0 \in J$ , and  $\pi(p) = \gamma(s_0)$ , due to Proposition 8.1. On another hand,  $\Delta_p^h$  consists of the vectors tangent at  $s_0$  to the paths  $\tilde{\gamma}_{s_0,p} : t \mapsto \mathsf{P}^{\gamma \mid [s_0,t]}(p)$ , by virtue of Definition 3.2. Equation (8.24b) says that both types of paths coincide,  $\tilde{\gamma}_{s_0,p} = \bar{\gamma}_{s_0,p}$ , so that their tangent vectors at  $t = s_0$  are identical and, consequently,  $\Delta_p^P$  and  $\Delta_p^h$  are equal as sets,  $\Delta_p^P = \Delta_p^h$ , for all  $p \in E$ .

Roughly speaking, the above series of results says that a connection  $\Delta^h$  is equivalent to a mapping P (the assigned to it parallel transport) satisfying (8.11)–(8.15) or to a mapping P (the assigned to it parallel transport along paths) satisfying (8.1)–(8.5) (with P for I). Besides, the smoothness of  $\Delta^h$  is equivalent to the one of P or P. Let us summarize these results as follows.

**Theorem 8.2.** Given a connection  $\Delta^h$  on a bundle  $(E, \pi, M)$ , there exists a unique parallel transport I along paths in  $(E, \pi, M)$  which generates  $\Delta^h$  via (8.8), i.e.,  $\Delta^I = \Delta^h$ . Besides, the parallel transport P defined by  $\Delta^h$  is given by (8.17), i.e., P = I.

*Proof.* See Propositions 8.1 and 8.5 and Theorem 8.1.

**Theorem 8.3.** Given a parallel transport I along paths in a bundle  $(E, \pi, M)$ , then there exists a unique connection  $\Delta^h$  on  $(E, \pi, M)$  such that the parallel transport P along paths assigned to  $\Delta^h$  coincides with I, P = I. Besides, the connection  $\Delta^I$ generated by I is identical with  $\Delta^h$ ,  $\Delta^I = \Delta^h$ .

*Proof.* Apply Proposition 8.1 and corollaries 8.2 and 8.3.

**Theorem 8.4.** Given a parallel transport along paths in a bundle, there is a unique (axiomatically defined) parallel transport generating it. Conversely, given a parallel transport, there is a unique parallel transport along paths generating it.

*Proof.* This statement is a reformulation of Theorem 8.1.

**Theorem 8.5.** Given a parallel transport  $\mathsf{P}$ , there exists a unique connection  $\Delta^h$  generating it. Besides, the parallel transport assigned to  $\Delta^h$  coincides with  $\mathsf{P}$ .

*Proof.* See Theorems 8.4 and 8.3 and Definitions 3.2 and 8.5.

**Theorem 8.6.** Given a connection  $\Delta^h$ , there is a unique parallel transport  $\mathsf{P}$  such that the defined by it parallel transport P along paths generates  $\Delta^h$ ,  $\Delta^P = \Delta^h$ . Besides,  $\mathsf{P}$  coincides with the parallel transport assigned to  $\Delta^h$ .

Proof. Apply Theorems 8.2 and 8.4.

The above results can be summarized in the commutative diagram shown on figure 8.1, the mappings in which are described via Theorems 8.2–8.6. Besides, if





one of these objects is smooth, so are the other ones corresponding to it via the bijections constructed in the present section.

We end with the main moral of this section. The concepts "connection," "(axiomatically defined) parallel transport" and "parallel transport along path"

 $\square$ 

 $\square$ 

 $\square$ 

(i.e., one that satisfies equations (8.4) and (8.5)) are equivalent in a sense that there are bijective mappings between the sets of these objects. Besides, if one of these objects is smooth, so are the other ones corresponding to it via the bijections constructed in the present section.

### 9. Conclusion

In this chapter we have presented a short (and partial) review of (one of the approaches to) the connection theory on bundles whose base and bundle spaces are  $(C^2)$  differentiable manifolds. Special attention was paid to connections, in particular linear ones, on vector bundles, which find wide applications in physics [146,155]. However, many other approaches, generalizations, alternative descriptions, particular methods, etc. were not mentioned at all. In particular, these include: connections on more general (e.g., topological) bundles, connections on principal bundles (which are important in the gauge field theories), holonomy groups, flat connections, Riemannian connections, etc., etc. The surveys [32, 156] contain essential information on these and many other items. Consistent and self-contained exposition of such problems can be found in [23, 110, 151, 157].

If additional geometric structures are added to the theory considered in Section 3, there will become important connections compatible with these structures. In this way arise many theories of particular connections; we have demonstrated that on the example of linear connections on vector bundles (Section 4).

The consideration of arbitrary (co)frames in Section 5 may seem slightly artificial as the general theory can be developed without them. However, this is not the generic case when one starts to apply the connection theory for investigation of particular problems. It may happen that some problem has solutions in general (co)frames but it does not possess solutions when (co)frames generated directly by local coordinates are involved. For example, local coordinates (holonomic frames) normal at a given point for a covariant derivative operator (linear connection)  $\nabla$  on a manifold exist if  $\nabla$  is torsionless at that point, but anholonomic frames normal at a given point for  $\nabla$  exist in a case of non-vanishing torsion.

In Section 6, we saw that the theory of normal frames in the most general case is quite trivial. This reflects the understanding that the more general a concept is, the less particular properties it has, but the more concrete applications it can find if it is restricted somehow. This situation was demonstrate when holonomic normal frames were considered; e.g., they exist at a given point or along an injective horizontal path, but on an open set they may exist only in the flat case. A feature of a vector bundle  $(E, \pi, M)$  is that the frames in T(E) over E are in bijective correspondence with pairs of frames in E over M and in T(M) over M. This result allows the normal frames in T(E), if any, to be 'lowered' to ones in E. From here a conclusion was made that the theory of frames in T(E) normal for linear connections on a vector bundle is equivalent to the existing one of frames in Enormal for covariant derivatives in  $(E, \pi, M)$ . The concept "parallel transport" precedes historically the one of a "connection" and was first clearly formulated in the work [29] of Levi Civita on a parallel transport of a vector in Riemannian geometry. The connection theory was formulated approximately during the period 1920–1949 in a series of works on particular connections and their subsequent generalizations and has obtained an almost complete form in 1950–1955 together with the clear formulation of the concepts "manifold" and "fibre bundle" [32]; the concept of connection on general fibre bundle was established at about 1970 [158]. During that period, with a few exceptional works, the 'parallel transport' was considered as a secondary concept, defined by means of the one of a 'connection'. Later, as we pointed at the beginning of Section 8, there appear several attempts for axiomatizing the concept of a parallel transport and by its means the connection theory to be constructed; e.g., this approach is developed deeply in vector bundles in [23]. The major classical results on axiomatization of parallel transport theory are presented in Table 9.1.

Table 9.1: Main	contributions i	n axiom	atizing the	e concept c	of "parallel	transport".
			0	±	-	-

Year	Person	Result and original reference
1917	T. Levi-Civita	Definition of a parallel transport of a vector in Riemannian geometry. [29]
1949	Willi Rinow	An axiomatic definition of parallel transport in tangent bundle is introduced in unpublished lectures at Humboldt university. (See [31] and [23, p. 46].)
1964	Ü.G. Lumiste	Definition of a connection in principal bundle (with homo- geneous fibres) as a parallel transport along canonical paths $\alpha: [0,1] \to M$ in its base $M$ . The parallel transport is defined as a mapping from the fibre over $\alpha(0)$ into the one over $\alpha(1)$ satisfying some axioms. [30, Section 2.2]
1964	C. Teleman	Definition of a connection in topological bundle as a parallel transport along canonical paths $\alpha: [0, 1] \to M$ in its base $M$ The parallel transport is defined as a lifting of these paths through a point in the fibre over their initial points $\alpha(0)$ . [17, Chapter IV, Section B.3]
1968	P. Dombrowski	Definition of a linear connection in vector bundle as a parallel transport along paths $\beta \colon [a,b] \to M$ , with $a,b \in \mathbb{R}$ and $a \leq b$ , in its base $M$ . The parallel transport is defined as a mapping from the fibre over $\alpha(a)$ into the one over $\alpha(b)$ satisfying certain axioms. The theory of covariant derivatives is constructed on that base. [31, § 1]
1981	Walter Poor	A detailed axiomatic definition of a parallel transport in vector bundles. The whole theory of linear connections in such bundles is deduced on that ground. [23]

The main merit from Section 8 is a necessary and sufficient conditions when an axiomatically defined parallel transport (or a parallel transport along paths) defines a unique connection (with suitable properties) and *vice versa*. Moreover, the concepts "connection," "(axiomatically defined) parallel transport," and "parallel transport along path" (i.e., one that satisfies equations (8.4) and (8.5)) are equivalent in a sense that there are bijective mappings between the sets of these objects. However, the concept of a parallel transport admits a generalization to the one of a transport along paths and there exist transports along paths that cannot be generated by connections or axiomatically defined parallel transports.

## Bibliography

- Bernard F. Schutz. Geometrical methods of mathematical physics. Cambridge University Press, Cambridge-London-New York-New Rochelle-Melbourne-Sydney, 1982. Russian translation: Mir, Moscow, 1984.
- [2] Chris J. Isham. Modern differential geometry for physicists, volume 32 of World scientific lecture notes in physics. World scientific, Singapore, 1989.
- [3] B. A. Dubrovin, S. P. Novikov, and A. T. Fomenko. Modern geometry: methods and applications, pt. 1, volume 93 of Graduate text in Mathematics: The geometry of surfaces, transformation groups, and fields. Springer, New York, 2 edition, 1992. Translation from the original Russian ed., Moscow, Nauka, 1979.
- [4] B. A. Dubrovin, S. P. Novikov, and A. T. Fomenko. Modern geometry: methods and applications, pt. 2, volume 104 of Graduate text in Mathematics: The geometry and topology of manifolds. Springer, New York, 1985. Translation from the original Russian ed., Moscow, Nauka, 1979.
- [5] M. Göckeler and T. Schücker. Differential geometry, gauge theories, and gravity. Cambridge Univ. Press, Cambridge, 1987.
- [6] C. Nash and S. Sen. Topology and Geometry for physicists. Academic Press, London-New York, 1983.
- [7] F. W. Warner. Foundations of differentiable manifolds and Lie groups. Springer-Verlag, New York-Berlin-Heidelberg-Tokyo, 1983. Russian translation: Mir, Moscow, 1987.
- [8] D. Gromoll, W. Klingenberg, and W. Meyer. *Riemannian geometry at large*. Mir, Moscow, 1971. Russian translation from the German original: Riemannsche Geometrie im großen, Springer, Berlin-Heidelberg-New York, 1968.
- [9] F. Brickell and R. S. Clark. Differentiable manifolds. An introduction. van Nostrand Reinhold Co. Ltd., London, 1979.
- [10] R. L. Bishop and R. J. Crittenden. Geometry of Manifolds. Academic Press, New York-London, 1964.
- [11] S. Kobayashi and K. Nomizu. Foundations of Differential Geometry, volume I. Interscience Publishers, New York-London, 1963. Russian translation: Nauka, Moscow, 1981.
- [12] Y. Choquet-Bruhat et al. Analysis, manifolds and physics. North-Holland Publ. Co., Amsterdam, 1982.

- S. Sternberg. Lectures on differential geometry. Chelsea Publ. Co., New York, 1983.
   First edition: Prentice Hall, Inc. Englewood Cliffs, N.J., 1964. Russian translation: Mir, Moscow, 1970.
- [14] Serge Lang. Differential manifolds. Springer Verlag, New York, 1985. (originally published: Addison-Wesley Pub. Co, Reading, Mass. 1972).
- [15] S. Helgason. Differential Geometry, Lie Groups and Symmetric Spaces. Academic Press, New York-San Francisco-London, 1978.
- [16] W. Greub, S. Halperin, and R. Vanstone. De Rham cohomology of manifolds and vector bundles, volume 1 of Connections, Curvature, and Cohomology. Academic Press, New York and London, 1972.
- [17] C. Teleman. Elements of topology and differentiable manifolds. Mir, Moscow, 1967. In Russian. Translation from the Rumanian original: Elemente de topologie si varietāti differentiabile, Bucuresti, 1964. German translation: Grundzüge der Topologie und differenzierbare Mannigfaltigkeiten (Autorisierte und vom Autor erg. Übersetzung aus dem Rumänischen: Horst Antelmann.) Mit 22 Abbildungen, Berlin, Deutscher Verlag der Wissenschaften, 1968, 411 p.
- [18] David Lovelock and Hanno Rund. Tensors, Differential Forms, and Variational Principals. Pure and applied mathematics. Wiley-Interscience Publication, John Wiley & Sons, New York-London-Sydney-Toronto, 1975.
- [19] J. A. Schouten. Ricci-Calculus: An Introduction to Tensor Analysis and its Geometrical Applications. Springer Verlag, Berlin-Göttingen-Heidelberg, second edition, 1954.
- [20] C. W. Misner, K. S. Thorne, and J. A. Wheeler. *Gravitation*. W. H. Freeman and Company, San Francisco, 1973. Russian translation in 3 volumes: Mir, Moscow, 1977.
- [21] S. Weinberg. Gravitation and cosmology. John Wiley & Sons, Inc., New York-London-Sydney-Toronto, 1972. Russian translation: Mir, Moscow, 1975.
- [22] W. Greub, S. Halperin, and R. Vanstone. Lie groups, principal bundles, and characteristic classes, volume 2 of Connections, Curvature, and Cohomology. Academic Press, New York and London, 1973.
- [23] Walter A. Poor. Differential geometric structures. McGraw-Hill Book Company Inc., New York, 1981.
- [24] Luther P. Eisenhart. *Riemannian geometry*. Princeton Univ. Press, Princeton, second edition, 1949. First edition: 1926.
- [25] P. K. Rashevskii. Riemannian Geometry and Tensor Analysis. Nauka, Moscow, 1967. In Russian.
- [26] S. Kobayashi and K. Nomizu. Foundations of Differential Geometry, volume II. Interscience Publishers, New York-London-Sydney, 1969. Russian translation: Nauka, Moscow, 1981.
- [27] A. S. Mishchenko and A. T. Fomenko. A course on differential geometry and topology. Moscow University Press, Moscow, 1980. In Russian.
- [28] N. J. Hicks. Notes on differential geometry. D. Van Nostrand Comp., Inc., Princeton, 1965.
- [29] T. Levi-Civita. Nazione di parallelismo in una variete qualunque e consequente speticazione geometrica della curvaturre riemanniano. *Rend. Palermo*, 42:73–205, 1917. In Italian.

- [30] Ulo G. Lumiste. To the foundations of the global connection theory. In Scientific writings of the Tartu state university, number 150 in Works on mathematics and mechanics, IV, pages 69–107. University of Tartu, Tartu, 1964. In Russian.
- [31] Peter Dombrowski. Krümmungsgrößen gleichungsdefinierter Untermannigfaltigkeiten Riemannscher Mannigfaltigkeiten. Mathematische Nachrichten, 38(3/4): 133–180, 1968.
- [32] Ülo G. Lumiste. Connection theory in fibre bundles. In Science review, Mathematics: Algebra. Topology. Geometry. 1969, pages 123–168. VINITI, Moscow, 1971. In Russian.
- [33] P. Nikolov. On the correspondence between infinitesimal and integral description of connections. Internal Report, IC/81/196, ICTP, Trieste, 1981.
- [34] Ph. Hartman. Ordinary Differential Equations. John Wiley & Sons, New York-London-Sydney, 1964.
- [35] Sawa S. Manoff. Geodesic and autoparallel equation over differentiable manifolds. Int. J. Mod. Phys. A, 11(21):3849–3874, 1996.
- [36] F. W. Hehl, J. D. McCrea, E. W. Mielke, and Y. Ne'eman. Metric-affine gauge theory of gravity: field equations, Noether identities, world spinors, and breaking of dilation invariance. *Phys. Rep.*, 258(1 & 2):1–171, July 1995.
- [37] F. Gronwald and F. W. Hehl. On gauge aspects of gravity. In Bergmann P. G., de Sabbata V., and Treder H.-J., editors, *Quantum Cosmology*, Proc. of the 14-th Course of the School of Cosmology and Gravitation, pages 148–198, Erice, Italy, May 1995, 1996. World Scientific, Singapore. http://arXiv.org e-Print archive, E-print No. gr-qc/9602013, 1996.
- [38] M. M. Postnikov. Introduction to the Morse theory. Nauka, Moscow, 1971. In Russian.
- [39] Josef A. Wolf. Spaces of constant curvature. Univ. of California, Berkley, California, 1972. Russian translation: Nauka, Moscow, 1982.
- [40] K. Kodaira. Complex manifolds and deformations of complex structures, volume 283 of Grundlehren der mathematischen Wissenschaften. Springer, New York, 1986.
- [41] A. P. Norden. Spaces with affine connection. Nauka, Moscow, second edition, 1976. In Russian.
- [42] Mathematical encyclopedia, chief editor Vinogradov I. M., volume 3, Moscow, 1982. Sovetskaya Entsiklopediya (Soviet encyclopedia). In Russian. English translation in: Encyclopedia of Mathematics, vol. I–VI, Kluwer, 1995.
- [43] J. A. Schouten and D. J. Struik. Einführung in die neueren Methoden der Differentialgeometrie, volume I. Noordhoff, Groningen, 1935. Vol. I by J. A. Schouten, Russian translation: Moscow, 1939.
- [44] M. J. Pinl. Geodesic coordinates and rest systems for general linear connections. Duke Mathematical Journal, 18:557–562, 1951.
- [45] A. M. L. Messiah. *Quantum mechanics*, volume II. North Holland, Amsterdam, 1962. Russian translation: Nauka, Moscow, 1979.
- [46] Physical encyclopedia, chief editor Prohorov A. M., volume 3, Moscow, 1992. Bol'shaya Rossiĭskaya Entsiklopediya (Big Russian encyclopedia). In Russian.
- [47] Agapitos Hatzinikitas. A note on Riemann normal coordinates. http://arXiv.org e-Print archive, E-print No. hep-th/0001078, 2000.

- [48] B. Riemann. Über die Hypothesen welche de Geometrie zugrunde liegen (On the hypotheses underlying the geometry). *Götingen Abh.*, 13:1–20, 1868. Habilitationsschrift (Ph.d. thesis), 1854. In German.
- [49] R. Bellman. Introduction to matrix analysis. McGraw-Hill book comp., New York-Toronto-London, 1960. Russian translation: Nauka, Moscow, 1978.
- [50] G. D. Birkhoff. *Relativity and modern physics*. Harvard Univ. Press, Cambridge, Mass., 1923.
- [51] V. A. Fock. Theory of space, time and gravitation. Nauka, Moscow, 1955, 1961 (second ed.). In Russian. English translation: Pergamon Press, New York-London, 1959; German Translation: V. Fock, Theorie von Raum, Zeit und Gravitation, Akademie-Verlag, Berlin, 1960.
- [52] E. Fermi. Sopra i fenomeni che avvengono in vicinonza di una linear oraria (On phenomena near a world line). Atti R. Accad Lincei Rend., Cl. Sci. Fis. Mat. Nat., 31(1):21–23, 51–52, 1922. In Italian. Russian translation in [159, pp. 64–71].
- [53] Luther P. Eisenhart. Non-Riemannian geometry, volume VIII of Colloquium Publications. American Mathematical Society, New York, 1927.
- [54] T. Levi-Civita. Absolute differential calculus. Blacckie, London, 1927.
- [55] L. O Raifeartaigh. Fermi coordinates. Proceedings of the Royal Irish Academy, 59 Sec. A(2):15–24, 1958.
- [56] Keith Burns and Marian Gidea. Differential geometry and topology with a view to dynamical systems. Champman & Hall/CRC, Roca Raton-London-New York-Singapore, 2005.
- [57] G. Korn and T. Korn. *Mathematical Handbook*. Mc Graw-Hill Book Company, New York, second edition, 1968. Russian translation: Nauka, Moscow, 1973,.
- [58] Richard C. Tolman. *Relativity, thermodynamics and cosmology.* Clarendon Press, Oxford, 1969. Russian translation: Nauka, Moscow, 1974.
- [59] Physical encyclopedia, chief editor Prohorov A. M., volume 1–5, Moscow, 1988, 1990, 1992, 1994, 1998. Sovetskaya Entsiklopediya (Soviet encyclopedia) and Bol'shaya Rossiĭskaya Entsiklopediya (Big Russian encyclopedia). In Russian.
- [60] R. K. Sachs and H. Wu. General Relativity for Mathematicians. Springer-Verlag, New York-Heidelberg-Berlin, 1977.
- [61] C. Møller. The theory of relativity. Clarendon Press, Oxford, second edition, 1972. Russian translation: Atomizdat, Moscow, 1975.
- [62] J. L. Synge. *Relativity: The General Theory.* North-Holland Publ. Co., Amsterdam, 1960. Russian translation: IL (Foreign Literature), Moscow, 1963.
- [63] N. N. Bogolyubov and D. V. Shirkov. Introduction to the theory of quantized fields. Nauka, Moscow, third edition, 1976. In Russian. English translation: Wiley, New York, 1980.
- [64] J. D. Bjorken and S. D. Drell. *Relativistic quantum mechanics*, volume 1 and 2. McGraw-Hill Book Company, New York, 1964, 1965. Russian translation: Nauka, Moscow, 1978.
- [65] A. G. Kurosh. *Higher algebra*. Mir Publ., Moscow, 1980. Translation from the second Russian edition, Nauka, Moscow, 1975.
- [66] Mathematical encyclopedia, chief editor Vinogradov I. M., volume 1, Moscow, 1977. Sovetskaya Entsiklopediya (Soviet encyclopedia). In Russian. English translation in: Encyclopedia of Mathematics, vol. I–VI, Kluwer, 1995.

- [67] S. Chandrasekhar. The mathematical theory of black holes. Clarendon Press, Oxford, 1983. Russian translation in two volumes: Mir, Moscow, 1986.
- [68] L. D. Landau and E. M. Lifshitz. Classical theory of fields, volume II of Course of theoretical physics. Pergamon Press, Oxford, 5 edition, 1967. Translation from Russian, Nauka, Moscow, 1973.
- [69] Mathematical encyclopedia, chief editor Vinogradov I. M., volume 1–5, Moscow, 1977–1985. Sovetskaya Entsiklopediya (Soviet encyclopedia). In Russian. English translation: Encyclopedia of Mathematics, vol. I–VI, Kluwer, 1995.
- [70] J. A. Schouten. Tensor analysis for physicists. Clarendon Press, Oxford, 1951. Russian translation: Nauka, Moscow, 1965.
- [71] R. Torretti. *Relativity and Geometry*. Pergamon Press, Oxford-New York-Toronto-Sydney-Paris-Frankfurt, 1983.
- [72] O. Veblen. Normal Coordinates for the Geometry of Paths. Proc. Nat. Acad., 8:192–197, 1922.
- [73] O. Veblen and T. Y. Thomas. Projective normal coordinates for the geometry of paths. Proc. Nat. Acad., 11:204–207, 1925.
- [74] T. Y. Thomas. Note on the projective geometry of paths. Bull. Am. Math. Soc., 31:318–322, 1925.
- [75] E. Bortolotti. Sulla geometria della varietá a connessione affine. teoria invariantiva delle transformazione che conservano il parallelismo. Ann. di Mat., 8:53–101, 1930.
- [76] Bozhidar Z. Iliev. Normal frames and the validity of the equivalence principle: II. The case along paths. *Journal of Physics A: Mathematical and General*, 30(12):4327–4336, 1997. DOI No.10.1088/0305-4470/30/12/019 http://arXiv.org e-Print archive, E-print No. gr-qc/9709053, September 1997.
- [77] Laurent Schwartz. Analyse mathématique, volume I. Hermann, Paris, 1967. In French; Russian translation: Mir, Moscow, 1972.
- [78] J. Dieudonné. Foundations of modern analysis. Academic Press, New York, 1960.
- [79] Walter Rudin. Principles of mathematical analysis. McGraw-Hill Book Company, New York, 1964. Bulgarian translation: Nauka i izkustvo, Sofia, 1973.
- [80] Bozhidar Z. Iliev. Normal frames and the validity of the equivalence principle: I. Cases in a neighborhood and at a point. *Journal of Physics A: Mathematical and General*, 29(21):6895–6901, 1996. DOI No. 10.1088/0305-4470/29/21/020 http://arXiv.org e-Print archive, E-print No. gr-qc/9608019, August 1998.
- [81] T. Levi-Civita. Sur l'écart géodésique. Math. Ann., 97:291-320, 1926.
- [82] F. R. Gantmacher. The theory of matrices, volume two. Chelsea Pub. Co., New York, N.Y., 1960 (reprinted 1964). Translation from Russian. The right English transliteration of the author's name is Gantmakher.
- [83] Bozhidar Z. Iliev. Normal frames and the validity of the equivalence principle: III. The case along smooth maps with separable points of self-intersection. *Journal of Physics A: Mathematical and General*, 31(4):1287–1296, January 1998. DOI No. 10.1088/0305-4470/31/4/016 http://arXiv.org e-Print archive, E-print No. gr-qc/9805088, May 1998.
- [84] Bozhidar Z. Iliev. Special bases for derivations of tensor algebras. I. Cases in a neighborhood and at a point. JINR Communication E5-92-507, Dubna, 1992. 19 pp.

http://arXiv.org e-Print archive, E-print No. math.DG/0303373, March 2003. See also [80].

- [85] Bozhidar Z. Iliev. Special bases for derivations of tensor algebras. II. Case along paths. JINR Communication E5-92-508, Dubna, 1992. 16 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0304157, April 2003. See also [76].
- [86] Bozhidar Z. Iliev. Special bases for derivations of tensor algebras. III. Case along smooth maps with separable points of selfintersection. JINR Communication E5-92-543, Dubna, 1992. 15 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0305061, May 2003. See also [83].
- [87] Bozhidar Z. Iliev. Normal frames for derivations and linear connections and the equivalence principle. Journal of Geometry and Physics, 45(1-2):24-53, February 2003. DOI No. 10.1016/S0393-0440(02)00119-5

http://arXiv.org e-Print archive, E-print No. hep-th/0110194, October 2001.

- [88] R. Hermann. Geometry, physics, and systems. Marcel Dekker, Inc., New York, 1973.
- [89] K. Yano. The theory of Lie derivatives and its applications. North-Holland Publ. Co., Amsterdam, 1957.
- [90] F. R. Gantmacher. The theory of matrices, volume one. Chelsea Pub. Co., New York, N.Y., 1960. Translation from Russian. The right English transliteration of the author's name is Gantmakher.
- [91] Mathematical encyclopedia, chief editor Vinogradov I. M., volume 4, Moscow, 1984. Sovetskaya Entsiklopediya (Soviet encyclopedia). In Russian. English translation in: Encyclopedia of Mathematics, vol. I–VI, Kluwer, 1995.
- [92] G. Magnano. Are there metric theories of gravity other than general relativity? http://arXiv.org e-Print archive, E-print No. gr-qc/9511027, 1995.
- [93] N. A. Chernikov. Two connections in the gravity theory. JINR preprint E2-96-250, Dubna, Russia, 1996.
- [94] M. N. Tentyukov. Gravitational theory with the dynamical affine connection. JINR preprint E2-92-491, Dubna, Russia, 1992.
- [95] Sawa S. Manoff. Kinematics of vector fields, pages 61–113. Complex Structure and Vector Fields. World Scientific Publ., Singapore, 1995. Proc. of "Int. Workshop on Complex Structure and Vector Fields", August 14–17, 1994.
- [96] Sawa S. Manoff. Lagrangian theory of tensor fields over spaces with contravariant and covariant affine connection and metrics and its application to Einstein's theory of gravitation in  $\overline{V}_4$  spaces. Acta Aplicandae Mathematicae, 55(1):51–125, 1999.
- [97] Bozhidar Z. Iliev. Is the principle of equivalence a principle? Journal of Geometry and Physics, 24(3):209–222, 1998. DOI No. 10.1016/S0393-0440(97)00011-9 http://arXiv.org e-Print archive, E-print No. gr-qc/9806062, June 1998.
- [98] R. Dandoloff and W. J. Zakrzewski. Parallel transport along a space curve and related phases. J. Phys. A: Math. Gen., 22:L461–L466, 1989.
- [99] N. V. Mitskevich. Physical fields in general theory of relativity. Nauka, Moscow, 1969. In Russian.

- [100] D. Hartley. Normal frames for non-Riemanian connections. Class. Quantum Grav., 12(11):L103–L105, November 1995. http://arXiv.org e-Print archive, E-print No. gr-qc/9510013, 1995.
- [101] Bozhidar Z. Iliev. Linear transports along paths in vector bundles. II. Some applications. JINR Communication E5-93-260, Dubna, 1993. 24 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0412010, December 1, 2004.
- [102] Bozhidar Z. Iliev. Normal frames and linear transports along paths in vector bundles. JP Journal of Geometry and Topology, 5(3):187–250, 2005. http://arXiv.org e-Print archive, E-print No. gr-qc/9809084, September 1998 (last revision: March 2005).
- [103] Bozhidar Z. Iliev. Linear transports along paths in vector bundles. III. Curvature and torsion. JINR Communication E5-93-261, Dubna, 1993. 12 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0502008, February 1, 2005.
- [104] Bozhidar Z. Iliev. Linear transports along paths in vector bundles. V. Properties of curvature and torsion. JINR Communication E5-97-1, Dubna, 1997. 11 pp. http://arXiv.org e-Print archive, E-print No. dg-ga/9709017, September 1997.
- [105] Bozhidar Z. Iliev. Parallel transports in tensor spaces generated by derivations of tensor algebras. JINR Communication E5-93-1, Dubna, 1993. 19 pp http://arXiv.org e-Print archive, E-print No. math.DG/0502008, February 1, 2005.
- [106] N. Steenrod. The topology of fibre bundles. Princeton Univ. Press, Princeton, ninth edition, 1974.
- [107] D. Husemoller. Fibre bundles. McGraw-Hill Book Co., New York-St. Louis-San Francisco-Toronto-London-Sydney, 1966. Russian translation: Mir, Moscow, 1970.
- [108] Sze-Tsen Hu. Homotopy Theory. Academic Press, New York-London, 1959.
- [109] I. M. James. General topology and homotopy theory. Springer-Verlag., New York-Berlin-Heidelberg-Tokio, 1984.
- [110] W. Greub, S. Halperin, and R. Vanstone. Connections, Curvature, and Cohomology, volume 1,2,3. Academic Press, New York and London, 1972, 1973, 1976.
- [111] Bozhidar Z. Iliev. Linear transports of tensors along curves: General S-transport. Comptes rendus de L'Academie bulgare des Sciences, 40(7):47–50, 1987.
- [112] Bozhidar Z. Iliev. General linear transport (I-transport). Comptes rendus de L'Academie bulgare des Sciences, 40(8):45–48, 1987.
- [113] Bozhidar Z. Iliev. Linear transports along paths in vector bundles. I. General theory. JINR Communication E5-93-239, Dubna, 1993. 22 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0411023, November 1, 2004.
- [114] Bozhidar Z. Iliev. Transports along paths in fibre bundles. General theory. JINR Communication E5-93-299, Dubna, 1993. 26 pp. http://arXiv.org e-Print archive, E-print No. math.DG/0503005, March 1, 2005.
- [115] Bozhidar Z. Iliev. Transports along paths in fibre bundles. II. Ties with the theory of connections and parallel transports. JINR Communication E5-94-16, Dubna, 1994. 24 pp.

http://arXiv.org e-Print archive, E-print No. math.DG/0503006, March 1, 2005.

[116] D. J. Saunders. The geometry of jet bundles. Cambridge Univ. Press, Cambridge, 1989.

- [117] R. Hermann. Vector bundles in mathematical physics, volume I. W. A. Benjamin, Inc., New York, 1970.
- [118] Bozhidar Z. Iliev. Links between connections, parallel transports, and transports along paths in differentiable fibre bundles. International Journal of Geometric Methods in Modern Physics, 2(5):823–838, 2005.
  DOI No: 10.1142/S0219887805000806
  Talk at the International Workshop on "Advanced Geometric Methods in Physics", Florence, Italy, 14 18 April, 2005.
  http://arXiv.org e-Print archive, E-print No. math.DG/0504010, April 1, 2005.
- [119] S. W. Hawking and G. F. R. Ellis. The large scale structure of space-time. Cambridge Univ. Press, Cambridge, 1973.
- [120] B. B. Walwadkar. Truesdell transport in general relativity. General Relativity and Gravitation, 15(12):1107–1116, 1983.
- [121] B. B. Walwadkar and K. V. Virkar. Truesdell invariance in relativistic electromagnetic fields. *General Relativity and Gravitation*, 16(1):1–7, 1984.
- [122] L. Radhakrishna, L. N. Katkar, and T. H. Date. Jaumann transport in relativistic continuum mechanics. *General Relativity and Gravitation*, 13(10):939–946, 1981.
- [123] Bozhidar Z. Iliev. Transports along maps in fibre bundles. JINR Communication E5-97-2, Dubna, 1997. 19 pp. http://arXiv.org e-Print archive, E-print No. dg-ga/9709016, September 1997.

[124] James D. Stasheff. "Parallel" transport in fibre spaces. Reimpreso del boletin de la sociedad matematica mexicana, Sociedad Matematica Mexicana, Printed in USA, 1966. pages 68–84.

- [125] Pham Mau Quan. Introduction a la géométrie des variétés différentiables. Dunod, Paris, 1969.
- [126] Richard G. Cooke. Infinite matrices and sequence spaces. London, 1950. Reprinted: Dover Publications, New York, 1966.
- Bozhidar Z. Iliev. Fibre bundle formulation of nonrelativistic quantum mechanics.
   I. Introduction. The evolution transport. Journal of Physics A: Mathematical and General, 34(23):4887–4918, 2001. DOI No. 10.1088/0305-4470/34/23/308 http://arXiv.org e-Print archive, E-print No. quant-ph/9803084, March 1998.
- Bozhidar Z. Iliev. Fibre bundle formulation of nonrelativistic quantum mechanics.
   II. Equations of motion and observables. Journal of Physics A: Mathematical and General, 34(23):4919–4934, 2001. DOI No. 10.1088/0305-4470/34/23/309 http://arXiv.org e-Print archive, E-print No. quant-ph/9804062, April 1998.
- [129] Bozhidar Z. Iliev. Fibre bundle formulation of nonrelativistic quantum mechanics. III. Pictures and integrals of motion. *Journal of Physics A: Mathematical and General*, 34(23):4935–4950, 2001. DOI No. 10.1088/0305-4470/34/23/310 http://arXiv.org e-Print archive, E-print No. quant-ph/9806046, June 1998.
- Bozhidar Z. Iliev. Fibre bundle formulation of nonrelativistic quantum mechanics. IV. Mixed states and evolution transport's curvature. International Journal of Modern Physics A, 17(2):229-243, 2002. DOI No. 10.1142/S0217751X02005669 http://arXiv.org e-Print archive, E-print No. quant-ph/9901039, January 1999.
- [131] Bozhidar Z. Iliev. Fibre bundle formulation of nonrelativistic quantum mechanics. V. Interpretation, summary, and discussion. *International Journal of Modern*

*Physics A*, 17(2):245–258, 2002. DOI No. 10.1142/S0217751X02005712

http://arXiv.org e-Print archive, E-print No. quant-ph/9902068, February 1999.

- [132] O. Ya. Viro and D. B. Fuks. Introduction to homotopy theory. In *Reviews of science and technique*, volume 24 of *sec. "Modern problems in mathematics*", pages 6–121. VINITI, Moscow, 1988. In Russian.
- [133] P. A. M. Dirac. The principles of quantum mechanics. Oxford at the Clarendon Press, Oxford, fourth edition, 1958. Russian translation in: P. Dirac, Principles of quantum mechanics, Moscow, Nauka, 1979.
- [134] V. A. Fock. Fundamentals of quantum mechanics. Mir Publishers, Moscow, 1978. Russian edition: Nauka, Moscow, 1976.
- [135] A. M. L. Messiah. *Quantum mechanics*, volume I and II. Interscience, New York, 1958. Russian translation: Nauka, Moscow, 1978 (vol. I) and 1979 (vol. II).
- [136] J. von Neumann. Mathematical foundations of quantum mechanics. Princeton Univ. Press, Princeton, New Jersey, 1955.
- [137] Bozhidar Z. Iliev. Connection theory on differentiable fibre bundles: A padagogical introduction. http://arXiv.org e-Print archive, E-print No. math-ph/0510004, October 1, 2005, 2005.
- [138] Maido Rahula. New problems in differential geometry, volume 8 of Series on Soviet and East European Mathematics. World Scientific, Singapore-New Jersey-London-Hong Kong, 1993. 172 p.
- Bozhidar Z. Iliev. Normal frames for general connctions on differentiable fibre bundles. Journal of Geometry and Physics, 56(5):780-812, 2006.
   DOI No. 10.1016/j.geomphys.2005.04.018
   http://arXiv.org e-Print archive, E-print No. math.DG/0405004, May 1, 2004.
- [140] S. Kobayashi. Theory of connections. Ann. mat. pure ed appl., 43:119–194, 1957.
- [141] K. Yano and M. Kon. Structures on Manifolds, volume 3 of Series in Pure Mathematics. World Scientific Publ. Co., Singapore, 1984.
- [142] R. Sulanke and P. Wintgen. Differential geometry and fibre bundles. Mir, Moscow, 1975. In Russian. Translation from the German original: Differentialgeometrie und Faserbündel, VEB Deutscher Verlag der Wissenschaften, Berlin, 1972.
- [143] A. S. Mishchenko. Vector fibre bundles and their applications. Nauka, Moscow, 1984. In Russian.
- [144] M. F. Atiyah. K-theory. Harvard Univ, Cambridge, Mass., 1965.
- [145] I. Tamura. Topology of foliations. Mir, Moscow, 1979. Russian translation from 1976 Japanese original.
- [146] L. Mangiarotti and G. Sardanashvily. Connections in classical and quantum field theory. World Scientific, Singapore-New Jersey-London-Hong Kong, 2000.
- [147] Ülo G. Lumiste. Homogeneous fibre bundles with connection and their immersions. In Works of the Geometrical seminar, volume 1, pages 191–237. VINITI, Moscow, 1966. In Russian.
- [148] B. Durhuus and J. M. Leinaas. On the loop space formulation of gauge theories. Preprint TH 3110, CERN, Geneva, 1981.
- [149] O. M. Khudaverdian and A. S. Schwarz. A new comments on the string representation of gauge fields. *Phys. Lett. B*, 91(1):107–110, 1980.

- [150] N. M. Ostianu, V. V. Rizhov, and P. I. Shveikin. Article on the scientific investigations of German Fedorovich Laptev. In Works of the Geometrical seminar, volume 4, pages 7–70. VINITI, Moscow, 1973. In Russian.
- [151] S. Kobayashi and K. Nomizu. Foundations of Differential Geometry, volume I and II. Interscience Publishers, New York-London-Sydney, 1963 and 1969. Russian translation: Nauka, Moscow, 1981.
- [152] Ivan Kolář, Peter W. Michor, and Jan Slovák. Natural operations in differential geometry. Springer, Berlin - Heidelberg, 1993.
- [153] M. Spivak. A comprehensive introduction to differential geometry, volume 1. Publish or Perish, Boston, 1970.
- [154] B. A. Dubrovin, S. P. Novikov, and A. T. Fomenko. Modern geometry: methods and applications, pt. 3, volume 124 of Springer series in Soviet mathematics: Introduction to homotopy theory. Springer, New York, 1990. Translation from the original Russian ed., Moscow, Nauka, 1984.
- [155] T. Eguchi, P. B. Gilkey, and A. J. Hanson. Gravitation, gauge theories and differential geometry. *Physics reports*, 66(6):213–393, December 1980.
- [156] D. V. Alekseevskii, A. M. Vinogradov, and V. V. Lychagin. Basic ideas and concepts of differential geometry. In *Science and technology reviews*, volume 28, of *Modern problems in Mathematics, Fundamental directions. Geometry-1*, pages 5–298. VINITI, Moscow, 1988. In Russian.
- [157] Andre Lichnerowicz. Global theory of connections and holonomy groups. Foreign literature, Moscow, 1960. Russian translation from the French original: Théorie globale des connexions et des groupes d'holonomie, Edizioni Cremonese, Roma, 1955.
- [158] A. Liebermann. Parallélismes. Journal of differential geometry, pages 511–519, 1973.
- [159] Enrico Fermi. Scientific papers, 1921–1938. Italy, volume I. Nauka, Moscow, 1971. In Russian.

# Subject List of Symbols

The following table contains a list of the main symbols which consistently refer to the some concept and frequently are employed. The page numbers denote the page where the symbols first appear. A more or less alphabetically ordered list of the same symbols is presented in the Notation index, beginning on page 423.

Symbol	Explanation	Page
Set theory		
Ø	empty set	5
$\in$	belongs to, included in	4
Э	which contains	7
$\{x:P\}$	set whose elements have the property ${\cal P}$	5
$\{x P\}$	the same as $\{x: P\}$	5
$\{a, b, c, \ldots\}$	set with elements $a, b, c, \ldots$	4
$\subset$	contained in, proper subset	8
$\supset$	proper superset (overset)	27
$\subseteq$	contained in or equal, proper subset or equal	5
$\supseteq$	proper superset (overset) or equal	177
U	union (cup) sing	5
$\cap$	intersection (cap) sign	5
=	equals to, equality sign	4
:= or $=:$	equals by definition: a quantity sitting from the side	4
	of the colon being defined	
\	subtraction of sets; $A \setminus B := \{a   a \in A, a \notin B\}$	27
$ _A$ or $ A$	restriction to a set $A$	8
×	Cartesian (direct) product	4

continued from the previous page

Symbol	Explanation	Page
$\times^k A$	$\begin{array}{l} A \times \cdots \times A \text{ where the set } A \text{ is taken } k \text{ times, } k \in \mathbb{N} \cup \{0\}; \\ \times^{0} A := \varnothing \end{array}$	4
$\iff$ or iff	if and only if, logical equivalence	34
$\rightarrow$	mapping sign	5
$f\colon A\to B$	mapping $f$ with domain $A$ and range $B$	5
$\mapsto$	maps to sign	6
$f\colon x\mapsto y$	mapping $f$ assigning to $x$ the value $y$	6
$f^{-1}$	inverse mapping to a mapping $f$	5
0	sign of the composition of mappings; $(f \circ g)(p) := f(g(p))$	5
$id_A$	identity mapping of set A; $id_A : a \mapsto a$ for all $a \in A$	5
Linear/vector spa	ces	
N	natural numbers, integers	4
$\mathbb{R}$	real numbers, (field of) reals, real line	4
$\mathbb{C}$	complex numbers, (field of) complex, complex plane	4
$\mathbb{K}$	$\mathbb{R} \text{ or } \mathbb{C}$	4
E	1-dimensional Euclidean space	34
$\mathbb{R}^{n}$	<i>n</i> -th Cartesian power of $\mathbb{R}$ , $n \in \mathbb{N}$	4
$\mathbb{C}^n$	<i>n</i> -th Cartesian power of $\mathbb{C}$ , $n \in \mathbb{N}$	4
$\mathbb{K}^n$	<i>n</i> -th Cartesian power of $\mathbb{K}$ , $n \in \mathbb{N}$ ; $\mathbb{R}^n$ or $\mathbb{C}^n$	4
$\mathbb{E}^n$	<i>n</i> -th Cartesian power of $\mathbb{E}, n \in \mathbb{N}$	34
J	real interval of arbitrary type (open or closed from one or both ends); the same is the meaning of $J$ with some attached to it indices if the opposite is not stated explicitly	6
[a,b]	close real interval with left (resp. right) end a (resp. b), with $a, b \in \mathbb{R}$ and $a \leq b$	27
(a,b]	closed from right and opened from left real interval with left (resp. right) end a (resp. b), with $a, b \in \mathbb{R}$ and $a < b$	27
[a,b)	closed from left and opened from right real interval with left (resp. right) end a (resp. b), with $a, b \in \mathbb{R}$	27

and a < b

continued from the previous page

Symbol	Explanation	Page
(a,b)	open real interval with left (resp. right) end $a$ (resp. 2	
	b), with $a, b \in \mathbb{R}$ and $a < b$	
$J^n$	neighborhood in $\mathbb{R}^n, n \in \mathbb{N}$	161
$\otimes$	tensor product sign	17
$\otimes^k V$	$V \otimes \cdots \otimes V$ where the vector space $V$ over $\mathbb{K}$ is taken $k$ times, $k \in \mathbb{N} \cup \{0\}; \otimes^0 V := \mathbb{K}$	17
$A^*$	object dual to an object $A$ , $A$ being a vector or vector space	15
$i, j, k, \dots$	indices running from 1 to the dimension of a vector space	5
$\{e_i\}$	basis in a vector space	83
$A_{(ij)}$	symmetrization over the indices $i$ and $j$ ; equals to $\frac{1}{2}(A_{ij} + A_{ji})$ (see the list of conventions, page xii)	24
$A_{[ij]}$	antisymmetrization over the indices $i$ and $j$ ; equals to $\frac{1}{2}(A_{ij} - A_{ji})$ (see the list of conventions, page xii)	24
$[A,B]_{\_}$	commutator of operators A and B on a linear space; equals to $A \circ B - B \circ A$	13
Matrices		
1	identity (unit) matrix of the corresponding size	89
$1_n$	the identity (unit) $n \times n$ , $n \in \mathbb{N}$ , matrix	97
$\delta_{ij}^{i} = \delta^{ij} = \delta_i^j$	Kronecker delta symbol(s), $\delta_i^j = 1$ for $i = j$ and $\delta_i^j = 0$ for $i \neq j$	15
$A = [A_i^j]$	square matrix A with elements $A_i^j$	11
$A = [A_i^j]_{i,j=1}^n$	the same, but the range of the indexes $i$ and $j$ is written explicitly	11
$\operatorname{diag}(a_1,\ldots,a_n)$	diagonal matrix with diagonal elements $a_1, \ldots, a_n$ ; equals to $[\delta_{ij}a_j]_{i,j=1}^n$	36
$\det(A) \equiv \det A$	determinant of square matrix A	36
$A^{-1}$	inverse matrix of non-degenerate square matrix A	11
$A^{ op}$	transposed matrix of a matrix $A = [A_{ij}], (A^{\top})_{ij} :=$	47
	$A_{ji}$ ; for $A = [A^{j}_{i}]$ , we put $(A^{\top})_{j}^{k} := A^{j}_{k}$ , i.e., the superscript is considered as 'first' index, which numbers the rows of a matrix	

continued from	the previous page	
Symbol	Explanation	Page
$\operatorname{GL}(n,\mathbb{K})$	the set of $n \times n, n \in \mathbb{N}$ , invertible matrices on a field $\mathbb{K}$	82
N		

#### Manifolds

М	differentiable manifold	4
$\dim M \equiv \dim(M)$	dimension of manifold $M$	5
$\dim_{\mathbb{C}} M$	dimension of a complex manifold $M; \dim_{\mathbb{C}} M := \dim M$	5
$\dim_{\mathbb{R}} M$	the real dimension of manifold $M$ : $\dim_{\mathbb{R}} M = \dim M$ if $M$ is real and $\dim_{\mathbb{R}} M = 2 \dim M = 2 \dim_{\mathbb{C}} M$ if $M$ is complex.	5
U	subset of $M$ , often an open set (neighborhood)	5
N	submanifold of $M$	7
$T_p(M)$	space tangent to $M$ at $p \in M$	9
$T_p^*(M)$	space cotangent to $M$ at $p \in M$	15
$\dot{T}(M)$	bundle space tangent to $M$ ; $T(M) := \bigcup_{p \in M} T_p(M)$	11
$T^*(M)$	bundle space cotangent to $M; T^*(M) := \bigcup_{p \in M} T^*_p(M)$	70
$T_{p_s}^{r}(M)$	tensor space of type $(r, s), r, s \in \mathbb{N} \cup \{0\}$ , over $p \in M$	17
$T^r_s(M)$	tensor bundle space of type	223
	$(r,s), T^r_s(M) := \bigcup_{p \in M} T^r_{p_s}(M)$	
$\lambda, \mu,  u, \dots$	indices running from 1 to the dimension of a differen- tiable manifold (see the list of conventions, page xi)	247
$[\cdot, \cdot]_{}$	commutator (of vector fields); $[X, Y]_{-} := X \circ Y - Y \circ X$	13
$\{r^i\}$	standard Cartesian coordinates on $\mathbb{K}^{\dim M}$	5
$\{E_i _p\}$	basis in $T_p(M)$	11
$\{E^i _p\}$	basis in $T_p^*(M)$ dual to $\{E_i _p\}; E^i _p := (E_i _p)^*$	15
$\{E_i\}$	basis in $T(M)$ ; frame on $U \subseteq M$	12
$\{E^i\}$	basis in $T^*(M)$ dual to $\{E_i\}$ ; coframe on $U \subseteq M$ ,	16
	$E^i := (E_i)^*$	
$C^i_{jk}$	structure functions of a frame $\{E_i\}; [E_j, E_k]_{-} =: C^i_{jk} E_i$	26
$(U, \varphi)$	(local) chart with domain $U$ and homomorphism $\varphi$	5
$\{\varphi_i\}$	(local) coordinate system assigned to a chart $(U,\varphi)$	5
$\left\{ \frac{\partial}{\partial x^i} \Big _p \right\}$	coordinate basis in $T_p(M), p \in M$	10

continued from the previous page

Symbol	Explanation	Page
$\left\{\frac{\partial}{\partial x^i}\right\}$	coordinate frame on $U \subseteq M$	12
$\left\{ \mathbf{d}x^i _p \right\}$	coordinate (co)basis in $T_p^*(M), p \in M$ dual to $\left\{ \frac{\partial}{\partial x^i} \Big _p \right\}$	15
$\left\{ \mathrm{d}x^{i}\right\}$	coordinate coframe on $U \subseteq M$ dual to $\left\{\frac{\partial}{\partial r^i}\right\}$	16
$C^k$ mapping	mapping having continuous derivative(s) up to k-th order including, $k \in \mathbb{N} \cup \{0, \infty, \omega\}$	7
$f_*$	differential (induced tangent) mapping of a differentiable mapping $\boldsymbol{f}$	13
$\mathfrak{F}(U)$	the algebra of functions on $U \subseteq M$	8
$\mathfrak{F}^r(U)$	the algebra of $C^r$ functions on $U \subseteq M$	8
$\mathfrak{X}(U)$	the $\mathfrak{F}(U)$ -module of vector fields on $U \subseteq M$	11
$\mathfrak{X}^{r}(U)$	the $\mathfrak{F}^r(U)$ -module of $C^r$ vector fields on $U \subseteq M$	13
$\mathfrak{T}^r_s(U)$	the $\mathfrak{F}(U)\text{-module}$ of tensor fields of type $(r,s)$ on $U\subseteq M$	18
$\mathfrak{T}^{r;k}_s(U)$	the $\mathfrak{F}^k(U)\text{-module}$ of $C^k$ tensor fields of type $(r,s)$ on $U\subseteq M$	19
$\boldsymbol{T}_p(M)$	the tensor algebra at $p \in M$	17
T(U)	the algebra of tensor fields on $U \subseteq M$	19
$T^{r}(U)$	the algebra of $C^r$ tensor fields on $U \subseteq M$	19
$C_s^r$	contraction operator of type $(r, s)$ , acting on the $r^{\text{th}}$ superscript and $s^{\text{th}}$ subscript	19
C	contraction operator of arbitrary type	19
$f_{,i}$	the action of a basic vector field $E_i$ on $f \in \mathfrak{F}^1(M)$ ; equals to $E_i(f)$ (in coordinate frame coincides with the $i^{\text{th}}$ partial derivative of $f$ )	26
g	Riemannian metric	34
$\gamma\colon J\to M$	path in $M$	9
$\dot{\gamma}(s)$	the vector tangent to a $C^1$ path $\gamma$ at parameter value $s \in J, \dot{\gamma}(s) \in T_{\gamma(s)}(M)$	9
$\dot{\gamma}$	the vector field tangent to $C^1$ injective path $\gamma$ , i.e., $\dot{\gamma}: \gamma(s) \mapsto \dot{\gamma}(s)$ ; the tangent lifting of arbitrary $C^1$ path $\gamma$ , i.e., $\dot{\gamma}: s \mapsto \dot{\gamma}(s)$	11
$\Delta$	distribution on manifold	338
$\Delta_p$	the value of $\Delta$ at $p \in M$ ; $\Delta_p \subseteq T_p(M)$	338
$\mathcal{L}_X$	Lie derivative along vector field $X$	144

Symbol	Explanation	Page
Linear connection	s on manifolds	
$\nabla$	linear connection; covariant (absolute) differential	21
$\nabla_X$	covariant derivative along vector field $X$	21
Т	torsion, torsion tensor field (operator)	25
R	curvature, curvature tensor field (operator)	25
$\Gamma^{i}_{\ jk}$	(local) coefficients of linear connection in a (local) frame $\{E_i\}$	22
$\Gamma_k := \left[\Gamma^i_{\ jk}\right]_{i,j=1}^{\dim M}$	matrices of the coefficients of a linear connection; con- nection matrices, coefficients' matrices	38
$T^i_{\ jk}$	(local) components of the torsion in a (local) frame $\{E_i\}$	26
$R^i_{\ jkl}$	(local) components of the curvature in a (local) frame $\{E_i\}$	26
$oldsymbol{R}_{kl}$	matrices of the curvature components, curvature matrices; $\mathbf{R}_{kl} := [R^{i}_{\ ikl}]_{i \ i=1}^{\dim M}$	105
${i \choose jk}$	Christoffel symbols (of a metric)	36

continued from the previous page

#### **Derivations on manifolds**

D	derivation or derivation along vector fields	143
$D_X$	derivation along (tangent) vector field $X$	144
$\Gamma_X{}^i{}_j$	components of derivation along vector field X in a (lo- col) frame $\{E_i\}$	147
$\mathbf{\Gamma}_X = \left[ \Gamma_X {}^i_{\ j} \right]_{i,j=1}^{\dim M}$	the matrix of the components of a derivation along a vector field $X$	147
R	curvature, curvature operator	147
Т	torsion, torsion operator	147
$R^D$	curvature of a derivation $D$ along vector fields	147
$T^D$	torsion of a derivation $D$ along vector fields	147
$\Gamma^{i}_{\ ik}$	coefficients of a linear derivation along vector fields	146
$\Gamma_k = \left[\Gamma^i_{\ jk}\right]_{i,j=1}^{\dim M}$	matrices of the coefficients of a linear derivation along vector fields; coefficients' matrices	145
D	(section-, vector-, tensor-)derivation along paths	191
$D^\gamma$	derivation along path $\gamma$	191

Symbol	Explanation	Page
$\Gamma_{\gamma i}^{i}$	components of derivation along $\gamma \colon J \to M$	192
$\Gamma^{i}_{j}(s;\gamma)$	the value of $\Gamma_{\gamma_{i}}^{i}$ at $s \in J$	192
$\Gamma_{\gamma} := [\Gamma_{\gamma}{}^{i}{}_{i}]$	the matrix of derivation along $\gamma$	192
$\Gamma(s;\gamma)$	the value of $\Gamma_{\gamma}$ at $s$	190
Fibre bundles		
$(E,\pi,B)$	(fibre) bundle with (total) bundle space $E$ , projection $\pi \colon E \to B$ , and base (space) $B$	217
$\gamma\colon J\to B$	path in $B$	218
$\operatorname{Sec}(\xi)$	the set of sections of bundle $\xi$	217
$\operatorname{Sec}^k(\xi)$	the set of $C^k$ sections of bundle $\xi,k\in\mathbb{N}\cup\{0,\infty,\omega\}$	219
$\mathrm{P}(A)$	the set of paths in a set $A$	218
$\operatorname{PLift}(\xi)$	the set of liftings of paths in a bundle $\xi$	218
$\operatorname{PLift}^k(\xi)$	the set of $C^k$ liftings of paths in a bundle $\xi$	219
$\operatorname{PF}(B)$	the set of functions along paths in bundle $(E, \pi, B)$	218
$\mathrm{PF}^k(B)$	the set of $C^k$ functions along paths in bundle $(E, \pi, B)$	219
$\operatorname{Lift}_g(E,\pi,B)$	the set of liftings of a mapping $g\colon X\to B$ to $E$ in bundle $(E,\pi,M)$	218
$\sigma$	section of a bundle	217
$\hat{\sigma}$	lifting of paths assigned to a section $\sigma$ of a bundle, $\hat{\sigma}: \gamma \mapsto \hat{\sigma}_{\tau}:= \sigma \circ \gamma$	221
D	derivation along paths in a bundle	219
$D^{\gamma}$	derivation along a path $\gamma: J \to B$ in bundle $(E, \pi, B)$	219
$\frac{-}{D}$	derivation along paths of sections in a bundle	221
D	section-derivation along paths in a bundle	221
$i, j, k, \ldots$	indices running from 1 to the (fibre) dimension of a vector bundle	218
$\alpha, \beta, \ldots, \mu, \nu, \ldots$	indices running from 1 to the dimension of the base of a bundle with a manifold as a base space	247
$I, J, K, \ldots$	indices running from 1 to the dimension $\dim E$ of the bundle space of a bundle	340
<i>a</i> , <i>b</i> , <i>c</i> ,	indices running from $1 + \dim M$ , $\dim M$ being the dimension of the base space, to the dimension $\dim E$ of the bundle space of a bundle	339

continued from the previous page

continued from the previous page

Symbol	Explanation	Page
$\{e_i\}$	frame in the bundle space of a vector bundle	218
$\{\hat{e}_i\}$	bases for liftings of paths	218
$\Gamma^{i}{}_{j}$	components (2-index coefficients) of derivation	220
$\Gamma := [\Gamma^i_{\ j}]$	matrix (of the components) of derivation	220
$\Gamma^{i}_{j\mu}$	(3-index) coefficients of derivation	248
$\Gamma_{\mu}^{i} := [\Gamma^{i}_{\ i\mu}]$	coefficients' matrices of derivation	247
$(T(M), \pi, M)$	tangent bundle over manifold $M$	222
$(T^*(M),\pi^*,M)$	cotangent bundle over manifold $M$	222
$(T^r_s(M), \pi^r_s, M)$	tensor bundle of type $(r, s)$ over manifold $M$	223
$(\mathbf{T}(M), \boldsymbol{\pi}, M)$	algebraic tensor bundle over manifold $M$	223
$\Delta^h$	connection on differentiable bundle; horizontal distribution on that bundle	342
$\Delta^v$	vertical distribution on differentiable bundle	342
Linear transports a	nd derivations along paths	
L	linear transport along paths in a vector bundle $(E, \pi, B)$	226
$L^{\gamma}$	linear transport along path $\gamma: J \to B$	226
$L_{s \rightarrow t}^{\gamma}$	linear transport along $\gamma$ from s to t	226
D	derivation along paths generated by linear transport along paths	230
$D^{\gamma}$	derivation along paths generated by linear transport along a path $\gamma: J \to B$	231
$D_s^\gamma$	derivation along paths generated by linear transport along a path $\gamma: J \to B$ at $s, s \in J$	231
$\mathcal{D}$	derivation along tangent vector fields	257
$\mathcal{D}_X$	derivation along a tangent vector field $X$	257
$\Gamma^{j}_{\ i}$	(2-index) coefficients of transport/derivation along paths	231
$\Gamma^{j}_{\ i\mu}$	(3-index) coefficients of transport/derivation along paths	248
$\Gamma := [\Gamma^j_{\ i}]$	matrix of the (2-index) coefficients of transport or derivation along paths	232

continued from the previous page

Symbol	Explanation	Page
$\Gamma_{\mu} := [\Gamma^{j}{}_{i\mu}]_{i,j=1}^{n}$	matrices of the 3-index coefficients of transport or	247
1 ,5	derivation along paths	
R	curvature (operator) of linear transport or derivations	265
	along paths	
$R^{\eta}(s,t)$	curvature along $\eta \colon J \times J' \to B$ at $(s,t) \in J \times J'$	262
$\left(R^{\eta}(s,t)\right)_{j}^{i}$	(2-index) components of $R^{\eta}(s,t)$ in a frame $\{e_i\}$	263
$R^i_{j\alpha\beta}$	4-index components of $R^{\eta}(s,t)$ in a pair of frames $\{e_i\}$	265
5	in the bundle space $E$ and $\{E_{\mu}\}$ in the bundle space	
	T(M) of the tangent bundle	
$oldsymbol{R}^{\eta}(s,t)$	matrix of $R^{\eta}(s,t)$ in a frame $\{e_i\}, \mathbf{R}^{\eta}(s,t)$ := $[(R^{\eta}(s,t))_i^i]$	264
$\boldsymbol{R}_{\alpha\beta} := \begin{bmatrix} R_{i\alpha\beta}^i \end{bmatrix}$	matrices of $R^{\eta}(s,t)$ in a pair of frames $\{e_i\}$ and $\{E_{\mu}\}$	266
T	torsion (vector, operator) of a linear transport or	272
	derivation along paths in the tangent bundle over a	
	manifold	
$T^{\eta}(s,t)$	torsion along $\eta \colon J \times J' \to M$ at $(s,t) \in J \times J'$ (in the	272
	tangent bundle over a manifold)	
$(T^{\eta}(s,t))^i$	components of $T^{\eta}(s,t)$ in a frame $\{E_i\}$	272
$T^i_{\ jk}$	components of the torsion tensor in $\{E_i\}$	272
Р	parallel transport in $(T(M), \pi, M)$ considered as a lin-	281
	ear transport along paths	
Parallel transports,	connections, covariant derivatives	
$(E,\pi,M)$	vector bundle, $E$ and $M$ being $C^\infty$ manifolds	301
N	$C^{\infty}$ manifold	301
$\mathbb{P}$	parallelism structure	302
Р	parallel transport	311
$P^{eta}$	parallel transport along a path $\beta \colon [a, b] \to M$	302
Р	parallel transport along paths	315
$P^{\beta}_{s \to t}$	parallel transport along $\beta \colon [a,b] \to M$ from s to t, $s,t \in [a,b]$	315
$T^h(E)$	connection on a vector bundle $(E, \pi, M)$ , horizontal subspace of $T(E)$	303

Symbol	Explanation	Page
$T^h_u(E)$	horizontal subspace of $T_u(E), u \in E$	303
$T^v(E)$	the vertical subspace of $T(E)$	303
$T^v_u(E)$	the vertical subspace of $T_u(E), u \in E$	303
$ abla^g$	covariant derivative in $(E,\pi,M)$ along a $C^\infty$ mapping $g\colon N\to M$	308
$\nabla$	the same as $\nabla^g$ if g is insignificant or a covariant derivative in $(E, \pi, M)$ ; $\nabla = \nabla^{id_M}$	224
$i, j, \ldots$	indices running from 1 to the (fibre) dimension of $(E,\pi,M)$	308
$\mu,  u, \ldots$	indices running from 1 to the dimension of $N$ (or $M$ if $N=M)$	308
$\Gamma^{i}_{\ j\mu}(\cdot;g)$	coefficients of $\nabla^g$	308
$\Gamma^{i}_{j\mu}$	coefficients of $\nabla^{id_M}$ ; $\Gamma^i_{j\mu} = \Gamma^i_{j\mu}(\cdot; id_M)$	309
$\Gamma_{\mu}(\cdot;g)$	matrices of the coefficients of $\nabla^g$ , i.e., $\left[\Gamma^i_{\ j\mu}(\cdot;g)\right]$	310
$\Delta^h$	connection (horizontal distribution) on a differentiable manifold $(E,\pi,M)$	342
$\Delta^v$	vertical distribution on a differentiable manifold $(E, \pi, M)$	342
$I, J, K, \ldots$	indices running from 1 to the dimension $\dim E$ of the bundle space of a bundle	340
$a, b, c, \ldots$	indices running from $1 + \dim M$ , $\dim M$ being the dimension of the base space, to the dimension $\dim E$ of the bundle space of a bundle	339
$\{u^I\}$	bundle coordinates	340
$\{u^{\mu}\}$	basic coordinates	340
$\{u^a\}$	fibre coordinates	340
$\{e_I\}$	frame in $T(E)$	341
$\{e^I\}$	coframe dual to $\{e_I\}$	341
$^{\circ}\Gamma^{a}_{b\mu}$	fibre coefficients of connection	345
$R^a_{\mu u}$	fibre components of curvature	345
$\{X_I\}$	adapted frame	346
$\{\omega^I\}$	adapted coframe dual to $\{X_I\}$	346
$\Gamma^a_\mu$	(2-index) coefficients of connection	346
$\Gamma^a_{b\mu}$	(3-index) coefficients of linear connection	355

## **Author Index**

#### В

Birkhoff George D.
$(1884 - 1944) \dots 48, 84$
С
Cartan Élie J.
$(1869-1951) \dots 36$
Christoffel Elvin B.
$(1829-1900) \dots 36$
D
Dombrowski Peter
$\mathbf{E}$

#### $\mathbf{F}$

Fermi Enrico (1901–1954).....48, 84, 138

### $\mathbf{L}$

Levi-Civita Tullio (1871–1941).....26, 84, 138 Lumiste Ülo G. (b. 1919).....225

#### Р

### R

O'Raifeartaigh Lochlainn S. (1933–2000) ..... 49, 112, 138 Riemann Bernhard G.W. (1826–1866) ..... 47, 138

Rinow Willi									
(1907 - 1979)	 	•				• •		22	5

#### $\mathbf{S}$

Schouten Jan A.	
$(1883 - 1971) \dots \dots$	. 69, 111
Struik Dirk J.	
$(1894-2000)\dots\dots\dots\dots\dots$	111

#### Т

Teleman C			
(b. 1933)	)	 	

#### V

## **Notation Index**

The frequently used symbols with a fixed meaning are listen below. They are arranged in more or less alphabetical order. Since the sorting of mathematical symbols is not unique, they are sorted according to up to three criterions: the symbol's name or pronunciations (if any), the name of symbol's kernel (root) letter, and the meaning of the symbol as a whole. For some symbols such a classification is not unique, due to which they are listen more than ones; e.g.,  $\nabla$  can be found under the letters "C", standing for Connection or Covariant derivative (sorting by meaning), and "N", standing for Nabla (sorting by name). Besides, a multiple appearance of a symbol under different sorting letters may mean that it has different meanings in different but similar contexts. The number(s) standing to the right of the symbols denote the page(s) where the symbols first appear or are defined.

#### Symbols

*
$:= \dots \dots$
=4
=:
\
$\cap \dots \dots$
o
$\cup \dots \dots$
Ø5
$\iff \dots \dots 34$
$\in \dots \dots$
$\mapsto \dots \dots$
$\nabla$
∋7
1

$\oplus$	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	1	47
$\otimes$	• •	•	·	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	T	1
<u> </u>	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	ი	87
2	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	1	27	1 7
∠	•••	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	T	(	1
×	•••	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	4 5
	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	5
) 	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	8
	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	0
										A																							
A	(i)	;)																														2	4

$A_{(ij)}$	• •	24
$A_{[ij]}$		24
$f_{,i}$		26
$\{\omega^I\}$		.346
$\{X_I\}$	• • •	346
$\mathfrak{F}(U)$		8
$\mathfrak{F}^r(U)$	8	
---	-------	
T(U)	. 19	
$T_p(M)$	17	
$\boldsymbol{T}^{r}(U) \dots \dots \dots \dots$	. 19	
$(\mathbf{T}(M), \boldsymbol{\pi}, M)$	.223	
*	5, 15	

## В

В	
$\setminus \dots \dots 27$	
$\{\hat{e}_i\}$	
$\{u^{\mu}\}\dots\dots340$	
$\{\mathrm{d}x^i\ldots\ldots\ldots\underline{p}\}$ 15	
$\left\{\frac{\partial}{\partial x^i}\Big _p\right\}$	
$\{\tilde{E}^i _p\}$ 15	
$\{E_i _p\}$	
$\{e_i\}$	
$\in \dots \dots \dots 4$	
$(E, \pi, B)$ 217	
$\{u^I\}$	
$(E,\pi,M)\ldots\ldots301$	

## С

$\mathbf{C}$	
$C^i_{ik}$	26
$C^{k}$	5, 7
×	4
$(U, \varphi) \dots \dots \dots$	5
$\left\{ {i\atop ik} \right\} \dots \dots$	36
$[a, b] \dots \dots \dots$	27
$\Gamma^{i}_{j\mu}$	248
$\Gamma^{a'}_{b\mu}$	355
$\Gamma^{j}_{i}$	248
$^{\circ}\Gamma^{a}_{b\mu}$	345
$\Gamma^a_\mu \overset{\circ \mu}{\dots} \dots \dots$	346
$\Gamma^{i}_{\ i\mu}$	309
$\Gamma^{i}_{j\mu}(\cdot;g)$	308
$\Gamma^{i}{}^{j}{}^{r}{}^{i}{}_{ik}$	22
$\Gamma^{i}_{ik}$	146
$\Gamma_k^{j\kappa}$	145
$\Gamma^{j}_{i}$	231
$\Gamma_{\mu}$	247
$\{E_i^i\}$	16
$\{\omega^I\}$	346
[·, ·] <u>.</u>	13

#### Notation Index

$R^i_{\ ikl}\dots$		 		26
$\boldsymbol{R}_{kl}^{jm}$	•••	 		105
$R^\eta(s,t)$	• • •	 	2	262, 263
$R \dots$	•••	 		147

## D

d 49	
$D_X \dots \dots \dots 144$	
$\Delta^h \dots 342$	
$\delta_i^j$	
$\dot{\Delta^v}$	
<i>D</i> 143, 146	
$\mathcal{D}_X \dots \dots$	
$D^{\gamma}$	
$D^{\gamma}$	
$D_s^{\gamma}$	
<i>D</i> <sup>°</sup>	
D191	
<i>D</i> 230	
<i>D</i> 219	
$\overline{D}$	
$\mathcal{D}$	
det 36	
$\operatorname{diag}(a_1,\ldots,a_n)\ldots\ldots\ldots 36$	
$C^k$	
<i>M</i> 4	
$f_*$	
dim5	
$\dim_{\mathbb{C}} \dots $	
$\dim_{\mathbb{R}}\ldots$	
$\oplus \dots \dots \dots 4$	
$\Delta$	
$\Delta_p \dots \dots$	,
$\Delta^{\tilde{h}}$	
$\Delta^v \dots 342$	
*	

## Е

$\{\hat{e}_i\}$ 2	18
$\{e_i\}$	18
$\{E^i\}$	16
$\{E_i\}$	12
$\{e_i\}$	83
$\{E^i _p\}$	15

$\{E_i _p\}$
Ø5
$\mathbb{S}^n \dots \dots$
$:= \dots \dots$
=:
=4
$\mathbb{E}^n$
$\wedge \dots \dots \dots 49, 338$

## $\mathbf{F}$

$f_*$	13
$f_{.i}$	26
$^{\circ}\Gamma^{a}_{b\mu}$	
$R^{a}_{\mu\nu}$	345
$\{u^{\mu\nu}\}$	
Ř	4
$\mathbb{K}^n$	4
$R^i_{i\alpha\beta}$	
$\{X_I\}$	346
$dx^i$	16
${E^i}$	16
$\{e_i\}$	
$\{E_i\}$	
$\mathfrak{F}(U)$	8
PF(B)	
$\mathrm{PF}^{\dot{k}}(B)$	
$\mathfrak{F}^r(U)$	
U (- )	

## G

<i>g</i>
$\Gamma^{i}_{\ j\mu}$
$\left\{ \stackrel{i}{jk} \right\} \dots \dots 36$
$\Gamma^{i}_{\ j\mu}$
$\Gamma^{i}_{j\mu}(\cdot;g)$
$\Gamma^{i}_{\ ik}$
$\Gamma_k \dots \dots 145$
$\Gamma^{j}_{i}$
$\Gamma_{\mu}$
$\Gamma^{i}_{\ i}(s;\gamma)$
$\Gamma_X^{i}{}^i{}_j$
$\Gamma_{\gamma i}^{i}$
$\Gamma^{i}_{j}(s;\gamma)$
$\dot{\gamma}$

$\Gamma_{\mu}$
$\Gamma_{\mu}(\cdot;g)$
$\Gamma_{\gamma}$
$\Gamma(s;\gamma)$
Γ232
$\gamma \colon J \to M \dots \dots 9$
$\gamma \colon J \to B \dots \dots$
$\Gamma^{j}_{\ i\mu}$
$\dot{\gamma}(s)$
$\Gamma_X$
$\operatorname{GL}(n,\mathbb{K})$
$\lambda, \mu, \nu, \dots \dots$
$\mu, \nu, \ldots \ldots 308$
$\alpha, \beta, \ldots, \mu, \nu, \ldots \ldots \ldots 247$
····

## H

$\Delta^h$																														342	
------------	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	--	-----	--

## I

$id_A \ldots \ldots \ldots 5$
1
$\mathbb{1}_n \dots \dots 97$
$\iff$
€
$I, J, K, \dots, 340$
<i>a</i> , <i>b</i> , <i>c</i> ,
$\lambda, \mu, \nu, \dots, \dots, 247$
$i, j, \ldots$ 308
$\mu, \nu, \ldots \ldots 308$
$i, j, k, \ldots \ldots 218$
$i, j, k, \ldots \ldots$
$\alpha, \beta, \ldots, \mu, \nu, \ldots \ldots \ldots 247$
ℕ
∩
J
[a, b]
$\begin{bmatrix} a & b \end{bmatrix}$ 27
[a, b] 27
(a, b) 27
$f^{-1}$ 5
$\Lambda^{-1}$ 11
$\sim$ 55
=

# J

	0	
$J \dots$		
$J^n \dots$	•••••	

## K

$\mathbb K$ .																	. 4	Ł
$\mathbb{K}^n$																	.4	ŀ
$\delta_i^j$ .							•										15	j

## $\mathbf{L}$

$I, J, K, \ldots$	.340
$a, b, c, \ldots$	. 339
$i, j, \ldots$	. 308
$\mathcal{L}_X$	. 144
$\operatorname{Lift}_{g}(E,\pi,B)$	. 218
$\operatorname{PLift}(E, \pi, B)$	. 218
$\operatorname{PLift}^k(E,\pi,B)$	. 219
$\nabla$	21
$L^{\gamma}$	. 226
$L_{s \to t}^{\gamma} \dots \dots \dots \dots$	. 226
	. 226
$(U, \varphi) \dots \dots \dots$	5
$\left\{\varphi^{i}\right\}$	5

## $\mathbf{M}$

$\begin{array}{llllllllllllllllllllllllllllllllllll$	$f: x \mapsto y \dots \dots$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$f: A \to B \dots \dots 5$
$\begin{array}{c} \mapsto & \dots & & 6 \\ \Gamma_{\mu} \dots & & 247 \\ R_{\alpha\beta} \dots & & 266 \\ \Gamma_{\mu} \dots & & 247 \\ \Gamma_{\mu}(\cdot;g) \dots & & 310 \\ \Gamma_{k} \dots & & 38 \\ R^{\eta}(s,t) \dots & & 264 \\ \Gamma_{X} \dots & & 147 \\ \Gamma_{\gamma} \dots & & 192 \\ \Gamma(s;\gamma) \dots & & 190 \\ \Gamma(s;\gamma) \dots & & 100 \\ \Gamma(s;\gamma) \dots & & 110 \\ I_{i}^{j}]_{i,j=1}^{n} \dots & & 111 \\ I_{i}^{j}]_{i,j=1}^{n} \dots & & 34 \\ g \dots & & & & & 34 \\ \end{array}$	$\rightarrow \dots \dots$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\mapsto \ldots \ldots$
$egin{arrge} \dot{R}_{lphaeta} &$	$\Gamma_{\mu}$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$\vec{R}_{\alpha\beta}$
$     \begin{split}       \Gamma_{\mu}^{'}(\cdot;g) & \dots & 310 \\       \Gamma_{k} & \dots & .38 \\       R^{\eta}(s,t) & \dots & .264 \\       \Gamma_{X} & \dots & .147 \\       \Gamma_{\gamma} & \dots & .192 \\       \Gamma(s;\gamma) & \dots & .190 \\       \Gamma & \dots & .220 \\       \Gamma & \dots & .220 \\       \Gamma & \dots & .232 \\       I_{i}^{j}] & \dots & .11 \\       I_{i}^{j}]_{i,j=1}^{n} & \dots & .11 \\       g & \dots & .34 \\       g & \dots & .34 \\       g^{\tau;k}(II) & 19 \end{split} $	$\Gamma_{\mu}$
$ \begin{array}{ll} \Gamma_k & & & & & & & & & & & & \\ \Gamma_k & & & & & & & & & & \\ R^{\eta}(s,t) & & & & & & & & & & \\ \Gamma_X & & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & & & & & & \\ \Gamma_{\gamma} & & \\ \Gamma_{\gamma} & & & \\ \Gamma_{\gamma} & & \\ \Gamma_$	$\Gamma_{\mu}^{\prime}(\cdot;g)$
$     \begin{array}{l}         R^{\eta}(s,t) \dots 264 \\         \Gamma_X \dots 147 \\         \Gamma_{\gamma} \dots 192 \\         \Gamma(s;\gamma) \dots 190 \\         \Gamma \dots 220 \\         \Gamma \dots 220 \\         \Gamma \dots 232 \\         [A_i^j]_{i,j=1}^n \dots 11 \\         [A_i^j]_{i,j=1}^n \dots 34 \\         T^{r;k}(II) 19 \\         19 \\         19 \\         19 \\         19 \\         11 \\         11 \\         11 \\         $	$\Gamma_k$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\mathbf{R}^{\eta}(s,t)$ 264
$ \begin{array}{c} \Gamma_{\gamma} & & & 192 \\ \Gamma(s;\gamma) & & 190 \\ \Gamma & & 220 \\ \Gamma & & 232 \\ [A_i^j] & & 11 \\ [A_i^j]_{i,j=1}^n & & 11 \\ g & & & 34 \\ g & & & 34 \\ g & & & 34 \end{array} $	$\Gamma_X$
$ \begin{split} & \Gamma(s;\gamma) \dots \dots 190 \\ & \Gamma \dots \dots 220 \\ & \Gamma \dots \dots 232 \\ & [A_i^j] \dots \dots 11 \\ & [A_i^j]_{i,j=1}^n \dots \dots 11 \\ & g \dots \dots \dots 34 \\ & g \dots \dots \dots 34 \\ & g^{\tau;k}(I^{\gamma}) \end{pmatrix} $	$\Gamma_{\gamma}$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$\Gamma(s;\gamma)$
$     \begin{array}{l}       \Gamma \dots & 232 \\       [A_i^j] \dots & 11 \\       [A_i^j]_{i,j=1}^n \dots & 11 \\       g \dots & 34 \\       \Upsilon^{r;k}(U) & 19 \\     \end{array} $	<b>Γ</b>
$ \begin{bmatrix} [A_i^j] & \dots & \dots \\ [A_i^j]_{i,j=1}^n & \dots & \dots \\ g & \dots & \dots & \dots \\ \mathfrak{T}^{r;k}(U) & \dots & 19 \end{bmatrix} $	Γ232
$\begin{bmatrix} A_i^j \end{bmatrix}_{i,j=1}^n \dots \dots 11$ $g \dots \dots 34$ $\mathfrak{T}^{r;k}(U) \qquad 19$	$[A_i^j]$ 11
$g \dots 34$ $\sigma^{r;k}(U)$	$[A_i^j]_{i \ i=1}^n \dots \dots$
$\overline{\mathfrak{T}}^{r;k}(U)$ 19	$q \dots 34$
$\boldsymbol{z}_{s}$ (0)	$\tilde{\mathfrak{T}}^{r;k}_{s}(U)$

#### Notation Index

$\mathfrak{X}^r(U)$																									1	.3
$\mathfrak{T}^r_s(U)$						•	•			•			•		•		•	•		•			•	•	1	8
$\mathfrak{X}(U)$ .	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	1	.1

## Ν

<i>N</i>	01
$ abla \dots \dots$	21
$\nabla$ 2	24
$\nabla^g$	08
$\nabla_X \dots \dots$	21
$\mathbb{N} \dots \dots$	. 4
$J^n \dots \dots$	61

## 0

$\{\omega^I\}$	3
$\mathbb{S}^1$	2
(a,b)2'	7
$\otimes \dots \dots \dots 1'$	7

## Р

## R

$R^i_{j\alpha\beta}$	265
$(R^{\eta}(s,t))^{i}_{j}$	263
$\hat{R}$	265
$R^D$	147
<i>R</i>	. 25

## 

### $\mathbf{S}$

D221
$\operatorname{Sec}(E,\pi,B)$
$\operatorname{Sec}^{k}(E,\pi,B)$
[a,b)
(a,b]
$\{x:P\}$
$\{x P\}$ 5
$\{a, b, c, \ldots\}$
$\hat{\sigma}$
$\sigma$
$C^k$
$\mathbb{S}^1$
$\mathbb{S}^2$
$\mathbb{S}^n \dots \dots$
$\mathbb{S}_q^n \dots \dots$
$\{r^i\}$
$C^i_{ik}$
<i>N</i>
$\subset$
<i>U</i>
$\subseteq$
\
$\dot{A}_{(ij)}$

### Т

$T^*(M)$	.70
$(\mathbf{T}(M), \boldsymbol{\pi}, M) \dots \dots$	223
$(T^{\eta}(s,t))^i$	272
$T^i_{ik}$	272
$(T^{\mathcal{H}}(M), \pi^*, M) \dots$	222

$T^h(E) \dots 303$
$T^{h}_{\cdot\cdot}(E)$
$T^u_{\pi}(M)$ 15
$T_{r}^{p}(M)$
T
$T^v(E)$
$T^v(E)$
T(M) 11
$\{E_i _{\pi}\} $ 11
$(T(M) \pi M)$ 222
$\frac{T(M)}{T(M)} = 11$
$\{E_i\}$ 12
T(M) 9
$\dot{p}(\mathbf{n})$
$\mathbf{T}(I)$ 10
$\mathbf{T}(0)$
$\mathbf{T}_{p}(W)$ 10
$\begin{array}{c} \mathbf{I}  (0)  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $
$T^{r}(M) \qquad \qquad$
$\mathcal{I}_{s}(\mathcal{W})$
$\mathcal{I}_s(U)$
$\mathcal{Z}_{s'}$ (0)
$\otimes$
$\otimes$
$ \begin{array}{c} I_{p_s}(M) \dots \dots$
$\Gamma_{j\mu}$
×
T $2/2$
$T^{\mathcal{D}}$
T25
$T^{i}_{jk}$ 26
$T^{\prime\prime}(s,t) \dots 272$
<i>T</i>
$\mathbb{T}^2 \dots \dots$

$A^{\top}$										 	•				47
$\Gamma^{j}_{i\mu}$	<i>ı</i>	••								 		• •	•	. 2	48
$(R^r)$	$^{\prime}(s,$	(t)	$\binom{i}{i}$		•	 	•		 	 •				. 2	63
$\mathbb{S}^2$ .										 					51
$\Gamma^{j}_{i}$		• •		• •	• •	• •		•	 •	 • •	•	• •	•	. 2	31

## U

	•	
$\{u^{\mu}\}\ldots\ldots$		
$\{u^I\}\ldots\ldots$		
$\{u^a\}\ldots\ldots$		
U		5
1		
$\mathbb{1}_n \dots$		

### v

$(E, \pi, M) \dots$	. 301
$\mathfrak{X}(U)$	11
$\mathfrak{X}^r(U)$	13
$\dot{\gamma}(s)$	9
$\Delta^v$	. 342
$T^{v}(E)$	. 303
$T_u^v(E)$	. 303

## $\mathbf{W}$

$\wedge$																4	49	9	,	3	33	8
$\ni$										•	•	•	•	•	•	•		•	•			7

## Х

	<b>21</b>	
$\{X_I\}$		
$\mathfrak{X}(U)$		11
$\mathfrak{X}^r(U)$		

# Subject Index

The **boldface** numbers refer to the page where a concept is defined. The letter 'n' after a page number means that the concept is mentioned in a footnote. The ligature 'ff' after a page number stands for "and following pages". Some important concepts are listen twice under different names, e.g., 'coefficients of linear connection' and 'linear connection coefficients of'.

#### Symbols

1-form ..... see covector field

#### A

adapted frame346–348
admissible changes
affine connection
2-index coefficients of $\dots 361-362$
criterion for
definition of
generalized on manifold 364
on manifold <b>364</b>
on vector bundle <b>362</b>
algebraic tensor bundle 223
atlas5
autoparallel path 30, 325–328
definition of <b>326</b>
autoparallels see autoparallel path
axiomatic definition of parallel
transport
in differentiable bundle <b>392</b> ff
in tangent bundle 283ff
in vector bundle
axiomatic description of parallel
transport

#### В

base of bundle 217	7
basic coordinates <b>340</b>	)
basis	
coordinate11	1
for distribution	9
of liftings of paths218	3
of vector space 68	3
bundle <b>21</b> 7	7
algebraic tensor	3
base (space) of 217	7
bundle coordinates of340	)
bundle property of 217	7
bundle space of	7
fibre of	7
fibre over a point	7
line <b>25</b> 4	1
local triviality of	7
projection of	7
restriction to set	7
section of $\dots \dots \dots$	7

tangent to manifold see tangent
bundle
tensorsee tensor bundle
topological $\dots 217$
vector
bundle coordinates $\dots 340$
bundle property 217
bundle Schrödinger equation 329
bundle space217
coframes on $\dots 339-342$
frames on

## С

$\mathrm{chart}\ldots\ldots\ldots.5$
normalsee normal chart
${\rm Christoffel\ symbols} \ldots \ldots .36$
${\rm circle} \ldots \ldots 52$
normal coordinate on $\dots 52$
cobasis for distribution
coefficients
2-index of affine
connection $\dots 361-362$
2-index of connection <b>346</b> , <b>368</b> ff
2-index of derivation
along paths <b>220</b> n
2-index of derivation along
tangent vector
2-index of linear transport along
paths <b>231</b> , 249, 251
3-index of derivation along
tangent vector fields 259
3-index of linear connection 355
3-index of linear transport along
paths
fibre of connection 345, 370
of linear connection 355
coefficients of
connection
covariant derivative in vector
bundle
transformation law 310
derivation along paths <b>220</b> n
derivation along tangent vector
fields 259

linear
connection $23-25, 354-356$
definition
transformation law23
linear derivation along vector
field $\dots \dots 146$
linear transport along
paths <b>231</b> ff, 249
coefficients' matrix of . $see \ {\rm matrix}$ of
coframe16
on bundle space $\dots 339-342$
specialized <b>343</b>
commutator of vector fields $\dots 13$
components fibre of curvature of
connection <b>345</b> , <b>370</b>
components of
anholonomicity object 69, 338
covariant derivative
covariant differential23
curvature of connection on vector
bundle <b>323</b> ff
curvature of covariant derivative
in vector bundle <b>323</b> ff
curvature of derivation along
paths264
curvature of linear transport
along paths266
2-index
4-index
derivation
along fixed vector field145
along paths 192, 220
along tangent
vector fields
along vector field
linear transport along paths228
tangent vector10
tangent vector field12
components' matrix of see matrix of
connection
(2-index) coemcients
01 <b>340</b> , <b>308</b> Π
assigned to transport
along paths <b>391</b> П

coefficients of	<b>346</b> , 3	<b>68</b> ff
differentiable	. 342,	347
fibre coefficients of	. <b>345</b> ,	370
fibre components of		
curvature of	. 345,	370
$\operatorname{flat}$		. 345
flat Riemannian		132
Levi-Civita	••••	35
linear $\dots$ see linear c	onnec	tion
links with transports alo	ong	
$\operatorname{paths}\ldots\ldots\ldots\ldots\ldots$	. 389-	-398
on differentiable bundle		<b>342</b>
on manifold		349
on vector bundle	350-	-364
Riemannian		. 35
Weyl		35
$connection\ mapping\ldots\ldots$		.306
connection on differentiable	Э	
bundle	342-	-350
connection on vector		
bundle <b>303</b>	, 350-	-364
determined by covariant		
derivative	••••	. 306
determined by parallelis:	m	
$\operatorname{structure} \ldots \ldots \ldots$		304
connection theory $\ldots \ldots$	. 342-	-350
$contraction operator(s) \dots$		19
$coordinate \ functions \ldots \ldots$		5
standard		5
$coordinate \ system \ldots \ldots \ldots$		5
$\operatorname{coordinates} \ldots \ldots \ldots \ldots$		5
basic		340
$bundle\ldots\ldots\ldots\ldots$		340
fibre		<b>340</b>
normalsee normal co	ordina	ates,
normal coordinates fo	r and	
also normal frame		
on light cone $\ldots \ldots$	66	6-68
$pseudospherical \dots$	55	5 - 56
Riemannian		$46 \mathrm{ff}$
Riemannian normal		<b>76</b> ff
spherical	52	2-53
vector bundle		350
vector fibre	••••	. 350

components of <b>323</b> ff
derivation along paths 263ff
components of
matrix elements of
matrix of
derivation along vector
field147–148
Euclidean linear transport along
paths
linear connection
linear transport along
paths
2-index components of $\dots$ 266
4-index components of $\dots$ 266
definition of $\dots \dots \dots 265$ ff
section-derivation
curvature operator
ofsee curvature of
curvature tensor
field
curve

## D

D
derivation
induced by tensor field of
type $(1, 1) \dots 144, 206$
along fixed vector
field 206–207
frames normal for $\dots 206-207$
of section along paths 221
of tensor algebra over
manifold 143–148
definition of <b>143</b>
derivation along fixed vector
field144
components of <b>145</b>
linear 145
matrix of145
derivation along
paths 189, 193, 219–222
components of <b>192</b> , <b>220</b>
curvature of
definition of <b>191</b> , <b>219</b>
Euclidean

flat <b>267</b>
generated by linear
connection190–191, 224, 234
generated by linear
$ ext{transport} \dots \dots \dots 230$
in tensor bundle $\dots 223-224$
normal frame for see normal
frame for
of section
torsion of <b>271</b> ff
derivation along paths in vector
bundle. see derivation along paths
derivation along tangent vector
fields <b>257</b> ff
2-index coefficients of $\dots 258$
3-index coefficients of
coefficients of259
components of <b>258</b>
Euclidean <b>258</b> ff
linear
matrix of <b>258</b>
normal frame for see normal
frame for
particular examples 297–298
derivation along vector field $\dots 146$
components of146
curvature of $\dots 147-148$
linear
coefficients of
coefficients' matrices of 146
torsion of
accurations and component designations
Formi and Formi derivative
Formi Wallson and Formi Wallson
dorivetive
Inumann ace Inumann derivative
Lio eee Lio derivative
Truesdoll eee Truesdoll derivative
do Sittor manifold 64
normal coordinates on 64
diffeomorphism
differentiable
342 347

g
6
3
n
9
9
9
2
8
2

## Е

Einstein manifold 63
normal coordinates on64
equation of
autoparallels 327
geodesics <i>see</i> geodesic equation
normal coordinates see normal
coordinates equation
normal frames <i>see</i> normal frame
equation
Euclidean
derivationsee derivation
linear transport see linear
transport
metric $34, 50$
of index $q \dots 54$
space
normal frames on $\dots \dots 50$
of index $q \dots 54$
${\rm exponential\ mapping}\ldots\ldots 32$
exterior derivative49
exterior product 49

## $\mathbf{F}$

Fermi coordinates	3
definition of	Ð
on submanifold $\dots \dots \dots 119$	9
Fermi derivative	7
along path $\dots 208$	8
Fermi-Walker derivative207, 297	7

fibre coefficients of
connection <b>345</b> , <b>370</b>
fibre components of curvature of
connection <b>345</b> , <b>370</b>
fibre coordinates 340
fibre of bundle 217
fibre over a point
flat
connection <b>345</b>
connection on vector bundle $324$
covariant derivative in vector
bundle
derivation along paths267
linear connection <b>26</b> , see also
linear connection
linear transport along paths $267$
parallel transport in vector
bundle324
frame
adapted $\dots 346-348, 370$ ff
adapted to coordinates 346
adapted to frame <b>367</b>
along paths $\dots \dots 220$
anholonomic12
coordinate12
geodesic $\dots 209-210$
holonomic
in physics
in vector bundle $\dots 218-219$
inertial $\dots 43$
noncoordinate
normal. see normal frame, normal
frame for
normal along great circle on
2-sphere 129
on bundle space 339–342
on differentiable bundle365–375
adapted to coordinates 346
adapted to frame 367
over manifold 12, 68–71
anholonomic
definition of
holonomic
locally holonomic $\dots 71$

reference	3
specialized <b>34</b>	3
function	8

G
generalized affine connection on
manifold <b>364</b>
geodesic equation 30, 31, 326,
349 - 350
geodesic frame 209–210
geodesic path <b>30</b> ff, 325ff, 349
as autoparallel path $\dots 326$
geodesicssee geodesic path

## н

holonomic normal frame. see normal
frame <i>holonomic</i>
holonomicity of frame over
manifold $\dots \dots \dots 69$ ff
horizontal
distribution <b>342</b>
lifting of mapping 303
lifting of paths <b>343</b> , 349
lifting of vector <b>342</b>
lifting of vector field 304, 342
path <b>343</b>
path in vector bundle 303
tangent space
tangent vector <b>303</b>
vector <b>342</b>
vector field $\dots 342$

## Ι

inertial frame
integral path13
inverse path 392

## J

Jaumann derivative ...... 207, 298

## $\mathbf{L}$

Levi-Civita connection.. **35**, see also Riemannian connection

Lie derivative $\dots 144, 205,$
207,  337 - 338
along fixed vector field 205
coordinates normal for $\dots 206$
frames normal for $\dots 205-206$
lifting of mapping218
horizontal <b>303</b>
parallel <b>310</b>
parallel along path <b>310</b>
lifting of path in vector bundle
parallel <b>303</b>
lifting of paths $\dots \dots 218-219$
generated by linear transport
along paths <b>231</b>
horizontal <b>343</b> , 349
tangent <b>32</b> 5
light cone65–68
coordinates on 66–68
definition of
induced metric on 67, 68
normal coordinates on $\dots 67, 68$
line bundle <b>25</b> 4
linear connection $\dots$ 19–33, 353–361
coefficients of $\dots 23-25$
definition
transformation law $\ldots 23, 356$
curvature of $\dots 25-26$
defined via parallel
$ ext{transport$
definition of <b>21</b> , <b>354</b>
derivation along paths generated
by see derivation along paths
flat <b>26</b>
in tangent bundle
as covariant derivative 309
motivation of $\dots \dots \dots 20-21$
of class $C^r \dots 24-25$
of mixed type see mixed linear
connection
on vector bundle <b>354</b>
symmetric24
torsion of
torsionless <b>26</b>

## Subject Index

linear transport along
$paths \dots 225-234$
assigned to covariant
derivative
coefficients $\dots \dots 231$ ff, 249
2-index <b>231</b> , 249, 251
3-index <b>249</b> , 251
${\rm components} \dots \dots 228$
curvature of
4-index components of $\dots 266$
definition of $\dots \dots 265$ ff
definition of <b>226</b>
derivation along paths generated
by <b>230</b> ff
equivalence with derivation along
paths $\dots 232-233$
Euclidean235ff
basic definitions <b>235</b> , <b>238</b>
curvature of $\dots 268-270$
generated by frame $\dots 243$
flat <b>267</b>
in line bundle $\dots 254-255$
lifting of paths generated by 231
matrix elements of <b>228</b>
matrix of <b>228</b> ff
path-independence $\dots 242, 270$
structure of
torsion of $\dots 271-276$
definition of <b>272</b> ff
torsionless $\dots 273$ ff
linear transport along paths in
tangent bundle 292–301
assigned to derivation along fixed
vector field
assigned to derivation along paths
293-294
definition of <b>293</b>
assigned to derivation along
vector fields 295
normal frame for $\dots 298-301$
local
coordinate system. see coordinate
system
coordinatessee coordinates

local triviality	217
locally injective	
mapping	168
$\operatorname{path}$	. 99

## $\mathbf{M}$

manifold $\dots 4-7$
1-dimensional 50
1-dimensional real132
normal frames/coordinates
on132–134
chart of (for) $\dots 5$
$\operatorname{complex} \ldots \ldots 4$
complex as real7
dimension of $\dots \dots \dots 7$
de Sitter $\dots 64$
differentiable $\dots 6$
dimension of5
Einstein
Einstein-de Sitter 136
Minkowski see Minkowski
spacetime
$\operatorname{real} \ldots \ldots 4$
Riemanniansee Riemannian
Schwarzschild 61–62
${\rm topological} \dots \dots 4$
with boundary 6
without boundary6
mapping
differentiable
differentiable of class $C^k \dots 7$
exponentialsee exponential
mapping
lifting of 218
locally injective 168
regular14
matrices of
coefficients of covariant derivative
in vector bundle
coefficients of linear connection 38
matrix of
curvature of derivation along
paths264

derivation along fixed vector
field
derivation along tangent vector
fields <b>258</b>
linear transport along
paths <b>228</b> ff
Minkowski spacetime 54, 63,
65, 131, 132
mixed linear connection 199–200
coefficients' matrices 199
contravariant coefficients 199
covariant coefficients 199
definition of <b>199</b>

## Ν

normal chartsee also normal
coordinates
for derivation along vector
$\mathrm{fields}\ldots\ldots151\mathrm{ff}$
for linear connection $\dots 40$
normal coordinates . see $also$ normal
coordinates for, normal frame,
normal frame <i>holonomic</i> ,
along geodesic in
$1$ -manifold $\dots \dots 134$ -136
history of
111-112, 138
onsee normal
coordinates on
normal coordinates
equation
normal coordinates for connection
on bundle
along injective horizontal
mapping $\dots 384-388$
along injective horizontal
$\operatorname{path}\ldots\ldots378$
at a point
on open set 380
normal coordinates for derivation
along fixed vector field
definition of174
existence/uniqueness along
path183

existence/uniqueness at a
point
on manifold 186
normal coordinates for
derivation along vector fields
along injective mapping
existence/uniqueness167
along injective mapping167–168
along locally injective
mapping
along mapping between
manifolds
existence/uniqueness172
along path
at a point
existence/uniqueness154
definition of
in neighborhood158
on submanifold
existence/uniqueness160
normal coordinates for linear
$\operatorname{connection} \ldots \ldots 40 \operatorname{ff}$
along path 84–104, see also Fermi
coordinates
complete description $\dots$ 93–104
existence/uniqueness103
at a point $\dots 75-84$ , see also
Riemannian normal
coordinates
complete description79–82
existence76
existence/uniqueness81
in neighborhood
existence/uniqueness108
on submanifold $\dots \dots \dots 128$
existence $\dots \dots \dots$
existence/uniqueness124
normal coordinates for linear
transport along paths $\dots 276-280$
definition of
uniqueness
normal coordinates on
1-dimensional real
manifold $\dots 132-134$

1-sphere
2-sphere
2-sphere for Weyl
$connection \dots 131$
de Sitter manifold 64
Einstein manifold 64
Einstein-de Sitter manifold 137
Euclidean space
flat Riemannian manifold132
light cone
Minkowski spacetime for Weyl
connection
<i>n</i> -sphere53
pseudosphere
Riemannian manifold 44–49
1-dimensional50
Schwarzschild manifold62–63
surface of revolution61
torus
normal frame. see also normal frame
for
along geodesic in
1-manifold 134–136
for Weyl connection 130–131
for Weyl connection on
2-sphere
for Weyl connection on
Minkowski spacetime 132
on 1-dimensional real
manifold $\dots 132-134$
on Euclidean space $\dots \dots 50$
on Riemannian manifold
1-dimensional50
strong see strong normal frame
terminology209–210
normal frame equation39, 149,
152, 174, 194, 240,
291,300,322,376
normal frame for connection on
bundle
adapted to holonomic
frame
along injective horizontal
path 378

at a point	378
ununiqueness	377
definition of	<b>375</b>
general case for	-377
normal frame for derivation alon	ng
fixed vector field	-189
along path $\dots 178$ -	-184
existence/uniqueness	.179
at a point 175-	-178
existence/uniqueness	175
definition of	172
general properties172-	-175
on manifold $\dots \dots 184$ -	-186
existence/uniqueness	185
on subset of manifold $\dots$ 186-	-189
normal frame for derivation alon	g
paths $\dots \dots 189-196$ ,	257
along path	193
existence/uniqueness	.193
on subset	196
existence	196
normal frame for derivation alon	g
tangent vector fields $\dots 2$	<b>58</b> ff
normal frame for derivation alon	g
vector fields	
along injective mapping162-	-168
existence/uniqueness165-	-166
along locally injective	
mapping 168-	-169
along mapping161-	-172
along mapping between	4 - 0
manifolds 169-	-172
existence/uniqueness170-	-171
along path154-	-157
existence/uniqueness	1 2 2
holonomicity	155
noiononnenty	$155 \\ 156$
at a point 153-	155 156 -154
at a point	.155 .156 .154 .153
at a point	155 156 154 153 <b>148</b>
at a point	.155 .156 .154 .153 <b>148</b> .153
at a point	.155 .156 -154 .153 <b>148</b> -153 151
at a point	.155 .156 .154 .153 <b>148</b> .153 .151 .151

necessary condition for
existence 149
on submanifold $\dots \dots 158-160$
existence/uniqueness158–159
uniqueness 150
normal frame for derivation in
vector bundle with manifold as
base
normal frame for linear connection
along path
complete description 93–104
existence/uniqueness 97–99
global
holonomicity
at a point
complete description
existence/uniqueness 81–82
definition of
fundamental ideas and
definitions 37–43
holonomicity 39
on neighborhood
existence
uniqueness 107
on submanifold
existence/uniqueness 120
holonomicity 127
review 202–204
uniqueness 30
normal frame for linear transport
along paths 235ff
basic definitions 235
existence theorems 239–246
general properties 234–239
holonomic 276–280
ovistonco 276
in tangent hundle assigned to
derivation along
water fields 208 201
when the bundle's base is
manifold 247 254
normal frame for mixed linear
connection 900ff
ovistonco/uniquonoss 2001
existence/uniqueness 201

#### 0

one-form ..... see covector field

Р
parallel lifting see lifting
parallel path $\dots \dots see$ path
parallel section see section
parallel tensor field see tensor field
parallel transport
as linear transport along
paths $\dots 233-234$
assigned to connection 343
assigned to linear
connection $\dots \dots 26-30$
definition of <b>27</b>
$path-independence\ldots 42$
assigned to parallel transport
along paths <b>394</b> ff
axiomatically defined <b>392</b> ff
history of 399
in tangent bundle280–289
along paths see parallel
transport along paths

pseudosphere
normal coordinates on 56–59
of index <i>q</i> <b>54</b>
$pseudospherical\ coordinates \dots 55{-}56$

## R

reference frame
restriction to set of bundle $\dots 217$
Riemannian
connection <b>35</b>
connection flat $\dots \dots 132$
coordinates normal for $\dots 132$
coordinates 46ff
manifold 34ff
metric
normal coordinates <b>76</b> ff, 84

## $\mathbf{S}$

scalar product	34
Schrödinger bundle equation 3	29
Schwarzschild manifold 61-	62
normal coordinates on $\dots 62-$	63
section	
along paths $\dots 2$	21
derivation of $\dots \dots 2$	<b>21</b>
derivation along paths of $\ldots 2$	21
of bundle $217, 2$	18
of tangent bundle 2	22
of tensor bundle2	23
of vector bundle	
parallel <b>3</b>	03
vertical lift of 3	<b>51</b>
section-curvature operator ofs	see
curvature of	
section-derivation along paths $2$	21
in tensor bundle $\dots 223-2$	24
section-derivation curvature of .26	<b>2</b> ff
specialized coframe 3	43
specialized frame3	43
sphere	
1-dimensional	52
normal coordinate on	52
2-dimensional	51

coordinates normal along
great circle $\dots 129$
coordinates normal
at point $\dots \dots \dots 128$
normal coordinates on $51-52$
even-dimensional 337, 367
n-dimensional
normal coordinates on $\dots 53$
spherical coordinates 52–53
standard fibresee fibre of bundle
strong normal frame
equation 291
for linear transport along paths
in bundle with manifold as
base <b>290</b> ff
base <b>290</b> ff for parallel transport in tangent
base <b>290</b> ff for parallel transport in tangent bundle <b>289</b>
base <b>290</b> ff for parallel transport in tangent bundle <b>289</b> in vector bundle for
base290ff for parallel transport in tangent bundle
base290ff for parallel transport in tangent bundle
base290ff for parallel transport in tangent bundle
base

## $\mathbf{T}$

tangent	
bundle	222
section of	222
field see tangent vect	or field
lifting of paths	325
mappingsee different	ntial of
mapping	
space	9
horizontal	303
vertical	303
vector	9 - 15
components of	10
components of definition of	10 <b>9</b>
components of definition of horizontal	10 <b>9</b> <b>303</b>

to path <b>9</b>
vertical <b>303</b>
vector field $\dots 11-15$
components of $\dots \dots 12$
definition of $\dots \dots 11$
to injective path 11
tensor 17–19
algebra18
bases 17–18
bundle222-224
derivation along
paths in $\dots 223-224$
of type $(r, s) \dots 223$
section of $\dots \dots 223$
section-derivation along paths
in 223–224
field 18–19
components of 18
definition of
parallel 29
frame
space17
tensor on manifold
components of
definition of17
torsion freesee torsionless
torsion of
derivation along paths 271ff
derivation along vector
field147–148
linear connection $\dots 25-26$
linear transport along
paths 271–276
definition of $\dots \dots 272$ ff
torsion vectorsee torsion
torsionless
linear connection 26, see also
linear connection
linear transport along
paths <b>273</b> ff
torus
<i>n</i> -dimensional 60
normal coordinates on60
total bundle space. see bundle space

#### Subject Index

transport
along mappings $\dots \dots 227n$
along paths $\dots 226$
assigned to parallel
$\mathrm{transport} \dots 394 \mathrm{ff}$
definition of <b>389</b>
linear see linear transport along
paths
links with
$connections \dots 389-398$
parallelsee parallel transport
typical fibresee fibre of bundle
typical fibresee fibre of bundle

## U

Universe model	
de Sitter	. 64
Einstein	63
Einstein-de Sitter	136
non-static spacially	
homogeneous	. 65
Schwarzschild	. 61

## $\mathbf{V}$

vectorsee also tangent vector
$horizontal \dots 342$
vertical <b>342</b>
vector bundle
connection on see connection
derivation along paths insee
derivation along paths

vector bundle coordinates 350
vector fibre coordinates
vector field see also tangent vector
field
horizontal <b>342</b>
horizontal lifting of <b>30</b> 4
vertical <b>342</b>
vector-derivation along paths 294
vertical
distribution
lift of section <b>351</b>
lifts350–353
path <b>34</b> 3
tangent space <b>303</b>
tangent vector <b>303</b>
vector
vector field

### W

wedge product 49
Weyl connection $\dots 35n$ , 129–130
normal frame for $\dots 130-131$
on 2-sphere 131
normal frame/coordinates
for 131
on Minkowski
spacetime $\dots \dots 131-132$
normal frame/coordinates
for $\dots \dots \dots 132$